DISS. ETH NO. 19177

Modeling and Optimization of Bidirectional Dual Active Bridge DC–DC Converter Topologies

A dissertation submitted to ETH ZURICH

> for the degree of Doctor of Sciences

> > presented by

FLORIAN KRISMER

Dipl. Ing., Technische Universität Wien born 11. April 1978 citizen of Austria

accepted on the recommendation of

Prof. Dr. J. W. Kolar, examiner Prof. Dr. P. Tenti, co-examiner

Acknowledgments

First of all I want to thank Prof. Dr. Johann W. Kolar for giving me the opportunity to do my Ph.D. thesis at the Power Electronic Systems Laboratory at ETH Zürich and for the numerous valuable ideas and advices he has provided to me. Under his guidance I have learned a lot in the field of power electronics and I have learned how to conduct high quality research and how to present research results in a comprehensible way.

Furthermore, I want to express my gratitude to Prof. Dr. Paolo Tenti for agreeing to be a co-examiner for my Ph.D. exam and for his effort to evaluate my thesis.

Many thanks go to all my colleagues at ETH, for the support and for creating such a pleasant working environment. I want to thank Dr. Gerold Laimer for all his support during the beginning of my Ph.D. and Dr. Simon Round for his numerous valuable advices. Besides, I want to thank Stefan Waffler, Dr. Dominik Bortis, Yanick Lobsiger, and Gabriel Ortiz for our talks and for our good time. I also want to thank Benjamin Wrzecionko for his ambitious work on the bidirectional DC–DC converter.

I would like to acknowledge the work of the PES Laboratory staff, Dr. Beat Seiler, Roswitha Coccia-Kunz, Monica Kohn-Müller, and Prisca Maurantonio, and the technicians Hansueli Altorfer, Peter Seitz, and Peter Albrecht.

I wish to express my sincere gratitude to my wife Olivia, my parents Margit and Eckhart, and my brother Ulrich. I want to thank Olivia for her love and understanding; this thesis would not have been possible without her support. I also want to thank my parents and my brother for their support and for encouraging me during hard times.

Contents

Kurzfassung

\mathbf{A}	bstra	act 3	ciii
1	Intr	roduction	1
	1.1	Applications of Bidirectional DC–DC Converters	1
		1.1.1 Automotive Applications	1
		1.1.2 Regenerative Power Systems	10
		1.1.3 Uninterruptible Power Supplies (UPS)	11
	1.2	Specifications of the Investigated DC–DC Converter	13
	1.3	Objectives and New Contributions of the Work	15
	1.4	Chapter Overview	17
2	Bid	irectional and Isolated DC–DC Converter Topologies	19
	2.1	Derivation of Bidirectional DC–DC Converters	19
	2.2	Topological Overview	24
		2.2.1 Single-Stage Topologies	25
		2.2.2 Two-Stage Topologies	48
	2.3	Conclusion	50
3	Stea	ady-State Operation of the Dual Active Bridge (DAB)	55
	3.1	Lossless DAB Model	55
		3.1.1 DAB Inductor Current and Power Transfer	56
		3.1.2 Phase Shift Modulation	58
		3.1.3 Alternative Modulation Methods	61
	3.2	DAB Model Regarding Conduction Losses	88
		3.2.1 Inductor Current and Power Transfer	91
		3.2.2 Implications on the Modulation Schemes	93
	3.3	Model Including the Magnetizing Inductance	101
		3.3.1 Inductor Current and Power Transfer	103
		3.3.2 Implications on the Modulation Schemes	106
	3.4	Conclusion	108

 \mathbf{xi}

4	Acc	urate DAB Loss Model	111	
	4.1	Losses Calculated with a Simple Loss Model	112	
		4.1.1 Power Switches	113	
		4.1.2 Transformer, Inductor	115	
		$4.1.3 {\rm Total \ Losses - Predicted \ Efficiency} \ .$	116	
	4.2	Accurate Conduction Losses, High Frequency	Effects 118	
	4.3	Accurate Switching Losses		
		4.3.1 Measurement Setups	120	
		4.3.2 HV Side Switching: Results		
		4.3.3 LV Side Switching: Results	136	
		4.3.4 Gate Driver Losses	146	
	4.4	Core Losses		
	4.5	Accurate Input and Output Power Calculatio	n 147	
	4.6	Calculated and Measured Efficiencies, Phase	Shift Modulation . 148	
		4.6.1 Numerical Efficiency Calculation		
		4.6.2 Comparison to Measured Efficiency Va	alues $\ldots \ldots \ldots 148$	
	4.7	Conclusion	154	
5	Adv	anced DAB Modulation Schemes	157	
	5.1	Extended Tria. and Trap. Current Mode Mod	lulation 157	
		5.1.1 Working Principle	159	
		5.1.2 Application of the DAB Loss Model	172	
		5.1.3 Measurement Results	175	
	5.2	Minimum Loss Modulation	175	
		5.2.1 Numerical Efficiency Optimization .	178	
		5.2.2 Suboptimal Modulation Schemes		
		5.2.3 Results	191	
		5.2.4 Conclusion	195	
6	Dyr	amic DAB Model and Control	207	
	6.1	Small-Signal Model		
		6.1.1 DAB Power Stage, Phase Shift Modul	ation $\ldots \ldots \ldots \ldots 210$	
		6.1.2 DAB Power Stage, Alternative Modul	ation Schemes 215	
		$6.1.3 Modulator \dots \dots$		
	6.2	DAB Digital Control Loop		
		6.2.1 System Sampling Rate		
		6.2.2 Time Delays		
		6.2.3 Moving Average Filters		
		6.2.4 DAB Control Plant G_{DAB}		
		6.2.5 Experimental Verification		
		6.2.6 Derivation of a Simplified G_{DAB}		
	6.3	Current and Voltage Controllers		
	6.4	Conclusion		

7	Two-Stage Converter Realizations 2		
	7.1	Motivation, Investigated Converter Topologies	235
	7.2	Buck-or-Boost Converter Loss Model	236
		7.2.1 Buck Operation	238
		7.2.2 Boost Operation	239
		7.2.3 Conduction Losses, Copper Losses, Dielectric Losses	240
		7.2.4 Switching Losses	242
		7.2.5 Core Losses	248
		7.2.6 Gate Driver Losses	249
		7.2.7 Total Buck-or-Boost Converter Losses	251
		7.2.8 Results	251
	7.3	Two-Stage Converters – Efficiencies	252
		7.3.1 Efficiencies for $V_i \leq 225 V$	253
		7.3.2 Efficiencies for $V_i \ge 325 V \dots \dots$	255
	7.4	Conclusion	256
8	Con	clusions and Outlook	259
	8.1	Bidirectional DC–DC Converter Topologies	261
	8.2	Unidirectional DC–DC Converter Topologies (HV \rightarrow LV) 2	263
	8.3	Unidirectional DC–DC Converter Topologies $(LV \rightarrow HV)$ 2	264
	8.4	Outlook	264
		8.4.1 DAB Converter Modelling and Optimization	264
		8.4.2 Possible Improvements of the DAB Converter Hardware	266
		8.4.3 Future Research on the Dynamic Behavior of the DAB . 2	266
A	open	dices	267
Α	Con	nverter Design	269
	A.1	Aim of the Converter Design	269
	A.2	Design of the DAB	270
		A.2.1 Calculation of n and L	271
		A.2.2 HV Side Full Bridge	274
		A.2.3 LV Side Full Bridge	276
		A.2.4 Transformer	290
		A.2.5 Inductor	304
		A.2.6 EMI Filters	308
	A.3	Three-Phase DAB Design	312
		A.3.1 Principle of Operation	312
		A.3.2 Lossless Electric Converter Model	313
		A.3.3 Model Parameterization	318
		A.3.4 Design Results	322
	A.4	LLC Converter Design	322
		A.4.1 Analytical Model of the LLC Converter	322
		A.4.2 HF Capacitor	329

		A.4.3	Design Results, Phase Shift Modulation	329
		A.4.4	Design Results, Optimal Modulation	330
	A.5	Full B	ridge Converter Design	335
		A.5.1	Mode of Operation	335
		A.5.2	Converter Design	339
	A.6	Currer	nt Doubler Converter Design	346
		A.6.1	Mode of Operation	346
		A.6.2	Converter Design	350
	A.7	Push-I	Pull Converter Design	356
		A.7.1	Mode of Operation	356
		A.7.2	Converter Design	359
	A.8	Design	of the Bidirectional Buck-or-Boost Converter	361
		A.8.1	Assumptions and Specifications	361
		A.8.2	Calculation of $L_{\rm DC}$	363
		A.8.3	Design Summary	364
	A.9	HF Inc	ductor Design	367
		A.9.1	Design Procedure	370
		A.9.2	Winding Configuration, Optimal Litz Wire Calculation	371
		A.9.3	Inductor Air Gap Length	373
		A.9.4	Inductor AC Resistance	376
в	Con	verter	Comparison: Calculation of the VA ratings	385
	B.1	Semico	onductor Switches	385
	B.2	Transf	ormer	385
	B.3	AC an	d DC Inductors	386
		B.3.1	Theoretical Relation between $W_{\text{mag,peak}}$ and Inductor	
			Volume V_L	387
	B.4	AC an	d DC Capacitors	389
		B.4.1	Theoretical Relation between $W_{\rm el,peak}$ and Capacitor	
			Volume V_C	390
a	T	. •		000
U	EXU.	racting	; the Stemmetz Coemcients	393
D	Eva	luation	of Power MOSFETs	399
	D.1	HV Sie	de	399
		D.1.1	Selection of the MOSFETs used for the DAB Converter	399
		D.1.2	Drain-Source Breakdown Voltages and On-State Resis-	
			tances	400
	D.2	LV Sid	le	404
		D.2.1	Drain-Source Breakdown Voltages and On-State Resis-	
			tances	404
\mathbf{E}	Pra	ctical l	Implementation	409
	E.1	Digital	l Control: Concept	409

	E.2 E.3	On-Line Calculation of the Modulation Parameters	$\begin{array}{c} 409\\ 411 \end{array}$
F	Sup F.1 F.2 F.3	plement to the Small-Signal Model of the DAB \sim Small-Signal Model \sim F.1.1Matrix ExpressionsF.1.2Small-Signal Transfer FunctionSampling Rate Reduction \sim Derivation of G_{DAB} \sim	417 417 417 418 419 419
Gl	ossai	ry	421
Bi	bliog	raphy	426
Cu	ırricı	ulum Vitae	443

Kurzfassung

Die vorliegende Arbeit beschreibt die Realisierung eines bidirektionalen Gleichspannungswandlers, welcher über einen Hochspannungsanschluss mit einer Anschlussspannung im Bereich zwischen 240 V und 450 V und über einen galvanisch getrennten Niederspannungsanschluss mit einer Anschlussspannung im Bereich zwischen 11 V und 16 V verfügt. Die Nennleistung des Konverters beträgt 2 kW, unabhängig von der vorliegenden Energieflussrichtung.

Für die in **Kapitel 2** dargestellte Topologieübersicht zeichnet sich unter Verwendung der gegebenen Konverterspezifikationen eine Einschränkung der geeigneten Schaltungskonzepte ab. Die resultierende Auswahl beinhaltet neben einstufigen Konvertertopologien auch zweistufige Realisierungsvarianten (d.h. die Serienschaltung eines galvanisch getrennten Gleichspannungswandlers und eines Gleichspannungswandlers ohne Potentialtrennung). Unter den einstufigen Topologien zeichnet sich der Dual Active Bridge (DAB) Konverter als die am besten geeignete Wahl hinsichtlich der gewünschten Effizienz und hinsichtlich der erzielbaren Leistungsdichte ab. Ausserdem lassen sich, aufgrund der weiten Ein- und Ausgangsspannungsbereiche, weitere Wirkungsgradverbesserungen für zweistufige Schaltungsvarianten erwarten.

Im Anschluss an Kapitel 2 erfolgt eine detaillierte Beschreibung der DAB.

- Kapitel 3 beschreibt die Funktionsweise der DAB; in diesem Zusammenhang werden drei verschiedene Konvertermodelle erläutert. Dies beinhaltet ein verlustfreies Konvertermodell, welches z.B. der vereinfachten Berechnung der Transformatorstromeffektivwerte dient; die resultierenden Zusammenhänge werden zur Synthese eines hinsichtlich Transformatorstromeffektivwerte optimalen Modulationsverfahrens verwendet. Die in diesem Kapitel ebenfalls erläuterten erweiterten Konvertermodelle berücksichtigen die Auswirkungen der Leitverluste und des Magnetisierungsstromes auf die primär- und sekundärseitigen Transformatorströme.
- In Kapitel 4 wird ein detailliertes Verlustmodell der DAB beschrieben,

welches der Berechnung der für eine zuverlässige Ermittlung des Wirkungsgrades relevanten Verlustanteile dient. Das Verlustmodell wird einerseits dazu verwendet, die in Kapitel 3 erstellten Konvertermodelle zu parametrisieren. Andererseits ermöglicht dieses Verlustmodell die in Kapitel 5 beschriebene Optimierung des Steuerverfahrens hinsichtlich maximalen Wirkungsgrades.

- Kapitel 5 erläutert eine systematische Untersuchung der in Kapitel 3 vorgestellten Modulationsverfahren unter Verwendung des in Kapitel 4 beschriebenen Verlustmodells. Die aus dieser Untersuchung resultierenden Ergebnisse dienen der Synthese eines neuartigen Steuerverfahrens, welches den Betrieb der DAB nahe des maximal möglichen Wirkungsgrades erzielt.
- Kapitel 6 beschreibt das transiente Verhalten des DAB Konverters. Dies beinhaltet neben einem arbeitspunktabhängigen Kleinsignalmodell der DAB auch die Übertragungsfunktionen des digitalen Steuerteils.
- In Kapitel 7 werden möglichen Wirkungsgradverbesserungen untersucht, die sich durch die Verwendung einer zweistufigen Konvertervariante erzielen lassen würden. Es stellt sich jedoch heraus, dass der für den zweistufigen Konverter resultierende mittlere Wirkungsgrad ($\overline{\eta} = 93.0\%$) kleiner ist, als der der einstufigen DAB ($\overline{\eta} = 93.5\%$; die Berechnung von $\overline{\eta}$ erfolgt gemäss Anhang A.1).
- Kapitel 8 schliesst die Arbeit mit einer Zusammenfassung der erarbeiteten Ergebnisse, abschliessenden Bemerkungen zu bidirektionalen und unidirektionalen DC–DC Wandlern und einem Ausblick bezüglich weiterer Forschungsmöglichkeiten auf dem Gebiet des DAB Konverters ab.

Messergebnisse bestätigen die in dieser Arbeit berechneten Ergebnisse.

Zur Ermittlung der experimentellen Ergebnisse wurde ein voll funktionsfähiger Prototyp des oben spezifizierten DAB Konverters entworfen (**Anhang A.2**) und realisiert. Der unter Verwendung des in Kapitel 5 erläuterten effizienzoptimierten Modulationsverfahrens gemessene Wirkungsgrad beträgt 94.5% (Eingangsspannung: 340 V, Ausgangsspannung: 12 V, Ausgangsleistung: 2 kW, Energietransfer in Richtung der Niederspannungsseite, Messung bei Raumtemperatur: 25°C).

Abstract

This thesis investigates different bidirectional and isolated DC–DC converters with a high voltage port (ranging from 240 V to 450 V), a low voltage port (11 V to 16 V), and a rated power of 2 kW.

The work starts with an overview on different bidirectional DC–DC converter topologies and investigates their applicability with respect to the given specifications (**Chapter 2**). Based on the findings of Chapter 2, the single-phase Dual Active Bridge (DAB) converter topology is considered most promising regarding the achievable converter efficiency and the achievable power density. Moreover, due to the wide input and output voltage ranges, two-stage DC–DC converters, i.e. the series connection of an isolated DC–DC converter and a DC–DC converter without galvanic isolation, are expected to facilitate an improved converter efficiency.

The subsequent Chapters discuss the single-phase DAB converter in detail.

- Chapter 3 explains the working principle of the DAB converter based on 3 different simplified converter models. This includes a lossless model of the DAB, which facilitates basic analytical investigations and allows for a simplified synthesis of optimal modulation schemes (calculated with respect to minimum transformer RMS current). Moreover, two different refinements of the electric DAB converter model are included in order to consider the impacts of the conduction losses and the magnetizing current on the transformer currents.
- **Chapter 4** discusses a detailed loss model of the DAB, needed in order to parameterize the models developed in Chapter 3 and to develop the efficiency optimized modulation presented in Chapter 5. This loss model is used to calculate the losses in the most relevant converter components.
- Chapter 5 presents a systematic investigation of efficiency improvements achievable for the given DAB converter, obtained with the use of optimized modulation schemes. Finally, with the suboptimal modulation schemes discussed in Section 5.2.2, converter operation close to its maximum possible efficiency is achieved (Figure 5.28).

- Chapter 6 details a dynamic model of the complete converter system (including the digital control platform).
- Chapter 7 investigates the most promising two-stage converter configurations in order to examine possible efficiency improvements. As a result, however, the average efficiency achieved with the two-stage converter ($\overline{\eta} = 93.0\%$) is less than the average efficiency achieved with the single-stage DAB ($\overline{\eta} = 93.5\%$; calculated according to Appendix A.1).
- Chapter 8 concludes this work and presents an outlook regarding future research in the field of the DAB converter.

The calculated results are verified using experimental data.

A hardware prototype of the above specified DAB converter is realized according to **Appendix A.2**. With this converter prototype and with the optimized modulation schemes detailed in Chapter 5, a converter efficiency of 94.5% is measured (input voltage: 340 V, output voltage: 12 V, output power: 2 kW, power being transferred to the low voltage port, room temperature: 25°C).

Chapter 1

Introduction

1.1 Applications of Bidirectional DC–DC Converters

1.1.1 Automotive Applications

Today, conventional cars use combustion engines which burn fossil fuels (e.g. gasoline, diesel) to generate the required propulsion power. The products of the combustion process – mainly the nontoxic compounds H_2O and CO_2 – are exhaled into the air. CO_2 , however, is known as a greenhouse gas and partly accounts for global warming.¹ Besides that, excessive burning of fossil fuels causes environmental pollution and the resources to exhaust [2].

Different initiatives have been taken in order to attain a reduction of CO_2 emissions; prominent examples are the Kyoto Protocol [3] and the California Zero Emission Vehicle (ZEV) program [4].

Regarding automobiles, the most promising vehicle technologies with respect to an effective reduction in fuel consumption are hybrid electric vehicles, electric vehicles, and fuel cell vehicles [2]. Hybrid electric vehicles utilize an internal combustion engine (ICE) in combination with one or more electric machines in order to improve the overall efficiency. With a hybrid power train and a further optimized ICE, the ambitious target of the European Commission of a CO₂ output value of 90 g CO₂/km (equivalent to a fuel consumption of 3.51/100 km or an efficiency of 67.2 MPG – US) after the year 2012 may become feasible [5]. The target of zero local CO₂ emissions can be achieved with battery powered electric vehicles and with fuel cell powered vehicles operated with hydrogen as fuel [6]; in that case, the fuel cell only emits water [7].

¹Water vapor acts as a greenhouse gas, too; however, the actual amount in the atmosphere is approximately 3%, much higher than that of CO₂ (\approx 385 ppm) [1].

Hybrid Electric Vehicles (HEV)

A hybrid electric vehicle utilizes two or more energy sources for propulsion to increase the overall system efficiency. The power train of a HEV typically consists of an ICE and one or more electrically powered machines [8]. Additionally, the hybrid electric drive train allows for improved vehicle acceleration, which is an important argument regarding customer satisfaction [9, 10, 11].

According to the degree of hybridization (i.e. the complexity of the system architecture and the provided amount of electric power), the different HEV architectures can be classified into 3 categories (Table 1.1, [12, 13]):

• Micro hybrid: belt-driven starter-generator system, recuperative braking can be implemented up to a certain degree. The electric system is designed to provide a comparably low amount of the maximum traction power, which is economically achievable with a relatively low system voltage. Still, a considerable effort is required to upgrade an existing

Category	Battery Power	Battery Voltage	Realization	Lit.
	(Nom. Capacity		(if available)	
	if available)			
Micro Hybrid	$4\mathrm{kW}\dots5\mathrm{kW}$	$30\mathrm{V}\ldots40\mathrm{V}$	_	[11]
	$10 \mathrm{kW}$	26 V	Saturn VUE	[14]
	10 K W	50 V	Hybrid	[14]
Mild Hybrid	$10kW\dots 20kW$	$144\mathrm{V}$	_	[11]
	up to $20\mathrm{kW}$	—	—	[12]
	_	$28V\ldots42V$	—	[15]
	$10\mathrm{kW}$	144 17	Honda Insight	[16]
	$(0.9\mathrm{kWh})$	144 V	$(year \ 2000)$	[10]
Full Hybrid	$30\mathrm{kW}\dots50\mathrm{kW}$	$288\mathrm{V}$	_	[11]
	$21\mathrm{kW}$	202 V	Toyota Prius	[19]
	$(6.5 \mathrm{Ah})$	202 V	$(year \ 2005)$	[13]
			Toyota	
	$45\mathrm{kW}$	$288\mathrm{V}$	Highlander	[13]
			Hybrid SUV	

Table 1.1: Electric power and typical voltages or voltage ranges of HV batteries employed for HEVs; if available, the nominal energy storage capacity is given in paranthesis (either in kWh or in Ah). conventional car to a micro hybrid vehicle, since a second battery and a second voltage bus are typically required (in addition to the existing 12 V battery and the 14 V bus) [11].

- Mild hybrid: starter-generator system, recuperative braking, and electric propulsion for a short time; the electric motor is directly flanged to the crankshaft [12]. The power provided from the electric system is typically between 10 kW and 20 kW. Again, a supply voltage higher than the installed 14 V is required in order to obtain technically reasonable battery and motor currents [17].
- Full hybrid or strong hybrid: starter-generator system, recuperative braking, and electric driving. There, the required electric power (up to 100 kW) demands for comparably high bus voltages of up to 800 V [15].

In another common approach, hybrid electric vehicles are classified into four different categories, with respect to the employed drive train [13,15]:

- Parallel HEV topology (Figure 1.1): both, the ICE and the electric machine are mechanically connected to the transmission. Thus, the system allows for the ICE and the electric motor to share the demanded torque while the revolution speed of the engine depends on the selected gear and the cruising speed what may implicate inefficient ICE operation [8, 18]. A clutch is installed between the ICE and the electric machine, which enables pure electric driving with the ICE being turned off [15].
- Series HEV topology: the ICE is connected to an electric generator which provides electric energy to a second electric machine for propulsion. Such a system allows for an optimized operation of the ICE at its best operating point, since the ICE revolution speed is independent of the cruising speed. However, compared to the parallel HEV drive topology, multiple stages of energy conversion are employed (mechanical → electrical, AC, with frequency f₁ → electrical, AC, with frequency f₂ → mechanical) what causes the system efficiency to decrease. Furthermore, a much higher rated power of the electric system (generator, motor, power electronics) is required [18].
- Series-parallel HEV topology: allows for the operation with both, the parallel hybrid drive train and the series hybrid drive train and thus combines the advantages of both systems. Disadvantageous is the high system complexity [8, 13].
- Complex HEV: subsumes all HEV drive train topologies, which cannot be classified into the 3 categories discussed above [13].



Figure 1.1: Example of a parallel HEV drive train architecture [8].

The electric power demand in a HEV is typically much higher than that of a conventional car (Table 1.1). Hence, a supply voltage considerably higher than the voltage provided from the existing 14 V bus (9 V...16 V at the battery terminals [6]) is required, whereas a high voltage (HV) DC bus² is typically employed in order to distribute the electric power (Figure 1.2). The HV DC bus (e.g. with a typical voltage between 200 V $\leq V_1 \leq 300$ V [10,13]³) primarily enables power transfer between the HV battery and the electric propulsion system. Moreover, high power ancillary loads (e.g. air conditioner, power assisted steering) are favorably supplied from the HV DC bus, since the ICE may be turned off during vehicle operation.

Even though, the proposed system architecture theoretically renders the

 $^{^2 \}rm Due$ to safety reasons, a voltage level of more than 60 V is often denoted a "high voltage" (HV) in automotive applications and for a voltage level below 60 V the term "low voltage" (LV) is used [9].

 $^{^3 \}rm Recently,$ many car manufacturers selected considerably higher voltages for the HV DC bus (e.g. 650 V for the Lexus GS 450h [13]) in order to reduce motor and inverter currents.



Figure 1.2: Typical HEV power system architecture [8].

conventional 14 V bus unnecessary, it is widely believed that the 14 V bus will coexist with the HV DC bus for a very long time, since all the conventional low power automotive loads (e.g. interior/exterior lighting, electric motor driven fans/pumps/compressors, instrumentation subsystems) are designed based on 14 V standards [6]. Thus, the 12 V battery will remain in order to provide a backup to the electric power supply of the 14 V bus.

Low Voltage DC–DC Converter. Instead of the conventional alternator – which cannot be used, since the ICE may be turned off during vehicle operation – a DC–DC converter is employed in order to provide electric power to the low power loads and the 12 V battery. For that application, a DC–DC converter designed for unidirectional power transfer would be sufficient. Still, practical reasons support the use of a bidirectional DC–DC converter [6,13]:

- Jump-starting the car: with a conventional battery-jumper-cable, the 12 V battery is connected to another car's 12 V battery in order to start the ICE [19]. For this mode of operation, a bidirectional DC–DC converter is required, since the starter is connected to the HV DC bus (e.g. the electric machine in Figure 1.2).
- Charging the HV battery: with a bidirectional DC–DC converter, the battery can be charged using an external power supply that is connected to the 12 V battery.
- Leaving a dangerous zone in case of a HV battery failure: the bidirectional DC-DC converter could be used to (slowly) drive the car out of a dangerous zone (e.g. road crossing) if the HV battery fails.

Current publications suggest a 14 V bus power demand of 1 kW...2.5 kW [6, 15, 20, 21], which defines the nominal output power $P_{\text{out,nom}} = P_{2,\text{nom}}$ [cf. (3.14)] of the DC–DC converter when operated in buck mode (i.e. power is transferred from the HV bus to the 14 V bus). However, the nominal output power in boost mode (i.e. power transfer from the 14 V bus to the HV bus) is not specified in literature; due to the aforementioned boost mode applications, an output power of $-P_{\text{out,nom}} = -P_{1,\text{nom}} \approx 1 \,\text{kW} \dots 1.6 \,\text{kW}$ seems reasonable e.g. with respect to the power requirements of conventional starters (up to 1.6 kW [12]).

Table 1.2 summarizes typical specifications of the bidirectional DC–DC converter employed to transfer electric power between the HV DC bus and the LV DC bus in a HEV.

Electric Vehicles (EV)

The drive train of an EV is exclusively powered from an electric machine and a HV battery with high energy density typically provides the required electric energy [26, 27]. Thus, EVs allow for zero (local) emissions what makes them most interesting for the use in urban areas where traffic emissions considerably pollute the local environment.

The basic architecture of an EV drive train is similar to the drive train of a series HEV without ICE and electric generator (cf. Figure 1.1 and Figure 1.2): besides the existing 14 V bus, a HV DC bus is required in order to provide the

HV port voltage:	$200 \mathrm{V} < V_1 < 250 \mathrm{V} \ [10]$	
	$200 \mathrm{V} < V_1 < 300 \mathrm{V}$ [13]	
	some publications mention even higher port voltages of up to 800 V [15]	
LV port voltage range:	$V_2 \approx 9 \mathrm{V} \dots 16 \mathrm{V}$	
Rated power, buck mode:	$P_{\rm out,nom} \approx 1 \rm kW \dots 2.5 \rm kW$	
Rated power, boost mode:	$-P_{\rm out,nom} \approx 1 \rm kW \dots 1.6 \rm kW$	
Galvanic isolation:	required due to safety reasons [15]	
Other requirements:	low cost, high efficiency, high power density,	
	low weight $[22, 23, 24, 25]$	

Table 1.2: Specification of a bidirectional DC–DC converter for HEVs used to transfer electric power between the LV DC bus and the HV DC bus; the output power is defined with (3.14) (using $P_{\text{out,nom}} = P_{\text{out}}$).

electric power for vehicle propulsion and all the high power ancillary loads. Similar to the HEV, the EV employs a 14 V bus to power the conventional low power automotive loads.

Low Voltage DC–DC Converter. Since the electric system architectures of EVs and HEVs are rather similar, the DC–DC converter which provides power to the conventional 14 V components during normal operation is specified according to Table 1.2. Bidirectional power transfer is needed in order to allow for jump-starting the car, leaving a dangerous zone using the starter, and charging the HV battery via the terminals of the 12 V battery.

Fuel Cell Vehicles (FCV)

In a FCV, a fuel cell provides the required traction power [in literature, Proton Exchange Membrane Fuel Cells (PEMFC) or Solid Oxide Fuel Cells (SOFC) are typically considered for automobiles [8]]. FCVs allow for zero (local) CO₂ emissions and achieve a cruising range comparable to cars with ICE.

Figure 1.3 depicts a simple power train architecture of a FCV. Depending on the load, the fuel cell generates a voltage which ranges from 230 V to 450 V during normal vehicle operation [20]. It provides power for vehicle propulsion, all the auxiliary fuel cell components (compressors, pumps, etc.) as well as for the conventional 14 V bus (using a DC–DC converter). In this simple design, the HV battery is avoided; during the starting process of the vehicle, the 12 V



Figure 1.3: Simple power system architecture for a FCV [20]; the HV battery is avoided and thus the HV bus voltage varies within a wide voltage range (e.g. between 230 V and 450 V).

battery provides the required electric power to the fuel cell system.

Most documented systems, however, include a HV battery in order to stabilize the HV DC bus voltage during transient load conditions (Figure 1.4, [8, 28]). The fuel cell, the HV battery, and the 14 V bus are all connected to the HV DC bus using separate DC–DC converters. In such a system, the HV battery provides power when the car is started and thus the DC–DC converter between the HV DC bus and the 14 V bus must not be bidirectional any more.



Figure 1.4: Extended power system architecture for a FCV [8,28], including a HV battery, which stabilizes the HV bus voltage during transient load conditions.

Low Voltage DC–DC Converter. For FCVs without a HV battery, the 12 V battery provides power to the HV DC bus and the fuel cell system when the vehicle is started. Therefore, a bidirectional DC–DC converter is needed to transfers power between the HV DC bus and the LV DC bus in both directions. Due to the slow dynamic properties of the fuel cell, a large voltage range applies for the HV DC bus voltage (Table 1.3).

In the case of an on-board HV battery, the electric system becomes similar to that of a series hybrid vehicle with the generator being replaced with a fuel cell. Thus, the specifications listed in Table 1.2 apply.

Boost mode		
HV port voltage range:	$V_1 = 230 \mathrm{V} \dots 450 \mathrm{V}$	
LV port voltage range:	$V_2 = 8.5 \mathrm{V} \dots 16 \mathrm{V}$	
Rated power, boost mode:	$P_{\rm out,nom} = 3 \rm kW$ for 20 seconds	
Buck mode		
HV port voltage range:	$V_1 = 230 \mathrm{V} \dots 450 \mathrm{V}$	
LV port voltage range:	$V_2 = 13.3 \mathrm{V} \dots 14.4 \mathrm{V}$	
Rated power, buck mode:	$P_{\rm out,nom} = 2 \rm kW,$ continuous	
Galvanic isolation:	required due to safety reasons [5]	
Other requirements:	low cost, high efficiency, high power density,	
	low weight $[22, 23, 24, 25]$	

Table 1.3: Specification of a bidirectional DC–DC converter for FCVs without HV battery (according to [20]); the output power, $P_{\text{out,nom}} = P_{\text{out}}$, is defined with (3.14).

1.1.2 Regenerative Power Systems

A variety of different applications of fuel cells (FCs), used to generate clean power from hydrogen and oxygen, exist. Besides the use in automobiles, typical applications include stand-alone residential power and small distributed power generation plants [29].

An ideal power generation system provides a sinusoidal voltage with constant magnitude and frequency, independent of load changes. The FC, however, represents a DC voltage source with a comparably high internal impedance and slow dynamics; varying power demand can only slowly be regulated by changing the flow-rate of supplied fuel. Thus, a typical power conditioning system (Figure 1.5) contains a battery which supports the fuel cell during transient power variations. FC and battery are both connected to a DC bus using different DC–DC converters in order to maintain a constant DC bus voltage (e.g. typically $\approx 400 \text{ V}$). For the FC, a unidirectional DC-DC converter is employed, for the battery, however, a bidirectional DC-DC converter is required in order to allow for charging and discharging modes of operations. The nominal battery voltage is typically selected according to the required power in order to obtain technically reasonable currents: for high power applications, e.g. higher than 5 kW, a HV battery may be selected, and for lower power requirements (e.g. below 5 kW), a low voltage battery with a nominal terminal voltage of 12 V, 24 V, 36 V or 48 V is typically used, due to higher reliability and lower cost [17, 27, 30, 31]. With a low voltage battery



Figure 1.5: Power system architecture of a FC power generation system; the battery supports the fuel cell during transient load conditions [29].

being employed, it is reasonable to consider a bidirectional DC–DC converter with a transformer, by reason of the large voltage ratio between the DC bus voltage and the battery voltage [32].

In literature, similar power conditioning systems are described for standalone photovoltaic (PV) power generation systems; e.g., in [33], a battery and an ultra-capacitor interface to the HV DC bus via bidirectional DC–DC converters is proposed in order to provide a stable bus voltage (Figure 1.6). Since the solar panel cannot accept electric energy, a unidirectional DC–DC converter transfers power from the solar panel to the DC bus, whereas a bidirectional converter is used to charge and discharge the battery. Again, the battery may be selected according to the required power level. For a nominal power of up to 5 kW, a low voltage battery is typically used.

Thus, a bidirectional DC–DC converter similar to the converter employed in automotive applications can be used for FC power plants and PV power systems. However, different to the automotive applications, equal nominal power requirements apply to the two directions of power transfer (i.e. $P_{2,\text{nom}} = -P_{1,\text{nom}}$, cf. (3.14)).

1.1.3 Uninterruptible Power Supplies (UPS)

A UPS reliably provides electric power to important loads. One prominent UPS system is the so-called on-line UPS which contains a mains connected rectifier, a battery-buffered HV DC bus, and an inverter (Figure 1.7, [34]). During normal operation of the mains, the rectifier provides electric power to



Figure 1.6: Power system architecture of a PV power generation system: the battery provides electric power during a shortage of electric power obtained from the PV panel, and the ultra-capacitor provides power during transient load conditions [33].

the HV DC bus and to the bidirectional DC–DC converter in order to charge the battery; a triac can be used to bypass the inverter. During a mains power failure, the battery provides power to the load (via the bidirectional DC–DC converter and the inverter).

Similar to regenerative power systems, different batteries may be used, depending on the required output power. If a low voltage battery is used (e.g. nominal voltage of 12 V, 24 V, 36 V, or 48 V), a bidirectional DC–DC converter with transformer is typically employed, due to the large ratio $V_1 : V_2$ of the DC bus voltages (e.g. 400 V : 12 V). Thus, for UPS applications and for regenerative power systems, basically the same specifications apply for the respective bidirectional DC–DC converters (cf. Section 1.1.2).



Figure 1.7: Power system architecture of the on-line UPS presented in [34].

1.2 Specifications of the Investigated DC–DC Converter

The specifications of the bidirectional DC–DC converter investigated in this work (Table 1.4) have been compiled in collaboration with industry partners. The DC–DC converter comprises of a high voltage (HV) port with the terminal voltage V_1 (240 V $\leq V_1 \leq 450$ V; nominal voltage $V_1 = 340$ V) and a low voltage (LV) port with the terminal voltage V_2 (11 V $\leq V_2 \leq 16$ V; nominal voltage $V_2 = 12$ V). A nominal output power P_{out} of 2 kW is required within the specified voltage ranges and in both directions of power transfer. Moreover, the low voltage port needs to be isolated from the high voltage port.

Further objectives regarding the selection and the design of the bidirectional DC–DC converter are (in the order of the considered priorities):

- 1. Converter efficiency > 90% at the nominal operating point ($V_1 = 340 \text{ V}$, $V_2 = 12 \text{ V}$, $P_{\text{out}} = \pm 2 \text{ kW}$),⁴
- 2. Converter efficiency > 90% within reasonable input and output voltage ranges and reasonable power ranges,
- 3. Low converter volume (i.e. high power density).

⁴Positive values P_{out} denote a power transfer from the HV port to the LV port and negative P_{out} denotes a power transfer from the LV port to the HV port, cf. (3.14).

An important design parameter to be defined in this Chapter is the switching frequency $f_{\rm S}$. The selected value, $f_{\rm S} = 100$ kHz, however, is rather based on general considerations than on a in-depth system optimization. Besides, a constant switching frequency is required by the industry partners in order to reduce potential EMI issues.

Provided that no thermal limitations apply, the volumes of inductors, transformers, and capacitors decrease with increasing switching frequency. Therefore, the power density of a converter can be increased by increasing the switching frequency [35]. However, the local conduction losses due to high frequency effects (skin and proximity effects), the local core losses, and the switching losses increase with increasing switching frequency. Thus, thermal limitations apply at high switching frequencies and the required total converter volume increases if the switching frequency becomes higher than the optimum switching frequency (calculated with respect to maximum total power density) [36, 37, 38, 39].

In [38, 39] two different DC–DC converters are optimized with respect to maximum power density. The following list summarizes the obtained results.

- 1. From [38]: resonant LCC DC–DC converter with unidirectional power flow, $V_1 = 320$ V, $V_2 = 26$ V, $P_2 = 3.9$ kW ($I_2 = 150$ A), considered heat sink: aluminum heat sink optimized with respect to low volume; calculated power density: p = 7.48 kW/dm³, calculated efficiency: $\eta =$ 93.9%, calculated optimum switching frequency: $f_{\rm S} = 147.1$ kHz;
- 2. from [38]: resonant LCC DC–DC converter with unidirectional power flow, $V_1 = 320$ V, $V_2 = 26$ V, $P_2 = 3.9$ kW ($I_2 = 150$ A), considered heat sink: copper heat sink optimized with respect to low volume; calculated power density: p = 13.3 kW/dm³, calculated efficiency: $\eta = 93.7\%$, calculated optimum switching frequency: $f_{\rm S} = 176$ kHz;
- 3. from [39]: DC–DC converter with unidirectional power flow (integrated current doubler topology), $V_1 = 400 \text{ V}$, $48 \text{ V} \le V_2 \le 56 \text{ V}$, $P_2 = 5 \text{ kW}$ (105 A > I_2 > 89 A), considered heat sink: copper heat sink optimized with respect to low volume; calculated power density: $p = 10.2 \text{ kW/dm}^3$, calculated efficiency: $\eta = 96.1\%$, calculated optimum switching frequency: $f_{\rm S} = 92.9 \text{ kHz}$ (for $V_2 = 54 \text{ V}$).

Besides the unidirectional power transfer capability, the specifications of the above listed DC–DC converters are similar to the required bidirectional DC–DC converter, i.e. optimum switching frequencies in the range $\approx 90 \text{ kHz} < f_{\rm S} < 200 \text{ kHz}$ result.

The considered switching frequency, $f_{\rm S} = 100$ kHz, is selected close to the lowest value of the above presented optimal switching frequencies, due to the expected impact of skin and proximity effects on the conduction losses of the

Property	Value	Description		
	$240 \mathrm{V} \le V_1 \le 450 \mathrm{V}$	minimum and maximum voltages at		
V_1		the high voltage (HV) port		
	$340\mathrm{V}$	nominal HV port voltage		
	$11\mathrm{V} \le V_2 \le 16\mathrm{V}$	minimum and maximum voltages at		
V_2		the low voltage (LV) port		
	$12\mathrm{V}$	nominal LV port voltage		
$P_{\rm out}$	$2\mathrm{kW}$	nominal output power, $\mathrm{HV} \to \mathrm{LV}$		
	$-2\mathrm{kW}$	kW nominal output power, $LV \rightarrow HV$		
$f_{ m S}$	$f_{\rm S}$ 100 kHz switching frequency			
Additional requirements:				
• galvanic isolation				
• high conversion efficiency $(>90\%$ at the nominal operating point)				
• low converter volume				
• constant switching frequency				

Table 1.4: Specifications of the investigated bidirectional DC-DC converter.

LV side (Section 4.2 and Appendix A.2.3) and due to the expected switching losses on the LV side (Section 4.3.3)

1.3 Objectives and New Contributions of the Work

The objective of this work is to select a converter topology which is suitable to fulfill the requirements specified in Section 1.2. Furthermore, the selected converter topology is optimized with respect to a high converter efficiency in order to meet the target of a required converter efficiency of more than 90% within reasonable input and output voltage ranges and reasonable power ranges.

The new contributions of this work are:

- a comprehensive overview on bidirectional DC–DC converter topologies, including a comparison of the different converters that are suitable to fulfill the given specifications (Chapter 2); in that context, the extensions of the unidirectional resonant LCC and LLC converters with respect to bidirectional operation are investigated;
- an optimal modulation scheme for the lossless DAB converter (with

respect to minimum transformer RMS current; Section 3.1.3);

- a detailed model used to predict the power losses of the investigated DAB converter, which includes the impact of high frequency harmonic components on the conduction and copper losses, switching losses, and core losses (Chapter 4);
- an examination of the results obtained from a numerical optimization of the modulation parameters D_1 and D_2 of the DAB converter with respect to maximum converter efficiency and a synthesis of an efficiency optimized modulation scheme of the given DAB converter (Section 5.2);
- a detailed small-signal control-oriented model of the DAB converter, which includes the dynamic properties of the EMI filters connected to the converter ports (Chapter 6);
- a design procedure to determine appropriate values for the turns ratio n and the inductance L of the DAB converter (Appendix A.2).

Six scientific papers and one patent have been published in the course of this dissertation:

- F. Krismer, J. W. Kolar, "Accurate Small-Signal Model for the Digital Control of an Automotive Bidirectional Dual Active Bridge," IEEE Transactions on Power Electronics, vol. 24, no. 12, pp. 2756–2768, Dec. 2009.
- F. Krismer, J. W. Kolar, "Accurate Power Loss Model Derivation of a High Current Dual Active Bridge Converter for an Automotive Application," IEEE Transactions on Industrial Electronics, vol. 57, no. 3, pp. 881–891, March 2010.
- F. Krismer, J. W. Kolar, "Efficiency-Optimized High Current Dual Active Bridge Converter for Automotive Applications," IEEE Transactions on Industrial Electronics, accepted for future publication, 2011.
- F. Krismer, J. Biela, J. W. Kolar, "A Comparative Evaluation of Isolated Bi-directional DC/DC Converters with Wide Input and Output Voltage Range," in Conf. Rec. of the IEEE 40th Annual Industry Applications Conference IAS 2005, Hong Kong, China, 2–6 Oct. 2005, pp. 599–606.
- F. Krismer, S. Round, J. W. Kolar, "Performance Optimization of a High Current Dual Active Bridge with a Wide Operating Voltage Range," in Proc. of the IEEE 37th Annual Power Electronics Specialists Conference PESC 2006, Jeju, Korea, 18–22 June 2006, pp. 1–7.

- F. Krismer, J. W. Kolar, "Accurate Small-Signal Model for an Automotive Bidirectional Dual Active Bridge Converter," in Proc. of the 11th Workshop on Control and Modeling for Power Electronics COMPEL, Zürich, Switzerland, 17–20 Aug. 2008, pp. 1–10.
- F. Krismer, J.W. Kolar: "Verfahren zur schaltverlustminimalen Steuerung eines bidirektionalen nicht potentialgetrennten Gleichspannungswandlers mit überlappendem Ein- und Ausgangsspannungsbereich."

1.4 Chapter Overview

Chapter 2 gives an overview on different bidirectional DC–DC converters and investigates the applicability of the different converter topologies with respect to the specifications given in Table 1.4. Based on the findings of Chapter 2, the single-phase DAB converter topology is considered most promising with respect to the achievable converter efficiency and the achievable power density. Moreover, due to the wide input and output voltage ranges, two-stage DC–DC converters, i.e. the series connection of an isolated DC–DC converter and a DC–DC converter without isolation, are expected to facilitate an improved converter efficiency.

In **Chapter 3**, the working principle of the DAB converter is explained based on different simplified converter models. This includes the lossless converter model (Figure 3.2), which facilitates basic analytical investigations and allows for a simplified synthesis of advanced modulation schemes. Moreover, two different refinements of the electric DAB converter model are presented in order to include the impacts of the conduction losses and the magnetizing current on the transformer currents.

Chapter 4 presents a detailed loss model of the DAB, needed in order to parameterize the models developed in Chapter 3 and to develop the efficiency optimized modulation presented in Chapter 5. This loss model is used to calculate the losses in the most relevant converter components, i.e. the semiconductor switches (conduction and switching losses), the transformer (copper and core losses), and the inductor (copper and core losses).

Chapter 5 details a systematic investigation of efficiency improvements achievable for the given DAB converter, obtained with the use of optimized modulation schemes. Finally, with the suboptimal modulation schemes discussed in Section 5.2.2, converter operation close to its maximum possible efficiency is achieved (Figure 5.28). The calculated results are verified using experimental data.

In **Chapter 6**, a detailed small-signal control-oriented model of the DAB and a dynamic model of the complete converter system (including the digital control platform) are derived. Moreover, Section 6.2.6 illustrates an effective method to derive a simplified transfer function of a given DAB converter. The calculated results are verified using experimental data.

Chapter 7 investigates the most promising two-stage converter configurations (out of the configurations depicted in Figure 2.22) in order to compare the efficiencies achieved with the two-stage topologies to the efficiencies obtained with the single-stage DAB converter. However, the average efficiency achieved with the two-stage converter ($\overline{\eta} = 93.0\%$; calculated according to Appendix A.1) is less than the average efficiency achieved with the single-stage DAB ($\overline{\eta} = 93.5\%$). Thus, in average, the additional losses of the non-isolated DC–DC converter cannot outweigh the loss reduction achieved for the DAB converter.

Chapter 8 summarizes the obtained results and concludes this work. Furthermore, Chapter 8 presents an outlook regarding future research in the field of the DAB converter.

The **Appendices** include:

- details on the converter designs (**Appendix A**);
- the methods used to calculate the VA ratings needed for the converter comparison (**Appendix B**);
- the calculation of the Steinmetz parameters k, α , and β (Appendix C);
- an evaluation of suitable power MOSFETs (**Appendix D**);
- information regarding the practical implementation of the proposed modulation schemes (**Appendix E**);
- supplemental information on the small-signal control-oriented model of the DAB converter (**Appendix F**).

Chapter 2

Bidirectional and Isolated DC–DC Converter Topologies

2.1 Derivation of Bidirectional DC–DC Converters

In principle, bidirectional power transfer between two unipolar DC voltage sources may be established with two unidirectional DC–DC converters C_1



Figure 2.1: Principle construction of a bidirectional DC–DC converter using two unidirectional DC–DC converters.



Figure 2.2: Unidirectional full bridge DC–DC converter with output inductor L_{DC2} . The converters C_1 and C_2 in Figure 2.1 can be replaced by the depicted unidirectional full bridge converter, however, for C_2 , the indices 1 and 2 need to be interchanged.



Figure 2.3: Bidirectional full bridge DC–DC converter topology, constructed from two unidirectional full bridge DC–DC converters C_1 and C_2 according to Figure 2.1 and Figure 2.2; the diodes D_A , D_B , D_C , and D_D are required in order to avoid reverse blocking voltages on $S_1 \dots S_8$ [40].

and C_2 according to Figure 2.1. There, C_1 is used to transfer power from port 1 to port 2 (forward direction, forward operating mode) and C_2 is needed to transfer power in the opposite direction (backward direction, backward operating mode). In order to illustrate an example of a practical converter realization including galvanic isolation, full bridge DC–DC converters with high frequency (HF) transformers and output inductors are employed for C_1 and C_2 (Figure 2.2).¹ According to [40], the inverter stages of C_1 and C_2 can

¹For the sake of clarity, the full bridge DC–DC converters C_1 and C_2 are considered lossless and without any parasitic inductive and capacitive components. Additionally, very large smoothing inductors (e.g. L_{DC2} in Figure 2.2) are considered and hence the characterization of the system functionality confines to the rectifiers being operated in


Figure 2.4: (a) Transformer voltage $v_{AC1}(t)$ generated with the inverter stage of the DC–DC converter depicted in Figure 2.2 and Figure 2.3 (for the assumption of an ideal HF transformer); (b) respective switching states of S_1 , S_2 , S_3 , and S_4 .

even be combined with the rectifier stages of C_2 and C_1 and thus, the resulting bidirectional DC–DC converter only contains a single HF transformer (Figure 2.3).

In forward mode of operation, i.e. power is transferred from port 1 to port 2 (cf. Figure 2.1), the converter C₁ is active and its inverter stage applies a rectangular AC voltage $v_{AC1}(t)$ with constant frequency $f_S = 1/T_S$ and arbitrary duty cycle 0 < D < 0.5 to the primary winding of the HF transformer (Figure 2.4). The (ideal) HF transformer simply alters the AC voltage amplitude according to its turns ratio n_{C_1} , i.e. $v_{AC2}(t) = v_{AC1}(t)/n_{C_1}$, and applies v_{AC2} to the rectifier. On the assumption of ideal low-pass filtering and continuous conduction mode of operation, the rectified output voltage V_2

continuous conduction mode.

results:

$$V_2 = 2 \cdot D \cdot V_1 / n_{C_1}.$$
 (2.1)

According to Figure 2.4, the duty cycle is limited to values between 0 and 0.5. Therefore, regarding (2.1), $V_2 < V_1/n_{C_1}$ must hold true for all operating points within the required voltage ranges (Table 1.4) and thus, the transformer turns ratio of C_1 is limited according to

$$n_{\rm C_1} < V_{1,\rm min} / V_{2,\rm max} = 240 \,{\rm V} / 16 \,{\rm V} = 15.$$
 (2.2)

In the backward operating mode, the converter C_2 is active and feeds power from port 2 to port 1 (Figure 2.1). Similar to (2.1), V_1 is calculated from

$$V_1 = 2 \cdot D \cdot V_2 / n_{C_2}.$$
 (2.3)

Hence, for operation within the specified voltage ranges,

$$n_{\rm C_2} < V_{2,\rm min}/V_{1,\rm max} = 11 \,{\rm V}/450 \,{\rm V} \approx 24.4 \cdot 10^{-3}$$
 (2.4)

applies to the transformer turns ratio of the converter C_2 .

According to (2.2) and (2.4), the turns ratio n of the HF transformer employed for the bidirectional DC–DC converter depicted in Figure 2.3 must be changed if the direction of power transfer changes: during forward mode of operation, n < 15 is required and during backward mode of operation, $n > 1/(24.4 \cdot 10^{-3}) \approx 41$ is needed. Thus, the converter shown in Figure 2.3 is considered to be inappropriate with respect to the required specifications (Table 1.4).

In a different approach, the switches and the diodes of a unidirectional full bridge DC–DC converter are modified according to Figure 2.5 in order to achieve bidirectional power transfer within the specified voltage ranges. There, the switches S_1 , S_2 , S_3 , and S_4 as well as the diodes D_5 , D_6 , D_7 , and



Figure 2.5: Bidirectional full bridge DC–DC converter, derived from the unidirectional DC–DC converter depicted in Figure 2.2.



Figure 2.6: (a) Transformer voltage $v_{AC2}(t)$ and (b) transformer current $i_{AC2}(t)$ generated in backward mode (for the assumption of an ideal HF transformer); (c) respective switching states of S₅, S₆, S₇, and S₈ (cf. Figure 2.5). During the time intervals II $(D \cdot T_{\rm S} < t_{\rm II} < T_{\rm S}/2)$ and IV $[(D + 1/2) \cdot T_{\rm S} < t_{\rm IV} < T_{\rm S}]$ the switches S₅, S₆, S₇, and S₈ remain in the on-state in order to magnetize $L_{\rm DC2}$; hence, $v_{\rm AC2}(t_{\rm II}) = 0$ and $v_{\rm AC2}(t_{\rm IV}) = 0$. Thus, the rectifier diodes D₁, D₂, D₃, and D₄ are blocking within the time intervals II and IV, i.e. $i_{\rm AC2}(t_{\rm II}) = i_{\rm AC2}(t_{\rm IV}) = 0$.

 D_8 are extended with respective diodes and switches connected in parallel. With this modification, the operation in forward mode remains unchanged and the output voltage is still determined with (2.1). In backward mode of operation, the switching states shown in Figure 2.6 (c) are employed in order to impress a rectangular AC current $i_{AC2}(t)$ [Figure 2.6 (b)] on the secondary winding of the HF transformer. The primary side transformer current $i_{AC1}(t) = i_{AC2}(t)/n$ is then rectified with $D_1 \dots D_4$ and filtered with C_{DC1} ,

$$I_1 = 2 \cdot D \cdot I_2/n. \tag{2.5}$$

Due to the absence of losses, the same input and output power levels result, i.e. $V_2 \cdot I_2 = V_1 \cdot I_1$. This, together with (2.5), yields the voltage transfer ratio

$$V_1 = V_2 \cdot n/(2 \cdot D) \tag{2.6}$$

in backward mode, which is identical to the forward mode voltage transfer ratio (2.1). Thus, (2.2) determines the required transformer turns ratio for backward operation, as well; the given circuit allows for bidirectional power transfer according to the specifications given in Table 1.4.

2.2 Topological Overview

The different components of a bidirectional DC–DC converter with galvanic isolation are depicted in Figure 2.7:

- The port 1 and port 2 filter networks provide smooth terminal voltages and currents. For each filter network, at least a single capacitor or a single inductor is employed.
- The *DC*-*AC* converter is a switch network which provides AC power to the HF transformer and the *AC*-*DC* converter supplies DC power to the receiving port; both converters must allow for bidirectional power



Figure 2.7: The different components required for an isolated, bidirectional DC–DC converter.

transfer. Typically, full bridge circuits, half bridge circuits, and pushpull circuits are employed. However, different solutions (e.g. the single switch networks used in a bidirectional flyback converter) are reported, as well [41,42].

- The *reactive HF networks* provide energy storage capability within the HF AC part and are used to modify the shapes of the switch current waveforms in order to achieve low switching losses (cf. Section 4.3.2, [35, 43]). Even though, these parts are not necessarily required for a fully functional bidirectional DC–DC converter, they will always be present in practice due to the parasitic components of the HF transformer (e.g. stray and magnetizing inductances, parasitic capacitances).
- The *HF transformer* is required in order to achieve electric isolation; it further enables large voltage and current transfer ratios. The HF transformer is considered superior over a low frequency transformer, since transformer and filter components become smaller (and often less expensive) at higher frequencies [35].²

Bidirectional DC–DC converter topologies with a system configuration according to Figure 2.7, are called *Single-Stage Topologies* [45, 46] (Section 2.2.1), since they contain a minimum number of conversion stages. Accordingly, the total number of required components is comparably low. However, the operation within wide input and output voltage ranges causes ineffective transformer and switch utilization. Improved transformer and switch utilization is achieved with multi-stage topologies, which contain an additional power converter in order to adjust voltage and current levels (Section 2.2.2; the efficiencies obtained from different two-stage topologies are presented in Section 7.2.8).

2.2.1 Single-Stage Topologies

In the presented approach, single-stage topologies are grouped into

- 1. converters with a low number of switches,
- 2. dual bridge converters without resonant HF network, and
- 3. dual bridge converters with resonant HF network,

according to the increasing complexity of the investigated systems (Figure 2.8).

²However, the converter losses (e.g. magnetizing losses, switching losses, losses due to skin- and proximity-effects) increase with increasing frequency; depending on the requirements and on the employed topology, an optimum switching frequency can be determined [44].



Figure 2.8: Topological overview of bidirectional DC–DC converter topologies. The topologies are classified according to the number of conversion stages, the employed switch circuits, the complexity of the HF network, and the port types (cf. Figure 2.7 and Figure 2.9).

The first group contains isolated and bidirectional flyback, forward, and Ćuk converter topologies which are typically employed for applications with an output power well below 2 kW. Main advantages of the forward and flyback converters are simple circuit structures and low numbers of required switches; however, ineffective transformer and switch utilizations render these topologies inappropriate with respect to the given specifications.

Improved utilization of the HF transformer is achieved with the Ćuk converter topology presented in [42], which has been designed according to the specifications given in Table 1.4 in order to investigate advantages and disadvantages (additional assumptions are listed in Table 2.1).³ The component stress values are calculated according to Appendix B and [42]; Table 2.1 lists the respective results. In anticipation of the results calculated for the final converter candidates (Table 2.2), very competitive VA ratings are found for the HF transformer and for the switches of the Ćuk converter. However, a number of major disadvantages render this converter less attractive:

- The switches are operated in a hard switching manner (cf. Section 4.3.2). In order to achieve a high efficiency and a high switching frequency, snubber networks are required [42].
- The switch stress is not distributed over a number of components: a single HV switch with a VA rating of 11.0 kVA and a single 10.2 kVA LV switch are needed. At the switching instant, high instantaneous voltages (HV: 754 V, LV: 40 V, with the voltage spikes during switching being neglected) and high instantaneous currents (HV: 21.4 A, LV: 407 A) occur.
- Two DC blocking capacitors are required and both carry the full transformer current (HV: 9.9 A, LV: 189 A). Thus, a capacitor with a very high current rating of 189 A is required on the LV side.
- Two filter inductors with a total magnetic energy storage capability of 89 mJ are required.

The converter topologies of the second group in Figure 2.8 contain DC–AC and AC–DC converters with more than a single switch: either full bridges, half bridges or push-pull converters are employed. Furthermore, a simple, non-resonant HF network – most often an inductor in series to the transformer stray inductance or the transformer stray inductance itself – is typically considered.

³The transformer turns ratio n = 19 has been selected in order to achieve an approximately balanced VA rating for the switches on the HV and on the LV side. Additionally, a maximum blocking voltage of 754 V results for the HV side switches and thus, MOSFETs with a breakdown voltage of 1000 V have to be employed.

Assumptions	
Assumed efficiency at full load:	90%
Selected turns ratio $n = N_1 : N_2$:	19
Peak-to-peak current ripple, HV side:	$\leq 40\%$ of full load DC curr.
Peak-to-peak current ripple, LV side:	$\leq 40\%$ of full load DC curr.
	(selected in accordance to
	Appendix A.5.2)
HV side switches	
Peak voltage:	$754\mathrm{V}$
Peak current:	$21.4\mathrm{A}$
Max. RMS current:	14.6 A
VA rating:	11.0 kVA
LV side switches	
Peak voltage:	$40\mathrm{V}$
Peak current:	407 A
Max. RMS current:	$258\mathrm{A}$
VA rating:	$10.2\mathrm{kVA}$
Transformer	
Max. equiv. value of the HV side	
transf. voltage [cf. (B.5)]:	403 V
Max. HV side RMS current:	9.9 A
Max. equiv. value of the LV side	91.9 W
Max LV side BMS current:	21.2 V 180 Δ
VA rating	4.0 kVA
HV side inductor	1.0 K VII
Employed inductance:	020 uH
Peak current:	10 0 Δ
Peak energy	46 m I
	10 110
LV side inductor	
Employed inductance:	1.7 μH
Peak current:	224 A
Peak energy:	$43\mathrm{mJ}$

 Table 2.1: Ćuk converter topology: assumed specifications and results.



Figure 2.9: DC-AC converter with different terminations of the DC port: (a) voltage sourced port, (b) current sourced port.

Compared to the flyback and forward converters, the converters of the second group achieve a much more effective converter utilization. In particular, the semiconductor switches can be operated with low switching losses (soft switching, Section 4.3.2) and hence, a converter with a high power density is feasible.

Most prominent members of this second group are the aforementioned bidirectional full bridge DC–DC converter (Section 2.1) and the Dual Active Bridge converter (Figure 2.10, Chapters 3, 4 and 5).

The third group contains DC–DC converters with resonant HF networks (so-called resonant DC–DC converters), which, in particular, offer increased capabilities regarding the utilization of the semiconductor switches, since the resonant network modifies the waveforms of the switch currents; this property can be used to achieve low switching losses (cf. Section 4.3.2, [43]). Disadvantages are the higher converter complexity and the additionally required power components.

Besides the classification with respect to the complexity of the regarded power converter (3^{rd} row in Figure 2.8), single-stage topologies may as well be classified with respect to the type of filter impedances which are directly connected to the DC ports of the DC–AC and AC–DC converters in Figure 2.7. In case of a capacitor, the port is denoted *voltage sourced* [Figure 2.9 (a)] and in case of an inductor, the port is denoted *current sourced* [Figure 2.9 (b), [80]]. If higher order filters are employed, the impedance directly connected to the switching network determines the port type. This second kind of classification is employed to further distinguish between the different converter topologies (4th row in Figure 2.8).

With respect to the given specifications (Table 1.4) and in anticipation of the findings obtained from the discussions given below, the following 6 converter topologies are considered to be the most relevant single-stage topologies.

- Single-phase Dual Active Bridge (DAB) converter (Figure 2.10, designed in Appendix A.2)
- Three-phase DAB converter (Figure 2.11, designed in Appendix A.3)
- LLC converter (Figure 2.18, designed in Appendix A.4)
- Bidirectional and isolated full bridge topology (Figure 2.12, designed in Appendix A.5)
- Bidirectional and isolated current doubler topology (Figure 2.13, designed in Appendix A.6)
- Bidirectional and isolated push-pull topology (Figure 2.14, designed in Appendix A.7)

In order to allow for a meaningful comparison, the different converter topologies have been designed separately. For this, the transformer turns ratios and the HF network component values, needed for the DAB converters (single-phase and three-phase) and the LLC converter, are optimized with respect to maximum average efficiency (cf. Appendix A.1). Moreover, for the single-phase DAB converter and the LLC converter, two different modulation schemes have been considered (conventional phase shift modulation and efficiency optimal modulation, cf. Section 3.1 and Section 5.2.1). For the isolated full bridge converter, the isolated current doubler topology, and the isolated push-pull converter, the transformer turns ratio is limited according to (2.2) and the assumptions listed in Table A.15 and Table A.17 are used to design the converters.

Table 2.2 lists the resulting component stress values obtained for the different DC–DC converters (only the power stages are considered; additionally required EMI filter components are discarded). The results of the converter designs outlined in Appendix A (i.e. component values, calculated average converter efficiencies) are given in Table 2.3.

Dual Bridge Converters without Resonant Network

Most prominent converter topologies within this group are the Dual Active Bridge (DAB) converter [49, 50, 51, 52, 53] and the bidirectional and isolated full bridge converter [46, 55, 56, 57] (including converter variants, e.g. with a current doubler circuit [58, 59] or a push-pull circuit [47, 62, 63, 65] being employed on the LV side).

Topological	Overview
-------------	----------

	DAB	DAB		TTC	TTC	Full	Bridge Conve	erter
	(Conv.	(Opt.	Three-	(Conv.	(Opt.		+	
Topologies	Mod., Sec-	Mod., Sec-	Phase	Mod.	Mod	Current	Commont	Current
	$tion \ 3.1.2)$	tion 5.2.1)	DAB [54]	Appendix A.4.3)	Appendix A.4.4)	Fed Full Bridge	Doubler	Fed Push Pull
Circuit	Fia	2 10	Fic 211	Fig.	9.18	Fig. 2.12	Fig. 2.13	Fig 214
0110	0	0.1		0.1				
No. of act. components	×	×	12	×	×	×	9	9
$HV \ side$								
Switch peak volt.	$450 \mathrm{V}$	$450\mathrm{V}$	$450\mathrm{V}$	$450\mathrm{V}$	$450\mathrm{V}$	$450\mathrm{V}$	450 V	$450 \mathrm{V}$
Switch peak curr.	$29.0\mathrm{A}$	$23.9\mathrm{A}$	$14.2\mathrm{A}$	$23.2\mathrm{A}$	$22.2\mathrm{A}$	$22.6\mathrm{A}$	$25.9\mathrm{A}$	$22.6\mathrm{A}$
Max. switch RMS current	$11.0 \mathrm{A}$	$9.6\mathrm{A}$	$6.4\mathrm{A}$	$10.0\mathrm{A}$	$9.8\mathrm{A}$	$14.2\mathrm{A}$	$15.9\mathrm{A}$	$14.2\mathrm{A}$
$\sum_{\rm VA \ rating}$	$19.8\mathrm{kVA}$	$17.3\mathrm{kVA}$	$17.4\mathrm{kVA}$	$18.0\mathrm{kVA}$	$17.6 \mathrm{kVA}$	$25.5 \mathrm{kVA}$	$28.6 \mathrm{kVA}$	$25.5\mathrm{kVA}$
$LV \ side$								
Switch peak volt.	$16 \mathrm{V}$	16 V	$16 \mathrm{V}$	16 V	16 V	$37.5\mathrm{V}$	75 V	75 V
Switch peak curr.	$550\mathrm{A}$	$407\mathrm{A}$	$285\mathrm{A}$	$465\mathrm{A}$	$377 \mathrm{A}$	$236\mathrm{A}$	251 A	$236\mathrm{A}$
Max. switch RMS current	$208\mathrm{A}$	$163\mathrm{A}$	$129\mathrm{A}$	$203\mathrm{A}$	$164\mathrm{A}$	$157 \mathrm{A}$	$159 \mathrm{A}$	$157 \mathrm{A}$
$\sum_{\rm VA \ rating}$	$13.3\mathrm{kVA}$	$10.5\mathrm{kVA}$	$12.4\mathrm{kVA}$	$13.0\mathrm{kVA}$	$10.5\mathrm{kVA}$	23.5 kVA	23.8 kVA	$23.5\mathrm{kVA}$

 Table 2.2: Converter topologies: semiconductor switches.

Bidirectional	\mathbf{and}	Isolated	DC–DC	Converter	Topologie	3

rter	Current Fed Push Pull	3	12:1	213 V, 22 imes 17.8 V	$\begin{array}{c} 20.0\mathrm{A},\\ 2\times157\mathrm{A}\end{array}$	$4.9 \mathrm{kVA}$	$\begin{array}{c} L: \ 0 \ \mathrm{A} \\ L_{\mathrm{DC2}}: 202 \ \mathrm{A} \end{array}$	$\substack{L: \ 20.0 \mathrm{A} \\ L_{\mathrm{DC2}}: 202 \mathrm{A} }$	$\substack{L: \ 22.6 \mathrm{A} \\ L_{\mathrm{DC2}}: 226 \mathrm{A} \\ \end{array}$	$\begin{array}{c} L: \ 1.6 \ \mathrm{mJ} \\ L_{\mathrm{DC2}}:21 \ \mathrm{mJ} \end{array}$	L: 1.2 kVA
Bridge Conve +	Current Doubler	3	12:2	213 V, 35.5 V	$22.5 \mathrm{A}, 135 \mathrm{A}$	$4.8\mathrm{kVA}$	$\begin{array}{c} L\colon 0\; \mathrm{A}\\ L_{\mathrm{DC2a}}\colon 101\; \mathrm{A}\\ L_{\mathrm{DC2b}}\colon 101\; \mathrm{A}\end{array}$	$\begin{array}{c} L:\ 22.5\ {\rm A} \\ L_{\rm DC2a}:102\ {\rm A} \\ L_{\rm DC2b}:102\ {\rm A} \end{array}$	$\begin{array}{c} L:\ 25.9\ {\rm A} \\ L_{\rm DC2a}:129\ {\rm A} \\ L_{\rm DC2b}:129\ {\rm A} \end{array}$	$L: 2.0 \text{ mJ} \\ L_{\text{DC2a}}:14 \text{ mJ} \\ L_{\text{DC2b}}:14 \text{ mJ}$	$L: 1.6 \mathrm{kVA}$
Full .	Current Fed Full Bridge	3	12:1	213 V, 17.8 V	$\begin{array}{c} 20.0\mathrm{A},\\ 241\mathrm{A} \end{array}$	$4.3 \mathrm{kVA}$	L: 0 A $L_{\mathrm{DC2}}:202 \text{ A}$	L: 20.0 A $L_{ m DC2}$:202 A	L: 22.6 A $L_{ m DC2}$:226 A	$L: 1.6 \mathrm{mJ}$ $L_{\mathrm{DC2}}:21 \mathrm{mJ}$	L: 1.2 kVA
LLC (Opt .	Mod., Appendix A.4.4)	3	17:1	$304 \mathrm{V}, 17.9 \mathrm{V}$	$^{13.8\mathrm{A},}_{231\mathrm{A}}$	$4.2 \mathrm{kVA}$	$L: 0 \mathrm{A}$ $L_{\mathrm{M}}: 0 \mathrm{A}$	L: 13.8 A $L_{\rm M}$: 1.0 A	$L: 22.2 \mathrm{A}$ $L_{\mathrm{M}}: 1.7 \mathrm{A}$	$L: 9.4 \mathrm{mJ}$ $L_{\mathrm{M}}: 0.6 \mathrm{mJ}$	$\begin{array}{c} L: \ 5.2 \mathrm{kVA} \\ L_{\mathrm{M}}: 0.3 \mathrm{kVA} \end{array}$
LLC (Conv.	Mod., Appendix A.4.3)	3	20:1	376 V, 18.8 V	$\begin{array}{c} 14.2\mathrm{A},\\ 287\mathrm{A} \end{array}$	$5.3 \mathrm{kVA}$	$L: 0 \mathrm{A}$ $L_{\mathrm{M}}: 0 \mathrm{A}$	L: 14.2 A $L_{ m M}$: 0.3 A	$L:~23.2\mathrm{A}$ $L_\mathrm{M}:~0.5\mathrm{A}$	$L: 14.9 \mathrm{mJ}$ $L_{\mathrm{M}}: 0.2 \mathrm{mJ}$	L:8.1 kVA $L_{\rm M}$: 115 VA
Three-	Phase DAB [54]	9	20:1	158 V, 7.9 V	$\begin{array}{c} 9.1\mathrm{A}, \\ 182\mathrm{A} \end{array}$	$4.3\mathrm{kVA}$	L: 0 A	$L: 9.1\mathrm{A}$	$L: 14.2{ m A}$	$3 \times L: 6.7 \mathrm{mJ}$	$3 \times L$: 3.8 kVA
DAB (Opt.	Mod., $Sec-tion 5.2.1)$	2	17:1	302 V, 17.8 V	$\begin{array}{c} 13.6\mathrm{A},\\ 231\mathrm{A} \end{array}$	$4.1 \mathrm{kVA}$	L: 0 A	L: 13.6 A	$L: 23.9\mathrm{A}$	$L: 6.2 \mathrm{mJ}$	L: 3.1 kVA
DAB (Conv.	Mod., Sec- tion $3.1.2$)	3	19:1	338V, 17.8V	$\begin{array}{c} 15.6\mathrm{A},\\ 294\mathrm{A} \end{array}$	$5.2 \mathrm{kVA}$	L: 0 A	$L: 15.6 \mathrm{A}$	$L: 29.0 \mathrm{A}$	$L: 11.2 \mathrm{mJ}$	L: 4.1 kVA
	Topologies	No. of magn. components	Transformer turns ratio	Max. equiv. transf. volt. (HV, LV) cf. (B.4), (B.5)	Max. RMS transf. curr. (HV, LV)	Transf. VA	Max. DC inductor currents	Max. RMS ind. currents	Peak inductor currents	Peak inductor energies	Inductor VA ratings

 Table 2.2 cont.:
 Converter topologies: magnetic components

Topological Overview

ppologies	$\begin{array}{c} DAB \\ (Conv. \\ Mod., Sec. \\ tion \ 3.1.2 \end{array} \end{array}$	$\begin{array}{c} DAB\\ (Opt.\\ Mod., Sec.\\ tion 5.2.1 \end{array}$	Three- Phase DAB [54]	LLC (Conv. Mod., Appendix A.4.3)	LLC (Opt. Mod., Appendix A.4.4)	Full Current Fed Full Bridge	Bridge Conve + Current Doubler	rter Current Fed Push Pull
o. of citors	2	2	2	3	3	2	2	2
eak acitor tages	$C_{ m DC1};450{ m V}$ $C_{ m DC2}:16{ m V}$	$C_{\mathrm{DC1}}^{\mathrm{CDC1}}$:450 V C_{DC2} : 16 V	$C_{ m DC1};450{ m V}$ $C_{ m DC2}:16{ m V}$	$C_{ m DC1}:450{ m V}$ $C_{ m DC2}:16{ m V}$ $C:342{ m V}$	$C_{\rm DC1}:450{\rm V}$ $C_{\rm DC2}:16{\rm V}$ $C:230{\rm V}$	C_{DC1} :450 V	C_{DC1} :450 V	$C_{\mathrm{DC1}}:450\mathrm{V}$
k cap. ergies	$C_{\rm DC1}:333{\rm mJ}$ $C_{\rm DC2}:128{\rm mJ}$	$C_{ m DC1}:333 m mJ$ $C_{ m DC2}:128 m mJ$	$C_{ m DC1}:333~mJ$ $C_{ m DC2}:128~mJ$	$C_{ m DC1}:333{ m mJ}$ $C_{ m DC2}:128{ m mJ}$ $C:5.4{ m mJ}$	$C_{\rm DC1}:333{ m mJ}$ $C_{\rm DC2}:128{ m mJ}$ $C:3.6{ m mJ}$	$C_{ m DC1}:333{ m mJ}$	$C_{ m DC1}:333{ m mJ}$	$C_{ m DC1}$:333 mJ
. RMS rrents	$C_{ m DC1}$:14.9 A $C_{ m DC2}$:244 A	C_{DC1} :9.2 A C_{DC2} :125 A	$C_{ m DC1}:4.1 m A$ $C_{ m DC2}:66 m A$	$C_{ m DC1}:11.4 m A$ $C_{ m DC2}:271 m A$ C:14.2 m A	$C_{ m DC1}, 9.3 { m A}$ $C_{ m DC2}, 122 { m A}$ $C, 13.8 { m A}$	$C_{ m DC1}$:10.1 A	$C_{ m DC1}$: 10.7 A	C_{DC1} : 10.1 A
acitor ratings				C: 3.4 kVA	C:2.2 kVA			
lulator plexity	low	high	low	low	high	low	low	low
. range tations	no	по	no	no	no	$V_1>nV_2$	$V_1>2nV_2$	$V_1>nV_2$
st. freq. ration	yes	yes	yes	yes	yes	yes	yes	yes
ubber juired	no	no	no	no	no	yes	yes	yes

Table 2.2 cont.: Conv. topologies: capacitors & ancillary information.

	DAB (Conv.	DAB (Opt.	Three-	LLC (Conv.	LLC (Opt.	Hull	$Bridge \ Conver \\ +$	ter
Topologies	Mod. Section $3.1.2$)	$Mod_{5,2.1}$ Section 5.2.1)	Phase Phase DAB [54]	$\substack{(Mod.)\\Appendix\\A.4.3)}$	Appendix A.4.4)	$Current \\ Fed Full \\ Bridge$	Current Doubler	$Current \\ Fed \ Pull \\ Pull \\$
Advantages	 lowest comp. count cound. avoids DC ind. simple modula- tion 	 lowest comp. count. avoids DC ind. reduced cricr. curr. reduced cap. RMS currents currents rented cap. range ZVS achiev- able 	 comparatively low capacitor RMS curr. RMS curr. avoids DC ind. reduced circ. curr. 	- slightly reduced comp. stresses stresses to the conv. DAB DAB DAB DAB - avoids DC ind. - simple tion tion	avoids DC ind. DC ind. - reduced circ. curr. RMS curr. RMS curr. RMS curr. RMS curr. Capacitor RMS curr. Curr. FUV side: full range switches: ZCS achiev- able	– LV side low cap.: RMS curr. – HV side: Tull range ZVS	 lowest transf. curr.: transf. curr.: lowest losses in contact resistances e.g. transf. terminal) lowest sw. MOSFETs lowest sw. curr.:better mOSFETs lower losses) LV ouer losses) LV ouer losses) LV ouer losses) low RMS curr. RMS curr. ZVS 	 LV side: no high- side gate drivers LV side DC cap.: low RMS RMS curr. HV side: full range ZVS
Disadvantages	 limited ZVS range transf: lar, edited transf: lar, edited transf: large RMS curt. 	- LV DC cap::large ge RMS curr. - complex tion tion	 limited ZVS range high component count LV DC capacion: rather large RMS curr. 	 large L & C limited ZVS range transf.: large eirc. curr. eirc. curr. very large RMS curr. 	- rather L & C L & C - LV DC - LV DC - Cap.: large - RMS curr. - complex tion	 large DC ind. required may be meeded limited operating volt. range 	- increased transf. VA rating DC ind. DC ind. PC equired required may be may be may be needed - limited operating volt. range	 increased VA rating VA rating VA rating Inarging Inarg

Table 2.2 cont.: Converter topologies: advantages and disadvantages.

The Single-phase Dual Active Bridge (DAB) converter (Figure 2.10) contains two voltage sourced full bridge circuits or half bridge circuits (or even push-pull circuits [72]) and a HF transformer. The reactive network simply consists of an inductor L connected in series to the HF transformer; hence, the DAB directly utilizes the transformer stray inductance. Due to the symmetric circuit structure, the DAB readily allows for bidirectional power transfer.

The main advantage of the DAB are the low number of passive components, the evenly shared currents in the switches, and its soft switching properties.⁴ With the DAB converter topology, high power density is feasible [50, 54].

However, the waveforms of the transformer currents $i_{AC1}(t)$ and $i_{AC2}(t)$ highly depend on the actual operating point (i.e. V_1 , V_2 , and the output power P_{out} ; cf. Chapter 3); for certain operating points, very high transformer RMS currents result. Moreover, high maximum capacitor RMS currents $I_{C_{DC1}}$ and $I_{C_{DC2}}$ occur.

⁴Soft switching denotes low-loss switching processes according to Section 4.3.2. For the DAB, the operating range with soft switching is limited; in particular at low load conditions, soft switching may be lost [50].

Topologies	DAB (Conv. Mod., Chap- ter 3.1.2)	DAB (Opt. Mod., Chap- ter 3.1.3)	Three- Phase DAB [54]	LLC (Conv. Mod., Ap- pendix A.4)	LLC (Opt. Mod., Ap- pendix A.4)
$N_1: N_2 = n$	19:1	17:1	20:1	20:1	17:1
L	$26.7 \mu H$	$21.7 \mu H$	$22.2\mu\mathrm{H}$	$55.4\mu\mathrm{H}$	$38.3\mu\mathrm{H}$
C	-	-	-	$93.3\mathrm{nF}$	$135\mathrm{nF}$
L_{M}	-	I		$1600 \mu \mathrm{H}$	$400 \ \mu H$
Average efficiency (cf. Appen- dix A.1)	89.5%	93.7%	90.9%	90.0%	93.9%
Maximum losses	$\begin{array}{l} 361{\rm W}{\rm at}\\ V_1=240{\rm V},\\ V_2=11{\rm V},\\ P_2=2{\rm kW},\\ {\rm HV}{\rightarrow}{\rm LV} \end{array}$	$\begin{array}{l} 188 {\rm W} \ {\rm at} \\ V_1 = 240 {\rm V}, \\ V_2 = 11 {\rm V}, \\ P_2 = 2 {\rm kW}, \\ {\rm HV} {\rightarrow} {\rm LV} \end{array}$	$\begin{array}{l} 283 \mathrm{W} \mathrm{at} \\ V_1 = 240 \mathrm{V}, \\ V_2 = 11 \mathrm{V}, \\ P_2 = 2 \mathrm{kW}, \\ \mathrm{HV} {\rightarrow} \mathrm{LV} \end{array}$	$\begin{array}{l} 336 \ {\rm W} \ {\rm at} \\ V_1 = 240 \ {\rm V}, \\ V_2 = 16 \ {\rm V}, \\ P_2 = 2 \ {\rm kW}, \\ {\rm HV} {\rightarrow} {\rm LV} \end{array}$	$\begin{array}{l} 167 {\rm W} {\rm at} \\ V_1 = 450 {\rm V}, \\ V_2 = 11 {\rm V}, \\ P_2 = 2 {\rm kW}, \\ {\rm LV} {\rightarrow} {\rm HV} \end{array}$

Table 2.3: Optimized results for the DAB and the LLC converters (cf. Appendix A).

Particularly high transformer and capacitor RMS currents result if the conventional modulation scheme – the so-called phase shift modulation – is employed. A considerable reduction of these RMS currents is achieved with the use of the efficiency optimized modulation schemes presented in Chapter 5.

According to Table 2.2, comparatively small VA ratings of the LV side switches, a low VA rating of the HF transformer, and a low total value of the magnetic energy storage capability already result for the DAB when operated with the conventional modulation scheme. Moreover, the DAB contains the lowest number of inductors. Expectedly, the total VA ratings and the required magnetic energy storage capability can even be reduced with the use of the efficiency optimal modulation scheme. However, the very large LV side RMS capacitor currents (≤ 244 A or ≤ 125 A, depending on the modulation scheme) present a considerable challenge regarding the practical realization of the DC capacitor $C_{\rm DC2}$.

The actual DAB employs full bridge circuits on the HV side and on the LV side. Within the regarded DC–AC converter topologies (full bridge, half bridge, and push-pull), the full bridge circuits allow for the best converter utilization due to the following reasons.

- Only the *full bridge* can generate zero output voltage (i.e. the voltage applied to the HF network in Figure 2.7) and therefore enables the use of improved modulation schemes, cf. Chapter 5.
- The advantage of the *half bridge* is the lower amount of required switches (reduced hardware effort regarding gate drivers). For the half bridge, the switch RMS current ratings are twice the RMS current ratings of the switches used for the full bridge. Thus, similar total VA ratings result for the switches of the half bridge and the switches of the full bridge. However, the magnitude of the half bridge AC voltage is half



Figure 2.10: Dual Active Bridge (DAB) converter topology.

of the magnitude of the full bridge AC voltage. This presents a serious disadvantage with respect to the LV side circuitry: if a half bridge is employed on the LV side, transformer RMS currents of up to 590 A and switch RMS currents of up to 420 A occur (DAB, phase shift modulation, cf. Table 2.2).

• In order to connect the *push-pull* circuit, two transformer windings are required. Since each winding only conducts current during half a switching period, ineffective HF transformer utilization results (i.e. the transformer VA rating increases). However, the push-pull circuit only requires two low-side switches and thus, the hardware needed for the gate drivers is less complex. The current ratings of the employed switches are the same for the push-pull circuit and the full bridge circuit; the voltage rating of the switches of the push-pull circuits is twice the voltage rating of the switches used for the full bridge. Thus, similar total VA ratings are calculated for the full bridge and for the push-pull circuit. A comparison (full bridge vs. push-pull) is presented in [81].

The Three-Phase DAB (Figure 2.11) uses three half bridges on the HV side and another 3 half bridges on the LV side. It requires 3 converter inductors and 3 HF transformers (which can be consolidated on a single three-phase HF transformer [49, 54]). The three-phase DAB is operated with a modulation scheme similar to the conventional modulation scheme employed for the single-phase DAB (phase shift modulation, [54]). However, different to the single-phase DAB, further performance enhancements using alternative modulation schemes are not feasible for the three-phase DAB.

According to Table 2.2, very good overall performance is achieved with the



Figure 2.11: Three-phase Dual Active Bridge (DAB) converter topology.

three-phase DAB: a low total transformer VA rating, low switch VA ratings and a low magnetic energy storage capability. In particular, compared to the single-phase DAB, considerably smaller RMS capacitor currents result (66 A vs. 125 A on the LV side, Table 2.2).

A disadvantage of the three-phase DAB is the high number of active components needed: 12 semiconductor switches and, accordingly, 12 gate drivers (6 high-side gate drivers) are required. Moreover, high conduction and switching losses result for certain operating points if the converter is operated within wide voltage and power ranges, due to the restrictions regarding the employed modulation scheme.

The Bidirectional and Isolated Full Bridge Converter (Figure 2.12) contains a voltage sourced full bridge on the HV side and a current sourced full bridge on the LV side (using the DC inductor L_{DC2}). The power flow is typically controlled with the duty cycle D [cf. Figure 2.4 and (2.1)]. Additionally, the LV side full bridge needs to be appropriately controlled in order to allow for bidirectional power flow (e.g. Appendix A.5, [57, 58, 59, 61]). The bidirectional and isolated full bridge converter facilitates ZVS operation of the switches of the HV side; the switches of the LV side switch at zero current. Thus, a high switching frequency and a high power density are feasible (at low power levels, however, additional circuitry is needed; cf. [82, 83]).

Different to the DAB and the LLC converters, usable values for the transformer turns ratio are limited according to (2.2) (i.e. $V_1 > nV_2$; for operation with $V_1 \leq nV_2$, e.g. during system start-up, additional circuitry is required; cf. [55, 56]). Regarding the specifications given in Table 1.4, $n \leq 15$ results; due the additionally required commutation time interval, n = 12 is calculated in Appendix A.5 (Table 2.4 lists the design results). For the isolated full bridge converter, the VA ratings of the semiconductor switches – in par-



Figure 2.12: Bidirectional and isolated full bridge converter topology.

ticular on the LV side – are considerably higher than the VA ratings obtained for the DAB and the LLC converters (Table 2.2).

The isolated full bridge converter achieves a smooth current $i_{L_{DC2}}(t)$ (cf. Figure A.39) and therefore, compared to the DAB, a considerably lower capacitor RMS current $I_{C_{DC2}}$ results (≈ 16 A for the full bridge converters designed in Appendix A.5). A disadvantage is the additional volume required for the DC inductor. Furthermore, the LV side switches repeatedly connect the DC inductor L_{DC2} in series to the transformer stray inductance L and thus, a snubber circuit is needed in order to avoid voltage spikes when switching.

Instead of the LV side full bridge converter, different AC–DC converter topologies can be used; typical substitutes are the current doubler topology (Figure 2.13) or the push-pull topology (Figure 2.14). The advantages and the disadvantages of the three different AC–DC converters are summarized in the following.

- The switch current ratings and the sum of the switch VA ratings are similar for all three circuits.
- For the *full bridge* circuit, the highest transformer currents $i_{AC2}(t)$ occur (instantaneous and RMS values).
- For the *current doubler* circuit, two inductors are required instead of one. Still, due to partial ripple cancellation, the sum of the magnetic energy storage capabilities (28 mJ) is comparable to that of the DC inductors of the full bridge and the push-pull circuits (21 mJ). However, the AC magnitudes of $i_{L_{DC2a}}(t)$ and $i_{L_{DC2b}}(t)$ are considerably higher for the current doubler (provided that identical design assumptions are used; cf. Figure A.39 and Figure A.42); consequently, an increased transformer VA rating results.



Figure 2.13: Bidirectional and isolated converter with a current doubler on the LV side (side 2).

Compared to the full bridge circuit, twice the transformer voltage $v_{AC2}(t)$ is needed for the current doubler and therefore, lowest transformer currents $i_{AC2}(t)$ are achieved. The current doubler requires two (low-side) switches with a rated voltage of twice the rated voltage of the full bridge switches.

• The *push-pull* circuit connects to a center-tapped transformer with two LV side transformer windings. Thus, each of these windings conducts current only during half a switching period and, consequently, ineffective transformer utilization and a higher VA rating occur (Table 2.2). For the push-pull circuit, two (low side) switches with a rated voltage equal to twice the rated voltage of the full bridge switches are required.

Other Converter Topologies without Resonant HF Network. In [68], a DC–DC converter with minimum number of devices is discussed. It consists of a voltage sourced half bridge on the HV side and a modified voltage sourced half bridge on the LV side (Figure 2.15). Thus, the converter is a DAB converter with the LV side half bridge being additionally used as a bidirectional buck or boost converter (depending on the direction of power transfer). The transferred power is controlled with the phase shift between $v_{AC1}(t)$ and $v_{AC2}(t)$ (cf. conventional DAB converter).

With this converter, unequal RMS switch currents and higher maximum blocking voltages result for T_3 and T_4 (e.g. $2 \cdot V_{2,max}$ for converter operation according to [68]). Additionally, the LV side capacitors are exposed to high RMS currents, similar to the DAB. Due to the employed half bridge circuits,



Figure 2.14: Bidirectional and isolated converter with a push-pull circuit on the LV side (side 2).

	Ful	l Bridge Conv	erter
Topologies	Current Fed Full Bridge	$\stackrel{+}{Current} Doubler$	Current Fed Push-Pull
$N_1: N_2 = n$	12:1	12:2	12:1
L (HV side referred)	$6.4\mu\mathrm{H}$	$6.1\mu\mathrm{H}$	$6.4\mu\mathrm{H}$
$L_{\rm DC2}$ (LV side)	$0.83\mu\mathrm{H}$	—	$0.83\mu\mathrm{H}$
$L_{\rm DC2a}$ and $L_{\rm DC2b}$ (LV side)	_	$1.65~\mu H$	_

 Table 2.4: Calculated component values for the isolated full bridge converter topologies.



Figure 2.15: Bidirectional DC–DC converter with minimum number of devices [68].

no optimized modulation schemes – like the efficiency optimal modulation scheme for the DAB – can be applied to the isolation stage of the given converter. However, asymmetrical PWM waveforms may be generated in order to adjust V_2 ,⁵ which allows for a certain converter optimization. Still, the complex interaction of the DAB and the buck/boost stage, a limited soft switching range, high capacitor RMS currents and the additionally required DC inductor render this topology less suitable for the given application.

In a different approach, a voltage sourced half bridge circuit is used on the HV side and a current doubler circuit on the LV side [69]; an asymmetrical PWM modulation scheme is proposed to enable the control of the transferred

⁵The capacitors $C_{\rm DC1a}$, $C_{\rm DC1b}$, $C_{\rm DC2a}$, and $C_{\rm DC2b}$ keep the resulting DC component away from the HF transformer.

power. Consequently, uneven RMS currents occur in the LV side switches and in the LV side DC inductors. With this, comparably high current stresses result for all power components.⁶

Dual Bridge Converters with Resonant Network

Even though, resonant DC–DC converters with unidirectional power flow are thoroughly described in literature, only very few is found on bidirectional resonant converters. Reasons might be the higher converter complexity and the additional HF components needed. This section briefly summarizes advantages and disadvantages of the most common resonant DC–DC converters and discusses the extensions required to achieve bidirectional converter operation.

Series Resonant, Parallel Resonant, and Series-Parallel Resonant DC–DC Converters with unidirectional power flow belong to the most popular resonant converter topologies [35, 43, 84, 85]. With these converters, nearly sinusoidal waveforms of the transformer currents result. Thus, low switching losses are feasible, which facilitates the use of a high switching frequency to obtain a high power density of the converter [43]. The conventional resonant DC–DC converters are operated with variable switching frequency, i.e. the actual switching frequency depends on the supply voltage and on the load.

The main advantages and disadvantages of the most common unidirectional, resonant DC–DC converters are summarized in the following (according to [35]).

• The Series Resonant Converter (SRC) contains a capacitor in series to the transformer stray inductance, which acts like a DC blocking capacitor and prevents saturation of the HF transformer. Furthermore, the RMS currents in the switches and in the resonant network decrease with decreasing load. The main disadvantages of the SRC are related to wide operating voltage and power ranges, since the actual switching frequency strongly varies in dependence on the employed supply voltage and on the load; even more, the SRC cannot be regulated in case of no load (requires infinite switching frequency). The voltage-sourced output (e.g. LV side) is as well disadvantageous with respect to high RMS capacitor currents for applications with low output voltages.

⁶The converter presented in [69] is obviously not designed for operation within the voltage ranges specified in [69]; with respect to these voltage ranges, a transformer turns ratio of n < 2.5 would be needed, which would considerably decrease the converter efficiency and the achievable power density; with respect to the voltage ranges specified in Table 1.4, n < 3.75 results.



Figure 2.16: Unidirectional, series-parallel resonant LCC converter [35].

- The advantage of the *Parallel Resonant Converter (PRC)* is its ability to control the output voltage at no load. Additionally, with the employed output DC inductor, the PRC is better suited for applications with low output voltages and high output currents than the SRC is. However, for the PRC, the currents through the switches and through the resonant network do not substantially decrease with decreasing load. Thus, this converter is less suited for applications with large operating voltage and power ranges or for applications where the converter is most often operated considerably below its rated power.
- With the Series-Parallel Resonant Converter (SPRC), the output voltage can be controlled at no load and the range of the required switching frequency is smaller than for the SRC. Still, compared to the PRC, considerably lower switch and transformer RMS currents are achieved at low load conditions.

The HF network of the SPRC may either consist of one inductor and two capacitors according to the circuit discussed in [35] (often called LCC converter, Figure 2.16) or of two inductors and one capacitor (LLC converter, Figure 2.17, [86]).

With a circuit modification according to Section 2.1, bidirectional power transfer can be achieved. For the resulting resonant DC–DC converters with two voltage sourced ports (i.e. the bidirectional SRC and the bidirectional LLC converter, Figure 2.18), no restrictions regarding the converter control emerge. In contrast, for the PRC and for the LCC converters (Figure 2.19 (a)), difficulties arise with respect to the LV side full bridge. Due to the inductor $L_{\rm DC2}$ and the capacitor $C_{\rm P}$, only 4 different switching states are allowed on the LV side:

• T_5 , T_6 : on; T_7 , T_8 : off;



Figure 2.17: Unidirectional, series-parallel resonant *LLC* converter (cf. [86]).

- T_7 , T_8 : on; T_5 , T_6 : off;
- T_5 , T_8 : on; T_6 , T_7 : off;
- T₆, T₇: on; T₅, T₈: off.

All other switching states are forbidden, since they would either interrupt the inductor current $I_{L_{DC2}}$ (e.g. T_5 , T_6 , T_7 , and T_8 are turned off, for $I_{L_{DC2}} < 0$), cause a short circuit on C_P (e.g. T_5 , T_6 , T_7 , and T_8 : on), or both (e.g. T_6 , T_8 : on; T_5 , T_7 : off)—in particular during a state change of the LV side full bridge, it is very likely for the actual converter hardware to switch into one of the forbidden states for a short time. This issue can be overcome with the use of 4-quadrant switches on the LV side, according to Figure 2.19 (b), and an appropriate modulation scheme. The additional hardware effort, however, renders the PRC and the LCC converter topologies less attractive than the SRC and the LCC converters are not considered any further.

The SRC can be considered as a special case of the LLC converter, therefore the presented discussion is confined to the bidirectional LLC converter. Due to the series capacitor, C, approximately sinusoidal transformer currents i_{AC1} and i_{AC2} are obtained (in contrast to the DAB, where triangular or trapezoidal currents occur); the parallel inductor $L_{\rm M}$ increases the reactive power generated with the HF network ($L_{\rm M}$ can be advantageously used to reduce the switching losses on the HV side and on the LV side). Compared to the DAB, reduced RMS transformer currents and reduced switching losses can be achieved with the LLC converter, especially with respect to the operation within wide input and output voltage ranges.

For the LLC converter, full bridge circuits are used on the HV side and on the LV side due to the same reasons as mentioned for the DAB above.



Figure 2.18: Bidirectional, series-parallel resonant *LLC* converter with two voltage sourced ports; for the depicted converter topology, no restrictions regarding the control of the converter apply.



Figure 2.19: Bidirectional, series-parallel resonant LCC converter: (a) with 2-quadrant switches T_5 , T_6 , T_7 , and T_8 , the full bridge on the LV side (side 2) may reside in forbidden states, which cause a short circuit of C_P and / or an interruption of I_{LDC2} ; (b) with the LV side full bridge being equipped with 4-quadrant switches, the aforementioned forbidden states can be avoided.

The converter performance is analyzed using two different modulation schemes, which both allow for fixed frequency operation:

- The conventional modulation scheme operates the two full bridges with maximum duty cycle in order to generate purely rectangular voltages $v_{\rm AC1}$ and $v_{\rm AC2}$. The power transfer is controlled with the phase angle φ between $v_{\rm AC1}$ and $v_{\rm AC2}$, identical to the conventional operation of the DAB converter (Appendix A.4.3).
- The efficiency optimal modulation scheme utilizes the ability of the full bridges to generate zero output voltage (i.e. $v_{AC1} = 0$ or $v_{AC2} = 0$) in order to further reduce the transformer RMS currents and to achieve lower switching losses. The modulation parameters $(D_1, D_2, \text{ and } \varphi)$ with respect to maximum converter efficiency are obtained in the same way as for the DAB converter (Section 5.2.1, Appendix A.4.4).

The component values for the LLC converter $(L, C, L_M, \text{ and } n)$ have been optimized with respect to maximum average efficiency (Appendix A.1), the given converter specifications (Table 1.4) and for both modulation schemes.

If the LLC converter is operated with the conventional modulation scheme, the VA ratings for the HF transformer and the switches are slightly below the VA ratings obtained for the conventional DAB converter. However, an inductor L with a higher maximum energy storage capability is required (14.9 mJ for the LLC converter and 11.2 mJ for the DAB converter) and a series capacitor C is needed (the parallel inductor $L_{\rm M}$ can be neglected, since the optimization results in a very large value for $L_{\rm M}$).

A better utilization of the LLC converter is achieved if the employed modulation parameters are optimized with respect to maximum converter efficiency. With this method, low VA ratings are obtained for the HF transformer (4.2 kVA) and for all switches (HV side: 17.6 kVA, LV side: 10.5 kVA, cf. Table 2.2). Among the considered DC–DC converters with two voltage sourced ports, the best average efficiency is achieved for the LLC converter (Table 2.3).

The LLC converter could as well be operated with a DC inductor on the LV side. The respective circuit is depicted in Figure 2.20 and is derived from the bidirectional, isolated full bridge converter. Similar to the isolated full bridge converter, this LLC converter repeatedly switches L and $L_{\rm M}$ in series to $L_{\rm DC2}$, which causes voltage spikes on $v_{\rm AC2}(t)$; a snubber circuit is needed to reduce these voltage spikes. Moreover, compared to the bidirectional full bridge converters, a considerably higher transformer VA rating is expected due to the increase of the reactive power in the high frequency network (in particular with respect to the required input and output voltage ranges and the required constant switching frequency operation, Table 1.4). Taking the additionally required power components C and $L_{\rm M}$ into account, the LLC

converter with DC inductor is considered to be less attractive than the bidirectional full bridge converter and is thus not investigated.

Other resonant DC–DC converters. Different to the SRC, PRC, and SPRC converter topologies, a bidirectional converter topology with switched resonance frequency is discussed in [75, 76] (Figure 2.21). The converter contains a voltage sourced half bridge on the HV side, a voltage sourced full bridge on the LV side, and a 4-quadrant switch to change the effective resonance capacitor.

With this configuration, high efficiency has been demonstrated for a wide load range.⁷ However, the LV port voltage V_2 of the proposed converter cannot exceed V_1/n [i.e. the constraint (2.2) applies]. Moreover, with the use of alternative modulation schemes, similar transformer currents (waveforms and RMS values) can be achieved with the LLC converter (Figure 2.18). In summary, a higher total VA rating is expected to result for the converter depicted in Figure 2.21 than for the LLC converter (in particular due to the additionally required 4-quadrant switch) and thus, a higher power density is expected to be achieved with the LLC converter.

In another approach, the capacitance values of $C_{\rm DC1}$ and $C_{\rm DC2}$ of the DAB converter are considerably reduced in order to form a resonant converter which allows for low switching losses. Therefore, however, additional EMI filter components are necessarily required, since the EMI filters become part of the HF resonant network [77]. The oscillation frequency of the currents in the filter capacitors is twice the switching frequency and, additionally, very low currents are desired at the switching instants at $t \approx k \cdot T_{\rm S}/2$. Based on the switch current depicted in Fig. 11(b) in [77], the transformer current $i_{\rm AC1}$ can be

⁷In [75, 76] the switching frequency is reduced at low load conditions.



Figure 2.20: Bidirectional, series-parallel resonant *LLC* converter with a current sourced LV port.

approximated with

$$i_{\rm AC1}(t) = \begin{cases} (I_{\rm AC1,peak}/2) \cdot [1 - \cos(2 \cdot 2\pi f_{\rm S} t)] & \forall \quad 0 \le t < 1/(2f_{\rm S}) \\ -(I_{\rm AC1,peak}/2) \cdot [1 - \cos(2 \cdot 2\pi f_{\rm S} t)] & \forall \ 1/(2f_{\rm S}) \le t < 1/(f_{\rm S}) \end{cases}$$
(2.7)

 $[I_{AC1,peak}$ is the amplitude of $i_{AC1}(t)$]. Accordingly, the RMS value $I_{AC1} = I_{AC1,peak} \cdot \sqrt{3/8}$ and the rectified average value $|I_{AC1}| = I_{AC1,peak}/2$ result. Compared to a purely rectangular transformer current $i_{AC1,rect}(t)$ (there, $I_{AC1,rect}/|I_{AC1,rect}|$ is equal to 1), the conduction losses in the HF transformer and in the semiconductor switches will increase by a factor of $(I_{AC1}/|I_{AC1}|)^2 = 1.5$ due to the employed transformer current waveform.⁸

The converter with resonating filter capacitors is optimized with respect to low switching losses [77], however, comparably high conduction losses occur. Moreover, on the LV side, the resonating capacitors are exposed to very high RMS currents and therefore difficult to implement. This renders the topology less suitable with respect to the considered application.

2.2.2 Two-Stage Topologies

The isolated DC–DC converter topologies discussed in Section 2.2.1 are operated most efficiently if the input to output voltage ratio V_1/V_2 is close to the transformer turns ratio n. Thus, within the specified input and output voltage ranges, a more evenly distributed converter efficiency is expected for a two-stage solution which employs an additional buck and / or boost converter

⁸For a triangular current $(I_{AC1}/|\overline{I_{AC1}}|)^2$ is equal to 1.33 (e.g. DAB HF transformer current at certain operating points, cf. Section 3.1.3) and for a sinusoidal current $(I_{AC1}/|\overline{I_{AC1}}|)^2 = 1.23$ (e.g. achieved for the LLC converter at certain operating points).



Figure 2.21: Bidirectional, resonant converter with switched resonance frequency [75, 76].

without galvanic isolation to adjust the voltage applied to the isolated DC– DC converter stage. However, the total number of required power components is higher for the two-stage converter. Thus, the additional DC–DC converter needs to be highly efficient in order to achieve a higher efficiency with the two-stage converter than with the respective single-stage converter.

In this thesis, a bidirectional buck or boost converter, which allows for buck operation in one power transfer direction and boost operation in the opposite direction, provides the converter stage without galvanic isolation. With this, the 4 different arrangements depicted in Figure 2.22 are feasible:⁹

- *HV side buck converter* [Figure 2.22 (a)]: with this arrangement, the given specifications, and a maximum duty cycle $D_{\text{buck,max}} = 95\%$ of the buck converter, the DC link voltage V_i approximately remains within the voltage range $150 \text{ V} \le V_i \le 225 \text{ V}$ and enables efficient operation of the isolated DC-DC converter stage at (or close to) $V_i/V_2 = n$ within the whole specified operation range (cf. Chapter 5). Due to the reduced DC link voltage V_i , MOSFETs with a maximum voltage rating of 300 V can be employed for the HV side DC-AC converter of the isolated converter stage (cf. Figure 2.7), however, a comparably high maximum port current max(I_i) $\approx 15 \text{ A}$ results (for the assumption of 90% efficiency).
- With a *HV side boost converter* [Figure 2.22 (b)], the DC link voltage is larger than the HV port voltage. For the assumption of a minimum duty cycle $D_{\text{boost,min}} = 5\%$ of the boost converter, the DC link voltage range is $475 \text{ V} \le V_i \le 700 \text{ V}$ in order to enable efficient operation of the isolated DC-DC converter stage (i.e. $V_i/V_2 \approx n$). This exceeds the maximum voltage rating of the employed MOSFETs ($V_{(\text{BR})\text{DSS}} = 600 \text{ V}$). However, only for $V_{(\text{BR})\text{DSS}} \approx 600 \text{ V}$ highly optimized MOSFETs are available (Appendix D.1.2); Section 7 therefore presents a trade-off study on how to efficiently operate the isolated DC-DC converter stage and still reduce the maximum DC link voltage to $\max(V_i) = 475 \text{ V}$.
- The configuration with the LV side buck converter [Figure 2.22 (c)] allows for reduced currents in the LV side DC-AC converter (e.g. full bridge) of the isolated converter stage. However, the realization of a highly efficient low voltage and high current buck converter (e.g. 32 V and 200 A) is considerably more difficult than the realization of a 450 V and 15 A buck converter (cf. [25, 89]).

⁹A bidirectional buck and boost converter (allows for buck and boost operation in either direction [87,88]) could be employed to achieve a more flexible converter topology. However, the 4 power switches required for the bidirectional buck and boost converter are considered as a disadvantage with respect to the converter cost and the achievable power density.



Figure 2.22: Four different arrangements of a bidirectional buck or boost converter and a single-stage isolated bidirectional DC–DC converter: (a) HV side buck converter, (b) HV side boost converter, (c) LV side buck converter, and (d) LV side boost converter.

• The LV side boost converter [Figure 2.22 (d)] requires the DC link voltage to be lower than the LV port voltage, $V_i < V_2$ which leads to increased currents in the LV side switches and in the HF transformer. This arrangement is thus less suitable with respect to high converter efficiency.

2.3 Conclusion

The advantages and the disadvantages of the considered single-stage, bidirectional, and isolated DC–DC converter topologies are summarized in the following.

• DAB converter: the DAB converter topology (Figure 2.10) contains only

one inductor, L, with a comparably low peak energy storage capability (6.2 mJ on the assumption of efficiency optimal modulation parameters, Table 2.2); it features low VA ratings of the semiconductor switches (HV side: 17.3 kVA, LV side: 10.5 kVA), ZVS operation (within certain limits, cf. [50,90,91]), and is highly flexible regarding the employed modulation scheme (cf. Section 3.1.3). The DAB converter enables power transfer for $V_1 \ge n V_2$ and for $V_1 \le n V_2$ (*n* is the transformer turns ratio). Due to its symmetric circuit structure, the converter efficiency characteristics is independent of the actual power transfer direction.¹⁰ Disadvantages of the DAB converter are: the high maximum value of the RMS current through $C_{\rm DC2}$ (125 A), the high maximum transformer RMS current through the LV side winding (231 A), and the high complexity of the modulation algorithm required to generate the gate signals for the switches.

- Three-phase DABconverter: the three-phase DAB converter (Figure 2.11) features a low total value of the required peak magnetic energy storage capability $(3 \cdot 2.23 \text{ mJ} = 6.7 \text{ mJ})$, ZVS operation (within a limited operating range [54]), reduced capacitor RMS currents (66 A through C_{DC2}), and a symmetric circuit structure; the three-phase DAB converter can be operated with $V_1 > n V_2$ and with $V_1 < n V_2$. However, the system requires a high number of components (3 transformers, 3 inductors, and 12 switches). Moreover, the expected average efficiency, calculated according to Appendix A.1 ($\overline{\eta} = 90.9\%$), is less than the average efficiencies calculated for the single-phase DAB converter $(\overline{\eta} = 93.7\%)$ and the LLC converter $(\overline{\eta} = 93.9\%)$.
- *LLC converter:* compared to the DAB converter, a slightly higher average efficiency is calculated for the LLC converter (Figure 2.18); the semiconductor's VA ratings are similar (HV side: 17.6 kVA, LV side: 10.5 kVA). However, a larger value of the required magnetic storage capability occurs (9.4 mJ + 0.6 mJ) and an additional resonance capacitor, C, is needed for the resonant circuit (peak energy: 3.6 mJ). Thus, the resonant circuit of the LLC converter requires a larger volume than the inductor of the DAB converter (cf. [74]). Again, a large RMS current through C_{DC2} is calculated (122 A).

In contrast to the DAB converter, the electric energy stored in the resonance capacitor C forces the transformer current to change during the freewheeling time interval (i.e. for $v_{AC1} = v_{AC2} = 0$, cf. Figure 3.7), which causes the transformer RMS current to increase. Thus, for converter operation according to Appendix A.4 (i.e. continuous conduction

¹⁰Provided that parasitic effects, e.g. the impact of switching losses on the transformer current, are negligible; cf. Section 4.6.2.

mode operation), the LLC converter is more suitable if a variable switching frequency is permitted.

• Full bridge converters with one or more DC inductors on the LV side: all full bridge converters considered in Table 2.2 (Figure 2.12, Figure 2.13, and Figure 2.14) feature ZVS operation of the HV side semiconductors (at low power levels, additional circuitry is needed, cf. [82, 83]) and a low RMS current through C_{DC2} (≈ 10 A); moreover, the required gate signals can be generated in a straightforward manner (Appendix A.5). However, the energy storage capabilities of the DC inductors required for the full bridge converters are comparably large (21 mJ for the current fed full bridge and the push pull circuit; $2 \cdot 14$ mJ for the current doubler). For the semiconductor switches, high VA ratings result (HV side: ≈ 26 kVA, LV side: ≈ 24 kVA). Moreover, the operating voltages are limited according to (2.2), i.e. $V_1 > nV_2$. For operation with $V_1 \leq nV_2$, e.g. during system start-up, additional circuitry is required (cf. [55, 56]).

For all considered single-stage converter topologies, similar VA ratings are calculated for the HF transformers and thus, similar transformer volumes result [92].

Based on these findings, the single-phase DAB converter topology is considered most promising with respect to the achievable converter efficiency (cf. Table 2.3) and the achievable power density (due to the low number of inductors and due to the employed capacitive filters on the HV side and on the LV side [39], i.e. the DC capacitor C_{DC2} is used instead of the DC inductor L_{DC2}). Therefore, the DAB is investigated in detail in the following Chapters, where Chapter 3 describes the working principles of the DAB; Chapter 4 presents a loss model for the DAB, needed to develop the efficiency optimized modulation schemes presented in Chapter 5; in Chapter 6, a detailed small-signal control-oriented model of the DAB converter is presented, which characterizes the dynamic converter properties.

Due to the wide input and output voltage ranges, all isolated single-stage DC–DC converters listed above could be used in a two-stage arrangement according to Figure 2.22 in order to achieve a more efficient converter operation.

- The efficiencies obtained with the single-phase DAB converter, the threephase DAB converter, and the LLC converter are very sensitive on the ratio of the input to output voltage: for $V_1/V_2 \ll n$ or $V_1/V_2 \gg n$, large transformer RMS currents occur (cf. Figure 3.6 and Figure 3.21 for $V_1 = 450$ V and $V_2 = 11$ V) and thus, large conduction losses result; this can be avoided in a two-stage arrangement which keeps the input voltage of the isolated DC-DC converter, V_i , close to nV_2 .
- For the isolated full bridge converters with one or two DC inductors, the peak voltage applied to the semiconductor switches of the LV side full

bridge can be reduced if a two-stage arrangement is used; consequently, MOSFETs with a lower on-state resistance can be used. Moreover, the VA ratings of the switches of the HV side full bridge, T_1 , T_2 , T_3 , and T_4 in Figure 2.12, become smaller.

In order to compare the efficiencies achieved with the two-stage topologies to the efficiencies achieved with the single-stage DAB converter, i.e. the two most promising converter configurations, the arrangements depicted in Figure 2.22 (a) and Figure 2.22 (b) are investigated in Chapter 7; there, the isolated DC–DC converter comprises of a DAB converter. As a result, it turns out that the average efficiency achieved with the two-stage converter $(\bar{\eta} = 93.0\%)$ is less than the average efficiency achieved with the single-stage DAB ($\bar{\eta} = 93.5\%$),¹¹ i.e. in average, the additional losses of the buck-or-boost converter are higher than the loss reduction achieved for the DAB converter.

¹¹The respective average efficiencies, $\overline{\eta}$, are calculated according to Appendix A.1; the calculations assume the suboptimal modulation schemes to be used for the DAB, cf. Section 5.2.2.

Chapter 3

Steady-State Operation of the Dual Active Bridge (DAB)

3.1 Lossless DAB Model

In Section 2.3, the DAB converter topology is regarded most promising with respect to the achievable converter efficiency and with respect to a low converter volume. The DAB converter contains two voltage sourced full bridge circuits (cf. Section 2.2.1) that are connected to the inductor L and the high frequency transformer (Figure 3.1). In order to transfer power, time varying voltages $v_{AC1}(t)$ and $v_{AC2}(t)$ must be provided by the full bridge circuits to



Figure 3.1: Dual Active Bridge (DAB) converter topology.



Figure 3.2: Lossless DAB model.

both, the high frequency transformer and the converter inductor L. Thus, the voltage sourced HV and LV side full bridge circuits can be replaced by the respective voltage sources v_{AC1} and v_{AC2} to simplify the analysis of the DAB converter; for the most simple electric DAB converter model

- 1. all losses are neglected,
- 2. the transformer magnetizing inductance and parasitic capacitances (e.g. transformer coupling capacitance between LV and HV sides) are neglected,
- 3. all LV side quantities are referred to the HV side, and
- 4. constant supply voltages V_1 and V_2 are considered.

The resulting lossless DAB model is depicted in Figure 3.2.

3.1.1 DAB Inductor Current and Power Transfer

With the HV side full bridge, three different voltage levels are possible for $v_{AC1}(t)$,

$$v_{\rm AC1}(t) = \begin{cases} +V_1 & \text{for state I:} & T_1, T_4 \text{ on}, T_2, T_3 \text{ off} \\ & \text{for state II:} & T_1, T_3 \text{ on}, T_2, T_4 \text{ off or} \\ & \text{for state III:} & T_2, T_4 \text{ on}, T_1, T_3 \text{ off} \\ -V_1 & \text{for state IV:} & T_2, T_3 \text{ on}, T_1, T_4 \text{ off} \end{cases}$$
(3.1)

(on the assumption of an ideal full bridge, i.e. switching transients are not considered). Similarly, $v_{AC2}(t)$ is equal to V_2 , 0, or $-V_2$ depending on the switching states of T_5 , T_6 , T_7 , and T_8 . The resulting voltage $v_R(t)$,

$$v_{\rm R}(t) = v_{\rm AC1}(t) - n \, v_{\rm AC2}(t) \,,$$
(3.2)
applied to the inductor of the simple DAB model (Figure 3.2), generates the current

$$i_L(t_1) = i_L(t_0) + \frac{1}{L} \int_{t_0}^{t_1} v_{\rm R} \, \mathrm{d}t \quad \forall \quad t_0 < t_1, \tag{3.3}$$

at the time t_1 , starting with an initial current $i_L(t_0)$ at time t_0 . The voltage sources v_{AC1} and v_{AC2} thus generate or receive the respective instantaneous powers

$$p_1(t) = v_{AC1}(t) \cdot i_L(t)$$
 and $p_2(t) = n v_{AC2}(t) \cdot i_L(t)$. (3.4)

In order to simplify the calculation of (3.3), it is reasonable to select t_0 and t_1 such that $v_{AC1}(t)$ and $v_{AC2}(t)$ are constant during the time interval $t_0 < t < t_1$.

The average power over one switching cycle $T_{\rm S}$, $T_{\rm S} = 1/f_{\rm S}$, is finally calculated with

$$P_1 = \frac{1}{T_{\rm S}} \int_{t_0}^{t_0 + T_{\rm S}} p_1(t) \mathrm{d}t$$
(3.5)

for the HV side and

$$P_2 = \frac{1}{T_{\rm S}} \int_{t_0}^{t_0 + T_{\rm S}} p_2(t) \mathrm{d}t$$
 (3.6)

for the LV side. Therefore, one switching period, $t_0 < t < t_0 + T_S$, is split up into *m* time intervals (numbered with the index counter *i*) with constant voltages $v_{AC1}(t)$ and $v_{AC2}(t)$,

time interval I
$$(i = 1)$$
:
time interval II $(i = 2)$:
final time interval $(i = m)$:
 $t_m = 1 < t < t_m = t_0 + T_S$.
(3.7)

According to (3.4), (3.5), and (3.6) the waveforms $v_{AC1}(t)$ and $i_L(t)$ determine P_1 and the waveforms $v_{AC2}(t)$ and $i_L(t)$ determine P_2 ; further,

$$P_1 = P_2 \tag{3.8}$$

applies, since the considered DAB model is lossless. The power level of the DAB converter is thus typically adjusted using one or more out of the 4 control parameters depicted in Figure 3.3:

- the phase shift, φ , between $v_{AC1}(t)$ and $v_{AC2}(t)$ with $-\pi < \varphi < \pi$,
- the duty cycle, D_1 , of $v_{AC1}(t)$ with $0 < D_1 < 1/2$,
- the duty cycle, D_2 , of $v_{AC2}(t)$ with $0 < D_2 < 1/2$, and
- the switching frequency $f_{\rm S}$.



Figure 3.3: The 4 DAB control parameters: $\varphi = \varphi_2 - \varphi_1$ denotes the phase shift between v_{AC1} and v_{AC2} , D_1 and D_2 are the respective duty cycles, and f_S is the switching frequency.

It is important to note, that the average values of $v_{AC1}(t)$ and $v_{AC2}(t)$, evaluated over one switching period in steady-state converter operation, are zero (i.e. the DC components of $v_{AC1}(t)$ and $v_{AC2}(t)$ are zero) in order to avoid saturation of the high frequency transformer.

3.1.2 Phase Shift Modulation

The most common modulation principle, the so called phase shift modulation, operates the DAB with a constant switching frequency and with maximum duty cycles, $D_1 = D_2 = 1/2$; it solely varies the phase shift φ in order to control the transferred power. Hence, $v_{AC1}(t)$ is either $-V_1$ or $+V_1$ and $v_{AC2}(t)$ is either $-V_2$ or $+V_2$ (Figure 3.4). During steady-state operation, the voltages $v_{AC1}(t)$ and $v_{AC2}(t)$ and the inductor current repeat every half-cycle with reversed signs,

$$v_{AC1}(t + T_S/2) = -v_{AC1}(t),$$

$$v_{AC2}(t + T_S/2) = -v_{AC2}(t), \text{ and }$$

$$i_L(t + T_S/2) = -i_L(t),$$

(3.9)



Figure 3.4: Transformer voltages and inductor current for phase shift modulation, $V_1 = 240$ V, $V_2 = 11$ V, and $P_2 = 2$ kW (LV port output power); the converter is designed according to Appendix A.2 with n = 19 and $L = 26.7 \,\mu$ H.

since the phase shift time T_{φ} and the DC supply voltages V_1 and V_2 remain the same during the first and during the second half-cycle (time intervals I, II and III, IV in Figure 3.4, respectively). For the calculation of the transferred power, therefore, only the first half-cycle (intervals I and II) needs to be considered. With $t_0 = 0$, (3.5) we have

$$P_{1} = \frac{1}{T_{\rm S}} \int_{0}^{T_{\rm S}} p_{1}(t) \mathrm{d}t = \frac{2}{T_{\rm S}} \int_{0}^{T_{\rm S}/2} v_{\rm AC1}(t) \cdot i_{L}(t) \mathrm{d}t = \frac{2V_{1}}{T_{\rm S}} \int_{0}^{T_{\rm S}/2} i_{L}(t) \mathrm{d}t;$$
(3.10)

thus, in order to obtain an analytical expression for P_1 , the current $i_L(t)$ needs to be determined. In steady-state operation, a certain current $i_{L,0} = i_L(t_0)$ is presumed. The two different time intervals I and II occur during $0 < t < T_S/2$ (Figure 3.4). On the assumption of a positive phase shift, $0 < \varphi < \pi$, the resulting expressions for the inductor current are:

time interval I: $i_L(t) = i_{L,0} + (V_1 + n V_2) t/L$ $\forall 0 < t < T_{\varphi}$, time interval II: $i_L(t) = i_L(T_{\varphi}) + (V_1 - n V_2) (t - t_1)/L \ \forall T_{\varphi} < t < T_S/2$. (3.11) Due to the half-cycle symmetry (3.9) and with $T_{\varphi} = \varphi/(2\pi f_{\rm S})$,

$$i_{L,0} = \frac{\pi \cdot (n V_2 - V_1) - 2\varphi \, n \, V_2}{4\pi f_{\rm S} \, L} \tag{3.12}$$

results for positive phase shift, $0 < \varphi < \pi$, and a similar result is obtained for negative phase shift, $-\pi < \varphi < 0$. With (3.8) and (3.10), and by extending the results to the full phase shift range, $-\pi < \varphi < \pi$, the transferred power

$$P = P_1 = P_2 = \frac{n \cdot V_1 \cdot V_2 \cdot \varphi \cdot (\pi - |\varphi|)}{2\pi^2 f_{\rm S} L} \quad \forall \quad -\pi < \varphi < \pi \tag{3.13}$$

results, where

P > 0 denotes a power transfer from the HV to the LV port and

P < 0 denotes a power transfer from the LV to the HV port. (3.14)

The achieved power transfer for the DAB converter designed in Appendix A.2 is depicted in Figure 3.5 and shows a maximum for a certain phase shift angle; maximum power occurs for $\partial P/\partial \varphi = 0$ with the solution

$$|P_{\rm PS,max}| = \frac{n \cdot V_1 \cdot V_2}{8f_{\rm S}L} \quad \text{for} \quad \varphi = \pm \pi/2, \tag{3.15}$$

which presents an upper limit to the employed converter inductance L with respect to the specified output power.

The expression for the phase shift required to achieve a given power transfer is derived by rearranging (3.13):

$$\varphi = \frac{\pi}{2} \left[1 - \sqrt{1 - \frac{8f_{\rm S} L |P|}{n V_1 V_2}} \right] \, \text{sgn}(P) \ \forall \ |P| < |P_{\rm PS,max}| \,. \tag{3.16}$$

The great advantage of the phase shift modulation is its simplicity: according to (3.13), only a single control variable, the phase shift angle φ , is required to adjust the DAB power level (cf. Figure 3.5). The simplicity of this modulation method and the possibility of using half bridge circuits to generate the high frequency transformer voltages $v_{AC1}(t)$ and $v_{AC2}(t)$ are the main reasons for the wide use of this modulation method. Disadvantages are a limited operating range with low switching losses (soft switching range [50], cf. Section 4.3.2) and large RMS currents in the high frequency transformer for most operating points when the DAB is operated in wide voltage ranges (Figure 3.6). Even though, the calculated results depicted in Figure 3.6 are based on an efficiency optimized DAB design (cf. Appendix A.2), the lossless DAB converter model predicts LV side transformer RMS currents of up to 283 A for $P = 2 \,\text{kW}$, $V_1 = 450 \,\text{V}$, and $V_2 = 11 \,\text{V}$ [Figure 3.6 (d)]. Figures 3.6 (a) and (b) depict the transformer RMS currents for $P = 1 \,\text{kW}$:



Figure 3.5: Transferred power for the DAB designed in Appendix A.2 (n = 19, $L = 26.7 \,\mu\text{H}$), phase shift modulation, $V_1 = 340 \,\text{V}$ and different voltages V_2 . At $\varphi = \pm \pi/2$, maximum power transfer is achieved.

still currents of up to 256 A occur on the LV side. Merely for operating points close to $V_1 = n V_2$ [marked with dashed-and-dotted lines in Figures 3.6 (a) to (d)], effective transformer utilization is obtained.

With phase shift modulation, a reduced RMS value of the transformer current is only achieved with a different selection of n and L, since it is not possible to modify the shape of the transformer current;¹ the transformer current waveform solely depends on the DC voltages V_1 and V_2 and on the phase shift φ (for the assumption of a constant switching frequency $f_{\rm S}$).

3.1.3 Alternative Modulation Methods

The disadvantages of phase shift modulation give reason to investigate alternative modulation methods which take advantage of the employed full bridges (Figure 3.1) and not only adjust the phase shift between $v_{AC1}(t)$ and $v_{AC2}(t)$

¹The presented design with n = 19 and $L = 26.7 \,\mu\text{H}$ is optimized with respect to maximum average efficiency (Appendix A.1, Appendix A.2). For a different selection of n and L, the maximum RMS value of the LV side transformer current can be reduced; as a result, however, the maximum switching losses increase.



Figure 3.6: Transformer RMS currents for phase shift modulation and operation within the specified voltage ranges: (a) HV side and (b) LV side transformer currents for P = 1 kW, (c) HV side and (d) LV side transformer currents for P = 2 kW; n = 19, $L = 26.7 \mu\text{H}$.

but as well change the duty cycles of $v_{AC1}(t)$ and $v_{AC2}(t)$ (Figure 3.3) in order to reduce the circulating transformer current and to achieve low switching losses. Thus, for the alternative modulation schemes, the desired attributes are:

- 1. Minimum RMS inductor current I_L in order to achieve low conduction losses, since the conduction losses dominate the total losses within a wide operating range.
- Zero switch current during a switching process on the LV side: on the LV side, high currents lead to excessive switching losses due to the parasitic stray inductances of the packages of the switches (cf. Section 4.3.3, [89,93]).
- 3. Zero voltage switching (ZVS) operation of the HV side switches in order

to obtain low switching losses (Section 4.3.2, [85]).

First investigations on alternative modulation schemes are given in [94] for a bidirectional AC–DC converter: the proposed alternative modulation schemes extend the zero voltage switching (ZVS) range of the DAB and simultaneously reduce the transformer RMS currents (the principle of operation of ZVS is discussed in [43]). Detailed investigations of the modulation schemes discussed in [94] with either $D_1 \leq 0.5 \wedge D_2 = 0.5$ or $D_1 = 0.5 \wedge D_2 \leq 0.5$ are presented in [95,96]. The 1-D optimization problem (either D_1 or D_2 changes) with respect to maximum converter efficiency is solved in [95] (the analysis is based on the lossless electric DAB model [50] and an analytical converter loss model).

In a different approach, highly efficient operation of the DAB is reported with modulation schemes employing $D_1 \leq 0.5$ and $D_2 \leq 0.5$ [91, 97, 98, 99]. However, compared to the above mentioned 1-D problem, it is considerably more complex to solve the resulting 2-D problem (D_1 and D_2 change simultaneously) with respect to maximal efficiency [100]. Therefore, a more intuitive method is typically used to determine D_1 and D_2 as described e.g. in [91,97], where D_1 and D_2 are selected in order to achieve a triangular or trapezoidal shape of the transformer current which results in low switching losses and low conduction losses. Further efficiency improvements are reported with the use of combined modulation schemes (in [101,102], either the conventional phase shift modulation or modulation schemes with triangular or trapezoidal transformer currents are employed, depending on the actual output power and the port voltages V_1 and V_2).

With the alternative modulation schemes discussed in [97], i.e. the triangular and trapezoidal current mode modulation schemes, considerable efficiency improvements are expected; moreover, comparably little computational effort is required to calculate the respective modulation parameters D_1 , D_2 , and φ . These modulation schemes are therefore first revisited in this Section; subsequently, a systematic optimization of the modulation parameters D_1 and D_2 with respect to minimal transformer RMS current is presented for the lossless DAB (with given D_1 and D_2 the operating voltages V_1 and V_2 and the required output power determine φ).²

Triangular Current Mode Modulation

In [97], the triangular current mode modulation is introduced; this modulation scheme enables LV side switching at zero transformer current. Furthermore, compared to phase shift modulation, a considerable reduction of the transformer RMS current is achieved.

 $^{^2\}mathrm{Improved}$ and optimized modulation schemes considering the DAB converter losses are detailed in Chapter 5.



Figure 3.7: Transformer voltages and inductor current for triangular current mode modulation [97], $V_1 = 340$ V, $V_2 = 12$ V, P = 2 kW, n = 12, $L = 8.8 \mu$ H, and $f_S = 100$ kHz. In order to enable the operation of the DAB with the triangular current mode modulation within the operating range specified in Table 1.4, a transformer turns ratio n < 15 is required [cf. (3.29)]. A converter optimization with respect to maximum average efficiency (Appendix A.1) gives n = 12 and $L = 8.8 \mu$ H.

Figure 3.7 depicts typical voltage and current waveforms for the triangular current mode modulation, $V_1 > n V_2$, and P > 0 [power transfer from the HV port to the LV port, cf. (3.14)]. There, at t = 0, the inductor current, i_L , is zero and the DAB full bridges apply $v_{AC1}(t) = V_1$ and $v_{AC2}(t) = V_2$ to the HF transformer and the DAB inductor L. Due to $i_L(0) = 0$, low switching losses are expected for the respective switching operations.³ Subsequently, during $0 < t < t_1$, the inductor current increases according to

$$i_L(t) = 0 + \frac{V_1 - n V_2}{L} t \quad \forall \quad 0 < t < t_1.$$
(3.17)

At $t = t_1$, the state of the HV side full bridge changes with low switching losses, since the condition for ZVS is fulfilled (cf. Section 4.3.2), and during $t_1 < t < t_2$, the HV side full bridge generates $v_{AC1}(t) = 0$ while the LV side full

 $^{^{3}}$ In practice this is true for the LV side; however, considerable HV side switching losses occur due to the effects discussed in Section 4.3.2 (cf. Figure 4.14).

bridge state remains unchanged. Thus, the inductor current changes according to:

$$i_L(t) = i_L(t_1) - \frac{n V_2}{L} (t - t_1) \quad \forall \quad t_1 < t < t_2.$$
(3.18)

At the time t_2 the inductor current is zero and the LV side full bridge switches to $v_{AC2} = 0$. Consequently, the inductor current remains zero during $t_2 < t < T_S/2$,

$$i_L(t) = 0 \quad \forall \quad t_2 < t < T_S/2.$$
 (3.19)

The durations T_1 and T_2 of the respective time intervals I and II determine the transferred power P according to

$$P = \frac{n V_2}{T_{\rm S} L} \left[V_1 T_1 \cdot (2T_2 - T_1) - n V_2 T_2^2 \right] \quad \forall \quad V_1 > n V_2 ; \qquad (3.20)$$

moreover, T_2 depends on T_1 in order to achieve $i_L(t_2) = 0$:

$$T_2 = T_1 \frac{V_1 - n V_2}{n V_2} \,. \tag{3.21}$$

With

$$\varphi = 2\pi f_{\rm S} \left[(T_1 + T_2)/2 - T_1/2 \right] = \pi f_{\rm S} T_2 \tag{3.22}$$

(cf. Figure 3.3), (3.20), and (3.21), the transferred power is solely controlled with φ ,

$$P = \frac{\varphi^2 V_1 (n V_2)^2}{\pi^2 f_{\rm S} L (V_1 - n V_2)} \quad \forall \quad V_1 > n V_2 \land 0 < \varphi < \varphi_{\triangle, a, \max}, \qquad (3.23)$$

so we have for φ , T_1 , and T_2 :

$$\varphi = \pi \sqrt{f_{\rm S} \, L \, P \, \frac{V_1 - n \, V_2}{V_1 \, (n \, V_2)^2}} \quad \forall \quad V_1 > n \, V_2 \quad \land \quad 0 < \varphi < \varphi_{\triangle, \rm a, max}, \tag{3.24}$$

$$T_1 = \frac{\varphi}{\pi f_{\rm S}} \frac{n V_2}{V_1 - n V_2}$$
, and (3.25)

$$T_2 = \frac{\varphi}{\pi f_{\rm S}} \tag{3.26}$$

in order to achieve a given power level. However, the maximum phase shift angle $\varphi_{\triangle,a,\max}$ must not be exceeded, since the duration $T_3 = T_S/2 - T_1 - T_2$ of time interval III decreases with increasing power P due to (3.21), (3.22), and (3.24). The maximum power transfer is achieved for $T_3 = 0$ and is equal to

$$P_{\Delta,\mathrm{a,max}} = \frac{n^2 V_2^2 (V_1 - n V_2)}{4 f_{\mathrm{S}} L V_1}; \qquad (3.27)$$

the respective maximum phase shift angle is

$$\varphi_{\triangle, \mathbf{a}, \max} = \varphi(P_{\triangle, \mathbf{a}, \max}) = \frac{\pi}{2} \left(1 - \frac{n V_2}{V_1} \right).$$
(3.28)



Figure 3.8: Transformer voltages and inductor current for triangular current mode modulation [97], $V_1 = 340$ V, $V_2 = 12$ V, P = -2 kW, n = 12, $L = 8.8 \mu$ H, and $f_S = 100$ kHz.

Obviously, $P_{\Delta,a,\max}$ is zero for $V_1 = n V_2$ and negative for $V_1 < n V_2$. The presented modulation scheme only allows for a power transfer for $V_1 > n V_2$ and for positive power $0 < P < P_{\Delta,a,\max}$. The transformer turns ratio n is thus limited to

$$n < \frac{V_{1,\min}}{V_{2,\max}} = 15.$$
 (3.29)

A converter optimization with respect to maximum average efficiency (Appendix A.1) gives n = 12 and $L = 8.8 \,\mu\text{H}$.

For $V_1 > n V_2$, power transfer from the LV port to the HV port, $0 > P > -P_{\Delta,a,\max}$, is achieved with negative φ (i.e. swapped time intervals I and II in Figure 3.7, cf. Figure 3.8):

$$P = -\frac{\varphi^2 V_1 (n V_2)^2}{\pi^2 f_{\rm S} L (V_1 - n V_2)} \quad \forall \quad V_1 > n V_2 \land -\varphi_{\triangle, a, \max} < \varphi < 0.$$
(3.30)

With known transformer turns ratio n, (3.27) limits the maximum allowable

DAB inductance L according to

$$L < \min\left(\frac{n^2 V_{2,\min}^2 \left(V_{1,\min} - n V_{2,\min}\right)}{4 f_{\rm S} P_{\Delta,\mathrm{a,max}} V_{1,\min}}, \frac{n^2 V_{2,\max}^2 \left(V_{1,\min} - n V_{2,\max}\right)}{4 f_{\rm S} P_{\Delta,\mathrm{a,max}} V_{1,\min}}\right).$$
(3.31)

The resulting transformer RMS currents are shown in Figure 3.9 for n = 12, $L = 8.8 \,\mu\text{H}$, and $P = 1 \,\text{kW}$ [Figures 3.9 (a) and (b)] or $P = 2 \,\text{kW}$ [Figures 3.9 (c) and (d)]. For $P = 2 \,\text{kW}$, the lossless DAB model predicts a maximum transformer RMS current of 242 A on the LV side, which is 14% less than the maximum RMS current obtained for phase shift modulation (283 A). For $P = 1 \,\text{kW}$, the maximum RMS current decreases to 144 A (compared to 256 A for phase shift modulation). However, the usable transformer turns ratio, n, is limited according to (3.29), i.e. $V_1 > n V_2$ applies. Consequently, comparably high transformer RMS currents occur on the HV side, which considerably increase the HV side conduction losses (cf. Section 4.1); moreover, the DAB fails to operate in case of the abnormal operating condition $V_1 < n V_2$.

If LV side switching at zero inductor current is not mandatory, the operation with $V_1 < n V_2$ is easily achieved with a different triangular current mode modulation scheme, which employs the AC voltages $v_{AC1}(t)$ and $v_{AC2}(t)$ shown in Figures 3.10 and 3.11. The respective DAB power levels are:

$$P = \begin{cases} \frac{\varphi^2 V_1^2 n V_2}{\pi^2 f_{\rm S} L (n V_2 - V_1)} & \forall V_1 < n V_2 \land 0 < \varphi < \varphi_{\triangle, \rm b, max} \text{ and} \\ v_{\rm AC1}(t), v_{\rm AC2}(t) \text{ according to Figure 3.10,} \\ -\frac{\varphi^2 V_1^2 n V_2}{\pi^2 f_{\rm S} L (n V_2 - V_1)} & \forall V_1 < n V_2 \land -\varphi_{\triangle, \rm b, max} < \varphi < 0 \text{ and} \\ v_{\rm AC1}(t), v_{\rm AC2}(t) \text{ according to Figure 3.11,} \end{cases}$$
(3.32)

with the maximum phase shift angle

$$\varphi_{\Delta,\mathrm{b,max}} = \frac{\pi}{2} \left(1 - \frac{V_1}{n V_2} \right) \,. \tag{3.33}$$

Maximum power transfer is again achieved for $t_2 = T_S/2$ (cf. Figure 3.10); with this, the upper power limit for the combined triangular current mode modulation schemes (Figures 3.7, 3.8, 3.10, and 3.11) becomes:

$$P_{\triangle,\max} = \begin{cases} \frac{n^2 V_2^2 (V_1 - n V_2)}{4 f_{\rm S} L V_1} & \text{for } V_1 > n V_2 ,\\ 0 & \text{for } V_1 = n V_2 ,\\ \frac{V_1^2 (n V_2 - V_1)}{4 f_{\rm S} L n V_2} & \text{for } V_1 < n V_2 . \end{cases}$$
(3.34)



Figure 3.9: Transformer RMS currents for the triangular current mode modulation scheme and operation within the specified voltage ranges: (a) HV side and (b) LV side transformer currents for P = 1 kW, (c) HV side and (d) LV side transformer currents for P = 2 kW; n = 12, $L = 8.8 \mu$ H.

Trapezoidal Current Mode Modulation

With the triangular current mode modulation schemes, $|P| < P_{\Delta,\max}$ [cf. (3.34)] limits the DAB power level; this is particularly undesirable for $V_1 \approx n V_2$ where $P_{\Delta,\max}$ is either zero or close to zero. If switching at zero inductor current is not mandatory on the LV side, the operation of the DAB with $|P| > P_{\Delta,\max}$ is possible, e.g. with the trapezoidal current mode modulation scheme presented in [97]. Typical current and voltage waveforms are depicted in Figures 3.12 and 3.13. Since the inductor current is again zero at $t = k \cdot T_S/2, k \in \mathbb{Z}$, a seamless transition between trapezoidal and triangular current mode modulation is achieved.

For the trapezoidal current mode modulation, three different time intervals can be distinguished. During $0 < t < t_1$, $v_{AC1}(t) = V_1$ and $v_{AC2}(t) = 0$ apply and the absolute value of the inductor current, $|i_L(t)|$, increases from zero; at $t = t_1$, the voltage $v_{AC2}(t)$ changes to $v_{AC2}(t) = n V_2$ and remains constant



Figure 3.10: Transformer voltages and inductor current for triangular current mode modulation (cf. [97]) in case of $V_1 < n V_2$: $V_1 = 100 \text{ V}, V_2 = 12 \text{ V}, P = 500 \text{ W}, n = 12, L = 8.8 \,\mu\text{H}, \text{ and } f_{\text{S}} = 100 \,\text{kHz}.$

during $t_1 < t < t_2$; during $t_2 < t < T_S/2$, $v_{AC1}(t) = 0$ and $v_{AC2}(t) = V_2$ apply and $|i_L(t)|$ decreases back to zero.

The transferred power is equal to

$$P = \operatorname{sgn}(\varphi) \frac{nV_1V_2 \cdot \left[2nV_1V_2(\pi^2 - 2\varphi^2) - (V_1^2 + (nV_2)^2) \cdot (\pi - 2|\varphi|)^2\right]}{4\pi^2 f_{\rm S}L \cdot (V_1 + nV_2)^2} \forall \varphi_{\triangle, \max} < |\varphi| < \varphi_{\triangle, \max}$$
(3.35)

with

$$\varphi_{\Box,\max} = \frac{\pi}{2} \left(1 - \frac{n V_1 V_2}{V_1^2 + n V_1 V_2 + (n V_2)^2} \right).$$
(3.36)

The solutions for the required phase shift angle φ and the durations T_1 , T_2 , and T_3 are:

$$\varphi = \frac{\pi}{2} \cdot \frac{\operatorname{sgn}(P)}{V_1^2 + nV_1V_2 + (nV_2)^2} \cdot \left\{ V_1^2 + (nV_2)^2 - (V_1 + nV_2)\sqrt{nV_1V_2 \left[1 - \frac{4f_{\rm S}L|P|(V_1^2 + nV_1V_2 + (nV_2)^2)}{(nV_1V_2)^2} \right]} \right\}, \quad (3.37)$$



Figure 3.11: Transformer voltages and inductor current for triangular current mode modulation (cf. [97]) in case of $V_1 < n V_2$: $V_1 = 100 \text{ V}$, $V_2 = 12 \text{ V}$, P = -500 W, n = 12, $L = 8.8 \,\mu\text{H}$, and $f_{\text{S}} = 100 \,\text{kHz}$.

$$T_1 = \frac{n V_2 - V_1 + 2 V_1 \varphi/\pi}{2 f_{\rm S} \cdot (V_1 + n V_2)},\tag{3.38}$$

$$T_2 = \frac{1 - 2\,\varphi/\pi}{2\,f_{\rm S}},\tag{3.39}$$

$$T_3 = \frac{V_1 - n \, V_2 + 2 \, n \, V_2 \, \varphi/\pi}{2 \, f_{\rm S} \cdot (V_1 + n \, V_2)}.\tag{3.40}$$

The maximum transferable power is:

$$P_{\Box,\max} = \frac{(n \, V_1 \, V_2)^2}{4f_{\rm S}L \cdot (V_1^2 + n \, V_1 \, V_2 + (n \, V_2)^2)}.$$
(3.42)

Expression (3.42) determines the upper limit for L:

$$L < \frac{(n V_{1,\min} V_{2,\min})^2}{4f_{\rm S} P_{\Box,\max} \cdot (V_{1,\min}^2 + n V_{1,\min} V_{2,\min} + (n V_{2,\min})^2)}.$$
 (3.43)

The RMS values of the HV side and LV side transformer currents predicted with the lossless DAB model are shown in Figure 3.14 for the alternative



Figure 3.12: Transformer voltages and inductor current for trapezoidal current mode modulation [97], $V_1 = 240$ V, $V_2 = 12$ V, P = 2 kW, n = 19, $L = 18.7 \,\mu\text{H}$, and $f_{\text{S}} = 100 \,\text{kHz}$ [n = 19 is selected in order to allow for a direct comparison with the results obtained for phase shift modulation and $L = 18.7 \,\mu\text{H}$ is required in order to achieve the specified power transfer, cf. (3.43)].

modulation scheme. In order to allow for a comparison with the RMS currents obtained for phase shift modulation (Figure 3.6), n = 19 was selected, which requires a converter inductance $L = 18.7 \,\mu\text{H}$.

Depending on the actual power level P and the operating voltages V_1 and V_2 , either the trapezoidal current mode modulation or the triangular current mode modulation is used; with (3.34) and (3.42), the appropriate modulation scheme is selected:

$$0 < |P| < P_{\Delta,\max}(V_1, V_2):$$
triangular current mode,
$$P_{\Delta,\max}(V_1, V_2) < |P| < P_{\Delta,\max}(V_1, V_2):$$
trapezoidal current mode. (3.44)

In Figure 3.14, the borders between triangular and trapezoidal current mode modulations are marked with dashed lines.

With the alternative modulation scheme (the combined trapezoidal and triangular current mode modulation), the maximum transformer RMS current (235 A) is only slightly below the maximum RMS current obtained for exclusive triangular current mode modulation (242 A). However, on the HV side, a



Figure 3.13: Transformer voltages and inductor current for trapezoidal current mode modulation [97], $V_1 = 240$ V, $V_2 = 12$ V, P = -2 kW, n = 19, $L = 18.7 \,\mu$ H, and $f_S = 100$ kHz.

considerably lower maximum RMS current of 12.4 A is obtained (as opposed to 20.1 A for the exclusive triangular current mode modulation). Compared to phase shift modulation, lower maximum RMS currents are achieved (phase shift modulation: HV side: 14.9 A, LV side: 283 A). For P = 1 kW, the maximum transformer RMS currents decrease to 7.3 A on the HV side and to 140 A on the LV side with the alternative modulation scheme. Thus, compared to the phase shift modulation (Figure 3.6) or to the exclusive triangular current mode modulation (Figure 3.9), a considerable improvement is achieved.

Optimal Modulation

For the given DAB converter, the trapezoidal and the triangular current mode modulation schemes have been intuitively selected due to the low expected conduction and switching losses. A more systematic investigation is required in order to discover the modulation scheme that leads to the lowest value of the RMS current I_L (and thus to the lowest conduction losses, cf. Chapter 4), the lowest switching losses, and / or the lowest total losses. In this section, the discussion focuses on the minimization of I_L .

With the two full bridge circuits, 12 basic voltage sequences can be dis-



Figure 3.14: Transformer RMS currents for the trapezoidal and triangular current mode modulation schemes and operation within the specified voltage

current mode modulation schemes and operation within the specified voltage ranges: (a) HV side and (b) LV side transformer currents for P = 1 kW, (c) HV side and (d) LV side transformer currents for P = 2 kW; n = 19, $L = 18.7 \,\mu\text{H}$.

tinguished with respect to the different sequences of rising and falling edges of v_{AC1} and v_{AC2} (Figure 3.15). The search towards the modulation scheme with minimum I_L , though, only considers the 6 sequences 1a, 1b, 2, 3b, 7b, and 8; the remaining 6 options lead to an increased RMS value of the inductor current I_L and do not result in a higher DAB power level (cf. Table 3.1 and Figure 3.15).

For the presented analysis, a set of 5 input parameters is required, which consists of V_1 , V_2 , P, D_1 , and D_2 . The optimization method further requires a certain mode discrimination (Table 3.2), since the expressions for the particular DAB power levels (Table 3.3), the phase shift angles (Table 3.4), and the inductor RMS currents (Table 3.5) are substantially different for the different

Sequence Number	D_1	D_2	φ	I_L	$P_1 = P_2$
1a	0.1	0.25	0°	$3.5\mathrm{A}$	$0\mathrm{W}$
2	0.1	0.25	45°	$6.9\mathrm{A}$	$653\mathrm{W}$
3a	0.1	0.25	90°	$10.8\mathrm{A}$	$726\mathrm{W}$
4	0.1	0.25	135°	$13.5\mathrm{A}$	$653\mathrm{W}$
5a	0.1	0.25	180°	$14.7\mathrm{A}$	$0\mathrm{W}$
6	0.1	0.25	-135°	$13.5\mathrm{A}$	$-653\mathrm{W}$
7a	0.1	0.25	-90°	$10.8\mathrm{A}$	$-726\mathrm{W}$
8	0.1	0.25	-45°	$6.9\mathrm{A}$	$-653\mathrm{W}$
1b	0.25	0.1	0°	$9.3\mathrm{A}$	$0\mathrm{W}$
$5\mathrm{b}$	0.25	0.1	180°	$17.0\mathrm{A}$	$0\mathrm{W}$
3b	0.25	0.4	90°	$17.7\mathrm{A}$	$2.58\mathrm{kW}$
7b	0.25	0.4	-90°	$17.7\mathrm{A}$	$-2.58\rm kW$

Table 3.1: Resulting inductor RMS currents and power levels for the different voltage sequences depicted in Figure 3.15 ($V_1 = 340$ V, $V_2 = 12$ V, $L = 26.7 \mu$ H, n = 19, and $f_S = 100$ kHz).

voltage sequences depicted in Figure 3.15.

Table 3.2 summarizes the conditions that must be fulfilled for the respective voltage sequence and Figure 3.16 shows the active sequences for $V_1 = 340$ V, $V_2 = 12$ V, different duty cycles, and different power levels. With the preselected 6 sequences and with a specified set of input parameters only a single valid sequence number occurs—ambiguities are avoided.

The optimization procedure then minimizes the cost function, e.g. the value of the RMS current I_L , with respect to D_1 and D_2 for a fixed operating point (Figure 3.17) in order to determine the optimal modulation scheme.

(The text continues on page 81.)



Figure 3.15: The 12 basic voltage sequences generated with the two full bridges (continued on the next page).



Figure 3.15 cont.: The 12 basic voltage sequences generated with the two full bridges. The respective transformer RMS currents and DAB power levels are given in Table 3.1; $V_1 = 340$ V, $V_2 = 12$ V, $L = 26.7 \,\mu\text{H}$, n = 19, and $f_S = 100 \,\text{kHz}$; t_0 denotes the time the steady-state inductor current expressions in Table 3.6 are given for. Again, $L = 26.7 \,\mu\text{H}$ and n = 19 are selected in order to allow for a straight comparison with the results obtained for phase shift modulation.

Mode	Condition
1a	$D_1 - D_2 < \varphi/\pi < -D_1 + D_2$
1b	$-D_1 + D_2 < \varphi/\pi < D_1 - D_2$
2	$ D_1 - D_2 < \varphi/\pi < \min[D_1 + D_2, 1 - (D_1 + D_2)]$
8	$ D_1 - D_2 < -\varphi/\pi < \min[D_1 + D_2, 1 - (D_1 + D_2)]$
3b	$1 - (D_1 + D_2) < \varphi/\pi < D_1 + D_2$
7b	$1 - (D_1 + D_2) < -\varphi/\pi < D_1 + D_2$

Table 3.2: Conditions required in order to enable the respective sequences.

Mode	DAB power level
All	$P = P_1 = P_2 = \frac{nV_1V_2}{f_{\rm S}L} e_P$
1a	$e_P = D_1 \frac{\varphi}{\pi}$
$1\mathrm{b}$	$e_P = D_2 \frac{\varphi}{\pi}$
2	$e_P = -\frac{1}{4} \left[\frac{\varphi^2}{\pi^2} - 2\frac{\varphi}{\pi} \left(D_1 + D_2 \right) + \left(D_1 - D_2 \right)^2 \right]$
8	$e_P = \frac{1}{4} \left[\frac{\varphi^2}{\pi^2} + 2\frac{\varphi}{\pi} \left(D_1 + D_2 \right) + \left(D_1 - D_2 \right)^2 \right]$
3b	$e_P = -\frac{1}{2} \left\{ \frac{\varphi^2}{\pi^2} - \frac{\varphi}{\pi} + \left[\frac{1}{2} - D_1 \left(1 - D_1 \right) - D_2 \left(1 - D_2 \right) \right] \right\}$
7b	$e_P = \frac{1}{2} \left\{ \frac{\varphi^2}{\pi^2} + \frac{\varphi}{\pi} + \left[\frac{1}{2} - D_1 \left(1 - D_1 \right) - D_2 \left(1 - D_2 \right) \right] \right\}$

Table 3.3: Expressions for the DAB power level P for the considered voltage sequences.



Figure 3.16: The employed voltage sequences, plotted against D_1 and D_2 , for fixed operating voltages ($V_1 = 340$ V, $V_2 = 12$ V) and 4 different power levels; the required power transfer cannot be accomplished if D_1 and D_2 fall below a certain limit. DAB parameters: n = 19, $L = 26.7 \,\mu\text{H}$, $f_{\text{S}} = 100 \,\text{kHz}$.



Figure 3.17: The resulting inductor RMS currents, I_L , for fixed operating voltages ($V_1 = 340$ V, $V_2 = 12$ V), 2 different power levels, and different combinations of D_1 and D_2 ; I_{opt} denotes the minimum RMS current. Employed DAB converter parameters: n = 19, $L = 26.7 \,\mu\text{H}$, $f_{\text{S}} = 100 \,\text{kHz}$.

Mode	φ
1a	$\varphi = \frac{\pi L f_{\rm S} P}{n D_1 V_1 V_2}$
$1\mathrm{b}$	$\varphi = \frac{\pi L f_{\rm S} P}{n D_2 V_1 V_2}$
2, 8	$\varphi = \pi \operatorname{sgn}(P) \cdot \left(D_1 + D_2 - 2\sqrt{D_1 D_2 - \frac{L P f_S}{nV_1 V_2}} \right)$
3b, 7b	$\varphi = \pi \operatorname{sgn}(P) \cdot \left[\frac{1}{2} - \sqrt{D_1 (1 - D_1) + D_2 (1 - D_2) - \frac{1}{4} - \frac{2f_{\mathrm{S}}L P }{nV_1V_2}} \right]$

Table 3.4: Required phase shift angle φ for a given power $P = P_1 = P_2$.

$$\begin{array}{|c|c|c|c|c|} \hline Mode & Inductor RMS \ current \ I_L \\ \hline \text{All} & I_L = \frac{1}{2f_{\text{S}}L} \sqrt{D_1^2 V_1^2 \left(1 - \frac{4}{3}D_1\right) + D_2^2 n^2 V_2^2 \left(1 - \frac{4}{3}D_2\right) + \frac{nV_1 V_2}{3} e_{\text{RMS}}} \\ \hline \text{Ia} & e_{\text{RMS}} = 6D_1 \left[\frac{\varphi^2}{\pi^2} + \left(D_2 \left(D_2 - 1\right) + \frac{D_1^2}{3}\right)\right] \\ \hline \text{Ib} & e_{\text{RMS}} = 6D_2 \left[\frac{\varphi^2}{\pi^2} + \left(D_1 \left(D_1 - 1\right) + \frac{D_2^2}{3}\right)\right] \\ 2 & e_{\text{RMS}} = D_1^3 + 3D_1^2 \left(D_2 - \frac{\varphi}{\pi}\right) + 3D_1 \left[\frac{\varphi^2}{\pi^2} - D_2 \left(2 - \frac{2\varphi}{\pi} - D_2\right)\right] + \\ & \left(D_2 - \frac{\varphi}{\pi}\right)^3 \\ 8 & e_{\text{RMS}} = D_1^3 + 3D_1^2 \left(D_2 + \frac{\varphi}{\pi}\right) + 3D_1 \left[\frac{\varphi^2}{\pi^2} - D_2 \left(2 + \frac{2\varphi}{\pi} - D_2\right)\right] + \\ & \left(D_2 + \frac{\varphi}{\pi}\right)^3 \\ 3b & e_{\text{RMS}} = \left(1 - \frac{2\varphi}{\pi}\right) \left\{1 - \frac{\varphi}{\pi} + \frac{\varphi^2}{\pi^2} - 3\left[D_1 \left(1 - D_1\right) + D_2 \left(1 - D_2\right)\right]\right\} \\ \hline \text{7b} & e_{\text{RMS}} = \left(1 + \frac{2\varphi}{\pi}\right) \left\{1 + \frac{\varphi}{\pi} + \frac{\varphi^2}{\pi^2} - 3\left[D_1 \left(1 - D_1\right) + D_2 \left(1 - D_2\right)\right]\right\} \end{array}$$

Table 3.5: Expressions for the inductor RMS current I_L for the considered voltage sequences.

Mode	Inductor current at $t = t_0$
1a, 8, 7b	$i_L(t_0) = \frac{1}{2Lf_S} \left[nV_2 \left(\frac{\varphi}{\pi} + D_1\right) - D_1 V_1 \right]$
1b, 2	$i_L(t_0) = \frac{1}{2Lf_{\rm S}} \left(nD_2V_2 - D_1V_1 \right)$
3b	$i_L(t_0) = \frac{1}{2Lf_S} \left[nV_2 \left(1 - \frac{\varphi}{\pi} \right) - D_1 \left(V_1 + nV_2 \right) \right]$

Table 3.6: Steady-state inductor current $i_L(t_0)$ (at t_0 , v_{AC1} changes from 0 to V_1 , cf. Figure 3.15).

The optimization with respect to minimum I_L suggests the triangular current mode modulation for $|P| < P_{\triangle,\max}$ [cf. (3.34)]. For $|P| > P_{\triangle,\max}$, however, a modulation scheme different to the trapezoidal current mode modulation results: the optimization identifies the voltage sequence 3b for $P > P_{\triangle,\max}$ (Figure 3.18) and 7b for $P < -P_{\triangle,\max}$ (Figure 3.19) as optimal. The respective values of the optimal duty cycles depend on the ratio $V_1/(nV_2)$:

 $\begin{array}{ll} 1. \quad V_{1} = n \, V_{2} \\ 2. \quad V_{1} > n \, V_{2} \ \land \ |P| > P_{\mathrm{a,max}} \\ 3. \quad V_{1} < n \, V_{2} \ \land \ |P| > P_{\mathrm{b,max}} \end{array} \right\} \text{ phase shift modulation: } \begin{cases} D_{1,\mathrm{opt}} = 1/2, \\ D_{2,\mathrm{opt}} = 1/2; \end{cases} \\ 4. \quad V_{1} > n \, V_{2} \ \land \ P_{\Delta,\mathrm{max}} < |P| < P_{\mathrm{a,max}}: \text{ optimal transition mode;} \\ D_{1,\mathrm{opt}} \text{ is calculated with } (3.45), \\ D_{2,\mathrm{opt}} = 1/2, \end{cases} \\ 5. \quad V_{1} < n \, V_{2} \ \land \ P_{\Delta,\mathrm{max}} < |P| < P_{\mathrm{b,max}}: \text{ optimal transition mode;} \\ D_{1,\mathrm{opt}} = 1/2, \\ 5. \quad V_{1} < n \, V_{2} \ \land \ P_{\Delta,\mathrm{max}} < |P| < P_{\mathrm{b,max}}: \text{ optimal transition mode;} \\ D_{1,\mathrm{opt}} = 1/2, \\ D_{2,\mathrm{opt}} = 1/2, \\ D_{2,\mathrm{opt}} \text{ is calculated with } (3.49) \end{array}$

 $[P_{a,max} \text{ and } P_{b,max} \text{ are defined with (3.47) and (3.50), respectively}].$

For $P_{\triangle,\max} < |P| < P_{a,\max}$, a complicated expression results for $D_{1,\text{opt}}$ $(e_{a,1} \dots e_{a,7}$ denote interim values):

$$\begin{split} e_{\mathrm{a},1} &= \frac{V_1^2}{V_1^2 + n^2 V_2^2} - 2, \\ e_{\mathrm{a},2} &= \frac{n^2 V_2^2}{V_1^2 + n^2 V_2^2} + \frac{2L|P|f_{\mathrm{S}}}{nV_1 V_2}, \\ e_{\mathrm{a},3} &= \frac{V_1^3}{f_{\mathrm{S}}^6} \left[2V_1^3 n^9 V_2^9 + 3L^2 P^2 f_{\mathrm{S}}^2 V_1 n^3 V_2^3 \left(V_1^2 + 4n^2 V_2^2 \right) \left(5V_1^2 + 8n^2 V_2^2 \right) + \\ &\quad 3L|P|f_{\mathrm{S}} V_1^2 nV_2 \left(-8n^7 V_2^7 - 2V_1^2 n^5 V_2^5 + \\ &\quad \sqrt{3} \left\{ L|P|f_{\mathrm{S}} \left[4V_1^5 n^7 V_2^7 - 256L^3 |P|^3 f_{\mathrm{S}}^3 \left(V_1^2 + n^2 V_2^2 \right)^3 - \\ &\quad L|P|f_{\mathrm{S}} V_1^2 n^4 V_2^4 \left(V_1^2 + 4n^2 V_2^2 \right) \left(13V_1^2 + 4n^2 V_2^2 \right) + \\ &\quad 32L^2 P^2 f_{\mathrm{S}}^2 nV_1 V_2 \left(V_1^2 + n^2 V_2^2 \right)^2 \left(V_1^2 + 4n^2 V_2^2 \right) \right] \right\}^{\frac{1}{2}} \right) + \\ &\quad 16L^3 |P|^3 f_{\mathrm{S}}^3 \left(V_1^2 + n^2 V_2^2 \right)^2 \left(V_1^2 - 8n^2 V_2^2 \right) \right], \\ e_{\mathrm{a},4} &= \frac{\sqrt[3]{2}}{3\sqrt[3]{e_{\mathrm{a},3}} f_{\mathrm{S}}^2 nV_2 \left(V_1^2 + n^2 V_2^2 \right)} \left[4L^2 P^2 f_{\mathrm{S}}^2 \left(V_1^2 + n^2 V_2^2 \right) \left(V_1^2 + 4n^2 V_2^2 \right) + \\ &\quad V_1^2 n^6 V_2^6 - 2L|P|f_{\mathrm{S}} V_1 n^3 V_2^3 \left(V_1^2 + 4n^2 V_2^2 \right) \right], \end{aligned}$$



Figure 3.18: Voltage and current waveforms for minimum RMS current I_L , $V_1 = 340$ V, $V_2 = 12$ V, and P = 2 kW (corresponds to voltage sequence 3b in Figure 3.15); DAB parameters: n = 19, $L = 26.7 \mu$ H, $f_S = 100$ kHz.

$$\begin{split} e_{\mathrm{a},5} &= -\frac{4\left[n^{3}V_{2}^{3}\left(V_{1}^{2}+2n^{2}V_{2}^{2}\right)+2L|P|f_{\mathrm{S}}V_{1}\left(V_{1}^{2}+n^{2}V_{2}^{2}\right)\right]}{nV_{2}\left(V_{1}^{2}+n^{2}V_{2}^{2}\right)^{2}}\\ e_{\mathrm{a},6} &= \frac{f_{\mathrm{S}}^{2}\sqrt[3]{e_{\mathrm{a},3}}}{3\sqrt[3]{2}\left(nV_{2}V_{1}^{4}+n^{3}V_{2}^{3}V_{1}^{2}\right)}+\frac{e_{\mathrm{a},1}^{2}}{4}+e_{\mathrm{a},4}-\frac{2e_{\mathrm{a},2}}{3},\\ e_{\mathrm{a},7} &= \frac{1}{4}\left(3e_{\mathrm{a},1}^{2}-8e_{\mathrm{a},2}-4e_{\mathrm{a},6}+\frac{e_{\mathrm{a},5}-e_{\mathrm{a},1}^{3}}{\sqrt{e_{\mathrm{a},6}}}\right), \end{split}$$

$$D_{1,\text{opt}} = \frac{\frac{V_1^2 + 2n^2 V_2^2}{V_1^2 + n^2 V_2^2} + 2\sqrt{e_{a,6}} - 2\sqrt{e_{a,7}}}{4}$$

$$\forall P_{\triangle,\text{max}} < |P| < P_{a,\text{max}} \land V_1 > n V_2, \qquad (3.45)$$

$$D_{2,\text{opt}} = \frac{1}{2} \quad \forall \quad P_{\triangle,\max} < |P| < P_{a,\max} \land V_1 > nV_2, \tag{3.46}$$

$$P_{\rm a,max}: D_{\rm 1,opt}(P_{\rm a,max}) = \frac{1}{2} \land P_{\triangle,max} < P_{\rm a,max} < P_{\rm PS,max}.$$
(3.47)



Figure 3.19: Voltage and current waveforms for minimum RMS current I_L , $V_1 = 340$ V, $V_2 = 12$ V, and P = -2 kW (corresponds to voltage sequence 7b in Figure 3.15); DAB parameters: n = 19, $L = 26.7 \mu$ H, $f_S = 100$ kHz.

$$\begin{aligned} & \text{For } P_{\Delta,\max} < |P| < P_{\text{b,max}}, \, D_{1,\text{opt}} \text{ and } D_{2,\text{opt}} \text{ are:} \\ & e_{\text{b},1} = \frac{n^2 V_2^2}{V_1^2 + n^2 V_2^2} - 2, \\ & e_{\text{b},2} = \frac{V_1^2}{V_1^2 + n^2 V_2^2} + \frac{2L|P|f_{\text{S}}}{nV_1 V_2}, \\ & e_{\text{b},3} = \frac{n^3 V_2^3}{f_{\text{S}}^6} \left[2n^3 V_2^3 V_1^9 + 3L^2 P^2 f_{\text{S}}^2 n V_2 V_1^3 \left(4V_1^2 + n^2 V_2^2 \right) \left(8V_1^2 + 5n^2 V_2^2 \right) + \\ & 3L|P|f_{\text{S}} n^2 V_2^2 V_1 \left(-8V_1^7 - 2n^2 V_2^2 V_1^5 + \\ & \sqrt{3} \left\{ L|P|f_{\text{S}} \left[4n^5 V_2^5 V_1^7 - 256L^3 |P|^3 f_{\text{S}}^3 \left(V_1^2 + n^2 V_2^2 \right)^3 - \\ & L|P|f_{\text{S}} n^2 V_2^2 V_1^4 \left(4V_1^2 + n^2 V_2^2 \right) \left(4V_1^2 + 13n^2 V_2^2 \right) + \\ & 32L^2 P^2 f_{\text{S}}^2 n V_1 V_2 \left(V_1^2 + n^2 V_2^2 \right)^2 \left(4V_1^2 + n^2 V_2^2 \right) \right] \right\}^{\frac{1}{2}} \right) + \\ & 16L^3 |P|^3 f_{\text{S}}^3 \left(V_1^2 + n^2 V_2^2 \right)^2 \left(n^2 V_2^2 - 8V_1^2 \right) \right], \\ & e_{\text{b},4} = \frac{\sqrt[3]{2}}{3\sqrt[3]{e_{\text{b},3}} f_{\text{S}}^2 V_1 \left(V_1^2 + n^2 V_2^2 \right)} \left[4L^2 P^2 f_{\text{S}}^2 \left(V_1^2 + n^2 V_2^2 \right) \left(4V_1^2 + n^2 V_2^2 \right) + \\ \end{aligned}$$

$$n^{2}V_{2}^{2}V_{1}^{6} - 2L|P|f_{\rm S}nV_{2}V_{1}^{3}\left(4V_{1}^{2} + n^{2}V_{2}^{2}\right)],$$

$$e_{\rm b,5} = -\frac{4\left[\left(2V_{1}^{2} + n^{2}V_{2}^{2}\right)V_{1}^{3} + 2Ln|P|f_{\rm S}V_{2}\left(V_{1}^{2} + n^{2}V_{2}^{2}\right)\right]}{V_{1}\left(V_{1}^{2} + n^{2}V_{2}^{2}\right)^{2}},$$

$$e_{\rm b,6} = \frac{f_{\rm S}^{2}\sqrt[3]{e_{\rm b,3}}}{\sqrt[3]{2}\left(3n^{4}V_{1}V_{2}^{4} + 3n^{2}V_{1}^{3}V_{2}^{2}\right)} + \frac{e_{\rm b,1}^{2}}{4} + e_{\rm b,4} - \frac{2e_{\rm b,2}}{3},$$

$$e_{\rm b,7} = \frac{1}{4}\left(3e_{\rm b,1}^{2} - 8e_{\rm b,2} - 4e_{\rm b,6} + \frac{e_{\rm b,5} - e_{\rm b,1}^{3}}{\sqrt{e_{\rm b,6}}}\right),$$

$$D_{\rm 1,opt} = \frac{1}{2} \quad \forall \quad P_{\Delta,\max} < |P| < P_{\rm b,max} \land V_{1} < nV_{2},$$

$$(3.48)$$

$$D_{2,\text{opt}} = \frac{\frac{2V_1^2 + n^2 V_2^2}{V_1^2 + n^2 V_2^2} + 2\sqrt{e_{\text{b},6}} - 2\sqrt{e_{\text{b},7}}}{4}$$
$$\forall P_{\triangle,\text{max}} < |P| < P_{\text{b},\text{max}} \land V_1 < nV_2, \quad (3.49)$$

$$P_{\rm b,max}: D_{2,\rm opt}(P_{\rm b,max}) = \frac{1}{2} \land P_{\Delta,\rm max} < P_{\rm b,max} < P_{\rm PS,max};$$
 (3.50)

Figure 3.20 depicts the resulting duty cycles $D_{1,\text{opt}}$ and $D_{2,\text{opt}}$ for $V_1 = 340 \text{ V}$, $V_2 = 12 \text{ V}$, and $0 < P < P_{\text{PS,max}}$ [cf. (3.15)]. With given $D_1 = D_{1,\text{opt}}$ and $D_2 = D_{2,\text{opt}}$, the required phase shift angle φ is calculated according to Table 3.4.

The upper limit for L is determined with (3.15), since phase shift modulation is employed for maximum power.

The transformer RMS currents obtained with the optimized modulation scheme are depicted in Figure 3.21 for the DAB with n = 19 and $L = 26.7 \,\mu\text{H}$, i.e. the DAB designed for phase shift modulation, Appendix A.2, in order to allow for a straight comparison with the results obtained for phase shift modulation. At $P = 2 \,\text{kW}$, the LV side transformer RMS current is between 113 A (at $V_1 = 335 \,\text{V}$, $V_2 = 16 \,\text{V}$) and 215 A ($V_1 = 450 \,\text{V}$, $V_2 = 11 \,\text{V}$); at $P = 1 \,\text{kW}$, the LV side transformer RMS current is between 65 A ($V_1 = 311 \,\text{V}$, $V_2 = 16 \,\text{V}$) and 128 A ($V_1 = 450 \,\text{V}$, $V_2 = 11 \,\text{V}$). Compared to phase shift modulation, the maximum RMS current is 24% lower (phase shift modulation: 283 A) and the respective maximum conduction losses are 42% lower. Compared to the combined triangular and trapezoidal current mode modulation scheme (maximum RMS current: 235 A), the maximum RMS current is 8.5% lower and, accordingly, 16% lower maximum conduction losses result. Table 3.7 lists the minimum and maximum transformer RMS current values, I_L , for the discussed modulation schemes.



Figure 3.20: Duty cycles D_1 and D_2 , which minimize the inductor RMS current for $V_1 = 340$ V and $V_2 = 12$ V: at low power levels, the triangular current mode modulation is optimal, at medium power levels, D_1 and D_2 are calculated according to (3.45) and (3.46), and at high power levels, phase shift modulation is optimal. Employed DAB parameters: n = 19, $L = 26.7 \,\mu\text{H}$, $f_{\rm S} = 100 \,\text{kHz}$.



Boundary between optimal transition mode modulation (opt.) and phase shift modulation (PS)

--- Boundary between optimal transition mode modulation (opt.) and triangular current mode modulation (tria.)

Figure 3.21: Transformer RMS currents for the modulation scheme with minimum inductor RMS current and operation within the specified voltage ranges: (a) HV side and (b) LV side transformer currents for P = 1 kW, (c) HV side and (d) LV side transformer currents for P = 2 kW; n = 19, $L = 26.7 \,\mu\text{H}$, $f_{\text{S}} = 100 \,\text{kHz}$.

	I_L	$n I_L$	Respect	ive ope	rating	conditions
			V_1	V_2	n	L
Phase shift modulation						
min. I_L for $P = 1 \mathrm{kW}$	$3.4\mathrm{A}$	$65\mathrm{A}$	$311\mathrm{V}$	$16\mathrm{V}$	19	$26.7\mu\mathrm{H}$
min. I_L for $P = 2 \mathrm{kW}$	$7.0\mathrm{A}$	$134\mathrm{A}$	$333\mathrm{V}$	$16\mathrm{V}$	19	$26.7 \mu H$
max. I_L for $P = 1 \mathrm{kW}$	$13.5\mathrm{A}$	$256\mathrm{A}$	$450\mathrm{V}$	$11\mathrm{V}$	19	$26.7\mu\mathrm{H}$
max. I_L for $P = 2 \mathrm{kW}$	$14.9\mathrm{A}$	$283\mathrm{A}$	$450\mathrm{V}$	$11\mathrm{V}$	19	$26.7 \ \mu H$
Solely triangular current mode modulation						
min. I_L for $P = 1 \mathrm{kW}$	$7.2\mathrm{A}$	87 A	$240\mathrm{V}$	$16\mathrm{V}$	12	$8.8\mu\mathrm{H}$
min. I_L for $P = 2 \mathrm{kW}$	$12.2\mathrm{A}$	$146\mathrm{A}$	$240\mathrm{V}$	$16\mathrm{V}$	12	$8.8\mu\mathrm{H}$
max. I_L for $P = 1 \mathrm{kW}$	$12.0\mathrm{A}$	$144\mathrm{A}$	$450\mathrm{V}$	$11\mathrm{V}$	12	$8.8\mu\mathrm{H}$
max. I_L for $P = 2 \mathrm{kW}$	$20.1\mathrm{A}$	$242\mathrm{A}$	$450\mathrm{V}$	$11\mathrm{V}$	12	$8.8\mu\mathrm{H}$
Combined triangular and trapezoidal current mode modulation						
min. I_L for $P = 1 \mathrm{kW}$	$3.4\mathrm{A}$	$65\mathrm{A}$	$308\mathrm{V}$	$16\mathrm{V}$	19	$18.7 \mu \mathrm{H}$
min. I_L for $P = 2 \mathrm{kW}$	$7.0\mathrm{A}$	$134\mathrm{A}$	$323\mathrm{V}$	$16\mathrm{V}$	19	$18.7 \mu \mathrm{H}$
max. I_L for $P = 1 \mathrm{kW}$	$7.3\mathrm{A}$	$140\mathrm{A}$	$450\mathrm{V}$	$11\mathrm{V}$	19	$18.7 \mu \mathrm{H}$
max. I_L for $P = 2 \mathrm{kW}$	$12.4\mathrm{A}$	$235\mathrm{A}$	$450\mathrm{V}$	$11\mathrm{V}$	19	$18.7 \mu \mathrm{H}$
Modulation scheme with minimum transformer and inductor RMS current						
min. I_L for $P = 1 \mathrm{kW}$	$3.4\mathrm{A}$	$65\mathrm{A}$	$311\mathrm{V}$	$16\mathrm{V}$	19	$26.7\mu\mathrm{H}$
min. I_L for $P = 2 \mathrm{kW}$	$7.0\mathrm{A}$	$133\mathrm{A}$	$335\mathrm{V}$	$16\mathrm{V}$	19	$26.7 \mu H$
max. I_L for $P = 1 \mathrm{kW}$	$6.7\mathrm{A}$	$128\mathrm{A}$	$450\mathrm{V}$	$11\mathrm{V}$	19	$26.7 \mu H$
max. I_L for $P = 2 \mathrm{kW}$	$11.3\mathrm{A}$	$215\mathrm{A}$	$450\mathrm{V}$	$11\mathrm{V}$	19	$26.7\mu\mathrm{H}$

Table 3.7: Comparison of the resulting inductor RMS currents I_L for different modulation schemes; $f_S = 100$ kHz.

3.2 DAB Model Regarding Conduction Losses

The simplified converter model (Figure 3.2) allows for basic analytical investigations on the DAB, however, compared to measurement results, the obtained expressions [e.g. the power transferred with phase shift modulation, (3.13)] may compute considerably different values, mainly due to converter losses. As for any power converter, conduction losses, switching losses, and core losses of magnetic components form the total losses, which is explicated in Chapter 4. In this Section, conduction losses are considered in order to obtain a more precise but still simple electric DAB converter model.

The idealized HV side full bridge circuit (Figure 3.22), connected to an inductive load L_1 and its copper resistance $R_{\text{Cu}1}$, resides in one of four possible states and applies the voltage $v_{\text{AC}1}$ to L_1 and $R_{\text{Cu}1}$, according to (3.1). On the HV side, only the switch resistances R_{S1} and the copper resistance $R_{\text{Cu}1}$ are considered;⁴ PCB conduction losses and conduction losses due to the

 4 MOSFETs form the semiconductor switches of the DAB on the HV side and on the LV side (Appendix A.2.2, Appendix A.2.3, and Appendix D) and therefore, resistors are



Figure 3.22: Idealized full bridge circuits, HV side; the switches S_1 to S_4 are composed of MOSFETs (Appendix A.2.2), the resistors R_{S1} represent the total of the respective MOSFET on-state resistances. With synchronous rectification and for the assumption of negligible dead time intervals, the MOSFET's body diodes never conduct and are thus neglected. The resistance R_{Cu1} models the inductor's AC (copper) resistance at f_S . The depicted full bridge is in state I (S_1 and S_4 are on); the dotted line indicates the associated current path.

capacitor's ESR are neglected.

For $v_{AC1} = V_1$, the switches S_1 and S_4 are turned on [state I in (3.1)] and thus, the inductor current path encloses $2R_{S1}$, L_1 , R_{Cu1} , C_{DC1} , the EMI filter, and the battery connected to the HV side. On the assumption of a constant voltage V_1 [i.e. negligible impedance of C_{DC1} at twice the switching frequency: $(2\pi(2f_S)C_{DC1})^{-1} \rightarrow 0$], the total resistance in series to L_1 ,

$$R_{1,\mathrm{I}} = 2R_{\mathrm{S1}} + R_{\mathrm{Cu1}},$$

results for state I. A similar switch configuration exists for state IV with $v_{AC1} = -V_1$, so $R_{1,IV} = R_{1,I}$. For states II and III ($v_{AC1} \approx 0$) the switch configuration is different to states I and IV. However, according to Figure 3.22, the same resistance values result: $R_{1,II} = R_{1,III} = R_{1,I}$. Thus, for the HV side full bridge, the equivalent resistance

$$R_1 = 2R_{\rm S1} + R_{\rm Cu1} \tag{3.51}$$

is considered in series to L_1 .

The electric model regarding the LV side full bridge (Figure 3.23) is slightly more complex, since measurements reveal considerable PCB conduction losses on the LV side (Section 4.2). Thus, during states I and IV, the resistance values

$$R_{2,I} = R_{2,IV} = 2R_{S2} + R_{Cu2} + R_{PCB,a} + R_{PCB,b}$$
(3.52)

result. However, the resistances $R_{2,\text{II}}$ and $R_{2,\text{III}}$ during the freewheeling states II and III are less than $R_{2,\text{I}}$ and $R_{2,\text{IV}}$,

$$R_{2,\rm II} = R_{2,\rm III} = 2R_{\rm S2} + R_{\rm Cu2} + R_{\rm PCB,a}.$$
(3.53)

Since the LV side full bridge mainly resides in states I and IV (cf. phase shift modulation) and since the difference due to $R_{\rm PCB,b}$ is small for the considered design (at DC, $R_{\rm PCB,b}$ is 160 $\mu\Omega$, Figure 4.3, and thus, at 200 A a voltage error of 32 mV occurs), the equivalent resistance

$$R_2 = R_{2,\mathrm{I}} = R_{2,\mathrm{IV}} \tag{3.54}$$

is considered.

For the actual DAB converter (Figure 3.24), the copper resistances R_{Cu1} and R_{Cu2} are replaced by the respective inductor and transformer winding resistances (cf. Appendix A.2.4 and Appendix A.9):

$$R_{\rm Cu1} = R_{L_{\rm HV}} + R_{\rm tr1}, \tag{3.55}$$

$$R_{\rm Cu2} = R_{\rm tr2}.$$
 (3.56)



Figure 3.23: Idealized full bridge circuits, LV side; the switches S_5 to S_8 are composed of MOSFETs, the resistors R_{S2} represent the total of the respective MOSFET on-state resistances. With synchronous rectification and on the assumption of negligible dead time intervals, the MOSFET's body diodes never conduct and are thus neglected. The resistance R_{Cu2} models the inductor's AC (copper) resistance at f_S , and $R_{PCB,a}$ and $R_{PCB,b}$ model the PCB's AC resistances (Figure 4.3).

The total loss resistance, R, for the DAB depicted in Figure 3.24 results with (3.51), (3.54), (3.55), and (3.56):

$$R = R_1 + n^2 R_2 = 2R_{\rm S1} + R_{L_{\rm HV}} + R_{\rm tr1} + n^2 \cdot (R_{\rm tr2} + R_{\rm PCB,a} + 2R_{\rm S2} + R_{\rm PCB,b})$$
(3.57)

(Section 4 gives the values for R_{S1} , R_{S2} , $R_{PCB,a}$ and $R_{PCB,b}$, cf. Figure 4.3; in Appendix A.2.4, R_{tr1} and R_{tr2} are determined, cf. Table A.9; Appendix A.9 outlines the calculation of $R_{L_{HV}}$). Moreover, according to Figure 3.24 and without the stray inductances (transformer, PCB, switches) being considered,

$$L = L_1 + n^2 L_2 \tag{3.58}$$

results.⁵ The corresponding DAB converter model considers conduction losses and is depicted in Figure 3.25.

used to model the respective semiconductor conduction losses; the limitation of the onstate voltage due to the MOSFET's body diodes during reverse current conditions is neglected. If IGBTs are selected, the forward voltage drops of the IGBTs, $V_{\rm s}$, and of the respective free-wheeling diodes, $V_{\rm d}$, need to be considered separately [103].

⁵The more detailed DAB converter model discussed in Section 3.3 includes the different parasitic stray inductances, besides the transformer magnetizing inductance.



Figure 3.24: DAB converter with idealized full bridges; the resistors $R_{\rm S1}$ and $R_{\rm S2}$ represent the total of the respective MOSFET on-state resistances, $R_{L_{\rm HV}}$ models the AC (copper) resistance of the inductor at $f_{\rm S}$, $R_{\rm tr1}$ and $R_{\rm tr2}$ model the respective AC resistances of the primary and secondary transformer windings (cf. Appendix A.2.4 and Table A.9), and $R_{\rm PCB,a}$ and $R_{\rm PCB,b}$ model the PCB's AC resistances (Figure 4.3).



Figure 3.25: Improved DAB model, which considers conduction losses.

3.2.1 Inductor Current and Power Transfer

The voltage $v_{\rm R}$ [cf. (3.2)], applied to the series connection of L and R, generates the current

$$i_L(t_1) = i_L(t_0) + \frac{1}{L} \int_{t_0}^{t_1} \left[v_R(t) - i_L(t) R \right] dt \quad \forall \quad t_0 < t_1$$
(3.59)

at the time t_1 , starting with an initial current $i_L(t_0)$ at t_0 . For constant supply voltages V_1 and V_2 and ideal full bridges, the inductor current

$$i_L(t_1) = e^{-\frac{T_1}{\tau}} i_L(t_0) + \left(1 - e^{-\frac{T_1}{\tau}}\right) \frac{v_R(t_{0^+})}{R}, \qquad (3.60)$$

$$T_1 = t_1 - t_0, \quad \tau = L/R,$$
 (3.61)

results.

With rectangular voltages v_{AC1} and v_{AC2} , the calculation of P_1 and P_2 is again carried out with piecewise integration of (3.60), using (3.5) and (3.6); e.g. for phase shift modulation (Figure 3.4), P_1 and P_2 are calculated with:

$$P_{1} = \frac{2}{T_{\rm S}} \left(V_{1} \int_{0}^{T_{\varphi}} i_{L}(t) \mathrm{d}t + V_{1} \int_{T_{\varphi}}^{T_{\rm S}/2} i_{L}(t) \mathrm{d}t \right) \text{ and }$$
(3.62)

$$P_{2} = \frac{2}{T_{\rm S}} \left(-n \, V_{2} \, \int_{0}^{T_{\varphi}} i_{L}(t) \mathrm{d}t + n \, V_{2} \, \int_{T_{\varphi}}^{T_{\rm S}/2} i_{L}(t) \mathrm{d}t \right) \tag{3.63}$$

[for the assumption of half-cycle symmetry, cf. (3.10)]. Thus, with constant voltages V_1 and V_2 , the solution to the piecewise average inductor current is needed:

$$\int_{t_0}^{t_1} i_L(t) dt = \frac{v_R(t_{0^+})}{R} \cdot T_1 + \tau \cdot \left(i_L(t_0) - \frac{v_R(t_{0^+})}{R} \right) \cdot \left(1 - e^{-\frac{T_1}{\tau}} \right) \quad (3.64)$$

[using (3.61)].

Besides P_1 and P_2 , the inductor RMS current, I_L , is an important quantity with respect to the dissipated power (cf. Section 4.1). Its calculation as well requires piecewise integration, e.g. for the current waveform depicted in Figure 3.4:

$$I_L = \sqrt{\frac{2}{T_{\rm S}} \left[\int_0^{T_{\varphi}} i_L(t)^2 \,\mathrm{d}t + \int_{T_{\varphi}}^{T_{\rm S}/2} i_L(t)^2 \,\mathrm{d}t \right]} \,. \tag{3.65}$$

The solutions to the integrals employed in (3.65) are obtained as

$$\int_{t_0}^{t_1} i_L(t)^2 dt = \tau \cdot \left[\frac{i_L(t_0)^2}{2} + \frac{i_L(t_0) v_R(t_{0^+})}{R} + \left(\frac{T_1}{\tau} - \frac{3}{2}\right) \cdot \left(\frac{v_R(t_{0^+})}{R}\right)^2 - 2 \cdot \frac{v_R(t_{0^+})}{R} \cdot \left(i_L(t_0) - \frac{v_R(t_{0^+})}{R}\right) \cdot e^{-\frac{T_1}{\tau}} - \frac{1}{2} \cdot \left(i_L(t_0) - \frac{v_R(t_{0^+})}{R}\right)^2 \cdot e^{-\frac{2T_1}{\tau}} \right]$$
(3.66)
and (3.61).

With the DAB model with conduction losses, the steady-state values of P_1 , P_2 , and I_L are always calculated based on (3.64) and (3.66), independent on the employed modulation scheme. However, the inductor current at t = 0 must be known in order to evaluate P_1 , P_2 , and I_L [i.e. (3.62), (3.63), and (3.65) for phase shift modulation with $0 < T_{\varphi} < T_S/2$, cf. Figure 3.4]. On the assumption of half-cycle symmetry, k different time intervals [e.g. k = 3 in Figure 3.7, cf. (3.7)], $t_0 = 0$, and with (3.61), the general expression

$$i_{L,0} = -\frac{i_L(T_{\rm S}/2)|_{i_L(0)=0}}{1 + \prod_{i=1}^k e^{-T_i/\tau}}$$
(3.67)

is derived for the steady-state inductor current $i_{L,0} = i_L(t = t_0 = 0)$. In (3.67), $i_L(T_S/2)|_{i_L(0)=0}$ is the inductor current that results after a halfcycle if the current at the beginning of the half-cycle, $i_L(0)$, was zero and T_i is the duration of the *i*-th time interval.

3.2.2 Implications on the Modulation Schemes

Phase Shift Modulation

With phase shift modulation, two different time intervals occur during one half-cycle (cf. Section 3.1.2) and the inductor current

$$i_{L}(t) = e^{-\frac{t}{\tau}} i_{L,0} + \left(1 - e^{-\frac{t}{\tau}}\right) \frac{V_{1} + nV_{2}}{R}, \quad \forall \quad 0 < t < T_{\varphi},$$

$$i_{L}(t) = e^{-\frac{t - T_{\varphi}}{\tau}} i_{L}(T_{\varphi}) + \left(1 - e^{-\frac{t - T_{\varphi}}{\tau}}\right) \frac{V_{1} - nV_{2}}{R}, \quad \forall \quad T_{\varphi} < t < T_{S}/2.$$
(3.68)

results $[i_{L,0}$ is calculated with (3.67)].

Due to the losses $R \cdot I_L^2$, different power values P_1 and P_2 result. With (3.4), (3.5), (3.6), (3.60), (3.61) and (3.67), the expressions

$$P_{\rm in} = \left\{ \frac{(V_{\rm in}')^2}{R} \left[1 - 4\tau f_{\rm S} \tanh\left(\frac{1}{4\tau f_{\rm S}}\right) \right] - \frac{V_{\rm in}' V_{\rm out}'}{R} \left[1 - \frac{2|\varphi|}{\pi} + 4\tau f_{\rm S} \sinh\left(\frac{|\varphi|}{2\pi\tau f_{\rm S}}\right) \right] - \frac{4V_{\rm in}' V_{\rm out}' \tau f_{\rm S}}{R} \left[2\sinh\left(\frac{|\varphi|}{4\pi\tau f_{\rm S}}\right) \sinh\left(\frac{|\varphi|}{4\pi\tau f_{\rm S}}\right) \right] + \frac{4V_{\rm in}' V_{\rm out}' \tau f_{\rm S}}{R} \left[e^{\frac{|\varphi|}{2\pi\tau f_{\rm S}}} \tanh\left(\frac{1}{4\tau f_{\rm S}}\right) \right] \right\} \cdot \operatorname{sgn}(\varphi)$$
(3.69)

and

$$P_{\text{out}} = \left\{ \frac{(V_{\text{out}}')^2}{R} \left[4\tau f_{\text{S}} \tanh\left(\frac{1}{4\tau f_{\text{S}}}\right) - 1 \right] + \frac{V_{\text{in}}' V_{\text{out}}'}{R} \left[1 - \frac{2|\varphi|}{\pi} + 4\tau f_{\text{S}} \sinh\left(\frac{|\varphi|}{2\pi\tau f_{\text{S}}}\right) \right] - \frac{4V_{\text{in}}' V_{\text{out}}' \tau f_{\text{S}}}{R} \left[2\sinh\left(\frac{|\varphi|}{4\pi\tau f_{\text{S}}}\right) \sinh\left(\frac{|\varphi|}{4\pi\tau f_{\text{S}}}\right) \right] - \frac{4V_{\text{in}}' V_{\text{out}}' \tau f_{\text{S}}}{R} \left[e^{-\frac{|\varphi|}{2\pi\tau f_{\text{S}}}} \tanh\left(\frac{1}{4\tau f_{\text{S}}}\right) \right] \right\} \cdot \text{sgn}(\varphi)$$
(3.70)

result for phase shift modulation, with

$$\begin{array}{l}
\left. \begin{array}{c} V_{\mathrm{in}}' = V_{1} \\
V_{\mathrm{out}}' = n V_{2} \\
P_{\mathrm{in}} = P_{1} \\
P_{\mathrm{out}} = P_{2} \end{array} \right\} \text{ power transfer HV} \rightarrow \mathrm{LV} \ (\varphi > 0), \\
\left. \begin{array}{c} (3.71) \\
V_{\mathrm{out}}' = n V_{2} \\
V_{\mathrm{out}}' = V_{1} \\
P_{\mathrm{out}} = P_{2} \\
P_{\mathrm{out}} = P_{1} \end{array} \right\} \text{ power transfer LV} \rightarrow \mathrm{HV} \ (\varphi < 0). \\
\end{array}$$

Figure 3.26 depicts P_1 and P_2 for the DAB with $L = 26.7 \,\mu\text{H}$, $R = 0.76 \,\Omega$ (according to the total conduction losses detailed in Sections 4.1 and 4.2), n = 19, $V_1 = 340 \,\text{V}$, and $V_2 = 12 \,\text{V}$. Obviously, maximum P_1 and P_2 are not obtained for $\varphi = \pm \pi/2$. Even more, the phase shift angle for maximum $|P_1|$, $\varphi_{1,\text{max}}$, is different to the phase shift angle $\varphi_{2,\text{max}}$ for maximum $|P_2|$. Depending on the power transfer direction, $\varphi_{1,\text{max}}$ and $\varphi_{2,\text{max}}$ are:

$$\text{HV} \to \text{LV: } \varphi_{1,\max,A} = \frac{2\pi f_{\text{S}}L \cdot \ln\left(\frac{1+e^{\frac{R}{2Lf_{\text{S}}}}}{2}\right)}{R}, \quad \varphi_{2,\max,A} = \pi - \varphi_{1,\max,A},$$

$$\text{LV} \to \text{HV: } \varphi_{2,\max,B} = -\frac{2\pi f_{\text{S}}L \cdot \ln\left(\frac{1+e^{\frac{R}{2Lf_{\text{S}}}}}{2}\right)}{R}, \quad \varphi_{1,\max,B} = -\pi - \varphi_{2,\max,B}.$$

$$(3.72)$$

The technically useful range for φ is $\varphi_{1,\max,B} < \varphi < \varphi_{2,\max,A}$. Outside these limits, the output power, P_{out} , decreases with increasing $|\varphi|$.

No closed-form expression exists for the phase shift angle φ with respect to a certain output power. Thus, φ is obtained with (3.70) using a numerical solver.



Figure 3.26: Transferred power for the DAB designed in Section A.2, phase shift modulation, $V_1 = 340$ V, $V_2 = 12$ V, n = 19, $L = 26.7 \,\mu\text{H}$, and $R = 0.76 \,\Omega$ (according to the total conduction losses detailed in Sections 4.1 and 4.2). Due to the conduction losses, different power levels P_1 and P_2 occur and the maxima of $|P_1|$ and $|P_2|$ occur for $\varphi \neq \pm \pi/2$ [cf. (3.72)].

Due to the conduction losses, the RMS value of the inductor current, I_L , depends on the direction of power transfer (Figure 3.27). Further, a slightly higher maximum RMS current of 293 A occurs at $V_1 = 450$ V, $V_2 = 11$ V, and $P_1 = -2$ kW, compared to the lossless DAB model (283 A, Figure 3.6).

Alternative Modulation Schemes

For the triangular and trapezoidal current mode modulation schemes (cf. Section 3.1.3), closed-form expressions exist for the duty cycles D_1 and D_2 with respect to the phase shift angle φ ; however, no closed-form expression exists for the phase shift angle φ , required to obtain a certain output power [i.e. P_2 or P_1 depending on the direction of power transfer, cf. (3.71)]. In general, for the derivation of D_1 , D_2 , and φ , six different cases need to be considered:

- 1. triangular current mode modulation with $V_1 > n V_2$, HV \rightarrow LV,
- 2. triangular current mode modulation with $V_1 < n V_2$, $HV \rightarrow LV$,



Figure 3.27: LV side transformer RMS currents for phase shift modulation and operation within the specified voltage ranges: (a) $P_2 = 1 \text{ kW}$, (b) $P_1 = -1 \text{ kW}$, (c) $P_2 = 2 \text{ kW}$, and (d) $P_1 = -2 \text{ kW}$; n = 19, $L = 26.7 \mu$ H, and $R = 0.76 \Omega$.

- 3. trapezoidal current mode modulation, $HV \rightarrow LV$,
- 4. triangular current mode modulation with $V_1 > n V_2$, $LV \rightarrow HV$,
- 5. triangular current mode modulation with $V_1 < n V_2$, $LV \rightarrow HV$,
- 6. trapezoidal current mode modulation, $LV \rightarrow HV$.

However, due to the symmetry of the regarded DAB model, only one power transfer direction needs to be investigated (e.g. $HV \rightarrow LV$), and the equations for the reverse direction are obtained with

$$\begin{bmatrix} V_1 \ V_2 \ P_1 \ P_2 \ D_1 \ D_2 \ \varphi \end{bmatrix}^{\mathrm{T}} \to \begin{bmatrix} n V_2 \ V_1/n \ -P_2 \ -P_1 \ D_2 \ D_1 \ -\varphi \end{bmatrix}^{\mathrm{T}}.$$
(3.73)

Triangular Current Mode Modulation with $V_1 > nV_2$ (**HV** \rightarrow **LV**). With the considered DAB model and with given phase shift angle φ , the duty cycles D_1 and D_2 become:

$$D_{1} = \tau f_{\rm S} \ln \left(\frac{V_{1} - nV_{2}}{V_{1} - nV_{2} \, \mathrm{e}^{\varphi/(\tau \pi f_{\rm S})}} \right), \tag{3.74}$$

$$D_2 = D_1 + \varphi/\pi.$$
 (3.75)

The maximum phase shift angle for this operation mode is:

$$\varphi_{\Delta,\max,A} = \frac{\pi}{2} \left[1 - 2\tau f_{\rm S} \ln \left(\frac{V_1 - n \, V_2 \cdot (1 - \mathrm{e}^{1/(2\tau f_{\rm S})})}{V_1} \right) \right]. \tag{3.76}$$

The port 1 power level P_1 , the port 2 power level P_2 , and the inductor RMS current I_L are equal to:

$$P_{1} = 2f_{\rm S}V_{1} \int_{0}^{D_{1}/f_{\rm S}} i_{L}(t)|_{v_{\rm R}=V_{1}-nV_{2}} dt, \qquad (3.77)$$

$$P_{2} = 2f_{\rm S}nV_{2} \left[\int_{0}^{D_{1}/f_{\rm S}} i_{L}(t)|_{v_{\rm R}=V_{1}-nV_{2}} dt + \int_{D_{1}/f_{\rm S}}^{D_{2}/f_{\rm S}} i_{L}(t)|_{v_{\rm R}=-nV_{2}} dt \right], \qquad (3.78)$$

$$I_{L} = \sqrt{2f_{\rm S} \left[\int_{0}^{D_{1}/f_{\rm S}} \left(i_{L}(t)|_{v_{\rm R}=V_{1}-nV_{2}} \right)^{2} \mathrm{d}t + \int_{D_{1}/f_{\rm S}}^{D_{2}/f_{\rm S}} \left(i_{L}(t)|_{v_{\rm R}=-nV_{2}} \right)^{2} \mathrm{d}t \right]}.$$
(3.79)

With (3.64), (3.66), and $i_L(0) = 0$, closed-form expressions are feasible, though, rather complicated expressions result. Moreover, neither for P_1 nor for P_2 closed-form expressions with respect to the phase shift angle φ exist and thus, the phase shift angle required for a certain output power P_2 is obtained using a numerical solver.

Triangular Current Mode Modulation with $V_1 < nV_2$ (HV \rightarrow LV). With given phase shift angle φ , the duty cycles D_1 and D_2 are equal to

$$D_1 = \frac{\varphi}{\pi} + \tau f_{\rm S} \ln \left(\frac{V_1 \,\mathrm{e}^{-\varphi/(\tau \pi f_{\rm S})} - nV_2}{V_1 - nV_2} \right) \quad \text{and} \tag{3.80}$$

$$D_2 = D_1 - \varphi/\pi; \tag{3.81}$$

the expression for the maximum phase shift angle is:

$$\varphi_{\Delta,\max,\mathrm{B}} = \pi \tau f_{\mathrm{S}} \ln \left(\frac{V_1}{n \, V_2} + \frac{(n \, V_2 - V_1) \cdot \mathrm{e}^{1/(2\tau f_{\mathrm{S}})}}{n \, V_2} \right). \tag{3.82}$$

 P_1 , P_2 , and I_L are calculated according to:

$$P_{1} = 2f_{\rm S}V_{1} \left[\int_{0}^{D_{1}/f_{\rm S}} i_{L}(t) \big|_{v_{\rm R}=V_{1}} \,\mathrm{d}t + \int_{D_{1}/f_{\rm S}}^{D_{2}/f_{\rm S}} i_{L}(t) \big|_{v_{\rm R}=V_{1}-nV_{2}} \,\mathrm{d}t \right], \quad (3.83)$$

$$P_2 = 2f_{\rm S}nV_2 \int_{D_1/f_{\rm S}}^{D_2/f_{\rm S}} i_L(t)|_{v_{\rm R}=V_1-nV_2} \,\mathrm{d}t, \qquad (3.84)$$

$$I_L = \sqrt{2f_{\rm S} \left[\int_0^{D_1/f_{\rm S}} \left(i_L(t)|_{v_{\rm R}=V_1} \right)^2 \mathrm{d}t + \int_{D_1/f_{\rm S}}^{D_2/f_{\rm S}} \left(i_L(t)|_{v_{\rm R}=V_1-nV_2} \right)^2 \mathrm{d}t \right]}$$
(3.85)

with (3.64), (3.66), and $i_L(0) = 0$ and again, a numerical solver is needed in order to determine the phase shift angle φ required to obtain a certain output power P_2 .

Trapezoidal Current Mode Modulation (HV \rightarrow LV). With given phase shift angle φ , the duty cycles

$$D_{1} = \frac{1}{2} - \frac{\varphi}{\pi} + \tau f_{\rm S} \ln \left(\frac{V_{1} + nV_{2} \,\mathrm{e}^{1/(2\tau f_{\rm S})}}{V_{1} \,\mathrm{e}^{(1/2 - \varphi/\pi)/(\tau f_{\rm S})} + nV_{2}} \right) \quad \text{and} \tag{3.86}$$

$$D_2 = \frac{1}{2} - \frac{\varphi}{\pi} + \tau f_{\rm S} \ln \left(\frac{V_1 + nV_2 \,\mathrm{e}^{-(1/2 - \varphi/\pi)/(\tau f_{\rm S})}}{V_1 \,\mathrm{e}^{-1/(2\tau f_{\rm S})} + nV_2} \right)$$
(3.87)

result; maximum power transfer is achieved for

$$\varphi_{\Box,\max} = \tau \pi f_{\rm S} \Biggl\{ \ln \Biggl[V_1 (2nV_2 - V_1) + nV_2 (nV_2 - V_1) \,\mathrm{e}^{1/(2\tau f_{\rm S})} \\ + \Biggl(\left(V_1^2 + nV_1 V_2 (\mathrm{e}^{1/(2\tau f_{\rm S})} - 2) - (nV_2)^2 \,\mathrm{e}^{1/(2\tau f_{\rm S})} \right)^2 \\ + 4nV_1^2 V_2 \left(V_1 + nV_2 \,\mathrm{e}^{1/(2\tau f_{\rm S})} \right) \Biggr)^{\frac{1}{2}} \Biggr] \\ - \ln (2nV_2) - \ln \left(V_1 \,\mathrm{e}^{-1/(2\tau f_{\rm S})} + nV_2 \right) \Biggr\}.$$
(3.88)

 P_1 , P_2 , and I_L are calculated according to:

$$P_{1} = 2f_{\rm S}V_{1} \left[\int_{0}^{(\frac{1}{2} - D_{2})/f_{\rm S}} i_{L}(t)|_{v_{\rm R} = V_{1}} \,\mathrm{d}t + \int_{(\frac{1}{2} - D_{2})/f_{\rm S}}^{D_{1}/f_{\rm S}} i_{L}(t)|_{v_{\rm R} = V_{1} - nV_{2}} \,\mathrm{d}t \right],$$
(3.89)

$$P_{2} = 2f_{\rm S}nV_{2} \left[\int_{(\frac{1}{2} - D_{2})/f_{\rm S}}^{D_{1}/f_{\rm S}} i_{L}(t)|_{v_{\rm R} = V_{1} - nV_{2}} \,\mathrm{d}t + \int_{D_{1}/f_{\rm S}}^{1/(2f_{\rm S})} i_{L}(t)|_{v_{\rm R} = -nV_{2}} \,\mathrm{d}t \right],$$
(3.90)

$$I_{L} = \left\{ 2f_{\rm S} \left[\int_{0}^{(\frac{1}{2} - D_{2})/f_{\rm S}} \left(i_{L}(t) |_{v_{\rm R} = V_{\rm I}} \right)^{2} \mathrm{d}t + \int_{(\frac{1}{2} - D_{2})/f_{\rm S}}^{D_{1}/f_{\rm S}} \left(i_{L}(t) |_{v_{\rm R} = V_{\rm I} - nV_{\rm 2}} \right)^{2} \mathrm{d}t + \int_{D_{1}/f_{\rm S}}^{1/(2f_{\rm S})} \left(i_{L}(t) |_{v_{\rm R} = -nV_{\rm 2}} \right)^{2} \mathrm{d}t \right] \right\}^{\frac{1}{2}}$$
(3.91)

with (3.64), (3.66), and $i_L(0) = 0$; a numerical solver is needed to determine the required phase shift angle φ for a certain output power P_2 .

Resulting Inductor RMS Currents. The operation with n = 19, $L = 18.7 \,\mu\text{H}$, and $R = 0.76 \,\Omega$ results in the LV side referred inductor RMS currents $n I_L$ depicted in Figure 3.28: compared to the lossless model (calculated maximum RMS current: 235 A), a maximum current of 245 A is calculated for $V_1 = 450 \text{ V}, V_2 = 11 \text{ V}$, and $P_1 = -2 \text{ kW}$. Moreover, due to the conduction losses, different inductor RMS currents result for the different directions of power transfer.

Optimal Modulation

Similar to the modulation with minimum inductor RMS current I_L , presented in Section 3.1.3, a modulation scheme with minimum I_L exists for the R-L model of the DAB (Figure 3.25). However, due to the higher model complexity, the control parameters D_1 and D_2 , required to obtain minimum I_L for a given phase shift angle φ , need to be determined using a numerical solver. With the numerical solver, though, not only the conduction losses but additional loss components (e.g. switching losses) can be considered in order to determine the control parameters required for minimum power dissipation. Thus, a more comprehensive optimization of the employed modulation scheme is presented in Section 5.2.



Figure 3.28: LV side transformer RMS currents for the combined triangular and trapezoidal current mode modulation and operation within the specified voltage ranges: (a) $P_2 = 1 \text{ kW}$, (b) $P_1 = -1 \text{ kW}$, (c) $P_2 = 2 \text{ kW}$, and (d) $P_1 = -2 \text{ kW}$; n = 19, $L = 18.7 \mu$ H, and $R = 0.76 \Omega$.

3.3 Model Including the Magnetizing Inductance

The transformer magnetizing inductance $L_{\rm M}$ causes reactive power in the high frequency AC part of the DAB, in addition to the reactive power that originates from the main converter inductance L, and may considerably influence the DAB transformer currents $i_{\rm AC1}$ and $i_{\rm AC2}$. The employed simplified transformer model depicted in Figure 3.29 disregards the effect of parasitic capacitances (winding capacitances, capacitance between primary and secondary side) and assumes equal (HV side referred) stray inductances

$$L_{\rm tr1} = n^2 L_{\rm tr2}; \tag{3.92}$$

the resistance values R_{tr1} and R_{tr2} are obtained from calculations and measurements (Appendix A.2.4, Table A.9). This transformer model replaces the transformer depicted in Figure 3.30. In Figure 3.30, however, two additional inductors, L_{LV} and L_{HV} , are shown: L_{LV} considers the LV side parasitic inductance of the high frequency path (Appendix A.2.4) and L_{HV} is needed in order to achieve the required total inductance L, i.e.

$$L_{\rm HV} \approx L - L_{\rm tr1} - n^2 \left(L_{\rm tr2} + L_{\rm LV} \right)$$
 (3.93)

(for the assumption $L_{tr1} + n^2(L_{tr2} + L_{LV}) \ll L_M$). Accordingly, in the extended DAB model depicted in Figure 3.31, the inductors L_1 , L_2 , and L_M model the inductive nature of the HF network, the resistor R_M models the transformer core losses (cf. Section 4.1.2), and the resistors R_1 and R_2 con-



Figure 3.29: The considered transformer model with HV side stray inductance L_{tr1} , LV side stray inductance L_{tr2} and magnetizing inductance L_{M} . The resistors R_{tr1} and R_{tr2} model the winding losses and the resistor R_{M} models the core losses (cf. Appendix A.2.4 and Table A.9).



Figure 3.30: The DAB circuit used to obtain the DAB model depicted in Figure 3.31; the transformer needs to be replaced with the transformer model shown in Figure 3.29.



Figure 3.31: The extended DAB model, which includes the effects of the magnetizing inductance, conduction and copper losses, and core losses.

sider the conduction losses (cf. Section 3.2):

$$L_1 = L_{\rm HV} + L_{\rm tr1}, \tag{3.94}$$

$$R_1 = 2R_{\rm S1} + R_{L_{\rm HV}} + R_{\rm tr1}, \qquad (3.95)$$

$$L_2 = L_{\rm tr2} + L_{\rm LV},\tag{3.96}$$

$$R_2 = R_{\rm tr2} + 2R_{\rm S2} + R_{\rm PCB,a} + R_{\rm PCB,b}$$
(3.97)

 $(R_{L_{\rm HV}}$ is the copper resistance of the additional inductor $L_{\rm HV}$).



Figure 3.32: The reconfigured Delta model of the reactive HF network of the DAB.

3.3.1 Inductor Current and Power Transfer

The DAB power values at the HV port and at the LV port are again calculated using the instantaneous power values,

$$p_1(t) = v_{AC1}(t) \cdot i_{L_1}(t) \text{ and } p_2(t) = v_{AC2}(t) \cdot i_{L_2}(t),$$
 (3.98)

and the expressions (3.5) and (3.6). Obviously, different inductor currents i_{L_1} and i_{L_2} occur on the HV side and on the LV side, which need to be known prior to the calculation of P_1 and P_2 . However, third order differential equations need to be solved to determine $i_{L_1}(t)$ and $i_{L_2}(t)$; consequently, very complicated expressions result for $i_{L_1}(t)$ and $i_{L_2}(t)$.

In a more practical approach, the given reactive HF network is reconfigured using the Wye-delta transformation (Figure 3.32). This considerably reduces the effort to calculate i_{L_1} and i_{L_2} , since the voltages v_{AC1} , v_{AC2} , and $v_R = v_{AC1} - v_{AC2}$, applied to \underline{Z}_{13} , \underline{Z}_{23} , and \underline{Z}_{12} , are known in advance. The expressions for \underline{Z}_{12} , \underline{Z}_{13} , and \underline{Z}_{23} ,

$$\underline{Z}_{12} = R_1 + n^2 R_2 + s \cdot (L_1 + n^2 L_2) + \frac{n^2 (R_1 + s L_1) \cdot (R_2 + s L_2)}{R_M ||(s L_M)}, \quad (3.99)$$

$$\underline{Z}_{13} = R_1 + s L_1 + (R_{\rm M} || (s L_{\rm M})) \cdot \left(1 + \frac{R_1 + s L_1}{n^2 (R_2 + s L_2)}\right), \tag{3.100}$$

$$\underline{Z}_{23} = n^2 (R_2 + s L_2) + (R_{\rm M} || (s L_{\rm M})) \cdot \left(1 + \frac{n^2 (R_2 + s L_2)}{R_1 + s L_1}\right), \qquad (3.101)$$

however, contain higher order elements and need to be simplified. For the assumptions

$$R_1 \ll s L_1, \ R_2 \ll s L_2, \ R_M \gg s L_M,$$
 (3.102)

 $\underline{Z}_{12}, \underline{Z}_{13}, \text{ and } \underline{Z}_{23}$ become

$$\underline{Z}_{12} \approx R_1 + n^2 R_2 + s \cdot \left(L_1 + n^2 L_2 + \frac{n^2 L_1 L_2}{L_M} \right) = R_{12} + s L_{12}, \quad (3.103)$$

$$\underline{Z}_{13} \approx (R_1 + s L_1) \left(1 + \frac{L_{\rm M}}{n^2 L_2} \right) + s L_{\rm M} = R_{13} + s L_{13}, \tag{3.104}$$

$$\underline{Z}_{23} \approx n^2 \left(R_2 + s \, L_2 \right) \, \left(1 + \frac{L_{\rm M}}{L_1} \right) + s \, L_{\rm M} = R_{23} + s \, L_{23}, \tag{3.105}$$

each being equal to a series connection of an inductor and a resistor. Consequently, the currents i_{L_1} and i_{L_2} are calculated with

$$i_{L_1} = i_{\underline{Z}_{12}} + i_{\underline{Z}_{13}}$$
 and $i_{L_2} = n (i_{\underline{Z}_{12}} - i_{\underline{Z}_{23}}),$ (3.106)

using the solutions to the first order differential equations for $i_{\underline{Z}_{12}}$, $i_{\underline{Z}_{13}}$, and $i_{\underline{Z}_{23}}$,

$$i_{\underline{Z}_{12}}(t_1) = e^{-\frac{T_1}{\tau_{12}}} i_{\underline{Z}_{12}}(t_0) + \left(1 - e^{-\frac{T_1}{\tau_{12}}}\right) \frac{v_{\mathrm{R}}(t_{0^+})}{R_{12}},\tag{3.107}$$

$$i_{\underline{Z}_{13}}(t_1) = e^{-\frac{T_1}{\tau_{13}}} i_{\underline{Z}_{13}}(t_0) + \left(1 - e^{-\frac{T_1}{\tau_{13}}}\right) \frac{v_{AC1}(t_{0^+})}{R_{13}},$$
(3.108)

$$i_{\underline{Z}_{23}}(t_1) = e^{-\frac{T_1}{\tau_{23}}} i_{\underline{Z}_{23}}(t_0) + \left(1 - e^{-\frac{T_1}{\tau_{23}}}\right) \frac{v_{AC2}(t_{0+})}{R_{23}},$$
(3.109)

$$T_1 = t_1 - t_0, \ \tau_{12} = L_{12}/R_{12}, \ \tau_{13} = L_{13}/R_{13}, \ \tau_{23} = L_{23}/R_{23}.$$
(3.110)

Due to the discussed simplifications, a certain error results for the currents $i_{L_1}(t)$ and $i_{L_2}(t)$, illustrated in Figure 3.33 for the inductor RMS currents I_{L_1} and I_{L_2} , phase shift operation, $V_1 = 340$ V, $V_2 = 12$ V, and $P_2 = 2$ kW $[L = 26.7 \,\mu\text{H}, \text{ cf. } (3.93), n = 19, f_{\rm S} = 100 \,\text{kHz}]$. There, for $L_{\rm M} = 2.7 \,\text{mH}$ and $R_{\rm M} = 13.6 \,\text{k}\Omega$, a relative error of -0.12% occurs for $i_{L_1}(t)$ and less than $\pm 0.01\%$ for $i_{L_2}(t)$;⁶ the error increases with decreasing $L_{\rm M}$.

With the extended DAB model (Figure 3.31), due to the lossy components R_1 , R_2 , and R_M , different power values P_1 and P_2 are obtained, which are calculated using (3.5), (3.6) and (3.98):

$$P_{1} = \frac{2}{T_{\rm S}} \int_{0}^{T_{\rm S}/2} v_{\rm AC1}(t) \, i_{L_{1}}(t) \mathrm{d}t \quad \text{and} \quad P_{2} = \frac{2}{T_{\rm S}} \int_{0}^{T_{\rm S}/2} v_{\rm AC2}(t) \, i_{L_{2}}(t) \mathrm{d}t.$$
(3.111)

 $^{{}^{6}}R_{\rm M}$ is calculated for the assumption that solely $n v_{\rm AC2}$ causes transformer core losses (cf. Appendix A.2.5). For phase shift operation and $V_2 = 12$ V, transformer core losses of 4.3 W are calculated and thus, the HV side referred resistance $R_{\rm M} = 13.6$ k Ω results. The magnetizing inductance $L_{\rm M}$ is given in Appendix A.2.4.



Figure 3.33: Relative errors of the calculated RMS currents I_{L_1} and I_{L_2} due to the simplifications (3.103), (3.104), and (3.105): (a) $e_1 = (I_{L_1,\text{calc}} - I_{L_1,\text{sim}})/I_{L_1,\text{sim}}$ and (b) $e_2 = (I_{L_2,\text{calc}} - I_{L_2,\text{sim}})/I_{L_2,\text{sim}}$. $I_{L_1,\text{calc}}$ and $I_{L_2,\text{calc}}$ denote the calculated inductor RMS currents that are subject to errors; the reference values $I_{L_1,\text{sim}}$ and $I_{L_2,\text{sim}}$ are obtained with a circuit simulator.

For the assumption of constant voltages V_1 and V_2 , the solutions to the integrals

$$\int_{0}^{T_{\rm S}/2} i_{L_1} dt = \int_{0}^{T_{\rm S}/2} i_{\underline{Z}_{12}} dt + \int_{0}^{T_{\rm S}/2} i_{\underline{Z}_{13}} dt \quad \text{and} \tag{3.112}$$

$$\int_{0}^{T_{\rm S}/2} i_{L_2} \mathrm{d}t = \int_{0}^{T_{\rm S}/2} i_{\underline{Z}_{12}} \mathrm{d}t - \int_{0}^{T_{\rm S}/2} i_{\underline{Z}_{23}} \mathrm{d}t$$
(3.113)

are sufficient to solve for P_1 and P_2 using piecewise integration [cf. Section 3.2.1 and (3.64)]. The steady-state values

$$i_{\underline{Z}_{12},0} = i_{\underline{Z}_{12}}(t)\big|_{t=0}, \qquad (3.114)$$

$$i_{\underline{Z}_{13},0} = \left. i_{\underline{Z}_{13}}(t) \right|_{t=0}, \qquad (3.115)$$

$$i_{\underline{Z}_{23},0} = i_{\underline{Z}_{23}}(t)\big|_{t=0}, \qquad (3.116)$$

required to evaluate (3.112) and (3.113), are calculated with (3.67).

The RMS currents I_{L_1} and I_{L_2} are given with:

$$I_{L_1} = \sqrt{\frac{2}{T_{\rm S}} \int_0^{T_{\rm S}/2} i_{L_1}^2 \mathrm{d}t} \quad \text{and} \quad I_{L_2} = \sqrt{\frac{2}{T_{\rm S}} \int_0^{T_{\rm S}/2} i_{L_2}^2 \mathrm{d}t}, \tag{3.117}$$

$$\int_{t_0}^{t_1} i_{L_1}^2 dt = \int_{t_0}^{t_1} \left(i_{\underline{Z}_{12}}^2 + 2i_{\underline{Z}_{12}} i_{\underline{Z}_{13}} + i_{\underline{Z}_{13}}^2 \right) dt, \qquad (3.118)$$

$$\int_{t_0}^{t_1} i_{L_2}^2 \mathrm{d}t = \int_{t_0}^{t_1} \left(i_{\underline{Z}_{12}}^2 - 2i_{\underline{Z}_{12}} i_{\underline{Z}_{23}} + i_{\underline{Z}_{23}}^2 \right) \mathrm{d}t.$$
(3.119)

The quadratic terms of (3.118) and (3.119) are solved with (3.66) and the solution to the remaining integral in (3.118) is:

$$\begin{split} \int_{t_0}^{t_1} i_{\underline{Z}_{12}} i_{\underline{Z}_{13}} \mathrm{d}t &= \frac{v_{\mathrm{R}}(t_{0^+}) v_{\mathrm{AC1}}(t_{0^+})}{R_{12} R_{13}} \left\{ T_1 + \frac{1}{\tau_{12} + \tau_{13}} \cdot \left[\\ \tau_{12}^2 \left(\frac{i_{\underline{Z}_{12}}(t_0) R_{12}}{v_{\mathrm{R}}(t_{0^+})} - 1 \right) \right. \\ &+ \tau_{12} \tau_{13} \left(\frac{i_{\underline{Z}_{12}}(t_0) i_{\underline{Z}_{13}}(t_0) R_{12} R_{13}}{v_{\mathrm{R}}(t_0 +) v_{\mathrm{AC1}}(t_{0^+})} - 1 \right) \\ &+ \tau_{12}^2 \left(\frac{i_{\underline{Z}_{13}}(t_0) R_{13}}{v_{\mathrm{AC1}}(t_{0^+})} - 1 \right) \right] \\ &- \tau_{12} \left(\frac{i_{\underline{Z}_{12}}(t_0) R_{12}}{v_{\mathrm{R}}(t_{0^+})} - 1 \right) \mathrm{e}^{-\frac{T_1}{\tau_{12}}} \\ &- \tau_{13} \left(\frac{i_{\underline{Z}_{13}}(t_0) R_{13}}{v_{\mathrm{AC1}}(t_{0^+})} - 1 \right) \mathrm{e}^{-T_1 \left(\frac{1}{\tau_{12}} + \frac{1}{\tau_{13}} \right)} \right\}; \\ &(3.120) \end{split}$$

the same expression results for $\int_{t_0}^{t_1} i_{\underline{Z}_{12}} i_{\underline{Z}_{23}} dt$ with τ_{13} , R_{13} , and $v_{AC1}(t_{0+})$ being replaced by τ_{23} , R_{23} , and $v_{AC2}(t_{0+})$.

3.3.2 Implications on the Modulation Schemes

Phase Shift Modulation

Similar to Section 3.2.2, analytical expressions exist for $P_1(\varphi)$ and $P_2(\varphi)$, however, these expressions are very complicated. Moreover, no closed-form expression exists for the phase shift angle φ required to obtain a certain value for P_1 or P_2 ; for this purpose, a numerical solver is needed.

With $L_{\rm M} = 2.7$ mH, the resulting inductor RMS currents (Figure 3.34) are similar to the inductor RMS currents obtained in Section 3.2.2 (Figure 3.27). The maximum RMS current is 15.5 A on the HV side and 292 A on the LV side and occurs for $V_1 = 450$ V, $V_2 = 11$ V, and $P_1 = -2$ kW.

Considerably different inductor RMS currents result if a reduced magnetizing inductance, $L_{\rm M} = 200 \,\mu\text{H}$, is employed (Figure 3.35). Compared to Figure 3.34, a reduction of the maximum current is achieved on the LV side $(I_{L_2} = 275 \,\text{A} \text{ at } V_1 = 450 \,\text{V}, V_2 = 11 \,\text{V}, \text{ and } P_1 = -2 \,\text{kW})$, however, the HV side inductor RMS current, I_{L_1} , increases (maximal 15.8 A at $V_1 = 450 \,\text{V},$ $V_2 = 11 \,\text{V}, \text{ and } P_1 = -2 \,\text{kW})$. Thus, with the selected transformer turns ratio,



Figure 3.34: Inductor RMS currents for phase shift operation within the specified voltage ranges: (a) and (b) HV side, (c) and (d) LV side; $P_2 = 2 \text{ kW}$ in (a) and (c); $P_1 = -2 \text{ kW}$ in (b) and (d); n = 19, $L = 26.7 \mu\text{H}$ [cf. (3.93)], $L_{\text{M}} = 2.7 \text{ mH}$.

the magnetizing inductance $L_{\rm M}$ facilitates a reduced maximum value of I_{L_2} at the expense of a higher maximum value of I_{L_1} .

Alternative Modulation Schemes

For the modulation parameters D_1 , D_2 , and φ , employed for the triangular and trapezoidal current mode modulation (cf. Section 3.1.3), no closedform expressions exist any more; D_1 , D_2 , and φ need to be determined using a numerical solver. The employed numerical solver algorithm discussed in Section 4.6.1, however, allows to include the additional effects discussed in Chapter 4 in order to increase the accuracy of the calculated modulation parameters. Therefore, the alternative modulation schemes (i.e. the triangular current mode modulation, the trapezoidal current mode modulation, and the



Figure 3.35: Inductor RMS currents for a reduced $L_{\rm M} = 200 \,\mu\text{H}$ and phase shift operation within the specified voltage ranges: (a) and (b) HV side, (c) and (d) LV side; $P_2 = 2 \,\text{kW}$ in (a) and (c); $P_1 = -2 \,\text{kW}$ in (b) and (d); n = 19, $L = 26.7 \,\mu\text{H}$ [cf. (3.93)].

modulation scheme with minimum losses) are revisited in Chapter 5.

3.4 Conclusion

This Chapter explains the working principle of the DAB converter and details 3 different electric converter models: the lossless converter model, which facilitates a simplified calculation of the transformer currents and the transferred power, and two extended converter models which consider the impacts of the conduction losses and the magnetizing currents on the transformer current and on the converter's power transfer characteristics. Different modulation schemes are detailed; in particular, in Section 3.1.3, optimal modulation parameters D_1 , D_2 , and φ (with respect to minimum transformer RMS currents) are derived for the lossless DAB converter model. The presented converter models enable the calculation of the transformer currents (instantaneous values and RMS value) and the calculation of the transformer flux density. The currents and the flux density calculated with the final electric model of the DAB (Figure 3.31) are input quantities to the converter loss model detailed in Chapter 4.

Chapter 4

Accurate DAB Loss Model

In order to facilitate a thorough optimization of the DAB (e.g. with respect to the employed modulation scheme and the converter design), the voltage and current stresses of the different DAB components (i.e. semiconductor switches, transformer, high current PCB, and DC capacitors) and the respective power dissipation need to be determined. Both, the voltage and the current stresses, are obtained using one of the previously presented DAB models (Chapter 3).

With a simple loss model, the calculated losses prove to be inaccurate (Section 4.1). Therefore, in Sections 4.2, 4.3, 4.4, and 4.5, different refinements of the loss model are discussed. In particular, the rapidly changing DAB transformer current is identified as the key property and needs to be determined precisely in order to allow for an accurate calculation of the converter losses (Section 4.6.1). In Section 4.6.2, a comparison of the predicted losses and the experimental results is presented and a high accuracy of the theoretical results is verified.

The converter loss model detailed in this Chapter is developed for the DAB converter designed in Appendix A.2. In summary, the basic technical data of the employed DAB converter is:

- PCB: four layer PCB, 200 µm copper on each layer,
- LV side:
 - DC capacitor: 96 \times 10 μ F/25 V/X7R in parallel,
 - Switches: 8 \times IRF2804 in parallel,

- HV side:
 - DC capacitor: $6 \times 470 \,\mathrm{nF}/630 \,\mathrm{V}$ in parallel,
 - Switches: SPW47N60CFD,
- Transformer core: two planar E58 cores (ferrite),
- Transformer turns ratio and DAB converter inductance:
 - phase shift modulation: $n = 19: 1, L = 26.7 \,\mu\text{H}.$
 - trapezoidal and triangular current mode modulation schemes: $n = 16: 1, L = 15.5 \,\mu\text{H}.$
 - minimal loss mode modulation: $n = 16: 1, L = 22.4 \,\mu\text{H}.$

For all calculations and measurements in this Chapter, a reference temperature $(T = 25^{\circ}\text{C})$ has been used to avoid that two effects, namely the electric losses and the additional influence on the total losses due to components with different temperatures, occur at the same time. With this, a clear distinction between the different losses (e.g. copper losses, conduction losses, switching losses) of the power components is achieved. The shown experimental efficiency results have therefore been obtained at 25°C with the run time of the converter being limited to 30 seconds in order to assure that no heating of the components occurs. In a future step, the discussed model can be extended with a coupled electro-thermal converter model that includes the influence of raised component temperatures on the efficiency in order to accurately calculate the expected component temperatures [104, 105].

4.1 Losses Calculated with a Simple Loss Model

The most simple loss model uses the lossless electric DAB model (Figure 3.2) to determine the converter stress values listed below.

- RMS currents in the semiconductor switches, the transformer windings, and the inductor windings;
- instantaneous currents during switching;
- peak inductor currents;
- voltage-time areas applied to the transformer core.

With known stress values, the respective losses are calculated.

4.1.1 Power Switches

The calculation of the power dissipated in the switches considers two different loss mechanisms: conduction losses and switching losses.

The RMS currents through the switches determine the respective conduction losses. In steady-state operation, every switch conducts current during half a switching cycle $T_{\rm S}$ and the waveform $i_L(t)$ (cf. Figure 3.2) repeats with negative sign after one half-cycle [i.e. $i_L(t) = -i_L(t - T_{\rm S}/2)$]. Thus, each of the 4 switches on the HV side (T₁, T₂, T₃, and T₄) carries the RMS current $I_{\rm S1}$,

$$I_{\rm S1} = \frac{I_L}{\sqrt{2}},\tag{4.1}$$

and each of the 4 switches on the LV side $(T_5, T_6, T_7, and T_8)$ carries the RMS current

$$I_{S2} = \frac{n I_L}{\sqrt{2}} \tag{4.2}$$

 $[I_L \text{ is the RMS value of } i_L(t)].$

According to (4.1), the switches on the HV side generate the total conduction losses

$$P_{\rm S1,cond} = 4R_{\rm S1}I_{\rm S1}^2 \tag{4.3}$$

and due to (4.2) all switches on the LV side cause the total conduction losses

$$P_{\rm S2,cond} = 4R_{\rm S2}I_{\rm S2}^2. \tag{4.4}$$

In power electronics, the influences of high frequency skin and proximity effects are typically neglected for power semiconductor devices. The DC switch resistances of the selected MOSFETs (cf. Appendix A.2) are obtained from the data sheet values:

$$R_{\rm S1} = R_{\rm DS(on), SPW47N60CFD} = 70\,\mathrm{m}\Omega,\tag{4.5}$$

$$R_{\rm S2} = R_{\rm DS(on), IRF2804} / 8 = 2.2 \,\mathrm{m}\Omega / 8 = 275 \,\mu\Omega \tag{4.6}$$

(on the LV side, 8 MOSFETs are operated in parallel; junction temperature considered for R_{S1} and R_{S2} : $T_j = 25^{\circ}$ C).

The calculation of the switching losses is more demanding, since these not only depend on the selected power MOSFETs themselves; in particular, the surrounding parasitic components (e.g. PCB stray inductances) considerably affect the switching losses.

If ZVS (Zero Voltage Switching, Section 4.3.2) is achieved on the HV side, very low switching losses result. These can be neglected within the most simple loss model: $P_{S1,sw} = 0$. In contrast, hard switching operation of the HV MOS-FETs leads to excessive semiconductor losses and must be avoided in any



Figure 4.1: (a) LV side full bridge with the parasitic MOSFET lead inductances $L_{\rm S}$ and $L_{\rm D}$; (b) estimated switching losses on the LV side.

case (e.g. using advanced modulation methods, cf. Section 5, or additional circuitry [82, 106]).

On the LV side, soft switching (i.e. ZVS) is effectively not achieved, due to the energy stored in the parasitic drain and source lead inductances $L_{\rm D}$ and $L_{\rm S}$. Accordingly, the switching losses which occur if the condition for ZVS is satisfied can be estimated with [93]

$$E_{\rm S2,sw} = \frac{1}{2} 2 \left(L_{\rm D} + L_{\rm S} \right) I_{\rm S2,sw}^2 \frac{V_{\rm pk}}{V_{\rm pk} - V_2} \tag{4.7}$$

 $(I_{S2,sw}$ denotes the instantaneous MOSFET current at the switching instant). The parameters $V_{\rm pk}$, $L_{\rm D}$, and $L_{\rm S}$, needed to evaluate (4.7), are determined with data sheet values, calculations (e.g. finite element simulations) or measurements; for the given setup, the total parasitic inductance $L_{\rm D}+L_{\rm S}=2.4$ nH and the peak voltage $V_{\rm pk}=32.7$ V have been identified using measured switching losses. If the LV MOSFETs are operated with hard switching, comparably low switching losses occur, which is detailed in Section 4.3.3. Thus, regarding the most simple loss model, hard switching losses are neglected, there. The total estimated switching losses are shown in Figure 4.1 (b); there, $I_{\rm S2,sw} \leq 0$ denotes hard switching and $I_{\rm S2,sw} > 0$ denotes the operation with ZVS, cf. Section 4.3 and, in particular, Table 4.1, Figure 4.5, and Figure 4.10.

In steady-state, i_{L_2} [cf. Figure 4.1] repeats with negative sign after one halfcycle; thus, for the assumption of equal parasitic component values, T₅ and T₆ generate equal switching losses $E_{S2,a,sw}$; accordingly, T₇ and T₈ generate equal switching losses $E_{S2,b,sw}$ (for phase shift modulation $E_{S2,a,sw} = E_{S2,b,sw}$ applies). Consequently, for the full bridge depicted in Figure 4.1 (a), the total switching losses

$$P_{\rm S2,sw} = 2f_{\rm S} \left(E_{\rm S2,a,sw} + E_{\rm S2,b,sw} \right) \tag{4.8}$$

result.

4.1.2 Transformer, Inductor

With the elementary electric DAB model (Figure 3.2), the transformer copper losses are obtained from

$$P_{\rm tr,cond} = (R_{\rm tr1} + n^2 R_{\rm tr2}) I_L^2 \tag{4.9}$$

 $(R_{\rm tr1} \text{ and } R_{\rm tr2} \text{ denote the respective resistances of the HV side winding and of the LV side winding, determined at the switching frequency, cf. Appendix A.2.4 and Table A.9).$

The transformer core losses are calculated with the Steinmetz equation,

$$P_{\rm tr,core} \approx V_{\rm tr,core} k f_{\rm S}^{\alpha} B_{\rm tr,peak}^{\beta}, \qquad (4.10)$$

with the Steinmetz parameters k, α , and β (Appendix C: Table C.2 and Figure C.4), the total core volume $V_{\rm tr,core}$, and the peak magnetic flux density $B_{\rm tr,peak}$,

$$B_{\rm tr,peak} = \frac{\max[\Phi_{\rm tr}(t)] - \min[\Phi_{\rm tr}(t)]}{2} \frac{1}{A_{\rm tr,core}} \quad \forall \quad 0 < t \le T_{\rm S}$$
(4.11)

using

$$\Phi_{\rm tr}(t) = \int_0^t \frac{v_{\rm M}(t_{\rm int})}{N_1} \,\mathrm{d}t_{\rm int} + \Phi(0) \tag{4.12}$$

 $(N_1 \text{ is the number of turns of the HV side winding, A_{tr,core}}$ is the core cross sectional area, and $v_{\rm M}$ denotes the voltage applied to the magnetizing inductance, i.e. $v_{\rm M} \approx n v_{\rm AC2}$ for the given DAB hardware prototype, since the predominant part of the DAB inductance L is placed on the HV side, cf. Appendix A.2.5).

The copper losses of the external inductor are calculated with

$$P_{L_{\rm HV},\rm cond} = R_{L_{\rm HV}} I_L^2; \tag{4.13}$$

the calculation of the inductor copper resistance, $R_{L_{\rm HV}}$, is outlined in Appendix A.9.4. The inductor core losses are estimated with the peak inductor current $I_{L_{1,\rm peak}}$ using the Steinmetz equation,

$$P_{L_{\rm HV,core}} \approx V_{L_{\rm HV,core}} k f_{\rm S}^{\alpha} B_{L_{\rm HV,peak}}^{\beta}, \qquad (4.14)$$

with

$$B_{L_{\rm HV},\rm peak} \approx \frac{\Phi_{L_{\rm HV},\rm peak}}{N_{L_{\rm HV}}A_{L_{\rm HV},\rm core}} = \frac{L_{\rm HV}I_{L_1,\rm peak}}{N_{L_{\rm HV}}A_{L_{\rm HV},\rm core}}$$
(4.15)

 $(N_{L_{HV}}: \text{ number of turns}, A_{L_{HV}, \text{core}}: \text{ inductor core cross sectional area}).$

4.1.3 Total Losses – Predicted Efficiency

The total losses include all the discussed power losses as well as the auxiliary power required on the HV side $(P_{1,aux})$ and on the LV side $(P_{2,aux})$: the HV side auxiliary supply powers the DSP plus the HV side gate drivers and the LV side auxiliary supply provides power to the LV side gate drivers. Approximately constant power consumption has been measured ($f_{\rm S} = 100 \text{ kHz} = \text{const.}$),

$$P_{1,\text{aux}} = 6.2 \text{ W} \text{ and } P_{2,\text{aux}} = 9.5 \text{ W}.$$
 (4.16)

Thus, the total losses $P_{\rm t}$ are calculated with

$$P_{\rm t} = P_{1,\rm aux} + P_{\rm S1,\rm cond} + P_{\rm S1,\rm sw} + P_{L_{\rm HV},\rm cond} + P_{L_{\rm HV},\rm core} + P_{\rm tr,\rm cond} + P_{\rm tr,\rm core} + P_{\rm S2,\rm cond} + P_{\rm S2,\rm sw} + P_{2,\rm aux}.$$
(4.17)

The most simple loss model evaluates all required charcteristics (e.g. RMS current values) at a given input power, $P_{\rm in}$, in order to include the impact of the losses on these quantities. The efficiency $\eta = P_{\rm out}/P_{\rm in}$ is then calculated with the output power $P_{\rm out} = P_{\rm in} - P_{\rm t}$. The result (Figure 4.2, dotted lines) shows, that a very poor matching is obtained between the losses calculated with this simple model and measured losses. Main reasons are:

- 1. The converter quantities (e.g. switch RMS currents) calculated with the lossless DAB model (Figure 3.2) significantly deviate from measured values, since losses are completely neglected, there.
- 2. Inaccurate modeling of the conduction losses leads to a wrong estimation of the dissipated power. This occurs mainly on the LV side, where high currents at high frequencies generate considerably more losses than predicted with (4.17).
- 3. Switching losses cause a large part of the total losses and therefore the switching losses need to be known in detail. It further turns out, that, depending on the operating point, the switching action may change the transformer current considerably.
- 4. Due to losses that are not considered in the electric DAB model (e.g. switching losses, core losses), the input and output power levels $P_{\rm in}$ and $P_{\rm out}$, calculated with the electric DAB model, are different to the input and output power levels obtained with the DAB hardware



Figure 4.2: Calculated and measured DAB efficiencies for different operating points [negative power: power transfer from the LV port to the HV port, cf. (3.14)], phase shift operation, $L = 26.7 \,\mu\text{H}$, and n = 19; dotted lines: conventional analysis (Section 4.1), dashed lines: includes accurate conduction losses (Section 4.2), dashed and dotted lines: accurate conduction and switching losses (Section 4.3); the boxes, \Box , indicate measured values.

prototype. Consequently, the calculation uses slightly different control parameters in order to obtain the specified output power, which results in inaccurate DAB converter quantities (e.g. switch RMS currents).

4.2 Accurate Conduction Losses, High Frequency Effects

The accurate calculation of the power dissipation employs the detailed DAB converter model depicted in Figure 3.31.

Still, the conduction losses of the HV MOSFETs are calculated using the onstate resistances of the selected MOSFETs and the RMS current I_{L_1} [cf. (4.1)]:

$$P_{\rm S1,cond} = 4R_{\rm S1}I_{\rm S1}^2; \quad I_{\rm S1} = \frac{I_{L_1}}{\sqrt{2}}.$$
 (4.18)

Measurements confirm a considerable impact of high frequency effects on the resistances of the inductor winding, the transformer windings, the PCB, and the LV MOSFETs (Figure 4.3, Figure 4.4, and Figure A.21, Figure A.23, and Figure A.59). However, the copper losses generated by the inductor have a low proportion of the total losses and are calculated with the RMS current I_{L_1} and the DC resistance of the litz wire winding (cf. Appendix A.9.4):

$$P_{L_{\rm HV},\rm cond} = R_{L_{\rm HV}} I_{L_1}^2.$$
(4.19)

The accurate loss model considers the current harmonics to determine the copper losses of the transformer and the PCB and the conduction losses of the LV MOSFETs:

$$P_{\rm tr1,cond} = \sum_{k=1}^{m} R_{\rm tr1}(kf_{\rm S}) [I_{L_1}(kf_{\rm S})]^2$$
(4.20)

$$P_{\rm tr2,cond} = \sum_{k=1}^{m} R_{\rm tr2} (kf_{\rm S}) [I_{L_2}(kf_{\rm S})]^2$$
(4.21)

$$P_{\rm PCB,a,cond} = \sum_{k=1}^{m} R_{\rm PCB,a}(kf_{\rm S}) [I_{L_2}(kf_{\rm S})]^2$$
(4.22)

$$P_{\rm S2,cond} = \sum_{k=0}^{m} R_{\rm S2}(kf_{\rm S}) [I_{\rm S2}(kf_{\rm S})]^2$$
(4.23)

$$P_{\rm PCB,b,cond} = \sum_{k=0}^{m} R_{\rm PCB,b} (kf_{\rm S}) [I_2(kf_{\rm S})]^2$$
(4.24)

using

$$I_{S2} = \frac{I_{L_2}}{\sqrt{2}}.$$
 (4.25)

[for k > 0, $I(kf_{\rm S})$ is the RMS value of the k-th harmonic of i(t), i.e. $I(kf_{\rm S}) = I_{\rm peak}(kf_{\rm S})/\sqrt{2}$; I(0) represents the DC component of i(t)]. The calculated



Figure 4.3: AC resistances measured at $T = 25^{\circ}$ C for (a) $R_{PCB,a}$ and (b) the sum $R_{PCB,a} + R_{PCB,b}$ ($R_{PCB,a}$ and $R_{PCB,b}$ are shown in Figure 3.24); the measurement employs a 13:1 measurement transformer in order to obtain reasonable impedance values for the Agilent 4294A Precision Impedance Analyzer; at 100 kHz the calculated maximum measurement error is $\pm 13\%$ for $R_{PCB,a}$ and $\pm 7\%$ for ($R_{PCB,a} + R_{PCB,b}$) (calculated according to [152]). The values shown for $R_{PCB,a}(f)$ are the average values of the resistances measured if T₅ and T₇ are replaced by a short circuit [$R_{PCB,a1}(f)$] and the resistances measured if T₆ and T₈ are replaced by a short circuit [$R_{PCB,a2}(f)$]; the measurement results obtained for $R_{PCB,a1}(f)$ and $R_{PCB,a2}(f)$ are nearly equal. The values shown for $R_{PCB,a}(f) + R_{PCB,b}(f)$ are the average values of $R_{PCB,ab1}(f)$ (T₅ and T₈ are replaced by a short circuit) and $R_{PCB,ab2}(f)$ (T₆ and T₇ are replaced by a short circuit); the measurement results obtained for $R_{PCB,ab1}(f)$ (T₆ and T₈ are replaced by a short circuit of $R_{PCB,ab2}(f)$ (T₆ and T₇ are replaced by a short circuit); the measurement results obtained for $R_{PCB,ab1}(f)$ and $R_{PCB,ab2}(f)$ are nearly equal.



Figure 4.4: AC resistances measured at $T = 25^{\circ}$ C for the on-state resistance of a single IRF2804 MOSFET with a gate to source voltage of 10 V (provided from a battery); the on-state resistance is measured at the drain and source pins of the MOSFET, with a distance of 6 mm to the package; the measurement employs a 13:1 measurement transformer in order to obtain reasonable impedance values for the Agilent 4294A Precision Impedance Analyzer; at 100 kHz the calculated maximum measurement error of $R_{DS(on),IRF2804}$ is $\pm 3\%$ (calculated according to [152]).

results include the harmonic components up to m = 20. The calculation of the transformer winding resistances $R_{tr1}(kf_S)$ and $R_{tr2}(kf_S)$ is outlined in Appendix A.2.4 and is based on measurement values (Figure A.21, Figure A.23).

4.3 Accurate Switching Losses

4.3.1 Measurement Setups

Besides the conduction losses, the switching losses amount to a significant part of the total losses. Therefore, a precise description of the switching losses – obtained from measurements – is required in order to accurately determine the power dissipated in the semiconductor switches.

HV Side

The switching losses of a single half bridge [MOSFETs T_1 and T_2 in Figure 4.5 (a)] are measured on the final converter PCB to obtain accurate results. Moreover, the MOSFETs are mounted on a heatable metal plate (temperature controlled) in order to measure the switching losses at different



Figure 4.5: (a) Switching loss measurement setup, HV side; (b) employed double pulse signal; (c) respective gate signals for T_1 and T_2 .

junction temperatures T_j (for the switching loss measurement, equal case and junction temperatures are assumed).

During the switching process, a significant amount of the stored energy may be transferred from one switch to another. Therefore, the drain currents i_{D,T_1} and i_{D,T_2} and the blocking voltages v_{DS,T_1} and v_{DS,T_2} of the respective half bridge need to be measured simultaneously in order to calculate the energies dissipated during switching:

$$E_{\rm T_1} = \int_{t_{\rm begin}}^{t_{\rm end}} v_{\rm DS,T_1} \, i_{\rm D,T_1} dt \quad \text{and} \quad E_{\rm T_2} = \int_{t_{\rm begin}}^{t_{\rm end}} v_{\rm DS,T_2} \, i_{\rm D,T_2} dt. \tag{4.26}$$

The instants t_{begin} and t_{end} denote the beginning and the end of the switching process, respectively; t_{begin} and t_{end} are selected such that a decrease of t_{begin} and an increase of t_{end} do not change the measured switching losses.

In steady-state operation each switch of the DAB converter is turned-on during one half-cycle; moreover, the signs of the transformer currents change after each half-cycle. Therefore, identical turn-on and turn-off switching operations of the MOSFETs of a single half bridge result (provided that equal MOSFETs are used for a single half bridge). The proposed switching loss measurement correctly accounts for the generated switching losses:

- if stored energy is transferred from one MOSFET to another (soft switching, Section 4.3.2) the released energy, which has been absorbed in the preceding half-cycle, appears as negative switching losses and the absorbed energy as positive losses (Figure 4.12); for the assumption of equal MOSFETs, the difference between released and absorbed energies is the dissipated energy;
- during hard switching operation (Section 4.3.2) the measurement according to (4.26) does not account for the dissipation of the energy stored in the MOSFET that is turned on (e.g. T₁). However (4.26) includes the energy stored in the remaining MOSFET (e.g. in T₂), which will be dissipated during the switching operation of the subsequent halfcycle. Thus, for the assumption of equal MOSFETs, the correct total switching losses are measured.

The measurement of the switching losses employs the double pulse signal depicted in Figure 4.5 (b), which consists of 4 different time intervals.

- Time interval I is used to generate the required current $i_{L_1}(t_{1a})$; at $t = t_{1a}$, the condition for Zero Voltage Switching (ZVS, cf. Section 4.3.2) is fulfilled.
- During the free wheeling time interval II, the current i_{L_1} remains approximately constant.
- During time interval III, i_{L_1} increases again; at $t = t_{2b}$, the condition for hard switching is fulfilled (Section 4.3.2).
- For $t > t_{3b}$, the current i_{L_1} free wheels through T₂ and eventually decreases to zero.

In order to avoid a current shoot-through in the half bridge, a dead time, $T_{\text{deadtime}} = 200 \text{ ns}$, is used [Figure 4.5 (c)].

The switching loss measurement employs a digital oscilloscope (LeCroy WavePro 950) to determine the energy dissipated during the switching process, high voltage probes (100:1, 2 kV) to measure the drain-to-source voltages, and wide bandwidth current sensors to measure the drain currents. The schematics of the employed current sensors is depicted in Figure 4.6. With the employed MOSFETs (SPW47N60CFD), very fast voltage and current transients occur (cf. Figure 4.12) and therefore, the frequency response of the employed current sensors is measured using the measurement setup depicted in Figure 4.7. Figure 4.8 illustrates the equivalent circuit of this measurement



Figure 4.6: Schematics of the current sensors employed to measure the drain currents i_{D,T_1} and i_{D,T_2} . The current transformer is made of two stacked ferrite cores (R6.3, material: T38) and 27 turns (wire diameter: 0.3 mm) on the secondary side. The depicted capacitors are used to compensate the frequency response of the current sensor and a common mode choke is used to suppress common mode noise; the $47 \Omega + 1 \Omega$ resistors approximately provide a 50Ω termination. The respective channels of the oscilloscope are operated with an input impedance of 50Ω .



Figure 4.7: Measurement setup used to determine the frequency response of the employed current sensors: a network analyzer generates a frequency sweep and measures the output power of the current sensor at the excitation frequency (port A). Prior to the measurement, the network analyzer is calibrated with a coaxial cable of the correct length (i.e. the length of the coaxial cable connected between the oscillator and the primary side of the current sensor) in order to calibrate the magnitude measurement and to eliminate the effect of the propagation time delay caused by the input cable; the cable connected between the current sensor and port A is part of the current sensor circuit itself and thus, the respective propagation time delay needs to be considered.



Figure 4.8: Equivalent circuit of the measurement setup depicted in Figure 4.7 including the current sensor circuit of Figure 4.6. The input impedance of measurement port A is 50Ω .

setup: the oscillator of the network analyzer generates a frequency sweep and at the measurement port A the output voltage $v_i(t)$ is measured with respect to magnitude and phase. The primary side of the current sensor is terminated with 50Ω since the primary side input impedance of the current sensor is much smaller than 50Ω . The sensor gain, which is expected to be measured with this setup (oscillator power: $P_{\rm RF} = 0 \, \text{dBmW}$) and with the employed current sensor, is calculated based on the equivalent circuit shown in Figure 4.8 using the RMS values I and V_i of i(t) and $v_i(t)$, respectively:

$$I = \sqrt{\frac{P_{\rm RF}}{50\,\Omega}} = \sqrt{\frac{0\,\mathrm{dBmW}}{50\,\Omega}} = \sqrt{\frac{1\,\mathrm{mW}}{50\,\Omega}} = 4.47\,\mathrm{mA},\tag{4.27}$$

$$V_i = \frac{1}{27} \cdot 1 \,\Omega \cdot \qquad \underbrace{\frac{1}{2}}_{2} = 82.8 \,\mu \text{V}, \qquad (4.28)$$

resistive voltage divider due to the

 50Ω termination

$$\frac{P_i}{P_{\rm RF}} = \frac{82.8\,\mu{\rm V}^2/50\,\Omega}{1\,\rm mW} = 137\,\rm nW = -68.6\,\rm dB. \tag{4.29}$$

The capacitors depicted in Figure 4.6 are used to improve the frequency response (Figure 4.9) of the employed current sensor: for frequencies below 50 MHz a gain error of less than ± 0.5 dB occurs; with a gain error of ± 3 dB being tolerated, the current sensor can be used for frequencies of up to 100 MHz.

The time skew between voltage and current measurements is measured with the same oscilloscope the switching losses are measured with. For the time skew, a value of 1 ns is determined (the current measurement slightly lags behind the voltage measurement).



Figure 4.9: Frequency response of the current sensors (Figure 4.6) measured with the setup depicted in Figure 4.7; (a) gain response, (b) phase response.

LV Side

The LV side switching losses are measured using one of the two full bridges of the final converter with 4 MOSFETs being operated in parallel (Figure 4.10);¹ the MOSFETs are mounted on a heatable and temperature controlled metal plate. Again, the double pulse signal [cf. Figure 4.5 (b)] is employed with a dead time of 240 ns. For each of the MOSFETs, the drain current and the

 $^{^1\}mathrm{On}$ the LV side, the final converter operates two full bridges in parallel, cf. Appendix A.2.3.



Figure 4.10: Switching loss measurement setup, LV side; for the switching loss measurement, one of the two full bridges (cf. Appendix A.2.3) is employed, i.e. 4 MOSFETs are connected in parallel to form one switch. On the assumption of equal MOSFETs $T_{5,A} \ldots T_{5,D}$ and $T_{6,A} \ldots T_{6,D}$, the parasitic inductances $2L_{D,T_5} = L_{D,T_5,A} ||L_{D,T_5,B}||L_{D,T_5,D}||L_{D,T_5,D}$, $2L_{S,T_5} = L_{S,T_5,A} ||L_{S,T_5,B}||L_{S,T_5,C}||L_{S,T_5,D}$ etc. result. Consequently, the total parasitic inductances are L_{D,T_5} , L_{S,T_5} , L_{D,T_6} , and L_{S,T_6} .



Figure 4.11: Schematics of the current sensors employed to measure the drain currents $i_{D,T_{5,A}}$, $i_{D,T_{5,B}}$, $i_{D,T_{5,C}}$, and $i_{D,T_{5,D}}$ and $i_{D,T_{6,A}}$, $i_{D,T_{6,B}}$, $i_{D,T_{6,C}}$, and $i_{D,T_{6,D}}$. The current transformer is made of a ferrite core (R6.3, material: T38) and 27 turns (wire diameter: 0.3 mm) on the secondary side. The 51 Ω ||2.2 k Ω resistors provide a 50 Ω termination. The respective channels of the oscilloscope are operated with an input impedance of 50 Ω .

drain-to-source voltage is measured simultaneously in order to determine the respective switching losses.

The switching loss measurement employs a digital oscilloscope (LeCroy WavePro 950) to determine the energy dissipated during the switching process, conventional voltage probes (10 : 1, 600 Vpk) to measure the drain-to-source voltages, and current sensors to measure the drain currents. The schematics of the employed current sensors is depicted in Figure 4.11.²

The time skew between voltage and current measurements is measured with the same oscilloscope the switching losses are measured with. For the time skew, a value of 2 ns is determined (the voltage measurement slightly lags behind the current measurement).

4.3.2 HV Side Switching: Results

Soft Switching (ZVS)

Soft switching occurs at $t = t_{1a}$ in Figure 4.5 (b). There, the upper switch, T_1 , is turned off with current i_{D,T_1} in forward direction and thus, its body diode blocks. Consequently, the inductor current, i_{L_1} , charges the drain-to-source capacitance of T_1 and discharges the respective capacitance of T_2 (Figure 4.12). This process continues until either the body diode of T_2 starts to conduct or until T_2 is turned on.

 $^{^2{\}rm The}$ frequency response of these current sensors has not been compensated, since the employed LV MOSFETs (IRF2804) cause considerably slower voltage and current transients than the employed HV MOSFETs

Very low switching losses are achieved with this kind of operation, since T_1 is turned off with $v_{DS,T_1} \approx 0$ (zero voltage turn-off),³ and – on the assumption of $T_{deadtime} = t_{1b} - t_{1a}$ being selected long enough – T_2 is turned on with $v_{DS,T_2} = -v_D \approx 0$ (zero voltage turn-on; typically, the losses due to the forward voltage drop v_D of the body diode can be neglected during the short time the diode conducts). Thus, this switching operation is termed Zero Voltage Switching (ZVS) or soft switching (Figure 4.12).

Hard Switching

At $t = t_{2a}$, the lower switch depicted in Figure 4.5, T₂, is turned off, however, its body diode continues to conduct due to the negative value of t_{D,T_2} . The MOSFET T₁ is turned on at $t = t_{2b}$ and the drain current of T₁ starts to increase after the turn-on time delay expired at $t = t_{2c} > t_{2b}$ (Figure 4.13). The current through the body diode of T₂ accordingly decreases, crosses zero at $t = t_{2d} > t_{2c}$, becomes negative due to reverse recovery effects, and reaches the peak of its reverse recovery current at $t = t_{2e}$ in Figure 4.13. At $t = t_{2e}$ the diode starts to block and the reverse current, which is still present in T₂, decreases to zero.

This switching operation is referred to as hard switching operation and typically generates high switching losses (cf. Figure 4.13: there, a maximum instantaneous power dissipation of more than 16 kW is measured at $t = t_{2e}$, $V_1 = 400$ V and $I_{S1,sw} = -10$ A). During hard switching, due to parasitic line inductance, quickly changing drain currents cause excess voltages superimposed on the nominal drain-to-source voltages of the blocking semiconductor switches, which may damage these semiconductors. Furthermore, high voltage slopes $dv_{DS,T_1}/dt$ and $dv_{DS,T_2}/dt$ (up to $80 \text{ kV}/\mu$ s are observed with the given hardware setup) and ringing on v_{DS,T_1} and v_{DS,T_2} causes serious EMI issues.

Hard switching as well occurs if the inductor current i_{L_1} is zero during switching, since the drain-to-source capacitances of the respective MOSFETs need to be charged or discharged. Therefore, with $i_{L_1} = 0$, the drain-to-source voltages v_{DS,T_1} and v_{DS,T_2} remain constant during the dead time interval; after the dead time interval has elapsed, the activated switch provides the required current to change v_{DS,T_1} and v_{DS,T_2} , which particularly causes turnon losses (Figure 4.14).

³Zero voltage turn-off is achieved due to the parasitic drain-to-source capacitances of the employed MOSFETs. The non-linear dependence of these capacitances on the respective drain-to-source voltages additionally supports the successful practical implementation of zero voltage turn-off.


Figure 4.12: HV side switch currents and voltage waveforms for soft switching operation, $V_1 = 400$ V and $I_{S1,sw} = 20$ A: (a) T₁ and (b) T₂. During the switching process, negative switching losses occur for T₂, since the inductor L discharges the parasitic drain-to-source capacitance of T₂.



Figure 4.13: HV side switch currents and voltage waveforms for hard switching operation, $V_1 = 400$ V and $I_{S1,sw} = -10$ A: (a) T₁ and (b) T₂. The vertical dashed line indicates the turn-on time of T₁, t_{2c} [> t_{2b} in Figure 4.5 (c)], identified from the rising switch currents. At $t = t_{2d}$ the switch current i_{D,T_2} approaches zero, thus diode reverse recovery occurs during $t_{2d} < t < t_{2e}$. At $t = t_{2e}$, the drain current $i_{D,T_1} \approx 44$ A causes an instantaneous power dissipation of more than 16 kW in T₁; the peak voltage slope of $dv_{DS,T_1}/dt$ reaches ≈ 80 kV/µs.



Figure 4.14: HV side switch currents and voltage waveforms for hard switching operation, $V_1 = 400$ V and $I_{S1,sw} = 0$: (a) T₁ and (b) T₂; a peak drain current $i_{D,T_1,max}$ of ≈ 14 A is required to charge the parasitic drain-to-source capacitance of T₂, which causes a maximal instantaneous power dissipation of ≈ 5 kW in T₁; the peak voltage slope $dv_{DS,T_1}/dt$ reaches ≈ 30 kV/µs (cf. Figure 4.13).

HV Side, Edge Type of $v_{\rm AC1}$	$I_{ m S1,sw}$	$s_{ m edge}$
Rising edge of $v_{AC1}(t = t_{sw})$	$I_{\rm S1,sw} = -i_{L_1}(t_{\rm sw})$	$s_{\rm edge} = -1$
Falling edge of $v_{\rm AC1}(t = t_{\rm sw})$	$I_{\rm S1,sw} = +i_{L_1}(t_{\rm sw})$	$s_{\rm edge} = +1$
$LV Side, Edge Type of v_{AC2}$	$I_{ m S2,sw}$	$s_{ m edge}$
$\frac{LV Side, Edge Type of v_{AC2}}{\text{Rising edge of } v_{AC2}(t = t_{sw})}$	$I_{\rm S2,sw}$ $I_{\rm S2,sw} = +i_{L_2}(t_{\rm sw})$	$\frac{s_{\rm edge}}{s_{\rm edge} = +1}$

Table 4.1: Expressions required to obtain $I_{S1,sw}$ and $I_{S2,sw}$ from the respective instantaneous inductor currents $i_{L_1}(t_{sw})$ and $i_{L_2}(t_{sw})$ at a switching instant t_{sw} ; e.g. in Figure 3.4, at t = 0, a rising edge of v_{AC1} occurs and thus, the HV side full bridge switches the current $I_{S1,sw} = -i_{L_1}(0) \approx i_L(0)$]. The respective sign adjustment [e.g. in (4.30)] employs s_{edge} .

Measured Switching Losses

In Figure 4.15, the sum $E_{\rm T_1} + E_{\rm T_2} = E_{\rm S1,a,sw}$ is shown for a single switching process and for different switching currents $I_{\rm S1,sw}$, different operating voltages V_1 , and a junction temperature, T_j , of 25°C. Negative switching currents in Figure 4.15 indicate hard switching operation of the half bridge circuit where significant losses occur due to the slow MOSFET body diodes. For positive switching currents greater than 2 A, the circuit is successfully operated with ZVS giving very low switching losses. Low currents (0 A $\leq I_{\rm S1,sw} \leq 2$ A) are insufficient to charge and discharge the respective drain-to-source capacitances within the dead time interval, which causes the turn-on losses to increase. Figure 4.16 depicts the switching losses measured at $T_j = 125^{\circ}$ C; due to the raised component temperatures, the switching losses increase by $\approx 30\% \dots 40\%$. The instants $t_{\rm begin}$ and $t_{\rm end}$, needed to evaluate (4.26), are selected such that neither a reduction of $t_{\rm begin}$ nor an increase of $t_{\rm end}$ changes the measured switching losses; the duration of the measurement time interval $t_{\rm end} - t_{\rm begin}$ is between 100 ns and 300 ns.

Table 4.1 summarizes the expressions required to determine $I_{\rm S1,sw}$ from the actual inductor current $i_{L_1}(t_{\rm sw})$ at the switching instant $t_{\rm sw}$.

Detailed Consideration of the Switching Losses

Depending on the switching current of the full bridge, two different polynomial functions – one for hard switching and one for soft switching – are fitted to the switching losses using a least mean square approximation. However, the switching losses are measured with constant inductor current $i_{L_1}(t)$ during



Figure 4.15: HV side switching losses measured at $T_{\rm j} = 25^{\circ}$ C for a single switching action. $I_{\rm S1,sw}$ is the instantaneous current during switching: $I_{\rm S1,sw} < 0$ denotes hard switching and for $I_{\rm S1,sw} > 0$ the ZVS condition is satisfied.



Figure 4.16: HV side switching losses measured at $T_{\rm j} = 125^{\circ}$ C for a single switching action. $I_{\rm S1,sw}$ is the instantaneous current during switching: $I_{\rm S1,sw} < 0$ denotes hard switching and for $I_{\rm S1,sw} > 0$ the ZVS condition is satisfied.

the dead time interval what may not be true in practice. Thus, wrong results are obtained for certain operating points, if the current calculated at the switching instant, $I_{S1,sw}$, is directly applied to the switching loss function [Figure 4.17 (c)]. In order to improve the results, the soft switching case and the hard switching case need to be looked at separately, due to essentially different switching processes.

For the soft switching case ($I_{S1,sw} > 2$ A in Figure 4.15), the voltage v_{AC1} changes within the dead time interval, whereas a certain charge must be supplied in order to achieve ZVS [i.e. to charge and discharge the respective MOSFET junction capacitances, Figures 4.17 (a) and (b)],

$$Q(V_1) = s_{\text{edge}} \cdot \int_{t_{\text{off}}}^{t_{\text{edge}}} i_{L_1}(t) \, \mathrm{d}t$$

$$(4.30)$$

($t_{\rm off}$ denotes the MOSFET turn-off time and $t_{\rm edge}$ is the time when $v_{\rm AC1}$ changes; $s_{\rm edge}$ denotes the sign adjustment according to Table 4.1). The amount of charge depends on the employed power MOSFETs, e.g. $Q(V_1) \approx 220 \,\mathrm{nC} + V_1 \cdot 218 \,\mathrm{pF}$ for the SPW47N60CFD at room temperature (estimated using data sheet information). With known $Q(V_1)$, $i_{L_1}(t)$, and $t_{\rm edge}$, the turn-off time $t_{\rm off}$, that is required in order to fully charge and discharge the MOS-FET drain-to-source capacitances, can be calculated.⁴

For the hard switching case ($I_{\rm S1,sw} < 0$ in Figure 4.15), the voltage $v_{\rm AC1}$ changes after the dead time interval has elapsed. On the assumption of negligible turn-on and turn-off delays, the MOSFET turn-off time is determined with $t_{\rm off} = t_{\rm edge} - T_{\rm deadtime}$.

It finally turns out that the calculated switching losses become considerably more accurate if the switching loss function employs the average of the current i_{L_1} , evaluated over the time interval $t_{\text{off}} < t < t_{\text{edge}}$ [Figure 4.17 (b) and (c)],

$$\overline{I}_{S1,sw} = \frac{s_{edge} \cdot \int\limits_{t_{off}}^{t_{edge}} i_{L_1}(t) dt}{t_{edge} - t_{off}} .$$

$$(4.31)$$

The resulting switching losses of the HV side full bridge comprise of the switching losses caused by the two half bridges T_1 , T_2 and T_3 , T_4 ,

$$P_{\rm S1,sw} = 2f_{\rm S} \left[E_{\rm S1,sw} \left(\overline{I}_{\rm S1,a,sw} \right) + E_{\rm S1,sw} \left(\overline{I}_{\rm S1,b,sw} \right) \right]$$
(4.32)

 $(\overline{I}_{S1,a,sw}$ denotes the average switching current of T_1 and T_2 ; $\overline{I}_{S1,b,sw}$ denotes the average switching current of T_3 and T_4 ; for phase shift modulation $\overline{I}_{S1,a,sw} = \overline{I}_{S1,b,sw}$ applies).

⁴Whenever $t_{\rm off} < t_{\rm edge} - T_{\rm deadtime}$ occurs, the injected charge during the dead time interval is not sufficient to achieve soft switching (0 < $I_{\rm S1,sw}$ < 2A in Figure 4.15); consequently, $t_{\rm off}$ needs to be limited to $t_{\rm edge} - T_{\rm deadtime}$.



Figure 4.17: Implemented HV side switch current estimation, illustrated for the results obtained from a first converter design with n = 24 and $L = 30.8 \,\mu\text{H}$. There, at low power levels ($|P_1| < 1000 \,\text{W}$), the instantaneous inductor current during HV side switching is insufficient to fully charge and discharge the respective MOSFET drain-to-source capacitances, which causes the HV side switching losses to increase. (a) Calculated voltage and current waveforms for $V_1 = 240 \,\text{V}$, $V_2 = 11 \,\text{V}$, and $P_1 = -2 \,\text{kW}$ (power is transferred from the LV port to the HV port); (b) the magnified time interval $t_0 < t < t_1$ depicts the turn-off process for a HV MOSFET with involved charge $Q(V_1)$ and the current $\overline{I}_{\text{S1,sw}}$: the MOSFET gate is turned off at t_{off} and the drainto-source voltage changes at t_{edge} ; (c) achieved improvement of the efficiency prediction (solid line: calculation uses $\overline{I}_{\text{S1,sw}}$, dashed line: $i_{L_1}(t_{\text{edge}})$ is used to determine the switching losses).

Measured Time Delays

Gate driver delays, internal delays of the semiconductor switches, and capacitive charging processes cause a total time delay which needs to be considered in order to achieve the designated transformer currents and the required output power with the modulation parameters $(D_1, D_2, \text{ and } \varphi)$ of the modulation schemes discussed in Chapter 3 and Chapter 5 [103]. The employed measurement setup [Figure 4.18 (a)] allows the measurement of the total time delay between the initiation of the switching process and the actual change of the respective drain-to-source voltage, using the control signals depicted in Figure 4.18 (d).

The measured switching time delays predominantly depend on the switching current $I_{\rm S1,sw}$ (= I_{L_1} in Figure 4.18); the minimal time delay, $T_{\rm S1,d,min} \approx$ 100 ns, is achieved for ZVS operation with maximal current (i.e. $I_{\rm S1,sw} = 25$ A in Figure 4.19); maximal time delay, $T_{\rm S1,d,max} \approx 350$ ns, occurs for hard switching operation with maximal negative current (i.e. $I_{\rm S1,sw} = -10$ A in Figure 4.19). In the range 2 A < $I_{\rm S1,sw} < 7$ A, the relation between $T_{\rm S1,d}$ and $I_{\rm S1,sw}$ is highly non-linear, since the time to charge and discharge the drain-tosource capacitances of the MOSFETs increases with reduced currents $I_{\rm S1,sw}$.

Measurements with different voltages V_1 within the specified voltage range, 240 V $\leq V_1 \leq 450$ V, show virtually unchanged time delays $T_{\rm S1,d}$; thus, $T_{\rm S1,d}$ is considered independent of V_1 . Increased semiconductor temperatures cause the time delays to increase slightly, i.e. if the junction temperature rises from $T_{\rm j} = 25^{\circ}$ C to $T_{\rm j} = 125^{\circ}$ C, the time delay increases by ≈ 20 ns in the soft switching range ($I_{\rm S1,sw} > 0$) and by ≈ 15 ns in the hard switching range (Figure 4.19).

4.3.3 LV Side Switching: Results

Soft Switching

With the double pulse signal being applied to the half bridge depicted in Figure 4.10, the switch T_5 is turned off at $t = t_{1a}$ with positive drain current, $i_{D,T_5} > 0$. Thus, the body diode of T_5 blocks and the inductor current \tilde{i}_{L_2} charges and discharges the parasitic drain-to-source capacitances of T_5 and T_6 , respectively. Due to the low voltage and high current operation, however, the parasitic stray inductances L_{D,T_5} , L_{S,T_5} , L_{D,T_6} , and L_{S,T_6} (sums of the lead inductances, the inductances of the bonding wires, and the inductances of the wiring on the PCB) cause the drain-to-source voltage of T_5 to considerably exceed the supply voltage (Figure 4.20). For T_5 , a large drain-to-source voltage and a large drain current occur simultaneously and thus, large turnoff losses result (Figure 4.22). Consequently, compared to the soft switching process on the HV side discussed in Section 4.3.2, ZVS operation is effec-



Figure 4.18: (a) Setup employed to measure the time delay between the turn-off gate signal (i.e. gate signal of T_1) and the edge of the bridge output voltage (i.e. v_{DS,T_1} or v_{DS,T_2}); (b) typical waveforms, ZVS operation; (c) typical waveforms, hard switching operation; (d) employed gate signals for T_1 and T_2 .

tively not achieved for the LV side full bridge.⁵ For high switching currents,

⁵The switching losses in the ZVS range can be reduced by increasing the drain-tosource capacitances of the MOSFETs. However, since the inductances of bonding wires and connecting leads are a large proportion of the total parasitic inductances $L_{\rm D,T_5}$ + $L_{\rm S,T_5}$, $L_{\rm D,T_6} + L_{\rm S,T_6}$, $L_{\rm D,T_7} + L_{\rm S,T_7}$, and $L_{\rm D,T_8} + L_{\rm S,T_8}$ (cf. Figure 4.10), the additional capacitances would be required to be closely placed to the chips of the MOSFETs. If the



Figure 4.19: Measured HV side switching time delays with respect to the switching current $I_{S1,sw}$, constant supply voltage, $V_1 = 340$ V, and different junction temperatures T_j .

 $i_{L_2}(t_{1a}) \gg 0$, the switching losses due to the parasitic inductances can be estimated with (4.7).

Hard Switching

At $t = t_{2a}$ [Figure 4.5 (b) and Figure 4.10], T₆ is turned off with negative drain current, $i_{D,T_6} < 0$; hence, the body diode of T₆ conducts during $t_{2a} < t < t_{2b}$. At $t = t_{2b}$, T₅ is turned on and, due to the parasitic stray inductances, $|i_{D,T_6}|$ decreases and $|i_{D,T_5}|$ increases (Figure 4.21), i.e. the magnetic energy stored in L_{D,T_6} and L_{S,T_6} is transferred to L_{D,T_5} and L_{S,T_5} . Reverse recovery effects still occur and mainly cause ringing; the resulting reverse recovery losses are comparatively small, due to the low operating voltage of the full bridge, the small MOSFET reverse recovery charge ($Q_{rr} \leq 100 \text{ nC}$ at room temperature and $di/dt = -100 \text{ A}/\mu\text{s}$), and the parasitic stray inductances, which limit the slew rates of i_{D,T_5} and i_{D,T_6} . Thus, in contrast to the HV side (Section 4.3.2), the full bridge on the LV side generates low losses during hard switching (cf. Figure 4.22).

additional capacitors are solely placed in parallel to the packaged MOSFETs, only a minor improvement of the switching losses is achieved with the existing hardware prototype (e.g. with $C = 4.7 \,\mathrm{nF}$ in parallel to each LV MOSFET, the switching losses decrease by $\approx 10\%$).



Figure 4.20: LV side switch currents and voltage waveforms for soft switching operation, $V_1 = 12$ V and $I_{S2,sw} = 200$ A (50 A per switch): (a) T₅ and (b) T₆; the drain-to-source voltage of T₆, v_{DS,T_6} , considerably exceeds the supply voltage due to parasitic stray inductances.



Figure 4.21: LV side switch current and voltage waveforms for hard switching operation, $V_2 = 12$ V and $I_{S2,sw} = -200$ A (-50 A per switch): (a) T₅ and (b) T₆; the vertical dashed line indicates the turn-on time of T₅, t_{2c} ($> t_{2b}$ in Figure 4.5), identified from the rising switch currents. At $t = t_{2d}$ the switch current i_{D,T_5} is zero and during $t_{2d} < t < t_{2e}$, diode reverse recovery occurs and causes ringing. The drain-to-source voltage of T₅, v_{DS,T_5} , is considerably larger than zero during $t_{2c} < t < t_{2e}$ (mainly due to the employed gate to source voltage of 10 V: with a higher gate voltage, a reduction of v_{DS,T_5} could be achieved during $t_{2c} < t < t_{2e}$ and therefore, lower turn-on losses would result).



Figure 4.22: Measured switching losses at $T_{\rm j} = 25^{\circ}$ C for a single LV side switching action. $I_{\rm S2,sw}$ is the instantaneous current during switching; $I_{\rm S2,sw} < 0$ denotes hard switching and for $I_{\rm S2,sw} > 0$ the ZVS condition is satisfied; $I_{\rm S2,sw,opt}$ denotes the switch current with respect to minimal switching losses, needed in Section 5.2.2.

Measured Switching Losses

Figure 4.22 presents the measured and scaled switching losses of one half bridge of the fully equipped LV side full bridge (i.e. 8 parallel MOSFETs per switch) for a junction temperature $T_{\rm j}$ of 25°C. Figure 4.23 presents the switching losses for $T_{\rm j} = 125$ °C. Hard switching occurs for $I_{\rm S2,sw} < 0$ and soft switching for $I_{\rm S2,sw} > 0$. For $I_{\rm S2,sw} > 0$, the switching losses are mainly caused by the energies stored in the parasitic lead inductances [cf. (4.7) and [93]] and are therefore almost independent of the MOSFETs' junction temperatures. The measurement results further show that, in case of hard switching operation ($I_{\rm S2,sw} < 0$), the switching losses are almost independent of the MOSFETs' junction temperatures, too.⁶

In accordance to the switching processes discussed above, high switching losses result for $I_{\rm S2,sw} > 0$ (soft switching) and comparably low switching losses occur for $I_{\rm S2,sw} < 0$ (hard switching). With the LV side full bridge, lowest switching losses are achieved for $I_{\rm S2,sw} \approx 0$.

⁶The lower and the upper integration limits t_{begin} and t_{end} , needed to determine the switching loss energy [cf. (4.26)], are selected such that neither a reduction of t_{begin} nor an increase of t_{end} changes the measured switching losses; the duration of the measurement time interval $t_{\text{end}} - t_{\text{begin}}$ is between 200 ns and 1 µs.



Figure 4.23: Measured switching losses at $T_{\rm j} = 125^{\circ}{\rm C}$ for a single LV side switching action. $I_{\rm S2,sw}$ is the instantaneous current during switching (cf. Figure 4.22). For $I_{\rm S2,sw} > 0$, the switching losses are mainly caused by the energies stored in the parasitic lead inductances [cf. (4.7) and [93]] and are therefore almost independent of the MOSFETs' junction temperatures. The depicted measurement results further show that, in case of hard switching operation ($I_{\rm S2,sw} < 0$), the switching losses are almost independent of the MOSFETs' junction temperatures.

Detailed Consideration of the Switching Losses

Similar to the HV side, two different polynomial functions are fitted to the measured switching losses. Again, the switching losses are measured using a constant inductor current $\tilde{i}_{L_2}(t)$ during the dead time interval. However, due to high currents and low voltages, the switching process either commences directly after turn-off ($I_{S2,sw} > 0$ in Figure 4.22) or directly after turn-on ($I_{S2,sw} < 0$) and thus, the switching losses can be accurately evaluated using the instantaneous current $\tilde{i}_{L_2}(t_{sw})$ at the switching instant t_{sw} [cf. Table 4.1 with $\tilde{i}_{L_2}(t_{sw}) = -i_{L_2}(t_{sw})$].

Impact of the Switching Process on the Transformer Current

Due to the parasitic inductances connected in series to the semiconductor switches T_5 , T_6 , T_7 , and T_8 , a considerable voltage spike occurs on v_{AC2} if the LV side full bridge of the DAB is operated with ZVS (Figure 4.24); this voltage spike causes the absolute values of the inductor currents $|i_{L_1}|$ and $|i_{L_2}|$ to drop and decreases the converter efficiency.



Figure 4.24: (a) LV side overvoltage due to the parasitic MOSFET and PCB inductances (magnified detail of the measured waveforms of a first design of the DAB with n = 24 and $L = 30.8 \,\mu\text{H}$, $V_1 = 240 \,\text{V}$, $V_2 = 11 \,\text{V}$, $P_2 = 2 \,\text{kW}$ and power being transferred from the HV port to the LV port); (b) measured voltage time areas for different switch currents $I_{\text{S2,sw}}$ and phase shift modulation. The bold line denotes the voltage time area approximated for phase shift modulation: $2V_{\text{over}} \cdot T_{\text{over}} = 3.66 \,\text{nVs/A} \cdot I_{\text{S2,sw}}$ (the factor 2 is due to the phase shift modulation: there, two half bridges switch simultaneously); for a switching process of a single half bridge, $V_{\text{over}} = 10.2 \,\text{V}$ is considered.

The effect of the voltage spike on v_{AC2} is derived for a full bridge [Figure 4.25 (a)] operated with ZVS, a constant supply voltage V_2 , and a sufficiently large inductor current $|i_{L_2}(t)| \gg 0$ within the considered time interval. The presented analysis considers the lossless electric DAB model depicted in Figure 3.2, i.e. $\tilde{i}_{L_2} = -i_{L_2} = -n i_L$, and assumes equal total parasitic inductances L_{DS} ,

$$L_{\rm DS} = L_{\rm D} + L_{\rm S},$$
 (4.33)

for all 4 switches T_5 , T_6 , T_7 , and T_8 ; the forward voltage drops due to the MOSFETs' body diodes are neglected.

Initially, T₅ and T₈ are in the on-state and T₆ and T₇ in the off-state. At $t = t_{1A}$, T₅ is turned off and v_{DS,T_6} quickly drops until the body diode starts to conduct at $t = t_{1B}$. However, sufficiently large currents and a small time interval $t_{1A} < t < t_{1B}$ are assumed in order to estimate the worst possible current drop in \tilde{i}_{L_2} , i.e. the time interval $t_{1A} < t < t_{1B}$ shown in Figure 4.24 (a) is neglected.



Figure 4.25: (a) Inductively loaded full bridge (LV side) with parasitic inductances $L_{\rm D}$ and $L_{\rm S}$; (b) equivalent circuit of the LV full bridge used to exclusively investigate the implication of $v_{\rm over}$ on the inductor current \tilde{i}_{L_2} .

During $t_{1B} < t < t_{1C}$, the excess voltage $v_{over}(t) > 0$ is superimposed on v_{DS,T_5} ,

$$v_{\text{DS},\text{T}_5} = V_2 + v_{\text{over}}(t).$$
 (4.34)

Consequently, $v_{\text{over}}(t) = v_{\text{DS},\text{T}_5}(t) - V_2$ is partly applied to the parasitic lead inductances of T₅ [Figure 4.25 (b)]:

$$v'_{L_{\text{DS},5}} = v'_{L_{\text{D},5}} + v'_{L_{\text{S},5}} = -\frac{L_{\text{DS}}}{L_{\text{total}}} v_{\text{over}} \quad \text{with}$$
 (4.35)

$$L_{\text{total}} = L_{\text{DS}} + L_{\text{DS}} || (L_{\text{DAB}} + L_{\text{DS}})$$
 (4.36)

(' denotes currents and voltages that are generated due to v_{over}). Furthermore, v_{over} causes a voltage drop across the lead inductors of T₆:

$$v'_{L_{\text{DS},6}} = v'_{L_{\text{D},6}} + v'_{L_{\text{S},6}} = -\left(1 - \frac{L_{\text{DS}}}{L_{\text{total}}}\right) v_{\text{over}}.$$
 (4.37)

The voltage $v'_{L_{DS,5}}$, applied to L_S and L_D , causes the current i'_{D,T_5} to decrease according to:

$$i'_{\rm D,T_5}(t) = i'_{\rm D,T_5}(t_{1\rm B}) - \frac{1}{L_{\rm DS}} \frac{L_{\rm DS}}{L_{\rm total}} \int_{t_{1\rm B}}^{t} v_{\rm over}(t_{\rm int}) \,\mathrm{d}t_{\rm int}, \quad t_{1\rm B} \le t \le t_{1\rm C}.$$
(4.38)

With the condition $i'_{D,T_5}(t_{1C}) = 0$, the integrated overvoltage becomes:

$$\int_{t_{1B}}^{t_{1C}} v_{\text{over}}(t_{\text{int}}) \, \mathrm{d}t_{\text{int}} = i'_{\text{D},\text{T}_5}(t_{1B}) \, L_{\text{total}} = i'_{\text{D},\text{T}_5}(t_{1B}) \, \frac{L_{\text{DS}} \cdot (2L_{\text{DAB}} + 3L_{\text{DS}})}{L_{\text{DAB}} + 2L_{\text{DS}}}.$$
(4.39)

With (4.37), the inductor current,

$$\tilde{i}'_{L_2}(t_{1C}) = \tilde{i}'_{L_2}(t_{1B}) - \frac{1}{L_{\text{DAB}} + L_{\text{DS}}} \cdot \left(1 - \frac{L_{\text{DS}}}{L_{\text{total}}}\right) \int_{t_{1B}}^{t_{1C}} v_{\text{over}}(t_{\text{int}}) \, \mathrm{d}t_{\text{int}}, \quad (4.40)$$

is derived at $t = t_{1C}$. Thus, the inductor current drops by

$$\Delta \tilde{i}'_{L_2} = \tilde{i}'_{L_2}(t_{1C}) - \tilde{i}'_{L_2}(t_{1B}) = -\frac{L_{\text{DS}}}{L_{\text{DAB}} + 2L_{\text{DS}}} \tilde{i}'_{L_2}(t_{1B}) = -\frac{L_{\text{DS}}}{L/n^2} \tilde{i}'_{L_2}(t_{1B})$$
(4.41)

[using (4.39), $i_{\rm D,T_5}(t_{1\rm B}) = \tilde{i}'_{L_2}(t_{1\rm B})$, and $L/n^2 = L_{\rm DAB} + 2L_{\rm DS}$], independent of $v_{\rm over}$. Moreover, according to (4.40), an arbitrary waveform $v_{\rm over}(t)$ that satisfies (4.39) can be used to accurately calculate $\Delta \tilde{i}'_{L_2}$. The employed numerical solver (Section 4.6.1) superimposes a rectangular voltage with the constant magnitude $V_{\rm over} = 10.2$ V and the variable duration $t_{1\rm C} - t_{1\rm B}$,

$$t_{1\rm C} - t_{1\rm B} = \begin{cases} 179\,\mathrm{ps/A} \cdot I_{\rm S2,sw} & \text{if } I_{\rm S2,sw} > 0, \\ 0 & \text{if } I_{\rm S2,sw} < 0, \end{cases}$$
(4.42)

on $v_{AC2}(t)$ after a single switching process on the LV side (Figure 4.24, Figure 4.28).

Switching Time Delays

The total time delays between the initiation of a switching process and the actual change of the respective drain-to-source voltage are measured with the LV counterpart of the measurement setup depicted in Figure 4.18.

The measured switching time delays predominantly depend on the switching current $I_{S2,sw}$; the minimal time delay, $T_{S2,d,min} \approx 220 \text{ ns}$, is achieved for maximal positive current (i.e. $I_{S2,sw} = 500 \text{ A}$ in Figure 4.26); maximal time delay, $T_{S2,d,max} \approx 560 \text{ ns}$, occurs for hard switching operation with maximal negative current (i.e. $I_{S2,sw} = -500 \text{ A}$ in Figure 4.26). In the range $0 \text{ A} < I_{S2,sw} < 50 \text{ A}$, the relation between $T_{S2,d}$ and $I_{S2,sw}$ is highly non-linear: for low current levels, the time to charge the drain-to-source capacitances of the MOSFETs increases (cf. Section 4.3.2).

Measurements with different voltages V_2 within the specified voltage range, 10 V $\leq V_2 \leq 16$ V, show slightly different time delays (i.e. the time delay varies



Figure 4.26: Measured LV side switching time delays with respect to the switching current $I_{S2,sw}$ and with constant supply voltage, $V_1 = 12$ V, and different junction temperatures T_j .

within $\pm 20 \text{ ns}$); however, compared to the dependence on $I_{\text{S2,sw}}$, the influence of V_2 is considered to be negligible. Increased semiconductor temperatures cause the time delays to increase slightly, i.e. if the junction temperature rises from $T_j = 25^{\circ}\text{C}$ to $T_j = 125^{\circ}\text{C}$, the time delay in the soft switching range, $I_{\text{S2,sw}} > 0$, increases by $\approx 30 \text{ ns}$; in the hard switching range, the time delay increases by $\approx 10 \text{ ns}$ (Figure 4.26).

4.3.4 Gate Driver Losses

The gate driver circuits handle the charging and discharging processes of the parasitic gate-to-source and gate-to-drain capacitances of the MOSFETs in order to turn the MOSFETs on or off. The employed gate drivers simply use series resistances to charge or discharge the effective input capacitances of the MOSFETs. With this circuit, the related power dissipation can be calculated using the total gate charge $Q_{\rm G}$ of the MOSFET (datasheet value), the steady-state gate-to-source voltage $V_{\rm GS}$ during turn-on, and the switching frequency [107]. The HV side auxiliary power supply thus provides the resulting total gate driver power $P_{\rm S1,gate}$,

$$P_{\rm S1,gate} = 4 \, Q_{\rm G,1} V_{\rm GS,1} f_{\rm S},\tag{4.43}$$

for all 4 HV MOSFETs. Similarly, the LV side auxiliary power supply provides the total gate driver power $P_{\rm S2,gate}$,

$$P_{\rm S2,gate} = 32 \, Q_{\rm G,2} V_{\rm GS,2} f_{\rm S},\tag{4.44}$$

for all 32 LV MOSFETs. Thus, $P_{1,aux}$ already includes $P_{S1,gate}$ and $P_{2,aux}$ includes $P_{S2,gate}$.

4.4 Core Losses

For the given converter, the losses predicted with (4.10) and (4.14) are considered accurate enough, since the core losses are found to account for less than 10% of the total losses (at rated output power). However, for different specifications, core losses may become more important and thus a more advanced method, e.g. the modified Steinmetz equation [108], finite element analysis or the use of measurement results [109], may be required.

4.5 Accurate Input and Output Power Calculation

According to Section 4.1.3, all quantities required to calculate the efficiency (e.g. RMS current values) are determined at a certain input power $P_{\rm in}$. However, the port power levels calculated with the electric DAB model [e.g. (3.111) for the DAB converter model with conduction losses and magnetizing inductance] are inaccurate, since the losses considered within the DAB model are different to the accurately calculated losses. Consequently, the calculated efficiency is inaccurately mapped to the considered input and output power levels.

A reference power level P_{ref} is introduced in order to achieve a more accurate calculation of the actual input and output power levels. P_{ref} is calculated based on the HV side DAB power obtained with the electric DAB model, $P_1 = V_1 \cdot I_1$, and the HV side conduction losses,⁷

$$P_{\rm ref} = P_1 - R_1 \cdot I_{L_1}^2. \tag{4.45}$$

The more precise values of the HV port power, $P_{1,\text{prec}}$, and the LV port power, $P_{2,\text{prec}}$, are then obtained with

$$P_{1,\text{prec}} = P_{\text{ref}} + P_{1,\text{aux}} + P_{\text{S1,cond}} + P_{\text{S1,sw}} + P_{L_{\text{HV},\text{cond}}} + P_{L_{\text{HV},\text{core}}} + P_{\text{tr1,cond}} + P_{\text{tr,cond}}/2 \text{ and } (4.46)$$
$$P_{2,\text{prec}} = P_{\text{ref}} - (P_{\text{tr,core}}/2 + P_{\text{tr2,cond}} + P_{\text{PCB,a,cond}} + P_{\text{S2,cond}} + P_{\text{S2,sw}} + P_{\text{PCB,b,cond}} + P_{2,\text{aux}}), \qquad (4.47)$$

 $^{^{7}}P_{\text{ref}}$ could as well be calculated using $P_{2} = V_{2} I_{2}$. However, overvoltage switching transients are regarded on the LV side and therefore P_{2} partly contains switching losses; it is thus more intuitive to use P_{1} for the presented calculation.

whereas evenly shared transformer core losses between LV side and HV side are assumed. The expressions (4.45), (4.46), and (4.47) are independent of the direction of power transfer, however, negative values result for $P_{\rm ref}$, $P_{\rm 1,prec}$, and $P_{\rm 2,prec}$ if power is transferred from the LV port to the HV port.

4.6 Calculated and Measured Efficiencies, Phase Shift Modulation

4.6.1 Numerical Efficiency Calculation

A numerical solver calculates the transformer currents $i_{L_1}(t)$ and $i_{L_2}(t)$ in steady-state operation using piecewise constant voltages $v_{AC1}(t)$ and $v_{AC2}(t)$, which may change several times during a switching period and which repeat with changed sign after one half-cycle. By way of an examples, Figure 4.27 (a) illustrates $v_{AC1}(t)$ and $v_{AC2}(t)$ for $V_1 = 240$ V, $V_2 = 16$ V, $D_1 = 0.45$, $D_2 =$ 0.4, and $\varphi = 60^{\circ}$.

The employed numerical solver algorithm applies v_{AC1} and v_{AC2} to the most complete electric DAB model discussed in Section 3.3 (Figure 3.31) in order to calculate the waveforms $i_{L_1}(t)$ and $i_{L_2}(t)$ [Figure 4.27 (b)].⁸ Furthermore, current dependent overvoltage spikes due to switching processes on the LV side (Section 4.3.3) are added to v_{AC2} to consider their impacts on the currents i_{L_1} and i_{L_2} (Figure 4.28; an iterative loop within the solver algorithm determines the actual duration of the overvoltage spikes).

With the proposed method, excellent matching to measured current waveforms is achieved (Figure 4.29). The RMS currents, harmonic current components, instantaneous current values, and input and output power levels obtained from $i_{L_1}(t)$ and $i_{L_2}(t)$ [cf. (3.111) and (3.117), Section 3.3], are then applied to the methods discussed in this Chapter in order to determine the converter losses.

4.6.2 Comparison to Measured Efficiency Values

With increasing level of detail, the calculated efficiency becomes more and more accurate (Figure 4.2): Table 4.2 compares the predicted losses P_t to the measured losses $P_{t,meas}$ (the absolute differences in Table 4.2 are equal to $|P_t - P_{t,meas}|$, the relative differences are calculated with $|P_t - P_{t,meas}|/P_{t,meas}|$. In particular, the accurate consideration of the conduction losses – including high frequency effects – and the accurate consideration of the switching losses result in a major improvement and reduces the relative error from 44.0% to 6.3%

 $^{^{8}{\}rm The}$ current waveform calculation employs (3.106)–(3.110), (3.114)–(3.116), and (3.67).



Figure 4.27: (a) Considered voltage waveforms and (b) calculated current waveforms after the first run of the numerical solver using $V_1 = 240$ V, $V_2 = 16$ V, $D_1 = 0.45$, $D_2 = 0.4$, $\varphi = 60^\circ$, n = 19, $L_1 + n^2 L_2 = 26.7 \,\mu$ H, $R_1 + n^2 R_2 = 0.76 \,\Omega$ and a low magnetizing inductance of $L_{\rm M} = 100 \,\mu$ H (to emphasize the effect of the magnetizing current); during the first run, overvoltage spikes due to LV side switching are not considered.

in average. Additional little improvement to 6.0% average error is achieved with the correct consideration of input and output power levels according to Section 4.5.

In Figure 4.30, measured efficiencies are compared to calculated efficiencies for different operating points. Very good agreement between the results obtained from the theoretical model and the experimental values is observed.



Figure 4.28: (a) The final voltage waveforms and (b) the final current waveforms calculated with the numerical solver algorithm include overvoltage switching spikes according to Section 4.3.3; the illustrated waveforms are calculated using the same parameters used to obtain Figure 4.27.

Moreover, Figure 4.30 illustrates the strong dependence of the efficiency values on the operating point. Interestingly, these efficiency values also depend on the direction of power transfer (e.g. for $V_1 = 240$ V and $V_2 = 11$ V), even though the DAB topology shows a symmetric structure. The main reason for this difference is the transformer current reduction due to LV side switching according to (4.41). For power being transferred to the LV port (mode A), this current reduction occurs at the beginning of the main energy transfer interval (interval II in Figure 4.29), which consequently reduces the amount of transferred power. Thus, the phase angle $\varphi = \varphi_A$ must be increased in order



Figure 4.29: (a) Calculated and (b) measured voltage and current waveforms for phase shift modulation, $V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW, and power being transferred from the HV port to the LV port; n = 19, $L = 26.7 \,\mu\text{H}$. At this operating point, the instantaneous switch currents during a switching process of the LV side are close to zero and therefore, the overvoltage switching spikes contained in the calculation of $v_{AC2}(t)$ (Section 4.3.3) are almost negligible.

to obtain the required output power $P_{2,A,prec}$. For power being transferred to the HV port (mode B), the current reduction due to LV side switching occurs at the end of the main energy transfer interval. Hence, a lower phase angle $|\varphi_{\rm B}| < |\varphi_{\rm A}|$ is required for the same output power level $|P_{1,{\rm B},{\rm prec}}| = P_{2,{\rm A},{\rm prec}}$ which results in less circulating currents and lower switching losses.



Figure 4.30: Predicted efficiencies (solid lines) and measured efficiencies (\Box) for different operating conditions; the efficiencies are calculated with the improved DAB loss model which includes all methods discussed in this Chapter.

The presented loss model facilitates the prediction of the power loss distribution in the converter. In Figure 4.31 the losses for nominal operation are presented for the PCB, the low and high voltage side MOSFETs, the transformer, the inductor, and the auxiliary power supplies that provide the gate driver power. The LV MOSFETs dissipate 31 W, for the HV MOSFETs, total losses of 24 W occur, and the transformer has 22 W of losses.

However, the presented loss model does not include the effect of the temperature rise on the efficiency. Thus, besides the discussed electric model, a thermal converter model, which contains accurate information on the temperature dependency of the losses, would be required in order to calculate the expected component temperatures [104, 105].

For nominal operation of the DAB, high efficiency of more than 90% is achieved. The efficiency decreases for operating points that diverge significantly from the nominal operating point (Figure 4.30). The main reasons for the additional power loss arise from the conduction losses due to circulating transformer currents and increased switching losses on the LV side. A considerable improvement of the converter performance is achieved with optimized modulation methods where circulating transformer currents and switching losses are reduced, which is addressed in Chapter 5.

The development of the loss model has shown that the critical parts of the converter are the resistive losses on the LV side as well as switching losses. The losses on the LV side are considerably higher than on the HV side, even though the LV side contains eight switches in parallel and occupies twice the volume of the high voltage side.

Therefore, the following issues must be considered carefully in the design of a low voltage and high current converter to achieve high efficiency:

- Circulating transformer currents should be as small as possible.
- The impact of high frequency skin and proximity effects must be considered (transformer and PCB).

	Maximum relative difference	Maximum absolute difference	Average relative difference	Average absolute difference
Simple Model	56.4%	$135\mathrm{W}$	44.0%	$56.3\mathrm{W}$
Accurate conduction losses	46.1%	$67\mathrm{W}$	31.1%	$39.8\mathrm{W}$
Accurate conduction and switching losses	17.4%	$47\mathrm{W}$	6.3%	$8.1\mathrm{W}$
Full model (Figure 4.30)	17.1%	$16\mathrm{W}$	6.0%	$7.1\mathrm{W}$

• LV side switching should occur at low currents.

 Table 4.2: Differences between calculated and measured losses.



Figure 4.31: Calculated distribution of the power losses for operation at $V_1 = 340$ V, $V_2 = 12$ V, and $P_2 = 2$ kW.

One important aspect is the layout of the LV side PCB in order to achieve good current distribution in the PCB and between the parallel connected MOSFETs (cf. Appendix A.2.3). On the LV side, the transformer winding termination needs to be designed carefully due to high frequency losses (Appendix A.2.4). A model with calculated resistance and switching loss values can be used in a first step to design the system and can be refined with measurement results from a prototype.

4.7 Conclusion

An accurate power loss model for a bidirectional DC-DC converter with a high current / low voltage port is developed. It shows that switching losses and the LV side conduction losses are most critical for the design of the converter. Calculated results are compared to measurements obtained from a 2kW experimental system in order to verify the loss model. Based on this model the expected converter efficiency as well as the distribution of the power losses in the converter are calculated. This in-depth knowledge facilitates the development of efficiency optimized modulation schemes (Section 5.2) and enables the efficiency optimized converter design presented in Appendix A.2.

Even though, the discussion's focus is on a high current DAB, the general nature of the presented methods enables an accurate prediction of the power losses of other DC-DC converter topologies operated at high switching frequencies; the presented approach is particularly suitable for power converters that exhibit large current ripples. The method could thus as well be applied to automotive power converters with high power density [88] or resonant DC-DC converters with a high current port (Appendix A.4).

Chapter 5

Advanced DAB Modulation Schemes

According to Chapter 3, high transformer and switch RMS currents result if the DAB converter employs the conventional phase shift modulation. Alternative modulation schemes, e.g. the triangular and trapezoidal current mode modulation schemes (Section 3.1.3, Section 3.1.3, [97]) and the modulation schemes with minimum transformer RMS current (Section 3.1.3), cause lower RMS currents and thus, lower converter losses are expected.

The DAB converter, however, not only generates conduction losses: depending on the actual operating point (given V_1 , V_2 , and P_{out}), switching losses may contribute a considerable percentage of the total losses (Chapter 4).

In the present chapter, in Section 5.1, a suitable extension of the triangular and trapezoidal current mode modulation schemes is discussed (Section 5.1.1) and applied to the DAB loss model (Section 5.1.2). The resulting modulation schemes are further improved in Section 5.2 based on the findings obtained from a thorough optimization of the DAB modulation method with respect to the total converter losses.

5.1 Extended Triangular and Trapezoidal Current Mode Modulation

Figure 5.1 depicts the converter efficiencies calculated for different modulation schemes, different operating voltages, and different power levels P_2 . Obviously, compared to phase shift modulation, a major improvement is achieved with the triangular and trapezoidal current mode modulation schemes [Figures 5.1 (c) and (d)].¹ With these modulation schemes, however,

 $^{^{1}}$ The efficiency calculation employed to obtain the results depicted in Figure 5.1 considers the effect of the linear interpolation, used by the DSP, to determine the actual



Figure 5.1: Comparison of the efficiencies calculated for different modulation schemes and different power levels P_2 ; (a) and (b) phase shift modulation $(n = 19, L = 26.7 \,\mu\text{H})$; (c) and (d) triangular and trapezoidal current mode modulation schemes according to [97] $(n = 19, L = 18.7 \,\mu\text{H})$; (e) and (f) extended triangular and trapezoidal current mode modulation schemes (Section 5.1; $n = 19, L = 18.7 \,\mu\text{H}$); (g) and (h) suboptimal modulation schemes (Section 5.2.2; $n = 19, L = 26.7 \,\mu\text{H}$). The calculation considers the effect of the linear interpolation used by the DSP to determine the actual modulation parameters (Appendix E.2). The efficiency is calculated for $T_i = 25^{\circ}\text{C}$ (cf. Chapter 4).

hard switching processes at zero transformer current occur, which cause high switching losses on the HV side (cf. Section 4.3.2, Figure 4.14). The extended triangular and trapezoidal current mode modulation schemes discussed in this Section operate the HV side full bridge with ZVS and thus, further efficiency improvements are achieved [Figures 5.1 (e) and (f)]. Finally, with the suboptimal modulation schemes discussed in Section 5.2.2, converter efficiencies close to the optimum efficiencies are achieved [Figures 5.1 (g) and (h), Figure 5.28].

5.1.1 Working Principle

According to Section 3.1.3, the triangular and trapezoidal current mode modulation schemes employ 3 different operating modes of the DAB:

- Mode a: triangular current mode, $V_1 > nV_2 \land 0 < |P| \le P_{\triangle,a,\max}$
- Mode b: triangular current mode, $V_1 < nV_2 \land 0 < |P| \le P_{\triangle, b, max}$
- Mode c: trapezoidal current mode, $(V_1 > nV_2 \land P_{\triangle,a,\max} < |P| \le P_{\triangle,\max}) \lor$ $(V_1 < nV_2 \land P_{\triangle,b,\max} < |P| \le P_{\triangle,\max})$

Within the subsequent sections, the 3 operating modes are separately improved with respect to converter efficiency; the resulting modulation schemes are termed extended triangular and trapezoidal current mode modulation schemes. Due to the implemented changes, however, the transition conditions change:

- Mode a: extended triangular current mode, $V_{2,\lim} > V_2 \land 0 < |P| \le P_{\triangle,a,\max}$
- Mode b: extended triangular current mode, $V_{2,\text{lim}} < V_2 \land 0 < |P| \le P_{\triangle,\text{b,max}}$
- Mode c: extended trapezoidal current mode, $(V_{2,\lim} > V_2 \land P_{\triangle,a,\max} < |P| \le P_{\triangle,\max}) \lor$ $(V_{2,\lim} < V_2 \land P_{\triangle,b,\max} < |P| \le P_{\triangle,\max})$

Thus, besides different values for $P_{\triangle,a,\max}$, $P_{\triangle,b,\max}$, and $P_{\triangle,\max}$, the new variable $V_{2,\lim}$ is introduced [cf. (5.7) and Figure 5.24].

Since this Section focuses on the working principles of the employed extended triangular and trapezoidal current mode modulation schemes, the lossless electric DAB model is used to generate all equations and Figures. Accordingly, equal power levels occur for port 1 and port 2, i.e. $P_1 = P_2 = P$.

modulation parameters (Appendix E.2). This causes the irregular characteristics of the depicted contour lines.

Again, positive power values denote a power transfer from port 1 to port 2 (HV to LV) and negative power values denote a power transfer from port 2 to port 1 [cf. (3.14)].

Triangular Current Mode Modulation Revisited

Mode a, $V_{2,\text{lim}} > V_2$. The unmodified triangular current mode modulation scheme generates transformer currents according to the currents depicted in Figure 3.7 and Figure 3.8, i.e. the LV side full bridge solely switches at zero transformer current causing low switching losses (cf. Section 4.3.3); once per half period, the HV side full bridge switches zero transformer current, too, which causes high switching losses (Section 4.3.2).

With respect to low total switching losses, the modulation scheme needs to be modified, such that the LV side full bridge continues to switch at zero current and the HV side full bridge is operated with ZVS. The required modification to the modulation scheme introduces circulating transformer currents (Figure 5.2), needed to achieve ZVS on the HV side (for the given hardware prototype, the reduced switching losses more than outweigh the slightly increased conduction losses, cf. Section 5.2.1, Figure 5.17).

With this modified triangular modulation scheme, the maximum achieved power is far below the maximum possible power of the DAB [cf. (3.27), Section 3.1.3]. However, different to the original triangular current mode modulation, no direct transition to a high power modulation scheme (e.g. trapezoidal current mode modulation) exists; the transition between low and high power operation rather includes an intermediate modulation scheme (Figure 5.3) which considerably increases the complexity of the modulator.

Figure 5.4 depicts a different extension of the triangular current mode modulation scheme which also enables the actual hardware prototype to achieve a high converter efficiency (i.e. considerably higher than the efficiency achieved with the unmodified triangular current mode modulation scheme). The modulation scheme shown in Figure 5.4 furthermore facilitates a seamless transition to high power modulation schemes (e.g. the extended trapezoidal current mode modulation scheme discussed below, cf. Figure 5.6). There, a certain transformer current remains during the freewheeling time interval, $t_3 < t < T_S/2$, which enables ZVS of the HV side full bridge at $t = T_S/2$; at $t = t_1$, the LV side full bridge switches at zero transformer current is achieved at $t = t_1$. The implemented modification, however, increases the transformer RMS current and causes the LV side full bridge to perform hard switching processes at $t = t_3$ and $t = t_7$. Still, with this extended triangular current mode modulation, increased converter efficiency is achieved (compared to the original triangular current mode modulation), since the full bridge on the



Figure 5.2: Modified triangular current mode modulation which enables HV side ZVS and LV side switching at zero transformer current; $V_1 = 340$ V, $V_2 = 12$ V, P = 1 kW, n = 19, $L = 18.7 \mu$ H, and $f_S = 100$ kHz.

LV side causes comparably low switching losses when operated with hard switching (Figure 4.22).

Based on the lossless electric DAB model, the required duty cycles and the required phase shift angle are determined with:

$$D_1 = \frac{1}{\pi} \frac{nV_2}{V_1 - nV_2} |\varphi| + \frac{2f_{\rm S}L(-I_0)}{V_1},\tag{5.1}$$

$$D_2 = \frac{1}{\pi} \frac{V_1}{V_1 - nV_2} |\varphi|, \tag{5.2}$$

$$\varphi = \frac{\pi f_{\rm S} L I_0}{n V_1 V_2} \left[V_1 - n V_2 - \sqrt{V_1 (V_1 - n V_2) \left(1 + \frac{|P|}{f_{\rm S} L I_0^2} \right)} \right] \operatorname{sgn}(P) \quad (5.3)$$

$$\forall \quad V_2 < V_{2,\lim} \land P_{\triangle,a,\min} < |P| \le P_{\triangle,a,\max} \tag{5.4}$$

(with given operating voltages V_1 and V_2 , the DAB power level P, and the current $I_0 = i_L(0)$ required to achieve ZVS on the HV side).



Figure 5.3: Sequence of DAB operating modes required to transit from low power levels (modified triangular current mode modulation scheme, Figure 5.2) to high power levels (extended trapezoidal current mode modulation scheme, Figure 5.6): (a) maximum power achieved with the operating mode depicted in Figure 5.2 and with $T_{\rm f} = 500$ ns; (b), (c), and (d) continuously increasing DAB power levels, achieved with the extended triangular current mode modulation depicted in Figure 5.4; (e) continuous transition to the extended trapezoidal current mode modulation (Figure 5.6); $V_1 = 340$ V, $V_2 = 12$ V, n = 19, $L = 18.7 \,\mu$ H, and $f_{\rm S} = 100$ kHz; a minimum freewheeling time of $T_{\rm f} = 500$ ns is selected in order to obtain a clear illustration.



Figure 5.4: Extended triangular current mode modulation scheme (mode a), which enables ZVS on the HV side and allows for a direct transition to the extended trapezoidal current mode modulation scheme; $V_1 = 340$ V, $V_2 = 12$ V, P = 1 kW, n = 19, $L = 18.7 \mu$ H, and $f_S = 100$ kHz.

The operating range is limited according to:

$$P_{\Delta,a,\min} = \frac{nV_2}{V_1 - nV_2} f_S L I_0^2,$$

$$P_{\Delta,a,\max} = \frac{(nV_2)^2 (V_1 - nV_2)}{4f_S L V_1} \left[4f_S L (1 - 2f_S T_f) (-I_0) \frac{V_1 - nV_2}{nV_1 V_2} + \frac{(1 - 2f_S T_f)^2 - 4(f_S L V_2 + 2^2 3V_1^2 - 3nV_1 V_2 + (nV_2)^2)}{(1 - 2f_S T_2)^2 - 4(f_S L V_2 + 2^2 3V_1^2 - 3nV_1 V_2 + (nV_2)^2)} \right]$$
(5.5)

$$(1 - 2f_{\rm S}T_{\rm f})^2 - 4(f_{\rm S}L)^2 I_0^2 \frac{3V_1^2 - 3nV_1V_2 + (nV_2)^2}{V_1^3 nV_2 - (nV_1V_2)^2} \Big],$$
(5.6)

$$nV_{2,\lim} = V_1 + \frac{1}{\frac{1}{V_1} + \frac{T_f}{L(-I_0)} - \frac{1}{2f_S(-I_0)L}}$$
(5.7)

 $(T_{\rm f}$ denotes the minimum length of the freewheeling time interval, cf. Figure 5.3 (e), [97]).

Mode b, $V_{2,\text{lim}} < V_2$. For triangular current mode modulation with $V_{2,\text{lim}} < V_2$, the HV side full bridge solely switches at zero transformer current and the LV side full bridge alternately switches at zero current, i.e. $I_{S2,\text{sw}} = 0$, and at $I_{S2,\text{sw}} > 0$ (Figure 3.10 and Figure 3.11; cf. Table 4.1 regarding $I_{S2,\text{sw}}$).



Figure 5.5: Extended triangular current mode modulation scheme (mode b), which enables ZVS on the HV side for $V_{2,\text{lim}} < V_2$; $V_1 = 240 \text{ V}$, $V_2 = 16 \text{ V}$, P = 1 kW, n = 19, $L = 18.7 \,\mu\text{H}$, and $f_{\text{S}} = 100 \,\text{kHz}$.

Similar to mode a, a reduction of the switching losses on the HV side (i.e. ZVS operation) is achieved with the extended triangular current mode modulation scheme (mode b) depicted in Figure 5.5: the freewheeling current during $t_2 < t < T_S/2$ enables ZVS operation on the HV side, however, increases the transformer RMS current and causes hard switching processes on the LV side at $t = t_2$ and at $t = t_5$.

For a given operating point (given V_1 , V_2 , and P), the required duty cycles and the phase shift angle are:

$$D_1 = \frac{1}{\pi} \frac{nV_2}{nV_2 - V_1} |\varphi| - \frac{2f_{\rm S}L(-I_0)}{nV_2 - V_1},\tag{5.8}$$

$$D_2 = \frac{1}{\pi} \frac{V_1}{nV_2 - V_1} |\varphi| - \frac{2f_{\rm S}L(-I_0)}{nV_2 - V_1},\tag{5.9}$$

$$\varphi = \frac{\pi f_{\rm S} L(-I_0)}{V_1} \left[1 + \sqrt{1 + \frac{nV_2 - V_1}{nV_2} \frac{|P|}{f_{\rm S} L I_0^2}} \right] \operatorname{sgn}(P)$$
(5.10)

$$\forall \quad (V_2 < V_{2,\lim} \land |P| \le P_{\triangle,a,\min}) \lor (V_2 \ge V_{2,\lim} \land |P| \le P_{\triangle,b,\max}) \quad (5.11)$$

[using the lossless electric DAB model, the current $I_0 = i_L(0)$ required to achieve ZVS, and (5.7)]. The useful range of the above equations is limited


Figure 5.6: Extended trapezoidal current mode modulation scheme, which enables HV side ZVS and high power operation of the DAB; $V_1 = 340$ V, $V_2 = 12$ V, P = 2.5 kW, n = 19, $L = 18.7 \mu$ H, and $f_S = 100$ kHz.

with (5.5) and

$$P_{\Delta,\mathrm{b,max}} = 4f_{\mathrm{S}}LI_{0}^{2} \frac{V_{1}}{nV_{2}} \left[V_{1} \frac{1-2f_{\mathrm{S}}T_{\mathrm{f}}}{4f_{\mathrm{S}}L(-I_{0})} - 1 \right] \left[(nV_{2}-V_{1})\frac{1-2f_{\mathrm{S}}T_{\mathrm{f}}}{4f_{\mathrm{S}}L(-I_{0})} + 1 \right].$$
(5.12)

Extended Trapezoidal Current Mode Modulation

The trapezoidal current mode modulation scheme (Figure 3.12, and Figure 3.13) requires the HV side full bridge to alternately switch at $I_{\rm S1,sw} = 0$ (hard switching process) and at $I_{\rm S1,sw} > 0$ (ZVS). Similarly, the LV side full bridge alternately switches at $I_{\rm S2,sw} = 0$ and at $I_{\rm S2,sw} > 0$.

In order to avoid hard switching processes on the HV side, the current $I_0 = i_L(0)$ is shifted accordingly, which, however, causes the transformer RMS current to increase and introduces hard switching processes on the LV side at $t = t_3$ and at $t = t_7$ (Figure 5.6). (Again, a minimum freewheeling time $T_f = T_S/2 - t_3$ may be used to avoid overlapping of the transformer voltages v_{AC1} and v_{AC2} at $t = kT_S/2$, $k \in \mathbb{Z}$, caused by different time delays of the employed semiconductors; cf. Section 4.3.2 and Section 4.3.3.)

Based on the lossless electric DAB model, the modulation parameters of the extended trapezoidal current mode modulation scheme (mode c) for a given operating point $(V_1, V_2, \text{ and } P)$ are calculated with:

$$D_1 = \frac{nV_2(1 - 2f_{\rm S}T_{\rm f}) + 2f_{\rm S}L(-I_0)}{V_1 + nV_2} - \frac{nV_2}{\pi(V_1 + nV_2)}|\varphi|,$$
(5.13)

$$D_2 = \frac{V_1(1 - 2f_{\rm S}T_{\rm f}) - 2f_{\rm S}L(-I_0)}{V_1 + nV_2} - \frac{V_1}{\pi(V_1 + nV_2)}|\varphi|, \tag{5.14}$$

$$\varphi = \frac{\pi}{V_1^2 + nV_1V_2 + (nV_2)^2} \left[-f_{\rm S}T_{\rm f}(V_1^2 + (nV_2)^2) + \frac{V_1^2 - 2f_{\rm S}L(-I_0)(V_1 - nV_2) + (nV_2)^2 - (V_1 + nV_2)\sqrt{f}}{2} \right] \operatorname{sgn}(P)$$
(5.15)

$$\forall \quad (V_2 < V_{2,\lim} \land P_{\triangle,a,\max} < |P| \le P_{\triangle,\max}) \lor (V_2 \ge V_{2,\lim} \land P_{\triangle,b,\max} < |P| \le P_{\triangle,\max}),$$
 (5.16)

using (5.6), (5.7), (5.12), and

$$f = nV_1V_2 + 4f_SL(-I_0)[V_1 - nV_2 - 3f_SL(-I_0)] - 4f_SL|P|\frac{V_1^2 + nV_1V_2 + (nV_2)^2}{nV_1V_2} + 4f_ST_f[2f_SL(-I_0)(nV_2 - V_1) + nV_1V_2(f_ST_f - 1)].$$
(5.17)

The maximum possible power is limited according to:

$$P_{\Box,\max} = \frac{nV_1V_2}{4f_{\rm S}L[V_1^2 + nV_1V_2 + (nV_2)^2]} \Big[4f_{\rm S}L(1 - 2f_{\rm S}T_{\rm f})(-I_0)(V_1 - nV_2) - 12(f_{\rm S}L)^2I_0^2 + nV_1V_2(1 - 2f_{\rm S}T_{\rm f})^2\Big].$$
(5.18)

Summary of the Extended Triangular and Trapezoidal Current Mode Modulation Schemes

Very low power operation. At DAB power levels close to zero, the operation using the *mode* b extended triangular current mode modulation is always possible [Figures 5.7 (a) and (b)].²

²According to Figures 5.7 (a) and (b) almost a trapezoidal current i_L results at very low power operation, due to the current I_0 , which is required to achieve ZVS.



Figure 5.7: Extended triangular current mode modulation (*mode b*) at very low power levels: (a) $V_1 = 400$ V, $V_2 = 11$ V, P = 10 W and (b) $V_1 = 240$ V, $V_2 = 16$ V, P = 10 W; $I_0 = -4$ A, n = 19, $L = 18.7 \mu$ H, and $f_S = 100$ kHz. Only the first half-cycle is shown; the waveforms repeat with opposite sign during the second half-cycle. The inductor current i_L is almost trapezoidal at very low power levels, due to the current I_0 , which is required to achieve ZVS.

Low and medium power operation. With increasing power levels, the current I_1 depicted in Figure 5.8 (a) decreases and becomes zero at $|P| = P_{\triangle a,\min}$ [Figure 5.8 (b)]; simultaneously, t_2 increases and becomes equal to $T_S/2-T_f$ at $|P| = P_{\triangle b,\max}$. For $V_2 = V_{2,\lim}$, the power limits $P_{\triangle a,\min}$, $P_{\triangle a,\max}$, and $P_{\triangle b,\max}$ are all equal, i.e. the useful range of mode a becomes zero. Consequently, mode a applies for $V_2 < V_{2,\lim}$ and $P_{\triangle a,\min} < |P| < P_{\triangle a,\max}$ [Figure 5.9 (a)] and mode b applies for $V_2 \ge V_{2,\lim}$ and $|P| < P_{\triangle b,\max}$ [Figure 5.9 (b)].

High power operation. For DAB power levels exceeding $P_{\Delta a, \max}$ or $P_{\Delta b, \max}$ (for $V_2 < V_{2, \lim}$ or $V_2 \ge V_{2, \lim}$, respectively), the trapezoidal current mode modulation scheme (*mode c*) is employed.

Selection of I_0

With the employed HV MOSFETS (CoolMOS SPW47N60CFD) and the selected dead time used to avoid a shoot-through in the half bridge ($T_{\text{deadtime}} = 200 \text{ ns}$), I_0 could be set to $\approx -2 \text{ A}$, based on the switching losses depicted in Figure 4.15. However, different to the switching loss measurement, the current in the actual DAB may considerably change during the dead time interval and thus, $I_0 < -2 \text{ A}$ is required to obtain ZVS (cf. Section 4.3.2).



Figure 5.8: Extended triangular current mode (mode b) modulation at low power levels: (a) $V_1 = 240$ V, $V_2 = V_{2,\text{lim}} = 11.8$ V, P = 50 W $< P_{\triangle b,\text{max}}$ and (b) $V_1 = 240$ V, $V_2 = V_{2,\text{lim}} = 11.8$ V, $P = P_{\triangle b,\text{max}} = 415$ W; $I_0 = -4$ A, n = 19, $L = 18.7 \,\mu\text{H}$, and $f_S = 100 \,\text{kHz}$. For $V_2 = V_{2,\text{lim}}$ and $P = P_{\triangle b,\text{max}}$, t_2 becomes equal to $T_S/2 - T_f$ and $I_1 = i_L(t_1)$ becomes zero. Only the first halfcycle is shown; the waveforms repeat with opposite sign during the second half-cycle.



Figure 5.9: Extended triangular current mode modulation at medium power levels: (a) mode a, $V_1 = 400$ V, $V_2 = 11$ V, P = 500 W and (b) mode b, $V_1 = 240$ V, $V_2 = 16$ V, P = 500 W; $I_0 = -4$ A, n = 19, $L = 18.7 \,\mu$ H, and $f_S = 100$ kHz. Only the first half-cycle is shown; the waveforms repeat with opposite sign during the second half-cycle.

The measurement setup depicted in Figure 5.10 (a) is used to determine reasonable values for I_0 . During $t_{0b} < t < t_{1a}$, the full bridge circuit applies



Figure 5.10: (a) Full bridge measurement setup used to obtain reasonable values for I_0 ; (b) measured waveforms $v_{AC1}(t)$ and $i_L(t)$ for $V_1 = 350$ V, $i_L(t_2) = 10$ A, $L = 20 \,\mu$ H, and $T_j = 25^{\circ}$ C: with the selected MOSFETs (SPW47N60CFD), $I_0 = i_L(t_3) \approx -4$ A occurs; (c) gate signals required to obtain the depicted voltage waveform $v_{AC1}(t)$.

positive voltage to the inductor L in order to increase i_L . The subsequent freewheeling time interval $(t_{1a} < t < t_2)$ is used to accurately adjust $i_L(t_2)$, since the inductor current slightly decreases due to resistive losses. At $t = t_2$, T_4 turns off and the body diode of T_3 conducts during $t_2 < t < t_3$ with $i_{D,T_3}(t) = -i_L(t)$. Consequently, negative voltage, $v_{AC1}(t) = -V_1$, is applied to L, which causes the inductor current to drop quickly. At $t = t_3$, the diode reverse recovery charge is removed, the drain-to-source capacitances of T_3 and T_4 are charged or discharged, and thus, $v_{AC1}(t)$ changes from $-V_1$ to zero.

The value of the peak reverse current determines the actual value of I_0 [i.e. $I_0 = i_L(t_3)$]; I_0 depends on the supply voltage V_1 , the current slope di_L/dt , the initial diode current $i_L(t_2)$, and the junction temperature T_j (Figure 5.11). Similar to the switching loss measurements, the polynomial function

$$I_{0}(\mathrm{d}i_{L}/\mathrm{d}t, V_{1}, i_{L}(t_{2}), T_{j}) = a + b_{1} \mathrm{d}i_{L}/\mathrm{d}t + b_{2} V_{1} + b_{3} i_{L}(t_{2}) + b_{4} T_{j} + c_{1} V_{1} \mathrm{d}i_{L}/\mathrm{d}t + c_{2} i_{L}(t_{2}) \mathrm{d}i_{L}/\mathrm{d}t + c_{3} T_{j} \mathrm{d}i_{L}/\mathrm{d}t + c_{4} V_{1} i_{L}(t_{2}) + c_{5} V_{1} T_{j} + c_{6} T_{j} i_{L}(t_{2})$$

$$(5.19)$$

with the constant coefficient a, 4 linear terms with the coefficients b_1 , b_2 , b_3 , and b_4 , and 6 product terms (coefficients $c_1 \ldots c_6$), is fitted to the measurement data by means of least mean square optimization. The minimum of the average over all absolute values of the relative error,

$$e_{\text{avg}} = \frac{1}{m} \sum_{i=1}^{m} |e_i|, \ e_i = \frac{I_{0,\text{calc},i} - I_{0,\text{meas},i}}{I_{0,\text{meas},i}},$$
(5.20)

(m = 99 denotes the total number of measurements; $I_{0,\text{meas},i}$ and $I_{0,\text{calc},i}$ denote the *i*-th measured and calculated values of I_0 , respectively), $\min(e_{\text{avg}}) \approx 2.5\%$, is obtained for

$$a = -269 \text{ mA},$$

$$b_1 = 132 \text{ ns}, \qquad b_2 = -5.09 \text{ mA/V}, \qquad b_3 = 79.8 \text{ mA/A}, \qquad b_4 = 0,$$

$$c_1 = -119 \text{ ps/V}, \quad c_2 = 5.08 \text{ ns/A}, \qquad c_3 = 222 \text{ ps/K},$$

$$c_4 = 0, \qquad c_5 = 7.72 \text{ }\mu\text{A/(VK)}, \quad c_6 = -1.39 \cdot 10^{-3} \text{ K}^{-1}. \qquad (5.21)$$

With $di_L/dt = -V_1/L$ and for constant L, T_j , and $i_L(t_2)$, the relation between V_1 and I_0 becomes almost linear (Figure 5.12), e.g. for $L = 18.7 \,\mu\text{H}$, $T_j = 125^{\circ}\text{C}$, and $i_L(t_2) = 2 \,\text{A}$,

$$I_0(V_1)|_{L=18.7\,\mu\text{H}} = -1.19\,\text{A} - 8.82\,\text{mA/V} \cdot V_1 \tag{5.22}$$



Figure 5.11: Measured peak reverse currents $i_L(t_3)$ for different supply voltages V_1 , different inductances L and different junction temperatures T_j ; the measured values are obtained from the full bridge circuit depicted in Figure 5.10 with CoolMOS SPW47N60CFD MOSFETs being employed. The solid lines are calculated with (5.19).



Figure 5.12: Evaluated polynomial function (5.19) using $di_L/dt = -V_1/L$, $L = 18.7 \,\mu\text{H}$, and different values for $i_L(t_2)$ and T_j : for constant L, $i_L(t_2)$, and T_j an almost linear relation between V_1 and I_0 results.

results;³ for $L = 26.7 \,\mu\text{H}$, $T_{\rm j} = 125^{\circ}\text{C}$, and $i_L(t_2) = 2\,\text{A}$, $I_0(V_1)$ is approximately given with

$$I_0(V_1)|_{L=26.7 \text{ uH}} = -0.97 \text{ A} - 7.41 \text{ mA/V} \cdot V_1.$$
(5.23)

5.1.2 Application of the DAB Loss Model

According to Chapter 4, the lossless DAB model (Figure 3.2) is insufficient to accurately predict the actual DAB output power P_{out} [cf. (3.71)]. Consequently, the equations to calculate the DAB control parameters D_1 , D_2 , and φ for a given output power presented in Section 5.1.1 (based on the lossless DAB model) cause the DAB hardware prototype to inaccurately set P_{out} and I_0 .

More precise values for D_1 , D_2 , and φ are achieved with the numerical solver discussed in Section 4.6.1. On-line parameter calculation, however, is not feasible due to the limited DSP processing power. Therefore, in a more

³The actually required value of $|I_0|$ may be lower due to a lower $|di_L/dt|$ at the switching instant (i.e. for power being transferred from the HV port to the LV port, cf. Figure 5.4 and Figure 5.5); however, for the opposite power transfer direction and once per half-cycle, a switching process similar to the process at $t = t_3$, depicted in Figure 5.10, occurs. Thus, in order to cover the worst case, I_0 is selected according to (5.19) by setting $di_L/dt = -V_1/L$.

practical approach, the DSP stores the control parameters D_1 , D_2 , and φ , precalculated for dedicated output power levels P_{out} and voltages V_1 and V_2 , in three-dimensional tables and employs a fast linear interpolation algorithm to compute the actually required control parameters (Appendix E.2).

The off-line calculation of the control parameters considers the 3 different operating modes of the DAB (i.e. 2 triangular current modes and 1 trapezoidal current mode), separately.

Mode a, triangular current mode (cf. Figure 5.4). In order to reduce the computational effort, the time $T_{I_0} = t_1 - 0$, required to enable ZVS, is approximated with

$$T_{I_0} = \frac{L |I_0|}{V_1} \tag{5.24}$$

and thus,

$$D_1 = D_2 - \frac{|\varphi|}{\pi} + 2T_{I_0} f_{\rm S}$$
(5.25)

applies (cf. Figure 5.4 and Figure 3.3). A numerical search algorithm seeks for suitable values D_2 and φ that meet the requirements

$$P_{\rm out}(D_1, D_2, \varphi) = P_{\rm out,d},\tag{5.26}$$

$$I_0(D_1, D_2, \varphi) = I_{0,d} \tag{5.27}$$

$$\forall V_2 < V_{2,\lim} \land |P_{\mathrm{out},\triangle,\mathrm{a,min}}| < |P_{\mathrm{out}}| \le |P_{\mathrm{out},\triangle,\mathrm{a,max}}| \quad (5.28)$$

in order to achieve the designated output power $P_{\rm out,d}$ and the designated freewheeling current $I_{0,\rm d}.^4$

The numerical solver as well computes $P_{\text{out},\triangle,a,\min}$ and $P_{\text{out},\triangle,a,\max}$: for $P_{\text{out},\triangle,a,\min}$, $D_1 - D_2 = T_{I_0} f_{\text{S}}$ applies; thus, using (5.25),

$$[I_0(D_1, D_2, \varphi) = I_{0,d} \land D_1 = D_2 + T_{I_0} f_{\mathrm{S}} \land \varphi = \pi T_{I_0} f_{\mathrm{S}} \operatorname{sgn}(P_{\mathrm{out},d})]$$

$$\Leftrightarrow P_{\mathrm{out}}(D_1, D_2, \varphi) = P_{\mathrm{out}, \triangle, \mathrm{a,min}}, \text{ search variable: } D_2, \quad (5.29)$$

needs to be solved with respect to $P_{\text{out}, \triangle, a, \min}$. At $P_{\text{out}} = P_{\text{out}, \triangle, a, \max}$,

$$[I_0(D_1, D_2, \varphi) = I_{0,d} \wedge D_1 = D_2 - \frac{|\varphi|}{\pi} + 2T_{I_0} f_{\mathrm{S}} \wedge D_2 = \frac{1}{2} - (T_{I_0} + T_{\mathrm{f}}) f_{\mathrm{S}}]$$

$$\Leftrightarrow P_{\mathrm{out}}(D_1, D_2, \varphi) = P_{\mathrm{out}, \triangle, \mathrm{a,max}}, \text{ search variable: } \varphi, \quad (5.30)$$

applies.

The calculation of $V_{2,\text{lim}}$ is discussed in the next paragraph, since $V_2 = V_{2,\text{lim}}$ causes the DAB to operate exactly at the border between *mode a* and *mode b*.

⁴Different to Section 5.1.1, the values of the limiting power levels $P_{\text{out}, \triangle, a, \min}$, $P_{\text{out}, \triangle, a, \max}$, $P_{\text{out}, \triangle, b, \max}$, and $P_{\text{out}, \triangle, \max}$ now depend on the direction of power transfer.

Mode b, triangular current mode (cf. Figure 5.5). With (5.24),

$$D_2 = D_1 - \frac{|\varphi|}{\pi}$$
(5.31)

results; a numerical search algorithm determines the appropriate values for D_1 and φ that satisfy (5.26) and (5.27) within

 $(V_2 < V_{2,\lim} \land |P_{\text{out}}| \le |P_{\text{out},\triangle,a,\min}|) \lor (V_2 \ge V_{2,\lim} \land |P_{\text{out}}| \le |P_{\text{out},\triangle,b,\max}|).$ (5.32)

The maximum allowable power $P_{out, \triangle, b, max}$ is obtained from

$$\begin{bmatrix} I_0(D_1, D_2, \varphi) = I_{0,d} \land D_1 = \frac{1}{2} - T_{\rm f} f_{\rm S} \land D_2 = D_1 - \frac{|\varphi|}{\pi} \end{bmatrix}$$

$$\Leftrightarrow P_{\rm out}(D_1, D_2, \varphi) = P_{\rm out, \triangle, b, max}, \text{ search variable: } \varphi. \quad (5.33)$$

A slightly more complex procedure is needed to obtain $V_{2,\text{lim}}$: unlike (5.7), $V_{2,\text{lim}}$ now as well depends on the actual value of $P_{\text{out},d}$, due to resistive losses and due to the magnetizing current. First, $P_{\text{out},\triangle,\text{V2lim,max}} = P_{\text{out},\triangle,\text{b,max}}(V_2 = V_{2,\text{lim}})$ is calculated; there (5.29), (5.30), and (5.33) are simultaneously valid (cf. Section 5.1.1) and with (5.24) and constant V_1 , a constant phase shift angle and constant duty cycles result. The numerical solver hence needs to vary V_2 in order to determine $P_{\text{out},\triangle,\text{V2lim,max}}$ and the corresponding $V_{2,\text{lim}}(P_{\text{out},\triangle,\text{V2lim,max}})$ in compliance with

For $|P_{\text{out},d}| \ge |P_{\text{out},\triangle,\text{V2lim,max}}|$, (5.34) already determines the correct $V_{2,\text{lim}}$. For $|P_{\text{out},d}| < |P_{\text{out},\triangle,\text{V2lim,max}}|$ and with (5.24) and (5.31),

applies, which is used to compute $V_{2,\lim}(P_{\text{out,d}})$.

-

Mode c, trapezoidal current mode (cf. Figure 5.6). The trapezoidal current mode modulation scheme employs

$$D_{1} + D_{2} = 1 - \frac{|\varphi|}{\pi} - 2T_{f}f_{S}$$

$$\forall \quad (V_{2} < V_{2,\lim} \land |P_{\text{out},\triangle,a,\max}| < |P| \le |P_{\text{out},\triangle,\max}|) \lor$$

$$(V_{2} > V_{2,\lim} \land |P_{\text{out},\triangle,b,\max}| < |P| < |P_{\text{out},\triangle,\max}|),$$

$$(5.36)$$

 $(V_2 \ge V_{2,\lim} \land |P_{\text{out},\triangle,\text{b,max}}| < |P| \le |P_{\text{out},\triangle,\text{max}}|),$ (5.37) whereas the maximum possible output power, $|P_{\text{out},\triangle,\text{max}}|$, is obtained using

whereas the maximum possible output power, $|P_{\text{out}, \Box, \text{max}}|$, is obtained using a numerical maximum search. For practical reasons, the sum $D_1 + D_2$ and the weighting factor a,

$$D_{12} = D_1 + D_2, (5.38)$$

$$a = \frac{D_1}{D_{12}},\tag{5.39}$$

are used instead of D_1 and D_2 . With (5.36) and (5.38), the numerical solver adjusts φ and a in order to meet (5.26) and (5.27).

5.1.3 Measurement Results

In Figure 5.13, the waveforms of $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L1}(t)$, calculated with the numerical solver outlined in Section 4.6.1, are compared to the measured waveforms at $V_1 = 340$ V, $V_2 = 12$ V, and $P_2 = 2$ kW (extended triangular current mode modulation). The very good matching of calculated and measured waveforms [in particular regarding the current $i_{L1}(t)$] is essential with respect to an accurate prediction of the expected converter efficiency.

The converter efficiencies achieved for the DAB converter designed in Appendix A.2 (i.e. n = 16 and $L = 15.5 \,\mu\text{H}$) and with the extended triangular and trapezoidal current mode modulation schemes are depicted in Figure 5.14 (measured at $T_{\rm j} \approx 25^{\circ}$ C, cf. Chapter 4): compared to phase shift modulation (Figure 4.30), increased converter efficiencies are achieved for most operating points, in particular at power levels considerably below the rated power of 2 kW.

5.2 Minimum Loss Modulation

The extended triangular and trapezoidal current mode modulation schemes, developed in Section 5.1, facilitate a considerably more efficient converter operation than phase shift modulation (for most operating points, in particular at low power levels). The extended triangular and trapezoidal current mode modulation schemes, however, are developed based on the findings discussed in Section 4 rather than on an extensive efficiency optimization; in retrospect



Figure 5.13: (a) Calculated (cf. Section 4.6.1) and (b) measured voltage and current waveforms obtained with the extended triangular and trapezoidal current mode modulation schemes for $V_1 = 340$ V, $V_2 = 12$ V, and $P_2 = 2$ kW; for all measurements, the efficiency optimized DAB design with n = 16 and $L = 15.5 \,\mu\text{H}$ is employed (cf. Appendix A.2).



Figure 5.14: Predicted efficiencies (solid lines) and measured efficiencies (\Box) obtained with the extended triangular and trapezoidal current mode modulation schemes ($T_{\rm f} = 50 \,\mathrm{ns}$) and for different operating conditions; the efficiencies are calculated with the full DAB loss model discussed in Section 4; for all measurements, the efficiency optimized DAB design with n = 16 and $L = 15.5 \,\mu\mathrm{H}$ is employed (cf. Appendix A.2); the efficiency is measured according to Chapter 4, i.e. $T_{\rm j} \approx 25^{\circ}\mathrm{C}$.

on the findings presented in Chapter 3 it is conceivable that more effective DAB modulation schemes exist.

For a particular operating point (i.e. given V_1 , V_2 , and P_{out}), the optimal DAB converter parameters φ , D_1 , and D_2 (with respect to maximum efficiency) are calculated according to the optimization procedure discussed in Section 3.1.3 using the numerical solver discussed in Section 4.6.1. The result of this optimization procedure reveals, that the optimal operation of the DAB within the specified operation ranges (V_1 , V_2 , and P_{out}) requires fundamental mode transitions, which cause steps in the control parameters (φ , D_1 , and D_2) and consequently increases the controller complexity.⁵ However, with a minor decrease of the DAB efficiency being tolerated, especially at low output power, suboptimal modulation schemes with continuous control parameter functions are feasible, and still, converter operation close to maximum efficiency is achieved.

5.2.1 Numerical Efficiency Optimization

In order to determine a modulation strategy with respect to maximum converter efficiency, 100 particular operating points (Table 5.1) are optimized using the numerical solver and the loss model discussed in Section 4. The presented discussion summarizes the obtained results based on 12 representative operating points:

- $V_1 = 340 \text{ V}, V_2 = 12 \text{ V}, P_{\text{out}} = \pm 200 \text{ W}$ (Figure 5.17),
- $V_1 = 340 \text{ V}, V_2 = 12 \text{ V}, P_{\text{out}} = \pm 1 \text{ kW}$ (Figure 5.15),
- $V_1 = 340 \text{ V}, V_2 = 12 \text{ V}, P_{\text{out}} = \pm 2 \text{ kW}$ (Figure 5.16),
- $V_1 = 240 \text{ V}, V_2 = 11 \text{ V}, P_{\text{out}} = \pm 2 \text{ kW}$ (Figure 5.19),
- $V_1 = 240 \text{ V}, V_2 = 16 \text{ V}, P_{\text{out}} = 200 \text{ W}$ [Figure 5.20 (a)],
- $V_1 = 240 \text{ V}, V_2 = 16 \text{ V}, P_{\text{out}} = 1 \text{ kW}$ [Figure 5.20 (b)],
- $V_1 = 240 \text{ V}, V_2 = 16 \text{ V}, P_{\text{out}} = \pm 2 \text{ kW}$ (Figure 5.21).

⁵Due to the discontinuities in the control parameter functions, the inductor currents, the switching currents, the transformer magnetization, and the DAB efficiency change discontinuously, as well. The mentioned mode transitions thus need to be carefully considered in order to achieve a stable DAB operation.

Optimal DAB operation at $V_1 = 340$ V and $V_2 = 12$ V

If the DAB is operated with maximum efficiency at $V_1 = 340$ V, $V_2 = 12$ V, and $P_{\text{out}} = \pm 1$ kW, a DAB inductor current waveform $i_{L_1}(t)$ similar to the inductor current achieved with the extended triangular current mode modulation scheme results (Figure 5.15); merely, the LV side switch current at $t = t_1$ in Figure 5.15 (a) is equal to

$$I_{\rm S2,sw,opt} \approx 20 \,\mathrm{A}$$
 (5.40)

giving minimum switching losses (Figure 4.22).

The DAB exceeds the power limit of the triangular current mode modulation at rated output power, $P_{\text{out}} = \pm 2 \,\text{kW}$ (Figure 5.16), however, the obtained optimal modulation strategy is different to the trapezoidal current mode modulation, but rather similar to the optimal transition mode modulation presented in Section 3.1.3. For $P_{\text{out}} = \pm 2 \,\text{kW}$, the optimization algorithm employs $D_2 = 0.5$ and adjusts D_1 and φ in order to simultaneously obtain the required output power, low conduction losses on the LV side, and low switching losses on the LV side (i.e. minimum LV side inductor RMS currents are achieved, taking $I_{\text{S2,sw,opt}} \approx 20 \,\text{A}$ into account).

At low output power levels, the most efficient modulation method (Figure 5.17) is similar to the method illustrated in Figure 5.2, due to low switching and conduction losses: compared to the extended triangular current mode modulation, only a small inductor current circulates during the

	V_1	V_2	P_{out}
1	$340\mathrm{V}$	$12\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
2	$240\mathrm{V}$	$11\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
3	$240\mathrm{V}$	$13.5\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
4	$240\mathrm{V}$	$16\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
5	$345\mathrm{V}$	$11\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
6	$345\mathrm{V}$	$13.5\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
7	$345\mathrm{V}$	$16\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
8	$450\mathrm{V}$	$11\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
9	$450\mathrm{V}$	$13.5\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$
10	$450\mathrm{V}$	$16\mathrm{V}$	$\pm 200 \mathrm{W}, \pm 500 \mathrm{W}, \pm 1 \mathrm{kW}, \pm 1.5 \mathrm{kW}, \pm 2 \mathrm{kW}$

Table 5.1: For these 100 operating points, the optimal control parameters D_1 , D_2 , and φ are calculated with respect to maximum efficiency; the results are given in Table 5.2.



Figure 5.15: Calculated waveforms $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L_1}(t)$ for maximum efficiency at (a) $P_{out} = P_2 = 1 \text{ kW}$, and (b) $P_{out} = P_1 = -1 \text{ kW}$; $V_1 = 340 \text{ V}$, $V_2 = 12 \text{ V}$, n = 19, $L = 26.7 \text{ \mu}\text{H}$, and $f_{\rm S} = 100 \text{ kHz}$; due to the large magnetizing inductance ($L_{\rm M} = 2.7 \text{ mH}$), $i_{L_1} \approx i_{L_2}/n$ applies.



Figure 5.16: Calculated waveforms $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L_1}(t)$ for maximum efficiency at (a) $P_{out} = 2 \text{ kW}$, and (b) $P_{out} = -2 \text{ kW}$; $V_1 = 340 \text{ V}$, $V_2 = 12 \text{ V}$, n = 19, $L = 26.7 \text{ }\mu\text{H}$, $L_{M} = 2.7 \text{ }\text{mH}$, and $f_{S} = 100 \text{ kHz}$.



Figure 5.17: Calculated waveforms $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L_1}(t)$ for maximum efficiency at (a) $P_{out} = 200$ W, and (b) $P_{out} = -200$ W; $V_1 = 340$ V, $V_2 = 12$ V, n = 19, $L = 26.7 \mu$ H, $L_M = 2.7 \text{ mH}$, and $f_S = 100$ kHz.

freewheeling time interval $[t_3 < t < T_S/2$ in Figures 5.17 (a) and (b)] and the LV side full bridge switches with $I_{S2,sw,opt}$. With this modulation scheme, however, discontinuous control parameters D_1 , D_2 , and φ result [Figure 5.18 (a) for P_{out} between 800 W and 900 W], which presents difficulties to the practical implementation (e.g. a particular sequence of different modulation schemes needs to be implemented in order to avoid the depicted discontinuities, cf. Figure 5.3). Therefore, at low power levels, the extended triangular current modulation scheme depicted in Figure 5.15 is preferred.

Optimal DAB operation at $V_1 = 240 \text{ V}, V_2 = 11 \text{ V}, \text{ and } P_{out} = \pm 2 \text{ kW}$

At $V_1 = 240$ V, $V_2 = 11$ V, and $P_{out} = \pm 2$ kW, the DAB operates close to its maximum possible power [cf. Table 1.4 and (3.15)]. There, according to the optimization procedure discussed in Section 3.1.3, phase shift modulation is employed to achieve minimum inductor RMS currents. The actual efficiency optimization based on the DAB loss model (Section 4), however, identifies $D_1 = 0.5$ and $D_2 < 0.5$ to yield most efficient converter operation at very high power levels (Figure 5.19), due to a reduction of the LV side switching losses.



Figure 5.18: (a) Duty cycles D_1 and D_2 required in order to achieve maximum converter efficiency for $V_1 = 340$ V, $V_2 = 12$ V, and power being transferred from the HV port to the LV port: for P_{out} between 800 W and 900 W, a large step occurs [a second step of D_2 occurs between 1.6 kW and 1.7 kW, which is, however, covered using linear interpolation, cf. (5.50)]; (b) D_1 and D_2 for maximum converter efficiency, $V_1 = 240$ V, $V_2 = 16$ V, power being transferred to the LV port: for P_{out} between 1.2 kW and 1.4 kW, $D_2(P_{out})$ changes considerably. Employed DAB: n = 19, $L = 26.7 \,\mu\text{H}$.



Figure 5.19: Calculated waveforms $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L_1}(t)$ for maximum efficiency at (a) $P_{out} = 2 \text{ kW}$, and (b) $P_{out} = -2 \text{ kW}$; $V_1 = 240 \text{ V}$, $V_2 = 11 \text{ V}$, n = 19, $L = 26.7 \text{ }\mu\text{H}$, $L_{M} = 2.7 \text{ mH}$, and $f_{S} = 100 \text{ kHz}$.

Optimal DAB operation at $V_1 = 240$ V and $V_2 = 16$ V

At $V_1 = 240$ V, $V_2 = 16$ V, and $P_{out} = 200$ W or $P_{out} = 1$ kW (Figure 5.20), the optimization algorithm suggests the DAB operation mode 1b [Figure 3.15 (i)] to be used, since minimum switching losses are achieved, there. However, similar to the DAB operation at $V_1 = 340$ V, $V_2 = 12$ V, and $P_{out} = 200$ W, difficulties arise with respect to the practical realization and the complexity required to calculate the modulation parameters increases [e.g. for P_{out} between 1.2 kW and 1.4 kW in Figure 5.18 (b)]. Since the efficiency achieved with the extended triangular current mode modulation (mode b, Section 5.1.1, Figure 5.5) is close to the maximum achievable efficiency [cf. Figure 5.28 (b) for $P_{out} < 1.2$ kW], the less complex extended triangular current mode modulation is preferred over the modulation method suggested by the numerical optimization algorithm.

The DAB exceeds the maximum possible output power of the extended triangular current mode modulation (mode b) at $V_1 = 240$ V, $V_2 = 16$ V, and $P_{\text{out}} = \pm 2$ kW. There, $D_1 = 0.5$ and $D_2 < 0.5$ result for maximum efficiency (low conduction losses and low switching losses are achieved, cf. optimal DAB operation at $V_1 = 240$ V, $V_2 = 11$ V, and $P_{\text{out}} = \pm 2$ kW).



Figure 5.20: Calculated waveforms $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L_1}(t)$ for maximum efficiency at (a) $P_{out} = 200$ W, and (b) $P_{out} = 1$ kW; $V_1 = 240$ V, $V_2 = 16$ V, n = 19, $L = 26.7 \mu$ H, $L_M = 2.7 \text{ mH}$, and $f_S = 100$ kHz.



Figure 5.21: Calculated waveforms $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L_1}(t)$ for maximum efficiency at (a) $P_{out} = 2 \text{ kW}$, and (b) $P_{out} = -2 \text{ kW}$; $V_1 = 240 \text{ V}$, $V_2 = 16 \text{ V}$, n = 19, $L = 26.7 \text{ }\mu\text{H}$, $L_M = 2.7 \text{ }\text{mH}$, and $f_S = 100 \text{ kHz}$.

5.2.2 Suboptimal Modulation Schemes

The discussions presented in Section 5.2.1 enable the development of suboptimal modulation schemes, which facilitate the DAB to be operated close to its maximum efficiency.

The proposed modulation schemes try to minimize the impact of the most important DAB loss mechanisms:

- conduction losses,
- HV side switching losses (ZVS is required), and
- LV side switching losses [during switching, large currents should be avoided; switching processes with I_{S2,sw} = I_{S2,sw,opt}, cf. (5.40), generate minimum switching losses].

Suboptimal DAB Operation at Low Power Levels

According to Section 5.2.1, the extended triangular current mode modulation scheme is employed at low output power levels. The results of the optimization procedure discussed in Section 5.2.1 suggest to select the value of the circulating current, I_0 , as small as possible to achieve low conduction losses and ZVS on the HV side; I_0 is therefore selected according to (5.23). The results of the optimization further suggest to employ LV side ZVS with $I_{S2,sw,opt}$, since lowest losses occur, there. Thus, a modified value is used for T_{I_0} [cf. (5.24)],

$$T_{I_0} = \frac{L\left(|I_0| + \frac{20\,\mathrm{A}}{n}\right)}{V_1} \tag{5.41}$$

(on the assumption of negligible magnetizing currents and negligible losses during $0 < t < T_{I_0}$).

The applicable power range of the extended triangular current mode modulation is given with (5.28) and (5.32):

$$(|P_{\text{out}}| \le |P_{\text{out},\triangle,\text{a,max}}| \land V_2 < V_{2,\text{lim}}) \lor (|P_{\text{out}}| \le |P_{\text{out},\triangle,\text{b,max}}| \land V_2 \ge V_{2,\text{lim}}).$$
(5.42)

Suboptimal DAB Operation at High Power Levels

For power levels exceeding the limits given with (5.42), the modulation modes 3b or 7b (depending on the direction of power transfer) are employed [cf. Figures 3.15 (k) and (l)].⁶ Thus, the minimum freewheeling time $T_{\rm f}$,

 $^{^{6}\}mathrm{The}$ presented explanation confines to a power transfer direction from the HV port to the LV port.

needed to obtain $P_{\text{out}, \Delta, a, \text{max}}$ and $P_{\text{out}, \Delta, b, \text{max}}$, is set to zero in order to facilitate a seamless transition between the extended triangular current mode modulation schemes and the suboptimal high power modulation schemes based on modulation modes 3b or 7b. The proposed high power modulation schemes furthermore require the calculations of D_1 , D_2 , and φ to be considered separately for $V_2 \geq V_{2,\text{lim}}$ and $V_2 < V_{2,\text{lim}}$.

 $V_2 \geq V_{2,\text{lim}}$ (Figure 5.22). At $P_{\text{out}} = P_{\text{out},\triangle,\text{b,max}}$ and with $T_{\text{f}} = 0$, the duty cycles $D_1 = 0.5$ and $D_2 < 0.5$ result [Figure 5.22 (a)]; furthermore, hard switching occurs on the LV side at $t = t_1 = 0$ with $I_{\text{S2,sw},\triangle,\text{b,max}} = i_{L_2}(0) < 0$. With increasing output power D_2 and φ increase, and the current $i_{L_2}(t_1)$ becomes equal to $I_{\text{S2,sw,opt}}$ at $P_{\text{out}} = P_{\text{out},\text{opt},\text{b,min}}$ [Figure 5.22 (b)]:

$$[i_{L_1}(D_1, D_2, \varphi, t = 0) = I_{0,d} \land i_{L_2}(D_1, D_2, \varphi, t = t_1) = I_{S2,sw,opt} \land D_1 = 0.5]$$

$$\Leftrightarrow P_{out}(D_1, D_2, \varphi) = P_{out,opt,b,min}, \text{ search variables: } D_2, \varphi. \quad (5.43)$$

For $|P_{\text{out},\triangle,b,\max}| < |P_{\text{out}}| \leq |P_{\text{out},\text{opt},b,\min}|$, a numerical search algorithm seeks for suitable values of D_2 and φ (D_1 remains equal to 0.5) in order to satisfy

$$P_{\text{out}}(D_1 = 0.5, D_2, \varphi) = P_{\text{out},d},$$

$$i_{L_2}(D_1 = 0.5, D_2, \varphi, t = t_1) = I_{\text{S2,sw,d}}$$

$$\forall V_2 \ge V_{2,\text{lim}} \land |P_{\text{out},\triangle,\text{b,max}}| < |P_{\text{out}}| \le |P_{\text{out},\text{opt,b,min}}| \quad (5.44)$$

employing a linear transition for the LV side current $i_{L_2}(t_1)$,

$$I_{S2,sw,d} = I_{S2,sw,\triangle,b,max} + (I_{S2,sw,opt} - I_{S2,sw,\triangle,b,max}) \cdot \frac{P_{out,d} - P_{out,\triangle,b,max}}{P_{out,opt,b,min} - P_{out,\triangle,b,max}}.$$
 (5.45)

The maximum achievable output power, $P_{\text{out,max}}$, is approximately determined using phase shift modulation and a numerical maximum search:

$$(D_1 = 0.5 \land D_2 = 0.5) \Leftrightarrow P_{\text{out,max}} \approx \max[P_{\text{out}}(D_1, D_2, \varphi)], \text{ search var.: } \varphi.$$
(5.46)

For output power levels between $P_{\text{out,opt,b,min}}$ and $P_{\text{out,max}}$, $D_1 = 0.5$ remains and D_2 and φ are adjusted in order to obtain the required output power and maximum efficiency, provided that the LV side switching current $i_{S2}(t_1) = i_{L_2}(t_1)$ remains greater or equal than $I_{S2,sw,opt}$ [Figure 5.22 (c)], i.e. a numerical search algorithm seeks for D_2 and φ in order to satisfy:

$$P_{\text{out}}(D_1, D_2, \varphi) = P_{\text{out},d},$$

$$\eta(D_1, D_2, \varphi) = \max[\eta(D_1, D_2, \varphi)],$$

$$i_{L_2}(D_1, D_2, \varphi, t = t_1) > I_{\text{S2,sw,opt}}$$

$$\forall V_2 \ge V_{2,\text{lim}} \land |P_{\text{out},\text{opt},\text{b,min}}| < |P_{\text{out}}| \le |P_{\text{out},\text{max}}|. \quad (5.47)$$



Figure 5.22: Calculated waveforms $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L_1}(t)$ for $V_1 = 240$ V and $V_2 = 16$ V, using the suboptimal modulation schemes: (a) $P_{out} = 1.4$ kW, (b) $P_{out} = 1.5$ kW, and (c) $P_{out} = 2.5$ kW; n = 19, $L = 26.7 \mu$ H, $L_M = 2.7$ mH (thus, $i_{L_1} \approx i_{L_2}/n$ applies), and $f_S = 100$ kHz; Figures (a) and (b) depict the limits of the respective operating modes: (a) $P_{out} = P_{out, \triangle, b, max}$; (b) $P_{out} = P_{out, opt, b, min}$.

 $V_2 < V_{2,\text{lim}}$ (Figure 5.23). At $P_{\text{out}} = P_{\text{out},\triangle,a,\max}$, $D_1 < 0.5$ and $D_2 < 0.5$ result (due to the time interval $0 < t < T_{I_0}$, required to generate the freewheeling current I_0). Moreover, hard switching occurs on the LV side, e.g. at t = 0, with $I_{\text{S2,sw},\triangle,a,\max} = i_{L_2}(0) < 0$ [Figure 5.23 (a)]. If $|P_{\text{out}}|$ exceeds $|P_{\text{out},\triangle,a,\max}|$, the conditions $D_2 = 0.5$, $i_{L_1}(0) \ge I_{0,d}$, and $i_{L_2}(t_1) = I_{\text{S2,sw,opt}}$ need to be satisfied in order to achieve low conduction losses and low switching losses; the respective minimum power level $P_{\text{out,opt,a,min}}$ is thus determined using

$$[i_{L_1}(D_1, D_2, \varphi, t = 0) = I_{0,d} \land i_{L_2}(D_1, D_2, \varphi, t = t_1) = I_{S2,sw,opt} \land D_2 = 0.5]$$

$$\Leftrightarrow P_{out}(D_1, D_2, \varphi) = P_{out,opt,a,min}, \text{ search variables: } D_1, \varphi. \quad (5.48)$$

The resulting voltage and current waveforms for $V_1 = 340$ V, $V_2 = 12$ V, and $P_{\text{out}} = P_{\text{out,opt,a,min}}$ are depicted in Figure 5.23 (b). For output power levels between $P_{\text{out,}\triangle,\text{a,max}}$ and $P_{\text{out,opt,a,min}}$, the modulation parameters D_1 , D_2 , and φ are calculated using

$$P_{\text{out}}(D_1, D_2, \varphi) = P_{\text{out},d},$$

$$i_{L_1}(D_1, D_2, \varphi, t = 0) = I_0,$$

$$i_{L_2}(D_1, D_2, \varphi, t = t_1) = I_{\text{S2,sw,d}}$$

$$\forall V_2 < V_{2,\text{lim}} \land |P_{\text{out}, \bigtriangleup, \text{max}}| < |P_{\text{out}}| \le |P_{\text{out},\text{opt},\text{a,min}}|. \quad (5.49)$$

For the LV side current $i_{L_2}(t_1)$, a linear transition is employed:

$$I_{S2,sw,d} = I_{S2,sw,\Delta,a,max} + (I_{S2,sw,opt} - I_{S2,sw,\Delta,a,max}) \cdot \frac{P_{out,d} - P_{out,\Delta,a,max}}{P_{out,opt,a,min} - P_{out,\Delta,a,max}}.$$
 (5.50)

With increasing output power, $D_2 = 0.5$ is kept constant and D_1 and φ are calculated in order to satisfy

$$P_{\text{out}}(D_1, D_2, \varphi) = P_{\text{out},d},$$

$$i_{L_2}(D_1, D_2, \varphi, t = t_1) = I_{\text{S2,sw,opt}}$$

$$\forall V_2 < V_{2,\text{lim}} \land |P_{\text{out,opt,a,min}}| < |P_{\text{out}}| \le |P_{\text{out,opt,a,hi}}|. \quad (5.51)$$

At $|P_{\text{out}}| = |P_{\text{out,opt,a,hi}}|$, D_1 is equal to 0.5 [Figure 5.23 (c)]:

$$[i_{L_2}(D_1, D_2, \varphi, t = t_1) = I_{S2, sw, opt} \land D_1 = 0.5 \land D_2 = 0.5]$$

$$\Leftrightarrow P_{out}(D_1, D_2, \varphi) = P_{out, opt, a, hi}, \text{ search variable: } \varphi. \quad (5.52)$$

If the required output power level exceeds $P_{\text{out,opt,a,hi}}$, $D_2 < 0.5$ is employed in order to reduce the LV side switching losses [Figure 5.23 (d)]. With



Figure 5.23: Calculated waveforms $v_{AC1}(t)$, $v_{AC2}(t)$, and $i_{L_1}(t)$ for DAB operation at $V_1 = 340$ V and $V_2 = 12$ V, using the suboptimal modulation scheme: (a) $P_{out} = 1.7$ kW, (b) $P_{out} = 1.8$ kW, (c) $P_{out} = 2.1$ kW, and (d) $P_{out} = 2.5$ kW; n = 19, $L = 26.7 \mu$ H, $L_M = 2.7$ mH (thus, $i_{L_1} \approx i_{L_2}/n$ applies), and $f_S = 100$ kHz; Figures (a), (b), and (c) depict the limits of the respective operating modes: (a) $P_{out} = P_{out,\triangle,a,max}$; (b) $P_{out} = P_{out,opt,a,min}$; (c) $P_{out} = P_{out,opt,a,hi}$.



Figure 5.24: DAB operating modes for $V_1 = 260$ V, different voltages V_2 , different output power levels P_{out} , n = 19, and $L = 26.7 \,\mu\text{H}$; mode d denotes the voltage sequences 3b or 7b [depending on the direction of power transfer; Figures 3.15 (k) and (l)]; mode c would denote the trapezoidal current mode modulation and is not used for the suboptimal current mode modulation.

decreasing D_2 , however, the inductor RMS currents I_{L_1} and I_{L_2} increase. Thus, D_2 and φ are determined with respect to maximum total efficiency:

$$P_{\text{out}}(D_1 = 0.5, D_2, \varphi) = P_{\text{out,d}},$$

$$\eta(D_1 = 0.5, D_2, \varphi) = \max[\eta(D_1 = 0.5, D_2, \varphi)],$$

$$i_{L_2}(D_1 = 0.5, D_2, \varphi, t = t_1) > I_{\text{S2,sw,opt}}$$

$$\forall V_2 < V_{2,\text{lim}} \land |P_{\text{out,opt,a,hi}}| < |P_{\text{out}}| \le |P_{\text{out,max}}|. \quad (5.53)$$

Summary on the Suboptimal Modulation Schemes

Figure 5.24 presents an overview over the different operating modes employed for the suboptimal modulation schemes detailed in this Section ($V_1 = 260 \text{ V}$, $11 \text{ V} \le V_2 \le 16 \text{ V}$, $0 \le P_{\text{out}} \le 1.5 \text{ kW}$). At low power levels (i.e. $|P_{\text{out}}| < P_{\text{out},\Delta,a,\max}$ or $|P_{\text{out}}| < P_{\text{out},\Delta,b,\max}$) the extended triangular current mode modulation schemes are employed (either mode a for $V_2 < V_{2,\text{lim}}$ or mode b for $V_2 > V_{2,\text{lim}}$). At high power levels, the voltage sequences 3b or 7b [depending on the direction of power transfer; cf. Figures 3.15 (k) and (l)], denoted with mode d in Figure 5.24, are used. The maximum power achieved with the extended triangular current mode modulation schemes, however, is slightly smaller than the minimum power achieved with the high power modulation schemes (i.e. $|P_{\text{out},\text{opt},a,\min}|$ or $|P_{\text{out},\text{opt},b,\min}|$, depending on $V_2/V_{2,\text{lim}}$) and therefore, the missing power levels are covered using mode d converter operation and linearly interpolated modulation parameters. Furthermore, for $V_2 < V_{2,\text{lim}}$, the cases $|P_{\text{out},\text{opt},a,\min}| <$ $|P_{\text{out}}| \leq |P_{\text{out},\text{opt},a,\text{hi}}|$ and $|P_{\text{out}}| > |P_{\text{out},\text{opt},a,\text{hi}}|$ are treated separately in order to achieve a reduction of the switching losses on the LV side.

5.2.3 Results

In order to compare the improvement achieved with the extended triangular and trapezoidal current mode modulation schemes and the suboptimal modulation schemes, the average efficiencies, calculated over the operating points listed in Table 5.1, are depicted in Figure 5.25 (the calculated efficiency values are given in Table 5.2). Furthermore, the losses calculated at the edges of the specified voltage operating range are depicted in Figure 5.26 for different output power levels and power being transferred to the LV port.

With respect to the specified voltage and power ranges (Table 1.4), phase shift modulation obviously results in the lowest average efficiency. A significant improvement is achieved with the extended triangular and trapezoidal current mode modulation being employed; there, the calculated average efficiency [Figure 5.25 (d)] increases from 83.5% to 89.0%. For most operating points, further efficiency improvements are achieved with the suboptimal modulation schemes (Section 5.2.2) and the optimal modulation schemes (Section 5.2.1): the respective average efficiencies increase to $\overline{\eta} = 90.3\%$ and $\overline{\eta} = 91.6\%$ [Figure 5.25 (d)]. However, for $V_1 = 240$ V, $V_2 = 11$ V, and $P_{\rm out} = 2 \,\rm kW$, a power dissipation of $P_{\rm t} = 253 \,\rm W$ is calculated for the extended trapezoidal current mode modulation scheme whereas $P_{\rm t} = 291$ W results with the suboptimal modulation scheme [cf. Figure 5.25 (c) and Figure 5.26 (a)]. The reason for this result is the reduced inductance required to achieve 2 kW output power for the extended trapezoidal current mode modulation $(L = 18.7 \,\mu\text{H})$. Thus, at a given operating point, fixed duty cycles D_1 and D_2 , and $L = 18.7 \,\mu\text{H}$, the employed phase shift angle φ is less than the phase shift angle required for $L = 26.7 \,\mu\text{H}$. Consequently, at high power levels, a decrease of the transformer RMS currents may be achieved by reducing the inductance values: with $L = 18.7 \,\mu\text{H}$, $V_1 = 240 \,\text{V}$, $V_2 = 11 \,\text{V}$, $P_{\text{out}} = 2 \,\text{kW}$, and the suboptimal modulation schemes being employed, the total losses de-



Figure 5.25: Calculated mean efficiencies for the different modulation schemes [phase shift modulation, extended triangular and trapezoidal current mode modulation (Section 5.1), suboptimal modulation (Section 5.2.2), and optimal modulation (Section 5.2.1)], averaged over all regarded voltages V_1 and V_2 (cf. Table 5.2) and different output power levels: (a) $P_{\text{out}} = \pm 200 \text{ W}$, (b) $P_{\text{out}} = \pm 1 \text{ kW}$, (c) $P_{\text{out}} = \pm 2 \text{ kW}$, (d) average efficiency regarding all 100 operating points given in Table 5.2; n = 19, $L_{\text{M}} = 2.7 \text{ mH}$, and $f_{\text{S}} = 100 \text{ kHz}$; $L = 26.7 \text{ \mu H}$ applies for Figures (a), (c), and (d); $L = 18.7 \text{ \mu H}$ applies for Figure (b).



Figure 5.26: Calculated total converter losses for the different modulation schemes [phase shift modulation, extended triangular and trapezoidal current mode modulation (Section 5.1), suboptimal modulation (Section 5.2.2), and optimal modulation (Section 5.2.1)], power being transferred to the LV port, and different operating points: (a) $V_1 = 240$ V, $V_2 = 11$ V, (b) $V_1 = 240$ V, $V_2 = 16$ V, (c) $V_1 = 450$ V, $V_2 = 11$ V, (d) $V_1 = 450$ V, $V_2 = 16$ V; n = 19, $L_M = 2.7$ mH, and $f_S = 100$ kHz; L = 26.7 µH applies for phase shift modulation, suboptimal modulation, and optimal modulation; L = 18.7 µH is required for the extended triangular and trapezoidal current mode modulation.



Figure 5.27: Efficiencies obtained with $V_1 = 240$ V, $V_2 = 11$ V, $P_{out} = 2$ kW, n = 19, and $f_S = 100$ kHz for different modulation schemes and different inductance values: at high power levels and low DC voltages V_1 and V_2 , the losses decrease with decreasing L, due to a reduction of the transformer RMS currents.

crease from 291 W to 162 W (Figure 5.27). A reduction of L, however, causes a reduced average efficiency at low and medium power levels if the DAB converter is operated within wide input and output voltage ranges.⁷ Therefore, L is designed with respect to maximum average efficiency (Appendix A.1).

In Figure 5.28, the efficiency achieved with the suboptimal modulation scheme is compared to the maximum achievable efficiency and the efficiency achieved with phase shift modulation. Moreover, two different operating voltages are considered in Figure 5.28: $V_1 = 340$ V, $V_2 = 12$ V [Figure 5.28 (a)] and $V_1 = 240$ V, $V_2 = 16$ V [Figure 5.28 (b)]. For $V_1 = 340$ V, $V_2 = 12$ V, and phase shift modulation, high efficiency is only achieved for $P_{\text{out}} > 2$ kW. In contrast, with the suboptimal modulation schemes (detailed in Section 5.2.2),

⁷Consider, for example, phase shift modulation at zero output power (i.e. Figure 3.4 with $T_{\varphi} = 0$). There, the inductor current changes according to the second equation given in (3.11) and therefore, the inductor RMS current is proportional to $|V_1 - nV_2|/L$.

the DAB converter operates close to the maximum achievable efficiency. Due to the use of the extended triangular current mode modulation at low and medium power levels the efficiency is approximately 1% to 2% below the maximum possible efficiency for $P_{\rm out} < 1000$ W and less than 1% below the maximum achievable efficiency for $P_{\rm out} > 1000$ W [Figure 5.28 (a)].

With the calculation procedure outlined in Section 4.6.1, employed to accurately predict the converter efficiency for the different operating points, good matching between calculated and measured waveforms is achieved (Figure 5.29). Thus, the calculated current RMS values (required to calculate the conduction and copper losses) and the instantaneous current values (needed to calculate the switching losses) are close to the values obtained with the DAB hardware prototype.

The converter efficiency is measured for the suboptimal modulation schemes and different operating points in order to verify the calculated results (for these measurements the optimized DAB designed in Appendix A.2 with n = 16 and $L = 22.4 \,\mu\text{H}$ is employed; the efficiency measurement is conducted at room temperature according to Chapter 4). Figure 5.30 illustrates the result; the \Box -symbols indicate measured values and solid lines denote the calculated efficiencies. The depicted results show good matching between calculated and measured results; accordingly, the employed loss model (Chapter 4) and the numerical calculation outlined in Section 4.6.1 are adequate for determining the suboptimal modulation schemes. Finally, the efficiencies calculated within the input and output voltage ranges specified in Table 1.4 are depicted in Figure 5.31 for the optimized DAB with n = 16 and $L = 22.4 \,\mu\text{H}$ and for different output power levels (cf. Figure 5.1).

5.2.4 Conclusion

With the presented modulation schemes, converter operation close to maximum converter efficiency is achieved:

- the extended triangular and trapezoidal current mode modulation schemes facilitate a comparably simple and intuitive method to obtain the modulation parameters D_1 , D_2 , and φ ; however, the achieved average converter efficiency [cf. (A.1)] is slightly less than the average efficiency achieved with the suboptimal modulation schemes (Appendix A.2);
- the suboptimal modulation schemes achieve a converter efficiency close to the maximum converter efficiency (Figure 5.28), though, compared to the extended trapezoidal current mode modulation scheme, the calculation of D_1 and D_2 is considerably more complex.

The key achievements with these modulation schemes are:



Figure 5.28: Efficiencies calculated for different modulation schemes and two different operating points: (a) $V_1 = 340$ V, $V_2 = 12$ V and (b) $V_1 = 240$ V, $V_2 = 16$ V; the solid line denotes the maximum achievable efficiency calculated according to Section 5.2.1, the dashed line denotes the efficiency achieved with the suboptimal modulation schemes (Section 5.2.2) and the line with alternating dashes and dots marks the efficiency obtained with phase shift modulation. With the optimized modulation scheme, highly efficient converter operation, close to maximum efficiency, is achieved (n = 19, $L = 26.7 \,\mu$ H, $T_j = 25^{\circ}$ C).



Figure 5.29: (a) Calculated and (b) measured voltage and current waveforms obtained with the suboptimal modulation schemes for $V_1 = 340$ V, $V_2 = 12$ V, and $P_2 = 2$ kW; for all measurements, the efficiency optimized DAB design with n = 16 and $L = 22.4 \,\mu\text{H}$ is employed (cf. Appendix A.2); at the nominal operating point, the instantaneous switch current during switching of the LV side switches is close to zero and therefore, the overvoltage switching spikes contained in the calculation of $v_{AC2}(t)$ (Section 4.3.3) are almost negligible.



Figure 5.30: Predicted efficiencies (solid lines) and measured efficiencies (\Box) obtained with the suboptimal modulation schemes and for different operating conditions; the efficiencies are calculated with the full DAB loss model discussed in Section 4; for all measurements, the efficiency optimized DAB design with n = 16 and $L = 22.4 \,\mu\text{H}$ is employed (cf. Appendix A.2).



Figure 5.31: Efficiencies predicted for the optimized DAB with n = 16and $L = 22.4 \,\mu\text{H}$ within the input and output voltage ranges specified in Table 1.4, different output power levels, and different directions of power transfer (cf. Appendix A.2): (a) $\text{LV} \leftarrow \text{LV}$, $P_{\text{out}} = 1 \,\text{kW}$; (a) $\text{HV} \leftarrow \text{LV}$, $P_{\text{out}} = 1 \,\text{kW}$; (a) $\text{LV} \leftarrow \text{HV}$, $|P_{\text{out}}| = 1 \,\text{kW}$; (c) $\text{HV} \leftarrow \text{LV}$, $P_{\text{out}} = 2 \,\text{kW}$; (d) $\text{LV} \leftarrow \text{HV}$, $|P_{\text{out}}| = 2 \,\text{kW}$. The efficiency calculation considers the effect of the linear interpolation used by the DSP to determine the actual modulation parameters (Appendix E.2); this causes the irregular characteristics of the depicted contour lines.

- 1. low transformer RMS currents (low conduction and copper losses);
- 2. ZVS on the HV side;
- 3. switch currents close to zero (if possible) on the LV side (low switching losses on the LV side).

With the use of these modulation schemes, efficiency improvements are achieved without changing (and in particular extending) the DAB hardware (except for the selection of n and L according to Section A.2). Further efficiency improvements may be achieved by reducing the magnetizing inductance $L_{\rm M}$: the resulting circulating transformer currents extend the ZVS range [50] and thus, a reduction of the HV side switching losses can be achieved, while the RMS transformer on the LV side can be reduced. An according investigation, however, confirms that this modification is only effective at medium power levels; at high power levels ZVS is naturally achieved with $L_{\rm M} \rightarrow \infty$ (thus, a reduced magnetizing inductance causes the conduction losses to increase) and at low power levels the voltage-time-area applied to $L_{\rm M}$ is insufficient to obtain the current level required for ZVS. For the presented converter, it was tried to achieve a high efficiency without the use of additional hardware components; if additional power components can be employed, a multitude of different hardware extensions, used to achieve ZVS for the full operating range at arbitrary currents $i_L(t_{sw})$, are discussed in literature (e.g. resonant pole circuits [110,111], auxiliary resonant pole circuits [85,112,113], and ZVS-FB converters using a modified HF transformer with center tapping [82, 83]).

Furthermore, the employed loss model is evaluated at room temperature $(T = T_j = 25^{\circ}C)$ and therefore, the presented calculations and measurements do not consider the impact of increased component temperatures (cf. Chapter 4). However, comparably low losses result with the suboptimal modulation schemes and thus, the efficiencies obtained during continuous converter operation are similar to the efficiencies obtained for a limited run time of 30 seconds (Figure 5.32).


□ Short-time operation (30 s per measurement)

 \triangle Continuous operation (> 1 h continuous run, > 10 min. per measurement)

Figure 5.32: Comparison of different efficiency measurements: the \Box -symbols denote the efficiencies measured after a short operating time of 30 seconds (i.e. $T_{\rm j} \approx 25^{\circ}$ C, cf. Chapter 4) and the \triangle -symbols denote the efficiencies measured during continuous converter operation (at room temperature); $V_1 = 340$ V, $V_2 = 12$ V, suboptimal modulation schemes; (a) power transfer from the HV port to the LV port; (b) opposite power transfer (LV \rightarrow HV). Due to the effective water cooler and the comparably low losses obtained with the suboptimal modulation schemes, similar efficiencies are obtained for a short operating time and for continuous operation.

			Extenu	ded tria	ngular						
	Phase	e shift	and	trapezo	idal	50	Suboptim	al		Optimal	
$P_{ m out}$	modu	lation	cn	rrent m_{ℓ}	$_{ode}$	r.	nodulatic	u	m	odulatic	u
	L = 2	$H\eta$ 7.9	ш	odulatic	u_{i}	Г	$i = 26.7 \mu$	H	T	$= 26.7 \mu$	H
			T	$= 18.7 \mu$	٤H						
	$P_{\rm t,PS}$	η_{PS}	$P_{\rm t,TT}$	η_{TT}	$\Delta \eta_{\mathrm{TT}}$	$P_{\rm t, subopt}$	$\eta_{ m subopt}$	$\Delta\eta_{ m subopt}$	$P_{\rm t,opt}$	η_{opt}	$\Delta\eta_{\rm opt}$
	$V_{1} = 3$	$40 V, V_2$	i = 12 V								
$200 \mathrm{W}$	92 W	68.4%	$54 \mathrm{W}$	78.9%	10.4%	$46 \mathrm{W}$	81.4%	12.9%	$34\mathrm{W}$	85.3%	16.9%
-200 W	88 W	69.5%	$49 \mathrm{W}$	80.3%	10.8%	$43\mathrm{W}$	82.1%	12.6%	$34\mathrm{W}$	85.6%	16.1%
500 W	$91 \mathrm{W}$	84.6%	60 W	89.3%	4.6%	$51 \mathrm{W}$	90.7%	6.0%	$43\mathrm{W}$	92.1%	7.5%
-500 W	$89 \mathrm{W}$	84.9%	$57 \mathrm{W}$	89.7%	4.8%	$50 \mathrm{W}$	90.8%	5.9%	$43\mathrm{W}$	92.1%	7.2%
$1 \mathrm{kW}$	92 W	91.6%	77 W	92.9%	1.3%	66 W	93.8%	2.2%	$61\mathrm{W}$	94.3%	2.7%
-1 kW	95 W	91.4%	78 W	92.8%	1.4%	68 W	93.7%	2.3%	$63 \mathrm{W}$	94.0%	2.7%
$1.5 \mathrm{kW}$	M 66	93.8%	M 66	93.8%	0.0%	86 W	94.6%	0.8%	$82\mathrm{W}$	94.8%	1.1%
-1.5 kW	$107 \mathrm{W}$	93.3%	$104\mathrm{W}$	93.5%	0.2%	M06	94.3%	1.0%	88 W	94.5%	1.2%
$2 \mathrm{kW}$	115 W	94.6%	$125 \mathrm{W}$	94.1%	-0.3%	$107 \mathrm{W}$	94.9%	0.3%	$107 \mathrm{W}$	94.9%	0.4%
-2 kW	$130 \mathrm{W}$	93.9%	$136 \mathrm{W}$	93.6%	-0.1%	$116 \mathrm{W}$	94.5%	0.6%	$116 \mathrm{W}$	94.5%	0.6%
	$V_{1} = 2$	$40 V, V_2$	= 11 V								
$200 \mathrm{W}$	36 W	84.7%	38 W	84.1%	-0.5%	35 W	85.3%	0.6%	31 W	86.5%	1.8%
-200 W	36 W	84.7%	$37 \mathrm{W}$	84.5%	-0.1%	$34\mathrm{W}$	85.6%	0.8%	$32 \mathrm{W}$	86.1%	1.4%
500 W	36 W	93.2%	$41\mathrm{W}$	92.4%	-0.8%	38 W	92.9%	-0.2%	$35 \mathrm{W}$	93.4%	0.2%
-500 W	37 W	93.0%	$42 \mathrm{W}$	92.3%	-0.6%	38 W	92.9%	-0.1%	$37 \mathrm{W}$	93.1%	0.1%
$1 \mathrm{kW}$	62 W	94.2%	58 W	94.5%	0.3%	$55 \mathrm{W}$	94.8%	0.6%	$55 \mathrm{W}$	94.8%	0.6%
-1 kW	$58 \mathrm{W}$	94.5%	$57 \mathrm{W}$	94.6%	0.0%	$55 \mathrm{W}$	94.8%	0.3%	$54\mathrm{W}$	94.9%	0.4%
$1.5 \mathrm{kW}$	$133 \mathrm{W}$	91.9%	106 W	93.4%	1.5%	$109 \mathrm{W}$	93.2%	1.3%	$109 \mathrm{W}$	93.2%	1.3%
-1.5 kW	116 W	92.8%	$100 \mathrm{W}$	93.7%	0.9%	$104 \mathrm{W}$	93.5%	0.7%	$103 \mathrm{W}$	93.6%	0.7%
$2 \mathrm{kW}$	354 W	84.9%	$253 \mathrm{W}$	88.8%	3.8%	$291 \mathrm{W}$	87.3%	2.4%	$291\mathrm{W}$	87.3%	2.4%
$-2 \mathrm{kW}$	$247 \mathrm{W}$	89.0%	$211\mathrm{W}$	90.5%	1.4%	$231\mathrm{W}$	89.6%	0.6%	$231\mathrm{W}$	89.7%	0.6%

Table 5.2: Calculated losses and efficiencies for different operating points and for phase shift modulation, extended triangular and trapezoidal current mode modulation (Section 5.1), optimized modulation (Section 5.2.2), and optimal modulation (Section 5.2.1)); n = 19, $L_{\rm M} = 2.7 \,\mathrm{mH}$, and $f_{\rm S} = 100 \,\mathrm{kHz}$; $L = 26.7 \,\mu\mathrm{H}$ applies for phase shift modulation, optimized modulation, and optimal modulation; $L = 18.7 \,\mu\mathrm{H}$ is required for the extended triangular and trapezoidal current mode modulation.

			Extend	ded tria	ngular						
	Phase	$e \ shift$	and	trapezo	idal	51	Suboptime	al		Optimal	
P_{out}	modu	lation	cn	rrent m_{u}	$_{ode}$	n	nodulatic	u	m	odulatic	u
	L = 2	$6.7 \mu { m H}$	m	codulatic	n	Γ	$i = 26.7 \mu$	Η	T	$= 26.7 \mu$	H
			L	$= 18.7 \mu$	٤H						
	$P_{\rm t,PS}$	η_{PS}	$P_{\rm t,TT}$	η_{TT}	$\Delta \eta_{\mathrm{TT}}$	$P_{\rm t, sub opt}$	$\eta_{ m subopt}$	$\Delta\eta_{ m subopt}$	$P_{\rm t,opt}$	η_{opt}	$\Delta\eta_{\rm opt}$
	$V_{1} = 2$	$40 \text{ V}, V_2$	i = 13.5	Λ							
$200 \mathrm{W}$	79 W	71.7%	63 W	76.1%	4.4%	52 W	79.4%	7.7%	36 W	84.7%	13.1%
-200 W	$98 \mathrm{W}$	67.1%	52 W	79.4%	12.4%	$46 \mathrm{W}$	81.2%	14.1%	35 W	85.1%	18.1%
500 W	$75 \mathrm{W}$	87.0%	71 W	87.5%	0.5%	59 W	89.4%	2.4%	$48 \mathrm{W}$	91.3%	4.2%
-500 W	$92 \mathrm{W}$	84.5%	60 W	89.3%	4.8%	$53 \mathrm{W}$	90.4%	5.9%	45 W	91.7%	7.2%
$1 \mathrm{kW}$	M 66	91.0%	89 W	91.9%	0.9%	M 69	93.5%	2.6%	09 W	93.5%	2.6%
-1 kW	$82 \mathrm{W}$	92.4%	75 W	93.0%	0.6%	$63 \mathrm{W}$	94.1%	1.7%	63 W	94.1%	1.7%
$1.5 \mathrm{kW}$	168 W	89.9%	$128 \mathrm{W}$	92.2%	2.2%	$114\mathrm{W}$	92.9%	3.0%	$114 \mathrm{W}$	92.9%	3.0%
-1.5 kW	$133 \mathrm{W}$	91.9%	109 W	93.2%	1.4%	$101 \mathrm{W}$	93.7%	1.8%	$101 \mathrm{W}$	93.7%	1.8%
$2 \mathrm{kW}$	293 W	87.2%	205 W	90.7%	3.5%	$209 \mathrm{W}$	90.5%	3.3%	$209 \mathrm{W}$	90.5%	3.3%
$-2 \mathrm{kW}$	215 W	90.3%	$169 \mathrm{W}$	92.2%	1.9%	$174 \mathrm{W}$	92.0%	1.7%	$174\mathrm{W}$	92.0%	1.7%
	$V_1 = 2$	$40 \text{ V}, V_2$	= 16 V								
$200 \mathrm{W}$	214 W	48.3%	82 W	70.9%	22.6%	M 73	75.0%	26.7%	47 W	80.8%	32.5%
-200 W	$238 \mathrm{W}$	45.6%	68 W	74.7%	29.1%	59 W	77.2%	31.5%	45 W	81.7%	36.0%
500 W	212 W	70.2%	105 W	82.6%	12.4%	85 W	85.5%	15.3%	68 W	88.0%	17.8%
-500 W	232 W	68.3%	$89 \mathrm{W}$	84.9%	16.6%	76 W	86.8%	18.5%	63 W	88.8%	20.5%
$1 \mathrm{kW}$	220 W	81.9%	146 W	87.2%	5.3%	117 W	89.5%	7.6%	$105 \mathrm{W}$	90.5%	8.5%
-1 kW	$224 \mathrm{W}$	81.7%	$124 \mathrm{W}$	89.0%	7.3%	$105 \mathrm{W}$	90.5%	8.8%	94 W	91.4%	9.7%
$1.5 \mathrm{kW}$	254 W	85.5%	191 W	88.7%	3.2%	$146 \mathrm{W}$	91.1%	5.6%	$141\mathrm{W}$	91.4%	5.9%
-1.5 kW	227 W	86.8%	160 W	90.4%	3.5%	$134\mathrm{W}$	91.8%	4.9%	$131\mathrm{W}$	92.0%	5.1%
$2 \mathrm{kW}$	364 W	84.6%	245 W	89.1%	4.5%	$212 \mathrm{W}$	90.4%	5.8%	$212 \mathrm{W}$	90.4%	5.8%
$-2 \mathrm{kW}$	$264 \mathrm{W}$	88.3%	200 W	90.9%	2.6%	$178 \mathrm{W}$	91.8%	3.5%	$178 \mathrm{W}$	91.8%	3.5%

Table 5.2 cont.: Calculated losses and efficiencies for different operating points and for phase shift modulation, extended triangular and trapezoidal current mode modulation (Section 5.1), optimized modulation (Section 5.2.2), and optimal modulation (Section 5.2.1)); n = 19, $L_{\rm M} =$ 2.7 mH, and $f_{\rm S} = 100$ kHz; $L = 26.7 \,\mu$ H applies for phase shift modulation, optimized modulation, and optimal modulation; $L = 18.7 \,\mu$ H is required for the extended triangular and trapezoidal current mode modulation.

			Extend	ded tria	ngular						
	Phase	e shift	and	trapezo	idal	01	Suboptime	al		Optimal	
$P_{ m out}$	modu	lation	cn	rrent ma	$_{ode}$	r	nodulatic	u	m	odulatic	u u
	L = 2	$6.7 \mu { m H}$	ш	odulatic	u	Г	$i = 26.7 \mu$	H	Γ	$= 26.7 \mu$	H
			L	$= 18.7 \mu$	Η						
	$P_{\rm t,PS}$	η_{PS}	$P_{\rm t,TT}$	η_{TT}	$\Delta \eta_{\mathrm{TT}}$	$P_{\rm t, subopt}$	$\eta_{ m subopt}$	$\Delta\eta_{ m subopt}$	$P_{\rm t,opt}$	η_{opt}	$\Delta\eta_{\rm opt}$
	$V_{1} = 3$	$45 V, V_2$	i = 11 V								
$200 \mathrm{W}$	107 W	65.0%	53 W	79.0%	13.9%	45 W	81.5%	16.5%	34 W	85.4%	20.3%
-200 W	103 W	66.0%	49 W	80.4%	14.4%	$43 \mathrm{W}$	82.2%	16.2%	34 W	85.5%	19.4%
500 W	107 W	82.4%	61 W	89.1%	6.6%	52 W	90.5%	8.1%	$44 \mathrm{W}$	91.9%	9.5%
-500 W	105 W	82.7%	59 W	89.4%	6.8%	52 W	90.6%	8.0%	$44 \mathrm{W}$	91.9%	9.2%
$1 \mathrm{kW}$	109 W	90.2%	82 W	92.4%	2.2%	71 W	93.4%	3.2%	65 W	93.9%	3.7%
-1 kW	$113 \mathrm{W}$	89.9%	$84\mathrm{W}$	92.3%	2.4%	72 W	93.3%	3.4%	68 W	93.6%	3.8%
$1.5 \mathrm{kW}$	118 W	92.7%	109 W	93.2%	0.5%	$94 \mathrm{W}$	94.1%	1.4%	90 W	94.4%	1.7%
-1.5 kW	$128 \mathrm{W}$	92.1%	115 W	92.9%	0.7%	M 66	93.8%	1.7%	97 W	94.0%	1.8%
$2 \mathrm{kW}$	137 W	93.6%	140 W	93.5%	-0.0%	$120 \mathrm{W}$	94.3%	0.8%	$120 \mathrm{W}$	94.4%	0.8%
$-2 \mathrm{kW}$	$156 \mathrm{W}$	92.8%	$153 \mathrm{W}$	92.9%	0.1%	$132 \mathrm{W}$	93.8%	1.0%	$132\mathrm{W}$	93.8%	1.0%
	$V_{1} = 3$	$45 \text{ V}, V_2$	i = 13.5	Λ							
$200 \mathrm{W}$	82 W	70.9%	56 W	78.2%	7.2%	48 W	80.7%	9.8%	35 W	84.9%	14.0%
-200 W	78 W	72.0%	51 W	79.8%	7.7%	$45 \mathrm{W}$	81.6%	9.6%	34 W	85.4%	13.4%
500 W	80 W	86.3%	60 W	89.3%	3.0%	52 W	90.6%	4.3%	$43~\mathrm{W}$	92.1%	5.9%
-500 W	78 W	86.6%	57 W	89.7%	3.2%	50 W	90.8%	4.3%	42 W	92.3%	5.7%
$1 \mathrm{kW}$	79 W	92.7%	73 W	93.2%	0.5%	63 W	94.1%	1.4%	57 W	94.6%	1.9%
-1 kW	$81\mathrm{W}$	92.5%	73 W	93.2%	0.7%	64 W	94.0%	1.5%	61 W	94.3%	1.8%
$1.5\mathrm{kW}$	$84\mathrm{W}$	94.7%	M 06	94.3%	-0.3%	78 W	95.0%	0.3%	74 W	95.3%	0.6%
-1.5 kW	90 W	94.3%	$94 \mathrm{W}$	94.1%	-0.1%	82 W	94.8%	0.5%	79 W	95.0%	0.6%
$2 \mathrm{kW}$	97 W	95.4%	110 W	94.8%	-0.5%	97 W	95.4%	0.0%	94 W	95.5%	0.2%
$-2 \mathrm{kW}$	$106 \mathrm{W}$	95.0%	$119 \mathrm{W}$	94.4%	-0.5%	$100 \mathrm{W}$	95.3%	0.3%	W 66	95.3%	0.3%

Table 5.2 cont.: Calculated losses and efficiencies for different operating points and for phase shift modulation, extended triangular and trapezoidal current mode modulation (Section 5.1), optimized modulation (Section 5.2.2), and optimal modulation (Section 5.2.1)); n = 19, $L_{\rm M} =$ 2.7 mH, and $f_{\rm S} = 100$ kHz; $L = 26.7 \,\mu$ H applies for phase shift modulation, optimized modulation, and optimal modulation; $L = 18.7 \,\mu$ H is required for the extended triangular and trapezoidal current mode modulation.

			Extend	ded tria	ngular						
	Phase	e shift	and	trapezo	idal		Suboptim	al		Optimal	
P_{out}	modu	lation	cn	rrent ma	ode	r.	nodulatic	u	m	vodulatic	n
	L = 2	$6.7 \mu \mathrm{H}$	m	odulatic	uc	Г	$i = 26.7 \mu$	H	Γ	$= 26.7 \mu$	H
			L	$= 18.7 \mu$	Η						
	$P_{\rm t, PS}$	η_{PS}	$P_{\rm t,TT}$	η_{TT}	$\Delta \eta_{\mathrm{TT}}$	$P_{\rm t, subopt}$	$\eta_{ m subopt}$	$\Delta\eta_{ m subopt}$	$P_{\rm t,opt}$	η_{opt}	$\Delta\eta_{\rm opt}$
	$V_{1} = 3$	$45 V, V_2$	i = 16 V								
200 W	00 W	76.8%	73 W	73.3%	-3.4%	29 W	77.3%	0.5%	40 W	83.3%	6.5%
-200 W	$59 \mathrm{W}$	77.3%	53 W	79.1%	1.7%	$48 \mathrm{W}$	80.7%	3.4%	38 W	83.9%	6.6%
500 W	$57 \mathrm{W}$	89.8%	59 W	89.4%	-0.3%	$52 \mathrm{W}$	90.6%	0.8%	$44~\mathrm{W}$	92.0%	2.2%
-500 W	$57 \mathrm{W}$	89.8%	57 W	89.8%	0.0%	$51\mathrm{W}$	90.8%	1.0%	46 W	91.5%	1.7%
$1 \mathrm{kW}$	$54\mathrm{W}$	94.9%	66 W	93.8%	-1.0%	59 W	94.4%	-0.3%	53 W	95.0%	0.1%
-1 kW	57 W	94.6%	66 W	93.8%	-0.7%	59 W	94.4%	-0.1%	56 W	94.7%	0.1%
$1.5 \mathrm{kW}$	66 W	95.8%	75 W	95.2%	-0.4%	$63 \mathrm{W}$	96.0%	0.2%	62 W	96.0%	0.3%
-1.5 kW	$64\mathrm{W}$	95.9%	79 W	95.0%	-0.8%	$63 \mathrm{W}$	96.0%	0.0%	62 W	96.0%	0.1%
$2 \mathrm{kW}$	107 W	94.9%	100 W	95.2%	0.3%	$91 \mathrm{W}$	95.6%	0.7%	91 W	95.7%	0.7%
$-2 \mathrm{kW}$	99 W	95.3%	98 W	95.3%	0.0%	90 W	95.7%	0.4%	89 W	95.7%	0.5%
	$V_1 = 4$	$50 \text{ V}, V_2$	s = 11 V								
$200 \mathrm{W}$	247 W	44.7%	67 W	75.0%	30.3%	55 W	78.4%	33.6%	42 W	82.7%	38.0%
-200 W	237 W	45.7%	60 W	77.0%	31.3%	$52 \mathrm{W}$	79.3%	33.6%	39 W	83.7%	37.9%
500 W	247 W	67.0%	77 W	86.6%	19.6%	64 W	88.6%	21.6%	50 W	90.9%	23.9%
-500 W	241 W	67.5%	73 W	87.3%	19.8%	62 W	88.9%	21.4%	51 W	90.8%	23.3%
$1 \mathrm{kW}$	$248 \mathrm{W}$	80.1%	104 W	90.6%	10.5%	87 W	92.0%	11.9%	76 W	93.0%	12.8%
-1 kW	249 W	80.0%	103 W	90.7%	10.6%	88 W	91.9%	11.9%	78 W	92.7%	12.7%
$1.5 \mathrm{kW}$	253 W	85.6%	138 W	91.6%	6.0%	$116 \mathrm{W}$	92.8%	7.3%	$107 \mathrm{W}$	93.3%	7.8%
-1.5 kW	263 W	85.1%	142 W	91.4%	6.3%	$120 \mathrm{W}$	92.6%	7.5%	$114 \mathrm{W}$	93.0%	7.9%
$2 \mathrm{kW}$	264 W	88.3%	177 W	91.9%	3.5%	$150 \mathrm{W}$	93.0%	4.7%	$142\mathrm{W}$	93.4%	5.0%
$-2 \mathrm{kW}$	285 W	87.5%	187 W	91.4%	3.9%	$158\mathrm{W}$	92.7%	5.1%	$154\mathrm{W}$	92.8%	5.3%

Table 5.2 cont.: Calculated losses and efficiencies for different operating points and for phase shift modulation, extended triangular and trapezoidal current mode modulation (Section 5.1), optimized modulation (Section 5.2.2), and optimal modulation (Section 5.2.1)); n = 19, $L_{\rm M} =$ 2.7 mH, and $f_{\rm S} = 100$ kHz; $L = 26.7 \,\mu$ H applies for phase shift modulation, optimized modulation, and optimal modulation; $L = 18.7 \,\mu$ H is required for the extended triangular and trapezoidal current mode modulation.

			Exten	ded tria	ngular						
	Phase	e shift	and	trapezo	idal	51	Suboptime	al		Optimal	
P_{out}	modu	lation	cn	rrent m_{u}	$_{ode}$	r	nodulatic	u	m	vodulatic	n
	L = 2	$6.7 \mu { m H}$	m	odulatic	u	Г	$i = 26.7 \mu$	H	Γ	$= 26.7 \mu$	Н
			L	$= 18.7 \mu$	Η						
	$P_{\rm t, PS}$	η_{PS}	$P_{\rm t,TT}$	η_{TT}	$\Delta \eta_{\mathrm{TT}}$	$P_{\rm t, subopt}$	$\eta_{ m subopt}$	$\Delta\eta_{ m subopt}$	$P_{\rm t,opt}$	η_{opt}	$\Delta\eta_{\rm opt}$
	$V_{1} = 4$	$50 \text{ V}, V_2$	i = 13.5	V							
$200 \mathrm{W}$	204 W	49.4%	71 W	73.9%	24.5%	59 W	77.3%	27.9%	39 W	83.7%	34.3%
-200 W	191 W	51.2%	62 W	76.2%	25.0%	55 W	78.5%	27.3%	39 W	83.8%	32.6%
500 W	202 W	71.2%	78 W	86.5%	15.3%	65 W	88.5%	17.3%	$48 \mathrm{W}$	91.2%	20.0%
-500 W	192 W	72.3%	72 W	87.5%	15.2%	62 W	89.0%	16.7%	$48 \mathrm{W}$	91.3%	19.0%
$1 \mathrm{kW}$	200 W	83.3%	97 W	91.2%	7.9%	81 W	92.5%	9.2%	67 W	93.7%	10.4%
-1 kW	196 W	83.6%	94 W	91.4%	7.8%	80 W	92.6%	8.9%	09 W	93.6%	10.0%
$1.5 \mathrm{kW}$	201 W	88.2%	121 W	92.5%	4.4%	$102 \mathrm{W}$	93.6%	5.5%	91 W	94.3%	6.1%
-1.5 kW	204 W	88.0%	122 W	92.5%	4.4%	$104\mathrm{W}$	93.5%	5.5%	94 W	94.1%	6.0%
$2 \mathrm{kW}$	206 W	90.6%	149 W	93.0%	2.4%	$127 \mathrm{W}$	94.0%	3.4%	$118 \mathrm{W}$	94.4%	3.8%
$-2 \mathrm{kW}$	217 W	90.2%	156 W	92.8%	2.6%	$132\mathrm{W}$	93.8%	3.6%	$128\mathrm{W}$	94.0%	3.8%
	$V_{1} = 4$	$50 V, V_2$	s = 16 V								
$200 \mathrm{W}$	167 W	54.5%	76 W	72.5%	18.0%	63 W	76.1%	21.6%	40 W	83.5%	28.9%
-200 W	$153 \mathrm{W}$	56.7%	66 W	75.1%	18.5%	58 W	77.5%	20.8%	39 W	83.7%	27.1%
$500 \mathrm{W}$	$163 \mathrm{W}$	75.5%	80 W	86.2%	10.8%	67 W	88.2%	12.8%	47 W	91.4%	16.0%
-500 W	152 W	76.7%	73 W	87.3%	10.6%	63 W	88.8%	12.1%	46 W	91.5%	14.8%
$1 \mathrm{kW}$	$158 \mathrm{W}$	86.4%	93 W	91.5%	5.1%	78 W	92.8%	6.4%	62 W	94.2%	7.8%
-1 kW	$153 \mathrm{W}$	86.8%	$89 \mathrm{W}$	91.9%	5.1%	76 W	92.9%	6.1%	63 W	94.1%	7.3%
$1.5 \mathrm{kW}$	156 W	90.6%	110 W	93.2%	2.6%	93 W	94.2%	3.6%	80 W	94.9%	4.4%
-1.5 kW	156 W	90.6%	109 W	93.2%	2.6%	$94 \mathrm{W}$	94.1%	3.6%	82 W	94.8%	4.3%
$2 \mathrm{kW}$	158 W	92.7%	130 W	93.9%	1.2%	111 W	94.7%	2.1%	$101 \mathrm{W}$	95.2%	2.5%
$-2 \mathrm{kW}$	$164~\mathrm{W}$	92.4%	$134 \mathrm{~W}$	93.7%	1.3%	$115 \mathrm{W}$	94.6%	2.2%	$108\mathrm{W}$	94.9%	2.5%

Table 5.2 cont.: Calculated losses and efficiencies for different operating points and for phase shift modulation, extended triangular and trapezoidal current mode modulation (Section 5.1), optimized modulation (Section 5.2.2), and optimal modulation (Section 5.2.1)); n = 19, $L_{\rm M} =$ 2.7 mH, and $f_{\rm S} = 100$ kHz; $L = 26.7 \,\mu$ H applies for phase shift modulation, optimized modulation, and optimal modulation; $L = 18.7 \,\mu$ H is required for the extended triangular and trapezoidal current mode modulation.

Chapter 6

Dynamic DAB Model and Control

In order for the presented converter to operate at a designated power level and a desired direction of power transfer, a two loop control structure is employed: the inner control loop adjusts the required current (e.g. on the high voltage side) and the superimposed voltage control loop regulates the respective output voltage.¹ Thus, in order to enable the controller design, the small-signal model of the DAB converter needs to be determined.

First small-signal models of switching power converters have been calculated with averaged switched circuits [114]. There, a converter circuit diagram results, which partly consists of the original circuit (except for switches and diodes) and additional current sources or voltage sources or both that model the low frequency system dynamics. It thus presents the small-signal system dynamics in a comprehensible way to the engineer. However, its construction requires circuit arrangements and is thus not accomplished in a straightforward way; with the introduction of state space averaging [115], the automated derivation of small-signal transfer functions has been achieved. Both methods employ the local average values of all independent system properties (e.g. inductor currents, capacitor voltages) and all input and output variables (e.g. actuating variable, output current), calculated over one switching period $T_{\rm S}$ in order to determine the desired transfer functions. A precise small-signal and continuous-time transfer function approximation is thus achieved for frequencies well below $1/T_{\rm S}$ [114,116] and well below the filter cut-off frequencies (negligible current and voltage ripples are assumed). These two methods presumably compute wrong transfer functions for the given DAB power converter

¹For power being transferred from the high voltage DC bus to the 14V DC bus, V_2 becomes the output voltage (cf. Figure 6.1). Hence, the voltage control loop may become very sensitive due to the high electric storage capability of the 12V battery and the voltage control loop may be replaced by a voltage inspection and battery charging algorithm and / or battery charging current control.



Figure 6.1: Dual active bridge (DAB) converter with EMI filters connected to the high voltage (HV) and to the low voltage (LV) ports.

(Figure 6.1), since the assumption of negligible current ripple is not satisfied for the transformer and inductor current i_L . A very accurate, but discretetime, small-signal model is obtained with the discrete modeling of switching regulators [117] which solely assumes a "small" input signal excitation around a steady-state value in order to reduce the nonlinear (exact) large signal model to a linear small-signal model.

Today, many applications still employ continuous-time regulators, e.g. low power and low cost converters that are used for standard applications and where integrated regulators are available. There, the discrete-time nature of the transfer function obtained with [117] is considered disadvantageous and thus, numerous extensions to the discrete-time modeling approach have been developed in order to achieve a highly accurate continuous-time small-signal converter model, for instance the so-called sampled data modeling [118]. For many high power converters, though, digital signal processors (DSPs) are increasingly employed, mainly because today a high computational performance is available at a comparably low cost. Advantages of a digital implementation are a considerably higher flexibility compared to analog electronics, a high EMI immunity, and the enhanced possibility of process and fault monitoring using an external interface or a network connection. Therefore, a digital control platform is employed for the control of the presented DAB and thus, the discrete-time transfer function obtained with discrete modeling could readily be used for the controller design. However, [117] only focuses on simple DC–DC converters (e.g. boost converter) operated with PWM modulation. An extension to resonant power converters including modulator constraints is presented in [119] and a straight forward summary on the construction of a discrete-time small-signal model for arbitrary resonant converters is discussed in [120]. Merely the time lag that arises due to software and A/D converter delay times is not considered in [120] and could be included using the procedure discussed in [121].

In this Chapter, an exact discrete-time DAB model including input and output filter dynamics is developed and verified with results obtained from an experimental system. Moreover, a flexible control structure, which allows for optimized modulation and control algorithms, is proposed and detailed. The effect of time delays, mainly due to software processing time, is identified to be crucial with respect to a sufficiently large phase margin. The obtained results facilitate a significantly simplified construction of the DAB controlto-output transfer function, which can be universally employed to design the digital controllers.

In Section 6.1, the small-signal model for the DAB is derived; the employed control loop is discussed in Section 6.2; and in Section 6.3, the design of the digital current and voltage controllers is detailed. The calculated results are verified with measured results in Sections 6.2 and 6.3.

6.1 Small-Signal Model

The proposed control structure, depicted in Figure 6.2, mainly consists of digital signal processing blocks: the voltage controller $(G_{C,V})$, the current controller $(G_{C,I})$, and the digital filters $(\mathbf{H_{filter}}, H_{avg})$ are fully implemented in a DSP; software and A/D converter time delays $(\mathbf{G_{Td,DSP}}, \mathbf{G_{Td,FPGA}}, \mathbf{G_{Td,meas}})$ are due to the digital implementation. Besides, the modulator function $(\mathbf{G_{mod,PS}} \text{ or } \mathbf{G_{mod,alt}})$ is realized in the DSP and determines the required timing values for the DAB [e.g. $T_{\varphi} = \varphi/(2\pi f_S)$ for phase shift modulation based on (3.13)] with respect to the desired controller set value $I_{1,mod}$ and the selected modulation scheme. Variations of these timing values dynamically alter the transferred power of the DAB power stage. The resulting changes of the filter currents $i_{f1}(t)$, $i_{f2}(t)$ and the filter voltages $v_{f1}(t)$, $v_{f2}(t)$ (cf. Figure 6.1) are obtained from the DAB small-signal model $\mathbf{G_{PE}}$ (the voltages V_1 and V_2 remain approximately constant during one switching period T_S , since very large filter capacitors C_{DC1} and C_{DC2} are considered).

The full control diagram (Figure 6.2) indicates two transfer functions, that need to be determined in order to further investigate the control loop: the control-to-output transfer function $\mathbf{G}_{\mathbf{PE}}$ of the DAB and the transfer function of the modulator.

6.1.1 DAB Power Stage, Phase Shift Modulation

Phase shift modulation is the most simple as well as the most common modulation method for the DAB. Therefore, the small-signal transfer function is first derived for this basic modulation scheme.

The small-signal calculation method outlined in [120] regards a single halfcycle (e.g. $0 < t \leq T_S/2$ in Figure 3.4) and separately considers the time intervals between two switching events (time intervals I and II in Figure 3.4). Within these time intervals, the time domain expressions for all system state variables (i.e. all time varying inductor currents and capacitor voltages) are required in order to determine their sensitivity on input signal variations. With the employed method, the state variables at the end of a half-cycle are compared to the state variables at the beginning of the corresponding half-cycle. This allows for the formulation of the control-to-output transfer function in discrete-time domain.

Prior to the derivation of the transfer function, system inputs and outputs must be defined. Clearly, the timing parameter T_{φ} is an input to the dynamic system. Additionally, the voltages V_1 and V_2 may be considered as inputs, too. The proposed system outputs are the filter current $i_{f1}(t) = i_{f1a}(t) + i_{f1b}(t)$ and the filter voltages $v_{f1}(t)$, $n v_{f2}(t)$. Further, it is reasonable to collect the



Figure 6.2: Investigated control structure including voltage and current control loops for power transfer from the HV to the LV port as well as the modulator function which generates the required timing signals $[T_{\rm M} = T_{\varphi}$ for the phase shift modulation or $\vec{T}_{\rm M} = (T_1 \ T_2 \ T_3)^{\rm T}$ for the alternative modulation]; gate signal generation (FPGA) and power electronics are sampled with sampling time $T_{\rm DAB}$, while the remaining transfer functions are sampled with $T = 10T_{\rm DAB}$ (due to computing time demands).

system state variables in a state vector \vec{x} :

$$\vec{x} = (i_L \ i_{f1a} \ i_{f1b} \ i_{f2a}/n \ i_{f2b}/n \ v_{f1} \ n v_{f2})^{\mathrm{T}}.$$
 (6.1)

In a next step, the time domain expressions for the system state variables need to be derived. These may change at each switching instant indicated in Figure 3.4. For the sake of clarity the time domain expressions during each time interval i are combined to one single function

$$\vec{\mathbf{f}}_{PS,i}(\vec{x}_{i-1},\Delta t_i) = \begin{bmatrix} i_{L,i}(\vec{x}_{i-1},\Delta t_i) & \\ i_{f1a,i}(\vec{x}_{i-1},\Delta t_i) & i_{f1b,i}(\vec{x}_{i-1},\Delta t_i) \\ i_{f2a,i}(\vec{x}_{i-1},\Delta t_i)/n & i_{f2b,i}(\vec{x}_{i-1},\Delta t_i)/n \\ v_{f1,i}(\vec{x}_{i-1},\Delta t_i) & n v_{f2,i}(\vec{x}_{i-1},\Delta t_i) \end{bmatrix}^{\mathrm{T}}.$$
(6.2)

In (6.2), Δt_i denotes the time within the considered time interval (i.e. Δt_i is zero at the beginning of the *i*-th time interval) and \vec{x}_{i-1} contains the initial values, so $\vec{f}_{\text{PS},i}(\vec{x}_{i-1},0) = \vec{x}_{i-1}$; the index "PS" denotes the phase shift modulation, the index *i* specifies the considered time interval. The analysis considers two different time intervals for phase shift modulation (Figure 3.4),²

Interval I (i = 1): $0 < t \le T_{\varphi} \to \Delta t_1 = t$, (6.3)

Interval II
$$(i = 2)$$
: $T_{\varphi} < t \le T_{\rm S}/2 \rightarrow \Delta t_2 = t - T_{\varphi}$. (6.4)

In steady-state operation, the system state vector $\vec{x}_{st}(t)$ (index "st" denotes steady-state) repeats cyclically every switching period. The steady-state values of $\vec{x}_{st}(t)$ are obtained from

$$\vec{x}_{\rm st}\left(t+\frac{T_{\rm S}}{2}\right) = \begin{bmatrix} -1 & \mathbf{0} \\ & {}^{1\times 6} \\ \mathbf{0} & \mathbf{I} \\ & {}^{6\times 1} & {}^{6\times 6} \end{bmatrix} \cdot \vec{x}_{\rm st}(t) = \mathbf{R} \cdot \vec{x}_{\rm st}(t), \tag{6.5}$$

since the DAB inductor current $i_L(t)$ changes its sign every half-cycle in steady-state [120], whereas the signs of the filter inductor currents and capacitor voltages remain the same after one half-cycle. In order to solve for $\vec{x}_{\rm st}(t)$ in (6.5), the time domain expressions for $\vec{x}(t + T_{\rm S}/2)$ are required as a function of $\vec{x}(t)$. The derivation of these expressions is described based on the switching sequence shown in Figure 3.4 and starts with the unknown values in $\vec{x}(t = 0)$. With (6.2), (6.3), and i = 1, the general expression for $\vec{x}(t)$ at $t = T_{\varphi}$ is equal to

$$\vec{x}(T_{\varphi}) = \vec{f}_{PS,1}(\vec{x}_0, T_{\varphi}) \text{ with } \vec{x}_0 = \vec{x}(0).$$
 (6.6)

²The derivation of the time domain expressions in $\vec{f}_{PS,i}(\vec{x}_{i-1}, \Delta t_i)$ is associated with a considerable calculation effort and is preferably carried out with a mathematical software tool that allows for symbolic evaluation (e.g. Mathematica, Maple).

At $t = T_{\varphi}$ the switches T₆, T₇ are turned off, T₅, T₈ are turned on, and time interval II starts. Since capacitor voltages as well as inductor currents cannot change instantaneously, the initial conditions of time interval II are equal to the final state variable values of time interval I, $\vec{x}_1 = \vec{x}(T_{\varphi})$. With the time domain expressions for all state variables during time interval II the state vector at the end of the half-cycle (cf. Figure 3.4) becomes

$$\vec{x}(T_{\rm S}/2) = \vec{f}_{\rm PS,2}(\vec{x}_1, T_{\rm S}/2 - T_{\varphi}) = \vec{f}_{\rm PS,2}[\vec{f}_{\rm PS,1}(\vec{x}_0, T_{\varphi}), T_{\rm S}/2 - T_{\varphi}].$$
(6.7)

The steady-state values are then obtained from the equation system (6.5) with (6.7), $\vec{x}_{st}(t) = \vec{x}(t)$, and t = 0.

Besides for the calculation of the steady-state, (6.7) forms the basis for the derivation of the discrete-time small-signal model with sampling time

$$T_{\rm DAB} = T_{\rm S}/2 = 5\,\mu {\rm s}\,.$$
 (6.8)

The proposed model considers small-signal deviations, $\vec{x}(t)$, of the system state variables $\vec{x}(t)$ from the steady-state $\vec{x}_{st}(t)$,

$$\vec{x}(t) = \vec{x}(t) - \vec{x}_{st}(t)$$
 (6.9)

(the symbol "^" denotes small-signal variables), by reason of three different kinds of excitations:

- variations of the state variables due to prior excitations of input variables: $\hat{\vec{x}}_0 = \hat{\vec{x}}(0) \vec{x}_{st}(0)$,
- HV and LV voltage changes: $\hat{\vec{v}}_{g,0} = [\hat{V}_1(0) \ n \hat{V}_2(0)]^T$,
- excitation of the control input: $\hat{c}_{\text{PS},0} = \hat{T}_{\varphi}(0)$.

It is important to note, that the proposed discrete-time model considers an excitation of any of these variables exactly at the beginning of the half-cycle and calculates the respective values of the system state variables at the end of the half-cycle (Figure 6.3). Consequently, all changes of the input variables $\hat{v}_{g}(t)$ and $\hat{c}_{PS}(t)$ that occur for $0 < t \leq T_{DAB}$ are not at all considered in $\hat{x}(T_{DAB})$. This limitation only regards transfer functions with continuous-time input signals such as $V_1(t)$, $V_2(t)$. It does not affect the control-to-output transfer function in a digitally controlled system, provided that the digital controller is synchronized to the power electronics, i.e. $T_{\varphi}(t)$ changes exactly at $t = k \cdot T_{DAB}$, $k \in \mathbb{N}_0$.

The small-signal deviations $\hat{\vec{x}}(T_{\text{DAB}})$ at the end of the half-cycle are then obtained from a linear approximation [120],

$$\hat{\vec{x}}(T_{\text{DAB}}) \approx \mathbf{A} \cdot \hat{\vec{x}}_0 + \mathbf{B}_{\mathbf{PS}} \cdot \hat{c}_{\text{PS},0} + \mathbf{C} \cdot \hat{\vec{v}}_{\text{g},0},$$
 (6.10)



Figure 6.3: Reaction of the DAB to a phase shift excitation $\hat{T}_{\varphi}(0)$ (thick lines): at the end of the half-cycle, $t = T_{\rm S}/2$, the inductor current deviation $\hat{i}_L(T_{\rm S}/2)$ results. The thin lines depict steady-state operation according to Figure 3.4 (index "st" denotes steady-state). The digital control electronics determines $T_{\varphi}(0)$ at t = 0, which is applied during $0 < t < T_{\rm S}/2$.

whereas the three terms $\mathbf{A} \cdot \hat{\vec{x}}_0$, $\mathbf{B}_{\mathbf{PS}} \cdot \hat{c}_{\mathrm{PS},0}$, and $\mathbf{C} \cdot \hat{\vec{v}}_{\mathrm{g},0}$ express the small-signal deviations of the state variables at $t = T_{\mathrm{DAB}}$ as a result of excitations in $\hat{\vec{x}}_0$, $\hat{c}_{\mathrm{PS},0}$, and $\hat{\vec{v}}_{\mathrm{g},0}$, respectively. The expressions for \mathbf{A} , $\mathbf{B}_{\mathbf{PS}}$, and \mathbf{C} are given in Appendix F.1.1.

Expression (6.10) allows for the derivation of the required small-signal transfer functions: according to [120] and Appendix F.1.2, the z-domain control-to-output transfer functions are calculated with

$$\mathbf{G}_{3\times1}\mathbf{PE,PS} = \begin{bmatrix} G_{\mathrm{PE,PS},I_{\mathrm{f1}}} \\ G_{\mathrm{PE,PS},V_{\mathrm{f1}}} \\ G_{\mathrm{PE,PS},V_{\mathrm{f2}}} \end{bmatrix} = \mathbf{E}^{\mathrm{T}} \cdot (z_{\mathrm{DAB}}\mathbf{I} - \mathbf{QRAQ})^{-1} \cdot \mathbf{QRB_{PS}}, \quad (6.11)$$

$$\mathbf{Q}_{7\times7} = \begin{bmatrix} \operatorname{sgn}(i_{L,0}) & \mathbf{0} \\ \mathbf{0} & 1\\ 6\times1 & 6\times6 \end{bmatrix}, \ \mathbf{R}_{7\times7} = \begin{bmatrix} -1 & \mathbf{0} \\ 1\times6 \\ \mathbf{0} & \mathbf{I} \\ 6\times1 & 6\times6 \end{bmatrix},$$
(6.12)

and $z_{\text{DAB}} = e^{sT_{\text{DAB}}}$. In order to collect the three control-to-output transfer functions with the output variables $I_{f1}(z_{\text{DAB}})$, $V_{f1}(z_{\text{DAB}})$, and $nV_{f2}(z_{\text{DAB}})$ in

one matrix **G**_{PE,PS},

$$\mathbf{G}_{3\times1}\mathbf{PE,PS} = \left[\frac{I_{f1a}(z_{DAB}) + I_{f1b}(z_{DAB})}{T_{\varphi}(z_{DAB})} \frac{V_{f1}(z_{DAB})}{T_{\varphi}(z_{DAB})} \frac{nV_{f2}(z_{DAB})}{T_{\varphi}(z_{DAB})}\right]^{\mathrm{T}}, \quad (6.13)$$

the matrix \mathbf{E}^{T} in (6.11) becomes

$$\mathbf{E}^{\mathrm{T}} = \begin{bmatrix} 0 & 1 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 0 & 1 \end{bmatrix}.$$
 (6.14)

The small-signal transfer functions in (6.11) have been verified for various operating points with results from a circuit simulator (e.g. the control-to-output transfer function $G_{\text{PE},\text{PS},I_{f1}}$ in Figure 6.4). Simulated and calculated results show very good agreement for all investigated operating points.

6.1.2 DAB Power Stage, Alternative Modulation Schemes

If the DAB converter is operated with phase shift modulation within wide voltage ranges, its efficiency drops considerably for certain operating points (i.e. for $V_1/V_2 \ll n$ or $V_1/V_2 \gg n$). The average DAB converter efficiency substantially increases with the use of alternative modulation schemes (the index "alt" denotes all variables regarding alternative modulation schemes, cf. Chapter 3 and Chapter 5).

The derivations of the small-signal transfer functions for phase shift modulation and the alternative modulation schemes are very similar. However, phase shift modulation requires only one control input T_{φ} whereas three control inputs T_1 , T_2 , and T_3 are needed for the alternative modulation schemes.

Figure 5.4 depicts typical voltage and current waveforms for the DAB with the extended triangular current mode modulation. Again, one single half-cycle is segmented into the time intervals where no switching occurs:

This, together with (6.1) and the time domain expressions for all state variables within time interval i, $\vec{f}_{alt,i}(\vec{x}_{i-1}, \Delta t_i)$ [cf. (6.2)], facilitates the derivation of the system state values at the end of the half-cycle (Figure 5.4),

$$\vec{x}(T_{\rm S}/2) = \vec{f}_{\rm alt,4} \big(\vec{f}_{\rm alt,3} \big\{ \vec{f}_{\rm alt,2} \big[\vec{f}_{\rm alt,1} \big(\vec{x}_0, T_1 \big), T_2 \big], T_3 \big\}, T_4 \big), \tag{6.15}$$



Figure 6.4: Simulated step response (solid lines) and calculated step response (dashed line) of $i_{f1}(t)$ for phase shift modulation, a time step, \hat{T}_{φ} , of 100 ns, nominal operating point, and power being transferred from the HV to the LV port. The circuit simulator generates a continuous-time waveform $i_{f1}(t)$ which is sampled after every half-cycle, $t = k \cdot T_{\text{DAB}}$, in order to facilitate the comparison to the step response obtained from the discrete-time small-signal transfer function $G_{\text{PE,PS},I_{f1}}$. The solid line forms a linear connection of the sampled values of the simulated waveform $i_{f1}(t)$ to facilitate the comparison with the calculated values $i_{f1}(kT_{\text{DAB}})$ (dashed line). This solid line is thus different from the local average of $i_{f1}(t)$.

whereas $T_4 = T_S/2 - (T_1 + T_2 + T_3)$.

The solution to the equation system formed with (6.5) and (6.15) determines the steady-state values for $\vec{x}_{st}(t) = \vec{x}(t)$ at t = 0.

Equation (6.15) as well denotes the starting point for the small-signal transfer function derivation. In contrast to the phase shift modulation, the control input variable is now vector valued,

$$\hat{\vec{c}}_{\text{alt},0} = [\hat{T}_1(0) \quad \hat{T}_2(0) \quad \hat{T}_3(0)]^{\mathrm{T}};$$
(6.16)

the small-signal vectors \vec{x}_0 and $\vec{v}_{g,0}$ remain unchanged. Consequently, the system states at the end of the half-cycle can be approximately calculated

with

$$\hat{\vec{x}}(T_{\text{DAB}}) \approx \mathbf{A} \cdot \hat{\vec{x}}_0 + \mathbf{B}_{\text{alt}} \cdot \hat{\vec{c}}_{\text{alt},0} + \mathbf{C} \cdot \hat{\vec{v}}_{\text{g},0}.$$
 (6.17)

 $(\mathbf{A}, \mathbf{B}_{alt}, and \mathbf{C}$ are derived in Appendix F.1.1). The control-to-output transfer functions are collected in the matrix

$$\begin{aligned} \mathbf{G}_{3\times3}\mathbf{PE,alt} &= \begin{bmatrix} \vec{G}_{\mathrm{PE,alt,1}} & \vec{G}_{\mathrm{PE,alt,2}} & \vec{G}_{\mathrm{PE,alt,3}} \end{bmatrix}, \end{aligned} \tag{6.18} \\ \vec{G}_{\mathrm{PE,alt,i}} &= \begin{bmatrix} I_{\mathrm{fla}}(z_{\mathrm{DAB}}) + I_{\mathrm{flb}}(z_{\mathrm{DAB}}) \\ \hline T_i(z_{\mathrm{DAB}}) & \hline T_i(z_{\mathrm{DAB}}) \\ \hline T_i(z_{\mathrm{DAB}}) & \hline T_i(z_{\mathrm{DAB}}) \end{bmatrix}^{\mathrm{T}}, \end{aligned}$$

(i = 1, 2, 3) which is derived based on (6.17) with (6.12) and (6.14),

$$\mathbf{G}_{3\times3}\mathbf{PE,alt} = \mathbf{E}^{\mathrm{T}} \cdot (z_{\mathrm{DAB}}\mathbf{I} - \mathbf{QRAQ})^{-1} \cdot \mathbf{QRB_{alt}}.$$
 (6.19)

The small-signal transfer functions in (6.18) have been verified with a circuit simulator. Very good agreement between simulation and calculation has been obtained for all investigated operating points [e.g. for nominal operation in Figure 6.5 and Figure 6.6 which show the step responses of the controlto-output transfer functions $G_{\text{PE,alt,1,I_{f1}}}$ for the extended triangular current mode modulation and the suboptimal modulation, respectively, with input $T_1(z_{\text{DAB}})$ and output $I_{\text{f1}}(z_{\text{DAB}})$].

6.1.3 Modulator

The modulator calculates the control variables depending on the applied modulation scheme, the set current $i_{1,\text{mod}}$, and the measured values \overline{v}_{f1} and $n\overline{v}_{f2}$, obtained from v_{f1} and nv_{f2} . The modulator function for phase shift modulation,

$$c_{\rm PS} = T_{\varphi} = f_{\rm mod, PS}[i_{1, \rm mod}(kT), \overline{v}_{\rm f1}(kT), n\overline{v}_{\rm f2}(kT)], \qquad (6.20)$$

can be derived from (3.13) for a lossless DAB [the sampling time T is defined with (6.24), $k \in \mathbb{Z}$, and variables with lowercase letters denote time domain functions, e.g. $i_{1,\text{mod}}(kT)$ is equal to the sum of the steady-state DC component $I_{1,\text{mod},\text{st}}$ and the small-signal disturbance $\mathcal{Z}^{-1}\{I_{1,\text{mod}}(z)\}$: $i_{1,\text{mod}}(kT) = I_{1,\text{mod},\text{st}} + \mathcal{Z}^{-1}\{I_{1,\text{mod}}(z)\}$]. The expressions for the alternative modulation schemes,

$$\vec{c}_{\text{alt}} = (T_1 \quad T_2 \quad T_3)^{\text{T}} = \vec{f}_{\text{mod,alt}}[i_{1,\text{mod}}(kT), \overline{v}_{\text{f1}}(kT), n\overline{v}_{\text{f2}}(kT)]$$
(6.21)

are rather complicated, hence their evaluation in the DSP is avoided. Instead, precalculated values stored in DSP memory tables are used together with the fast linear interpolation algorithm presented in Appendix E.2. Since the



Figure 6.5: Simulated step response (solid lines) and calculated step response (dashed line) of $i_{f1}(t)$ for the extended triangular current mode modulation if a time step $\hat{T}_1 = 100 \text{ ns} \cdot \sigma(t)$ is applied and $\hat{T}_2 = \hat{T}_3 = 0$, at nominal operation, and for power being transferred from the HV to the LV port. The continuous-time waveform $i_{f1}(t)$ is again sampled after every half-cycle in order to facilitate the comparison to the step response obtained from the discrete-time small-signal model. The solid line forms a linear connection of the sampled values of the simulated waveform $i_{f1}(t)$ to facilitate the comparison with the calculated values $i_{f1}(kT_{\text{DAB}})$ (dashed line). This solid line is thus different from the local average of $i_{f1}(t)$.

modulator function is static, the small-signal transfer function is simply determined with the respective derivatives at the operating point. Thus, the transfer functions for phase shift modulation are

$$\mathbf{G}_{1\times3}^{\mathbf{mod},\mathbf{PS}} = \begin{bmatrix} \frac{\partial f_{\mathrm{mod},\mathrm{PS}}}{\partial i_{1,\mathrm{mod}}} & \frac{\partial f_{\mathrm{mod},\mathrm{PS}}}{\partial \overline{v}_{\mathrm{f1}}} & \frac{\partial f_{\mathrm{mod},\mathrm{PS}}}{\partial (n\overline{v}_{\mathrm{f2}})} \end{bmatrix}, \quad (6.22)$$

and for the alternative modulation schemes

$$\mathbf{G}_{3\times3}^{\mathbf{mod},\mathbf{alt}} = \begin{bmatrix} \frac{\partial \vec{\mathbf{f}}_{\mathrm{mod},\mathrm{alt}}}{\partial i_{1,\mathrm{mod}}} & \frac{\partial \vec{\mathbf{f}}_{\mathrm{mod},\mathrm{alt}}}{\partial \overline{v}_{\mathrm{f}1}} & \frac{\partial \vec{\mathbf{f}}_{\mathrm{mod},\mathrm{alt}}}{\partial (n\overline{v}_{\mathrm{f}2})} \end{bmatrix}.$$
 (6.23)

 $\mathbf{G}_{\text{mod}, PS}$ and $\mathbf{G}_{\text{mod}, \text{alt}}$ are evaluated at the operating point with the steadystate values of $i_{1, \text{mod}}$, \overline{v}_{f1} , and \overline{v}_{f2} .



Figure 6.6: Simulated step response (solid lines) and calculated step response (dashed line) of $i_{f1}(t)$ for the suboptimal modulation if a time step $\hat{T}_1 = 100 \text{ ns} \cdot \sigma(t)$ is applied and $\hat{T}_2 = \hat{T}_3 = 0$, at nominal operation, and for power being transferred from the HV to the LV port. The continuous-time waveform $i_{f1}(t)$ is again sampled after every half-cycle in order to facilitate the comparison to the step response obtained from the discrete-time small-signal model. The solid line forms a linear connection of the sampled values of the simulated waveform $i_{f1}(t)$ to facilitate the comparison with the calculated values $i_{f1}(kT_{\text{DAB}})$ (dashed line). This solid line is thus different from the local average of $i_{f1}(t)$.

6.2 DAB Digital Control Loop

In the given laboratory setup, the average DAB output voltage is controlled (e.g. \overline{V}_{f2} in Figure 6.2 for power transfer from HV to LV). The proposed control loop consists of an inner loop with a PI-controller $G_{C,I}$ that controls $\overline{I}_{f1}(z)$ and an outer loop with another PI-controller $G_{C,V}$, which controls the output voltage. Depending on the power transfer direction, the average output voltage can be either $\overline{V}_{f2}(z)$ for power transfer from the HV port to the LV port or $\overline{V}_{f1}(z)$ for the opposite direction.

According to Figure 6.2, the current controller $G_{C,I}$ outputs $I_{1,mod}(z)$ and connects to the modulator \mathbf{G}_{mod} . The modulator then calculates the DAB timing parameters in order to achieve the required power transfer. Since the current controller solely operates on the difference between $\overline{I}_{f1}(z)$ and $I_{1,ref}(z)$, G_{DAB} is a single input, single output transfer function with set current input $I_{1,\text{mod}}(z)$ and HV side current output $\overline{I}_{f1}(z)$. However, there is internal feedback in the transfer function G_{DAB} , since $\overline{V}_{f1}(z)$ and $n\overline{V}_{f2}(z)$ are required for the modulator; this must be considered in order to determine G_{DAB} .

Except for the DAB small-signal transfer function matrix \mathbf{G}_{PE} (equal to $\mathbf{G}_{PE,PS}$ or $\mathbf{G}_{PE,alt}$), all transfer functions in Figure 6.2 are part of the digital system and either reside in the DSP or in the FPGA. Relatively simple z-domain transfer functions result for the modulator \mathbf{G}_{mod} (Section 6.1.3), the time delays $\mathbf{G}_{Td,DSP}$, $\mathbf{G}_{Td,FPGA}$, and $\mathbf{G}_{Td,meas}$, the moving average filters \mathbf{H}_{filter} and H_{avg} as well as for the controllers $G_{C,V}$ and $G_{C,I}$. This section focuses on these transfer functions in order to complete the derivation of G_{DAB} . The obtained transfer function is finally compared to a transfer function based on a simplified DAB model that allows for a significantly reduced calculation effort.

6.2.1 System Sampling Rate

The small-signal model of the power stage results in a z-domain transfer function with sampling time T_{DAB} [cf. (6.8)]. Due to the required DSP calculation time, the DAB timing parameters are only updated every $10 T_{\text{DAB}}$, therefore the DSP sampling time is

$$T = 10 T_{\text{DAB}} = 50 \,\mu\text{s}, \ z = e^{sT}.$$
 (6.24)

Hence, all transfer functions with faster update rate, such as $\mathbf{G}_{\mathbf{PE}}$ and voltage and current measurements (Figure 6.2), need to be resampled (e.g. with the method outlined in Appendix F.2).

6.2.2 Time Delays

The implemented software acquires three measurements during one calculation period T in order to achieve higher noise immunity. The resulting time delay in $\mathbf{G}_{\mathbf{Td},\mathbf{meas}}$ is $10 T_{\text{DAB}}$ for the first measurement, $8 T_{\text{DAB}}$ for the second and $6 T_{\text{DAB}}$ for the third measurement. The FPGA ($\mathbf{G}_{\mathbf{Td},\mathbf{FPGA}}$) causes an additional time delay of $2T_{\text{DAB}}$ in order to apply the new timing values to the power electronics. FPGA and measurement time delays sum up to a total time delay of $12 T_{\text{DAB}}$ for the first measurement, and $10 T_{\text{DAB}}$ and $8 T_{\text{DAB}}$ for the two subsequent measurements. The DSP calculates the average of the three measured values and

$$\mathbf{G}_{\mathbf{PE,measure}}(z_{\text{DAB}}) = \frac{z_{\text{DAB}}^{-12} + z_{\text{DAB}}^{-10} + z_{\text{DAB}}^{-8}}{3} \cdot \mathbf{G}_{\mathbf{PE}}(z_{\text{DAB}})$$
(6.25)

results for that part of G_{DAB} which is updated with the higher sampling rate T_{DAB} (Figure 6.2). The sampling rate of this transfer function finally needs

to be converted to the system sampling rate T (cf. Appendix F.2),

$$\mathbf{G}_{\mathbf{PE},\mathbf{measure}}(z_{\mathrm{DAB}}) \xrightarrow[\mathrm{resample}]{\rightarrow} \mathbf{G}_{\mathbf{r},\mathbf{PE},\mathbf{measure}}(z).$$
 (6.26)

The DSP causes another time delay of T in order to carry out all calculations, therefore

$$G_{\rm Td,DSP,PS}(z) = z^{-1},$$
 (6.27)

$$\mathbf{G}_{\mathbf{Td},\mathbf{DSP},\mathbf{alt}}(z) = \operatorname{diag}\left(z^{-1}, z^{-1}, z^{-1}\right),$$
 (6.28)

for phase shift or alternative modulation, respectively.

6.2.3 Moving Average Filters

The z-domain transfer function of the implemented N-th order moving average filter,

$$H_{\rm avg}(z) = \frac{1}{N} \sum_{i=0}^{N-1} z^{-i}, \qquad (6.29)$$

calculates the average over N previously measured values x(0), $x(T) \dots x[(N-1) \cdot T]$; the present software implementation uses N = 5. The moving average filter is applied to $I_{f1}(z)$, $V_{f1}(z)$, and $nV_{f2}(z)$. The filter regarding I_{f1} is included in G_{DAB} (Section 6.2.4), and the feedback transfer function $\mathbf{H_{filter}}$ contains the remaining two filter functions,

$$\mathbf{H_{filter}}(z) = \operatorname{diag}\left(H_{\operatorname{avg}}, H_{\operatorname{avg}}\right). \tag{6.30}$$

6.2.4 DAB Control Plant G_{DAB}

The transfer function G_{DAB} is calculated according to Appendix F.3,

$$G_{\text{DAB}} = \overline{I}_{\text{f1}} / I_{1,\text{mod}} = H_{\text{avg}} \left[G_{00} + \mathbf{G}_{0\mathbf{r}} \mathbf{H} \left(\mathbf{I} - \mathbf{G}_{\mathbf{sr}} \mathbf{H} \right)^{-1} \mathbf{G}_{\mathbf{s0}} \right], \qquad (6.31)$$

with G_{00} , $\mathbf{G_{0r}}$, $\mathbf{G_{sr}}$, and $\mathbf{G_{s0}}$ summarized in the matrix \mathbf{G} ,

$$\mathbf{G} = \begin{bmatrix} G_{00} & \mathbf{G_{0r}} \\ & 1 \times n \\ \mathbf{G_{s0}} & \mathbf{G_{sr}} \\ & m \times 1 & m \times n \end{bmatrix}$$
(6.32)

[cf. (F.13) in Appendix F.3]. According to Figure F.1 and Figure 6.2, ${\bf G}$ and ${\bf H}$ become

$$\mathbf{G} = \mathbf{G}_{\mathbf{r},\mathbf{PE},\mathbf{measure},\mathbf{PS}} \cdot G_{\mathrm{Td},\mathrm{DSP},\mathrm{PS}} \cdot \mathbf{G}_{\mathbf{mod},\mathbf{PS}},$$

$$\mathbf{H} = \mathbf{H}_{\mathrm{filter}}$$
(6.33)

for phase shift modulation and

$$\begin{aligned} \mathbf{G} &= \mathbf{G}_{\mathbf{r},\mathbf{PE},\mathbf{measure},\mathbf{alt}} \cdot \mathbf{G}_{\mathbf{Td},\mathbf{DSP},\mathbf{alt}} \cdot \mathbf{G}_{\mathbf{mod},\mathbf{alt}}, \\ \mathbf{H} &= \mathbf{H}_{\mathbf{filter}} \end{aligned}$$
 (6.34)

if the alternative modulation is employed.

6.2.5 Experimental Verification

The control-to-output transfer function $I_{f1}/I_{1,mod}$ (cf. Figure 6.2) is measured using a sinusoidal current sequence superimposed on the steady-state modulator current value $I_{1,mod,st}$ (Figure 6.7 and Figure 6.8). The proposed measurement sequence consists of three different time intervals (Figure 6.7):

- *Time interval I* is used to settle the desired converter operating point. Therefore, controlled converter operation is needed during time interval I.
- At the beginning of time interval II, the steady-state value $I_{1,\text{mod,st}}$ is stored in the DSP and thereafter, current and voltage controllers are turned off. Hence, the converter is operated in open loop during time interval II. The stored value $I_{1,\text{mod,st}}$ and a superimposed sinusoidal time series $\mathcal{Z}^{-1}{I_{1,\text{mod}}(z)}$, calculated in the DSP, with a given amplitude and the desired excitation frequency are used to generate a sinusoidal current excitation around the steady-state operating point. The DSP measures the current i_{f1} which results in the time series $\overline{I}_{f1,\text{st}} + \mathcal{Z}^{-1}{I_{f1}(z)}$. During time interval II, however, the DSP discards the measured current values in order to eliminate transient effects.
- During time interval III, the DSP continues to generate the DC shifted sinusoidal time series (Figure 6.7 and Figure 6.8) and stores the generated modulator current values $I_{1,\text{mod},\text{st}} + Z^{-1}{I_{1,\text{mod}}(z)}$ and the measured current values $\overline{I}_{\text{f1,st}} + Z^{-1}{I_{\text{f1}}(z)}$ in an on-chip table; the measurement sequence completes after time interval III has elapsed. In a post processing pass, the tabulated current values are refined with a FIR filter in order to suppress noise and to accurately obtain gain and phase of the control-to-output transfer function at the employed excitation frequency. The outlined procedure is repeated for each data point depicted in Figure 6.9, Figure 6.10, and Figure 6.11.

The results obtained for phase shift modulation (Figure 6.9), and for the extended triangular current mode modulation scheme (Figure 6.10) show a good matching between calculated and measured transfer functions. A noticeable difference between measured and calculated gain response is observed for the suboptimal modulation scheme (Figure 6.11). This error is attributed to the instantaneous inductor currents $i_L(t_1)$ and $i_L(t_2)$ being close to zero [cf. Figures 5.23 (a) and (b) and Figure 5.29]: there, the characteristics of the delay of the LV switches is highly non-linear (Figure 4.26). For all depicted results and for frequencies below 7 kHz, a maximum gain difference of less than ± 1.5 dB is achieved; the phase differences are smaller than $\pm 10^{\circ}$.



Figure 6.7: Current waveforms for the proposed measurement of the controlto-output transfer functions: during time interval I, the converter settles the desired operating point, time interval II is used to avoid transient effects, and during time interval III gain and phase are measured (cf. Figure 6.8). The converter is voltage and current controlled during time interval I and is operated in open loop during time intervals II and III with a sinusoidal waveform $Z^{-1}{I_{1,mod}(z)}$ being superimposed on $I_{1,mod,st}$. The depicted example employs an excitation amplitude of 500 mA at a frequency of 51 Hz and has been measured on a previous setup of the DAB with $L = 31 \,\mu\text{H}$ and n = 24; $V_1 = 340 \,\text{V}$, $V_2 = 12 \,\text{V}$, $P_2 = 2 \,\text{kW}$, phase shift modulation. The waveform $i_{f1}(t)$ is measured with an analog current probe in order to allow for a comparison between digital and analog signals. Time interval III is not fully depicted; its duration is equal to 50 ms.



Figure 6.8: Measurement method employed to determine the DAB transfer function: during time interval III (cf. Figure 6.7), amplitude and phase of the generated sinusoidal time series $\mathcal{Z}^{-1}{I_{1,\text{mod}}(z)}$ are compared to amplitude and phase of the measured time series $\mathcal{Z}^{-1}{I_{\text{fl}}(z)}$ in order to obtain gain and phase shift for one frequency of the control-to-output transfer function (for this, only AC components are regarded – the DC components $I_{1,\text{mod},\text{st}}$ and $\overline{I}_{\text{fl},\text{st}}$ are not considered). The depicted example employs an excitation amplitude of 500 mA at a frequency of 1 kHz and nominal converter operation. The waveform $i_{\text{fl}}(t)$ is measured with an analog current probe in order to allow for a comparison between digital and analog signals.



Figure 6.9: Calculated and measured DAB frequency response for phase shift operation [(a) gain and (b) phase]; the Nyquist frequency 1/(2T) = 10 kHz limits the maximum possible frequency; the solid line denotes the calculated frequency response and the boxes mark the measured values. The DAB is operated with $V_1 = 340$ V, $V_2 = 12$ V, and with a reduced output power $P_2 = 1.5$ kW [at $P_2 = 2$ kW, the inductor current $i_L(T_{\varphi})$ is approximately zero (Figure 3.4) and, consequently, the time delays of the LV side switches become highly dependent on the actual current $i_L(T_{\varphi})$ (Figure 4.26); even though, these time delays are considered in the employed modulator function, a reduced gain has been measured for phase shift modulation at $V_1 = 340$ V, $V_2 = 12$ V, and $P_2 = 2$ kW; the effect of these time delays is detailed in [122]].



Figure 6.10: Calculated and measured DAB frequency response for the extended triangular current mode modulation [(a) gain and (b) phase]; the Nyquist frequency 1/(2T) = 10 kHz limits the maximum possible frequency; the solid line denotes the calculated frequency response and the boxes mark the measured values. The DAB is operated with $V_1 = 340$ V, $V_2 = 12$ V, and with 2 kW output power at the LV port.



Figure 6.11: Calculated and measured DAB frequency response [(a) gain and (b) phase] for the suboptimal modulation scheme depicted in Figures 5.29; the Nyquist frequency 1/(2T) = 10 kHz limits the maximum possible frequency; the solid line denotes the calculated frequency response and the boxes mark the measured values. The DAB is operated with $V_1 = 340$ V, $V_2 = 12$ V, and with 2 kW output power at the LV port.

The error between measured and calculated frequency response is attributed to instantaneous inductor currents $i_L(t_1)$ and $i_L(t_2)$ close to zero [cf. Figures 5.23 (a) and (b) and Figure 5.29]. There, the characteristics of the delay of the LV switches is highly non-linear (Figure 4.26). Still, for frequencies below 8 kHz, the maximum absolute gain error is less than 1.5 dB and the relative gain error is less than 20%.



Figure 6.12: Simplified circuit of the DAB: the two current sources replace the switches, the inductor L, and the transformer. For a lossless DAB, the accurate current for the source on the right hand side would be equal to $\mathcal{Z}\{i_{1,\text{mod}}(kT) \cdot v_{\text{f1}}(kT)/v_{\text{f2}}(kT)\} \cdot G_{\text{Td}}$ which is approximately equal to $I_{1,\text{mod}}(z) \cdot G_{\text{Td}} \cdot \overline{V}_{\text{f1}}/\overline{V}_{\text{f2}}$. On the assumption of a constant transfer ratio $\overline{V}_{\text{f1}}/\overline{V}_{\text{f2}}$, the simplified circuit can be analyzed by means of Laplace- or Z-transform.

6.2.6 Derivation of a Simplified G_{DAB}

In the presented system, all time constants of the EMI filters are significantly smaller than the time delay caused by the digital controller. The question may now arise, whether a simple DAB converter model which neglects the couplings between the input filter, the output filter, and the DAB power stage itself would be sufficiently accurate in order to design current and voltage controllers. For steady-state operation, the employed modulator functions $\mathbf{G}_{\mathbf{mod},\mathbf{PS}}$ or $\mathbf{G}_{\mathbf{mod},\mathbf{alt}}$ already generate almost correct timing parameters, i.e. $I_{f1} \approx I_{1,\text{mod}}$ and $I_{f2} \approx I_{1,\text{mod}} \cdot \overline{V}_{f1}/\overline{V}_{f2}$. Additionally, the total time delay G_{Td} (DSP, FPGA, and measurement) needs to be regarded according to (6.25), (6.27), and (6.28). Thus, the DAB converter in Figure 6.1 may be replaced with the two current sources depicted in Figure 6.12 (cf. [96]).

In Figure 6.13, the frequency response of G_{DAB} (phase shift modulation, nominal operation) is compared to the frequency response of the simplified transfer function $G_{\text{DAB},\text{simp}}$ derived from the circuit depicted in Figure 6.12. The frequency response obtained from the simplified model differs only slightly from the accurate model which is mainly due to the linearization achieved with the modulator table (regarding the gain response) and due to the digital controller's time delays (regarding the phase response).



Figure 6.13: Frequency response of the detailed transfer function G_{DAB} (solid lines) and the simplified transfer function $G_{\text{DAB,simp}}$ (dashed lines); (a) gain and (b) phase diagrams for $V_1 = 340 \text{ V}$, $V_2 = 12 \text{ V}$, $P_2 = 2 \text{ kW}$, suboptimal modulation; the boxes mark measured values.

6.3 Current and Voltage Controllers

Due to the large capacitance values of $C_{\rm DC1}$ and $C_{\rm DC2}$, constant input and output voltages V_1 and V_2 are assumed for the design of the current controller. Furthermore, according to the considerations discussed in Section 6.2.6, the transfer function $G_{\rm DAB}$ is regarded to be independent of the actual operating point.

The proposed current controller (Figure 6.2) is a discrete-time PI con-

troller,³

$$G_{\rm C,I} = K_{\rm p,I} \frac{z - (1 - T/T_{\rm i,I})}{z - 1}$$
(6.35)

with gain $K_{\rm p,I}$ and cut-off frequency $\omega_{\rm i,I} = 1/T_{\rm i,I}$. Consequently, the open loop transfer function

$$F_{\rm o,I} = G_{\rm C,I} \cdot G_{\rm DAB} \tag{6.36}$$

results. Due to the low pass behavior of G_{DAB} , the following design method is proposed in order to achieve a high bandwidth of the closed current loop transfer function:

- 1. Calculation of the controller cut-off frequency such that $|G_{\text{DAB}}|$ at $\omega_{i,I}$ is 3 dB lower than the DC gain, i.e. $|G_{\text{DAB}}(z_{i,I})| = |G_{\text{DAB}}(e^{j \cdot 0})|/\sqrt{2}$ with $z_{i,I} = e^{j\omega_{i,I}T}$.
- 2. The controller gain $K_{\rm p,I}$ is determined in order to achieve a given phase margin $\Phi_{\rm R}$. First, the z-Parameter $z_{\Phi_{\rm R}}$ where the open loop phase is equal to $-180^{\circ} + \Phi_{\rm R}$ needs to be determined: $\arg(F_{\rm o,I}(z_{\Phi_{\rm R}})/K_{\rm p,I}) =$ $-180^{\circ} + \Phi_{\rm R}$. With this, the controller gain $K_{\rm p,I} = |1/F_{\rm o,I}(z_{\Phi_{\rm R}})|$ can be calculated.

With $\Phi_{\rm R} = 60^{\circ}$, the controller parameters $K_{\rm p,I} = 0.29$ and $T_{\rm i,I} = 93 \,\mu \rm s$ result for the given system setup. The calculated and the measured step response of the closed current control loop are depicted in Figure 6.14. Since the measured current signals are superimposed by noise, the average of 128 singular current step responses is shown; calculated and measured results match closely. The proposed current controller achieves the rise time $t_{\rm r} \approx 250 \,\mu \rm s = 5T$ and the delay time $t_{\rm d} \approx 200 \,\mu \rm s = 4T$ with a percentage overshoot of approximately 7% ($\Delta I_{\rm f1,p} \approx 70 \,\mathrm{mA}$, $\hat{I}_{1,\rm ref} = 1 \,\mathrm{A}$). For the given application, a rise time of less than 500 $\mu \rm s$ and a delay time of less than 500 $\mu \rm s$ are required, which has been achieved with the given controller design. If a faster response is required, the DSP time delays should be reduced (optimized algorithm, faster DSP), instead of using a smaller phase margin, to obtain an adequate control performance (fast response and small overshoot), since the time delays caused by the digital controller account for the main part of the presented converter's phase lag.

The direction of power transfer determines the voltage that needs to be controlled: this is V_1 for power transfer from the LV to the HV port, and V_2 for the opposite direction. However, V_1 and V_2 are not measured in the given system, only v_{f1} and v_{f2} are (Figure 6.1). Therefore, according to the direction of power transfer, either \overline{V}_{f1} or \overline{V}_{f2} is controlled and $V_1 \approx \overline{V}_{f1}$ as

³The transfer function of the digital PI controller is obtained from its continuous-time counterpart [123].



Figure 6.14: Measured and calculated unity step response of the closed current control loop; a rise time t_r of approximately 250 µs and a delay time t_d of approximately 200 µs are achieved.

well as $V_2 \approx \overline{V}_{\rm f2}$ is assumed. For the voltage controller, again a PI controller with the transfer function

$$G_{\rm C,V} = K_{\rm p,V} \frac{z - (1 - T/T_{\rm i,V})}{z - 1}$$
(6.37)

is employed. For power being transferred from the LV to the HV port, the controlled current $I_{\rm f1}$ flows into the output capacitor $C_{\rm DC1}$ and the load. The plant transfer function for the voltage controller $(\overline{V}_{f1}/I_{1,ref}, cf. Figure 6.2)$ therefore consists of the closed current loop transfer function, the filter impedances $L_{\rm f1a}$, $L_{\rm f1b}$, $R_{\rm f1b}$, and $C_{\rm DC1}$ (Figure 6.12) and the load connected to the HV port. If the direction of power transfer changes (HV to LV), then the current $I_{\rm f2}$, which is not measured in the given system, flows into $C_{\rm DC2}$ and into the load connected to the LV port. However, I_{f2} can be approximated with $I_{f1} \cdot \overline{V}_{f1}/\overline{V}_{f2}$ for control purposes, since losses are rather low and the capacitance C_{DC2} is comparably large. The transfer function of the closed current loop again exhibits low-pass characteristics but the output capacitor adds an integration stage to the plant. For this type of plant, the symmetric optimum design method [124] is selected in order to design the voltage controller (in the case of a resistive load, the no-load operation is the worst case for the voltage controller design). A phase margin of $\Phi_{\rm R} = 75^{\circ}$ leads to $K_{\rm p,V} = 79.5 \, {\rm A/V}$ and $T_{\rm i,I} = 24.7 \, {\rm ms}$.



Figure 6.15: Step response of the closed voltage control loop for no-load operation and for 2 kW output power at 12 V output voltage; the input voltage is 340 V and the output voltage step amplitude is 250 mV.

In Figure 6.15, the measured and the calculated step responses of the voltage controlled system are shown for no load and for a load resistance of $72 \text{ m}\Omega = (12 \text{ V})^2/2 \text{ kW}$; again a good matching between measured and calculated results is achieved.

The proposed voltage controller achieves the rise time $t_{\rm r} \approx 6.5 \,\mathrm{ms} = 130T$ and the delay time $t_{\rm d} \approx 1.8 \,\mathrm{ms} = 36T$ with a maximum percentage overshoot of approximately 7% at no-load operation ($\Delta V_{\rm fl,p,max} \approx 18 \,\mathrm{mV}$, $\hat{V}_{\rm l,ref} = 250 \,\mathrm{mV}$).

6.4 Conclusion

To develop an accurate small-signal model for a DAB converter, a precise knowledge of the modulation method is required and the EMI filters need to be included in order to consider their interactions with the DAB. However, simplified converter models may be used in order to facilitate a less laborious controller design, since the time delay of a digitally controlled system causes a significant phase lag, which is considerably larger than the power converter's phase lag. In this Chapter, the small-signal models for the DAB including EMI filter are derived for different modulation schemes. Further, the structure of the digital control system (including the most relevant algorithms), a simplified DAB model and the controller design are detailed. The resulting control-tooutput transfer functions are verified using measurements (obtained from an experimental setup) and a good matching between measured and calculated results is shown. Besides, in this Chapter, a measurement method to obtain the control-to-output transfer functions is proposed; this method may even be applied automatically in order to enable the autonomous calibration of the current and voltage controllers.

Chapter 7

Two-Stage Converter Realizations

7.1 Motivation, Investigated Converter Topologies

According to the results obtained in Chapter 3 and Chapter 5, the DAB converter efficiency considerably depends on the employed input and output voltages (cf. Figure 5.1). With the given DAB hardware prototype, the highest efficiency is achieved for V_1/V_2 close to the transformer turns ratio n. Thus, in order to effectively utilize the DAB converter, an additional DC–DC converter (without galvanic isolation) can be used to provide the required ratio V_1/V_2 . In order to limit the total system complexity, a simple buck-or-boost converter forms the additional converter stage (Figure 7.1: buck conversion)



Figure 7.1: Investigated two-stage circuits: DAB converter and a HV side buck-or-boost converter (without galvanic isolation); (a) $V_i < V_1$, (b) $V_i > V_1$.

in one direction of power transfer, boost conversion for the opposite power transfer direction).

According to Chapter 2, 4 different circuit arrangements are possible. However, the presented investigation is limited to the two arrangements depicted in Figure 7.1; there, the additional DC–DC converter is placed on the HV side. With the additional DC–DC converter being placed on the HV side, considerably lower total losses are expected compared to an additional LV side DC–DC converter [Figures 2.22 (c) and (d); Section 2.2.2].

The topology shown in Figure 7.1 (a) proposes a DC link voltage $V_i < V_1$;¹ consequently, the HV side peak voltages of the DAB decrease and the respective HV side currents increase (switch currents, transformer current, and inductor current on the HV side of the DAB). For $V_i < V_1$, two different situations are investigated: constant $V_i = 225$ V (on the assumption of a maximum duty cycle of the buck-or-boost converter of

$$D_{\rm b,max} = D_{\rm buck,max} = 1 - D_{\rm boost,min} = 0.95,$$
 (7.1)

cf. Appendix A.8) and variable V_i , $225 \text{ V} \cdot 11 \text{ V}/16 \text{ V} \approx 150 \text{ V} \leq V_i \leq 225 \text{ V}$. With constant V_i , the additional DC–DC converter generates lower losses and with variable V_i , the DAB is expected to operate at a higher efficiency.

The second investigated topology [Figure 7.1 (b)] employs $V_i > V_1$ to achieve lower currents on the HV side of the DAB. Again, constant and variable V_i are considered (Table 7.1). However, regarding a variable DC link voltage V_i , the exclusive DAB operation close to V_i/V_2 requires 475 V $\leq V_i \leq$ 475 V $\cdot 16$ V/11 V \approx 700 V, which exceeds the drain-to-source breakdown voltage $V_{(BR)DSS}$ of the employed MOSFETs ($V_{(BR)DSS} = 600$ V). In order to maintain the MOSFETs with $V_{(BR)DSS} = 600$ V,² the variable voltage range is reduced to 475 V $\cdot 11$ V/16 V ≈ 325 V $\leq V_i \leq 475$ V. This selection enables highly efficient operation of the DAB for $V_1/D_{b,max} < nV_2$ and prevents low DAB converter efficiency which would occur for $V_i/V_2 < n$ (e.g. $V_1 = 240$ V and $V_2 = 16$ V in Figure 5.1). Table 7.1 summarizes the voltage ranges employed for V_i .

7.2 Buck-or-Boost Converter Loss Model

In order to achieve a highly efficient and compact design of the additionally required buck-or-boost converter, the circuit depicted in Figure 7.2 is employed. There, the high side switch is composed of 3 power semiconductors: the MOSFET T_I , the Schottky diode $D_{I,s}$, and the silicon-carbide diode D_I . Similarly, the low side switch is composed of the MOSFET T_{II} , the Schottky

¹The index "i" originates from intermediate link voltage [79].

 $^{^2 {\}rm For}~V_{\rm (BR)DSS}=600\,{\rm V},$ optimized MOSFETs with very low on-state resistances are available (Section D.1.2).
No.	Circuit Arrangement	Voltage Range Employed for $V_{\rm i}$
1	Figure 7.1 (a), constant $V_{\rm i}$	$V_{\rm i} = 225 {\rm V}$
2	Figure 7.1 (a), variable $V_{\rm i}$	$V_{\rm i} = 225{\rm V}\cdot\frac{V_2}{16{\rm V}}$
		$\Rightarrow 150 \mathrm{V} \le V_{\mathrm{i}} \le 225 \mathrm{V}$
3	Figure 7.1 (b), constant $V_{\rm i}$	$V_{\rm i} = 475 {\rm V}$
4	Figure 7.1 (b), variable $V_{\rm i}$	$V_{\rm i} = \max\left(\frac{V_1}{D_{\rm b,max}}, 475{\rm V}\cdot\frac{V_2}{16{\rm V}} ight)$
		$\Rightarrow 325 \mathrm{V} \le V_{\mathrm{i}} \le 475 \mathrm{V}$

 Table 7.1: Two-stage converters (Figure 7.1): investigated DC link voltage ranges.

diode $D_{II,s}$, and the silicon-carbide diode D_{II} (the silicon-carbide diodes eliminate reverse recovery effects and the Schottky diodes prevent the internal body diodes of the MOSFETs to conduct). With the proposed switches, low switching losses are achieved on the expense of increased conduction losses. Due to the use of MOSFETs and silicon-carbide diodes, a high switching frequency f_S of 100 kHz is selected in order to achieve both, low converter volume and low converter losses [125].

Depending on the employed mode of operation (i.e. buck or boost operation), either T_I , $D_{I,s}$, and D_{II} or T_{II} , $D_{II,s}$, and D_I are operated; thus, the resulting converter efficiency depends on the actual operating mode.



Figure 7.2: Circuit diagram of the employed buck-or-boost converter [annotation with respect to Figure 7.1 (a)].

7.2.1 Buck Operation

The average and the RMS values of the currents through T_{I} , $D_{I,s}$, D_{II} , and L_{DC} are required in order to calculate the converter losses. Since very low losses are expected for the considered buck-or-boost converter, the lossless converter model is considered to accurately predict these currents. Moreover, constant voltages V_1 and V_i are considered.

The discussed model is used in the subsequent Section 7.3 to predict the expected converter efficiencies at medium and high output power (i.e. $P_{\text{out}} = 1 \text{ kW}$ and $P_{\text{out}} = 2 \text{ kW}$). There, the buck-or-boost converters designed in Appendix A.8 are operated in continuous conduction mode (CCM, [116]); thus, in order to simplify the outlined analysis, only CCM operation is considered.

Figure 7.3 depicts the employed converter model, typical voltage and current waveforms for $(V_1 = 340 \text{ V}, V_i = 225 \text{ V}, \text{ and } P_{\text{out}} = 2 \text{ kW})$, and the relative turn on time of T_I, i.e. the duty cycle D_{buck} .

Based on the above assumptions, the duty cycle D_{buck} is equal to

$$D_{\rm buck} = \frac{V_{\rm i}}{V_1} \tag{7.2}$$

during steady-state converter operation. The average inductor current is

$$\overline{I}_{L_{\rm DC},\rm buck} = \frac{P_{\rm out,buck}}{V_{\rm i}},\tag{7.3}$$



Figure 7.3: (a) Equivalent circuit diagram employed to calculate the currents through T_I, D_{I,s}, D_{II}, and $L_{\rm DC}$ during buck operation; (b) typical voltage and current waveforms during buck operation ($V_1 = 340$ V, $V_i = 225$ V, $P_{\rm out,buck} = 2$ kW, $L_{\rm DC} = 150 \,\mu$ H, $f_{\rm S} = 100$ kHz).

the peak-to-peak value of the inductor current ripple is

$$\Delta I_{L_{\rm DC},\rm buck} = \frac{V_{\rm i}(1 - D_{\rm buck})}{L_{\rm DC}f_{\rm S}},\tag{7.4}$$

and the inductor RMS current is

$$I_{L_{\rm DC},\rm buck} = \sqrt{\overline{I}_{L_{\rm DC},\rm buck}^2 + \frac{\Delta I_{L_{\rm DC},\rm buck}^2}{12}}.$$
(7.5)

Consequently, the average switch current $\overline{I}_{T_{I,buck}}$ and the switch RMS current $I_{T_{I,buck}}$ are:

$$\overline{I}_{T_{I,buck}} = D_{buck} \overline{I}_{L_{DC},buck}, \qquad (7.6)$$

$$I_{\rm T_I,buck} = \sqrt{D_{\rm buck}} I_{L_{\rm DC},\rm buck}.$$
(7.7)

The current through the diode D_{II} causes the current stresses:

$$\overline{I}_{D_{\rm II,buck}} = (1 - D_{\rm buck})\overline{I}_{L_{\rm DC},\rm buck},\tag{7.8}$$

$$I_{\rm D_{II}, buck} = \sqrt{1 - D_{\rm buck}} I_{L_{\rm DC}, \rm buck}$$
(7.9)

 $(\overline{I}_{D_{II},buck}: average diode current; I_{D_{II},buck}: diode RMS current).$

The discussed expressions are valid for the circuit depicted in Figure 7.1 (a); V_1 and V_i need to be interchanged in order to determine the currents of the buck-or-boost converter shown in Figure 7.1 (b).

7.2.2 Boost Operation

On the assumptions given in Section 7.2.1 (lossless buck-or-boost converter, constant voltages V_1 and V_i , CCM operation), $P_{\text{out,buck}} = P_{\text{out,boost}}$ applies; moreover, the relative turn-on time of T_{II} , D_{boost} , is equal to

$$D_{\text{boost}} = 1 - \frac{V_{\text{i}}}{V_{1}} = 1 - D_{\text{buck}}$$
 (7.10)

during steady-state operation (Figure 7.4).

The inductor current for boost operation is thus equal to the inductor current for buck operation, i.e.:

$$\overline{I}_{L_{\rm DC},\rm boost} = \overline{I}_{L_{\rm DC},\rm buck},\tag{7.11}$$

$$\Delta I_{L_{\rm DC},\rm boost} = \Delta I_{L_{\rm DC},\rm buck},\tag{7.12}$$

$$I_{L_{\rm DC},\rm boost} = I_{L_{\rm DC},\rm buck}.$$
(7.13)



Figure 7.4: (a) Equivalent circuit diagram employed to calculate the currents through T_I, D_{I,s}, D_{II}, and $L_{\rm DC}$ during boost operation; (b) typical voltage and current waveforms during boost operation ($V_1 = 340$ V, $V_i = 225$ V, $P_{\rm out,boost} = 2$ kW, $L_{\rm DC} = 150 \,\mu$ H, $f_{\rm S} = 100$ kHz).

During boost operation, the switch and diode currents are:

$$\overline{I}_{T_{\rm II},\rm boost} = D_{\rm boost} \overline{I}_{L_{\rm DC},\rm boost}, \qquad (7.14)$$

$$I_{\rm T_{II},boost} = \sqrt{D_{\rm boost}} I_{L_{\rm DC},boost}, \tag{7.15}$$

$$\overline{I}_{D_{I},\text{boost}} = (1 - D_{\text{boost}})\overline{I}_{L_{DC},\text{boost}}, \qquad (7.16)$$

$$I_{\rm D_I,boost} = \sqrt{1 - D_{\rm boost}} I_{L_{\rm DC},\rm boost}$$
(7.17)

 $(\overline{I}_{T_{II},boost}, I_{T_{II},boost}; average value and RMS value of <math>i_{T_{II},boost}(t)$, respectively; $\overline{I}_{D_{I},boost}, I_{D_{I},boost}; average value and RMS value of <math>i_{D_{I},boost}(t)$, respectively).

The expressions discussed in Section 7.2.1 and Section 7.2.2 are valid for the circuit depicted in Figure 7.1 (a); in order to determine the currents of the buck-or-boost converter in Figure 7.1 (b), V_1 and V_i need to be interchanged.

7.2.3 Conduction Losses, Copper Losses, Dielectric Losses

Buck Operation

During buck operation, T_I, D_{I,s}, and D_{II} cause conduction losses:

$$P_{\text{cond},\text{T}_{\text{I}},\text{buck}} = R_{\text{T}_{\text{I}}} I_{\text{T}_{\text{I}},\text{buck}}^2, \tag{7.18}$$

$$P_{\text{cond},\text{D}_{\text{I},\text{s}},\text{buck}} = V_{\text{D}_{\text{I},\text{s}}} \overline{I}_{\text{T}_{\text{I}},\text{buck}} + R_{\text{D}_{\text{I},\text{s}}} I_{\text{T}_{\text{I}},\text{buck}}^2,$$
(7.19)

$$P_{\text{cond},\text{D}_{\text{II}},\text{buck}} = V_{\text{D}_{\text{II}}} \overline{I}_{\text{D}_{\text{II}},\text{buck}} + R_{\text{D}_{\text{II}}} I_{\text{D}_{\text{II}},\text{buck}}^2.$$
(7.20)

The MOSFET's on-state resistance $R_{\text{T}_{\text{I}}}$, the diodes' forward voltage drops $V_{\text{D}_{\text{I}s}}$, $V_{\text{D}_{\text{I}}}$ and the diodes' bulk resistances $R_{\text{D}_{\text{I}s}}$, $R_{\text{D}_{\text{I}}}$ are given in Table 7.2.

Variable	Value	Description				
CoolMOS SPW47N60C3:						
$R_{\mathrm{T_{I}}}, R_{\mathrm{T_{II}}}$	$70\mathrm{m}\Omega$	on-state resistances of T_{I} and T_{II}				
DSS25-0025B	DSS25-0025B					
$V_{\mathrm{D}_{\mathrm{I,s}}}, V_{\mathrm{D}_{\mathrm{II,s}}}$	$0.3\mathrm{V}$	cut-in voltage of $D_{I,s}$ and $D_{II,s}$				
$R_{\mathrm{D}_{\mathrm{I},\mathrm{s}}},R_{\mathrm{D}_{\mathrm{II},\mathrm{s}}}$	$8.6\mathrm{m}\Omega$	diode series resistances of $D_{I,s}$ and $D_{II,s}$				
SDT12S60	SDT12S60					
$V_{\mathrm{D}_{\mathrm{I}}}, V_{\mathrm{D}_{\mathrm{II}}}$	$1.0\mathrm{V}$	cut-in voltage of D_{I} and D_{II}				
$R_{\mathrm{D_{I}}}, R_{\mathrm{D_{II}}}$	$43\mathrm{m}\Omega$	diode series resistances of D_{I} and D_{II}				
$L_{\rm DC} = 150 \mu {\rm H}$	[employed in	Figure 7.1 (a); cf. Figure A.49]:				
$R_{L_{\rm DC},{ m DC}}$	$26\mathrm{m}\Omega$	inductor's DC resistance				
$R_{L_{\rm DC},{ m AC}}$	$290\mathrm{m}\Omega$	inductor's AC resistance at $100\rm kHz$				
$L_{\rm DC} = 200 \mu\text{H}$ [employed in Figure 7.1 (b); cf. Figure A.50]:						
$R_{L_{\rm DC},\rm DC}$	$45\mathrm{m}\Omega$	inductor's DC resistance				
$R_{L_{\rm DC},{ m AC}}$	$390\mathrm{m}\Omega$	inductor's AC resistance at $100\rm kHz$				
Electrolytic capacitors used for $C_{\rm DCi}$ [cf. Figure A.51]:						
$R_{C_{\rm DCi}}$	$250\mathrm{m}\Omega$	total equivalent series resistance (100 kHz)				

Table 7.2: Converter parameters used to predict the conduction losses of the buck-or-boost converter at room temperature ($T_i = 25^{\circ}$ C; cf. Chapter 4).

The copper losses in the DC inductor are approximately calculated with

$$P_{\text{cond},L_{\text{DC}},\text{buck}} = R_{L_{\text{DC}},\text{DC}}\overline{I}_{L_{\text{DC}},\text{buck}}^2 + R_{L_{\text{DC}},\text{AC}}\frac{\Delta I_{L_{\text{DC}},\text{buck}}^2}{12}$$
(7.21)

 $(R_{L_{\rm DC},\rm DC})$ is the inductor's DC resistance and $R_{L_{\rm DC},\rm AC}$ is the AC resistance of the inductor at $f = 100 \,\rm kHz$.

The losses generated in $C_{\rm DC1}$ are neglected, since film capacitors are employed. For $C_{\rm DCi}$, however, electrolytic capacitors are used (cf. Figure A.51) with a measured total equivalent series resistance of 250 m Ω at 100 kHz. The respective losses are:

$$P_{\text{ESR},C_{\text{DCi}},\text{buck}} = R_{C_{\text{DCi}}} \frac{\Delta I_{L_{\text{DC}},\text{buck}}^2}{12}.$$
(7.22)

Boost Operation

During boost operation, T_{II}, D_{II,s}, and D_I cause conduction losses:

$$P_{\text{cond},\text{T}_{\text{II}},\text{boost}} = R_{\text{T}_{\text{II}}} I_{\text{T}_{\text{II}},\text{boost}}^2, \qquad (7.23)$$

$$P_{\text{cond},\text{D}_{\text{II},\text{s}},\text{boost}} = V_{\text{D}_{\text{II},\text{s}}} \overline{I}_{\text{T}_{\text{II}},\text{boost}} + R_{\text{D}_{\text{II},\text{s}}} I_{\text{T}_{\text{II}},\text{boost}}^2, \qquad (7.24)$$

$$P_{\text{cond},\text{D}_{\text{I}},\text{boost}} = V_{\text{D}_{\text{I}}} \overline{I}_{\text{D}_{\text{I}},\text{boost}} + R_{\text{D}_{\text{I}}} I_{\text{D}_{\text{I}},\text{boost}}^2.$$
(7.25)

The MOSFET's on-state resistance and the diodes' forward voltage drops and bulk resistances are given in Table 7.2.

The copper losses in the DC inductor are approximately calculated with

$$P_{\text{cond},L_{\text{DC}},\text{boost}} = R_{L_{\text{DC}},\text{DC}}\overline{I}_{L_{\text{DC}},\text{boost}}^2 + R_{L_{\text{DC}},\text{AC}}\frac{\Delta I_{L_{\text{DC}},\text{boost}}^2}{12}.$$
 (7.26)

The losses in $C_{\rm DCi}$ are:

$$P_{\text{ESR},C_{\text{DCi}},\text{boost}} = R_{C_{\text{DCi}}} \frac{\Delta I_{L_{\text{DC}},\text{boost}}^2}{12}.$$
(7.27)

7.2.4 Switching Losses

Buck Operation

During buck operation, $T_{\rm I}$ is actively switched and $T_{\rm II}$ remains in the offstate. The switching losses of $T_{\rm I}$ comprise of the turn-on losses and the turnoff losses of this MOSFET; the switching losses of the Schottky diode $D_{\rm I,s}$ and the silicon-carbide diode $D_{\rm II}$ are neglected. According to the discussions given in Section 4.3.2, the MOSFET's effective output capacitance enables ZVS and thus, very low turn-off losses result. Accordingly, the turn-off losses of the buck-or-boost converter are neglected.

In contrast, during turn-on, the involved MOSFET actively charges or discharges the parasitic capacitances depicted in Figure 7.5, which causes turn-on losses. The equivalent circuit used to calculate the turn-on losses (Figure 7.5) contains:

- the effective output capacitances of the employed MOSFETs $[C_{DS,TI}(v_{TI}), C_{DS,TII}(v_{TII});$ Figure 7.6 (a)],
- the junction capacitances of the silicon-carbide diodes $[C_{D_{I}}(v_{D_{I}}), C_{D_{II}}(v_{D_{II}});$ Figure 7.6 (b)],
- the backside to heat sink capacitances of the MOSFETs and the diodes (Table 7.3), and

• the winding capacitance of $L_{\rm DC}$, $C_{L_{\rm DC}}$, estimated using the measured impedances depicted in Figures A.49 and A.50. For $L_{\rm DC} = 150 \,\mu\text{H}$, a parallel resonant frequency of $f_{0,\rm p} = 2.9 \,\text{MHz}$ is measured and thus, $C_{L_{\rm DC}} = 20 \,\text{pF}$ results; for $L_{\rm DC} = 200 \,\mu\text{H}$, parallel resonance occurs for 3.5 MHz and $C_{L_{\rm DC}}$ is approximately equal to 11 pF.

The heat sink is left on floating potential (not connected to ground) and thus, the backside to heat sink capacitances are connected in star between the nodes (D, (Q), (Q), (Q), (Q), (Q)) [depicted in Figure 7.7 (a)]:

- $C_{\rm Y1} = C_{\rm TO220} + C_{\rm TO247},$
- $C_{\rm Y2} = 2C_{\rm TO220} + C_{\rm TO247},$
- $C_{Y3} = C_{TO220}$.

The outlined method used to predict the turn-on losses further considers a very simple MOSFET model: each MOSFET is replaced by the series con-



Figure 7.5: Parasitic capacitors considered to predict the turn-on losses of $T_{\rm I}$: $C_{\rm DS,T_{\rm I}}$ and $C_{\rm DS,T_{\rm I}}$ denote the output capacitances of the MOSFETs [Figure 7.6 (a)], $C_{\rm D_{\rm I}}$, and $C_{\rm D_{\rm II}}$ are the silicon-carbide diodes' junction capacitances [Figure 7.6 (b)], $C_{\rm TO220}$ denotes the capacitance of a single TO-220 package (capacitances between the diodes' cathodes and the heat sink), and $C_{\rm TO247}$ is the capacitance of a TO-247 package (capacitances between the MOSFETs' drain connections and the heat sink).



Figure 7.6: (a) Effective output capacitance (i.e. drain-to-source capacitance) of the employed MOSFET (CoolMOS, SPW47N60C3; obtained from [153]); (b) junction capacitance of the employed silicon-carbide diode (SDT12S60; obtained from [153]).

nection of an ideal switch and its on-state resistance (the body diodes of the MOSFETs never conduct and are thus not included).

Turning on $S_{\rm I}$ starts the charging or discharging of all the capacitors shown in Figure 7.5 via $R_{\rm T_{I}}$. During the turn-on process, the diodes $D_{\rm I,s}$ and $D_{\rm II,s}$ conduct and the diodes $D_{\rm I}$ and $D_{\rm II}$ block. Therefore, the circuit shown in Figure 7.5 can be replaced by the simplified circuit depicted in Figure 7.7 (b). During the duration of the turn-on process, constant inductor current $I_{L_{\rm DC}}$ is assumed and the voltage drops across $D_{\rm I,s}$ and $D_{\rm II,s}$ are neglected.

In order to determine $C_{\rm I}$ and $C_{\rm II}$, the star connection of $C_{\rm Y1}$, $C_{\rm Y2}$, and $C_{\rm Y2}$ needs to be transformed to a delta circuit:

- $C_{\text{D12}} = (C_{\text{TO220}} + C_{\text{TO247}})/2,$
- $C_{\text{D23}} = C_{\text{TO220}}/2,$
- $C_{\text{D13}} = C_{\text{TO220}} \left(C_{\text{TO220}} + C_{\text{TO247}} \right) / \left[2 \left(2 C_{\text{TO220}} + C_{\text{TO247}} \right) \right]$

[cf. Figure 7.7 (a)]. The resulting capacitances $C_{\rm I}$ and $C_{\rm II}$ [Figure 7.7 (b)] comprise of:

- $C_{\rm I} = C_{\rm DS,T_{\rm I}} + C_{\rm D_{\rm I}} + C_{\rm D12},$
- $C_{\text{II}} = C_{\text{DS},\text{T}_{\text{II}}} + C_{\text{D}_{\text{II}}} + C_{\text{D}_{23}} + C_{L_{\text{DC}}}$ (on the assumption $C_{L_{\text{DC}}} \ll C_{\text{DCi}}$).

Variable	Value	Description		
KOOL-PADS K200 (thermally conductive and electr. insulating foil):				
$\lambda_{ m iso}$	$1.30 \frac{W}{mK}$	thermal conductivity		
$\epsilon_{ m r,iso}$	2.5	relative permittivity (measured at $1 \mathrm{kHz}$)		
$d_{\rm iso}$	$0.2\mathrm{mm}$	thickness (typical)		
$V_{\rm iso,br}$	$1000\mathrm{VRMS}$	breakdown voltage at $50\mathrm{Hz}$		
TO-220:				
$A_{\rm TO220}$	$13.5\mathrm{mm}\times10\mathrm{mm}$	considered capacitor surface		
$C_{\rm TO220}$	$15\mathrm{pF}$	backside to heat sink capacitance		
TO-247:				
$A_{\rm TO247}$	$20.9\mathrm{mm}\times15.9\mathrm{mm}$	considered capacitor surface		
C_{TO247}	$37\mathrm{pF}$	backside to heat sink capacitance		

Table 7.3: Parasitic capacitances formed by the diodes' backsides and the heat sink and the MOSFETs' backsides and the heat sink (cf. Figure 7.5). Between the backsides of the power semiconductors and the heat sink, a thermally conductive and electrically insulating foil (KOOL-PADS K200) is used.

The energy $E_{\text{on},\text{T}_{I},\text{buck}}$, dissipated in $R_{\text{T}_{I}}$ during discharging or charging C_{I} and C_{II} , is used as an estimate of the turn-on losses. $E_{\text{on},\text{T}_{I},\text{buck}}$ can be obtained with an energy balance calculation:

$$E_{\text{stored,final}} + E_{\text{dissipated}} = E_{\text{stored,initial}} + E_{\text{delivered}},$$
 (7.28)

where $E_{\text{stored,initial}}$ denotes the energy stored in C_{I} and C_{II} at the beginning of the turn-on switching operation, $E_{\text{stored,final}}$ is the energy stored in C_{I} and C_{II} at the end of the switching operation, $E_{\text{delivered}}$ denotes the energy delivered by the source V_1 , and $E_{\text{dissipated}}$ is the energy dissipated during turn-on. According to [127] the energy $E_C(V)$, which is stored in a non-linear capacitor C(v) at a capacitor voltage V, is obtained from:

т.

$$E_C(V) = \int_0^V v C(v) dv = \frac{1}{2} C_{\mathrm{E,eq}}(V) V^2, \qquad (7.29)$$

where $C_{\rm E,eq}(V)$ denotes the energy equivalent capacitance value at a given capacitor voltage [Figure 7.8 (a) depicts the energy equivalent output capacitance of the SPW47N60C3 CoolMOS]. Since C(v) is a differential capacitance,

$$C(v) = \frac{\mathrm{d}q}{\mathrm{d}v},\tag{7.30}$$



Figure 7.7: (a) Star connection of the parasitic backside to heat sink capacitances of the MOSFETs and the diodes (dashed lines: delta connection obtained with the wye-delta transformation). (b) simplified circuit employed to estimate the turn-on losses (buck operation). The electric capacitance of the heat sink with respect to earth is neglected; the presented calculation and the equivalent circuit need to be modified if a considerably high capacitance with respect to earth is present, e.g. by using the star to polygon transformation given in [126].

the charge $Q_C(V)$, which is stored in C(v) at a capacitor voltage V, is given with [107]:

$$Q_C(V) = \int_{0}^{V} C(v) dv = C_{Q,eq}(V)V;$$
(7.31)

 $C_{Q,eq}(V)$ represents the charge equivalent capacitance value at a given capacitor voltage [Figure 7.8 (b)]. The particular energies used in (7.28) can be expressed with (7.29) and (7.31):

- $E_{\text{stored,initial}} = C_{\text{I,E,eq}}(V_1)V_1^2/2$, since C_{I} is charged to V_1 and C_{II} is fully discharged at the beginning of the turn-on switching operation of S_{I} ;
- $E_{\text{stored,final}} = C_{\text{II,E,eq}}(V_1)V_1^2/2$, since C_{I} is fully discharged and C_{II} is charged to V_1 at the end of the turn-on switching operation of S_{I} ;
- $E_{\text{delivered}} = V_1 \Delta Q = V_1 C_{\text{II},\text{Q},\text{eq}}(V_1) V_1 = V_1^2 C_{\text{II},\text{Q},\text{eq}}(V_1)$, since the source V_1 delivers the charge stored in C_{II} after the switching operation has completed (the charge of C_{I} is dissipated in $R_{\text{T}_{\text{I}}}$);



Figure 7.8: (a) Energy equivalent output capacitance of the SPW47N60C3 CoolMOS calculated with (7.29); (b) charge equivalent output capacitance of the SPW47N60C3 CoolMOS calculated with (7.31). The effective output capacitance of the SPW47N60C3 CoolMOS (Figure 7.6 (a)) is substituted for C(v) used in (7.29) and (7.31).

• $E_{\text{dissipated}} = E_{\text{on}, T_{\text{I}}, \text{buck}}$.

Thus, the result for $E_{\text{on},T_{I},\text{buck}}$ is:

$$E_{\text{on},\text{T}_{\text{I}},\text{buck}} = \frac{1}{2} C_{\text{I},\text{E},\text{eq}}(V_1) V_1^2 - \frac{1}{2} C_{\text{II},\text{E},\text{eq}}(V_1) V_1^2 + V_1^2 C_{\text{II},\text{Q},\text{eq}}(V_1). \quad (7.32)$$

The turn-on losses are finally calculated using:

$$P_{\rm on,T_I,buck} = f_{\rm S} E_{\rm on,T_I,buck}.$$
(7.33)

The energy and charge equivalent capacitances calculated for $C_{\rm I}(V_1)$ and $C_{\rm II}(V_1)$ are shown in Figure 7.9.

Boost Operation

During boost operation, $T_{\rm II}$ is actively switched and $T_{\rm I}$ remains in the offstate. Again, the turn-off losses of $T_{\rm II}$ are neglected and the turn-on losses are estimated according to the discussion given in the Section above. However, since the direction of $I_{L_{\rm DC}}$ reverses [cf. Figure 7.7 (b)], $C_{\rm I}$ and $C_{\rm II}$ need to be interchanged in order to calculate the turn-on losses $P_{\rm on,T_{II},boost}$:

$$E_{\rm on,T_{II},boost} = \frac{1}{2} C_{\rm II,E,eq}(V_1) V_1^2 - \frac{1}{2} C_{\rm I,E,eq}(V_1) V_1^2 + V_1^2 C_{\rm I,Q,eq}(V_1), \quad (7.34)$$

$$P_{\rm on,T_{II},boost} = f_{\rm S} E_{\rm on,T_{II},boost}.$$
(7.35)



Figure 7.9: (a) Energy equivalent capacitance values calculated for $C_{\rm I}(v)$; (b) charge equivalent capacitance values calculated for $C_{\rm I}(v)$; (c) energy equivalent capacitance values calculated for $C_{\rm II}(v)$; (d) charge equivalent capacitance values calculated for $C_{\rm II}(v)$; (d) charge equivalent capacitance values calculated for $C_{\rm II}(v)$. $C_{\rm I}(v)$ and $C_{\rm II}(v)$ of the investigated buck-or-boost converter are nearly equal.

7.2.5 Core Losses

The core losses of the DC inductor $L_{\rm DC}$ are calculated with the Steinmetz equation,

$$P_{\text{core},L_{\text{DC}}} = dpf(H_{L_{\text{DC}},\text{DC}}) \cdot V_{\text{core},L_{\text{DC}}} \cdot k \cdot f_{\text{S}}^{\alpha} \cdot B_{L_{\text{DC}},\text{AC},\text{peak}}^{\beta}; \qquad (7.36)$$

the displacement factor $dpf(H_{L_{\rm DC},\rm DC})$ considers the increase of the core losses due to the DC inductor current $\overline{I}_{L_{\rm DC},\rm buck}$ or $\overline{I}_{L_{\rm DC},\rm boost}$



Displacement Factor (Ferrite Material N87)

Figure 7.10: Relative increase of the core losses by reason of a DC biased inductor core (ferrite material N87); $H_{\rm DC}$ denotes the DC component of the magnetic field in the inductor core, $B_{\rm AC,peak}$ is the amplitude of the AC component of the flux density (from [128]; measured at a core temperature of 100°C and at a frequency of 100 kHz).

[(7.3), (7.11); Figure 7.10, [128]]. The actual value of $dpf(H_{L_{\rm DC},\rm DC})$ is obtained based on the data given in Figure 2 in [128] and linear approximation.³ The DC component of the magnetic field in the ferrite core is:

$$H_{L_{\rm DC},\rm DC} \approx \frac{L_{\rm DC} \,\overline{I}_{L_{\rm DC}}}{\mu_0 \mu_{\rm r, L_{\rm DC}} N_{L_{\rm DC}} A_{\rm core, L_{\rm DC}}} \tag{7.37}$$

and the amplitude of the AC component of the magnetic flux density is:

$$B_{L_{\rm DC},\rm AC,peak} = \frac{L_{\rm DC} \Delta I_{L_{\rm DC}}/2}{N_{L_{\rm DC}} A_{\rm core,L_{\rm DC}}}.$$
(7.38)

Table 7.4 summarizes the remaining parameters used in (7.36), (7.37), and (7.38).

7.2.6 Gate Driver Losses

With the employed power MOSFETs (SPW47N60C3), at rated output power ($P_{out} = 2 \text{ kW}$), $V_1 = 340 \text{ V}$, $V_i = 225 \text{ V}$, buck operation (only T_I is actively

³In [128] the displacement factor depicted in Figure 7.10 is measured for the ferrite material N87 at 100 kHz and for a core temperature $T_{\rm core}$ of 100°C. The presented converter model, however, considers a core temperature of 25°C. Still, with the data given in [128] being directly adopted for a core temperature of $T_{\rm core} = 25^{\circ}$ C, acceptable total losses are calculated (cf. Figure 7.11). However, if a more accurate calculation of the core losses is required, new measurements have to be conducted at $T_{\rm core} = 25^{\circ}$ C.

Variable	Value	Description					
Currents:							
$\overline{I}_{L_{\mathrm{DC}}}$	$\overline{I}_{L_{\rm DC}, \rm buck}$ (buck operation)	average inductor current					
	$\overline{I}_{L_{\rm DC},\rm boost}$ (boost operation)						
$\Delta I_{L_{\rm DC}}$	$\Delta I_{L_{\rm DC}, \rm buck}$ (buck operation)	peak-to-peak current value					
	$\Delta I_{L_{\rm DC},\rm boost}$ (boost operation)						
Inductor	Inductor parameters:						
$A_{\mathrm{core},L_{\mathrm{DC}}}$	$519 \mathrm{mm^2} \ (L_{\mathrm{DC}} = 150 \mathrm{\mu H})$	core cross section area					
$A_{\mathrm{core},L_{\mathrm{DC}}}$	$310 \mathrm{mm}^2 \ (L_{\mathrm{DC}} = 200 \mu\mathrm{H})$						
$V_{\mathrm{core},L_{\mathrm{DC}}}$	$36.2 \mathrm{cm}^3 \left(L_{\mathrm{DC}} = 150 \mathrm{\mu H}\right)$	total core volume					
$V_{\mathrm{core},L_{\mathrm{DC}}}$	$20.8 \mathrm{cm}^3 \ (L_{\mathrm{DC}} = 200 \mathrm{\mu H})$						
$N_{L_{\rm DC}}$	$16 \ (L_{\rm DC} = 150 \ \mu {\rm H})$	employed number of turns					
$N_{L_{\rm DC}}$	$24 \ (L_{\rm DC} = 200 \ \mu {\rm H})$						
Core material parameters (ferrite, N87):							
$\mu_{\mathrm{r},L_{\mathrm{DC}}}$	2200	relative initial permeability					
k	14.1	Steinmetz-parameters for a core					
α	1.275	temperature of 25° C, obtained					
β	2.416	from data sheet values, App. C.					

Table 7.4: Additional data used to calculate the core losses of $L_{\rm DC}$.

switched), and 100 kHz switching frequency, an input current of the high side gate driver of $\approx 40 \text{ mA}$ is measured at a supply voltage of 12 V. On the assumption of an efficiency of 75% of the isolated DC–DC converter, used to supply the high side gate driver, and an efficiency of 75% of the auxiliary power supply, needed to provide the supply voltages for the converter electronics (5 V, 12 V) from one of the DC voltages (e.g. V_1), the gate driver's power demand is

$$P_{\text{gate,buck}} = 12 \,\mathrm{V} \cdot 40 \,\mathrm{mA} / (0.75 \cdot 0.75) = 0.85 \,\mathrm{W}.$$
 (7.39)

During boost operation, only $T_{\rm II}$ is actively switched, and the measured input current of the respective gate driver is again ≈ 40 mA at 12 V. For the low side gate driver, the 12 V DC voltage provided from the auxiliary power supply is directly used to supply the gate driver. On the assumption of a power supply

efficiency of 75%, the gate driver's power demand is

$$P_{\text{gate,boost}} = 12 \,\mathrm{V} \cdot 40 \,\mathrm{mA}/0.75 = 0.64 \,\mathrm{W}.$$
 (7.40)

7.2.7 Total Buck-or-Boost Converter Losses

Buck Operation

During buck operation, the proposed loss model considers the total losses according to:

$$P_{t,buck} = P_{cond,T_{I},buck} + P_{on,T_{I},buck} + P_{cond,D_{I,s},buck} + P_{cond,D_{II},buck} + P_{cond,L_{DC},buck} + P_{ESR,C_{DC},buck} + P_{gate,buck}.$$
 (7.41)

Boost Operation

During boost operation, the total losses are:

$$P_{t,boost} = P_{cond,T_{II},boost} + P_{on,T_{II},boost} + P_{cond,D_{II,s},boost} + P_{cond,D_{I},boost} + P_{cond,L_{DC},boost} + P_{ESR,C_{DCi},boost} + P_{gate,boost}.$$
 (7.42)

7.2.8 Results

In Figure 7.11 the measured losses of the buck-or-boost converter prototype, designed in Appendix A.8, are compared to the losses calculated with (7.41) [Figure 7.11 (a)] or (7.42) [Figure 7.11 (b)]. There, the nominal input voltage, $V_1 = 340$ V, and the DC link voltage $V_i = 225$ V are used [according to the converter setup depicted in Figure 7.1 (a); cf. Table 7.1]. Similar to the DAB, the influence of raised component temperatures is not included (cf. Chapter 4); the losses are calculated for component and junction temperatures of 25° C (in Figure 7.11, the \Box -symbols denote the efficiencies measured after a short run time of 30 seconds and the \triangle -symbols denote the efficiencies measured at continuous converter operation).

With the simple converter model discussed in this Chapter, acceptable matching between measured and calculated losses is achieved. The accuracy achieved with the discussed buck-or-boost converter model is considered to be sufficient, if buck-or-boost converter and DAB are connected in series (Figure 7.1) since the DAB converter losses are considerably larger than the buck-or-boost converter losses.⁴

⁴The model improvements discussed in Chapter 4, in particular the accurate consideration of conduction losses and the use of measured switching losses, could be applied to the buck-or-boost converter if a better matching between measured and calculated losses would be needed.



Figure 7.11: Measured and calculated buck-or-boost converter losses for (a) buck operation and (b) boost operation; $V_1 = 340$ V, $V_i = 225$ V [Figure 7.1 (a)], $L_{\rm DC} = 150 \,\mu\text{H}$, $f_{\rm S} = 100 \,\text{kHz}$, and $T = 25^{\circ}\text{C}$. The solid line denotes the calculated losses (CCM operation), the rectangles mark the measured losses for a short run time of 30 seconds (cf. Chapter 4) and the Δ -symbols mark the losses measured at continuous operation ($\approx 10 \,\text{minutes}$). The dashed line denotes discontinuous conduction mode (DCM) operation; there, the calculations are not valid.

7.3 Two-Stage Converters – Efficiencies

The efficiency calculations for the converters depicted in Figure 7.1 employ the DAB converter model detailed in Chapter 4 and the buck-or-boost converter model discussed in Section 7.2, whereas the power transfer direction determines the input and output power levels of the converters:

- HV → LV: the DAB input power is equal to the output power of the buck-or-boost converter, P_{in,DAB} = P_{out,b} [Figure 7.12 (a)],
- LV \rightarrow HV: the input power of the buck-or-boost converter is equal to the DAB output power, $P_{\text{in,b}} = P_{\text{out,DAB}}$ [Figure 7.12 (b)].

The DAB is operated with the optimized modulation scheme discussed in Section 5.2.2 in order to achieve maximum efficiency. Moreover, for each of the 4 circuit variants listed in Table 7.1, the DAB is separately designed according to Appendix A.2 using the voltage ranges specified in Table 1.4 and Table 7.1. The resulting DAB converter parameters n and L for maximum average efficiency $\overline{\eta}$ [calculated with (A.1), (A.2), (A.3), and (A.4)] of the twostage converters are listed in Table 7.5. Besides, the value of the DC inductor $L_{\rm DC}$ depends on the circuit arrangement [i.e. buck/boost in Figure 7.1 (a) or boost/buck in Figure 7.1 (b); cf. Appendix A.8 and Table 7.5].



Figure 7.12: The calculation of the total converter efficiency employs the loss models of the buck-or-boost converter (Section 7.2) and of the DAB converter (Chapter 4). The considered input and output power levels depend on the direction of power transfer: (a) power transfer from the HV to the LV port, (b) power transfer from the LV to the HV port.

Figure 7.13 and Figure 7.14 depict the calculated total converter efficiencies for power being transferred from the HV port to the LV port, $P_{\text{out}}(=P_{\text{out},\text{DAB}}) = 1 \text{ kW}$ and $P_{\text{out}} = 2 \text{ kW}$, and the 4 circuit variants listed in Table 7.1:

- Figure 7.13 (a), (b): $V_i = 225 V_i$
- Figure 7.13 (c), (d): $V_i = 225 \text{ V} \cdot \frac{V_2}{16 \text{ V}}$, i.e. $150 \text{ V} \le V_i \le 225 \text{ V}$,
- Figure 7.14 (a), (b): $V_i = 475 V_i$
- Figure 7.14 (c), (d): $V_{\rm i} = \max\left(\frac{V_{\rm i}}{D_{\rm b,max}}, 475 \,\mathrm{V} \cdot \frac{V_{\rm 2}}{16 \,\mathrm{V}}\right)$, i.e. $325 \,\mathrm{V} \le V_{\rm i} \le 475 \,\mathrm{V}$.

7.3.1 Efficiencies for $V_i \leq 225 \text{ V}$ [Figure 7.1 (a)]

The converter setup depicted in Figure 7.1 (a) achieves highest efficiencies at low HV port voltages, i.e. for V_1 close to 240 V (Figure 7.13). Moreover, regarding the operation with $P_{\text{out}} = 2 \text{ kW}$ [Figures 7.13 (b) and (d)], the use of a constant DC link voltage, $V_i = 225 \text{ V}$, facilitates higher converter efficiencies than the operation with a variable DC link voltage, $150 \text{ V} \le V_i \le 225 \text{ V}$: with decreasing DC link voltages, the MOSFET currents and the diode currents of the buck-or-boost converter increase, which causes the conduction losses to increase and the efficiency to decrease. Consequently, the average efficiency achieved with constant DC link voltage ($\overline{\eta} = 93.0\%$) is higher than the average efficiency achieved with variable DC link voltage ($\overline{\eta} = 92.7\%$, Table 7.5).



Figure 7.13: Efficiencies achieved with the two-stage circuit variants 1 and 2 listed in Table 7.1 [Figure 7.1 (a)], $P_{\text{out}} = 1 \text{ kW}$ or $P_{\text{out}} = 2 \text{ kW}$, and power being transferred from the HV port to the LV port: (a), (b) constant $V_i = 225 \text{ V}$; (c), (d) variable $150 \text{ V} \leq V_i \leq 225 \text{ V}$. The efficiency calculation considers the effect of the linear interpolation used by the DSP to determine the actual modulation parameters (Appendix E.2), which causes the irregular characteristics of the depicted contour lines. Moreover, the irregular efficiency characteristics at $V_2 \approx 16 \text{ V}$ in Figure (a) and at $V_2 \approx 13 \text{ V}$ in Figure (c) are caused by the suboptimal modulation scheme: at these port voltages V_2 , P_{out} migrates from values less than $P_{\text{out},\triangle,a,\max}$ (extended triangular current mode modulation, i.e. the efficiency is slightly below the maximum achievable efficiency) to values greater than $P_{\text{out},\triangle,a,\max}$ [operation with almost maximum achievable efficiency, cf. Section 5.2.2 and Figure 5.28 (a)].

Circuit No.:	1	2	3	4		
DC-Link Voltage or DC-Link Voltage Range (according to Table 7.1):						
$V_{\rm i}$	$225\mathrm{V}$	$150 \dots 225 V$	$475\mathrm{V}$	$325 \dots 475 \mathrm{V}$		
Design Results for the DAB:						
L	11.0 µH	$6.1\mu\mathrm{H}$	$50.0 \mu H$	$36.1 \mu \mathrm{H}$		
n	13	13	28	26		
Design Results for the Buck-or-Boost Converter:						
$L_{ m DC}$	$150\mu\mathrm{H}$	$150 \mu \mathrm{H}$	$200 \mu H$	$200\mu\mathrm{H}$		
Calculated Total Average Converter Efficiencies (Appendix A.1)						
$\overline{\eta}$	93.0%	92.7%	92.7%	92.8%		

Table 7.5: Design summary for the two-stage converters depicted in Figure 7.1 and characterized in Table 7.1.

7.3.2 Efficiencies for $V_i \ge 325 \text{ V}$ [Figure 7.1 (b)]

With the converter setup depicted in Figure 7.1 (b) and constant DC link voltage ($V_i = 475$ V), the highest efficiencies are achieved for high HV port voltages (V_1 close to 450 V). At $P_{out} = 2$ kW, high converter efficiency is achieved within the defined input and output voltage ranges [Figure 7.14 (b)], due to the reduced MOSFET currents in the HV side full bridge of the DAB. At $P_{out} = 1$ kW, however, reduced efficiencies result [Figure 7.14 (a)], since the turn-on losses of the MOSFET T_{II} increase with increasing DC link voltage V_i (Section 7.2.4).

With variable DC link voltage and rated output power, $P_{\rm out} = 2 \, \rm kW$, highest efficiency is obtained for a high DC voltage V_2 [i.e. close to 16 V, Figure 7.14 (d)]; the obtained efficiency decreases with a reduced DC voltage V_2 close to 11 V, due to increasing currents in the buck-or-boost converter and in the HV side full bridge of the DAB. At $P_{\rm out} = 1 \, \rm kW$, however, the average efficiency achieved with a variable DC link voltage V_i [Figure 7.14 (c)] is higher than the average efficiency obtained with a constant V_i [Figure 7.14 (a)], since the turn-on losses of $T_{\rm II}$ and the conduction losses due to the circulating transformer currents decrease with decreasing DC link voltage V_i . Besides, according to Figures 7.14 (a) and (c), the characteristics of the efficiencies achieved with constant $V_i = V_2/D_{\rm b,max}$ is clearly different to the efficiency characteristics achieved with variable V_i .



Figure 7.14: Efficiencies achieved with the two-stage circuit variants 3 and 4 listed in Table 7.1 [Figure 7.1 (b)], $P_{\text{out}} = 1 \text{ kW}$ or $P_{\text{out}} = 2 \text{ kW}$, and power being transferred from the HV port to the LV port: (a), (b) constant $V_i = 475 \text{ V}$; (c), (d) variable $325 \text{ V} \leq V_i \leq 475 \text{ V}$. Irregular characteristics of the contour lines are due to the reasons given in the caption of Figure 7.13.

The average efficiencies achieved with $V_i \ge 325 \text{ V}$ ($\overline{\eta} = 92.7\%$ or $\overline{\eta} = 92.8\%$, Table 7.5) are slightly below the average efficiency achieved with $V_i = 225 \text{ V}$ ($\overline{\eta} = 93.0\%$). Thus, out of the converter configurations depicted in Figure 7.1 and detailed in Table 7.1, the converter setup with a constant DC link voltage of 225 V [Figure 7.1 (a)] is preferred.

7.4 Conclusion

In this Chapter, 4 different bidirectional, two-stage DC–DC converter configurations are investigated. The aim of the presented analysis is to determine the two-stage DC–DC converter with the highest average efficiency (calculated according to Appendix A.1).

The two-stage converters considered in Figure 7.1 consist of an optimized DAB converter, which provides galvanic isolation and facilitates highly efficient, bidirectional power transfer between an LV port and a HV DC link port, and a highly efficient buck-or-boost converter (Figure 7.2). Moreover, constant and variable DC link voltages are investigated (Table 7.1).

In order to facilitate the calculation of the total converter efficiency, a loss model is developed for the buck-or-boost converter. The loss model is validated using measurement results obtained from an actual hardware prototype.

At rated output power, $P_{\text{out}} = 2 \text{ kW}$, and within the specified input and output voltage ranges (Table 1.4), the distribution of the efficiencies obtained for the two-stage solutions 1 and 3 (constant DC link voltage, cf. Table 7.1) is more balanced than the efficiency distribution achieved for a single-stage DAB converter (Figure 5.31). However, the calculated average efficiencies (92.7% $\leq \overline{\eta} \leq 93.0\%$) are below the average efficiency values calculated for the singlestage DAB (93.5%). This, together with the more complex hardware setup required, renders the proposed two-stage solutions less attractive than the single-stage DAB.

Chapter 8

Conclusions and Outlook

In the presented work, a bidirectional DC–DC converter with a low voltage / high current port, specified in Table 1.4, is investigated. Based on an evaluation of 6 different suitable single-stage converter topologies, the DAB converter is judged to be most promising with respect to a high achievable converter efficiency and a high achievable power density (Chapter 2.3). Accordingly, the DAB converter is investigated in detail; the results obtained for the DAB converter are summarized in the following list.

- In Chapter 3, the working principle of the DAB converter is explained based on different simplified converter models. This includes the lossless converter model (Figure 3.2), which facilitates basic analytical investigations on the DAB converter. In this context, in Section 3.1.3, closedform analytical expressions for the modulation parameters D_1 , D_2 , and φ of an optimal modulation scheme, optimized with respect to minimum inductor RMS current, are derived. Moreover, two different refinements of the electric DAB converter model are presented: Section 3.2 investigates the impact of the conduction losses on the transformer currents and on the power transfer characteristics; Section 3.3 discusses a simplified method to consider the magnetizing current in the transformer currents.
- Chapter 4 presents a detailed loss model of the DAB converter, used to calculate the losses in the most relevant converter components, i.e. the semiconductor switches (conduction and switching losses), the transformer (winding and core losses), and the inductor (copper and core losses). The efficiencies calculated with this loss model are verified using experimental results: for phase shift modulation, an average relative error of the calculated losses of 6.0% is achieved (Table 4.2). The de-

tailed loss model of the DAB converter is needed in order to develop the efficiency optimized modulation schemes presented in Chapter 5.

• Chapter 5 details a systematic investigation of efficiency improvements achievable for the given DAB converter, obtained with the use of optimized modulation schemes. Finally, with the suboptimal modulation schemes discussed in Section 5.2.2, converter operation close to its maximum possible efficiency is achieved (Figure 5.28). However, in order to calculate modulation parameters D_1 , D_2 , and φ suitable for the experimental setup, the detailed converter loss model discussed in Chapter 4 is used. Due to the high complexity of this model, no closed-form expressions can be given for D_1 , D_2 , and φ ; the modulation parameters are calculated with the equation systems given in Section 5.2.2, using a multidimensional numerical solver.

The most critical loss mechanisms of the given DAB converter, identified in Section 5, are:

- conduction losses,
- HV side switching losses (ZVS is required), and
- LV side switching losses (large currents should be avoided during switching).

The calculated results are verified using experimental data.

• In Chapter 6, a small-signal model of the DAB and a dynamic model of the complete converter system (including the digital control platform) are derived. The calculated results are verified using experimental data; good matching between measured and calculated results is achieved. Moreover, Section 6.2.6 illustrates an effective method to derive a simplified transfer function of a given DAB converter. According to the obtained results, the time delays caused by the digital control platform mainly limit the dynamic properties of the given DAB converter. The achieved step response of the current control loop is depicted in Figure 6.14: the proposed current controller achieves a rise time $t_{\rm r} \approx 250 \,\mu{\rm s}$ and a delay time $t_{\rm d} \approx 200 \,\mu{\rm s}$ with a percentage overshoot of approximately 7%.

Chapter 7 considers 4 different circuit variants of the series connection of a DAB converter and a buck-or-boost converter (without galvanic isolation) in order to quantify possible efficiency improvements obtained with a two-stage solution. Consequently, a loss model of the buck-or-boost converter is developed to facilitate the calculation of the efficiency of the two-stage converter. However, the calculated average efficiency of the two-stage converters is less

than the average efficiency calculated for the single-stage DAB converter (twostage converter variants: $\overline{\eta} \leq 93.0\%$, single-stage DAB converter: $\overline{\eta} = 93.5\%$; $\overline{\eta}$ is calculated according to Appendix A.1; the DAB converters employ the suboptimal modulation schemes discussed in Section 5.2.2). Thus, in average, the additional losses of the buck-or-boost converter cannot outweigh the efficiency improvements achieved with the DAB converter.

8.1 Concluding Remarks Regarding Suitable Bidirectional DC–DC Converter Topologies

With the given DAB converter, the optimized modulation schemes, and the design procedure presented in Appendix A.2, high converter efficiency is achieved for a constant switching frequency, within wide voltage ranges $(240 \text{ V} \le V_1 \le 450 \text{ V} \text{ and } 11 \text{ V} \le V_2 \le 16 \text{ V})$, and within a wide power range (Figure 5.31). However, the construction of the LV side full bridge is challenging with respect to low conduction losses and low switching losses; furthermore, the LV side DC capacitor C_{DC2} is subject to large RMS currents and the calculation of applicable modulation parameters $(D_1, D_2, \text{ and } \varphi)$ is considerably complex. These difficulties can be avoided with the full bridge converter topologies with one or more DC inductors on the LV side (Section 2.2.1): the DC inductors reduce the RMS current through the DC capacitor C_{DC2} and with the modulation schemes discussed in Appendix A.5, the switches of the LV side full bridge do not actively turn off current. However, due to the large volume of the filter inductors L_{DC2} or L_{DC2a} and L_{DC2b} , the achievable power density is considered to be less than the power density that can be achieved with the DAB converter (following [39]). Furthermore, with the modulation schemes presented in Appendix A.5, the resulting efficiency depends on the direction of power transfer, i.e. the efficiency for power being transferred from the HV port to the LV port is higher than for power being transferred in the opposite direction.

Among the full bridge converter topologies with one or more DC inductors on the LV side, the current doubler circuit features the lowest transformer currents (advantageous with respect to low losses on the connecting paths between the LV port of the HF transformer and the switches) whereas the pushpull converter only requires a single DC inductor. With respect to the voltage range specified for V_2 , $11 \text{ V} \leq V_2 \leq 16 \text{ V}$, and the power rating of 2 kW, the current doubler (Figure 2.13) and the push-pull converter (Figure 2.14) will both outperform the LV side full bridge with DC inductor (Figure 2.12), due to the available semiconductor switches (Appendix D.2).

According to the specifications given in Table 1.4, a constant switching frequency $f_{\rm S}$ is required. However, if a variable switching frequency would be permitted, $f_{\rm S}$ could be used to modify the transferred power (besides D_1 , D_2 ,



Figure 8.1: Short-circuit impedances of the HF networks of the DAB and the LLC converters; $|\underline{Z}_{s,\text{DAB}}|$ considers $L = 21.7 \,\mu H$ (determined in case of optimal modulation parameters being used, cf. Appendix A.2.1); $|\underline{Z}_{s,\text{LLC}}|$ considers $L = 38.3 \,\mu\text{H}$ and $C = 135 \,\text{nF}$ (optimal modulation parameters, cf. Appendix A.4.4).

and φ). The absolute value of the short-circuit impedance of the HF network of the LLC converter,

$$|\underline{Z}_{s,LLC}(\omega)| = \frac{1 - \omega^2 LC}{\omega C},$$
(8.1)

varies more than the short-circuit impedance of the HF network of the DAB converter (Figure 8.1),

$$|\underline{Z}_{s,\text{DAB}}(\omega)| = \omega L, \qquad (8.2)$$

if two different frequencies ω_1 and ω_2 and operation above resonance of the LLC converter are regarded:

$$|\underline{Z}_{s,\text{DAB}}(\omega_2)/\underline{Z}_{s,\text{DAB}}(\omega_1)| < |\underline{Z}_{s,\text{LLC}}(\omega_2)/\underline{Z}_{s,\text{LLC}}(\omega_1)| \forall (\omega_1 > \frac{1}{\sqrt{LC}}) \lor (\omega_2 > \omega_1).$$
(8.3)

According to (3.13), $|\underline{Z}_{s,\text{DAB}}(\omega)|$ limits the maximum output power of the DAB converter and according to (16) in [129], $|\underline{Z}_{s,LLC}(\omega)|$ limits the maximum possible output power of the LLC converter.¹ Thus, for the LLC converter, a given increase of the switching frequency results in a larger reduction of the maximum output power than for the DAB converter (this is particularly true if the LLC converter is operated close to the resonance frequency of $Z_{\rm s,LLC}$). Figure 8.1 depicts the short-circuit impedances $|\underline{Z}_{s,DAB}(\omega)|$ and $|\underline{Z}_{s,LLC}(\omega)|$ for the DAB converter with $L = 21.7 \,\mu H$ (Appendix A.2.1) and for the LLC converter with $L = 38.3 \,\mu\text{H}$ and $C = 135 \,\text{nF}$ (Appendix A.4.4): for $\omega_1 =$ $2\pi \cdot 100 \cdot 10^3 \,\mathrm{s}^{-1} \text{ and } \omega_2 = 2\pi \cdot 200 \cdot 10^3 \,\mathrm{s}^{-1}, \ |\underline{Z}_{\mathrm{s,DAB}}(\omega_2)/\underline{Z}_{\mathrm{s,DAB}}(\omega_1)| = 2.0$ and $|\underline{Z}_{\mathrm{s,LLC}}(\omega_2)/\underline{Z}_{\mathrm{s,LLC}}(\omega_1)| = 3.44$ result, i.e. due to the increased switching frequency, the maximum output power of the DAB converter reduces by a factor of 2.0 and the maximum output power of the LLC converter reduces by a factor of 3.44. Thus, with variable switching frequency, a more effective operation of the LLC converter can be achieved than of the DAB converter; consequently, the achievable efficiency is higher for the LLC converter than for the DAB converter [130].

8.2 Concluding Remarks Regarding Suitable Unidirectional DC–DC Converters (HV \rightarrow LV)

If solely unidirectional power transfer capability is required, the output switches of the DAB topology (T_5 , T_6 , T_7 , and T_8) could be replaced by diodes in order to reduce the number of gate drivers (phase shifted DC–DC converter, [54]); the modulation schemes discussed in [131] could be used to control the output power (at low power levels, the employed modulation scheme is similar to the triangular current mode modulation scheme presented in Section 3.1.3). However, the RMS currents in the transformer, in the switches, and in the DC capacitors are larger than those of the DAB converter. With the output switches of the DAB converter being replaced by diodes, the achievable power density is less than the power density that can be achieved with the DAB converter [54].

If a variable switching frequency is permitted, a resonant LLC or LCC converter requires a lower variation of the switching frequency than the DAB converter in order to achieve the same change of the output power (Section 8.1). Thus, with variable switching frequency, a higher power density may be achieved for a resonant converter, than for the DAB.

A reduced RMS current through C_{DC2} is achieved with a DC–DC converter with output inductor. With respect to the specification given in Table 1.4,

¹The parallel inductor $L_{\rm M}$, not considered in [129], only causes an increase of the reactive power provided by $v_{\rm AC2}$ and does not affect the transferred power.

the current doubler converter or the push-pull converter are most promising (cf. Section 8.1 and Appendix D). Due to the volume of the employed DC inductor, however, the achievable power density decreases [39, 132].

8.3 Concluding Remarks Regarding Suitable Unidirectional DC–DC Converters (LV \rightarrow HV)

If power is solely transferred to the HV port, again a DAB converter or a phase shifted DC–DC converter [54, 131] could be used, whereas a higher power density can be achieved with the DAB converter [54]. Even higher power density may be achieved with a resonant LLC or LCC converter, if a variable switching frequency is permitted.

Again, difficulties due to large capacitor RMS currents occur on the LV side. However, different to Section 8.2, a large capacitor current through $C_{\rm DC2}$ results, if a converter topology with one or more DC inductors on the output side (i.e. on the HV side) is selected. A reduction of the respective capacitor current can be achieved with a topology with one or more DC inductors on the input side. According to the specification given in Table 1.4, either a current doubler converter or a push-pull converter, operated in reverse direction, are considered to be most suitable. Reliable converter operation is either achieved with the modulation scheme presented in Appendix A.5, with a passive snubber, or an active clamping circuit [57, 133, 134, 135].²

8.4 Outlook

Future research on the investigated DAB DC–DC converter may be related to the DAB converter modelling and optimization, the DAB converter hardware, or the dynamic behavior of the DAB converter. Besides, the suitability of the DAB converter as a bidirectional AC–DC converter could be investigated, including a comparison to existing bidirectional AC–DC converters (for AC– DC operation, the AC side full bridge of the DAB needs to be equipped with 4-quadrant switches to allow for the operation with positive and negative voltages at the AC port).

8.4.1 DAB Converter Modelling and Optimization

The loss model presented in Chapter 4 neglects the impact of raised component temperatures on the losses. Thus, in a future step, the discussed

²The modulation scheme presented in Appendix A.5 requires semiconductor switches on the HV side. With a passive snubber or an active clamping circuit, the semiconductor switches on the HV side (i.e. T_1 , T_2 , T_3 , and T_4 in Figure A.38) can be replaced by diodes.

model needs to be extended with a coupled electro-thermal converter model, which allows to cover the effect of raised component temperatures on the efficiency [104, 105].

Future investigation may as well consider a complete converter optimization with respect to maximum power density and / or maximum efficiency, using the suboptimal modulation schemes developed in Section 5.2.2 and the design procedure discussed in Appendix A.2 (i.e. calculation of n and L). Therefore, volume models of all converter components and thermal models of the converter, the HF transformer, the HF inductor L, and the DC capacitor $C_{\rm DC2}$ need to be developed besides the loss model detailed in Chapter 4. In this context, the optimization of the LV side full bridge with respect to low volume needs to be necessarily addressed, since the volume of the currently realized LV side full bridge is large volume of the LV side full bridge including gate driver circuits and DC filter capacitor: $9.0 \text{ cm} \cdot 9.2 \text{ cm} \cdot 5.3 \text{ cm} = 439 \text{ cm}^3$, volume of the HV side full bridge including gate driver circuits and DC filter capacitor: $9.0 \text{ cm} \cdot 7.5 \text{ cm} \cdot 2.5 \text{ cm} = 169 \text{ cm}^3$]. In doing so, low HF resistances of the PCB, a low HF resistance of the transformer contacts, and low parasitic inductances in series to the semiconductor switches need to be maintained in order to achieve low conduction losses (Section 4.2 and Appendix A.2.3) and low switching losses [cf. (4.7) and Section 4.3.3].

The presented investigation only considers a single DAB converter module. A reduction of the currents in a single converter module would be achieved if multiple modules were used. Moreover, the interleaved operation of multiple converter modules enables a considerable reduction of the DC capacitor currents [136]. Thus, future research may detail the effect of the use of multiple converter modules on the total converter efficiency and / or power density. Furthermore, new possibilities due to the use of multiple converter modules arise and could be analyzed (e.g. at low power operation increased efficiency can be achieved by turning off single modules [137]).

Future research may also address the detailed calculation of the HF losses on the PCB. According to Appendix A.2.3 the switching operations cause spectral components of the currents in the PCB tracks which depend on the position on the PCB: the fundamental frequency component is mainly present at the transformer connection whereas a large DC current and increased harmonic components are observed at the connecting pads of the MOSFETs. Therefore, the accurate calculation of the HF losses on the PCB needs to consider this effect caused by the switching operations.

Furthermore, continuative refinements of the electric DAB model could be investigated, e.g. the impact of non-constant DC voltages V_1 and V_2 on the transformer currents i_{AC1} and i_{AC2} could be considered. One possible solution would be the use of an electric circuit simulator, however, it is difficult to realiably calculate the steady state currents and voltages of the simulated circuit. Moreover, the calculation time needed for an electric circuit simulation is considerably larger than the calculation time needed to evaluate the simplified model presented in Chapter 3.

8.4.2 Possible Improvements of the DAB Converter Hardware

In future, different steps could be undertaken to improve the hardware prototype designed in Appendix A.2. Suggestions for potential hardware improvements are listed below.

- An evaluation of different transformer concepts could be carried out, e.g. the suitability of different transformer cores with respect to the high current operation and a comparison of the efficiencies and / or power densities achievable with different transformer cores (e.g. planar cores, E-cores) may be analyzed. Furthermore, the impact of the integration of the DAB converter inductance L into the HF transformer on the achievable converter efficiency and / or power density could be investigated.
- The impact of component tolerances on the converter's efficiency and on the reliable converter operation may be analyzed.
- The expected life cycle of the low voltage DC capacitor C_{DC2} needs to be investigated, since C_{DC2} is subject to large RMS currents. Moreover, possible consequences of a device failure need to be considered.
- The design procedure of the EMI filters may be accomplished with respect to a certain optimum, e.g. maximum power density.

8.4.3 Future Research on the Dynamic Behavior of the DAB

The small-signal model derived in Chapter 6 considers constant port voltages V_1 and V_2 and therefore neglects the impact of power components and converters that are connected to the DAB on the dynamic converter behavior. Future investigations may consider this effect, e.g. according to [116]. The implications of component tolerances and timing inaccuracies [e.g. due to a limited time resolution of the gate signals (time resolution of the employed gate signal generation unit is 10 ns) and / or due to strongly non-linear time delays of the MOSFETs] on the dynamic converter behavior form further possibilities regarding future research on the DAB.

Appendices

Appendix A

Converter Design

A.1 Aim of the Converter Design

The objectives of the presented converter design are (in the order of the considered priorities):

- 1. converter efficiency > 90% at the nominal operating point ($V_1 = 340 \text{ V}$, $V_2 = 12 \text{ V}$, $P_{\text{out}} = \pm 2 \text{ kW}$),
- converter efficiency > 90% within reasonable input and output voltage ranges and reasonable power ranges,
- 3. low converter volume.

The selectable converter parameters (i.e. n and L for the DAB) are thus calculated in order to guarantee $\eta > 90\%$ at $V_1 = 340$ V, $V_2 = 12$ V, and $P_{\text{out}} = \pm 2$ kW. Furthermore, if possible (e.g. for the DAB), the design process maximizes the average efficiency $\overline{\eta}_{\text{design}}$,

$$\overline{\eta}_{\text{design}} = \frac{1}{36} \sum_{i=1}^{3} \sum_{j=1}^{3} \sum_{k=1}^{4} \left(\eta \Big|_{V_1 = \vec{V}_{1,i}, V_2 = \vec{V}_{2,j}, P_{\text{out}} = \vec{P}_{\text{out},k}} \right) , \qquad (A.1)$$

that considers 36 different operating points using

$$\vec{V}_1 = (240 \text{ V} \quad 340 \text{ V} \quad 450 \text{ V})^{\mathrm{T}},$$
 (A.2)

$$\vec{V}_2 = (11 \text{ V} \ 12 \text{ V} \ 16 \text{ V})^{\mathrm{T}},$$
 (A.3)

$$\vec{P}_{\text{out}} = \begin{pmatrix} -2\,\mathrm{kW} & -1\,\mathrm{kW} & 1\,\mathrm{kW} & 2\,\mathrm{kW} \end{pmatrix}^{\mathrm{T}}.$$
 (A.4)

Equation (A.4) is defined in order to consider the operation at different output power levels (rated power and half of the rated power) in either direction of power transfer. Moreover, with (A.2) and (A.3), the operation with different combinations of DC port voltages V_1 and V_2 is considered in order to evaluate the predicted efficiency at:¹

- the nominal operating point $(V_1 = 340 \text{ V}, V_2 = 12 \text{ V}),$
- the edges of the specified voltage ranges,
- nominal HV port voltage and minimum or maximum LV port voltage $(V_1 = 340 \text{ V}, V_2 = 11 \text{ V} \text{ and } V_1 = 340 \text{ V}, V_2 = 16 \text{ V})$, and
- nominal LV port voltage and minimum or maximum HV port voltage $(V_1 = 240 \text{ V}, V_2 = 12 \text{ V} \text{ and } V_1 = 450 \text{ V}, V_2 = 12 \text{ V}).$

Finally, the employed converter components are arranged in a way that a low converter volume is achieved.

A.2 Design of the DAB

The design of the DAB is an iterative process: in a first step, approximate values of the component stress values (i.e. maximum blocking voltages, maximum RMS currents, maximum magnetic flux densities) are calculated with the lossless DAB converter model (Section 3.1). The obtained results enable a first selection of the employed DAB converter components (e.g. semiconductor switches, transformer core). With the lossless converter model, however, an inaccurate efficiency results and thus, the presented DAB loss model (Chapter 4) is used in a second step in order to improve the calculated stress values; thereafter, the DAB converter design is accordingly refined. Due to the changed DAB converter design, however, the parameters employed in the DAB loss model change; thus, the design procedure needs to be repeated until the specified requirements are met.

In order to avoid iterations in the presented documentation, the final DAB loss model is employed in Appendix A.2.1 to determine the DAB converter parameters n and L. Based on the calculated component stress values, the DAB converter components are designed:

- HV side semiconductor switches (Appendix A.2.2),
- LV side semiconductor switches (Appendix A.2.3),
- HF transformer (Appendix A.2.4),

¹If a different evaluation criteria is specified, then (A.1) (A.2), (A.3), and (A.4) need to be modified accordingly; consequently, different values for n and L may result. The converter design procedures discussed in the subsequent sections, however, remain the same.



Figure A.1: Picture of the 2 kW DAB converter prototype ($273 \text{ mm} \times 90 \text{ mm} \times 53 \text{ mm}$).

- DAB converter inductor L (Appendix A.2.5),
- EMI filters (Appendix A.2.6).

Figure A.1 shows a picture of the final DAB hardware prototype: it contains the 2 full bridges (HV side and LV side), the HF transformer, the digital control platforms (DSP: TMS320F2808, FPGA: LCMXO2280), the auxiliary power supplies (HV side and LV side; only the HF transformer of the HV side auxiliary power supply – a flyback converter – is visible in Figure A.1), and 2 water coolers on the top and on the bottom side of the converter.

A.2.1 Calculation of *n* and *L*

With the full electric DAB converter model (Section 3.3) and the loss model discussed in Chapter 4 an accurate prediction of the converter efficiency is feasible and thus, the converter parameters n and L can be calculated with respect to maximum average efficiency, $\max(\overline{\eta}_{design})$, [cf. (A.1)].

Extending the DAB Loss Model for $n \neq 19$

The existing loss model is established for n = 19 and is not readily applicable for $n \neq 19$. Still, for $n \neq 19$, most components of the discussed DAB loss model remain unchanged (i.e. conduction losses, switching losses, transformer core losses, and copper losses of the PCB).

Since the predominant part of the converter inductance L is placed on the HV side (cf. Appendix A.2.5), the voltage v_{AC2} , generated by the LV side full bridge, mainly excites the transformer core and generates the peak flux density $B_{tr,peak}$. Therefore, the LV side winding of the HF transformer remains independent of n (i.e. $N_2 = 1$; R_{tr2} remains constant) and the HV winding is adapted: $N_1 = n N_2 = n$. In order to enable a simple calculation of R_{tr1} a constant cross section surface of the HV winding, A_{HV} , and a constant average length of a single turn, l_{single} , are assumed. Thus, R_{tr1} is scaled according to

$$R_{\rm tr1}(N_1) = f_{\rm AC} \,\rho_{\rm Cu} \,\frac{N_1 \, l_{\rm single}}{A_{\rm HV}/N_1} = N_1^2 \, f_{\rm AC} \,\rho_{\rm Cu} \frac{l_{\rm single}}{A_{\rm HV}} = N_1^2 \frac{R_{\rm tr1}(19)}{19^2} \qquad (A.5)$$

(the implication of N_1 on the AC resistance factor $f_{AC} > 1$ is neglected; cf. Appendix A.2.4).

The parameters $L_{\rm HV}$, $R_{L_{\rm HV}}$, $V_{L_{\rm HV},\rm core}$, and $N_{L_{\rm HV}}$, used to calculate the inductor conduction and core losses (Section 4.1.2), change with varying n and L. Consequently, separate inductor designs are required for each combination of n and L (Appendix A.9).

Results for n and L

In order to reduce the required calculation time, the DAB converter is designed for a discrete number of turns ratios n_i and inductances L_j (Figure A.2, Figure A.3, Figure A.4):

$$\vec{n} = (11 \ 12 \ 13 \ 14 \ 15 \ 16 \ \dots \ 23 \ 24)^{\mathrm{T}},$$

 $\vec{L} = L_{\max}(n) \cdot (1.0 \ 0.975 \ 0.95 \ 0.925 \ 0.90 \ \dots \ 0.525 \ 0.5)^{\mathrm{T}}.$ (A.6)

With phase shift modulation, maximum average efficiency, max($\overline{\eta}_{\text{design,PS}}$) = 89.5%, is achieved for n = 19 and $L = 26.7 \,\mu\text{H}$ (Figure A.2). There, the maximum inductor RMS currents are: $I_{L_1} = 15.6 \,\text{A}$ and $I_{L_2} = 294 \,\text{A}$. On the LV side, capacitor RMS currents of up to 244 A occur (Table A.1).

The achievable maximum average efficiency considerably increases with the extended triangular and trapezoidal current mode modulation schemes being employed (Figure A.3, Section 5.1): for n = 16 and $L = 15.5 \,\mu\text{H}$, $\max(\overline{\eta}_{\text{design,TT}}) = 92.6\%$ is obtained. Compared to phase shift modulation,


Figure A.2: Average DAB efficiency $\overline{\eta}_{\text{design,PS}}$, calculated with the DAB loss model (Chapter 4), obtained for phase shift modulation and different *n* and *L*. L_{max} denotes the maximum possible DAB inductance which allows for the specified output power within the specified voltage ranges. Its value depends on the turns ratio *n*; for n = 19, $L_{\text{max}} = 26.7 \,\mu\text{H}$ is calculated.

reduced maximum inductor RMS currents and reduced capacitor RMS currents are achieved, as well (Table A.1). Moreover, the HV side full bridge operates with ZVS within the complete specified operating range.²

The suboptimal modulation schemes (Section 5.2.2) allow for an even higher average efficiency of 93.5% for n = 16 and $L = 22.4 \,\mu\text{H}$ (Figure A.4) and enable a further reduction of the maximum inductor RMS currents (HV side: 15.1 A, LV side: 240 A) and a reduction of the maximum capacitor RMS currents on the LV side (138 A; cf. Table A.1). Again, ZVS operation of the HV side full bridge is achieved within the complete specified operating range.

With optimal modulation parameters (Section 5.2.1) $\max(\overline{\eta}_{\text{design,opt}}) = 93.7\%$ is achieved for n = 17 and $L = 21.7 \,\mu\text{H}$ (Figure A.5). The design with optimal modulation parameters considers a reduced number of turns

²With phase shift modulation, hard switching occurs on the HV side for certain operating points [50], e.g. for $V_1 = 240$ V, $V_2 = 16$ V, $P_{out} = 200$ W, n = 19, L = 26.7 µH, and $f_S = 100$ kHz.



 $L/L_{\text{max,TT}}$ Ext. Triangular and Trapezoidal Current Mode Modulation: $\overline{\eta}_{\text{design,TT}}$

Figure A.3: Average DAB efficiency $\overline{\eta}_{\text{design,TT}}$, calculated with the DAB loss model (Chapter 4), obtained for the extended triangular and trapezoidal current mode modulation schemes and different n and L; L_{max} denotes the maximum possible DAB inductance which allows for the specified output power within the specified voltage ranges. For n = 16, $L_{\text{max}} = 15.5 \,\mu\text{H}$ is calculated.

ratios and inductance values in order to further reduce the calculation effort:

$$\vec{n} = (14 \quad 15 \quad 16 \quad 17 \quad 18 \quad 19 \quad 20 \quad 21 \quad 22 \quad 23 \quad 24)^{\mathrm{T}}, \vec{L} = L_{\mathrm{max}}(n) \cdot (0.975 \quad 0.95 \quad 0.925 \quad 0.90 \quad \dots \quad 0.775 \quad 0.75)^{\mathrm{T}}.$$
(A.7)

The result depicted in Figure A.5 justifies the step sizes selected for \vec{n} and \vec{L} : around the maximum average efficiency of 93.7% a single step change causes $\bar{\eta}_{\text{design,opt}}$ to change by less than 0.1% [around the maximum, the maximum largest change (0.081%) occurs between n = 17 and n = 18 in Figure A.5 (a)].

A.2.2 HV Side Full Bridge

On the HV side, a full bridge with a maximum supply voltage of 450 V (Table 1.4) and with a maximum output RMS current of 15.9 A (cf. Table A.1) is needed. In order to enable a high switching frequency and



Figure A.4: Average DAB efficiency $\overline{\eta}_{\text{design,subopt}}$, calculated with the DAB loss model (Chapter 4), obtained for the suboptimal modulation schemes and different *n* and *L*; L_{max} denotes the maximum possible DAB inductance which allows for the specified output power within the specified voltage ranges; for n = 16, $L_{\text{max}} = 23.6 \,\mu\text{H}$ is calculated.



Figure A.5: Average DAB efficiency $\overline{\eta}_{\text{design,opt}}$, calculated with the DAB loss model (Chapter 4), obtained for efficiency optimal modulation parameters D_1 and D_2 and for different n and L; maximum average converter efficiency of 93.7% is achieved for n = 17 and $L = 21.7 \,\mu\text{H}$.

Property	Phase Shift Mod.	Extended Tria./Trap.	Suboptimal Mod.	Optimal Mod.		
		Mod.				
$\max(\overline{\eta}_{\mathrm{design}})$	89.5%	92.6%	93.5%	93.7%		
L	$26.7\mu\mathrm{H}$	$15.5\mu\mathrm{H}$	$22.4\mu\mathrm{H}$	$21.7\mu\mathrm{H}$		
n	19	16	16	17		
$\max(V_{\mathrm{DS},1,\mathrm{peak}})$		450	V			
$\max(V_{\mathrm{DS},2,\mathrm{peak}})$	$16 \mathrm{V}$					
$\max(I_{L_1,\text{peak}})$	$29.0\mathrm{A}$	29.1 A	24.9 A	$23.9\mathrm{A}$		
$\max(I_{L_2,\text{peak}})$	$550\mathrm{A}$	$467\mathrm{A}$	$399\mathrm{A}$	$407\mathrm{A}$		
$\max(I_{L_1})$	$15.6\mathrm{A_{RMS}}$	$15.9\mathrm{A_{RMS}}$	$15.1\mathrm{A_{RMS}}$	$13.6A_{\rm RMS}$		
$\max(I_{L_2})$	$294A_{\rm RMS}$	$253A_{\rm RMS}$	$240A_{\rm RMS}$	$231A_{\rm RMS}$		
$\max(I_{C_{\mathrm{f1}}})$	$14.9A_{\rm RMS}$	$8.4A_{\rm RMS}$	$12.1A_{\rm RMS}$	$9.2A_{\rm RMS}$		
$\max(I_{C_{\mathbf{f}2}})$	$244A_{\rm RMS}$	$153\mathrm{A_{RMS}}$	$138A_{\rm RMS}$	$125A_{\rm RMS}$		
$\max(B_{\rm tr,peak})$	$142\mathrm{mT}$	$126\mathrm{mT}$	$129\mathrm{mT}$	$130\mathrm{mT}$		

Table A.1: DAB inductances, transformer turns ratios, and DAB stress values calculated for the different modulation schemes; n and L are selected with respect to maximum $\overline{\eta}_{\text{design}}$ [cf. (A.1)].

simultaneously achieve low losses, the SPW47N60CFD (CoolMOS) has been selected due to its low on-state resistance [cf. (4.5)], fast switching, and the enhanced and robust body diode. With its non-linear characteristic of the parasitic drain-to-source capacitance the proposed device is well suited for converters operating with ZVS.

A.2.3 LV Side Full Bridge

The construction of the LV side full bridge is considerably more challenging than the construction of the HV side full bridge, due to the high RMS values and the high frequency (100 kHz, Table 1.4) of the full bridge output currents.

The DAB hardware prototype therefore employs 2 full bridges which are operated in parallel [Figure A.6 (a)], such that the current rating of a single full bridge is cut into half and thus, a converter realization using printed circuit boards becomes feasible.

In order to achieve good current distribution at high frequencies, the component arrangement depicted in Figure A.7 is employed (considering 4 MOS-FETs per switch on a single LV full bridge, cf. "Selecting the LV MOSFETs" on p. 289). There, the low side MOSFETs (T_6, T_8) are placed on the inside and the high side MOSFETs (T_5, T_7) on the outside. This arrangement splits the full bridge up into 2 circuits and introduces a second instance of parallelization. The 2 output tracks of the bridge and the solder pads connecting to the transformer are in layers on top of each other in order to achieve a low inductive conduction path to the transformer and a low resistive transformer contact. The current flow paths related to the different switching states of the LV side full bridge are shown in Figure A.8, Figure A.9, Figure A.10, and Figure A.11. Two layers of the PCB are used for the output tracks of the full bridge and one layer forms the minus connection of the LV port. The remaining layer of the PCB (layer 1), however, is needed for the wiring of the gate driver circuits. Therefore, the bus bar used for the plus connection of the LV port (Figure A.7) is used as a fifth layer and forms a solid plane below the PCB.

FEM simulation has been used to determine the current distribution among the MOSFETs for the different switching states of the LV side full bridge shown in Figure A.6 (b). For this purpose a sinusoidal current with a frequency of $f = f_{\rm S} = 100 \, \rm kHz$ has been employed. The obtained simulation results give a first insight into the current distribution on the PCB, however, the impact of the switching operations on the current distribution is not accounted for: Figure A.12 shows a typical current waveform of the DAB and the related spectrum of the transformer current; Figure A.13 and Figure A.14 show the respective currents in T_5 and T_7 including the related spectra. Obviously, a distinct fundamental frequency component occurs for the transformer current, whereas a DC component and considerably large harmonic components result for the switch currents. Thus, due to the switching operations, different spectra of the PCB currents occur (depending on the position on the PCB), which is not accounted for in the presented approach. Still, the results obtained from the FEM simulations allow for an estimation of the expected current distributions among the MOSFETs, since a distinct fundamental frequency component results due to the fact that each MOSFET conducts for one half-cycle in steady-state operation.

Figure A.15 (a) shows the FEM simulation results of the RMS current distribution for a current with an RMS value of 100 A in the output conductor of the half bridge leg formed by T_5 and T_6 with the switches T_5 and T_8



Figure A.6: (a) The final converter prototype employs two full bridges on the LV side, which are operated in parallel; (b) equivalent circuits for the different switching states of a single LV side full bridge, used to determine the current distributions given in Table A.2 using FEM simulations (for high frequency excitations the DC capacitor can be replaced by a shorting bar).



Figure A.7: Proposed arrangement of the power electronic components of a single LV full bridge employing 4 MOSFETs per switch.



Figure A.8: Current paths in the LV full bridge if T_5 and T_8 are turned on and T_6 and T_7 are turned off.



Figure A.9: Current paths in the LV full bridge if T_6 and T_7 are turned on and T_5 and T_8 are turned off.



Figure A.10: Current paths in the LV full bridge if T_5 and T_7 are turned on and T_6 and T_8 are turned off.



Figure A.11: Current paths in the LV full bridge if T_6 and T_8 are turned on and T_5 and T_7 are turned off.



Figure A.12: (a) Current waveform $i_{AC2}(t)$ in the LV side transformer winding of the DAB, calculated for $V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW, $D_1 = 0.3$, $D_2 = 0.4$, L = 22.4 µH, and $f_S = 100$ kHz; (b) frequency spectrum calculated for $i_{AC2}(t)$.



Figure A.13: (a) Waveform of the current through T₅, calculated for $V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW, $D_1 = 0.3$, $D_2 = 0.4$, L = 22.4 µH, and $f_S = 100$ kHz; (b) frequency spectrum calculated for $i_{D,T_5}(t)$.



Figure A.14: (a) Waveform of the current through T₇, calculated for $V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW, $D_1 = 0.3$, $D_2 = 0.4$, L = 22.4 µH, and $f_S = 100$ kHz; (b) frequency spectrum calculated for $i_{D,T_7}(t)$.

Results for closed switches T_5 , T_8 and open switches T_6 , T_7 :								
$I_{\mathrm{T}_{5,\mathrm{A}}}$	$I_{\mathrm{T}_{5,\mathrm{B}}}$	$I_{\mathrm{T}_{5,\mathrm{C}}}$	$I_{\mathrm{T}_{5,\mathrm{D}}}$	$I_{\mathrm{T}_{8,\mathrm{A}}}$	$I_{\rm T_{8,B}}$	$I_{\rm T_{8,C}}$	$I_{\rm T_{8,C}}$	
$25.8\mathrm{A}$	$24.3\mathrm{A}$	$25.8\mathrm{A}$	$24.3\mathrm{A}$	$24.9\mathrm{A}$	$25.1\mathrm{A}$	$24.9\mathrm{A}$	$25.1\mathrm{A}$	
(+3.0%)	(-3.0%)	(+3.0%)	(-3.0%)	(-0.3%)	(+0.3%)	(-0.3%)	(+0.3%)	
Results f	Results for closed switches T_6 , T_7 and open switches T_5 , T_8 :							
$I_{T_{6,A}}$	$I_{\rm T_{6,B}}$	$I_{T_{6,C}}$	$I_{T_{6,D}}$	$I_{\mathrm{T}_{7,\mathrm{A}}}$	$I_{\rm T_{7,B}}$	$I_{\rm T_{7,C}}$	$I_{\mathrm{T}_{7,\mathrm{D}}}$	
$26.1\mathrm{A}$	$24.0\mathrm{A}$	$26.1\mathrm{A}$	$24.0\mathrm{A}$	$24.5\mathrm{A}$	$25.6\mathrm{A}$	$24.5\mathrm{A}$	$25.6\mathrm{A}$	
(+4.2%)	(-4.2%)	(+4.3%)	(-4.2%)	(-2.2%)	(+2.2%)	(-2.3%)	(+2.3%)	
Results for closed switches T_5 , T_7 and open switches T_6 , T_8 :								
Results f	or closed	switches '	T_5, T_7 ai	nd open s	witches T ₆	3, T ₈ :		
Results f $I_{T_{5,A}}$	or closed $I_{T_{5,B}}$	switches ' $I_{T_{5,C}}$	$\mathrm{T}_5,\mathrm{T}_7$ as $I_{\mathrm{T}_{5,\mathrm{D}}}$	nd open sv $I_{\mathrm{T}_{7,\mathrm{A}}}$	witches T_{6} $I_{T_{7,B}}$	з, Т ₈ : I _{Т7,С}	$I_{\mathrm{T}_{7,\mathrm{D}}}$	
Results f $I_{T_{5,A}}$ 26.3 A	for closed $I_{T_{5,B}}$ 23.7 A	switches $I_{T_{5,C}}$ 26.3 A	${f T_5,T_7}$ an $I_{T_{5,D}}$ 23.7 A	nd open sv $I_{\mathrm{T}_{7,\mathrm{A}}}$ 25.1 A	witches T_{6} $I_{T_{7,B}}$ 24.9 A	5, T ₈ : $I_{T_{7,C}}$ 25.1 A	$I_{\rm T_{7,D}}$ 24.9 A	
Results f $I_{T_{5,A}}$ 26.3 A (+5.3%)	or closed $I_{T_{5,B}}$ 23.7 A (-5.3%)	switches ' $I_{\rm T_{5,C}}$ 26.3 A (+5.3%)	${f T_5,T_7}$ an ${I_{T_{5,D}}\over 23.7A}$ (-5.3%)	nd open sv $I_{\rm T_{7,A}}$ 25.1 A (+0.5%)	witches T_{6} $I_{T_{7,B}}$ 24.9 A (-0.5%)	${}_{3}, { m T_8:} \ {I_{{ m T}_{7,{ m C}}}} \ {25.1 { m A}} \ (+0.5\%)$	$I_{\rm T_{7,D}}$ 24.9 A (-0.5%)	
$\begin{array}{c} \text{Results f} \\ I_{\mathrm{T}_{5,\mathrm{A}}} \\ 26.3 \mathrm{A} \\ (+5.3\%) \\ \text{Results f} \end{array}$	or closed $I_{T_{5,B}}$ 23.7 A (-5.3%) or closed	switches ' $I_{T_{5,C}}$ 26.3 A (+5.3%) switches '	$egin{array}{c} { m T}_5,{ m T}_7{ m an}\ I_{{ m T}_5,{ m D}}\ 23.7{ m A}\ (-5.3\%)\ { m T}_6,{ m T}_8{ m an} \end{array}$	nd open sv $I_{\mathrm{T}_{7,\mathrm{A}}}$ 25.1 A (+0.5%) nd open sv	witches Te $I_{T_{7,B}}$ 24.9 A (-0.5%) witches T ₅	$\begin{array}{c} \text{5, T_8:} \\ I_{\text{T}_{7,\text{C}}} \\ \hline 25.1 \text{ A} \\ (+0.5\%) \\ \hline \text{5, T_7:} \end{array}$	$\frac{I_{\rm T_{7,D}}}{24.9\rm A}$ (-0.5%)	
$ Results f I_{T_{5,A}} 26.3 A(+5.3%)Results fI_{T_{6,A}} $	for closed $I_{T_{5,B}}$ 23.7 A (-5.3%) for closed $I_{T_{6,B}}$	switches $^{\prime}$ $I_{T_{5,C}}$ 26.3 A (+5.3%) switches $^{\prime}$ $I_{T_{6,C}}$	T_5, T_7 an $I_{T_{5,D}}$ 23.7 A (-5.3%) T_6, T_8 an $I_{T_{6,D}}$	Ind open system $I_{T_{7,A}}$ $I_{T_{7,A}}$ $I_{25.1 A}$ (+0.5%) I_{0} $I_{T_{8,A}}$	witches T_{e} $I_{T_{7,B}}$ 24.9 A (-0.5%) witches T_{5} $I_{T_{8,B}}$	$\begin{array}{c} _{5}, \mathrm{T}_{8} \\ I_{\mathrm{T}_{7,\mathrm{C}}} \\ 25.1 \mathrm{A} \\ (+0.5\%) \\ _{5}, \mathrm{T}_{7} \\ I_{\mathrm{T}_{8,\mathrm{C}}} \end{array}$	$I_{\rm T_{7,D}}$ 24.9 A (-0.5%) $I_{\rm T_{8,C}}$	
$\begin{array}{c} \text{Results f} \\ I_{\text{T}_{5,A}} \\ \hline 26.3 \text{ A} \\ (+5.3\%) \\ \text{Results f} \\ I_{\text{T}_{6,A}} \\ \hline 26.3 \text{ A} \end{array}$	or closed $I_{T_{5,B}}$ 23.7 A (-5.3%) or closed $I_{T_{6,B}}$ 23.7 A		${f T_5, T_7}$ an ${I_{T_5,D}}$ 23.7 A (-5.3%) ${f T_6, T_8}$ an ${I_{T_6,D}}$ 23.7 A	$\begin{array}{c} \text{nd open s}\\ I_{\text{T}_{7,\text{A}}}\\ 25.1 \text{ A}\\ (+0.5\%)\\ \text{nd open s}\\ I_{\text{T}_{8,\text{A}}}\\ 25.2 \text{ A} \end{array}$	witches T_{e} $I_{T_{7,B}}$ 24.9 A (-0.5%) witches T_{5} $I_{T_{8,B}}$ 24.8 A	$\begin{array}{c} {}_{5},{\rm T_8}\colon\\ I_{{\rm T_{7,C}}}\\ {}_{25.1{\rm A}}\\ (+0.5\%)\\ {}_{5},{\rm T_7}\colon\\ I_{{\rm T_{8,C}}}\\ {}_{25.2{\rm A}}\end{array}$	$\frac{I_{\rm T_{7,D}}}{24.9\rm A}$ (-0.5%) $\frac{I_{\rm T_{8,C}}}{24.8\rm A}$	

Table A.2: MOSFET currents obtained from a FEM simulation of a single LV side full bridge PCB; the percentage values in parentheses denote the respective relative current deviation from the average value. For the presented results, the 4 different situations shown in Figure A.6 (b) are considered. The simulation is carried out with a RMS current of 100 A and f = 100 kHz.

being closed and T_6 and T_7 being open. With the proposed PCB layout, a good current distribution among the MOSFETs is achieved: according to Table A.2, the current deviation from the average current value is less than $\pm 6\%$. Furthermore, the transformer termination needs to be designed carefully: for an RMS current of 100 A and with the side-by-side arrangement depicted in Figure A.15 (b), a large hot spot with exceedingly high current densities of more than 100 A/mm² (RMS value) occurs. A considerably better current distribution is achieved at the transformer terminations if the respective solder pads are arranged on top of each other [Figure A.15 (a)].

Due to the high current stress, a 4 layer PCB with $200 \,\mu\text{m}$ copper on each layer is employed to obtain low resistive losses. The copper resistances of one



Figure A.15: (a) FEM simulation of the current distribution on a single LV full bridge using an AC RMS current of 100 A with a frequency of 100 kHz (the depicted current densities are RMS values); switches T_5 and T_8 conduct, switches T_6 and T_7 are off [cf. Figure A.6 (b) and Figure A.7]; in (a) the transformer terminations are arranged on top of each other; (b) simulated current distribution at the transformer terminations with a side-by-side arrangement of the respective solder pads: there, a large hot spot with current densities of more than 100 A/mm² occurs.

PCB is measured with an impedance analyzer (cf. Section 4.2 and Figure 4.3):

$$2R_{\rm PCB,a}(f = 100 \,\text{kHz}) = 520 \,\mu\Omega,$$

$$2R_{\rm PCB,b}(f = 0) = 240 \,\mu\Omega,$$

$$2R_{\rm PCB,b}(f = 200 \,\text{kHz}) = 580 \,\mu\Omega.$$

(A.8)

Two full bridges are operated in parallel on the LV side. Therefore, the effective resistance values $R_{\text{PCB},a}$ and $R_{\text{PCB},b}$ are half of the resistance values given above and the values depicted in Figure 4.3 are half of the resistance values measured for a single PCB.

Large currents also occur in the full bridge filter capacitor C_{f2} , therefore ceramic capacitors have been selected as they are superior to film capacitors with respect to energy density and to electrolytic capacitors with respect to ripple currents [138]. The DC capacitor C_{f2} is formed by the parallel connection of 96 ceramic capacitors, each with a value of $10 \,\mu\text{F}$, and this keeps the peak-to-peak voltage ripple below $\approx 600 \,\text{mV}$. In the case of ideal current distribution, a maximum RMS ripple current of 2.5 A flows through each capacitor which causes less than 25 mW per capacitor (calculated with a dissipation factor of 5%) and is well below the specified maximum RMS value of 6 A [154].

Selecting the LV MOSFETs

On the LV side, the maximum blocking voltage of each switch is 16 V; however, due to overvoltage spikes during switching (cf. Section 4.3.3), MOSFETs with a rated voltage of 40 V are selected. For the LV side full bridge, the MOSFET IRF2804 is selected due to its low on-state resistance of $2.2 \text{ m}\Omega$. Moreover, each switch on the single LV side full bridge is composed of a certain number of MOSFETs connected in parallel in order to achieve low conduction losses and low switching losses. In order to achieve a symmetric arrangement of the MOSFETs according to Figure A.7, a multiple of 4 MOSFETs needs to be employed. The number of required MOSFETs per switch, $N_{\rm IRF2804}$, is determined based on the efficiency calculated for different values for $N_{\rm IRF2804}$. This calculation employs the loss model discussed in Chapter 4 and considers the scaling laws listed below:

- the MOSFET currents are reciprocally proportional to the number of employed MOSFETs;
- the switching losses for a given number of MOSFETs per switch are evaluated based on the measured switching losses (Section 4.3.3) and

scaled according to

$$E_{\rm S2,sw,scaled}\left(I_{\rm S2,sw}, N_{\rm IRF2804}\right) = \frac{N_{\rm IRF2804}}{8} E_{\rm S2,sw}\left(\frac{8}{N_{\rm IRF2804}} I_{\rm S2,sw}\right);$$
(A.9)

- the leakage inductance of each LV switch (T₅, T₆, T₇, and T₈ in Figure 3.1) is reciprocally proportional to the number of employed MOSFETs per switch; consequently, the durations of the overvoltage spikes discussed in Section 4.3.3 are reciprocally proportional to the number of employed MOSFETs;
- the gate driver power demand scales proportional to the number of employed MOSFETs (Section 4.3.4).

In Figure A.16 and Figure A.17, the efficiencies calculated for $N_{\rm IRF2804} = 4$ and $N_{\rm IRF2804} = 8$ and different MOSFET junction temperatures, $T_{\rm j}$, are depicted (for the HV side MOSFETs and for the LV side MOSFETs the same junction temperatures are assumed: $T_{\rm j,HV} = T_{\rm j,LV} = T_{\rm j}$), whereas optimal transformer and inductor designs are used (i.e. n and L are adapted depending on $N_{\rm IRF2804}$ and $T_{\rm j}$). At elevated temperatures, the conduction losses of the MOSFETs increase considerably. Accordingly, with $N_{\rm IRF2804} = 4$ and $T_{\rm j} = 125^{\circ}$ C, the efficiency calculated at low port voltages and at rated output power drops below 84% [Figure A.17 (b)]. Thus, $N_{\rm IRF2804} = 8$ is selected in order to maintain a high efficiency of more than 90% at elevated temperatures (e.g. at $T_{\rm j} = 125^{\circ}$ C) and within a wide operating range.

A.2.4 Transformer

A planar core (E 58: 58 mm \times 38 mm \times 11 mm)³ is employed for the power transformer, mainly because of its advantageous properties with respect to high power density and excellent electromagnetic and thermal characteristics [105]. Its height (21 mm) is significantly lower than the total height of the converter (50 mm) what leaves space for a surrounding heat sink. For the selected ferrite core (material 3F3) a maximum peak flux density of 142 mT is calculated for phase shift modulation (Table A.1). The large safety margin between 142 mT and the saturation level of more than 300 mT has been selected to achieve low core losses and to prevent transformer saturation, since no DC coupling capacitors are implemented. DC coupling capacitors would show a significant volume, especially on the low voltage side due to the high current ratings.

 $^{^{3}}$ Since the ELP 58 core (which was initially considered) was not available, the E 58 core was used, instead. However, the dimensions of the ELP 58 and the E 58 cores are nearly identical, cf. [155, 156].



Figure A.16: Calculated efficiencies for an output power of 1 kW, a power transfer from the HV port to the LV port, different junction temperatures $T_{\rm j}$ (equal $T_{\rm j}$ is assumed for all MOSFETs), and for different numbers of parallel MOSFETs used per switch on the LV side: (a) $N_{\rm IRF2804} = 4$, $T_{\rm j} = 25^{\circ}$ C, (b) $N_{\rm IRF2804} = 4$, $T_{\rm j} = 125^{\circ}$ C, (c) $N_{\rm IRF2804} = 8$, $T_{\rm j} = 25^{\circ}$ C, (a) $N_{\rm IRF2804} = 8$, $T_{\rm j} = 125^{\circ}$ C; *n* and *L* of each configuration are optimized with respect to maximum average efficiency.

Typically, flat and wide windings are used in planar core transformers to achieve a low AC resistance. Hence, a considerably large parasitic coupling capacitance results between the primary side (HV side) and the secondary side (LV side) of the HF transformer if the transformer comprises of a large number of interleaved HV and LV windings that are placed on top of each other. In order to achieve a small coupling capacitance and a low AC resis-



Figure A.17: Calculated efficiencies for an output power of 2 kW, a power transfer from the HV port to the LV port, different junction temperatures $T_{\rm j}$ (equal $T_{\rm j}$ is assumed for all MOSFETs), and for different numbers of parallel MOSFETs used per switch on the LV side: (a) $N_{\rm IRF2804} = 4$, $T_{\rm j} = 25^{\circ}$ C, (b) $N_{\rm IRF2804} = 4$, $T_{\rm j} = 125^{\circ}$ C, (c) $N_{\rm IRF2804} = 8$, $T_{\rm j} = 25^{\circ}$ C, (a) $N_{\rm IRF2804} = 8$, $T_{\rm j} = 125^{\circ}$ C; *n* and *L* of each configuration are optimized with respect to maximum average efficiency.

tance, litz wire windings and an interleaved winding arrangement with a low number of alternating winding layers are employed: the HV winding with either 16 or 19 turns is placed in the middle and 2 single turn LV windings, which are operated in parallel, are placed above and below the HV winding (Figure A.18). Table A.3 lists the geometric dimensions of the HF transformer core and Table A.4 lists the employed litz wires, which are optimized accord-



Figure A.18: Planar HF transformer employed for the DAB. The transformer contains 2 windings on the LV side, one above and one below the HV winding, which are operated in parallel. The HV winding is placed in the middle.

ing to [139] in order to achieve a minimum high frequency resistance [140,141].

On the LV side, the winding terminations need to be designed with care in order to avoid high HF losses. The winding terminations depicted in Figure A.19 (a) (i.e. side-by-side arrangements of the wire ends) cause an in-

Property	Value	Description
$b_{\rm tr,core}$	$58.4\mathrm{mm}$	core width
$h_{ m tr,core}$	$21.1\mathrm{mm}$	core height
$l_{ m tr,core}$	$38.1\mathrm{mm}$	core depth
$p_{ m tr,c}$	$8.1\mathrm{mm}$	width of the inner core leg
$b_{ m tr,c}$	$21.5\mathrm{mm}$	width of the window area
$p_{ m tr,s}$	$3.7\mathrm{mm}$	width of each outer core leg

Table A.3: Dimensions of the HF transformer core ELP 58 (according to Figure A.18, [155]).



Figure A.19: (a) Side-by-side arrangement of the LV side transformer winding termination; (b) LV side winding ends being placed on top of each other (the HF transformer employs 2 windings on the LV side, which are operated in parallel). The shown copper bars are used to short circuit the LV port of the transformer in order to enable the measurement of the stray inductance and the AC resistance of the winding; the copper bars are removed once the measurement is done. According to the measurement results depicted in Figure A.20, the total AC resistance, $R_{tr1} + n^2 R_{tr2}$, achieved with arrangement (b) is 26% less than the AC resistance achieved with (a).



Figure A.20: Measured HV side referred AC resistance, $R_{\rm tr1,19} + n^2 R_{\rm tr2,19}$, of the HF transformer with n = 19 and the 2 winding terminations depicted in Figure A.19 (using the transformer model depicted in Figure 3.29). At f = 100 kHz, a total AC resistance of either 208 m Ω (side-by-side arrangement) or 153 m Ω (winding ends on top of each other) is measured. The measurement is accomplished using the Agilent 4294A Precision Impedance Analyzer. At f = 100 kHz, the accuracies of the Impedance Analyzer are $\pm 3.8\%$ for $R_{\rm tr1,19} + n^2 R_{\rm tr2,19} = 208$ m Ω and $\pm 4.1\%$ for $R_{\rm tr1,19} + n^2 R_{\rm tr2,19} = 153$ m Ω (calculated according to [152]).

crease of the total stray inductance [142] and generate additional HF losses (cf. Appendix A.2.3). Figure A.20 depicts the measured HV side referred copper resistance of the HF transformer with n = 19, using the side-by-side arrangements shown in Figure A.19 (a) and 2 shorting bars being soldered to the wire ends of the 2 transformer windings on the LV side: at f = 100 kHz, a total AC resistance of 208 m Ω results.

Winding	Employed Litz Wire
LV windings	9 parallel connected litz wires: 245 strands, single strand copper diameter: 0.1 mm
HV winding, $n = 16$	parallel connected litz wires: $245 \times 0.1 \text{ mm}$ and $120 \times 0.071 \text{ mm}$
HV winding, $n = 19$	$245\times0.1\mathrm{mm}$

Table A.4: Litz wires employed for the HF transformer windings.

The HF resistances of the transformer terminations clearly decrease if the winding terminations are placed on top of each other [Figure A.19 (b), Figure A.20]: the total HV side referred copper resistance is now 153 m Ω (at f = 100 kHz), i.e. 26% less than the AC resistance measured for the side-byside arrangement.

Furthermore, for the transformer with n = 19, a total HV side referred stray inductance of 7.9 µH, a magnetizing inductance of 2.7 mH and a parasitic coupling capacitance of 65 pF between HV side and LV side are measured at 100 kHz (Figure A.21 and Figure A.22). For n = 16 and f = 100 kHz, the measured total HV side referred copper resistance is 110 m Ω , the total stray inductance is 5.0 µH, the magnetizing inductance is 1.9 mH, and the coupling capacitance between HV side and LV side is 69 pF (Figure A.23 and Figure A.24).

The measurement results presented in Figure A.21 and Figure A.23 depict the total stray inductances $L_{tr1} + n^2 L_{tr2}$ and the total AC resistances $R_{tr1} + n^2 R_{tr2}$. For the employed transformer model (Figure 3.29 in Section 3.3), however, distinct values are required for L_{tr1} , L_{tr2} , R_{tr1} , and R_{tr2} .

The inductances $L_{\rm tr1}$ and $L_{\rm tr2}$ are determined from the measurement results (Figure A.21 and Figure A.23) at f = 100 kHz and based on the assumption that the magnetic energy stored in the transformer stray inductance is equally shared by the LV side winding and the HV side winding, i.e. $L_{\rm tr1} \approx n^2 L_{\rm tr2}$. The respective results for $L_{\rm tr1}$ and $L_{\rm tr2}$ are listed in Table A.9.

The resistances R_{tr1} and R_{tr2} are determined from the measurement results, too; there, however, the AC resistance R_{tr1} is calculated according to Appendix A.9.4; R_{tr2} is then obtained from (A.17). Due to the essentially different characteristics of the magnetic field lines inside the conductors for the part of the winding which is surrounded by the transformer core and for the transformer's end turns, the calculation of R_{tr1} distinguishes between these 2 parts.

For the winding portion which is surrounded by the transformer core, the respective AC resistance of the *i*-th turn, $R_{tr1,inside,i}$, is calculated according to Appendix A.9.4 and [38]:

$$R_{\rm tr1,inside,i} = R_{\rm tr1,inside,DC} \cdot \left(F_{\rm R} + \frac{4G_{\rm R}n_{\rm strands}^2}{\pi d_{\rm tl}^2} \underbrace{\int_{0}^{d_{\rm tl}/2} \int_{0}^{d_{\rm tl}/2} |\vec{H_i}(r,\varphi)|^2 r d\varphi dr}_{\rm obtained from FEM simulation,} \right)_{\rm obtained from FEM simulation,}$$

$$Table A.5 and Table A.6$$
(A.10)

 $[R_{tr1,inside,DC}$ is calculated with (A.12); d_{tl} denotes the total diameter of



Figure A.21: (a) Measured HV side referred total stray inductance of the HF transformer with n = 19 (cf. Figure 3.29); (b) and (c) measured HV side referred AC resistance. In order to obtain the depicted measurement results, the LV winding of the transformer is shorted and thus, the implication of the transformer magnetizing inductance (2.7 mH at 100 kHz) on $L_{\rm tr2,19}$ and $R_{\rm tr2,19}$ is neglected. The measurement is accomplished using the Agilent 4294A Precision Impedance Analyzer. At f = 100 kHz, the accuracies of the Impedance Analyzer are $\pm 0.13\%$ for $L_{\rm tr1,19} + n^2 L_{\rm tr2,19}$ and $\pm 4.1\%$ for $R_{\rm tr1,19} + n^2 R_{\rm tr2,19}$ (calculated according to [152]).



Figure A.22: (a) Measured magnetizing inductance, $L_{\rm M,19}$ of the HF transformer with n = 19; at f = 900 kHz a parallel resonance occurs ($L_{\rm M,19} = 2.7 \,\mu$ H in parallel to a capacitance of 11.6 pF); (b) coupling capacitance between the HV side and the LV side of the HF transformer for n = 19. The measurement is accomplished using the Agilent 4294A Precision Impedance Analyzer. At f = 100 kHz, the accuracies of the Impedance Analyzer are $\pm 0.1\%$ for $L_{\rm M,19}$ and $\pm 0.1\%$ for $C_{\rm HVLV,19}$ (calculated according to [152]).



Figure A.23: (a) Measured HV side referred total stray inductance of the HF transformer with n = 16 (cf. Figure 3.29); (b) and (c) measured HV side referred AC resistance. In order to obtain the depicted measurement results in (a), (b), and (c), the LV winding of the transformer is shorted and thus, the implication of the transformer magnetizing inductance (1.9 mH at 100 kHz) on $L_{\rm tr2,16}$ and $R_{\rm tr2,16}$ is neglected. The measurement is accomplished using the Agilent 4294A Precision Impedance Analyzer. At f = 100 kHz, the accuracies of the Impedance Analyzer are $\pm 0.15\%$ for $L_{\rm tr1,16} + n^2 L_{\rm tr2,16}$ and $\pm 4.4\%$ for $R_{\rm tr1,16} + n^2 R_{\rm tr2,16}$ (calculated according to [152]).



Figure A.24: (a) Measured magnetizing inductance, $L_{\rm M,16}$ of the HF transformer with n = 16; at f = 920 kHz a parallel resonance occurs ($L_{\rm M,19} = 1.9 \,\mu$ H in parallel to a capacitance of 15.8 pF); (b) coupling capacitance between the HV side and the LV side of the HF transformer for n = 16. The measurement is accomplished using the Agilent 4294A Precision Impedance Analyzer. At f = 100 kHz, the accuracies of the Impedance Analyzer are $\pm 0.1\%$ for $L_{\rm M,16}$ and $\pm 0.1\%$ for $C_{\rm HVLV,16}$ (calculated according to [152]).

the litz wire, calculated with (A.178); n_{strands} denotes the number of single strands]. FEM simulation is used to calculate $\int_{0}^{d_{\text{tl}}/2} \int_{0}^{2\pi} |\vec{H}_{i}(r,\varphi)|^{2} r d\varphi dr$ (Table A.5, Table A.6).

The AC resistance of the *i*-th turn of the end turns is also calculated using FEM simulation results [Figure A.25 (a)]:

$$R_{\rm tr1,outside,i} = R_{\rm tr1,outside,DC,i} \cdot \left(F_{\rm R} + \frac{4G_{\rm R}n_{\rm strands}^2}{\pi d_{\rm t1}^2} \underbrace{\int\limits_{0}^{d_{\rm t1}/2} \int\limits_{0}^{2\pi} |\vec{H}_i(r,\varphi)|^2 r \mathrm{d}\varphi \mathrm{d}r}_{\text{obtained from FEM simulation,}} \right)_{\rm Table A.7 and Table A.8}$$
(A.11)

 $[R_{tr1,outside,DC,i}$ is calculated with (A.13)]. The additional resistance increase due to the winding termination is neglected on the HV side.

With the DC resistances of the employed litz wires,

$$R_{\rm tr1,inside,DC} = \frac{4 \, l_{\rm tr1,inside}}{\sigma_{\rm Cu} n_{\rm strands} \pi d_{\rm c}^2},\tag{A.12}$$

$$R_{\rm tr1,outside,DC,i} = \frac{4 \, l_{\rm tr1,outside,i}}{\sigma_{\rm Cu} n_{\rm strands} \pi d_{\rm c}^2},\tag{A.13}$$

the particular wire lengths (cf. Figure A.18, Table A.7, Table A.8),

$$l_{\rm tr1,inside} = 2l_{\rm tr,core} = 2 \cdot 38.1 \,\mathrm{mm} = 76.2 \,\mathrm{mm},$$
 (A.14)

$$l_{\rm tr1,outside} = d_{\rm tr1,i} \cdot \pi, \tag{A.15}$$

and the equations (A.10) and (A.11), the total AC resistance R_{tr1} can be calculated:⁴

$$R_{\rm tr1} = \sum_{i=1}^{N_1} \left(R_{\rm tr1, inside, i} + R_{\rm tr1, outside, i} \right).$$
(A.16)

The remaining winding resistance is attributed to the LV side:

 $^{^{4}}$ In order to simplify the calculations, the 2 litz wires employed for the HF transformer with n = 16 are combined to a single litz wire with a single strand copper diameter of 0.1 mm. The number of strands is adjusted such that the total copper cross section area remains the same, i.e. $n_{\rm strands} = 305$.

$n = 19, d_{\rm tl} = 2.3 {\rm mm}$					
Conductor Number:	1	2	3	4	5
$\int\limits_{0}^{d_{\mathrm{tl}}/2} \int\limits_{0}^{2\pi} ec{H_i}(r,arphi) ^2 r \mathrm{d}arphi \mathrm{d}r$	$37.8\mathrm{A}^2$	$16.1 \mathrm{A}^2$	$12.7\mathrm{A}^2$	$52.2\mathrm{A}^2$	$50.6 \mathrm{A}^2$
Conductor Number:	6	7	8	9	10
$\int_{0}^{d_{\rm tl}/2} \int_{0}^{2\pi} \vec{H}_i(r,\varphi) ^2 r \mathrm{d}\varphi \mathrm{d}r$	$15.2\mathrm{A}^2$	$14.5\mathrm{A}^2$	$54.2\mathrm{A}^2$	$52.3\mathrm{A}^2$	$15.8\mathrm{A}^2$
Conductor Number:	11	12	13	14	15
$\int\limits_{0}^{d_{\mathrm{tl}}/2} \int\limits_{0}^{2\pi} ec{H_i}(r,arphi) ^2 r \mathrm{d}arphi \mathrm{d}r$	$15.2\mathrm{A}^2$	$55.2\mathrm{A}^2$	$51.7\mathrm{A}^2$	$17.3\mathrm{A}^2$	$15.3\mathrm{A}^2$
Conductor Number:	16	17	18	19	
$\int\limits_{0}^{d_{\mathrm{tl}}/2} \int\limits_{0}^{2\pi} ec{H_i}(r,arphi) ^2 r \mathrm{d}arphi \mathrm{d}r$	$57.1 \mathrm{A}^2$	$43.8\mathrm{A}^2$	$24.7\mathrm{A}^2$	$13.8\mathrm{A}^2$	

Table A.5: FEM simulator results for $\int_{0}^{d_{t1}/2} \int_{0}^{2\pi} |\vec{H}_i(r,\varphi)|^2 r d\varphi dr$ obtained for the winding portion inside the core [*i*-th turn of the HF transformer with n = 19, cf. Figure A.25 (b)].

$n=16,d_{\rm tl}=2.6\rm mm$					
Conductor Number:	1	2	3	4	5
$\int\limits_{0}^{d_{ ext{tl}}/2} \int\limits_{0}^{2\pi} ec{H_i}(r,arphi) ^2 r \mathrm{d}arphi \mathrm{d}r$	$32.7\mathrm{A}^2$	$32.7\mathrm{A}^2$	$37.1\mathrm{A}^2$	$37.1\mathrm{A}^2$	$38.2\mathrm{A}^2$
Conductor Number:	6	7	8	9	10
$\int_{0}^{d_{\mathrm{tl}}/2} \int_{0}^{2\pi} \vec{H}_i(r,\varphi) ^2 r \mathrm{d}\varphi \mathrm{d}r$	$38.2\mathrm{A}^2$	$38.5\mathrm{A}^2$	$38.5\mathrm{A}^2$	$38.5\mathrm{A}^2$	$38.5\mathrm{A}^2$
Conductor Number:	11	12	13	14	15
$\int_{0}^{d_{\mathrm{tl}}/2} \int_{0}^{2\pi} \vec{H_i}(r,\varphi) ^2 r \mathrm{d}\varphi \mathrm{d}r$	$38.2\mathrm{A}^2$	$38.2\mathrm{A}^2$	$37.1\mathrm{A}^2$	$37.1\mathrm{A}^2$	$32.7\mathrm{A}^2$
Conductor Number:	16				
$\int\limits_{0}^{d_{ ext{tl}}/2} \int\limits_{0}^{2\pi} ec{H_i}(r,arphi) ^2 r \mathrm{d}arphi \mathrm{d}r$	$32.7\mathrm{A}^2$				

Table A.6: FEM simulator results for $\int_{0}^{d_{t1}/2} \int_{0}^{2\pi} |\vec{H}_i(r,\varphi)|^2 r d\varphi dr$ obtained for the winding portion inside the core [*i*-th turn of the HF transformer with n = 16, cf. Figure A.25 (b)].



Figure A.25: (a) FEM simulation of the magnetic field (RMS value) generated by the transformer end turns ($N_1 = 19$, $N_2 = 1$). For the depicted field plot, $I_{AC1} = 10.5$ A, $I_{AC2} = 200$ A, and f = 100 kHz are selected. This Figure also shows the 4 radiuses $d_{tr1,1/2}$, $d_{tr1,9/2}$, $d_{tr1,10/2}$, and $d_{tr1,19/2}$ of the first, the ninth, the tenth, and the 19th end turn to illustrate the calculation of $d_{tr1,i}$; $d_{tr1,i}$ is needed to calculate the DC resistances $R_{tr1,outside,DC,i}$ [cf. (A.13) and (A.15)]. (b) Numbering used for the turns of the HV winding.

Table A.9 lists the component values of the equivalent transformer circuit depicted in Figure 3.29 (however, in order to improve the model accuracy, the transformer stray inductance determined for the fully assembled converter setup is used to obtain L_{tr1} and L_{tr2} , cf. Appendix A.2.5 and Table A.10).

$n=19,d_{\rm tl}=2.3\rm mm$					
Conductor Number:	1	2	3	4	5
$\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} \vec{H}_i(r,\varphi) ^2 r \mathrm{d}\varphi \mathrm{d}r$	$45.7\mathrm{A}^2$	$43.8\mathrm{A}^2$	$26.4\mathrm{A}^2$	$51.6\mathrm{A}^2$	$51.0\mathrm{A}^2$
$d_{{ m tr}1,i}$	$11.2\mathrm{mm}$	$11.2\mathrm{mm}$	$15.4\mathrm{mm}$	$15.4\mathrm{mm}$	$19.5\mathrm{mm}$
Conductor Number:	6	7	8	9	10
$\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} \vec{H}_i(r,\varphi) ^2 r \mathrm{d}\varphi \mathrm{d}r$	$21.1\mathrm{A}^2$	$17.9\mathrm{A}^2$	$51.7\mathrm{A}^2$	$50.2\mathrm{A}^2$	$16.3 \mathrm{A}^2$
$d_{{ m tr}1,i}$	$19.5\mathrm{mm}$	$23.7\mathrm{mm}$	$23.7\mathrm{mm}$	$27.9\mathrm{mm}$	$27.9\mathrm{mm}$
Conductor Number:	11	12	13	14	15
$\int\limits_{0}^{d_{tl}/2}\int\limits_{0}^{2\pi} ec{H}_{i}(r,arphi) ^{2}r\mathrm{d}arphi\mathrm{d}r$	$14.5 \mathrm{A}^2$	$50.5\mathrm{A}^2$	$47.1 \mathrm{A}^2$	$14.2\mathrm{A}^2$	$11.9\mathrm{A}^2$
$d_{{ m tr}1,i}$	$32.1\mathrm{mm}$	$32.1\mathrm{mm}$	$36.2\mathrm{mm}$	$36.2\mathrm{mm}$	$40.4\mathrm{mm}$
Conductor Number:	16	17	18	19	
$\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} \vec{H}_{i}(r,\varphi) ^{2} r \mathrm{d}\varphi \mathrm{d}r$	$47.4\mathrm{A}^2$	$36.2\mathrm{A}^2$	$18.3\mathrm{A}^2$	$14.2\mathrm{A}^2$	
$d_{{ m tr}1,i}$	$40.4\mathrm{mm}$	$44.6\mathrm{mm}$	$44.6\mathrm{mm}$	$48.7\mathrm{mm}$	

Table A.7: FEM simulator results for $\int_{0}^{d_{\rm tl}/2} \int_{0}^{2\pi} |\vec{H}_i(r,\varphi)|^2 r d\varphi dr$ and the respective diameters $d_{{\rm tr}1,i}$ obtained for the *i*-th winding of the end turns of the HF transformer with n = 19.

A.2.5 Inductor

The converter inductor L can be placed in series to the LV side and / or the HV side of the HF transformer without affecting the converter functionality. For the given converter specifications it is easier to place the inductor on the HV side due to the lower current rating, despite the larger inductance value required.

Since the transformer stray inductances (L_{tr1}, L_{tr2}) and the LV side stray inductances of MOSFETs and PCB (L_{LV}) contribute to L, reduced additional inductance values L_{HV} are needed for the DAB, depending on the employed modulation scheme. The total stray inductance, $L_{\sigma} = L_{tr1} + n^2(L_{tr2} + L_{LV})$, is measured on the final DAB hardware prototype in order to correctly determine the additionally required inductance L_{HV} . Therefore, the HV side transformer winding is shorted and the LV side full bridge generates rectan-

$n = 16, d_{\rm tl} = 2.6 {\rm mm}$					
Conductor Number:	1	2	3	4	5
$\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} \vec{H}_{i}(r,\varphi) ^{2} r \mathrm{d}\varphi \mathrm{d}r$	$42.6\mathrm{A}^2$	$42.6\mathrm{A}^2$	$39.4\mathrm{A}^2$	$39.4\mathrm{A}^2$	$38.5\mathrm{A}^2$
$d_{{ m tr}1,i}$	$11.5\mathrm{mm}$	$11.5\mathrm{mm}$	$16.8\mathrm{mm}$	$16.8\mathrm{mm}$	$22.1\mathrm{mm}$
Conductor Number:	6	7	8	9	10
$\int\limits_{0}^{d_{tl}/2}\int\limits_{0}^{2\pi} ec{H}_{i}(r,arphi) ^{2}r\mathrm{d}arphi\mathrm{d}r$	$38.5\mathrm{A}^2$	$37.8\mathrm{A}^2$	$37.8\mathrm{A}^2$	$36.9\mathrm{A}^2$	$36.9\mathrm{A}^2$
$d_{{ m tr}1,i}$	$22.1\mathrm{mm}$	$27.4\mathrm{mm}$	$27.4\mathrm{mm}$	$6\mathrm{mm}$	$32.6\mathrm{mm}$
Conductor Number:	11	12	13	14	15
$\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} \vec{H}_i(r,\varphi) ^2 r \mathrm{d}\varphi \mathrm{d}r$	$35.4\mathrm{A}^2$	$35.4\mathrm{A}^2$	$31.7\mathrm{A}^2$	$31.7\mathrm{A}^2$	$24.0\mathrm{A}^2$
$d_{{ m tr}1,i}$	$37.9\mathrm{mm}$	$37.9\mathrm{mm}$	$43.2\mathrm{mm}$	$43.2\mathrm{mm}$	$48.5\mathrm{mm}$
Conductor Number:	16				
$\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} \vec{H}_{i}(r,\varphi) ^{2} r \mathrm{d}\varphi \mathrm{d}r$	$24.0 \mathrm{A}^2$				
$d_{{ m tr}1,i}$	$48.5\mathrm{mm}$				

Table A.8: FEM simulator results for $\int_{0}^{d_{\rm tl}/2} \int_{0}^{2\pi} |\vec{H}_i(r,\varphi)|^2 r d\varphi dr$ and the respective diameters $d_{{\rm tr}1,i}$ obtained for the *i*-th winding of the end turns of the HF transformer with n = 16.

n	L_{tr1}	$L_{\mathrm{tr}2}$	L_{M}	$R_{ m tr1}$	$R_{ m tr2}$	$R_{\rm M}$
16	$2.1\mu\mathrm{H}$	$8.0\mathrm{nH}$	$1.9\mathrm{mH}$	$27\mathrm{m}\Omega$	$83\mu\Omega$	∞
19	$3.3\mu\mathrm{H}$	$9.1\mathrm{nH}$	$2.7\mathrm{mH}$	$39\mathrm{m}\Omega$	$114 \ \mu\Omega$	∞

Table A.9: Parasitic transformer component values $(f = f_{\rm S} = 100 \text{ kHz})$ employed for the detailed DAB converter model (Section 3.3: Figure 3.29, Figure 3.30). The resistance values $R_{\rm tr1}$ and $R_{\rm tr2}$ are calculated with (A.16) and (A.17), the stray inductances are obtained from Table A.10 and on the assumption $L_{\rm tr1} = n^2 L_{\rm tr2}$, and $L_{\rm M}$ is measured (Figure A.22 and Figure A.24).



Figure A.26: Measurement setup used to determine the stray inductance L_{σ} : the LV side full bridge generates a rectangular voltage v_{AC2} with a duty cycle D_2 according to Figure 3.3, which causes a linear change of the currents i_{L_2} and i_{L_1} ; the inductance L_{σ} is determined according to (A.18).

gular voltage pulses (Figure A.26), which excite linearly changing inductor currents i_{L_1} and i_{L_2} . On the assumption of a negligible transformer magnetizing current, i.e. $i_{L_1} \approx i_{L_2}/n$, L_{σ} is calculated with the measured input voltage v_{f2} of the full bridge and the measured current slope $di_{L_1}/dt \approx \Delta i_{L_1}/\Delta t$ (Figure A.27):

$$L_{\sigma} = L_{\rm tr1} + n^2 (L_{\rm tr2} + L_{\rm LV}) \approx \frac{n v_{\rm f2}}{|\Delta i_{L_1} / \Delta t|}.$$
 (A.18)

Moreover, with the average LV side transformer voltage, \overline{V}_{AC2} , being measured (e.g. according to Figure A.27: in order to eliminate the effect of the resistive voltage drop, the average is evaluated over the time interval $t_1 < t < t_3$, which is symmetrical around the time of the zero crossing of the inductor current, t_2), the stray inductances of the transformer $(L_{tr1} + n^2 L_{tr2})$ and of the PCB $(n^2 L_{LV})$ can be determined separately:

$$L_{\rm tr1} + n^2 L_{\rm tr2} \approx \frac{n \overline{V}_{\rm AC2}}{|\Delta i_{L_1}/\Delta t|},\tag{A.19}$$

$$n^2 L_{\rm LV} \approx \frac{n \left(v_{\rm f2} - \overline{V}_{\rm AC2} \right)}{|\Delta i_{L_1} / \Delta t|}.$$
 (A.20)

Table A.10 summarizes the results obtained for n = 16 and n = 19 and lists the respective inductance values required for $L_{\rm HV}$. Compared to the stray inductances measured for the stand-alone transformers (Figures A.23 and A.21), considerably smaller inductance values result; this effect is addressed to the high impedance transfer ratio, e.g. for n = 19 the impedance



Figure A.27: (a) Measured voltages v_{f2} and v_{AC2} ; at $t = t_0$ the LV side full bridge changes its output voltage from $v_{AC2} = 0$ to $v_{AC2} = V_2$, i.e. T₆ is turned off, T₅ is turned on, and T₈ remains in its on-state; at $t = t_4$ the LV side full bridge changes its output voltage from $v_{AC2} = V_2$ to $v_{AC2} = 0$, i.e. T₈ is turned off, T₇ is turned on, and T₅ remains in its on-state; v_{AC2} decreases due to resistive losses in the semiconductors and in the PCB; moreover, the overvoltage on v_{AC2} (generated due to the effects detailed in Section 4.3.3) causes the dip in the waveform of v_{f2} ; (b) measured current $-i_{L_1}$ in the HV side shorting bar for n = 19 and an input voltage of $v_{f2} = 10$ V.

transfer ratio is $n^2 = 361$, which makes the measured stray inductance very sensitive on the impedance of the shorting bar connected to the LV transformer terminations.

L	n	$\begin{aligned} \Delta i_{L_1}/\Delta t \\ (v_{\rm f2} = 10{\rm V}) \end{aligned}$	$\overline{V}_{ m AC2}$	$\frac{L_{\rm tr1}}{n^2 L_{\rm tr2}}$	$n^2 L_{\rm LV}$	$\begin{array}{c} Remaining \\ L_{\rm HV} \end{array}$
$15.5\mu\mathrm{H}$	16	$33.3\mathrm{A}/\mathrm{\mu s}$	$8.45\mathrm{V}$	$4.1\mu\mathrm{H}$	$0.7\mu\mathrm{H}$	$10.7\mu\mathrm{H}$
$22.4\mu\mathrm{H}$	16	$33.3\mathrm{A}/\mathrm{\mu s}$	$8.45\mathrm{V}$	$4.1\mu\mathrm{H}$	$0.7\mu\mathrm{H}$	$17.6\mu\mathrm{H}$
$26.7\mu\mathrm{H}$	19	$24.2\mathrm{A}/\mathrm{\mu s}$	$8.46\mathrm{V}$	$6.6\mu\mathrm{H}$	$1.2\mu\mathrm{H}$	$18.9\mu\mathrm{H}$

Table A.10: Measured stray inductances of the transformer and the PCB; employed inductance values $L_{\rm HV}$.

$L_{\rm HV}$	Employed Cores	Employed Litz Wires	$N_{L_{\rm HV}}$	$l_{ m g}$
$10.7\mu\mathrm{H}$	$4 \times E$ 32/6/20, $4 \times PLT$ 32/20/3, material: 3F3	parallel connected litz wires: $2 \times (200 \times 0.071 \text{ mm})$	5	$2.41\mathrm{mm}$
17.6 µH	$5 \times E$ 32/6/20, $5 \times PLT$ 32/20/3, material: 3F3	parallel connected litz wires: $2 \times (200 \times 0.071 \text{ mm})$	5	$1.67\mathrm{mm}$
18.9 µH	$6 \times E$ 32/6/20, $6 \times PLT$ 32/20/3, material: 3F3	parallel connected litz wires: $2 \times (200 \times 0.071 \text{ mm})$	5	$1.93\mathrm{mm}$

Table A.11: Design data of the inductors placed in series to the transformer (HV side) in order to obtain the total DAB converter inductance calculated for the different modulation schemes; the inductors are designed according to Appendix A.9.

The required inductors are designed according to Appendix A.9 using ferrite cores and litz wire windings; Table A.11 summarizes the key parameters of the employed inductors.

A.2.6 EMI Filters

Employed EMI Filters

The DC filter capacitors which are directly connected to the DAB (i.e. $C_{\rm f1}$ and $C_{\rm f2}$ in Figure 6.1) are exposed to high RMS currents (Table A.1). Therefore, and due to DC voltages of up to 450 V, metalized polypropylene film capacitors are selected for $C_{\rm f1}$ (470 nF / 630 VDC, 7 connected in parallel).
The low voltage capacitor $C_{\rm f2}$ is subject to very high RMS currents; moreover, a very low parasitic inductance (< 1 nH) is required. Thus, 96 ceramic capacitors (10 μ F / 25 V / X7R) are connected in parallel (Appendix A.2.3).

Both filter capacitors C_{f1} and C_{f2} are selected with respect to the capacitor RMS current; accordingly, the implemented capacitance values are small. In order to reduce the resulting DC port voltage ripples, the EMI filters depicted in Figure 6.1 are employed: the filter inductors L_{f1a} , L_{f1b} , L_{f2a} , and L_{f2b} suppress conducted EMI, the resistors R_{f1b} and R_{f2b} are providing a proper damping of the filters, and the DC capacitors C_{DC1} and C_{CD2} provide constant port voltages V_1 and V_2 . For C_{DC1} and C_{DC2} electrolytic capacitors are selected, since the RMS current stresses are very low in steady-state operation.

In order to simplify the design of the EMI filters, sinusoidal AC components of the inductor currents with a frequency of twice the switching frequency, $\hat{i}_{f1}(t) = \hat{I}_{f1,peak} \sin(4\pi f_S t)$ and $\hat{i}_{f2}(t) = \hat{I}_{f2,peak} \sin(4\pi f_S t)$, are considered and the maximum peak values of these AC components are set to:

- $\hat{I}_{f1,peak,max} = 500 \, mA$,
- $\hat{I}_{f2,peak,max} = 10 \, A.$

In a more comprehensive approach, the EMI filters could be designed with respect to a certain optimum, e.g. maximum power density.

HV Filter Design

For the design of $L_{\rm fla}$, $L_{\rm flb}$, $R_{\rm flb}$, $L_{\rm fla}$, the HV side full bridge is replaced by the current source $i_{\rm FB1}(t)$ [Figure A.28 (a), cf. Section 6.2.6]. On the assumption $C_{\rm DC1} \gg C_{\rm fl}$, the capacitor $C_{\rm DC1}$ can be replaced by a constant voltage source. Accordingly, the employed equivalent circuit of the EMI filter disregards $C_{\rm DC1}$ since the filter design solely considers AC signals [Figure A.28 (b)]. Moreover, the design only considers the fundamental component of the filter currents and voltages, i.e. $f = 2f_{\rm S}$; the AC component $\hat{i}_{\rm FB1}(t)$ is approximated with:

$$\hat{i}_{\rm FB1}(t) \approx \sqrt{2} \hat{I}_{\rm FB1,RMS} \cdot \sin\left(4\pi f_{\rm S}\right). \tag{A.21}$$

Besides the main filter inductor L_{f1a} (L_{f1a} carries the converter DC current), the components L_{f1b} and R_{f1b} , depicted in Figure A.28 (b), form a damping network. The design of this filter structure closely follows [116]. There, the ratio

$$n_{L_{\rm f1}} = L_{\rm f1b} / L_{\rm f1a}$$
 (A.22)

sets the filter damping. However, the filter attenuation decreases with increasing damping. According to [116], $n_{L_{f1}} = 0.5$ gives a good compromise



Figure A.28: (a) EMI filter employed on the HV side; the HV side full bridge is replaced by a current source (cf. Section 6.2.6); (b) equivalent AC circuit of the EMI filter used to calculate L_{f1} ; due to $C_{DC1} \gg C_{f1}$, the capacitor C_{DC1} can be replaced by a short circuit.

between filter degradation and the achievable damping. The filter resistance $R_{\rm f1b}$ is selected with respect to the optimal Q factor, $Q_{\rm opt,f1}$ [116],

$$Q_{\text{opt,f1}} = \sqrt{\frac{n_{L_{f1}}(3+4n_{L_{f1}})(1+2n_{L_{f1}})}{2(1+4n_{L_{f1}})}},$$
(A.23)

$$R_{\rm f1b} = Q_{\rm opt, f1} Z_{0, f1}, \tag{A.24}$$

using

$$Z_{0,f1} = \sqrt{\frac{L_{f1a}}{C_{f1}}}.$$
 (A.25)

With the specified peak current $\hat{I}_{f1,peak,max}$ and the maximum AC component of the input current on the HV side ($\hat{I}_{FB1,RMS,max} = 14.9$ A for phase shift modulation), the expression of the filter damping, given with

$$d(\omega = 4\pi f_{\rm S}) = \left| \frac{1 + j\omega \frac{L_{\rm f1a} + L_{\rm f1b}}{R_{\rm f1b}}}{1 - \omega^2 L_{\rm f1a} C_{\rm f1} + j\omega \frac{L_{\rm f1a} + L_{\rm f1b}}{R_{\rm f1b}} \left[1 - \omega^2 \left(L_{\rm f1a} \right) C_{\rm f1} \right] \right| = \frac{\hat{I}_{\rm FB1,RMS,max}}{\hat{I}_{\rm f1,peak,max}/\sqrt{2}} = 32.5 \, \rm{dB}, \quad (A.26)$$

is solved with respect to L_{f1a} , L_{f1b} , and R_{f1b} at $\omega = 4\pi f_S$.



Figure A.29: Bode plot of the HV EMI filter transfer function, $G_{f1} = \hat{I}_{FB1,RMS}/\hat{I}_{f1,RMS}$, with and without filter damping; $C_{f1} = 3.3 \,\mu\text{F}$, $L_{f1a} = 25 \,\mu\text{H}$, $L_{f1b} = 12.5 \,\mu\text{H}$, $R_{f1b} = 2.5 \,\Omega$. With the proposed damping network, the maximum gain is 6.2 dB at $f = 23 \,\text{kHz}$ and the filter attenuation at $f = 200 \,\text{kHz}$ decreases from 42.2 dB to 32.5 dB.

For $C_{\rm f1} = 7 \cdot 470 \,\mathrm{nF} = 3.3 \,\mu\mathrm{F}$, the specified attenuation and optimal damping are achieved for:

- $L_{\rm f1a} = 25 \,\mu{\rm H},$
- $L_{\rm f1b} = 12.5 \,\mu{\rm H},$
- $R_{\rm f1b} = 2.5 \,\Omega.$

The Bode plots of the respective filter transfer functions, $G_{f1} = \hat{I}_{FB1,RMS}/\hat{I}_{f1,RMS}$, with and without damping are shown in Figure A.30.

The DC capacitor C_{DC1} is designed with respect to the output impedance of the connected hardware in order to achieve a stable converter operation [116] and with respect to the required electric energy storage capabilities; for the given DAB converter, $C_{\rm DC1} = 220 \,\mu F$ is selected.

LV Filter Design

On the HV side and on the LV side, the same filter structures are employed. Therefore, the LV filter design method is identical to the method discussed above for the HV filter. For $C_{\rm f2} = 96 \cdot 10 \ \mu \rm F = 960 \ \mu \rm F$, the specified attenuation and optimal damping are achieved for:

- $L_{f2a} = 70 \text{ nH},$
- $L_{\rm f2b} = 35 \, \rm nH$,
- $R_{\rm f2b} = 8 \,\mathrm{m}\Omega$.

The Bode plots of the respective filter transfer functions, $G_{f2} = \hat{I}_{FB2,RMS}/\hat{I}_{f2,RMS}$, with and without damping are shown in Figure A.30.

For $C_{\rm DC2}$ a very large capacitance of 200 mF is employed in order to decouple the LV port of the DAB and the LV DC supply, since the impedance of the connecting cable to the LV supply considerably influences the control-to-output transfer function, derived in Section 6, if a smaller capacitance (e.g. $C_{\rm DC2} = 20 \,\mathrm{mF}$) is used.

A.3 Three-Phase DAB Design

This Section summarizes the principle of operation and the design equations employed for the three-phase DAB. A more detailed discussion on this power converter and the derivations of the presented equations are given in [49,54].

A.3.1 Principle of Operation

The three-phase DAB converter consists of 6 half bridge circuits and 3 HF transformers [Figure A.31 (a)]. During steady-state operation, the HV side half bridges are operated with a respective phase shift of $\pi/3$ [equal to a time shift of $T_S/6$, Figure A.31 (c)]. Moreover, each HF transformer connects a HV side half bridge to a LV side half bridge. According to Figures A.31 (b) and (c) [there, $T_{\varphi} = \varphi T_S/(2\pi)$ applies], equal phase shifts φ between the output voltages of the respective half bridges on the HV side and on the LV side are used to achieve the required power transfer between the DC ports of the power converter.



Figure A.30: Bode plot of the LV side EMI filter transfer function, $G_{f2} = \hat{I}_{FB2,RMS}/\hat{I}_{f2,RMS}$, with and without filter damping; $C_{f2} = 960 \,\mu\text{F}$, $L_{f2a} = 70 \,\text{nH}$, $L_{f2b} = 35 \,\text{nH}$, $R_{f2b} = 8 \,\text{m}\Omega$. With the proposed damping network, the maximum gain is 6.2 dB at $f = 26 \,\text{kHz}$ and the filter attenuation at $f = 200 \,\text{kHz}$ decreases from 40.4 dB to 30.8 dB.

A.3.2 Lossless Electric Converter Model

In order to simplify the design process, the lossless electric model of the threephase DAB converter is employed (cf. Section 3.1, [54]).

The phase shift, φ , required to achieve a certain power transfer, is:

$$\varphi = \begin{cases} \frac{2\pi - 2\pi\sqrt{1 - \frac{9f_{\rm S}L|P|}{nV_1V_2}}}{3} \operatorname{sgn}(P) & \forall \quad |P| \le \frac{nV_1V_2}{12f_{\rm S}L} \\ \frac{3\pi - \pi\sqrt{7 - \frac{72f_{\rm S}L|P|}{nV_1V_2}}}{6} \operatorname{sgn}(P) & \forall \quad \frac{nV_1V_2}{12f_{\rm S}L} < |P| \le P_{\rm max} \end{cases}$$
(A.27)



Figure A.31: (a) Three-phase DAB converter topology according to [49,54]; (b) voltage waveforms $v_{AC1}(t)$ and $v_{AC2}(t)$ and current waveforms $i_{AC1a}(t)$, $i_{AC1b}(t)$, and $i_{AC1c}(t)$ for $V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW (power transfer direction: HV \rightarrow LV), n = 19, $L = 21.4 \,\mu$ H, and $f_S = 100$ kHz; (c) respective gate signals inclusive dead time intervals. At the particular operating point, soft switching is achieved on the HV side. On the LV side, hard switching transitions occur (cf. Section 4.3.2 and Section 4.3.3).

using:

$$P_{\rm max} = \frac{7nV_1V_2}{72f_{\rm S}L} \tag{A.28}$$

(maximum power occurs for $\varphi = \pm \pi/2$; for the lossless model, $P = P_1 = P_2$ applies).

Similar to the single-phase DAB, the conduction losses $P_{\rm S1,cond}$ (HV side switches) and $P_{\rm S2,cond}$ (LV side switches), the copper losses $P_{\rm tr1,cond}$, $P_{\rm tr2,cond}$ (HF transformer), and $P_{L_{\rm HV},cond}$ (HF inductor), and the AC losses on the LV side of the PCB are calculated using the transformer RMS currents $I_{\rm AC1a}$, $I_{\rm AC1b}$, and $I_{\rm AC1c}$,

$$P_{\rm S1,cond} = \left(I_{\rm AC1a}^2 + I_{\rm AC1b}^2 + I_{\rm AC1c}^2\right) R_{\rm 3DAB,S1},\tag{A.29}$$

$$P_{\rm S2,cond} = n^2 \left(I_{\rm AC1a}^2 + I_{\rm AC1b}^2 + I_{\rm AC1c}^2 \right) R_{\rm 3DAB,S2},\tag{A.30}$$

$$P_{L_{\rm HV},\rm cond} = \left(I_{\rm AC1a}^2 + I_{\rm AC1b}^2 + I_{\rm AC1c}^2\right) R_{\rm 3DAB, L_{\rm HV}},\tag{A.31}$$

$$P_{\rm tr1,cond} = \left(I_{\rm AC1a}^2 + I_{\rm AC1b}^2 + I_{\rm AC1c}^2\right) R_{\rm 3DAB,tr1},\tag{A.32}$$

$$P_{\rm tr2,cond} = n^2 \left(I_{\rm AC1a}^2 + I_{\rm AC1b}^2 + I_{\rm AC1c}^2 \right) R_{\rm 3DAB, tr2}, \tag{A.33}$$

$$P_{\rm PCB,a,cond} = n^2 \left(I_{\rm AC1a}^2 + I_{\rm AC1b}^2 + I_{\rm AC1c}^2 \right) R_{\rm 3DAB,PCB,a};$$
(A.34)

the respective resistance values are obtained at the switching frequency, $f_{\rm S} = 100$ kHz. In steady-state operation, the transformer RMS currents $I_{\rm AC1a}$, $I_{\rm AC1b}$, and $I_{\rm AC1c}$ are all equal, i.e. $I_{\rm 3DAB,L} = I_{\rm AC1a} = I_{\rm AC1b} = I_{\rm AC1c}$, and are calculated with:

$$I_{3\text{DAB},L} = \begin{cases} \frac{\sqrt{-9nV_1V_2|\varphi|^3 + 18\pi nV_1V_2|\varphi|^2 + \frac{5}{3}\pi^3(V_1 - nV_2)^2}}{18\pi^{3/2}f_5L} &\forall |\varphi| \le \frac{\pi}{3}, \\ \frac{\sqrt{nV_1V_2}}{18\pi^{3/2}f_5L} &\cdot \left[-18|\varphi|^3 + 27\pi|\varphi|^2 - 3\pi^2|\varphi| + \frac{\pi^3}{3}\frac{5V_1^2 - 9nV_1V_2 + 5(nV_2)^2}{nV_1V_2} \right]^{\frac{1}{2}} &\forall \frac{\pi}{3} < |\varphi| \le \frac{\pi}{2}. \end{cases}$$
(A.35)

The calculation of the switching losses $P_{S1,sw}$ (HV side) and $P_{S2,sw}$ (LV side) requires the instantaneous currents at the switching instants:

$$P_{\rm S1,sw} = 6 f_{\rm S} E_{\rm 3DAB,S1,sw} (I_{\rm 3DAB,S1,sw}), \tag{A.36}$$

$$P_{\rm S2,sw} = 6 \, f_{\rm S} \, E_{\rm 3DAB,S2,sw}(I_{\rm 3DAB,S2,sw}) \tag{A.37}$$

with:

$$I_{3\text{DAB},\text{S1,sw}} = \begin{cases} \frac{2\pi(V_1 - nV_2) + 3nV_2|\varphi|}{18\pi f_{\text{S}}L} & \forall \quad |\varphi| \le \frac{\pi}{3}, \\ \frac{2\pi\left(V_1 - \frac{3nV_2}{2}\right) + 6nV_2|\varphi|}{18\pi f_{\text{S}}L} & \forall \quad \frac{\pi}{3} < |\varphi| \le \frac{\pi}{2}, \end{cases}$$
(A.38)

$$I_{3\text{DAB},S2,\text{sw}} = \begin{cases} n \frac{2\pi (nV_2 - V_1) + 3V_1|\varphi|}{18\pi f_S L} & \forall \quad |\varphi| \le \frac{\pi}{3}, \\ n \frac{2\pi (nV_2 - \frac{3V_1}{2}) + 6V_1|\varphi|}{18\pi f_S L} & \forall \quad \frac{\pi}{3} < |\varphi| \le \frac{\pi}{2}. \end{cases}$$
(A.39)

The transformer core loss calculation employs the Steinmetz equation (4.10) and thus, the peak flux density inside the transformer core, $B_{3DAB,tr,peak}$ is required. According to Appendix A.2.5, the HF inductor is placed on the HV side, so

$$B_{3\text{DAB,tr,peak}} = \frac{V_2}{9f_S N_2 A_{3\text{DAB,tr,core}}}$$
(A.40)

results (N_2 denotes the number of turns on the LV side and $A_{3DAB,tr,core}$ is the core cross section employed for a single HF transformer of the three-phase DAB). The total core losses are:

$$P_{3\text{DAB,tr,core}} \approx 3V_{3\text{DAB,tr,core}} k f_{\text{S}}^{\alpha} B_{3\text{DAB,tr,peak}}^{\beta}$$
(A.41)

(i.e. 3 HF transformers; $V_{3DAB,tr,core}$ denotes the core volume of a single transformer). The Steinmetz parameters k, α , and β are given in Appendix C (Table C.2, Figure C.4).

The inductors' core losses $P_{3\text{DAB},L_{\text{HV}},\text{core}}$ are calculated according to (4.14) and (4.15), using the peak inductor currents $I_{3\text{DAB},L,\text{peak}} = I_{\text{AC1a},\text{peak}} = I_{\text{AC1a},\text{peak}}$. The peak inductor currents are approximately calculated with

$$I_{3DAB,L,peak} \approx \sqrt{2} \cdot I_{3DAB,L},$$
 (A.42)

since the waveforms of the transformer currents are close to a sinusoidal waveform [Figure A.31 (b)].

In order to calculate the RMS currents through the DC capacitors C_{DC1} and C_{DC2} , the RMS values of the input current $i_{3\text{DAB},1}$ and of the output current $i_{3DAB,2}$ [cf. Figure A.31 (a)] need to known:

$$I_{3\text{DAB},1} = \begin{cases} \frac{1}{18\pi^{3/2} f_{\text{S}}L} \left[-9nV_2(2V_1 + 3nV_2)|\varphi|^3 + 9\pi nV_2(V_1 + 3nV_2)|\varphi|^2 + \frac{\pi^3}{3}(V_1 - nV_2)^2 \right]^{\frac{1}{2}} \\ \forall \quad |\varphi| \leq \frac{\pi}{3}, \\ \frac{1}{18\pi^{3/2} f_{\text{S}}L} \left[-36nV_1V_2|\varphi|^3 + 27\pi(2V_1 - nV_2)nV_2|\varphi|^2 + 3\pi^2 nV_2(9nV_2 - 8V_1)|\varphi| + \frac{\pi^3}{3}(V_1^2 + 9nV_1V_2 - 11(nV_2)^2) \right]^{\frac{1}{2}} \\ \forall \quad \frac{\pi}{3} < |\varphi| \leq \frac{\pi}{2}, \\ (A.43) \end{cases}$$

$$I_{3\text{DAB},2} = \begin{cases} \frac{n}{18\pi^{3/2} f_{\text{S}}L} \left[-9V_1(3V_1 + 2nV_2)|\varphi|^3 + 9\pi V_1(3V_1 + nV_2)|\varphi|^2 + \frac{\pi^3}{3}(V_1 - nV_2)^2 \right]^{\frac{1}{2}} \\ \forall \quad |\varphi| \leq \frac{\pi}{3}, \\ \frac{n}{18\pi^{3/2} f_{\text{S}}L} \left[-36nV_1V_2|\varphi|^3 - 27\pi V_1(V_1 - 2nV_2)|\varphi|^2 + 3\pi^2 V_1(9V_1 - 8nV_2)|\varphi| + \frac{\pi^3}{3}(-11V_1^2 + 9nV_1V_2 + (nV_2)^2) \right]^{\frac{1}{2}} \\ \forall \quad \frac{\pi}{3} < |\varphi| \leq \frac{\pi}{2}. \end{cases}$$

$$(A.44)$$

With known DC port currents $I_1 = P/V_1$ and $I_2 = P/V_2$, the capacitor RMS currents $I_{3DAB,C_{DC1}}$ and $I_{3DAB,C_{DC2}}$ result:

$$I_{3\text{DAB},C_{\text{DC1}}} = \sqrt{I_{3\text{DAB},1}^2 - I_1^2},$$
 (A.45)

$$I_{3\text{DAB},C_{\text{DC2}}} = \sqrt{I_{3\text{DAB},2}^2 - I_2^2}.$$
 (A.46)

The second part of the losses generated by the PCB (connection between the DC capacitors C_{DC2} and the half bridges) is calculated according to (4.24) using (A.44) and $I_2 = P/V_2$:

$$P_{\text{PCB},\text{b,cond}} = I_2^2 R_{3\text{DAB},\text{PCB},\text{b}}(f=0) + I_{3\text{DAB},C_{\text{DC2}}}^2 R_{3\text{DAB},\text{PCB},\text{b}}(f=6f_{\text{S}}).$$
(A.47)

For the single-phase DAB and for the three-phase DAB, the same semiconductor chip areas are considered and, consequently, the same gate driver power demand is assumed. Therefore, the power demands of the auxiliary power supplies remain equal to (4.16).

A.3.3 Model Parameterization

The three-phase DAB converter model is parameterized using data from the single-phase DAB detailed in Chapter 4 and Appendix A.2.

Resistance Values

The single-phase DAB employs 2 half bridge circuits on the HV side and 2 half bridge circuits on the LV side. In contrast, the three-phase DAB uses 3 half bridge circuits on either side. With respect to a constant semiconductor chip area, the employed on-state resistances of the switches are scaled according to:

$$R_{3\text{DAB},\text{S1}} = \frac{3}{2}R_{\text{S1}} = 105\,\text{m}\Omega,\tag{A.48}$$

$$R_{\rm 3DAB,S2} = \frac{3}{2}R_{\rm S2} = 0.52\,\mathrm{m}\Omega\tag{A.49}$$

(A.50)

[on the assumption of a junction temperature of $T_{\rm j} = 25^{\circ}$ C; cf. (4.5), and (4.6); in order to include increased losses due to HF effects on the LV side, the MOS-FET channel resistance $R_{\rm DS(on), IRF2804}(f = 100 \text{ kHz}) \approx 280 \text{ m}\Omega$ is considered, cf. Figure 4.4].

In order to achieve low HF losses, 3 parallel copper planes are needed on the LV side to provide an electric connection between each HF transformer and the HF port of the respective half bridge (similar to Figure A.7, cf. Appendix A.2.3). Thus, compared to the single-phase DAB, one additional copper layer is required on the LV side of the PCB of the three-phase DAB. Provided that the PCB geometry remains unmodified and that the PCB again employs copper layers with a thickness of $200 \,\mu\text{m}$ [cf. Section 4.1, (A.8)], then the following resistances apply:

$$R_{3\text{DAB,PCB,a}}(f = 100 \text{ kHz}) = \frac{1}{2} R_{\text{PCB,a}}(f = 100 \text{ kHz}) = 130 \,\mu\Omega,$$
 (A.51)

$$R_{3\text{DAB,PCB,b}}(f=0) = R_{\text{PCB,b}}(f=0) = 120 \,\mu\Omega,$$
 (A.52)

$$R_{3\text{DAB,PCB,b}}(f = 600 \,\text{kHz}) = R_{\text{PCB,b}}(f = 600 \,\text{kHz}) = 530 \,\mu\Omega$$
 (A.53)

[for the single-phase DAB, $R_{PCB,a}$ considers the resistance of the transformer feed line and the respective return line; therefore the factor 1/2 is needed in (A.51)].



Figure A.32: Planar transformer employed for the single-phase DAB; in order to improve the efficiency, 2 windings are operated in parallel on the LV side. The depicted setup is used to estimate the expected resistance values $R_{3DAB,tr1}$ and $R_{3DAB,tr2}$ for the three-phase DAB.

The calculations of the expected transformer resistance values $R_{3DAB,tr1}$ and $R_{3DAB,tr2}$ are based on the transformer setup depicted in Figure A.32 and on the product of $A_{tr,W}$ (winding window area) and $A_{tr,core}$ (core cross section area).

Provided that the HF inductor L solely resides on the HV side and that the peak magnetic flux densities in the transformer cores of the single-phase DAB ($B_{\text{DAB,tr,peak}}$) and the three-phase DAB ($B_{3\text{DAB,tr,peak}}$) are equal, the ratio of the core cross sections employed for the three-phase DAB and for the single-phase DAB, $A_{3\text{DAB,tr,core}}/A_{\text{DAB,tr,core}}$, is

$$B_{\text{DAB,tr,peak}} = \frac{V_2}{4f_{\text{S}}N_2A_{\text{DAB,tr,core}}} \\ B_{3\text{DAB,tr,peak}} = \frac{V_2}{9f_{\text{S}}N_2A_{3\text{DAB,tr,core}}} \\ \Rightarrow \frac{A_{3\text{DAB,tr,core}}}{A_{\text{DAB,tr,core}}} = \frac{4}{9}. \quad (A.54)$$

According to [92] and in anticipation of the VA ratings presented in Section 2.2.1 ($S_{\text{DAB,tr}} = 4.6 \text{ kVA}$ and $S_{3\text{DAB,tr}} = 4.2 \text{ kVA}$, whereas $S_{3\text{DAB,tr}}$ denotes the total VA rating of all 3 transformers of the three-phase DAB), the winding window of a single transformer of the three-phase DAB, $A_{3\text{DAB,tr,W}}$ can be calculated according to:

$$\frac{S_{3\text{DAB,tr}}/3}{S_{\text{DAB,tr}}} = \frac{A_{3\text{DAB,tr,core}} A_{3\text{DAB,tr,W}}}{A_{\text{DAB,tr,core}} A_{\text{DAB,tr,W}}} \Rightarrow \frac{A_{3\text{DAB,tr,W}}}{A_{\text{DAB,tr,W}}} = \frac{A_{\text{DAB,tr,core}}}{A_{3\text{DAB,tr,core}}} \frac{S_{3\text{DAB,tr}}/3}{S_{\text{DAB,tr}}} \approx 0.68. \quad (A.55)$$

Furthermore, on the assumption of the core length, $l_{\rm tr,core}$, being proportional to $\sqrt{A_{\rm tr,core}}$ (cf. Figure A.32) and with the core length of the transformer used for the single-phase DAB, $l_{\rm DAB,tr,core} = 38.1$ mm, the core length of a single transformer of the three-phase DAB, $l_{\rm 3DAB,tr,core}$, can be estimated:

$$l_{3\text{DAB,tr,core}} = l_{\text{DAB,tr,core}} \sqrt{\frac{A_{3\text{DAB,tr,core}}}{A_{\text{DAB,tr,core}}}} = \frac{2}{3} \cdot 38.1 \text{ mm} = 25.4 \text{ mm}. \quad (A.56)$$

Besides $l_{3DAB,tr,core}$, the length of a single turn, $l_{3DAB,tr,s}$, needed in (A.59) and (A.60), additionally includes the average length of the end turns, $\overline{l}_{3DAB,tr,e}$. With the dimensions of the ferrite core used for the DAB (ELP 58: $p_c = 8.1 \text{ mm}, b_c = 21.5 \text{ mm},$ Figure A.52), $\overline{l}_{3DAB,tr,e}$ becomes

$$\bar{l}_{3\text{DAB,tr,e}} = \pi \left(p_{c} \sqrt{\frac{A_{3\text{DAB,tr,core}}}{A_{\text{DAB,tr,core}}}} + b_{c} \sqrt{\frac{A_{3\text{DAB,tr,W}}}{A_{\text{DAB,tr,W}}}} \right) = 23.2 \text{ mm.} \quad (A.57)$$

With this, the ratio $l_{3DAB,tr,s}/l_{DAB,tr,s}$ can be determined ($l_{DAB,tr,s}$ denotes the length of a single turn of the windings used for the HF transformer of the single-phase DAB):

$$\frac{l_{\rm 3DAB,tr,s}}{l_{\rm DAB,tr,s}} = \frac{2l_{\rm 3DAB,tr,core} + \bar{l}_{\rm 3DAB,tr,e}}{2l_{\rm DAB,tr,core} + \bar{l}_{\rm DAB,tr,e}} = \frac{2 \cdot 25.4 \,\mathrm{mm} + 72.9 \,\mathrm{mm}}{2 \cdot 38.1 \,\mathrm{mm} + 93.0 \,\mathrm{mm}} = 0.731.$$
(A.58)

Finally, with the data given in Table A.9 and (A.55) and (A.58), the transformer resistances $R_{3DAB,tr1}$ and $R_{3DAB,tr2}$ can be calculated:

$$R_{3\text{DAB,tr1}} = \rho \frac{N_1 l_{3\text{DAB,tr,s}}}{kA_{3\text{DAB,tr,W}}/(2N_1)} = \rho \frac{0.731N_1 l_{\text{DAB,tr,s}}}{0.68kA_{\text{DAB,tr,W}}/(2N_1)}$$

= 1.07R_{DAB,tr1} \approx n² \cdot 0.12 m\Omega, (A.59)

$$R_{3\text{DAB,tr2}} = 1.07R_{\text{DAB,tr2}} \approx 0.34\,\text{m}\Omega\tag{A.60}$$

(winding temperature: 25° C).

Depending on the actual converter parameters n and L, separate HF inductors $L_{\rm HV}$ are designed (Appendix A.2), according to the inductor design procedure outlined in Appendix A.9. Subsequently, the respective conduction and core losses are calculated according to (A.31) and (4.14).

The dielectric losses generated by the DC capacitors C_{DC1} and C_{DC2} are very small and therefore neglected.

Switching Losses

The switching losses are calculated using the results discussed in Sections 4.3.2 and 4.3.3. In order to facilitate a meaningful comparison of the single-phase DAB and the three-phase DAB, the particular loss energies $E_{\rm S1,sw}$ and $E_{\rm S2,sw}$ are scaled with respect to a constant semiconductor chip area.

In case of a constant current density in the semiconductor chip, the switching losses increase with increasing chip area (on the assumption of a homogeneous distribution of the switching losses across the chip surface):

$$E_{3DAB,S1,sw}(A_{S1}J_{3DAB,S1,sw}) = aE_{S1,sw}(A_{S1}J_{3DAB,S1,sw}),$$
(A.61)

$$E_{3\text{DAB},S2,sw}(A_{S2}J_{3\text{DAB},S2,sw}) = aE_{S2,sw}(A_{S2}J_{3\text{DAB},S2,sw})$$
(A.62)

(a denotes the relative chip size, i.e. the ratio of the chip size used for each switch of the three-phase DAB to the chip size used for the respective switch of the single-phase DAB; A_{S1} and A_{S2} denote the chip areas of the switches on the HV side and on the LV side, respectively). However, with constant currents $I_{3DAB,S1,sw}$ and $I_{3DAB,S2,sw}$ and with increasing chip areas, the current densities $J_{3DAB,S1,sw}$ and $J_{3DAB,S2,sw}$ decrease. Thus, the switching loss energies are calculated according to:

$$E_{3\text{DAB},\text{S1,sw}}(I_{3\text{DAB},\text{S1,sw}}) = aE_{\text{S1,sw}}\left(\frac{1}{a}I_{3\text{DAB},\text{S1,sw}}\right),\qquad(A.63)$$

$$E_{3\text{DAB},\text{S2,sw}}(I_{3\text{DAB},\text{S2,sw}}) = aE_{\text{S2,sw}}\left(\frac{1}{a}I_{3\text{DAB},\text{S2,sw}}\right),\tag{A.64}$$

$$a = \frac{2}{3}.\tag{A.65}$$

With a = 2/3, equal semiconductor chip areas are used for the switches of the three-phase DAB and for the switches of the single-phase DAB.

Transformer Core Volume

The core volume of a single transformer of the three-phase DAB, $V_{3DAB,tr,core}$, required to calculate the transformer core losses, is equal to the total transformer volume (except for the volume of the end turns) minus the volume of the winding window:

 $V_{3\text{DAB,tr,core}} = b_{3\text{DAB,tr,core}} h_{3\text{DAB,tr,core}} l_{3\text{DAB,tr,core}} - 2A_{3\text{DAB,tr,W}} l_{3\text{DAB,tr,core}}.$ (A.66)

Provided that the respective widths and heights of $A_{3DAB,tr,W}$ and $A_{3DAB,tr,core}$ scale proportional to $\sqrt{A_{3DAB,tr,W}}$ and $\sqrt{A_{3DAB,tr,core}}$, the ex-

pected transformer height and width can be determined with:

$$h_{3\text{DAB,tr,core}} = 13 \,\text{mm} \sqrt{\frac{A_{3\text{DAB,tr,W}}}{A_{\text{DAB,tr,W}}}} + 8.1 \,\text{mm} \sqrt{\frac{A_{3\text{DAB,tr,core}}}{A_{\text{DAB,tr,core}}}} = 16.2 \,\text{mm},$$
(A.67)
$$b_{3\text{DAB,tr,core}} = 43 \,\text{mm} \sqrt{\frac{A_{3\text{DAB,tr,W}}}{A_{\text{DAB,tr,W}}}} + 15.4 \,\text{mm} \sqrt{\frac{A_{3\text{DAB,tr,core}}}{A_{\text{DAB,tr,core}}}} = 45.8 \,\text{mm}$$
(A.68)

(using the dimensions of the ELP 58 core [155]). Thus, the estimated core volume of one transformer of the three-phase DAB is:

$$V_{3DAB,tr,core} = 9.12 \,\mathrm{cm}^3.$$
 (A.69)

A.3.4 Design Results

The three-phase DAB converter is designed with respect to maximum average efficiency, according to Appendix A.1. For n = 20 and $L = 22.2 \,\mu\text{H}$ the maximum average efficiency $\overline{\eta}_{\text{design}} = 90.9\%$ results. Table A.12 summarizes the expected converter currents and voltages.

A.4 LLC Converter Design

A.4.1 Analytical Model of the LLC Converter

Lossless Electric Converter Model

With the investigations on the LLC converter being confined to the continuous conduction mode (CCM), the analytical converter model (Figure A.33) becomes very similar to the analytical model of the DAB: again, the full bridge circuits are replaced by voltage sources and all LV side quantities are referred to the HV side. Thus, the LLC converter [Figure A.33 (a)] can be replaced by the circuit shown in Figure A.33 (b), with the high frequency AC part being composed of a series resonant circuit (inductor L and capacitor C) and the additional inductance $L_{\rm M}$. In contrast to the DAB, second order differential equations need to be considered regarding the inductor current i_L and the capacitor voltage v_C . With the inductor current $i_{L,i-1}$ and the capacitor voltage $v_{C,i-1}$ at the beginning of the *i*-th time interval $(t = t_{i-1})$,

$$i_{L,i-1} = i_L(t_{i-1}), \quad v_{C,i-1} = v_C(t_{i-1}),$$
(A.70)

collected in the vector

$$\vec{x}_{i-1} = (i_{L,i-1} \quad v_{C,i-1})^{\mathrm{T}},$$
 (A.71)

Property	Value	Description		
Maximum Average Efficiency:				
$\overline{\eta}_{ m design}$	90.9%	maximum average converter efficiency		
-		calculated with $(A.1)$, $(A.2)$, $(A.3)$,		
		and $(A.4)$		
<i>HV Side Switches</i> T_1 , T_2 , T_3 , T_4 , T_5 , T_6 ($i = 16$):				
$\max\left(V_{\rm 3DAB,DS,T_{\it i},peak}\right)$	$450\mathrm{V}$	maximum HV switch voltages		
$\max\left(I_{3\text{DAB},\text{D},\text{T}_{i},\text{peak}}\right)$	$14.2\mathrm{A}$	max. instantaneous HV switch current		
$\max\left(I_{3\text{DAB},\text{D},\text{T}_{i}}\right)$	$6.4\mathrm{A}$	maximum HV switch RMS current		
<i>LV Side Switches</i> T_7 , T_8 , T_9 , T_{10} , T_{11} , T_{12} ($i = 7 \dots 12$):				
$\max\left(V_{3\text{DAB},\text{DS},\text{T}_{i},\text{peak}}\right)$	$16\mathrm{V}$	maximum LV switch voltages		
$\max\left(I_{3\text{DAB},\text{D},\text{T}_{i},\text{peak}}\right)$	$285\mathrm{A}$	max. instantaneous LV switch current		
$\max\left(I_{3\text{DAB},\text{D},\text{T}_{i}}\right)$	$129\mathrm{A}$	maximum LV switch RMS currents		
Transformer:				
n	20	turns ratio		
$\max\left(I_{3\mathrm{DAB},L} ight)$	$9.1\mathrm{A}$	HV windings, max. RMS curr.		
$\max\left(n I_{3\text{DAB},L}\right)$	$182\mathrm{A}$	LV windings, max. RMS curr.		
$\max\left(B_{3\mathrm{DAB,tr,core,peak}}\right)$	$129\mathrm{mT}$	peak transformer flux densities		
HF Inductors L:				
L	$22.2\mu\mathrm{H}$	inductance values		
$\max\left(I_{3\mathrm{DAB},L,\mathrm{peak}}\right)$	$14.2\mathrm{A}$	max. instantaneous inductor current		
$\max\left(I_{3\mathrm{DAB},L} ight)$	$9.1\mathrm{A}$	maximum inductor RMS current		
HV Side DC Capacitor $C_{\rm DC1}$:				
$\max\left(I_{3\mathrm{DAB},C_{\mathrm{DC1}}}\right)$	4.1 A	maximum capacitor RMS current		
$LV Side DC Capacitor C_{DC2}$:				
$\max\left(I_{3\mathrm{DAB},C_{\mathrm{DC2}}}\right)$	66 A	maximum capacitor RMS current		

 Table A.12: Design summary for the three-phase DAB converter.



Figure A.33: (a) Power stage of the considered LLC converter; (b) lossless electric model of the LLC converter, provided that the converter is operated in continuous conduction mode (CCM).

and on the assumption of constant voltage being applied to L and C during the *i*-th time interval (defined in order to be consistent with the definitions used in Chapter 6),

$$v_{\mathrm{R},i-1} = v_{\mathrm{AC1}}(t_{(i-1)^+}) - n \, v_{\mathrm{AC2}}(t_{(i-1)^+}), \tag{A.72}$$

the respective time domain solutions for $i_{L,i}$ and $v_{C,i}$ at the end of the regarded time interval with the duration T_i are:

$$i_{L,i} = i_L(\vec{x}_{i-1}, T_i) = i_{L,i-1} \cos(\omega_0 T_i) + \frac{v_{\mathrm{R},i-1} - v_{C,i-1}}{Z_0} \sin(\omega_0 T_i),$$
(A.73)

$$v_{C,i} = v_C(\vec{x}_{i-1}, T_i) = Z_0 \, i_{L,i-1} \sin(\omega_0 \, T_{i-1}) + v_{R,i-1} - (v_{R,i-1} - v_{C,i-1}) \cos(\omega_0 \, T_i) \,, \tag{A.74}$$

$$\omega_0 = \frac{1}{\sqrt{L C}}, \quad Z_0 = \sqrt{\frac{L}{C}}.$$
(A.75)

The voltage $n v_{AC2}$ generates the current i_M through L_M [Figure A.33 (b)] and therefore $i_M(t)$ can be calculated separately using the solution of a first order differential equation.

Improved Converter Model

In order to increase the accuracy of the calculated inductor current waveform $i_L(t)$, conduction losses need to be considered (cf. Section 3.2). The final LLC converter model further includes the improved transformer model depicted in Figure A.34 (cf. Appendix A.2.4 and Table A.9) and LV side stray inductances (Table A.10). The resulting LLC converter model is shown in Figure A.35 (a).

The HF AC circuit of the LLC converter model depicted in Figure A.35 (a) is of 4th-order and thus, the expressions for $i_{L_1}(t)$ and $i_{L_2}(t)$ become very involved. Similar to the DAB, the wye-delta transformation is employed in order to reduce the calculation effort [Figure A.35 (b); cf. Section 3.3].

The expressions for \underline{Z}_{12} , \underline{Z}_{13} , and \underline{Z}_{23} ,

$$\underline{Z}_{12} = R_1 + n^2 R_2 + s \cdot (L_1 + n^2 L_2) + \frac{1}{sC} + \frac{n^2 (R_1 + sL_1 + \frac{1}{sC}) \cdot (R_2 + sL_2)}{R_{\rm M} || (sL_{\rm M})}, \quad (A.76)$$

$$\underline{Z}_{13} = R_1 + s L_1 + \frac{1}{s C} + [R_{\rm M}||(s L_{\rm M})] \cdot \left[1 + \frac{R_1 + s L_1 + \frac{1}{s C}}{n^2 (R_2 + s L_2)}\right], \quad (A.77)$$

$$\underline{Z}_{23} = n^2 (R_2 + s L_2) + [R_{\rm M}||(s L_{\rm M})] \cdot \left[1 + \frac{n^2 (R_2 + s L_2)}{R_1 + s L_1 + \frac{1}{s C}}\right], \qquad (A.78)$$



Figure A.34: Transformer model employed for the improved electric model of the LLC converter.



Figure A.35: (a) Improved electric model of the LLC converter; (b) the wye-delta transformation is employed to simplify the calculations.

contain higher order components and need to be simplified. On the assumptions

$$R_1 \ll s L_1 + \frac{1}{sC}, \ R_2 \ll s L_2, \ R_M \gg s L_M$$
 (A.79)

[cf. (3.102)], \underline{Z}_{12} and \underline{Z}_{13} become:

$$\underline{Z}_{12} \approx R_1 + n^2 R_2 + s \cdot \left(L_1 + n^2 L_2 + \frac{n^2 L_1 L_2}{L_M} \right) + \frac{1}{s C} \left(1 + \frac{n^2 L_2}{L_M} \right) \\
= R_{12} + s L_{12} + \frac{1}{s C_{12}}, \quad (A.80)$$

$$\underline{Z}_{13} \approx \left(R_1 + sL_1 + \frac{1}{sC}\right) \left(1 + \frac{L_{\rm M}}{n^2 L_2}\right) + sL_{\rm M}$$
$$= R_{13} + sL_{13} + \frac{1}{sC_{13}}, \quad (A.81)$$

i.e. equal to the impedances of 2 separate series resonant circuits. For the simplification of the remaining impedance \underline{Z}_{23} , the assumption $s^2 L_1 C \gg 1$ is additionally required, i.e. converter operation above resonance. With this, \underline{Z}_{23} reduces to the impedance of an inductor and a resistor connected in series,

$$\underline{Z}_{23} \approx n^2 \left(R_2 + s L_2 \right) \left(1 + \frac{L_{\rm M}}{L_1} \right) + s L_{\rm M} = R_{23} + s L_{23}, \tag{A.82}$$

and thus, $i_{L_{23}}$ can be calculated using (3.60).

The expressions for the inductor current and the capacitor voltage of \underline{Z}_{12} are:

$$i_{L_{12}}(\vec{x}_{12,i-1},t) = e^{-\frac{d_{12}\omega_{0,12}}{2}t} \left\{ i_{L_{12},i-1}\cos\left(\omega_{r,12}t\right) + \left[\frac{v_{R,i-1} - v_{C_{12},i-1}}{Z_{0,12}} - \frac{d_{12}}{2}i_{L_{12},i-1}\right] \frac{\omega_{0,12}}{\omega_{r,12}}\sin\left(\omega_{r,12}t\right) \right\}$$
(A.83)

$$v_{C_{12}}(\vec{x}_{12,i-1},t) = v_{\mathrm{R},i-1} + \mathrm{e}^{-\frac{d_{12}\,\omega_{0,12}}{2}t} \left\{ (v_{C_{12},i-1} - v_{\mathrm{R},i-1})\cos\left(\omega_{\mathrm{r},12}\,t\right) + \left[Z_{0,12}\,i_{L_{12},i-1} + \frac{d_{12}}{2}\left(v_{C_{12},i-1} - v_{\mathrm{R},i-1}\right)\right] \frac{\omega_{0,12}}{\omega_{\mathrm{r},12}}\sin\left(\omega_{\mathrm{r},12}\,t\right) \right\}, \quad (A.84)$$

$$\vec{x}_{12,i-1} = \begin{pmatrix} i_{L_{12},i-1} & v_{C_{12},i-1} \end{pmatrix}^{\mathrm{T}},$$
(A.85)

$$\omega_{0,12} = \frac{1}{\sqrt{L_{12} C_{12}}}, \quad Z_{0,12} = \sqrt{\frac{L_{12}}{C_{12}}}, \quad (A.86)$$

$$Q_{12} = \frac{Z_{0,12}}{R_{12}}, \ d_{12} = \frac{1}{Q_{12}}, \ \omega_{r,12} = \frac{\omega_{0,12}\sqrt{4 - d_{12}^2}}{2}.$$
 (A.87)

The expressions regarding the inductor current and the capacitor voltage of \underline{Z}_{13} are similar to (A.83) and (A.84), however, besides the respective component values, $v_{\mathrm{R},i-1}$ needs to be replaced by $v_{\mathrm{AC1},i-1} = v_{\mathrm{AC1}}(t_{(i-1)+})$. The currents $i_{L_1}(t)$ and $i_{L_2}(t)$ are finally calculated with (3.106).

The converter power levels P_1 and P_2 are calculated with (3.111) and the conduction losses are calculated based on the RMS currents I_{L_1} and I_{L_2} [cf. (3.117) and Chapter 4].

Steady-State Inductor Current and Capacitor Voltage

In order to determine the optimal converter parameters n, L, C, and $L_{\rm M}$ with respect to maximum average converter efficiency (Appendix A.1), the numerical solver algorithm discussed in Section 4.6.1 is used to predict the expected total losses of the LLC converter. The employed algorithm calculates the different stress values (RMS currents, RMS voltages, peak currents, peak voltages, and peak flux densities) for the converter components. It assumes steady-state converter operation, CCM, and constant AC voltages $v_{\rm AC1}(t)$ and $v_{\rm AC2}(t)$ during the considered time intervals.

The numerical solver requires the initial steady-state values of all inductor currents and all capacitor voltages of the employed simplified resonant tank [Figure A.35 (b)] to be known at one particular point in time (e.g. at t = 0) in order to calculate the converter stress values [i.e. to evaluate (3.60), (A.83), and (A.84) and the expressions derived therefrom, e.g. the expressions for the RMS transformer currents; cf. Section 3.2.1]. For \underline{Z}_{23} , the initial value $i_{L_{23}}(0)$ is calculated with (3.67). Considerably more complex expressions result for the initial inductor currents and the initial capacitor voltages of the series resonant circuits represented by \underline{Z}_{12} and \underline{Z}_{13} . The respective expressions are derived for $i_{L_{12}}(0)$ and $v_{C_{12}}(0)$ in the following paragraph; (A.92) summarizes the result for \underline{Z}_{12} .

Each half-cycle is assumed to consist of m different time intervals (i.e. m time intervals with constant voltages $v_{\rm R}$ applied to \underline{Z}_{12} and constant voltages $v_{\rm AC1}$ applied to \underline{Z}_{13}). For \underline{Z}_{12} , the inductor currents and the capacitor voltages at the end of each time interval are calculated with:

$$\begin{pmatrix} i_{L_{12},i} \\ v_{C_{12},i} \end{pmatrix} = \mathbf{A}_{12,i} \cdot \begin{pmatrix} i_{L_{12},i-1} \\ v_{C_{12},i-1} \end{pmatrix} + \vec{b}_{12,i} \cdot v_{\mathrm{R},i-1}, \quad 1 \le i \le m, \quad (i,m) \in \mathbb{N};$$
(A.88)

 $\mathbf{A}_{12,i}$ and $\vec{b}_{12,i}$ are obtained from (A.83) and (A.84).

The final current and voltage values after one half period with $m (\geq 2)$ different time intervals are:

$$\begin{pmatrix} i_{L_{12},m} \\ v_{C_{12},m} \end{pmatrix} = \mathbf{A}_{\mathbf{p},12,m,1} \cdot \begin{pmatrix} i_{L_{12},0} \\ v_{C_{12},0} \end{pmatrix} + \\ \begin{pmatrix} \sum_{i=1}^{m-1} \mathbf{A}_{\mathbf{p},12,m,i+1} \cdot \vec{b}_{12,i} \cdot v_{\mathbf{R},i-1} \end{pmatrix} + \vec{b}_{12,m} \cdot v_{\mathbf{R},m-1}, \quad (A.89)$$

with

$$\mathbf{A}_{\mathrm{p},12,j,i} = \mathbf{A}_{12,j} \cdot \mathbf{A}_{12,j-1} \cdot \mathbf{A}_{12,j-2} \cdot \dots \mathbf{A}_{12,i+1} \cdot \mathbf{A}_{12,i}.$$
(A.90)

The steady-state solution for $i_{L_{12},0}$ and $v_{C_{12},0}$ is then obtained from

$$\begin{pmatrix} i_{L_{12},m} \\ v_{C_{12},m} \end{pmatrix} = \begin{pmatrix} -i_{L_{12},0} \\ -v_{C_{12},0} \end{pmatrix}$$
(A.91)

and calculated with

$$\begin{pmatrix} i_{L_{12},0} \\ v_{C_{12},0} \end{pmatrix} = (-\mathbf{I} - \mathbf{A}_{\mathbf{p},12,m,1})^{-1} \cdot \\ \left[\left(\sum_{i=1}^{m-1} \mathbf{A}_{\mathbf{p},12,m,i+1} \cdot \vec{b}_{12,i} \cdot v_{\mathbf{R},i-1} \right) + \vec{b}_{12,m} \cdot v_{\mathbf{R},m-1} \right].$$
(A.92)

For the initial inductor current and the initial capacitor voltage regarding the impedance \underline{Z}_{13} an expression similar to (A.92) results, however, $v_{\mathrm{R},i}$, $A_{\mathrm{p},12,j,i}$, and $\vec{b}_{12,i}$ need to be replaced by $v_{\mathrm{AC1},i}$, $A_{\mathrm{p},13,j,i}$, and $\vec{b}_{13,i}$.

A.4.2 HF Capacitor

Compared to the DAB converter, the resonant tank of the LLC converter additionally contains the HF capacitor C, which carries the HV side transformer RMS current I_{L_1} . In order to achieve low losses, film capacitors with a typical dissipation factor tan δ of

$$\tan \delta \approx 1.5 \cdot 10^{-3} \tag{A.93}$$

are used [157]. The corresponding ESR of the capacitor is:

$$R_C = \frac{\tan \delta}{2\pi f C}.\tag{A.94}$$

A.4.3 Design Results, Phase Shift Modulation

The LLC and the DAB converter topologies are almost identical, except for the impedance network employed in the HF path. Thus, the modulation methods investigated for the DAB can also be applied to the LLC converter. However, different transformer currents $i_{L_1}(t)$ and $i_{L_2}(t)$ result and, consequently, a different converter efficiency is obtained.

Compared to the DAB, the design of the LLC converter requires the 4 converter parameters n, L_1 , C, and L_M to be varied⁵ in order to determine the best configuration regarding maximum $\overline{\eta}_{\text{design}}$ [Figure A.35 (a)]. The LLC converter is operated above resonance to achieve low switching losses [35] and thus, it is more convenient to consider Z_0 and ω_0 instead of L_1 and C to design the LLC converter:

$$Z_0 \approx \sqrt{\frac{L_1 + n^2 L_2}{C}}, \qquad \omega_0 \approx \frac{1}{\sqrt{(L_1 + n^2 L_2)C}}.$$
 (A.95)

 $^{{}^{5}}L_{2}$ only considers parasitic inductances and remains approximately constant.

The design algorithm searches for the maximum average efficiency within a discrete number of turns ratios n_i , characteristic impedances $\vec{Z}_{0,j}$, resonance frequencies $\omega_{0,k}$ and magnetizing inductances $L_{M,l}$.⁶

$$\vec{n} = (14 \ 15 \ 16 \ 17 \ 18 \ 19 \ 20 \ 21 \ 22 \ 23 \ 24)^{\mathrm{T}},$$

$$\vec{Z}_{0} = Z_{0,\max}(n,\omega_{0},L_{\mathrm{M}}) \cdot (0.975 \ 0.95 \ 0.925 \ 0.90 \ 0.875 \ 0.85)^{\mathrm{T}},$$

$$\vec{\omega}_{0} = 2\pi f_{\mathrm{S}} \cdot (0.95 \ 0.9 \ 0.85 \ 0.8 \ 0.7 \ 0.6 \ 0.4 \ 0.2 \ 0.1)^{\mathrm{T}},$$

$$\vec{L}_{\mathrm{M}} = (1600\,\mu\mathrm{H} \ 800\,\mu\mathrm{H} \ 400\,\mu\mathrm{H} \ 200\,\mu\mathrm{H} \ 100\,\mu\mathrm{H})^{\mathrm{T}}.$$

(A.96)

With phase shift modulation (cf. Section 3.1.1), the maximum average efficiency, $\overline{\eta}_{\text{design}} = 90.0\%$, is achieved for:

- n = 20,
- $L_1 + n^2 L_2 = 55.4 \,\mu\text{H} \ (\approx L),$
- $C = 93.3 \,\mathrm{nF} \ (\omega_0 = 2\pi \cdot 70 \cdot 10^3 \,\mathrm{s}^{-1}, \ Z_0 = 0.975 Z_{\mathrm{max}} = 24.4 \,\Omega),$
- $L_{\rm M} = 1.6 \,\mathrm{mH}$ (Figure A.36).

The result depicted in Figure A.36 justifies the step sizes selected for \vec{n} , \vec{Z}_0 , $\vec{\omega}_0$, and $\vec{L}_{\rm m}$: around the maximum average efficiency of 90.0% a single step change causes $\overline{\eta}_{\rm design,PS}$ to change by less than 0.2% [around the maximum, the largest change (0.17%) occurs between $\omega_0 = 2\pi \cdot 70 \cdot 10^3 \, {\rm s}^{-1}$ and $\omega_0 = 2\pi \cdot 80 \cdot 10^3 \, {\rm s}^{-1}$ in Figure A.36 (c)].

The average efficiency achieved for the LLC converter is better than the average efficiency achieved for the DAB converter if phase shift modulation is employed (DAB: $\overline{\eta}_{\text{design,PS}} = 89.5\%$, Section A.2.1). It is, however, significantly below the average efficiency of the DAB if optimal modulation parameters are employed (DAB: $\overline{\eta}_{\text{design,opt}} = 93.7\%$).

A.4.4 Design Results, Optimal Modulation

With optimal duty ratios D_1 and D_2 (with respect to minimal losses, Section 5.2.1), the maximum average efficiency of 93.9% is achieved for:

- n = 17,
- $L_1 + n^2 L_2 = 38.3 \,\mu\text{H} \ (\approx L),$

⁶The design considers $Z_0 \leq 0.975 \cdot Z_{0,\max}$ instead of $Z_0 \leq 1.0 \cdot Z_{0,\max}$ in order to maintain a certain margin to the theoretically calculated maximum characteristic impedance $Z_{0,\max}$.

Property	Value	Description			
Maximum Average Efficiency:					
$\overline{\eta}_{ m design}$	90.0%	maximum average converter efficiency			
		calculated with $(A.1)$, $(A.2)$, $(A.3)$,			
		and $(A.4)$			
HV Side Switches	<i>HV Side Switches</i> T_1 , T_2 , T_3 , T_4 ($i = 14$):				
$\max\left(V_{\mathrm{DS},\mathrm{T}_{i},\mathrm{peak}}\right)$	$450\mathrm{V}$	maximum HV switch voltages			
$\max\left(I_{\mathrm{D},\mathrm{T}_{i},\mathrm{peak}}\right)$	$23.2\mathrm{A}$	max. instantaneous HV switch current			
$\max\left(I_{\mathrm{D},\mathrm{T}_{i}}\right)$	10.0 A	maximum HV switch RMS current			
LV Side Switches $\overline{T_5, T_6, T_7, T_8}$ $(i = 58)$:					
$\max\left(V_{\mathrm{DS},\mathrm{T}_{i},\mathrm{peak}}\right)$	$16\mathrm{V}$	maximum LV switch voltages			
$\max\left(I_{\mathrm{D},\mathrm{T}_{i},\mathrm{peak}}\right)$	$465\mathrm{A}$	max. instantaneous LV switch current			
$\max\left(I_{\mathrm{D},\mathrm{T}_{i}}\right)$	$203\mathrm{A}$	maximum LV switch RMS currents			
Transformer:					
n	20	turns ratio			
L_{M}	$1.6\mathrm{mH}$	magn. ind. value (HV side referred)			
$\max\left(I_{\mathrm{AC1}} ight)$	$14.2\mathrm{A}$	HV windings, max. RMS curr.			
$\max\left(I_{\mathrm{AC2}} ight)$	$287\mathrm{A}$	LV windings, max. RMS curr.			
$\max\left(B_{\mathrm{tr,core,peak}}\right)$	$138\mathrm{mT}$	peak transformer flux densities			
HF Inductor L:					
L	$55.4\mu\mathrm{H}$	inductance value			
$\max\left(I_{L,\text{peak}}\right)$	$23.2\mathrm{A}$	max. instantaneous inductor current			
$\max\left(I_L\right)$	$14.2\mathrm{A}$	maximum inductor RMS current			
HF Capacitor C:					
C	$93.3\mathrm{nF}$	capacitance value			
$\max\left(V_{C,\text{peak}}\right)$	$342\mathrm{V}$	max. instantaneous capacitor voltage			
$\max\left(I_C\right)$	$14.2\mathrm{A}$	maximum capacitor RMS current			
$HV Side DC Capacitor C_{DC1}$:					
$\max\left(I_{C_{\mathrm{DC1}}}\right)$	11.4 A	maximum capacitor RMS current			
LV Side DC Capacitor C_{DC2} :					
$\max\left(I_{C_{\rm DC2}}\right)$	$271\mathrm{A}$	maximum capacitor RMS current			

Table A.13: Design summary for the LLC converter operated with phaseshift modulation.



Figure A.36: Average efficiency, $\overline{\eta}_{\text{design,PS}}$, calculated for the LLC converter if phase shift modulation is employed; the maximum average converter efficiency is 90.0% for n = 20, $Z_0 = 0.975 Z_{\text{max}} = 24.4 \Omega$, $\omega_0 = 70 \cdot 10^3 \text{ s}^{-1}$, and $L_{\text{M}} = 400 \,\mu\text{H}$.

- $C = 135 \,\mathrm{nF} \ (\omega_0 = 2\pi \cdot 70 \cdot 10^3 \,\mathrm{s}^{-1}, \ Z_0 = 0.775 Z_{\mathrm{max}} = 16.9 \,\Omega),$
- $L_{\rm M} = 400 \,\mu {\rm H}$ (Figure A.37).

The calculation employs discrete values for n, ω_0 , and $L_{\rm M}$ according to (A.96). However, a higher number of considered impedance values is needed to find the maximum average converter efficiency:

$$\vec{Z}_0 = Z_{0,\max}(n,\omega_0,L_{\rm M}) \cdot (0.975 \quad 0.95 \quad 0.925 \quad 0.90 \quad \dots \quad 0.525 \quad 0.5)^{\rm T}.$$
(A.97)

The result depicted in Figure A.37 justifies the step sizes selected for $\vec{n}, \vec{Z}_0, \vec{\omega}_0$, and $\vec{L}_{\rm m}$: around the maximum average efficiency of 93.9% a single step



Figure A.37: Average efficiency $\overline{\eta}_{\text{design,opt}}$ of the LLC converter, obtained for efficiency optimal modulation parameters D_1 and D_2 and for different n, Z_0 , ω_0 , and L_{M} ; maximum average converter efficiency is 93.9% for n = 17, $\omega_0 = 70 \cdot 10^3 \text{ s}^{-1}$, $Z_0 = 0.775 Z_{\text{max}} = 16.9 \Omega$, and $L_{\text{M}} = 400 \,\mu\text{H}$.

change causes $\overline{\eta}_{\text{design,PS}}$ to change by less than 0.2% [around the maximum, the largest change (0.19%) occurs between $\omega_0 = 2\pi \cdot 70 \cdot 10^3 \,\text{s}^{-1}$ and $\omega_0 = 2\pi \cdot 80 \cdot 10^3 \,\text{s}^{-1}$ in Figure A.37 (c)].

Compared to the DAB converter, only little improvement is achieved: for the DAB, a maximum average efficiency of 93.7% is calculated if optimal modulation parameters are employed; with the suboptimal modulation schemes still $\overline{\eta}_{\text{design}} = 93.5\%$ is achieved for the DAB (Section A.2). Thus, the increased complexity of the LLC converter, the increased inductance required for *L* and the additionally required capacitor *C* render the LLC converter less attractive than the DAB converter.

Property	Value	Description			
Maximum Average Efficiency:					
$\overline{\eta}_{ m design}$	93.9%	maximum average converter efficiency			
		calculated with $(A.1), (A.2), (A.3),$			
		and $(A.4)$			
$HV \ Side \ Switches$	<i>HV Side Switches</i> T_1 , T_2 , T_3 , T_4 (<i>i</i> = 14):				
$\max\left(V_{\mathrm{DS},\mathrm{T}_{i},\mathrm{peak}}\right)$	$450\mathrm{V}$	maximum HV switch voltages			
$\max\left(I_{\mathrm{D},\mathrm{T}_{i},\mathrm{peak}}\right)$	$22.2\mathrm{A}$	max. instantaneous HV switch current			
$\max\left(I_{\mathrm{D},\mathrm{T}_{i}}\right)$	$9.8\mathrm{A}$	maximum HV switch RMS current			
<i>LV Side Switches</i> T_5 , T_6 , T_7 , T_8 ($i = 58$):					
$\max\left(V_{\mathrm{DS},\mathrm{T}_{i},\mathrm{peak}}\right)$	$16\mathrm{V}$	maximum LV switch voltages			
$\max\left(I_{\mathrm{D},\mathrm{T}_{i},\mathrm{peak}}\right)$	$377\mathrm{A}$	max. instantaneous LV switch current			
$\max\left(I_{\mathrm{D},\mathrm{T}_{i}}\right)$	$164\mathrm{A}$	maximum LV switch RMS currents			
Transformer:					
n	17	turns ratio			
L_{M}	$400 \mu H$	magn. ind. value (HV side referred)			
$\max\left(I_{\mathrm{AC1}} ight)$	$13.8\mathrm{A}$	HV windings, max. RMS curr.			
$\max\left(I_{\mathrm{AC2}} ight)$	$231\mathrm{A}$	LV windings, max. RMS curr.			
$\max\left(B_{\rm tr,core,peak}\right)$	$131\mathrm{mT}$	peak transformer flux densities			
HF Inductor L:					
L	$38.4\mu\mathrm{H}$	inductance value			
$\max\left(I_{L,\text{peak}}\right)$	$22.2\mathrm{A}$	max. instantaneous inductor current			
$\max\left(I_L\right)$	$13.8\mathrm{A}$	maximum inductor RMS current			
HF Capacitor C:					
C	$135\mathrm{nF}$	capacitance value			
$\max\left(V_{C,\text{peak}}\right)$	$230\mathrm{V}$	max. instantaneous capacitor voltage			
$\max\left(I_C ight)$	$13.8\mathrm{A}$	maximum capacitor RMS current			
$HV Side DC Capacitor C_{DC1}$:					
$\max\left(I_{C_{\mathrm{DC1}}}\right)$	$9.3\mathrm{A}$	maximum capacitor RMS current			
LV Side DC Capacitor C _{DC2} :					
$\max\left(I_{C_{\rm DC2}}\right)$	$122\mathrm{A}$	maximum capacitor RMS current			

Table A.14: Design summary for the LLC converter operated with optimal duty ratios D_1 and D_2 (with respect to minimum converter losses, cf. Section 5.2.1).

A.5 Full Bridge Converter Design

A.5.1 Mode of Operation

Figure A.38 depicts the bidirectional and isolated full bridge converter with a voltage sourced full bridge on the HV side (using the DC capacitor C_{DC1}) and a current sourced full bridge on the LV side (using the DC inductor L_{DC2}) [135]. The circuit structure of the full bridge converter is not symmetric (as opposed to the DAB) and thus, depending on the actual direction of power transfer, 2 different modulation schemes are employed.

- HV \rightarrow LV (forward mode, Figure A.39): the HV side full bridge applies the AC voltage v_{AC1} to the series connection of the inductance L and the primary side winding of the HF transformer. The LV side full bridge rectifies the secondary side transformer voltage v_{AC2} and the second order filter, formed by the DC inductor L_{DC2} and the DC capacitor C_{DC2} , filters the rectified transformer voltage.
- LV \rightarrow HV (reverse mode, Figure A.40): the LV side full bridge alternately forces the secondary side transformer current i_{AC2} to become equal to either $+i_{L_{DC2}}$ or $-i_{L_{DC2}}$. The HV side full bridge rectifies the primary side transformer current i_{AC1} and the DC capacitor C_{DC1} provides filtering of the output voltage V_1 .

In particular the HF inductance L, needed to achieve soft switching for the MOSFETs T_1 , T_2 , T_3 , and T_4 [143], causes a major difficulty during the reverse mode of operation: then, the LV full bridge repeatedly connects L_{DC2} in series to L which causes voltage spikes on v_{AC2} and large losses. Therefore,



Figure A.38: Bidirectional full bridge converter with LV side DC inductor.



Figure A.39: (a) Simulated voltage and current waveforms of the full bridge converter; (b) simulated currents through the DC inductor $L_{\rm DC2}$ and through the LV switches T_6 and T_7 [the currents through T_5 and T_8 are given with $i_{\rm D,T_5}(t) = i_{\rm D,T_8}(t) = i_{\rm D,T_6}(t - T_{\rm S}/2)$]; (c) employed gate signals on the assumption of zero turn-on time delays and zero turn-off time delays; $V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW, $f_{\rm S} = 100$ kHz.



Figure A.40: (a) Simulated voltage and current waveforms of the full bridge converter; (b) simulated currents through the DC inductor $L_{\rm DC2}$ and through the LV switches T_6 and T_7 [the currents through T_5 and T_8 are given with $i_{\rm D,T_5}(t) = i_{\rm D,T_8}(t) = i_{\rm D,T_6}(t - T_{\rm S}/2)$]; (c) employed gate signals on the assumption of zero turn-on time delays and zero turn-off time delays; $V_1 = 340$ V, $V_2 = 12$ V, $P_1 = -2$ kW, $f_{\rm S} = 100$ kHz.

snubber circuits are typically used to achieve high converter efficiency [135]; in this Section, however, a special modulation scheme is proposed in order to avoid additional circuitry.

Power Transfer from the HV Port to the LV Port (Forward Mode)

Figure A.39 shows typical current and voltage waveforms of the full bridge converter if power is transferred to the LV port ($V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW). At t = 0, the HV full bridge switches from $v_{AC1} \approx 0$ to $v_{AC1} \approx V_1$. During $0 < t < t_1$ the current i_{AC1} rises until it reaches the primary side referred DC inductor current $i_{L_{DC2}}(t)/n$ at $t = t_1$. During the subsequent time interval, $t_1 < t < t_2$, the LV full bridge applies $v_{AC2} \approx V_2$ while $v_{AC1} \approx V_1$ remains and the converter transfers power from the HV side to the LV side. Consequently, the relative duration of the active power transfer time interval, T_{rel} , is:

$$T_{\rm rel} = \frac{t_2 - t_1}{T_{\rm S}/2}.$$
 (A.98)

The respective duty cycle, defined according to Figure A.39, is thus:

$$D = \frac{T_{\rm rel}}{2} = \frac{t_2 - t_1}{T_{\rm S}}.$$
 (A.99)

The remaining time interval of the regarded half-cycle, $t_2 < t < T_S/2$, is the freewheeling time interval and is used to adjust V_2 and to control the actual output power. During the second half-cycle, $T_S/2 < t < T_S$, the transformer voltages and the transformer currents repeat with negative sign.

In summary, during the converter operation in forward mode, each halfcycle consists of 3 different time intervals:

- 1. $0 < t < t_1$: current commutation time interval used to reverse the current in the HF inductance L,
- 2. $t_1 < t < t_2$: active power transfer; similar to a buck converter, the current through the DC inductor L_{DC2} increases,
- 3. $t_2 < t < T_S/2$: freewheeling time interval.

Power Transfer from the LV Port to the HV Port (Reverse Mode)

During reverse mode of operation, at t = 0, the HV full bridge applies $v_{AC1} \approx V_1$. Moreover, the transformer current $i_{AC1}(0)$ needs to be more negative than the current through the DC inductor at t = 0, i.e. $i_{AC1}(0) < i_{L_{DC2}}(0)/n$, in order to achieve low switching losses (Figure A.40). Thus, the absolute value of the transformer current, $|i_{AC1}(t)|$, decreases during $0 < t < t_1$ until it

becomes equal to $|i_{L_{\rm DC2}}(t)/n|$ at $t = t_1$. Since the switches T_6 and T_7 are turned off at t = 0 (i.e. diode operation), the body diodes of T_6 and T_7 stop to conduct at $t = t_1$ (provided that the MOSFETs' body diodes are ideal). During $t_1 < t < t_2$, the current through L remains equal to the primary side referred DC inductor current,⁷ $i_{\rm AC1}(t) = i_{L_{\rm DC2}}(t)/n$, and the converter transfers power from the LV port to the HV port. At $t = t_2$, the switches T_6 and T_7 are turned on and the LV full bridge generates $v_{\rm AC2} \approx 0$. The subsequent time interval, $t_2 < t < t_3$, is the current for the beginning of the next half-cycle. The freewheeling time interval ($t_3 < t < T_{\rm S}/2$) is finally used to adjust V_1 and to control the output power of the converter. Again, the transformer voltages and the transformer currents repeat with negative sign during the second half-cycle ($T_{\rm S}/2 < t < T_{\rm S}$).

The duration of the time interval of active power transfer defines the duty cycle:

$$D = \frac{t_2 - t_1}{T_{\rm S}}.$$
 (A.100)

In summary, during reverse mode of operation, each half-cycle consists of 4 different time intervals:

- 1. 0 < t < t₁: current commutation time interval used to achieve $i_{AC1}(t_1) = i_{LDC2}(t_1)/n$,
- 2. $t_1 < t < t_2$: active power transfer similar to a boost converter,
- 3. $t_2 < t < t_3$: current commutation time interval used to reverse the current through the HF inductance L,
- 4. $t_3 < t < T_S/2$: freewheeling time interval.

A.5.2 Converter Design

According to Figure A.40, the semiconductor and transformer current stresses are higher for reverse mode of operation. Therefore, the converter is designed for power being transferred from the LV port to the HV port. Table A.16 summarizes the design results.

Assumptions and Definitions

In order to simplify the calculations, the impact of conduction and copper losses on the transformer currents i_{AC1} and i_{AC2} is neglected. Consequently,

⁷On the assumption of a negligible transformer magnetizing current.

(A.101)

constant transformer currents are considered during the freewheeling time intervals (e.g. $t_3 < t < T_S/2$). Moreover, during the freewheeling time intervals, the switch currents slightly change (Figure A.40), due to the current ripple on $i_{L_{DC2}}$:

$$i_{\rm D,T_5}(t) = i_{\rm D,T_8}(t) = \frac{i_{\rm AC2}(t) - i_{L_{\rm DC2}}(t)}{2} \approx \frac{i_{\rm AC2}(t) - I_2}{2} \quad \forall \ t_3 < t < T_{\rm S}/2,$$

$$i_{\rm D,T_6}(t) = i_{\rm D,T_7}(t) = \frac{-i_{\rm AC2}(t) - i_{L_{\rm DC2}}(t)}{2} \approx \frac{-i_{\rm AC2}(t) - I_2}{2} \ \forall \ t_3 < t < T_{\rm S}/2.$$
(A.102)

At rated power, the change of the switch currents during the freewheeling interval is comparably small [cf. (A.104)] and thus, its impact on the RMS values of the switch currents is neglected, i.e. the approximations given in (A.101) and (A.102) are used to design the full bridge converter.

Throughout the design, three important current values are needed (defined with respect to Figure A.40):

$$I_{L_{DC2},0} = i_{AC2}(0), I_{L_{DC2},1} = i_{AC2}(t_1), I_{L_{DC2},2} = i_{AC2}(t_2).$$
(A.103)

Moreover, the magnetizing current of the transformer is neglected, i.e. $i_{AC1} = i_{AC2}/n$ is considered.

At rated output power, the maximum peak-to-peak value of the current ripple on $i_{L_{DC2}}(t)$ is assumed to be equal or less than 40% of the output current [116],

$$\Delta I_{L_{\rm DC2},\rm max} = I_{L_{\rm DC2},2} - I_{L_{\rm DC2},1} \le 40\% |I_2| \quad \forall \ P_1 = -2.22 \,\rm kW, \quad (A.104)$$

and thus, the currents $I_{L_{\mathrm{DC2}},1}$ and $I_{L_{\mathrm{DC2}},2}$ are calculated according to:

$$I_{L_{DC2},1} = I_2 - \frac{\Delta I_{L_{DC2}}}{2},$$

$$I_{L_{DC2},2} = I_2 + \frac{\Delta I_{L_{DC2}}}{2}.$$
(A.105)

For the current $I_{L_{DC2},0}$, a value of

$$I_{L_{\rm DC2},0} = 1.2 \cdot I_{L_{\rm DC2},1} \tag{A.106}$$

is assumed in order to ensure low switching losses on the LV side.

The commutation times $T_{C,II}$ and $T_{C,II}$, depicted in Figure A.40, are given with:

$$T_{\rm C,I} = \frac{L}{V_1} \left(\frac{|I_{L_{\rm DC2},0}| - |I_{L_{\rm DC2},1}|}{n} \right), \tag{A.107}$$

$$T_{\rm C,II} = \frac{L}{V_1} \left(\frac{|I_{L_{\rm DC2},0}| + |I_{L_{\rm DC2},2}|}{n} \right).$$
(A.108)

The duration $T_{\rm C,I} + T_{\rm C,II}$ is specified for the maximum voltage $V_2 = V_{2,\rm max} = 16 \,\rm V$ and for the minimum voltage $V_1 = V_{1,\rm min} = 240 \,\rm V$ [there, maximum duty cycle occurs, cf. (A.111)]:⁸

$$T_{\rm C,max} = (T_{\rm C,I} + T_{\rm C,II})_{\rm max} \Big|_{V_1 = V_{1,\min}, V_2 = V_{2,\max}} = 7.5\% T_{\rm S}.$$
(A.109)

For the freewheeling time interval, a minimum duration of $2.5\% T_{\rm S}$ is used:

$$T_{\rm free,min} = T_{\rm S}/2 - t_3 \le 2.5\% T_{\rm S}.$$
 (A.110)

The full bridge converter is designed for an assumed converter efficiency of 90%. Table A.15 summarizes all the assumptions needed to carry out the converter design.

Transformer Turns Ratio

In steady-state operation, the duty cycle defined in Figure A.40 is equal to:

$$D = \frac{nV_2}{2V_1} \tag{A.111}$$

(on the assumptions of a lossless converter and constant DC voltages V_1 and V_2). Thus, the maximum duty cycle occurs for $V_1 = V_{1,\min} = 240$ V and $V_2 = V_{2,\max} = 16$ V; there, (A.111) represents an upper limit to the transformer turns ratio n. Furthermore, the commutation times $T_{C,I}$ and $T_{C,II}$ and the minimum freewheeling time $T_{\text{free,min}}$ need to be included into the calculation of n, since $T_{C,I}$, $T_{C,II}$, and $T_{\text{free,min}}$ reduce the maximum duration of the active power transfer time interval (e.g. $t_1 < t < t_2$ in Figure A.40):

$$D_{\rm max} = 0.5 - f_{\rm S} \cdot (T_{\rm C,I} + T_{\rm C,II} + T_{\rm free,min}).$$
(A.112)

The transformer turns ratio results from (A.111) and (A.112):

$$n = \left\lfloor \frac{V_1}{V_2} \cdot \left[1 - 2f_{\rm S} \cdot (T_{\rm C,I} + T_{\rm C,II} + T_{\rm free,min}) \right] \right\rfloor \Big|_{V_1 = V_{1,\min}, \ V_2 = V_{2,\max}} = 12.$$
(A.113)

⁸For $V_1 = V_{1,\min}$ and $V_2 = V_{2,\max}$, the maximum duty cycle results. With decreasing values V_2 (and constant $V_1 = V_{1,\min}$), the duty cycle decreases [cf. (A.111)], while $T_{C,I}$ and $T_{C,II}$ increase (due to the increasing port current I_2). Within the specified voltage ranges (Table 1.4), however, the maximum value of $(D/f_S) + T_{C,I} + T_{C,II}$ occurs for $V_1 = V_{1,\min}$ and $V_2 = V_{2,\max}$. On the assumptions given in Table A.15, $(D/f_S) + T_{C,I} + T_{C,II} = T_S/2$ would occur for $V_2 \approx 3$ V (provided that $V_1 = V_{1,\min}$).

Property	Assumption	Description
$\eta_{ m est}$	90%	estimated converter efficiency
$\Delta I_{L_{\mathrm{DC2}},\mathrm{max}}$	$40\% I_2 $	maximum peak-to-peak value of the
	$(P_1 = -2 \mathrm{kW}/\eta_{\mathrm{est}})$	current ripple on $i_{L_{DC2}}(t)$
	$\approx -2.22 \mathrm{kW})$	
$I_{L_{\rm DC2},0}$	$1.2 \cdot I_{L_{\rm DC2},1}$	inductor current at $t = 0$, used to achieve
		low switching losses on the LV side
$T_{\rm C,max}$	$7.5\%T_{ m S}$	duration of $T_{I,C} + T_{II,C}$ at $V_1 = 240 \text{ V}$
		and $V_2 = 16 \mathrm{V}$
$T_{\rm free,min}$	$2.5\%T_{ m S}$	minimum duration of the freewheeling
		time interval

Table A.15: Assumptions used to design the full bridge converter.

DC Inductor L_{DC2}

The specified maximum peak-to-peak value of the current ripple, $\Delta I_{L_{DC2},max}$, represents a lower boundary to the inductance value of L_{DC2} , whereas $\Delta I_{L_{DC2}}$ is equal to

$$\Delta I_{L_{\rm DC2}} = \left(\frac{V_1}{n} - V_2\right) \frac{D}{f_{\rm S} L_{\rm DC2}} \le 40\% \frac{|P_1|}{V_2}.$$
 (A.114)

Thus, the inductance value of L_{DC2} is determined with:

$$L_{\rm DC2} = \min\left[\left(\frac{V_1}{n} - V_2\right) \frac{D}{40\% \frac{|P_1|}{V_2} f_{\rm S}}\right] \begin{vmatrix} V_{1,\min} \le V_1 \le V_{1,\max}, \\ V_{2,\min} \le V_2 \le V_{2,\max}, \\ P_1 = -2.22 \,\rm{kW} \end{vmatrix} = 0.83 \,\mu\rm{H}.$$
(A.115)

The peak current through $L_{\rm DC2}$ is obtained from

$$\min(i_{L_{\rm DC2}}) = \min\left(\frac{P_1}{V_2} - \frac{\Delta I_{L_{\rm DC2}}}{2}\right) = -226\,\mathrm{A} \tag{A.116}$$

in reverse mode of operation [accordingly, $\max(i_{L_{DC2}}) = 226$ A applies in forward mode]. The respective maximum RMS current is

$$\max(I_{L_{\rm DC2}}) = \max\left[\sqrt{\left(\frac{P_1}{V_2}\right)^2 + \left(\frac{\Delta I_{L_{\rm DC2}}}{2\sqrt{3}}\right)^2}\right] = 202 \,\mathrm{A}.\tag{A.117}$$

HF Inductance L

The HF inductance L is calculated with (A.107), (A.108), and with the specified maximum commutation time $T_{C,max}$:

$$L = \frac{T_{\rm C,max}}{\left[(T_{\rm C,I} + T_{\rm C,II})/L \right] \Big|_{V_1 = V_{1,min}, V_2 = V_{2,max}, P_1 = -2.22 \text{ kW}}} = 6.4 \,\mu\text{H.} \quad (A.118)$$

The maximum peak inductor current, $\max(I_{L,\text{peak}})$, is determined for $P_1 = -2.22 \text{ kW}$ within the specified input and output voltage ranges:

$$\max(I_{L,\text{peak}}) = \max\left(\frac{|I_{L_{\text{DC2}},0}|}{n}\right) = 22.6 \text{ A.}$$
 (A.119)

Transformer Currents

The primary side transformer RMS current is obtained from:

$$I_{AC1} = \left\{ \frac{2f_{S}}{n^{2}} \left[\frac{T_{C,I}}{3} \left(I_{L_{DC2},0}^{2} + I_{L_{DC2},0} I_{L_{DC2},1} + I_{L_{DC2},1}^{2} \right) + \frac{D}{3f_{S}} \left(I_{L_{DC2},1}^{2} + I_{L_{DC2},1} I_{L_{DC2},2} + I_{L_{DC2},2}^{2} \right) + \frac{T_{C,II}}{3} \left(I_{L_{DC2},0}^{2} - I_{L_{DC2},0} I_{L_{DC2},2} + I_{L_{DC2},2}^{2} \right) + \frac{0.5 - D - f_{S} \left(T_{C,I} + T_{C,II} \right)}{f_{S}} I_{L_{DC2},0}^{2} \right] \right\}^{\frac{1}{2}}.$$
 (A.120)

The maximum values of the primary and secondary side transformer RMS currents are:

$$\max(I_{AC1}) = 20.0 \text{ A},$$

$$\max(I_{AC2}) = n \max(I_{AC1}) = 241 \text{ A}.$$
 (A.121)

Transformer Peak Flux Density

The peak flux density in the transformer core is:

$$B_{\rm tr,core,peak} = \frac{V_1}{N_1} \frac{D}{2f_{\rm S}A_{\rm tr,core}}$$
(A.122)

(on the assumption of a homogeneously distributed magnetic flux inside the transformer core; stray flux is neglected).

HV Side Switch Currents and Voltages

On the HV side, the peak switch voltages are equal to the maximum port voltage,

$$I_{D,T_1,\max} = I_{D,T_2,\max} = I_{D,T_3,\max} = I_{D,T_4,\max} = \frac{\max(I_{AC1})}{\sqrt{2}} = 14.2 \text{ A},$$

(A.124)
and the peak switch currents are equal to $\max(I_{L,\text{peak}}) = 22.6 \text{ A}$ [cf. (A.119)].

LV Side Switch Currents and Voltages

On the assumption of negligible overvoltage spikes, the peak switch voltages applied to the LV side switches are obtained from:

$$V_{\text{DS},\text{T}_{5},\text{peak,max}} = V_{\text{DS},\text{T}_{6},\text{peak,max}} = V_{\text{DS},\text{T}_{7},\text{peak,max}} = V_{\text{DS},\text{T}_{8},\text{peak,max}} = \frac{450 \text{ V}}{n} = 37.5 \text{ V}.$$
 (A.125)

The calculation of the switch RMS currents requires 4 interim current values (cf. Figure A.40),

$$I_{\rm D,T_{LV},1} = -I_{L_{\rm DC2},0} + \frac{I_{L_{\rm DC2},0} - P_1/V_2}{2} \ \left[\approx i_{\rm D,T_6}(t_3) \approx i_{\rm D,T_6}(T_{\rm S}/2)\right],\tag{A.126}$$

$$I_{\rm D,T_{\rm LV},2} = i_{\rm D,T_6}(t_5) = -I_{L_{\rm DC2},1},\tag{A.127}$$

$$I_{\rm D,T_{\rm LV},3} = i_{\rm D,T_6}(t_6) = -I_{L_{\rm DC2},2},\tag{A.128}$$

$$I_{\rm D,T_{LV},4} = \frac{I_{L_{\rm DC2},0} - P_1/V_2}{2} \ [\approx i_{\rm D,T_6}(t_7) \approx i_{\rm D,T_6}(T_{\rm S})]; \tag{A.129}$$
the switch RMS currents are:

$$\begin{split} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}}} &= I_{\mathrm{D},\mathrm{T}_{5}} = I_{\mathrm{D},\mathrm{T}_{6}} = I_{\mathrm{D},\mathrm{T}_{7}} = I_{\mathrm{D},\mathrm{T}_{8}} = \\ &\left\{ f_{\mathrm{S}} \left[\frac{T_{\mathrm{C},\mathrm{II}}}{3} \left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},1}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},3}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},3} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},4} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},4}^{2} \right) + \\ & \frac{0.5 - D - f_{\mathrm{S}} \left(T_{\mathrm{C},\mathrm{I}} + T_{\mathrm{C},\mathrm{II}} \right)}{f_{\mathrm{S}}} \left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},1}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},4}^{2} \right) + \\ & \frac{T_{\mathrm{C},\mathrm{I}}}{3} \left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},1}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},1} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},4}^{2} \right) + \\ & \frac{D}{3 f_{\mathrm{S}}} \left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},3} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},3}^{2} \right) \right] \right\}^{\frac{1}{2}}. \quad (\mathrm{A}.130) \end{split}$$

Within the specified input and output voltage ranges and for $P_1 = -2.22$ kW, the maximum switch RMS currents are:

$$I_{\rm D,T_{LV},max} = \max(I_{\rm D,T_{LV}}) = 157 \,\mathrm{A.}$$
 (A.131)

HV Side Capacitor AC Current

On the assumption of a constant HV port current I_1 , the RMS value of the current through C_{DC1} is:

$$I_{C_{\text{DC1}}} = \left\{ \frac{2f_{\text{S}}}{n^2} \left[\frac{T_{\text{C,I}}}{3} \left(I_{L_{\text{DC2}},0}^2 + I_{L_{\text{DC2}},0} I_{L_{\text{DC2}},1} + I_{L_{\text{DC2}},1}^2 \right) + \frac{D}{3f_{\text{S}}} \left(I_{L_{\text{DC2}},1}^2 + I_{L_{\text{DC2}},1} I_{L_{\text{DC2}},2} + I_{L_{\text{DC2}},2}^2 \right) + \frac{T_{\text{C,II}}}{3} \left(I_{L_{\text{DC2}},0}^2 - I_{L_{\text{DC2}},0} I_{L_{\text{DC2}},2} + I_{L_{\text{DC2}},2}^2 \right) \right] - \left(\frac{P_1}{V_1} \right)^2 \right\}^{\frac{1}{2}}.$$
 (A.132)

For $V_1 = 240$ V, $V_2 = 11$ V, and $P_1 = -2.22$ kW the maximum capacitor current occurs:

$$I_{C_{\rm DC1},\rm max} = 10.1 \,\rm A.$$
 (A.133)

LV Side Capacitor AC Current

On the assumption of a constant LV port current I_2 the RMS value of the current through C_{DC2} is:

$$I_{C_{\rm DC2}} = \frac{\Delta I_{L_{\rm DC2}}}{2\sqrt{3}};$$
 (A.134)

the maximum value of $I_{C_{DC2}}$ is:

$$I_{C_{\rm DC2},\rm max} = 16 \,\rm A.$$
 (A.135)

Table A.16 summarizes the design results.

A.6 Current Doubler Converter Design

A.6.1 Mode of Operation

Figure A.41 depicts the bidirectional and isolated current doubler topology [144]. Similar to the bidirectional full bridge converter, the actually employed modulation scheme depends on the direction of power transfer.

- HV \rightarrow LV (forward mode, Figure A.42): the HV side full bridge applies the AC voltage v_{AC1} to the series connection of the inductance L and the HF transformer. The diodes on the LV side (operated using synchronous rectification) rectify the secondary side transformer voltage v_{AC2} . Finally, the output filter, formed with the DC inductors L_{DC2a} and L_{DC2b} and the DC capacitor C_{DC2} , filters the rectified transformer voltage.
- LV \rightarrow HV (*reverse mode*, Figure A.43): the LV side switches alternately force the secondary side transformer current i_{AC2} to become equal to either $+i_{L_{DC2a}}$ or $-i_{L_{DC2b}}$. The HV side full bridge rectifies the primary side transformer current i_{AC1} and the DC capacitor C_{DC1} provides filtering of the output voltage V_1 .



Figure A.41: Bidirectional current doubler topology with LV side DC inductors.

Property	Value	Description			
HV Side Switches T_1 , T_2 , T_3 , T_4 :					
$\max\left(V_{\mathrm{DS,T_{HV},peak}}\right)$	$450\mathrm{V}$	maximum HV switch voltages			
$\max\left(I_{\mathrm{D},\mathrm{T}_{\mathrm{HV}},\mathrm{peak}}\right)$	$22.6\mathrm{A}$	max. HV switch curr. (instantaneous value)			
$\max\left(I_{\mathrm{D,T}_{\mathrm{HV}}}\right)$	$14.2\mathrm{A}$	maximum HV switch currents (RMS value)			
LV Side Switches T	$_{5}, T_{6}, T_{7}$, T ₈ :			
$\max\left(V_{\rm DS,T_{\rm LV},\rm peak}\right)$	$37.5\mathrm{V}$	maximum LV switch voltages			
$\max\left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},\mathrm{peak}}\right)$	$236\mathrm{A}$	max. LV switch curr. (instantaneous value)			
$\max\left(I_{\mathrm{D,T_{LV}}}\right)$	$157\mathrm{A}$	maximum LV switch currents (RMS value)			
Transformer:					
n	12	turns ratio			
$\max\left(I_{ m AC1} ight)$	$10.0\mathrm{A}$	HV transformer winding, max. RMS current			
$\max\left(I_{\mathrm{AC2}} ight)$	$241\mathrm{A}$	LV transformer winding, max. RMS current			
$\max\left(B_{\mathrm{tr,core,peak}}\right)$	$130\mathrm{mT}$	for $N_1 = 12$, $N_2 = 1$, and $A_{tr,core} = 308 \text{ mm}^2$			
		(ELP 58 core, cf. Section $A.2.4$)			
DC Inductor L_{DC2} :					
$L_{\rm DC2}$	$0.83\mu\mathrm{H}$	inductance value			
$\max\left(I_{L_{\rm DC2,peak}}\right)$	$226\mathrm{A}$	maximum instantaneous inductor current			
$\max\left(I_{L_{\rm DC2}}\right)$	$202\mathrm{A}$	maximum inductor RMS current			
HF Inductor L:					
L	$6.4\mu\mathrm{H}$	inductance value			
$\max\left(I_{L_{\rm DC2,peak}}\right)$	$22.6\mathrm{A}$	maximum instantaneous inductor current			
$\max\left(I_{L_{\rm DC2}}\right)$	$20.0\mathrm{A}$	maximum inductor RMS current			
HV Side DC Capacitor $C_{\rm DC1}$:					
$\max\left(I_{C_{\rm DC1}}\right)$	10.1 A	maximum capacitor RMS current			
LV Side DC Capacitor C_{DC2} :					
$\max\left(I_{C_{\rm DC2}}\right)$	16 A	maximum capacitor RMS current			

 Table A.16: Design summary for the bidirectional full bridge converter.



Figure A.42: (a) Simulated voltage and current waveforms of the bidirectional current doubler DC–DC converter; (b) simulated currents through the DC inductors L_{DC2a} and L_{DC2b} and through the LV switch T_5 [the current through T_6 is given with $i_{D,T_6}(t) = i_{D,T_5}(t-T_S/2)$]; (c) employed gate signals for the assumption of zero turn-on time delays and zero turn-off time delays; $V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW, $f_S = 100$ kHz.



Figure A.43: (a) Simulated voltage and current waveforms of the bidirectional current doubler DC–DC converter; (b) simulated currents through the DC inductors L_{DC2a} and L_{DC2b} and through the LV switch T_5 [the current through T_6 is given with $i_{D,T_6}(t) = i_{D,T_5}(t-T_S/2)$]; (c) employed gate signals on the assumption of zero turn-on time delays and zero turn-off time delays; $V_1 = 340 \text{ V}, V_2 = 12 \text{ V}, P_1 = -2 \text{ kW}, f_S = 100 \text{ kHz}.$

The employed modulation schemes are identical to the modulation schemes used for the bidirectional full bridge converter (Appendix A.5.1).

A.6.2 Converter Design

According to Figure A.43, the semiconductor and transformer current stresses are higher for reverse mode of operation. Therefore, the converter is designed for power being transferred from the LV port to the HV port. Table A.18 summarizes the design results.

Assumptions and Definitions

In order to simplify the calculations, the impact of conduction and copper losses on the transformer currents i_{AC1} and i_{AC2} is neglected. Consequently, constant transformer currents are considered during the freewheeling time intervals (e.g. $t_3 < t < T_S/2$).

Throughout the design, three important current values are needed (defined with respect to Figure A.43):

$$I_{L_{DC2a,0}} = i_{AC1}(0),$$

$$I_{L_{DC2a,1}} = i_{AC1}(t_1),$$

$$I_{L_{DC2a,2}} = i_{AC1}(t_2).$$
(A.136)

Moreover, the magnetizing current of the transformer is neglected, i.e. $i_{AC1} = i_{AC2}/n$ is considered.

The peak-to-peak value of the current ripple on $i_{L_{DC2a}}(t)$ [and on $i_{L_{DC2b}}(t)$] is equal to

$$\Delta I_{L_{\rm DC2a}} = \Delta I_{L_{\rm DC2b}} = I_{L_{\rm DC2a},2} - I_{L_{\rm DC2a},1} \tag{A.137}$$

and thus, the currents $I_{L_{DC2a},1}$ and $I_{L_{DC2a},2}$ are calculated according to:

$$I_{L_{\rm DC2a},1} = \frac{I_2}{2} - \frac{\Delta I_{L_{\rm DC2a}}}{2},$$

$$I_{L_{\rm DC2a},2} = \frac{I_2}{2} + \frac{\Delta I_{L_{\rm DC2a}}}{2}.$$
(A.138)

At rated output power, the maximum peak-to-peak value of the current ripple on $i_{L_{DC2a}}(t) + i_{L_{DC2b}}(t)$ is assumed to be equal or less than 40% of the output current,

$$\Delta I_{L_{\text{DC2a+b}},\text{max}} = \max \left[i_{L_{\text{DC2a}}}(t) + i_{L_{\text{DC2b}}}(t) \right] - \min \left[i_{L_{\text{DC2a}}}(t) + i_{L_{\text{DC2b}}}(t) \right]$$

$$\leq 40\% \left| I_2 \right| \quad \forall \quad P_1 = -2.22 \,\text{kW}. \quad (A.139)$$

For the current $I_{L_{DC2a},0}$, a value of

$$I_{L_{\rm DC2a},0} = 1.2 \cdot I_{L_{\rm DC2a},1} \tag{A.140}$$

is assumed in order to ensure low switching losses on the LV side.

The commutation times $T_{C,I}$ and $T_{C,II}$ depicted in Figure A.43 are given with:

$$T_{\rm C,I} = \frac{L}{V_1} \left(\frac{|I_{L_{\rm DC2a},0}| - |I_{L_{\rm DC2a},1}|}{n} \right), \tag{A.141}$$

$$T_{\rm C,II} = \frac{L}{V_1} \left(\frac{|I_{L_{\rm DC2a},0}| + |I_{L_{\rm DC2a},2}|}{n} \right).$$
(A.142)

The duration $T_{C,I} + T_{C,II}$ is specified for the maximum voltage $V_2 = V_{2,max} = 16 \text{ V}$ and the minimum voltage $V_1 = V_{1,min} = 240 \text{ V}$ [there, the maximum duty cycle occurs, cf. (A.145) and Appendix A.5.2]:

$$T_{\rm C,max} = (T_{\rm C,I} + T_{\rm C,II})_{\rm max} \Big|_{V_1 = V_{1,\min}, V_2 = V_{2,\max}} = 7.5\% T_{\rm S}.$$
(A.143)

For the freewheeling time interval, a minimum duration of $2.5\% T_{\rm S}$ is used:

$$T_{\rm free,min} = T_{\rm S}/2 - t_3 \le 2.5\% T_{\rm S}.$$
 (A.144)

The bidirectional current doubler DC–DC converter is designed for an assumed converter efficiency of 90%. Table A.17 summarizes all the assumptions needed to carry out the converter design.

Transformer Turns Ratio

In steady-state operation, the duty cycle defined in Figure A.43 is equal to:

$$D = \frac{nV_2}{V_1} \tag{A.145}$$

(on the assumptions of a lossless converter and constant DC voltages V_1 and V_2). Thus, the maximum duty cycle occurs for $V_1 = V_{1,\min} = 240$ V and $V_2 = V_{2,\max} = 16$ V; there, (A.145) represents an upper limit to the transformer turns ratio n. Furthermore, the commutation times $T_{C,I}$ and $T_{C,II}$ and the minimum freewheeling time $T_{\text{free,min}}$ need to be included into the calculation of n, since $T_{C,I}$, $T_{C,II}$, and $T_{\text{free,min}}$ reduce the maximum duration of the active power transfer time interval (e.g. $t_1 < t < t_2$ in Figure A.43):

$$D_{\rm max} = 0.5 - f_{\rm S} \cdot (T_{\rm C,I} + T_{\rm C,II} + T_{\rm free,min}).$$
(A.146)

The transformer turns ratio results from (A.145) and (A.146):

$$n = \left\lfloor \frac{V_1}{2V_2} \cdot \left[1 - 2f_{\rm S} \cdot (T_{\rm C,I} + T_{\rm C,II} + T_{\rm free,min}) \right] \right\rfloor \Big|_{V_1 = V_{1,\min}, \ V_2 = V_{2,\max}} = 6.$$
(A.147)

Property	Assumption	Description
$\eta_{ m est}$	90%	estimated converter efficiency
$\Delta I_{L_{\rm DC2a+b},\rm max}$	$40\% I_2 $	maximum peak-to-peak value of the
	$(P_1 = -2 \mathrm{kW}/\eta_{\mathrm{est}})$	current ripple on $i_{L_{DC2a}}(t) + i_{L_{DC2b}}(t)$
	$\approx -2.22 \mathrm{kW}$	
$I_{L_{\rm DC2a},0}$	$1.2 \cdot I_{L_{\mathrm{DC2a}},1}$	ind. current at $t = 0$, used to achieve
		low switching losses on the LV side
$T_{ m C,max}$	$7.5\%T_{ m S}$	duration of $T_{I,C} + T_{II,C}$ at $V_1 = 240 \text{ V}$
		and $V_2 = 16 \mathrm{V}$
$T_{\rm free,min}$	$2.5\%T_{ m S}$	minimum duration of the freewheeling
		time interval

 Table A.17: Assumptions used to design the bidirectional current doubler

 DC-DC converter.

DC Inductor L_{DC2}

The specified maximum peak-to-peak value of the current ripple, $\Delta I_{L_{\text{DC2a+b},\text{max}}}$, represents a lower boundary to the inductance values of L_{DC2a} and L_{DC2b} , whereas $\Delta I_{L_{\text{DC2a+b}}}$ is equal to

$$\Delta I_{L_{\rm DC2a+b}} = \left(\frac{V_1}{n} - V_2\right) \frac{D}{f_{\rm S} L_{\rm DC2a}} - V_2 \frac{D}{f_{\rm S} L_{\rm DC2b}} \le 40\% \frac{|P_1|}{V_2}.$$
 (A.148)

Provided that $L_{DC2a} = L_{DC2b}$ applies, the inductance value of L_{DC2a} can be determined using:

$$L_{\rm DC2a} = \min\left[\left(\frac{V_1}{n} - 2V_2\right) \frac{D}{40\% \frac{|P_1|}{V_2} f_{\rm S}}\right] \left| \begin{array}{l} V_{1,\min} \le V_1 \le V_{1,\max}, \\ V_{2,\min} \le V_2 \le V_{2,\max}, \\ P_1 = -2.22 \,\rm{kW} \end{array} \right| = 1.7 \,\mu\rm{H}.$$
(A.149)

The peak currents through $L_{\rm DC2a}$ and $L_{\rm DC2b}$ are obtained from

$$\min(i_{L_{\rm DC2a}}) = \min(i_{L_{\rm DC2b}}) = \min\left(\frac{P_1}{V_2} - \frac{\Delta I_{L_{\rm DC2a}}}{2}\right) = -129\,\text{A} \quad (A.150)$$

in reverse mode of operation [accordingly, $\max(i_{L_{DC2a}}) = \max(i_{L_{DC2b}}) =$

129 A applies in forward mode]. The respective maximum RMS current is

$$\max(I_{L_{\rm DC2a}}) = \max(I_{L_{\rm DC2b}}) = \max\left[\sqrt{\left(\frac{P_1}{2V_2}\right)^2 + \left(\frac{\Delta I_{L_{\rm DC2a}}}{2\sqrt{3}}\right)^2}\right] = 102 \,\mathrm{A}.$$
(A.151)

HF Inductance L, Transformer Currents

The calculation of the HF inductance L and of the transformer RMS currents is identical to the calculation presented in Appendix A.5.2. The results are:

$$L = 6.1 \,\mu\text{H},$$
 (A.152)

$$\max(I_{L,\text{peak}}) = 25.9 \text{ A},$$
 (A.153)

$$\max(I_{\rm AC1}) = 22.5 \,\mathrm{A},\tag{A.154}$$

$$\max(I_{AC2}) = n \max(I_{AC1}) = 135 \text{ A.}$$
(A.155)

Transformer Peak Flux Density

The peak flux density in the transformer core is calculated with (A.122).

HV Side Switch Currents and Voltages

On the HV side, the peak switch voltages are equal to the maximum port voltage,

$$V_{\text{DS},\text{T}_1,\text{peak},\text{max}} = V_{\text{DS},\text{T}_2,\text{peak},\text{max}} = V_{\text{DS},\text{T}_3,\text{peak},\text{max}} = V_{\text{DS},\text{T}_4,\text{peak},\text{max}} = 450 \text{ V}.$$
(A.156)

The maximum switch RMS currents are calculated according to (4.1),

$$I_{\rm D,T_1,max} = I_{\rm D,T_2,max} = I_{\rm D,T_3,max} = I_{\rm D,T_4,max} = \frac{\max\left(I_{\rm AC1}\right)}{\sqrt{2}} = 15.9 \,\rm A,$$
(A.157)

and the peak switch currents are [cf. (A.153)]:

$$I_{D,T_1,\text{peak},\text{max}} = I_{D,T_2,\text{peak},\text{max}} = I_{D,T_3,\text{peak},\text{max}} = I_{D,T_4,\text{peak},\text{max}} = \max(I_{L,\text{peak}}) = 25.9 \text{ A.} \quad (A.158)$$

LV Side Switch Currents and Voltages

On the assumption of negligible overvoltage spikes, the peak switch voltages applied to the LV side switches are obtained from:

$$V_{\text{DS},\text{T}_5,\text{peak,max}} = V_{\text{DS},\text{T}_6,\text{peak,max}} = \frac{450 \text{ V}}{n} = 75 \text{ V}.$$
 (A.159)

The calculation of the switch RMS currents requires 6 interim current values (cf. Figure A.40),

$$I_{\rm D,T_{LV},1} = i_{\rm D,T_5}(t_3) = -I_{L_{\rm DC2a},0} - I_{L_{\rm DC2a},2} + \frac{V_2}{L_{\rm DC2a}} T_{\rm C,II},$$
(A.160)

$$I_{\rm D,T_{LV,2}} = i_{\rm D,T_5}(t_4) = -I_{L_{\rm DC2a},0} - I_{L_{\rm DC2a},2} + \frac{V_2}{L_{\rm DC2a}} \frac{1 - 2D - 2f_{\rm S} T_{\rm C,I}}{2f_{\rm S}},$$
(A.161)

$$I_{\rm D,T_{\rm LV},3} = i_{\rm D,T_5}(t_5) = -2I_{L_{\rm DC2a},1} - \frac{V_2}{L_{\rm DC2a}} \frac{1}{2f_{\rm S}},\tag{A.162}$$

$$I_{\rm D,T_{LV},4} = i_{\rm D,T_5}(t_6) = -I_{L_{\rm DC2a},2} - I_{L_{\rm DC2a},1} - \frac{V_2}{L_{\rm DC2a}} \frac{1-2D}{2f_{\rm S}},\qquad(A.163)$$

$$I_{\rm D,T_{\rm LV},5} = i_{\rm D,T_5}(t_7) = I_{L_{\rm DC2a},0} - I_{L_{\rm DC2a},1} - \frac{V_2}{L_{\rm DC2a}} \frac{D + f_{\rm S} T_{\rm C,I}}{f_{\rm S}}, \quad (A.164)$$

$$I_{\rm D,T_{\rm LV},6} = i_{\rm D,T_5}(t_8) = I_{L_{\rm DC2a},0} - I_{L_{\rm DC2a},1} - \frac{V_2}{L_{\rm DC2a}} T_{\rm C,I};$$
(A.165)

the switch RMS currents are:

$$\begin{split} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}}} &= I_{\mathrm{D},\mathrm{T}_{6}} = \\ & \left\{ f_{\mathrm{S}} \left[\frac{T_{\mathrm{C},\mathrm{II}}}{3} \left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},1}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},4}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},4} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},5} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},5}^{2} \right) + \\ \frac{0.5 - D - f_{\mathrm{S}} \left(T_{\mathrm{C},\mathrm{I}} + T_{\mathrm{C},\mathrm{II}} \right)}{3 f_{\mathrm{S}}} \left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},1}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},1} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},5} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},6} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},6}^{2} \right) + \\ \frac{T_{\mathrm{C},\mathrm{I}}}{3} \left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},2} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},3} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},3}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},6}^{2} \right) + \\ & \frac{D}{3 f_{\mathrm{S}}} \left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},3}^{2} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},3} I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},4} + I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},4}^{2} \right) \right] \right\}^{\frac{1}{2}}. \quad (A.166)$$

Within the specified input and output voltage ranges and for $P_1 = -2.22$ kW, the maximum switch RMS currents are:

$$I_{\rm D,T_{LV},max} = \max(I_{\rm D,T_{LV}}) = 159 \,\mathrm{A}.$$
 (A.167)

HV Side and LV Side Capacitor AC Currents

The calculation of the capacitor AC currents is identical to the calculation presented in Appendix A.5.2 [solely, in (A.134), $\Delta I_{L_{DC2}}$ needs to be replaced by $\Delta I_{L_{DC2a+b}}$]. The results are:

$$I_{C_{\rm DC1},\rm max} = 10.7 \,\rm A,$$
 (A.168)

$$I_{C_{\rm DC2},\rm max} = 16 \,\rm A.$$
 (A.169)

Property	Value	Description			
HV Side Switches T_1 , T_2 , T_3 , T_4 :					
$\max\left(V_{\mathrm{DS,T_{HV},peak}}\right)$	$450\mathrm{V}$	maximum HV switch voltages			
$\max\left(I_{\mathrm{D},\mathrm{T}_{\mathrm{HV}},\mathrm{peak}}\right)$	$25.9\mathrm{A}$	max. HV switch curr. (instantaneous value)			
$\max\left(I_{\mathrm{D,T}_{\mathrm{HV}}}\right)$	$15.9\mathrm{A}$	maximum HV switch currents (RMS value)			
LV Side Switches T	$_{5}, T_{6}:$				
$\max\left(V_{\rm DS,T_{\rm LV},\rm peak}\right)$	$75\mathrm{V}$	maximum LV switch voltages			
$\max\left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},\mathrm{peak}}\right)$	$251\mathrm{A}$	max. LV switch curr. (instantaneous value)			
$\max\left(I_{\mathrm{D,T_{LV}}}\right)$	$159\mathrm{A}$	maximum LV switch currents (RMS value)			
Transformer:					
n	6	turns ratio			
$\max\left(I_{\mathrm{AC1}} ight)$	$22.5\mathrm{A}$	HV transformer winding, max. RMS current			
$\max\left(I_{\mathrm{AC2}} ight)$	$135\mathrm{A}$	LV transformer winding, max. RMS current			
$\max\left(B_{\mathrm{tr,core,peak}}\right)$	$130\mathrm{mT}$	for $N_1 = 12$, $N_2 = 2$, and $A_{tr,core} = 308 \text{ mm}^2$			
		(ELP 58 core, cf. Section A.2.4)			
DC Inductor L_{DC2} :					
$L_{\rm DC2}$	$1.7\mu\mathrm{H}$	inductance value			
$\max\left(I_{L_{\rm DC2,peak}}\right)$	$129\mathrm{A}$	maximum instantaneous inductor current			
$\max\left(I_{L_{\rm DC2}}\right)$	$102\mathrm{A}$	maximum inductor RMS current			
HF Inductor L:	HF Inductor L:				
L	$6.1\mu\mathrm{H}$	inductance value			
$\max\left(I_{L_{\mathrm{DC2,peak}}}\right)$	$25.9\mathrm{A}$	maximum instantaneous inductor current			
$\max\left(I_{L_{\mathrm{DC2}}}\right)$	$22.5\mathrm{A}$	maximum inductor RMS current			
HV Side DC Capacitor $C_{\rm DC1}$:					
$\max\left(I_{C_{\rm DC1}}\right)$	$10.7\mathrm{A}$	maximum capacitor RMS current			
$LV Side DC Capacitor C_{DC2}$:					
$\max\left(I_{C_{\rm DC2}}\right)$	16 A	maximum capacitor RMS current			

 Table A.18: Design summary for the bidirectional current doubler topology.

A.7 Push-Pull Converter Design

A.7.1 Mode of Operation

Figure A.44 depicts the bidirectional and isolated push-pull converter. Again, the actually employed modulation scheme depends on the direction of power transfer (cf. Appendix A.5.1):

- HV \rightarrow LV (forward mode, Figure A.45): the HV side full bridge applies the AC voltage v_{AC1} to the series connection of the inductance L and the HF transformer. The diodes on the LV side (operated using synchronous rectification) rectify the secondary side transformer voltage $v_{AC2a} - v_{AC2b}$ and the output filter formed with the DC inductor L_{DC2} and the DC capacitor C_{DC2} filters the rectified transformer voltage.
- LV \rightarrow HV (*reverse mode*, Figure A.46): the LV side switches alternately force one of the secondary side transformer currents to become equal to $i_{L_{DC2}}$ (i.e. either $i_{AC2a} = i_{L_{DC2}}$ or $i_{AC2b} = i_{L_{DC2}}$). The HV side full bridge rectifies the primary side transformer current i_{AC1} and the DC capacitor C_{DC1} provides filtering of the output voltage V_1 .

The employed modulation schemes are identical to the modulation schemes used for the bidirectional full bridge converter (detailed in Appendix A.5.1).



Figure A.44: Bidirectional push-pull converter with LV side DC inductor.



Figure A.45: (a) Simulated voltage and current waveforms of the bidirectional push-pull converter; (b) simulated currents through the DC inductor $L_{\rm DC2}$ and through the LV switch T_5 [the current through T_6 is given with $i_{\rm D,T_6}(t) = i_{\rm D,T_5}(t - T_{\rm S}/2)$]; (c) employed gate signals on the assumption of zero turn-on time delays and zero turn-off time delays; $V_1 = 340$ V, $V_2 = 12$ V, $P_2 = 2$ kW, $f_{\rm S} = 100$ kHz.



Figure A.46: (a) Simulated voltage and current waveforms of the bidirectional push-pull converter; (b) simulated currents through the DC inductor $L_{\rm DC2}$ and through the LV switch T_5 [the current through T_6 is given with $i_{\rm D,T_6}(t) = i_{\rm D,T_5}(t - T_{\rm S}/2)$]; (c) employed gate signals on the assumption of zero turn-on time delays and zero turn-off time delays; $V_1 = 340$ V, $V_2 = 12$ V, $P_1 = -2$ kW, $f_{\rm S} = 100$ kHz.

A.7.2 Converter Design

The design of the push-pull converter is almost identical to the design of the full bridge converter detailed in Appendix A.5.2 except for 2 differences:

- 1. The transformer of the push-pull converter employs 2 windings on the LV side (however, the turns ratio n is the same as for the full bridge converter).
- 2. The maximum voltage applied to the switches T_5 and T_6 of the pushpull converter is (approximately) twice the maximum voltage applied to the LV side switches of the full bridge converter.

Table A.19 summarizes the design results obtained for the push-pull converter.

Property	Value	Description		
HV Side Switches T ₁ , T ₂ , T ₃ , T ₄ :				
$\max\left(V_{\mathrm{DS,T}_{\mathrm{HV}},\mathrm{peak}}\right)$	$450\mathrm{V}$	maximum HV switch voltages		
$\max\left(I_{\mathrm{D},\mathrm{T}_{\mathrm{HV}},\mathrm{peak}}\right)$	$22.6\mathrm{A}$	max. HV switch curr. (instantaneous value)		
$\max\left(I_{\mathrm{D,T_{HV}}}\right)$	$14.2\mathrm{A}$	maximum HV switch currents (RMS value)		
LV Side Switches T	$_{5}, T_{6}:$			
$\max\left(V_{\mathrm{DS,T_{LV},peak}}\right)$	$75\mathrm{V}$	maximum LV switch voltages		
$\max\left(I_{\mathrm{D},\mathrm{T}_{\mathrm{LV}},\mathrm{peak}}\right)$	$236\mathrm{A}$	max. LV switch curr. (instantaneous value)		
$\max\left(I_{\mathrm{D,T_{LV}}} ight)$	$157\mathrm{A}$	maximum LV switch currents (RMS value)		
Transformer:				
n	12	turns ratio		
$\max\left(I_{\mathrm{AC1}}\right)$	$20.0\mathrm{A}$	HV transformer winding, max. RMS current		
$\max\left(I_{AC2a}\right)$	$157\mathrm{A}$	LV transformer winding 1, max. RMS curr.		
$\max\left(I_{\rm AC2b}\right)$	$157\mathrm{A}$	LV transformer winding 2, max. RMS curr.		
$\max\left(B_{\mathrm{tr,core,peak}}\right)$	$130\mathrm{mT}$	for $N_1 = 12$, $N_2 = 1$, and $A_{tr,core} = 308 \text{ mm}^2$		
		(ELP 58 core, cf. Section A.2.4)		
DC Inductor L_{DC2} :				
$L_{\rm DC2}$	$0.83\mu\mathrm{H}$	inductance value		
$\max\left(I_{L_{\mathrm{DC2,peak}}}\right)$	$226\mathrm{A}$	maximum instantaneous inductor current		
$\max\left(I_{L_{\rm DC2}}\right)$	$202\mathrm{A}$	maximum inductor RMS current		
HF Inductor L:	HF Inductor L:			
L	$6.4\mu\mathrm{H}$	inductance value		
$\max\left(I_{L_{\mathrm{DC2,peak}}}\right)$	$22.6\mathrm{A}$	maximum instantaneous inductor current		
$\max\left(I_{L_{\rm DC2}}\right)$	$20.0\mathrm{A}$	maximum inductor RMS current		
HV Side DC Capacitor $C_{\rm DC1}$:				
$\max\left(I_{C_{\mathrm{DC1}}}\right)$	10.1 A	maximum capacitor RMS current		
$LV Side DC Capacitor C_{DC2}$:				
$\max\left(I_{C_{\rm DC2}}\right)$	16 A	maximum capacitor RMS current		

 Table A.19: Design summary for the bidirectional push-pull converter.

A.8 Design of the Bidirectional Buck-or-Boost Converter

A.8.1 Assumptions and Specifications

In order to simplify the design of the bidirectional buck-or-boost converter used in Chapter 7 (Figure A.47), the maximum peak-to-peak value $\Delta I_{L_{\rm DC}}$ of the inductor current $i_{L_{\rm DC}}(t) = -i_{L_{\rm DC}}(t)$ is specified. Typically, $\Delta I_{L_{\rm DC}} \approx 20\% \dots 40\% \overline{I}_{L_{\rm DC}}$ [cf. (7.3)] is selected at rated load to limit the increase of the RMS value of the inductor current [116]. A larger current ripple decreases the inductor volume (provided that the converter operates in the continuous conduction mode), however, the inductor RMS current and – in particular – the current through the DC capacitor $C_{\rm DCi}$ increase. The relative increase of the inductor RMS current $I_{L_{\rm DC},\rm rel}$ with respect to the relative current ripple $\Delta I_{L_{\rm DC}}$, rel = $\Delta I_{L_{\rm DC}}/\overline{I}_{\rm L_{\rm DC}}$ is given with:

$$I_{L_{\rm DC},\rm rel} = \sqrt{1 + \frac{\Delta I_{L_{\rm DC},\rm rel}^2}{12}};$$
 (A.170)

Figure A.48 (a) depicts the relative increase. Obviously, values below 40% cause the RMS inductor current to increase by less than 0.7%. In order to achieve a low inductor core volume, a higher maximum value of the relative peak-to-peak current ripple of 80% is selected (Table A.20). Still, the relative increase of the inductor RMS current is low, $I_{L_{\rm DC},\rm rel} < 2.7\%$. Furthermore, the RMS current through the DC capacitor $C_{\rm DCi}$ increases linearly with increasing



Figure A.47: Bidirectional buck-or-boost converter employed for the twostage converters discussed in Chapter 7.



Figure A.48: (a) Relative increase of the RMS current through the DC inductor $L_{\rm DC}$, $I_{L_{\rm DC},\rm rel} = I_{L_{\rm DC}}/\overline{I}_{L_{\rm DC}}$ due to an increase of the relative current ripple $\Delta I_{L_{\rm DC},\rm rel} = \Delta I_{L_{\rm DC}}/\overline{I}_{L_{\rm DC}}$; (b) calculated maximum RMS current through the DC capacitor $C_{\rm DCi}$, plotted against $\Delta I_{L_{\rm DC},\rm rel}$.

 $I_{L_{\rm DC},\rm rel}$ [Figure A.48 (b)].⁹ With the selected limit, $\Delta I_{L_{\rm DC},\rm rel} \leq 80\%$, the maximum capacitor RMS current results: $I_{C_{\rm DCi}} \approx 2.2$ A.

In order to facilitate the practical realization of the bidirectional buck-orboost converter, the applicable duty cycles are limited to $0 \leq D_{\text{buck}} \leq 95\%$ during buck operation and $5\% \leq D_{\text{boost}} \leq 95\%$ during boost operation.

⁹For the calculation of the maximum capacitor RMS current depicted in Figure A.48 (b), the converter inductance $L_{\rm DC}$ is first calculated according to (A.171) and subsequently, max $(I_{C_{\rm DC}i})$ is determined.

Property	Assumption	Description
$\eta_{ m est}$	95%	estimated converter efficiency
$D_{ m buck,min}$	0	min. duty cycle, buck operation
$D_{ m buck,max}$	95%	max. duty cycle, buck operation
$D_{\rm boost,min}$	5%	min. duty cycle, boost operation
$D_{ m boost,max}$	95%	max. duty cycle, boost operation
$\max\left(\Delta I_{L_{\mathrm{DC}}}\right)$	$\leq 80\% \left \overline{I}_{L_{\rm DC}} \right $	maximum peak-to-peak value of
		the current ripple on $i_{L_{\rm DC}}(t)$
$f_{ m S}$	$100 \mathrm{kHz}$	switching frequency

Table A.20: Assumptions and specifications used to design the bidirectional buck-or-boost converter (in addition to the HV side voltage range specified in Table 1.4).

Furthermore, a switching frequency, $f_{\rm S}$, of 100 kHz is specified in order to achieve both, low converter volume and low converter losses.¹⁰ The bidirectional buck-or-boost converter is designed for an assumed converter efficiency of 95%. Table A.20 summarizes the specifications and assumptions needed (in addition to the HV side voltage range specified in Table 1.4) to carry out the converter design.

A.8.2 Calculation of $L_{\rm DC}$

With the specifications given in Table 1.4 and in Table A.20 and with the analytical results presented in Section 7.2, the DC inductors can be calculated. For the configurations 1 and 2 (i.e. $V_1 > V_i$, Table 7.1) the inductance value $L_{\rm DC} = 150 \,\mu\text{H}$,

$$L_{\rm DC} \ge \max \left[\frac{V_{\rm i}^2(V_1 - V_{\rm i})}{f_{\rm S}V_1 I_{L_{\rm DC}, \rm rel} P/\eta_{\rm est}} \right] \left| \begin{array}{l} V_{1,\min} \le V_1 \le V_{1,\max} \\ V_{i,\min} \le V_i \le V_{1,\max} \\ P = 2.1 \, \rm kW \end{array} \right| \approx 150 \, \mu \rm H,$$
(A.171)

 $^{^{10}\}mathrm{The}$ employed silicon-carbide diodes facilitate low switching losses; cf. Figure 7.11 and [125].

results and for the configurations 3 and 4 ($V_1 < V_i$) L_{DC} is:

$$L_{\rm DC} \ge \max \begin{bmatrix} \frac{V_1^2(V_i - V_1)}{f_{\rm S} V_i I_{L_{\rm DC}, \rm rel} P / \eta_{\rm est}} \end{bmatrix} \begin{vmatrix} V_{1,\min} \le V_1 \le V_{1,\max} \\ V_{i,\min} \le V_i \le V_{i,\max} \\ P = 2.1 \, \rm kW \end{vmatrix} \approx 200 \, \mu \rm H.$$
(A.172)

With the expressions given in Section 7.2, the peak inductor currents and the inductor RMS currents can be calculated:

$$L = 150 \,\mu\text{H}: \,\max\left(I_{L_{\text{DC}},\text{peak}}\right) = \max\left(\overline{I}_{L_{\text{DC}}} + \frac{\Delta I_{L_{\text{DC}}}}{2}\right) = 17.0\,\text{A},\,\,(\text{A}.173)$$

$$L = 200 \,\mu\text{H}: \,\max\left(I_{L_{\text{DC}},\text{peak}}\right) = \max\left(\overline{I}_{L_{\text{DC}}} + \frac{\Delta I_{L_{\text{DC}}}}{2}\right) = 11.7\,\text{A},\,\,(\text{A}.174)$$

$$L = 150 \,\mu\text{H}: \,\max\left(I_{L_{\text{DC}}}\right) = 13.7 \,\text{A},\tag{A.175}$$

$$L = 200 \,\mu\text{H}: \, \max\left(I_{L_{\text{DC}}}\right) = 8.9 \,\text{A}. \tag{A.176}$$

A.8.3 Design Summary

Semiconductor Switches and Diodes

According to the voltage and current ratings calculated for the semiconductor switches and diodes (Section 7.2 and Table A.21), appropriate MOSFETs and diodes are selected:

- T_I, T_{II}: CoolMOS SPW47N60C3 (max. drain-to-source voltage: 600 V, max. drain current at T_i = 25°C: 47 A; obtained from [153])
- $D_{\rm I}$, $D_{\rm II}$: Silicon-carbide Diodes SDT12S60 (max. blocking voltage: 600 V, max. diode current at $T_{\rm i} = 25^{\circ}$ C: 23 A; obtained from [153])
- $D_{I,s}$, $D_{II,s}$: Schottky Diodes DSS25-0025B (max. blocking voltage: 25 V, max. diode current at $T_i = 25^{\circ}$ C: 35 A; obtained from [158])

DC Inductor 1: $L_{DC} = 150 \,\mu\text{H}, \, \max(I_{L_{DC}, \text{peak}}) = 17.0 \,\text{A}.$

The design of the DC inductors is very close to the inductor design presented in Appendix A.9, however, multilayer windings are used to effectively utilize the available winding window. For the inductor design, a peak flux density of $B_{L_{\rm DC, Peak, max}} = 310 \,\mathrm{mT}$ and a maximum RMS current density of $J_{L_{\rm DC, max}} =$ $7 \,\mathrm{A/mm^2}$ are assumed.¹¹ Furthermore, a litz wire with a single strand copper

 $^{^{11}}$ With respect to the employed ferrite material, a rather large value of the flux density can be used; compared to a HF transformer, the ferrite core of the DC inductor is not subject to a full reversal of the flux at the rate of the switching frequency and thus, low core losses result (cf. Section 7.2.5).

Property	Value	Description			
Semiconductor Switches T_I , T_{II} (CoolMOS SPW47N60C3):					
$\max(V_{\text{DS},\text{T}_{\text{I},\text{II}},\text{peak}})$	$475\mathrm{V}$	maximum switch voltages			
$\max(I_{\mathrm{T}_{\mathrm{I}},\mathrm{peak}})$	$17.0\mathrm{A}$	max. instantaneous switch current, T_{I}			
$\max(I_{\mathrm{T_{II},peak}})$	$17.0\mathrm{A}$	max. instantaneous switch current, T_{II}			
$\max(I_{\mathrm{T}_{\mathrm{I}}})$	$11.0 \mathrm{A}$	maximum switch RMS current, T_I			
$\max(I_{\mathrm{T_{II}}})$	11.1 A	maximum switch RMS current, T_{II}			
Silicon-carbide Diode	$es D_{I}, D_{II}$	(SDT12S60):			
$\max(V_{CA,D_{I,II},peak})$	$475\mathrm{V}$	maximum blocking voltages			
$\max(I_{D_{I},peak})$	$17.0 \mathrm{A}$	max. instantaneous diode current, D_I			
$\max(I_{\mathrm{D_{II}, peak}})$	$17.0 \mathrm{A}$	max. instantaneous diode current, D_{II}			
$\max(\overline{I}_{D_{I}})$	$8.8\mathrm{A}$	maximum average diode current, D_I			
$\max(\overline{I}_{\mathrm{D_{II}}})$	$8.9\mathrm{A}$	maximum average diode current, D_{II}			
$\max(I_{\mathrm{D}_{\mathrm{I}}})$	$11.0 \mathrm{A}$	maximum diode RMS current, D_I			
$\max(I_{\mathrm{D_{II}}})$	11.1 A	maximum diode RMS current, D_{II}			
Series Diodes D _{I,s} , D _{II,s} (DSS25-0025B):					
$\max(V_{CA,D_{I,II,s},peak})$	$< 10 \mathrm{V}$	maximum blocking voltages			
$\max(I_{\mathrm{D}_{\mathrm{I},\mathrm{s}},\mathrm{peak}})$	$17.0\mathrm{A}$	max. instantaneous diode current, $D_{I,s}$			
$\max(I_{\mathrm{D}_{\mathrm{II,s}},\mathrm{peak}})$	$17.0 \mathrm{A}$	max. instantaneous diode current, $\rm D_{II,s}$			
$\max(\overline{I}_{D_{I,s}})$	$8.8\mathrm{A}$	maximum average diode current, $D_{I,s}$			
$\max(\overline{I}_{D_{II,s}})$	$8.9\mathrm{A}$	maximum average diode current, $D_{II,s}$			
$\max(I_{\mathrm{D}_{\mathrm{I},\mathrm{s}}})$	$11.0 \mathrm{A}$	maximum diode RMS current, $D_{I,s}$			
$\max(I_{\mathrm{D}_{\mathrm{II,s}}})$	11.1 A	maximum diode RMS current, $D_{II,s}$			
DC Inductor 1, $L_{\rm DC}$	$= 150\mu\mathrm{H}$	(config. 1 and 2, i.e. $V_1 > V_i$; Table 7.1):			
$L_{\rm DC}$	$150 \mu H$	inductance value			
$\max(I_{L_{\rm DC},{\rm peak}})$	$17.0 \mathrm{A}$	maximum instantaneous inductor current			
$\max(I_{L_{\mathrm{DC}}})$	$13.7\mathrm{A}$	maximum inductor RMS current			
DC Inductor 2, $L_{\rm DC} = 200 \mu \text{H}$ (config. 3 and 4, i.e. $V_1 < V_i$; Table 7.1):					
$L_{\rm DC}$	$200 \mu\text{H}$	inductance value			
$\max(I_{L_{\rm DC},{\rm peak}})$	$11.7\mathrm{A}$	maximum instantaneous inductor current			
$\max(I_{L_{\mathrm{DC}}})$	$8.9\mathrm{A}$	maximum inductor RMS current			
$DC \ Capacitor \ C_{\rm DC1}$:					
$C_{\rm DC1}$	$3\mu F$	type of capacitor: 2 film capacitors			
$\max(I_{C_{\mathrm{DC1}}})$	$6.9\mathrm{A}$	maximum capacitor RMS current			
$DC \ Capacitor \ C_{\rm DCi}$:					
$C_{\rm DCi}$	$47\mu\mathrm{F}$	type of capacitor: 4 electrolytic capacitors			
$\max(I_{C_{DC2}})$	$2.2\mathrm{A}$	maximum capacitor RMS current			

 Table A.21: Design summary for the bidirectional buck-or-boost converter.



Figure A.49: Inductance and series resistance of the converter inductor $L_{\rm DC} = 150 \,\mu\text{H}$, measured with the Agilent 4294A Precision Impedance Analyzer: (a) series inductance, (b) series resistance; at $f = 100 \,\text{kHz}$, an AC resistance of approximately 290 m Ω is measured (according to [152] the relative error of the resistance measurement is $\pm 24\%$ at $f = 100 \,\text{kHz}$). At $f = f_{L_{\rm DC},\rm res} = 2.9 \,\text{MHz}$ a parallel resonance occurs; the effective capacitance in parallel to $L_{\rm DC}$ is 20 pF.

diameter of 0.2 mm is selected and a realization with one of the planar cores ELP 32, ELP 38, ELP 43, ELP 58, and ELP 64 is considered.

The design algorithm suggests the planar ferrite core ELP 64 (material N87) to be used with respect to minimum inductor volume. The following list summarizes the inductor design results.

• Total air gap length: 1.51 mm

- Number of turns: 16
- Employed litz wire: 65×0.2 mm.

Figure A.49 depicts the measured inductance and the measured resistance of this DC inductor, plotted against the measurement frequency.

DC Inductor 2: $L_{DC} = 200 \,\mu\text{H}, \, \max(I_{L_{DC}, \text{peak}}) = 11.7 \,\text{A}.$

The second inductor is again designed according to Appendix A.9 (except for the employed multilayer winding) and employs a planar ferrite core. The selected peak flux density and the maximum RMS current density are: $B_{L_{\rm DC, Peak, max}} = 310 \,\mathrm{mT}, J_{L_{\rm DC, max}} = 7 \,\mathrm{A/mm^2}.$

Minimum inductor volume is achieved with the ELP 58 ferrite core; since the ELP 58 core was not available, an E 58 core was used, instead (the dimensions of the ELP 58 and the E 58 cores are nearly identical, cf. [155, 156]; employed ferrite material: 3F3).

- Total air gap length: 1.62 mm
- Number of turns: 24
- Employed litz wire: 50×0.2 mm.

Figure A.50 depicts the measured inductance and the measured resistance of this DC inductor, plotted against the measurement frequency.

DC Capacitors

Due to the large value of the maximum RMS current through $C_{\rm DC1}$ [max($I_{C_{\rm DC1}}$) = 6.9 A], 2 film capacitors, each with a capacitance of 1.5 µF and a rated DC voltage of 630 V, are connected in parallel to form $C_{\rm DC1}$. A considerably lower RMS current rating is required for $C_{\rm DCi}$ [max($I_{C_{\rm DCi}}$) = 2.2 A]; there, 4 electrolytic capacitors, each with a capacitance of 47 µF and a rated voltage of 250 V, are forming the DC capacitor with a total capacitance of 47 µF and a rated voltage of 500 V. Figure A.51 depicts the impedance of $C_{\rm DCi}$, measured with the Agilent 4294A Precision Impedance Analyzer.

A.9 HF Inductor Design

The additional HF inductor employed for the DAB ($L_{\rm HV}$ in Figure 3.30) consists of a ferrite core, 3 air gaps, and a litz wire winding in order to achieve a compact inductor design and a low AC resistance $R_{L_{\rm HV}}$ [Figure A.52 (a)]. The



Figure A.50: Inductance and series resistance of the converter inductor $L_{\rm DC} = 200 \,\mu\text{H}$, measured with the Agilent 4294A Precision Impedance Analyzer: (a) series inductance, (b) series resistance; at $f = 100 \,\text{kHz}$, an AC resistance of approximately 390 m Ω is measured (according to [152] the relative error of the resistance measurement is $\pm 24\%$ at $f = 100 \,\text{kHz}$). At $f = f_{L_{\rm DC},\rm res} = 3.5 \,\text{MHz}$ a parallel resonance occurs; the effective capacitance in parallel to $L_{\rm DC}$ is 10.3 pF.

proposed design procedure (Appendix A.9.1), required for the design of the DAB (Appendix A.2) is thus confined to inductors with a litz wire winding and a magnetic core. Apart from that, the design procedure is flexible regarding the geometric properties of the employed core. For the DAB hardware prototype, however, there is space for 2 inductors, each made of 3 stacked planar ELP 32 cores (Table A.22). Therefore, the design of $L_{\rm HV}$ is carried



Figure A.51: Capacitance and series resistance of the electrolytic DC capacitor $L_{\rm DCi} \approx 47 \,\mu\text{F}$, measured with the Agilent 4294A Precision Impedance Analyzer: (a) series capacitance, (b) series resistance; at $f = 100 \,\text{kHz}$, an AC resistance of approximately $250 \,\text{m}\Omega$ is measured (according to [152] the relative error of the resistance measurement is $\pm 1.1\%$ at $f = 100 \,\text{kHz}$). For $f > 10 \,\text{kHz}$ the capacitance value drops considerably; at $f_{C_{\rm DCi},\rm res} = 850 \,\text{kHz}$ the series resonance frequency, which is due to the parasitic equivalent series inductance, occurs.

out with these ELP 32 cores (ferrite material: 3F3). 12

 $^{^{12}}$ Since the ELP 32 core was not available, an E 32 core was used, instead (the dimensions of the ELP 32 and the E 32 cores are nearly identical, cf. [155, 156]).



Figure A.52: (a) Cross-sectional drawing of the employed planar HF inductor, composed of an E-core / I-core combination; (b) modified position of the inductor air gap considered in Appendix A.9.3 to calculate the fringing coefficients; dimensioning according to Table A.22.

A.9.1 Design Procedure

The proposed design procedure, outlined in Figure A.53, starts with a single turn, $N_{L_{\rm HV}} = 1$, and a single core, $N_{\rm cores} = 1$, and increases the number of stacked cores in order to keep the peak flux density, calculated with (4.15), below the specified maximum value: $B_{L_{\rm HV},{\rm Peak}} < B_{L_{\rm HV},{\rm Peak},{\rm max}}$. Thereafter, the optimal winding arrangement with respect to minimum AC resistance (Appendix A.9.2) is calculated in order to determine the resulting maximum current density, $J_{\rm RMS}$, and subsequently the number of turns is increased until the maximum current density exceeds the specified maximum value: $J_{\rm RMS} > J_{\rm RMS,max}$. The proposed design procedure finally returns the inductor design with the highest number of turns, $N_{L_{\rm HV}}$, and the lowest number of cores, $N_{\rm cores}$, which fulfills the conditions $B_{L_{\rm HV},{\rm Peak}} < B_{L_{\rm HV},{\rm Peak,max}}$ and $J_{\rm RMS} \leq J_{\rm RMS,max}$. It further returns the optimal winding arrangement with

Property	Value	Description		
ELP 32				
$b_{ m t}$	$31.8\mathrm{mm}$	core width		
b_{c}	$9.5\mathrm{mm}$	breadth of the window area		
$b_{ m b}$	$7.5\mathrm{mm}$	breadth of the window area used for the		
		winding $(1 \text{ mm space on either side})$		
h	$3.2\mathrm{mm}$	height of the window area		
$h_{ m t,ELP32}$	$6.4\mathrm{mm}$	total core height		
$q_{ m c}$	$20.3\mathrm{mm}$	core depth		
$p_{ m c}$	$6.4\mathrm{mm}$	width of the inner core leg		
p_{s}	$3.2\mathrm{mm}$	width of each outer core leg		
PLT 32				
$b_{ m t}$	$31.8\mathrm{mm}$	core width		
$h_{ m t,PLT32}$	$3.2\mathrm{mm}$	core height		
$q_{ m c}$	$20.3\mathrm{mm}$	core depth		
Combination ELP 32 / PLT 32				
$l_{ m m}$	$35.9\mathrm{mm}$	average length of the magnetic path		
$A_{\rm c}$	$130\mathrm{mm}^2$	core cross section		

Table A.22: Geometric properties of the planar ELP 32 core and the PLT 32 plate (from [155]).

respect to minimum AC resistance: for the given copper diameter of a single strand, d_c , the optimal number of strands, n_{strands} , and the optimal number of parallel wires in horizontal direction, $n_{\text{parallel,h}}$, and vertical direction, $n_{\text{parallel,h}}$, (cf. Figure A.54) are given. In a final step, the required inductor air gap (Appendix A.9.3) and the expected AC resistance (Appendix A.9.4) are calculated; the total inductor core cross-sectional area, $A_{L_{\text{HV}},\text{core}}$, is equal to:

$$A_{L_{\rm HV,core}} = N_{\rm cores} A_{\rm c}.$$
 (A.177)

A.9.2 Winding Configuration, Optimal Litz Wire Calculation

Based on the given number of turns $N_{L_{HV}}$ and the geometric properties of the winding area, this subroutine identifies the best possible choice of litz wires, i.e. the number of litz wire strands and the number of parallel wires [either in



Figure A.53: Inductor design flowchart: the proposed procedure uses the constraints $B_{L_{\rm HV},{\rm peak}} < B_{L_{\rm HV},{\rm peak},{\rm max}}$ and $J_{\rm RMS} \leq J_{\rm RMS,{\rm max}}$ to search for the optimal number of turns $N_{L_{\rm HV}}$ and the optimal number of stacked cores $N_{\rm cores}$, in order to achieve a most effective core utilization.

horizontal direction, Figure A.54 (a), or vertical direction, Figure A.54 (b)]¹³ in order to obtain the constrained minimum for $J_{\rm RMS}$.

The procedure therefore requires the outer diameter of the litz wire (including the covering with thread), d_{t1} , to be known as a function of the single strand copper diameter, d_c , and the number of strands, $n_{strands}$. A suitable approximation is discussed in [139]:

$$d_{\rm tl} = \alpha_{\rm litz} \, d_{\rm c}^{\beta_{\rm litz}} \, d_{\rm ref}^{1-\beta_{\rm litz}} \left(\frac{F_{\rm lp}}{n_{\rm strands}}\right)^{-\frac{1}{2}},\tag{A.178}$$

$$\alpha_{\text{litz}} = 1.16, \tag{A.179}$$

$$\beta_{\rm litz} = 0.915,$$
 (A.180)

$$F_{\rm lp} = 0.6,$$
 (A.181)

 $d_{\rm ref} = 0.079 \,\mathrm{mm.}$ (A.182)

The coefficients α_{litz} and β_{litz} are obtained from manufacturers data [159] using least mean square approximation; the selected litz packing factor F_{lp} and the reference diameter d_{ref} are adopted from [139].

A.9.3 Inductor Air Gap Length

Compared to the core dimensions (Table A.22), the total air gap length, $l_{\rm g}$, of the employed inductors is relatively large ($l_{\rm g} > 1 \,\mathrm{mm}$) and thus, the calculation based on the well-known equation $L_{\rm HV} = N_{L_{\rm HV}}^2 \mu_0 A_{L_{\rm HV},\rm core}/l_{\rm g}$ predicts a considerably wrong air gap length.

A more advanced approach employs the reluctance model depicted in Figure A.55 and the fringing coefficients F_1 , F_2 , and F_3 given in [145] in order to determine the air gap reluctances $\mathcal{R}_{g,mid}$ and $\mathcal{R}_{g,side}$. For the fringing coefficients given in [145], however, 2 E-cores are considered instead of the E-core / I-core combination depicted in Figure A.52 (a). Still, good results are obtained for the inductance value if the inductor air gap is considered in the middle of the inductor, according to Figure A.52 (b).

With the arrangement depicted in Figure A.52 (b), the fringing coefficients F_1 , F_2 , and F_3 become [145]:

¹³The employed method only considers $(n_{\text{parallel},h} > 1 \land n_{\text{parallel},v} = 1) \lor (n_{\text{parallel},h} = 1 \land n_{\text{parallel},v} > 1)$ to avoid a large number of parallel wires.



(a)



(b)

Figure A.54: Considered winding configurations for round litz wire (the numbers inside the wire denote the turn number the wire belongs to); (a) $N_{L_{\rm HV}} = 1$: in order to utilize the winding area the best possible way, 3 wires are placed in parallel ($n_{\rm parallel,h} = 3$, $n_{\rm parallel,v} = 1$); (b) $N_{L_{\rm HV}} = 5$: best possible utilization of the winding area is achieved with 2 parallel litz wires ($n_{\rm parallel,h} = 1$, $n_{\rm parallel,v} = 2$).



Figure A.55: Reluctance model of the HF inductor: \mathcal{R}_{c1} and \mathcal{R}_{c2} denote the reluctances of the core, $\mathcal{R}_{g,mid}$ is the reluctance of the middle air gap, and $\mathcal{R}_{g,side}$ denote the reluctances of the outer air gaps.

$$F_1 = \frac{2}{\pi} \ln\left(\frac{\frac{1}{b_c} + \frac{1}{d}}{\frac{1}{b_c} + \frac{1}{h/2}}\right) + \frac{h/2 - 0.26d - 0.5b_c}{3b_c(h/2)^2}(h/2 - d)^2 + \frac{b_c}{3h/2}, \quad (A.183)$$

$$F_2 = \frac{2}{\pi} \ln \left[\left(\frac{0.44[(h/2)^2 + b_c^2] - 0.218dh/2 + 0.67b_c d}{d^2} + \right) \right]$$
(A.184)

$$\frac{0.33b_{\rm c}h/2 + 0.7825d^2}{d^2}\Big)^{\frac{1}{2}}\Big], \quad (A.185)$$

$$F_3 = \frac{1}{\pi}\operatorname{arccosh}\left\{3.395\left[\frac{(h_{\rm t,ELP32} + h_{\rm t,PLT32})/2}{d}\right]^2 + 0.15\frac{(h_{\rm t,ELP32} + h_{\rm t,PLT32})/2}{d} + 1.1155\right\} \quad (A.186)$$

(using the parameters listed in Table A.22 and $d = l_g/4$). With these fringing coefficients, an increased effective air gap area is considered for the calculation of $\mathcal{R}_{g,mid}$ and $\mathcal{R}_{g,side}$:

$$\mathcal{R}_{\rm g,mid} = \frac{1}{\mu_0} \frac{2d}{A_{L_{\rm HV,core}} + 2 \cdot F_1 \cdot dq_c N_{\rm cores} + 2 \cdot F_2 \cdot dp_c},\tag{A.187}$$

$$\mathcal{R}_{\rm g,side} = \frac{1}{\mu_0} \frac{2d}{A_{L_{\rm HV},\rm core}/2 + (F_1 + F_3)dq_{\rm c}N_{\rm cores} + 2 \cdot F_3 \cdot dp_{\rm s}}.$$
 (A.188)

The reluctances of the left core leg and of the right core leg are equal to:

$$\mathcal{R}_{c1} + \mathcal{R}_{c2} = \frac{1}{\mu_0 \mu_r} \frac{l_m}{A_{L_{HV},core}/2}.$$
 (A.189)

The total reluctance, $\mathcal{R}_{L_{HV}}$, becomes:

$$\mathcal{R}_{L_{\rm HV}} = \mathcal{R}_{\rm g,mid} + \frac{1}{2} \left(\mathcal{R}_{\rm g,side} + \mathcal{R}_{\rm c1} + \mathcal{R}_{\rm c2} \right)$$
(A.190)

and the result for the inductance is:

$$L_{\rm HV}(d = l_{\rm g}/4) = \frac{N_{L_{\rm HV}}^2}{\mathcal{R}_{L_{\rm HV}}}.$$
 (A.191)

Finally, a numerical solver is used to solve for the air gap length $l_{\rm g}$, required to obtain a given inductance $L_{\rm HV}$.

A.9.4 Inductor AC Resistance

The DC resistance of a litz wire with n_{strands} strands, single strand copper diameter d_c , and length l_{litz} is equal to:

$$R_{\rm DC,litz} = \frac{4 \, l_{\rm litz}}{\sigma_{\rm Cu} n_{\rm strands} \pi d_{\rm c}^2}; \tag{A.192}$$

 $\sigma_{\rm Cu}$ denotes the conductivity of copper [146, 147],

$$\sigma_{\rm Cu} \approx \frac{56.2 \cdot 10^6 \,(\Omega {\rm m})^{-1}}{1 + 3.9 \cdot 10^{-3} \,(^{\circ}{\rm C})^{-1} \cdot \Delta T - 0.6 \cdot 10^{-6} \,(^{\circ}{\rm C})^{-2} \cdot \Delta T^2}, \qquad (A.193)$$

$$\Delta T = T_{\rm Cu} - 20^{\circ} \text{C}. \tag{A.194}$$

The increase of the wire resistance due to the skin effect at the excitation frequency f is determined with [38, 148]:¹⁴

$$\delta = \frac{1}{\sqrt{\pi\mu_0 \sigma_{\rm Cu} f}},\tag{A.195}$$

$$\xi = \frac{d_c}{\sqrt{2} \cdot \delta},\tag{A.196}$$

$$F_{\rm R} = \frac{\sqrt{2} \cdot \xi}{4} \left[\frac{\text{Ber}_0(\xi) \text{Bei}_1(\xi) - \text{Ber}_0(\xi) \text{Ber}_1(\xi)}{\text{Ber}_1(\xi)^2 + \text{Bei}_1(\xi)^2} - \frac{\text{Bei}_0(\xi) \text{Ber}_1(\xi) + \text{Bei}_0(\xi) \text{Bei}_1(\xi)}{\text{Ber}_1(\xi)^2 + \text{Bei}_1(\xi)^2} \right], \quad (A.197)$$

$$R_{\rm AC,litz,skin} = F_{\rm R} \cdot R_{\rm DC,litz} \tag{A.198}$$

and the increase due to the proximity effect by reason of the currents in the surrounding conductors and an external magnetic field \vec{H}_{ext} (RMS value)

 $[\]overline{{}^{14}\operatorname{Ber}_n(\xi)}$ and $\operatorname{Bei}_n(\xi)$ are the n^{th} order Kelvin functions obtained from the n^{th} order Bessel function $J_n(\zeta)$: $\operatorname{Ber}_n(\xi) = \operatorname{Re}[J_n(\xi e^{j \cdot 3\pi/4})]$ and $\operatorname{Bei}_n(\xi) = \operatorname{Im}[J_n(\xi e^{j \cdot 3\pi/4})]; \delta$ denotes the skin depth.

is [38, 148]:

$$G_{\rm R} = -\frac{\sqrt{2} \cdot \xi \pi^2 d_{\rm c}^2}{2} \left[\frac{\text{Ber}_2(\xi) \text{Ber}_1(\xi) + \text{Ber}_2(\xi) \text{Bei}_1(\xi)}{\text{Ber}_0(\xi)^2 + \text{Bei}_0(\xi)^2} + \frac{\text{Bei}_2(\xi) \text{Bei}_1(\xi) - \text{Bei}_2(\xi) \text{Ber}_1(\xi)}{\text{Ber}_0(\xi)^2 + \text{Bei}_0(\xi)^2} \right], \quad (A.199)$$

$$P_{\rm AC,litz,prox} = \frac{4}{\pi d_{\rm tl}^2} G_{\rm R} \, n_{\rm strands}^2 R_{\rm DC} \int_0^{d_{\rm tl}/2} \int_0^{2\pi} |\vec{H}(r,\varphi)|^2 r \mathrm{d}\varphi \mathrm{d}r, \qquad (A.200)$$

$$\vec{H}(r,\varphi) = \frac{2Ir}{\pi d_{\rm tl}^2} (-\sin\varphi \,\vec{e}_{\rm x} + \cos\varphi \,\vec{e}_{\rm y}) + \vec{H}_{\rm ext}(r,\varphi), \tag{A.201}$$

$$R_{\rm AC,litz,prox} = \frac{P_{\rm AC,litz,prox}}{I_{L_1}^2}; \tag{A.202}$$

[each strand is assumed to carry current with the same RMS value I/n_{strands} ; moreover, in (A.201), the conductor current I is oriented along the positive z-axis].

The challenge in the calculation of the AC resistance is to correctly determine the external magnetic field $\dot{H}_{ext}(r,\varphi)$ (in particular close to the air gap, where high field densities considerably deteriorate the HF properties of the inductor). Different possibilities to approximate $H_{\text{ext}}(r,\varphi)$ include the Schwarz-Christoffel transformation [38], field approximations [149], or FEM simulation. The Schwarz-Christoffel transformation results in a rather complex mathematical problem and the field approximation given in [149] fails due to the large air gaps of the employed inductors. Thus, FEM simulation is used to determine $H(r,\varphi)$ for a sample inductor with $L_{\rm HV} = 15\,\mu{\rm H}$, a peak current of 30 A and an RMS current of 15 A (the specifications used for the sample inductor, Table A.23, are close to the specifications of the inductors required for the DAB, cf. Table A.1). The sample inductor incorporates a space of 1 mm (approximately equal to $l_{\rm g}/2$) between the winding and the sides of the inductor core [Figure A.52: $(b_{\rm c} - b_{\rm b})/2 = 1 \,\mathrm{mm}$] in order to avoid excessive magnetic fields inside the winding; the winding employs 2 parallel litz wires in order to effectively utilize the remaining winding area.

Figure A.56 depicts the simulated magnetic field strength (RMS values) by reason of an RMS current of 15 A: close to the air gap, the magnetic field reaches values of up to 35 kA/m; moreover, the lower row of conductors (1a ... 5a) is exposed to considerably higher magnetic fields than the upper row (1b ... 5b). Numerical integration confirms the higher values of $\int_{0}^{d_{t1}/2} \int_{0}^{2\pi} |\vec{H}(r,\varphi)|^2 r d\varphi dr$ for the lower conductor row (Table A.24). Consequently, different AC resistances result for the 2 parallel conductors, which causes a considerable increase of the inductor HF losses (between f = 5 kHz



Figure A.56: Cross-sectional plot of the magnetic fields (RMS values) in the sample inductor, $L = 15 \,\mu\text{H}$, Table A.23, by reason of an RMS current of 15 A (FEM simulation, $f = 100 \,\text{kHz}$); the winding is optimized according to Appendix A.9.2: $n_{\text{parallel,h}} = 1$, $n_{\text{parallel,v}} = 2$ [cf. Figure A.54 (b); the circles indicate the litz wires]; (a) full inductor cross-sectional view: close to the air gap, the magnetic field reaches values of up to $35 \,\text{kA/m}$; (b) and (c) magnified winding window areas [to the left and to the right of the middle leg shown in (a), respectively]: the numbers inside the litz wires denote the turn number the conductor belongs to; "a" and "b" are used to distinguish between the 2 parallel wires.

Specifications:				
Property	Value	Description		
L	$15\mu\mathrm{H}$	inductance		
$I_{ m peak}$	$30\mathrm{A}$	maximum peak current		
Ι	$15\mathrm{A}$	maximum RMS current		
$B_{ m peak,max}$	$150\mathrm{mT}$	maximum peak flux density excitation of		
		the ferrite core		
J_{\max}	$10\mathrm{A/mm^2}$	maximum RMS current density		
Inductor Desig	n Properties:			
Property	Value	Description		
$l_{ m g}$	$2.06\mathrm{mm}$	total air gap length		
$N_{L_{\rm HV}}$	5	number of turns		
$N_{\rm parallel,h}$	1	number of horizontally arranged parallel		
		conductors		
$N_{\rm parallel,v}$	2	number of vertically arranged parallel		
		conductors		
$N_{ m cores}$	5	number of employed cores (ELP 32 $/$		
		PLT 32 combinations)		
$l_{ m wdg,core}$	$5\cdot 203\mathrm{mm}$	litz wire length inside the core		
$l_{\rm wdg,end\ turns}$	$5\cdot49\mathrm{mm}$	litz wire length due to the end turns		
$d_{ m c}$	$0.071\mathrm{mm}$	litz wire, single strand copper diameter		
n_{strands}	200	litz wire, number of strands		

Table A.23: Specifications and properties of the sample inductor used to verify the calculations discussed in Appendix A.9.

and f = 500 kHz in Figure A.57). The slightly modified winding configuration depicted in Figure A.58 cancels this effect: there, the 2 parallel litz wires are twisted by 180° on one side in order to achieve an equal increase of the AC resistance for both litz wires. The measured AC resistance thus decreases from $33 \text{ m}\Omega$ to $22 \text{ m}\Omega$ (at f = 100 kHz in Figure A.59).

The evaluation of $\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} |\vec{H}(r,\varphi)|^2 r d\varphi dr$ enables the calculation of the winding's AC resistance for different litz wires (Table A.25), whereas the AC resistance of the end turns is calculated on the assumption $\vec{H}_{ext}(r,\varphi) = \vec{0}$. Thus, in a final step, the optimal litz wire with respect to minimum AC re-

Conductor Number	1a	1b	2a	2b
$\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} \vec{H}(r,\varphi) ^2 r \mathrm{d}\varphi \mathrm{d}r$	$135.9\mathrm{A}^2$	$28.7\mathrm{A}^2$	$32.9\mathrm{A}^2$	$9.3\mathrm{A}^2$
Conductor Number	3a	3b	4a	4b
$\int_{0}^{d_{tl}/2} \int_{0}^{2\pi} \vec{H}(r,\varphi) ^2 r \mathrm{d}\varphi \mathrm{d}r$	$18.5\mathrm{A}^2$	$6.5\mathrm{A}^2$	$71.5\mathrm{A}^2$	$29.1\mathrm{A}^2$
Conductor Number	5a	5b		
$\int\limits_{0}^{d_{tl}/2}\int\limits_{0}^{2\pi}\int\limits_{0}ert ec{H}(r,arphi)ert^{2}r\mathrm{d}arphi\mathrm{d}r$	$236.7\mathrm{A}^2$	$53.6\mathrm{A}^2$		

Table A.24: FEM simulator results for $\int_{0}^{d_{t1}/2} \int_{0}^{2\pi} |\vec{H}(r,\varphi)|^2 r d\varphi dr$ with respect to the different conductors depicted in Figure A.56; the outer conductor diameter used in the simulation is $d_{t1} = 1.51$ mm.

sistance is to be determined: with an increasing single strand copper diameter d_c , the effective copper area increases and the DC resistance decreases; however, with increasing d_c , the impact of high frequency effects increases, too (Table A.25). With the given data (Tables A.22, A.23, and A.24), minimum AC resistance is achieved with single strand copper diameters of either 0.071 mm or 0.1 mm. For the inductor winding, $d_c = 0.071$ mm and $n_{\text{strands}} = 200$ is selected due to the availability of the respective litz wire.¹⁵

For $d_c = 0.071$ mm, the difference between DC and AC winding resistances is rather small (14.5 m Ω vs. 15.2 m Ω ; cf. Table A.25) and the impact of this difference on the total converter losses is negligible. Therefore, the presented calculation of the AC resistance is used to determine the appropriate single strand diameter $d_c = 0.071$ mm of the litz wire and the DC resistance, calculated for $d_c = 0.071$ mm, is used to approximately determine the resistance $R_{L_{\rm HV}}$ used to calculated the inductor copper losses.

¹⁵The difference between the measured AC resistance of $22 \text{ m}\Omega$ and the calculated AC resistance of $15 \text{ m}\Omega$ is addressed to the precision of the Agilent 4294A Precision Impedance Analyzer: at f = 100 kHz, the accuracy of the resistance measurement is $\pm 42\%$, since the phase angle of the inductor's impedance is close to 90° (calculated according to [152]). Moreover, the measured total series resistance includes (small-signal) core losses, as well.


Figure A.57: Inductance and series resistance of the sample inductor (Table A.23) with straight end turns, measured with the Agilent 4294A Precision Impedance Analyzer: at f = 100 kHz, the accuracy of the resistance measurement is $\pm 30\%$, since the phase angle of the total impedance is very close to 90° (calculated according to [152]). Still, a considerable increase of $R_{L_{\rm HV}}$ can be observed for f > 5 kHz due to different AC losses in each of the 2 parallel wires (cf. Table A.24).



Figure A.58: (a) Picture of the sample inductor (Table A.23) with the modified winding configuration; an insulating tape holds the 2 parallel litz wires together. The picture shows the employed end turns: on the right side, the end turns are twisted by 180° in order to decrease the AC resistance of the inductor. (b) Cross-sectional drawing of the inductor to illustrate the resulting arrangement of the litz wires (cf. Figure A.56).



Figure A.59: Inductance and series resistance of the sample inductor (Table A.23) with twisted end turns on one side, according to Figure A.58, measured with the Agilent 4294A Precision Impedance Analyzer: at f = 100 kHz, the accuracy of the resistance measurement is $\pm 42\%$, since the phase angle of the total impedance is very close to 90° (calculated according to [152]). Still, compared to Figure A.57, a reduced AC resistance is observed for 5 kHz < f < 500 kHz.

$d_{\rm c}$	$n_{\rm strands}$	$d_{ m tl}$	DC resistance $R_{\rm DC}$	AC resistance R_{AC}
$0.03\mathrm{mm}$	964	$1.51\mathrm{mm}$	$16.8\mathrm{m}\Omega$	$16.9\mathrm{m}\Omega$
$0.04\mathrm{mm}$	569	$1.51\mathrm{mm}$	$16.0\mathrm{m}\Omega$	$16.2\mathrm{m}\Omega$
$0.05\mathrm{mm}$	378	$1.51\mathrm{mm}$	$15.4\mathrm{m}\Omega$	$15.7\mathrm{m}\Omega$
$0.071\mathrm{mm}$	200	$1.51\mathrm{mm}$	$14.5\mathrm{m}\Omega$	$15.2\mathrm{m}\Omega$
$0.1\mathrm{mm}$	106	$1.51\mathrm{mm}$	$13.8\mathrm{m}\Omega$	$15.2\mathrm{m}\Omega$
$0.2\mathrm{mm}$	29	$1.49\mathrm{mm}$	$12.6\mathrm{m}\Omega$	$19.1 \mathrm{m}\Omega$
$0.355\mathrm{mm}$	10	$1.48\mathrm{mm}$	$11.6\mathrm{m}\Omega$	$33.1\mathrm{m}\Omega$

Table A.25: Number of strands, $n_{\rm strands}$, and outer litz wire diameter, $d_{\rm tl}$, calculated with the algorithm discussed in Appendix A.9.2. The DC resistance increases with decreasing values of the single strand copper diameter $d_{\rm c}$ due to a decrease of the effective copper area [i.e. $\beta_{\rm litz} < 1$ in (A.178)]; the ratio $R_{\rm AC}/R_{\rm DC}$ increases with increasing $d_{\rm c}$ due to increasing AC losses (evaluated at f = 100 kHz). Regarding the shown numerical precision, minimum AC resistance is achieved with $d_{\rm c} = 0.071$ mm and with $d_{\rm c} = 0.1$ mm.

Appendix B

Converter Comparison: Calculation of the VA ratings

B.1 Semiconductor Switches

Following [116], the VA rating of each semiconductor switch is expressed with

$$S_{\text{switch}} = \max(V_{\text{switch,peak}}) \cdot \max(I_{\text{switch}})$$
(B.1)

using the maximum voltage applied to the switch, $\max(V_{\text{switch,peak}})$, and the maximum RMS current through the switch, $\max(I_{\text{switch}})$.

B.2 Transformer

The VA rating of a single-phase transformer with m windings (i.e. m = 2 for the transformer used for the DAB, Figure 2.10), operated with sinusoidal voltages and currents, is calculated with

$$S_{\rm tr} = \frac{1}{2} \sum_{i=1}^{m} \hat{V}_{\rm tr} \cdot \hat{I}_{\rm tr}, \qquad (B.2)$$

whereas \hat{V}_{tri} denotes the RMS voltage applied to the *i*-th winding and \hat{I}_{tri} denotes the RMS current through the *i*-th winding [92]. In order to consider the increased component stress due to the wide operating range, (B.2) is

modified accordingly:¹

$$S_{\rm tr} = \frac{1}{2} \sum_{i=1}^{m} \max(\hat{V}_{\rm tr}i) \cdot \max(\hat{I}_{\rm tr}i); \qquad (B.3)$$

there, $\max(\hat{V}_{\text{tr}i})$ and $\max(\hat{I}_{\text{tr}i})$ denote the maximum RMS values of the voltages and currents applied to the *i*-th winding, which are calculated within the operating ranges defined in Table 1.4.

However, none of the DC–DC converters discussed in Section 2.2 employs purely sinusoidal voltages and currents; accordingly the modified expression

$$S_{\rm tr} = \frac{1}{2} \sum_{i=1}^{m} \max(V_{\rm tri,eq}) \cdot \max(I_{\rm tri,eq})$$
(B.4)

with

$$\frac{V_{\text{tr}i,\text{eq}}}{\hat{V}_{\text{tr}i}} = \frac{\max\left(\int_{0}^{T_{\text{S}}} v_{\text{tr}i}(t) \, \mathrm{d}t\right) - \min\left(\int_{0}^{T_{\text{S}}} v_{\text{tr}i}(t) \, \mathrm{d}t\right)}{\max\left(\int_{0}^{T_{\text{S}}} \hat{v}_{\text{tr}i}(t) \, \mathrm{d}t\right) - \min\left(\int_{0}^{T_{\text{S}}} \hat{v}_{\text{tr}i}(t) \, \mathrm{d}t\right)} = \\
= \frac{\max\left(\int_{0}^{T_{\text{S}}} v_{\text{tr}i}(t) \, \mathrm{d}t\right) - \min\left(\int_{0}^{T_{\text{S}}} v_{\text{tr}i}(t) \, \mathrm{d}t\right)}{\frac{\sqrt{2}}{\pi f_{\text{S}}} \, \hat{V}_{\text{tr}i}} = \tag{B.5}$$

$$= \frac{\pi D}{\sqrt{2}} \frac{V_1}{\hat{V}_{\text{tr}i}} \text{ (for rectangular voltages according to Figure 3.3), (B.6)}$$

$$I_{\rm tri,eq} = I_{\rm tri},\tag{B.7}$$

is used instead of (B.3). There, the currents $I_{\text{tri,eq}}$ and the voltages $V_{\text{tri,eq}}$ are selected with respect to the resulting losses (on the assumptions of negligible high frequency effects and negligible stray and magnetizing inductances). Consequently, the employed current RMS values remain unchanged. The voltages $V_{\text{tri,eq}}$, however, are adjusted such that the peak magnetic flux density caused by $v_{\text{tri}}(t)$ becomes equal to the peak magnetic flux density caused by the sinusoidal voltage $\hat{v}_{\text{tri}}(t)$ [92].

B.3 AC and DC Inductors

Similar to (B.3), the VA rating of an AC inductor can be defined:

$$S_{\text{ind}} = \max(\hat{V}) \cdot \max(\hat{I}) \tag{B.8}$$

¹On the assumptions of a limited value of the current density J_{max} in the winding and a limited value of the magnetic flux density $B_{\text{peak,max}}$ in the core, i.e. no thermal limitations apply.

 $[\max(\hat{V})$ denotes the maximum RMS voltage applied to the inductor and $\max(\hat{I})$ denotes the maximum RMS value of the AC component of the inductor current, calculated within the operating ranges defined in Table 1.4]. However, none of the DC–DC converters discussed in Section 2.2 employs purely sinusoidal inductor voltages and currents, and the modified expression

$$S_{\rm ind} = \max(V_{\rm eq}) \cdot \max(I_{\rm eq}) \tag{B.9}$$

with

$$\frac{V_{\text{eq}}}{\hat{V}} = \frac{\max\left(\int_{0}^{T_{\text{S}}} v(t) \, \mathrm{d}t\right) - \min\left(\int_{0}^{T_{\text{S}}} v(t) \, \mathrm{d}t\right)}{\max\left(\int_{0}^{T_{\text{S}}} \hat{v}(t) \, \mathrm{d}t\right) - \min\left(\int_{0}^{T_{\text{S}}} \hat{v}(t) \, \mathrm{d}t\right)} = \\
= \frac{\max\left(\int_{0}^{T_{\text{S}}} v(t) \, \mathrm{d}t\right) - \min\left(\int_{0}^{T_{\text{S}}} v(t) \, \mathrm{d}t\right)}{\frac{\sqrt{2}}{\pi f_{\text{S}}} \hat{V}}, \quad (B.10)$$

$$\frac{I_{\text{eq}}}{\hat{I}} = 1 \quad (B.11)$$

is used instead of (B.8), cf. Appendix B.2.

For DC inductors,² the VA rating defined with (B.8) is of no relevance: with increasing inductance L and unchanged inductor voltage v(t), the RMS value of the AC component of the inductor current, \hat{I} , decreases and converges to 0 for $L \to \infty$. It is thus more meaningful to specify the peak magnetic energy $W_{\text{mag,peak}}$ stored in the DC inductor, since the inductor volume increases with increasing $W_{\text{mag,peak}}$ (Appendix B.3.1).

B.3.1 Theoretical Relation between $W_{mag,peak}$ and Inductor Volume V_L

The inductance L of an inductor with a magnetic core and an air gap (Figure B.1) is approximately:

$$L \approx \mu_0 N^2 \frac{A_{\rm Fe}}{l}.\tag{B.12}$$

On the assumption of a maximum allowable peak current density $J_{\text{peak,max}}$ inside the winding window A_{W} and with a filling factor k_{W}

$$k_{\rm W}A_{\rm W} = \frac{N I_{\rm peak,max}}{J_{\rm peak,max}} \tag{B.13}$$

²For a DC inductor, the DC component of the inductor current is predominant.



Figure B.1: Principle setup of an inductor with magnetic core and air gap.

applies and thus, the maximum allowable peak value of the winding current is:

$$I_{\text{peak,max}} = \frac{k A_{\text{W}} J_{\text{peak,max}}}{N}.$$
 (B.14)

The permissible peak flux density $B_{\text{peak,max}}$ defines a second requirement:

$$B_{\text{peak,max}} = \mu_0 \frac{N I_{\text{peak,max}}}{l}.$$
 (B.15)

With this, the stored magnetic energy is proportional to the product $A_{\rm Fe} \cdot A_{\rm W},$

$$W_{\text{mag,peak}} = \frac{1}{2} L I_{\text{peak,max}}^2 \stackrel{(B.12)}{=} \frac{1}{2} N \cdot \underbrace{\left(\mu_0 N \frac{I_{\text{peak,max}}}{l}\right)}_{=B_{\text{peak,max}, \text{ cf. (B.15)}}} A_{\text{Fe}} I_{\text{peak,max}} = \underbrace{\left(\frac{B.14}{2}\right)}_{=B_{\text{peak,max}, \text{ cf. (B.15)}}} A_{\text{Fe}} I_{\text{peak,max}} = \underbrace{\left(\frac{B.14}{2}\right)}_{2} A_{\text{peak,max}} = \underbrace{\left(\frac{B.14}{2}\right)}_{2} A_{\text{peak,max}} = \underbrace{\left(\frac{B.14}{2}\right)}_{2} A_{\text{peak,max}} = \underbrace{\left(\frac{B.14}{2}\right)}_{2} A_$$

Provided that $B_{\rm peak,max}$, k, and $J_{\rm peak,max}$ are constant, $W_{\rm mag,peak} \propto A_{\rm Fe} A_{\rm W}$

applies. With the scaling laws given in [92],

$$\frac{A_{\rm Fe}}{A_{\rm Fe}^*} = \left(\frac{Lin}{Lin^*}\right)^2 \tag{B.17}$$

$$\frac{A_{\rm W}}{A_{\rm W}^*} = \left(\frac{Lin}{Lin^*}\right)^2 \tag{B.18}$$

$$\frac{V_{\rm L}}{V_{\rm L}^*} = \left(\frac{Lin}{Lin^*}\right)^3 \tag{B.19}$$

(*Lin* denotes a linear reference dimension, quantities without * denote the reference design values, and quantities with * denote all values obtained from the scaled design), the expression for $V_{\rm L}/V_{\rm L}^*$ can be rewritten [92]:

$$\frac{V_L}{V_L^*} = \left(\frac{W_{\rm mag,peak}}{W_{\rm mag,peak}^*}\right)^{\frac{3}{4}}.$$
 (B.20)

Thus,

$$\frac{w_{\rm mag,peak}}{w_{\rm mag,peak}^*} = \frac{W_{\rm mag,peak}/V_L}{W_{\rm mag,peak}^*/V_L^*} = \left(\frac{V_L}{V_L^*}\right)^{\gamma}, \, \gamma = 1/3, \tag{B.21}$$

results, i.e. the energy density of an inductor increases with increasing volume (the presented calculation, however, neglects thermal aspects; in case of a thermally limited design γ decreases [92]).

B.4 AC and DC Capacitors

Similar to the AC inductor, a VA rating can be defined for an AC capacitor with a given capacitance C:

$$S_{\text{cap}} = \max(\hat{V}) \cdot \max(\hat{I}) = \max(\hat{V} \cdot \hat{I}) = \frac{\max(\hat{I}^2)}{B_C}, B_C = 2\pi f_S C. \quad (B.22)$$

The VA rating is meaningless for DC capacitors and the peak electric energy storage capability $W_{\rm el,peak}$ and the peak RMS capacitor current $\max(\hat{I})$ are used, instead:

- $W_{el,peak}$ is a figure for the expected capacitor volume (cf. Appendix B.4.1),
- depending on the capacitor technology (e.g. electrolytic capacitors, ceramic capacitors, film capacitors), $\max(\hat{I})$ may have a considerable impact on the required number of capacitors [138].



Figure B.2: Principle setup of a parallel-plate capacitor.

B.4.1 Theoretical Relation between $W_{el,peak}$ and Capacitor Volume V_C

The capacitance C of the parallel-plate capacitor depicted in Figure B.2 is given with:

$$C = \epsilon_0 \epsilon_r \frac{A}{d}; \tag{B.23}$$

a dielectric material with $\epsilon = \epsilon_0 \epsilon_r$ fills the space between the two metal plates (provided that fringing effects are negligible). The respective capacitor volume is equal to:

$$V_C = A \cdot d. \tag{B.24}$$

On the assumption of a maximum permissible peak value of the electric field strength $E_{\text{peak,max}}$ inside the dielectric material, the minimum distance d_{\min} between the plates is:

$$d_{\min} = \frac{V_{\text{peak,max}}}{E_{\text{peak,max}}} \tag{B.25}$$

and the maximum allowable voltage is:

$$V_{\text{peak,max}} = d_{\min} E_{\text{peak,max}}.$$
 (B.26)

With $d \rightarrow d_{\min}$ the energy stored in the parallel-plate capacitor becomes:

$$W_{\rm el,peak} = \frac{1}{2} C V_{\rm peak,max}^2 = \frac{1}{2} \epsilon_0 \epsilon_r \frac{A}{d_{\rm min}} \left(d_{\rm min} E_{\rm peak,max} \right)^2 = \frac{1}{2} \epsilon_0 \epsilon_r \underbrace{A d_{\rm min}}_{=V_{C,\rm min}} E_{\rm peak,max}^2. \quad (B.27)$$

Thus, in theory, the volume of a parallel-plate capacitor is proportional to the stored electric energy, $W_{\rm el} \propto V_{C,{\rm min.}}^3$

³According to (B.27), the capacitor volume should be directly proportional to C and $V_{\text{peak,max}}^2$, i.e. $V_C \propto C$ for constant $V_{\text{peak,max}}$ and $V_C \propto V_{\text{peak,max}}^2$ for constant C. Data sheet values confirm $V_C \propto C$; however, with increasing peak voltage $V_{\text{peak,max}}$, the capacitor volume rather increases according to $V_C \propto V_{\text{peak,max}}^{\gamma}$ with $1 < \gamma < 2$ (γ needs to be extracted from data sheet values for the selected type of capacitor).

Appendix C

Extracting the Steinmetz Coefficients

According to Chapter 4, the Steinmetz equation is used to estimate the core losses P_{core} of the transformer and the inductor:

$$P_{\rm core} \approx V_{\rm core} k f^{\alpha} B^{\beta}_{\rm peak} \tag{C.1}$$

(V_{core} denotes the respective core volume, B_{peak} is the respective peak flux density, and f is the excitation frequency). The Steinmetz coefficients k, α , and β are obtained from data sheet values using a least mean square (LMS) approximation. In order to obtain a polynomial function suitable for the LMS approximation, the natural logarithm is first applied to (C.1),

$$\ln(p_{\text{core}}) = \ln(P_{\text{core}}/V_{\text{core}}) = \ln(k) + \alpha \ln(f) + \beta \ln(B_{\text{peak}})$$
(C.2)

 $(p_{\text{core}} \text{ denotes the core loss density})$, and the LMS approximation is accomplished with respect to $\ln(k)$, α , and β . The employed input data $p_{\text{core}}(f, B_{\text{peak}})$ and the calculated Steinmetz coefficients are given in Table C.1 for the N87 ferrite [160] and in Table C.2 for the 3F3 ferrite [161].

In Figure C.1 and Figure C.2 the input data listed in Table C.1 is compared to the values calculated with (C.1). For $T_{\text{core}} = 25^{\circ}$ C the maximum relative error due to the Steinmetz approximation (C.1) is 21% and the average relative error is 9.3%; for $T_{\text{core}} = 100^{\circ}$ C the maximum relative error is 30% and the average relative error is 11%. Figure C.3 compares the input data listed in Table C.2 to the values calculated with (C.1); the resulting maximum relative error is 19% and the average relative error is 8.8%.

In [161] the core losses of the 3F3 ferrite material are comprehensively characterized for $T_{\rm core} = 100^{\circ}$ C. Since the DAB converter model presented in Chapter 4 is parameterized for $T_{\rm core} = 25^{\circ}$ C, the Steinmetz parameters listed in Table C.2 need to be modified. According to Table C.1, all 3 Steinmetz

f	B_{peak}	$p_{ m core}$	k	α	β		
Ferrite material: N87, core temperature: $T_{\rm core} = 25^{\circ} C$ [160]							
$50\mathrm{kHz}$	$50\mathrm{mT}$	$11\mathrm{kW/m^3}$					
$50\mathrm{kHz}$	$100\mathrm{mT}$	$67\mathrm{kW/m^3}$					
$50\mathrm{kHz}$	$200\mathrm{mT}$	$285\mathrm{kW/m}^3$					
$100\mathrm{kHz}$	$25\mathrm{mT}$	$4.1\mathrm{kW/m^3}$					
$100\mathrm{kHz}$	$50\mathrm{mT}$	$67\mathrm{kW/m^3}$					
$100\mathrm{kHz}$	$100\mathrm{mT}$	$128\mathrm{kW/m^3}$					
$100\mathrm{kHz}$	$200\mathrm{mT}$	$650\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$25\mathrm{mT}$	$9.9\mathrm{kW/m}^3$	14.1	1.27	2.42		
$200\mathrm{kHz}$	$50\mathrm{mT}$	$50\mathrm{kW/m^3}$			2.72		
$200\mathrm{kHz}$	$100\mathrm{mT}$	$320\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$200\mathrm{mT}$	$1.4\mathrm{MW/m^3}$					
$300\mathrm{kHz}$	$25\mathrm{mT}$	mT $19 \mathrm{kW/m^3}$					
$300\mathrm{kHz}$	$50\mathrm{mT}$	$100{ m kW/m^3}$					
$300\mathrm{kHz}$	$100\mathrm{mT}$	$627\mathrm{kW/m^3}$					
$300\mathrm{kHz}$	$200\mathrm{mT}$	$3.1\mathrm{MW/m^3}$					
Ferrite material: N87, core temperature: $T_{\rm core} = 100^{\circ} C$ [160]							
$50\mathrm{kHz}$	$50\mathrm{mT}$	$2.4\mathrm{kW/m^3}$					
$50\mathrm{kHz}$	$100\mathrm{mT}$	$17 \mathrm{kW/m^3}$					
$50\mathrm{kHz}$	$200\mathrm{mT}$	$141\mathrm{kW/m^3}$					
$50\mathrm{kHz}$	$300\mathrm{mT}$	$426\mathrm{kW/m^3}$					
$100\mathrm{kHz}$	$25\mathrm{mT}$	$1.2\mathrm{kW/m^3}$					
$100\mathrm{kHz}$	$50\mathrm{mT}$	$7.8\mathrm{kW/m^3}$					
$100\mathrm{kHz}$	$100\mathrm{mT}$	$51\mathrm{kW/m^3}$	0.0786	1.72	2.74		
$100\mathrm{kHz}$	$200\mathrm{mT}$	$200\mathrm{mT}\qquad 393\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$25\mathrm{mT}$	$6.2\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$50\mathrm{mT}$	$30\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$100\mathrm{mT}$	$175\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$200\mathrm{mT}$	$1.2\mathrm{MW/m^3}$					

Table C.1: Core loss densities of the N87 ferrite material for different excitation frequencies and different peak flux densities (from [160]); the Steinmetz parameters are obtained with a least mean square approximation.



Figure C.1: Comparison of the core loss densities calculated with (C.1) (solid lines) to the core loss densities obtained from the data sheet [160]; ferrite material: N87, core temperature: 25° C.



Figure C.2: Comparison of the core loss densities calculated with (C.1) (solid lines) to the core loss densities obtained from the data sheet [160]; ferrite material: N87, core temperature: 100° C.

f	B_{peak}	$p_{ m core}$	k	α	β		
Ferrite material: 3F3, core temperature: $T_{\text{core}} = 100^{\circ} \text{C}$ [161]							
$25\mathrm{kHz}$	$100\mathrm{mT}$	$12.5\mathrm{kW/m^3}$					
$25\mathrm{kHz}$	$200\mathrm{mT}$	$85\mathrm{kW/m^3}$					
$25\mathrm{kHz}$	$300\mathrm{mT}$	$220\rm kW/m^3$					
$100\mathrm{kHz}$	$60\mathrm{mT}$	$17{ m kW/m^3}$					
$100\mathrm{kHz}$	$70\mathrm{mT}$	$27\mathrm{kW/m^3}$					
$100\mathrm{kHz}$	$100\mathrm{mT}$	$70\mathrm{kW/m^3}$	6.79	1.34	2.63		
$100\mathrm{kHz}$	$200\mathrm{mT}$	$480\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$40\mathrm{mT}$	$20\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$70\mathrm{mT}$	$88\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$100\mathrm{mT}$	$215\mathrm{kW/m^3}$					
$200\mathrm{kHz}$	$200\mathrm{mT}$	$1.15\mathrm{MW/m^3}$					

Table C.2: Core loss densities of the 3F3 ferrite material for different excitation frequencies and different peak flux densities (from [161]); the Steinmetz parameters are obtained with a least mean square approximation.



Figure C.3: Comparison of the core loss densities calculated with (C.1) (solid lines) to the core loss densities obtained from the data sheet [161]; ferrite material: 3F3, core temperature: $100^{\circ}C$.



Figure C.4: Relative temperature dependency of the core losses (ferrite: 3F3) for an excitation frequency of 100 kHz and for a peak flux density of 100 mT [161].

parameters are expected to change due to the change of the core temperature; however, in [161], the temperature dependency of the core losses is only specified for selected combinations of B_{peak} and f (e.g. for $B_{\text{peak}} = 100 \text{ mT}$ and f = 100 kHz in Figure C.4), which is insufficient to calculate k, α , and β . For the DAB converter, the switching frequency is 100 kHz and the peak magnetic flux densities in the transformer core and in the inductor core are typically close to 100 mT. Thus, solely the Steinmetz parameter k is scaled according to the function depicted in Figure C.4 and α and β remain unmodified. This approximation is considered accurate enough for the given DAB converter, since, at rated output power, the total core losses (inductor and transformer) are found to account for less than 10% of the total losses. If more accurate Steinmetz parameters are required, the core losses first need to be measured and, subsequently, k, α , and β are obtained from a least mean square optimization using (C.2).

Appendix D

Evaluation of Power MOSFETs

The decisions on the power semiconductors used for the DAB are mainly based on the maximum blocking voltages and the maximum switch RMS currents calculated in Appendix A.2. On the LV side, MOSFETs are most suitable due to low voltage and high current ratings [162]. On the HV side, either IGBTs or MOSFETs could be used: at high currents, IGBTs offer lower on-state losses than MOSFETs, however, the turn-off losses of IGBTs are considerably higher than the turn-off losses of MOSFETs. Regarding the peak switch currents of up to 24 A during turn-off,¹ the high switching frequency of 100 kHz (Table 1.4), and the required voltage and current ratings of the semiconductor switches, MOSFETs are found to be superior to IGBTs; therefore, the HV side full bridge also employs MOSFETs.

D.1 HV Side

D.1.1 Selection of the MOSFETs used for the DAB Converter

The HV side full bridge of the DAB converter requires MOSFETs with a maximum blocking voltage of 450 V. Considering a safety margin of 30%, the drain-source breakdown voltage of the selected MOSFET, $V_{(BR)DSS}$, should be at least 585 V. The maximum switch RMS current is 11 A for phase shift modulation [Table A.1 and (4.18)].

With the alternative modulation schemes, detailed in Section 5, hard switching operation does not occur during steady-state operation of the DAB. During transient operation, however, sporadic hard switching processes may occur. Therefore, acceptable reverse recovery characteristics of the MOSFETs' body diodes is required in order to ensure reliable converter operation.

¹Efficiency optimal operation of the DAB, Table A.1.

Considering the MOSFETs listed in Table D.1, three devices are most suitable: IXFH36N60P (IXYS), SPW47N60CFD (Infineon), and STW77N65M5 (ST Microelectronics). Amongst these MOSFETs, the STW77N65M5 facilitates minimum conduction losses (≈ 3.6 W per MOSFET at $I_{\rm D} = 11$ A and $T_{\rm j} = 25^{\circ}$ C; typ. $R_{\rm DS,on} = 30$ m Ω); however, the reverse recovery charge of the body diode is comparably large (Table D.2). Conversely, the IXFH36N60P contains a body diode with a low reverse recovery charge but the device generates high conduction losses. The SPW47N60CFD presents a trade-off; its low on-state resistance (typ. 70 m Ω at $I_{\rm D} = 29$ A, $V_{\rm GS} = 10$ V, and $T_{\rm j} = 25^{\circ}$ C) allows for comparably low conduction losses (≈ 8.5 W per MOSFET at $I_{\rm D} = 11$ A and $T_{\rm j} = 25^{\circ}$ C) and the low reverse recovery charge ensures reliable converter operation.

D.1.2 Drain-Source Breakdown Voltages and On-State Resistances

On the HV side, all single-stage topologies regarded in Appendix A require semiconductor switches with a rated voltage of 450 V. However, for the twostage topologies discussed in Chapter 7, semiconductors with a different voltage rating may be considered. Therefore, in Table D.1, a selection of MOS-FETs with different drain-source breakdown voltages is presented. In order to allow for a comparison, all considered MOSFETs employ the TO-247 package. Moreover, each table entry gives the device with the maximum available continuous drain current amongst all devices available in the TO-247 package for a certain type series.

Most commonly, the vertical MOSFET structure is used to realize power MOSFETs in order to achieve both: a low on-state resistance and a high blocking voltage capability. According to [150], the on-state resistance of the device is almost identical to the resistance of the epitaxial layer, $R_{\rm DS,on} \approx$ $R_{\rm epi}$, for high blocking voltages. The relative resistance of the epitaxial layer, $A_{\rm chip} \cdot R_{\rm epi}$, scales according to:

$$A_{\rm chip} \cdot R_{\rm epi} \propto V_{\rm (BR)DSS}^{\frac{5}{2}}$$
 (D.1)

 $(A_{\rm chip})$ is the surface of the semiconductor) [150]. With increasing breakdown voltage the relative resistance $A_{\rm chip} \cdot R_{\rm epi}$ increases by more than $V_{\rm (BR)DSS}^2$. Thus, a MOSFET with a higher breakdown voltage requires a larger chip area in order to maintain a given current carrying capability for a given maximum junction temperature.

The so-called super-junction devices facilitate a more effective utilization of the available chip surface. There, the relative resistance of the epitaxial layer scales according to [150]:

$$A_{\rm chip} \cdot R_{\rm epi} \propto V_{\rm (BR)DSS}^{\overline{6}}$$
 (D.2)



Figure D.1: On-state resistances of selected MOSFETs in the TO-247 package with drain-source breakdown voltages $V_{(BR)DSS}$ between 200 V and 1200 V.

Thus, the chip area required to maintain a given current carrying capability (for a given maximum junction temperature) becomes smaller if the breakdown voltage increases.

Figure D.1 depicts the on-state resistances of the MOSFETs listed in Table D.1 at $T_{\rm j} = 25^{\circ}$ C. For $V_{\rm (BR)DSS} > 300$ V, the on-state resistances of the IXYS power MOSFETs increase approximately proportional to $V_{\rm (BR)DSS}^{2.39}$. Conversely, the on-state resistances of the super-junction CoolMOS MOS-FETs scale approximately proportional to $V_{\rm (BR)DSS}^{1.14}$.²

Except for the CoolMOS with $V_{(BR)DSS} = 900 \text{ V}$, IPW90R120C3, highly optimized MOSFETs are only available for $V_{(BR)DSS}$ close to 600 V. Besides, the IPW90R120C3 is not considered to be used for the DAB due to the large reverse recovery charge of the body diode (Table D.2).

²Provided that all the listed power MOSFETs employ the same chip areas, the relations between $R_{\text{DS,on}}$ and $V_{(\text{BR})\text{DSS}}$, given in Figure D.1, directly apply to $A_{\text{chip}} \cdot R_{\text{epi}}$. Thus, a vertical MOSFET structure seems to be employed for the IXYS MOSFETs; the CoolMOS MOSFETs employ the super-junction technology.

Manufacturer's part designation	$\begin{array}{c} Drain-source\\ breakdown\\ voltage\\ V_{\rm (BR)DSS} \end{array}$	$\begin{array}{c} Continuous\\ drain\ current\\ I_{\rm D}\ (case\\ temp.:\ 25^{\circ}{\rm C}) \end{array}$	Max. on-state resistance $R_{\rm DS,on}$ (V _{GS} = 10 V, $T_{\rm j}$ = 25°C)				
IXYS, Power MOSFETs with fast intrinsic diode, TO-247 package							
IXFH120N20P	$200\mathrm{V}$	120 A	$22\mathrm{m}\Omega~(I_\mathrm{D}=60\mathrm{A})$				
IXFH100N25P	$250\mathrm{V}$	$100\mathrm{A}$	$27 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 50 \mathrm{A})$				
IXFH88N30P	$300\mathrm{V}$	88 A	$40 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 44 \mathrm{A})$				
IXFH52N50P2	$500\mathrm{V}$	$52 \mathrm{A}$	$120 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 26 \mathrm{A})$				
IXFH36N60P	$600\mathrm{V}$	$36\mathrm{A}$	$190 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 18 \mathrm{A})$				
IXFH24N80P	$800\mathrm{V}$	$24\mathrm{A}$	$400\mathrm{m}\Omega~(I_\mathrm{D}=12\mathrm{A})$				
IXFH24N90P	$900\mathrm{V}$	$24\mathrm{A}$	$420\mathrm{m}\Omega~(I_\mathrm{D}=12\mathrm{A})$				
IXFH20N100P	$1000\mathrm{V}$	$20 \mathrm{A}$	$570 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 10 \mathrm{A})$				
IXFH16N120P	$1200\mathrm{V}$	$16\mathrm{A}$	$950 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 8 \mathrm{A})$				
CoolMOS, C3 series, TO-247 package							
SPW52N50C3	$500\mathrm{V}$	$52\mathrm{A}$	$70 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 30 \mathrm{A})$				
SPW47N60C3 600 V		$47\mathrm{A}$	$70 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 30 \mathrm{A})$				
SPW47N65C3 650 V		$47\mathrm{A}$	$70 \mathrm{m}\Omega ~(I_\mathrm{D} = 30 \mathrm{A})$				
IPW90R120C3	$900 \mathrm{V}$	$36\mathrm{A}$	$120\mathrm{m}\Omega~(I_\mathrm{D}=26\mathrm{A})$				
Selected miscellaneous power MOSFETs, TO-247 package							
IPW60R041C6	$600\mathrm{V}$	$77.5\mathrm{A}$	$41 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 44.4 \mathrm{A})$				
SPW47N60CFD	$600\mathrm{V}$	$46\mathrm{A}$	$83 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 29 \mathrm{A})$				
STW77N65M5	$650\mathrm{V}$	69 A	$38 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 34.5 \mathrm{A})$				

Table D.1: Drain-source breakdown voltages $V_{(BR)DSS}$, continuous drain currents I_D , and maximum on-state resistances $R_{DS,on}$ of the power MOSFETs used for the survey depicted in Figure D.1 (data sheet information obtained from [153,158,163]). All listed MOSFETs employ the TO-247 package. Moreover, amongst all devices available in the TO-247 package for a certain type series, each table entry gives the device with the maximum available continuous drain current.

Manufacturer's part	$V_{\rm (BR)DSS}$	Body diode: reverse recovery charge $Q_{\rm rr}$, peak reverse recovery current $I_{\rm rrm}$				
designation		$Q_{ m rr}$	$I_{ m rrm}$	$\begin{array}{l} Measurement \ conditions \\ (besides \\ -\mathrm{d}i_{\mathrm{D}}/\mathrm{d}t = 100 \ \mathrm{A}/\mathrm{\mu s}, \\ T_{\mathrm{j}} = 25^{\circ}\mathrm{C}) \end{array}$		
IXYS, Power MO	SFETs with	n fast intr	insic di	ode, TO-247 package		
IXFH120N20P	$200\mathrm{V}$	$0.4\mu C$	6 A	$I_{\rm D} = -25 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		
IXFH100N25P	$250\mathrm{V}$	$0.6\mu\mathrm{C}$	$10\mathrm{A}$	$I_{\rm D} = -25 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		
IXFH88N30P	$300\mathrm{V}$	$0.6\mu\mathrm{C}$		$I_{\rm D} = -25 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		
IXFH52N50P2	$500\mathrm{V}$	$1.3\mu C$	$14\mathrm{A}$	$I_{\rm D} = -26 \mathrm{A}, V_{\rm DS} = 85 \mathrm{V}$		
IXFH36N60P	$600\mathrm{V}$	$0.8\mu\mathrm{C}$	$6\mathrm{A}$	$I_{\rm D} = -25 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		
IXFH24N80P	$800\mathrm{V}$	$0.8\mu\mathrm{C}$	$6 \mathrm{A}$	$I_{\rm D} = -25 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		
IXFH24N90P	$900\mathrm{V}$	$1.1\mu\mathrm{C}$	$11\mathrm{A}$	$I_{\rm D} = -12 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		
IXFH20N100P	$1000\mathrm{V}$	$0.9\mu\mathrm{C}$	$9\mathrm{A}$	$I_{\rm D} = -10 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		
IXFH16N120P	$1200\mathrm{V}$	$0.75\mu\mathrm{C}$	$7.5\mathrm{A}$	$I_{\rm D} = -8 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		
Infineon, CoolMC	OS, C3 serie	s, TO-247	packag	ge		
SPW52N50C3	$500\mathrm{V}$	$20\mu C$	$70\mathrm{A}$	$I_{\rm D} = -52 \mathrm{A}, V_{\rm DS} = 380 \mathrm{V}$		
SPW47N60C3	$600\mathrm{V}$	$23\mu\mathrm{C}$	$73\mathrm{A}$	$I_{\rm D} = -47 \mathrm{A}, V_{\rm DS} = 350 \mathrm{V}$		
SPW47N65C3	$650\mathrm{V}$	$19\mu\mathrm{C}$	$56\mathrm{A}$	$I_{\rm D} = -47 \mathrm{A}, V_{\rm DS} = 480 \mathrm{V}$		
IPW90R120C3	$900\mathrm{V}$	$30\mu\mathrm{C}$	$65\mathrm{A}$	$I_{\rm D} = -26 \mathrm{A}, V_{\rm DS} = 400 \mathrm{V}$		
Selected miscellaneous power MOSFETs, TO-247 package						
IPW60R041C6	$600\mathrm{V}$	$32\mu\mathrm{C}$	$62\mathrm{A}$	$I_{\rm D} = -44.4 \mathrm{A}, V_{\rm DS} = 400 \mathrm{V}$		
SPW47N60CFD	$600\mathrm{V}$	$2\mu C$	18 A	$I_{\rm D} = -46 \mathrm{A}, V_{\rm DS} = 480 \mathrm{V}$		
STW77N65M5	$\overline{650\mathrm{V}}$	$12\mu C$	46 A	$I_{\rm D} = -34.5 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$		

Table D.2: Reverse recovery characteristics of the body diodes of the power MOSFETs listed in Table D.1 (data sheet information obtained from [153, 158, 163]).

D.2 LV Side

At the time the LV side full bridge of the DAB was designed, the IRF2804 MOSFET offered the lowest on-state resistance in the TO-220 package (max. $R_{\rm DS,on} = 2.3 \,\mathrm{m\Omega}$ at $T_{\rm j} = 25^{\circ}\mathrm{C}$) for a drain-source breakdown voltage of 40 V (for the DAB, the selection of the LV MOSFETs is detailed in Appendix A.2 on page 289).³ Meanwhile, suitable MOSFETs with an on-state resistance below 2.3 m Ω are available (Table D.3), which could be used instead of the IRF2804.

D.2.1 Drain-Source Breakdown Voltages and On-State Resistances

Some of the topologies designed in Appendix A require switches with blocking voltages different to $16 \,\mathrm{V}$.

- The bidirectional full bridge converter with LV side DC inductor (Appendix A.5) requires switches with a maximum blocking voltage of 37.5 V on the LV side.
- The LV side switches of the bidirectional current doubler (Appendix A.6) and of the bidirectional push pull converter (Appendix A.7) are exposed to a maximum blocking voltage of 75 V (on the assumption of negligible voltage spikes).

Table D.3 and Figure D.2 present suitable MOSFETs (TO-220 package) for this low voltage and high current application. For drain-source breakdown voltages between 100 V and 200 V, the on-state resistance of the selected devices are approximately proportional to $V_{\rm (BR)DSS}^2$ (Figure D.2). However, for $V_{\rm (BR)DSS} < 100$ V, the on-state resistances are considerably higher than the on-state resistances predicted with the least mean square approximations. Thus, for $V_{\rm (BR)DSS} < 100$ V, MOSFETs with a higher breakdown voltage are capable of a higher VA rating, which is demonstrated below.

According to Table D.3, a new converter design may consider the MOS-FETs listed below.

- DAB converter: IPP015N04N with $R_{DS,on} = 1.5 \text{ m}\Omega$;
- full bridge converter: IPP023NE7N3 with R_{DS,on} = 2.3 mΩ (due to possible overvoltage spikes, a drain-source breakdown voltage of at least 1.5 · 37.5 V = 56 V is considered);

³At the time the LV side full bridge of the DAB was designed, only surface-mounted MOSFETs offered a higher power density (in particular the IRF2804S-7P with a maximum on-state resistance of 1.6 m Ω at $T_j = 25^{\circ}$ C). However, compared to through-hole components, it is considerably more difficult to attach surface-mounted devices to a heat sink and therefore, through-hole components are preferred.

• current doubler, push pull converter: IPP041N12N3 with $R_{\rm DS,on} = 4.1 \,\mathrm{m}\Omega$ (due to possible overvoltage spikes, a drain-source breakdown voltage of at least $1.5 \cdot 75 \,\mathrm{V} = 113 \,\mathrm{V}$ is considered).

With the selected MOSFETs, the following conduction losses are calculated for the different converter topologies ($T_{\rm j} = 25^{\circ}$ C is assumed).

• DAB converter: the maximum switch RMS current is 163 A [suboptimal modulation scheme, Table A.1 and (4.25)]; with 4 switches, the total conduction losses are:

$$P_{\rm S2,cond} = 4 \cdot (163 \,\text{A})^2 \cdot 1.5 \,\text{m}\Omega = 159 \,\text{W}.$$
 (D.3)

• Full bridge converter: the maximum switch RMS current is 157 A and with 4 switches, the total conduction losses are:

$$P_{\rm S2,cond} = 4 \cdot (157 \,\text{A})^2 \cdot 2.3 \,\text{m}\Omega = 227 \,\text{W}.$$
 (D.4)

• Current doubler: each switch operates 2 MOSFETs in parallel in order to allow for a comparison with the DAB and with the full bridge converter; the maximum switch RMS current is 159 A. Thus, the total conduction losses are:

$$P_{\rm S2,cond} = 2 \cdot 2 \cdot (79.5 \,\text{A})^2 \cdot 4.1 \,\text{m}\Omega = 104 \,\text{W}. \tag{D.5}$$

• Push pull converter: each switch operates 2 MOSFETs in parallel in order to allow for a comparison with the DAB and with the full bridge converter. The maximum switch RMS current is 157 A and the total conduction losses are:

$$P_{\rm S2,cond} = 2 \cdot 2 \cdot (78.5 \,\mathrm{A})^2 \cdot 4.1 \,\mathrm{m}\Omega = 101 \,\mathrm{W}.$$
 (D.6)

Thus, lowest conduction losses result with the push pull converter, even though the VA rating of the respective switches is higher than for the DAB (push pull converter: 23.5 kVA, DAB: 10.5 VA).

Manufacturer's part designation	$\begin{array}{c} Drain-source\\ breakdown\\ voltage\\ V_{\rm (BR)DSS} \end{array}$	Continuous drain current $I_{\rm D}$ (case temp.: 25°C)	Max. on-state resistance $R_{\rm DS,on}$ (V _{GS} = 10 V, $T_{\rm j} = 25^{\circ}{\rm C}$)				
International Rectifier, HEXFET, TO-220 package							
AUIRF1324	$24\mathrm{V}$	$195\mathrm{A}$	$1.5 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 195 \mathrm{A})$				
IRLB3034PBF	$40\mathrm{V}$	$195\mathrm{A}$	$1.7 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 195 \mathrm{A})$				
IRLB3036PBF	$60\mathrm{V}$	$195\mathrm{A}$	$2.4 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 165 \mathrm{A})$				
IRFB3077GPBF	$75\mathrm{V}$	$120 \mathrm{A}$	$3.3 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 75 \mathrm{A})$				
IRFL4030PBF	$100\mathrm{V}$	$180\mathrm{A}$	$4.3 \mathrm{m}\Omega \ (I_{\rm D} = 110 \mathrm{A})$				
IRFB4115GPBF	$150\mathrm{V}$	$104\mathrm{A}$	$11 \mathrm{m}\Omega (I_{\mathrm{D}} = 62 \mathrm{A})$				
IRFB4127PBF	$200\mathrm{V}$	$76\mathrm{A}$	$20 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 44 \mathrm{A})$				
IRFB4332PBF	$250\mathrm{V}$	$60\mathrm{A}$	$33 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 35 \mathrm{A})$				
Infineon, OptiMOS, TO-220 package							
IPP015N04N	$40\mathrm{V}$	120 A	$1.5 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 100 \mathrm{A})$				
IPP023NE7N3	$75\mathrm{V}$	$120 \mathrm{A}$	$2.3 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 100 \mathrm{A})$				
IPP028N08N3	$80\mathrm{V}$	$100 \mathrm{A}$	$2.8 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 100 \mathrm{A})$				
IPP030N10N3	$100 \mathrm{V}$	$100 \mathrm{A}$	$3.0 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 100 \mathrm{A})$				
IPP041N12N3	$120\mathrm{V}$	$120 \mathrm{A}$	$4.1 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 100 \mathrm{A})$				
IPP075N15N3	$150\mathrm{V}$	$100 \mathrm{A}$	$7.5 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 100 \mathrm{A})$				
IPP110N20N3	$200\mathrm{V}$	88 A	$11 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 88 \mathrm{A})$				
IPP200N25N3	$250\mathrm{V}$	$64\mathrm{A}$	$20 \mathrm{m}\Omega \ (I_{\mathrm{D}} = 64 \mathrm{A})$				

Table D.3: Drain-source breakdown voltages $V_{(BR)DSS}$, continuous drain currents I_D , and maximum on-state resistances $R_{DS,on}$ of the power MOSFETs used for the survey depicted in Figure D.2 (data sheet information obtained from [153,164]). All listed MOSFETs employ the TO-220 package. Moreover, amongst all devices available in the TO-220 package for a certain type series, each table entry gives the device with the maximum available continuous drain current.

Manufacturer's part	$V_{\rm (BR)DSS}$	Body diode: reverse recovery charge $Q_{\rm rr}$, peak reverse recovery current $I_{\rm rrm}$			
designation		$Q_{ m rr}$	$I_{\rm rrm}$	$\begin{array}{l} Measurement \ conditions \\ (besides \\ -di_{\rm D}/dt = 100 \ {\rm A}/\mu{\rm s}, \\ T_{\rm j} = 25^{\circ}{\rm C}) \end{array}$	
International Rec	tifier, HEXI	FET, TO	-220 pa	ckage	
AUIRF1324	$24\mathrm{V}$	$160\mathrm{nC}$	$7.7\mathrm{A}$	$I_{\rm D} = -195 \mathrm{A}, V_{\rm DS} = 20 \mathrm{V}$	
IRLB3034PBF	$40\mathrm{V}$	$39\mathrm{nC}$	$1.7\mathrm{A}$	$I_{\rm D} = -195 \mathrm{A}, V_{\rm DS} = 34 \mathrm{V}$	
IRLB3036PBF	$60\mathrm{V}$	$310\mathrm{nC}$	$4.4\mathrm{A}$	$I_{\rm D} = -165 \mathrm{A}, V_{\rm DS} = 51 \mathrm{V}$	
IRFB3077GPBF	$75\mathrm{V}$	$89\mathrm{nC}$	$2.5\mathrm{A}$	$I_{\rm D} = -75 \mathrm{A}, V_{\rm DS} = 64 \mathrm{V}$	
IRLB4030PBF	$100\mathrm{V}$	$88\mathrm{nC}$	$3.3\mathrm{A}$	$I_{\rm D} = -110 \mathrm{A}, V_{\rm DS} = 85 \mathrm{V}$	
IRFB4115GPBF	$150\mathrm{V}$	$300\mathrm{nC}$	$6.5\mathrm{A}$	$I_{\rm D} = -62 \mathrm{A}, V_{\rm DS} = 130 \mathrm{V}$	
IRFB4127PBF	$200\mathrm{V}$	$458\mathrm{nC}$	$8.3\mathrm{A}$	$I_{\rm D} = -44 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$	
IRFB4332PBF	$250\mathrm{V}$	$820\mathrm{nC}$		$I_{\rm D} = -35 \mathrm{A}, V_{\rm DS} = 50 \mathrm{V}$	
Infineon, OptiMO	S, TO-220	package			
IPP015N04N	$40\mathrm{V}$	$141\mathrm{nC}$		$I_{\rm D} = -120 \mathrm{A}, V_{\rm DS} = 15 \mathrm{V}$	
IPP023NE7N3	$75\mathrm{V}$	$129\mathrm{nC}$		$I_{\rm D} = -120 \mathrm{A}, V_{\rm DS} = 37.5 \mathrm{V}$	
IPP028N08N3	$80\mathrm{V}$	$317\mathrm{nC}$		$I_{\rm D} = -100 \mathrm{A}, V_{\rm DS} = 40 \mathrm{V}$	
IPP030N10N3	$100\mathrm{V}$	$232\mathrm{nC}$		$I_{\rm D} = -100 \mathrm{A}, V_{\rm DS} = 50 \mathrm{V}$	
IPP041N12N3	$120\mathrm{V}$	$356\mathrm{nC}$		$I_{\rm D} = -120 \mathrm{A}, V_{\rm DS} = 60 \mathrm{V}$	
IPP075N15N3	$150\mathrm{V}$	$478\mathrm{nC}$		$I_{\rm D} = -100 \mathrm{A}, V_{\rm DS} = 75 \mathrm{V}$	
IPP110N20N3	$200\mathrm{V}$	$640\mathrm{nC}$	_	$I_{\rm D} = -88 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$	
IPP200N25N3	$250\mathrm{V}$	$780\mathrm{nC}$	—	$I_{\rm D} = -64 \mathrm{A}, V_{\rm DS} = 100 \mathrm{V}$	

Table D.4: Reverse recovery characteristics of the body diodes of the power MOSFETs listed in Table D.3 (data sheet information obtained from [153, 164]).



Figure D.2: On-state resistances of selected MOSFETs in the TO-220 package with drain-source breakdown voltages $V_{(BR)DSS}$ between 24 V and 250 V.

Appendix E

Practical Implementation

E.1 Digital Control: Concept

A digital signal processor (DSP) controls the converter currents and voltages according to the block diagram depicted in Figure 6.2. The DSP runs the PI controllers, interfaces to current and voltage measurement peripherals (A/D converters), monitors overcurrent and overvoltage conditions in order to ensure safe converter operation, and communicates with a control PC which is used to start and stop the bidirectional DC-DC converter and to adjust the output voltage. The DSP as well calculates the different timing parameters, i.e. the phase shift time $\varphi/(2\pi f_S)$, the HV side duty cycle duration D_1/f_S , and the LV side duty cycle duration D_2/f_S (cf. Section E.2), and interfaces to a programmable logic device (field programmable gate array, FPGA) that generates the dedicated gate signals (Section E.3).

The employed DSP is a TMS320F2808 (Texas Instruments) operated with a clock frequency of 100 MHz; the software is implemented in the computer language C. For the FPGA, the device LCMXO2280, Lattice Semiconductor, operated with a clock frequency of 100 MHz, is employed; the hardware description language VHDL is used to program the FPGA.

E.2 On-Line Calculation of the Modulation Parameters

Except for phase shift modulation, elaborate numerical calculations are required in order to determine the 3 timing parameters φ , D_1 , and D_2 depending on the operating point defined with V_1 , V_2 , and P_{out} (Section 5). With the given computational power of the DSP, on-line parameter calculation is not feasible. In order to still run the discussed extended modulation methods, the respective timing parameters are calculated off-line for a set of basic values and stored in a table on the DSP. The controller software then uses linear interpolation to approximately determine the actual timing parameters.

Each timing parameter depends on the current operating point, characterized by V_1 , V_2 , and P_{out} : $\varphi(V_1, V_2, P_{out})$, $D_1(V_1, V_2, P_{out})$, and $D_2(V_1, V_2, P_{out})$. Consequently, a 3-D table is required for every timing parameter to store the basic values; an appropriate algorithm then interpolates the function values between the basic values.

In order to develop the 3-D interpolation algorithm, a simplified 2-D version is first discussed based on a table with x- and y-axis and is then extended to 3-D. For the sake of clarity, the input values to the underlying 2-D function f(x, y) are confined to $0 \le x \le 1$ and $0 \le y \le 1$ with four surrounding basic values f(0, 0), f(0, 1), f(1, 0), and f(1, 1), obtained from the 2-D table. The interpolated function values at x = 0 and x = 1 are calculated using linear expressions,

$$\begin{split} & f_i(0,y) = f(0,0) + [f(0,1) - f(0,0)] \cdot y, \\ & f_i(1,y) = f(1,0) + [f(1,1) - f(1,0)] \cdot y. \end{split}$$

The obtained values $f_i(0, y)$ and $f_i(1, y)$ are then used to calculate the interpolated function value $f_i(x, y)$,

$$f_{i}(x,y) = f_{i}(0,y) + [f_{i}(1,y) - f_{i}(0,y)] \cdot x.$$
(E.1)

This method is extended to a 3-D function f(x, y, z) with $0 \le x \le 1$, $0 \le y \le 1$, $0 \le z \le 1$ and 8 surrounding table values $f(0, 0, 0) \ldots f(1, 1, 1)$ in a straight forward manner:

$$\begin{aligned} & f_{i}(0,0,z) = f(0,0,0) + [f(0,0,1) - f(0,0,0)] \cdot z, \\ & f_{i}(0,1,z) = f(0,1,0) + [f(0,1,1) - f(0,1,0)] \cdot z, \\ & f_{i}(1,0,z) = f(1,0,0) + [f(1,0,1) - f(1,0,0)] \cdot z, \\ & f_{i}(1,1,z) = f(1,1,0) + [f(1,1,1) - f(1,1,0)] \cdot z, \\ & f_{i}(0,y,z) = f_{i}(0,0,z) + [f_{i}(0,1,z) - f_{i}(0,0,z)] \cdot y, \\ & f_{i}(1,y,z) = f_{i}(1,0,z) + [f_{i}(1,1,z) - f_{i}(1,0,z)] \cdot y, \\ & f_{i}(x,y,z) = f_{i}(0,y,z) + [f_{i}(1,y,z) - f_{i}(0,y,z)] \cdot x. \end{aligned}$$
(E.2)

The calculation of $f_i(x, y, z)$ requires a total of 7 multiplications and 14 additions or subtractions.

The current implementation uses 16 basic values $V_{1,k}$ and $V_{2,l}$,

$$V_{1,k} = V_{1,\min} + k \frac{V_{1,\max} - V_{1,\min}}{15}, \ k \in \mathbb{N}_0 \land 0 \le k \le 15,$$
(E.3)

$$V_{2,l} = V_{2,\min} + l \frac{V_{2,\max} - V_{2,\min}}{15}, \ l \in \mathbb{N}_0 \land 0 \le l \le 15,$$
(E.4)

and 32 values $P_{\text{out},m}$:

$$P_{\text{out},m} = -P_{\text{out},\max} + m \frac{2P_{\text{out},\max}}{31}, \ m \in \mathbb{N}_0 \land 0 \le m \le 31.$$
 (E.5)

In total, the tables for D_1 , D_2 , and φ require $3 \cdot 16^2 \cdot 32 = 24576$ memory cells; this fits into the on-chip FLASH of the TMS320F2808.

Depending on the operating point, the actual converter output power may deviate from the set output power by reason of the linear interpolation. The maximum output power errors occur approximately in the middle between 8 adjacent basic values, i.e.

$$V_{1,e,k} = \frac{V_{1,k} + V_{1,k+1}}{2}, \ k \in \mathbb{N}_0 \land 0 \le k < 15,$$
(E.6)

$$V_{2,e,l} = \frac{V_{2,l} + V_{2,l+1}}{2}, \ l \in \mathbb{N}_0 \land 0 \le l < 15, \text{ and}$$
 (E.7)

$$P_{\text{out},e,m} = \frac{P_{\text{out},e,m} + P_{\text{out},e,m+1}}{2}, \ m \in \mathbb{N}_0 \land 0 \le m < 31,$$
(E.8)

is considered in order to calculate the absolute and relative errors:

$$e_{\mathrm{P}}(V_{1,\mathrm{e},k}, V_{2,\mathrm{e},l}, P_{\mathrm{out},\mathrm{e},m}) = P_{\mathrm{out}} - P_{\mathrm{out},\mathrm{i}}(V_{1,\mathrm{e},k}, V_{2,\mathrm{e},l}, P_{\mathrm{out},\mathrm{e},m}), \quad (E.9)$$

$$e_{P,r}(V_{1,e,k}, V_{2,e,l}, P_{out,e,m}) = \frac{e_P(V_{1,e,k}, V_{2,e,l}, P_{out,e,m})}{P_{out}}$$
(E.10)

 $(P_{\text{out,i}}$ denotes the output power calculated with the interpolated duty cycles and the interpolated phase shift angle). Table E.1 summarizes the errors calculated for the different modulation schemes. The maximum absolute error of 107 W occurs for phase shift modulation; with the extended triangular and trapezoidal current mode modulation, a maximum relative error of 15% occurs. In order to eliminate the error due to the linear interpolation, the current controller discussed in Section 6.3 is required.

E.3 Flexible Gate Signal Generation Unit

The voltage sequences required for the different modulation schemes demand for a highly flexible gate signal generation unit. The gate signal generation is therefore performed with finite state machines (FSMs) being executed in the FPGA; this approach allows for very high flexibility regarding the generation of different gate signal pulse patterns.

The gate signal generation unit consists of 6 different state machines: the master FSM (Figure E.1) and its supporting subsidiary FSM (Figure E.2) control 4 subordinate FSMs which in turn generate the gate signals (Figure E.3). Let us assume, that the state "POS1" of the master FSM becomes active with the configured direction of power transfer being "HV \rightarrow LV".

Phase shift modulation:							
$\max e_{\mathrm{P}} $	$ e_{\mathrm{P}} $	$\max e_{\mathrm{P,r}} $	$ e_{\mathrm{P,r}} $				
$107\mathrm{W}$	107 W 4.7 W 5.7% 0.32%						
Extended tria	Extended triangular and trapezoidal current mode modulation:						
$51\mathrm{W}$	51 W 4.7 W 15% 0.77%						
Suboptimal modulation scheme:							
$58\mathrm{W}$	$6.2\mathrm{W}$	6.0%	0.83%				

Table E.1: Output power error by reason of the linear interpolation: $|e_{\rm P}|$ and $\overline{|e_{\rm P,r}|}$ denote the average absolute and relative errors. The error values are calculated according to (E.9) and (E.10), whereas the calculation of the relative error excludes zero output power.

The FPGA thus resets the half-cycle counter (required in order to detect the end of the half-cycle), updates the FSM timing parameters, and generates an external DSP interrupt to synchronize the DSP controllers to the FPGA. It further sets the switch signal events $S_{1,2}$ to 1 which triggers the half bridge FSM assigned to T₁ and T₂ (Figure E.3 with i = 1 and j = 2). At the time the half-cycle counter reaches T_1 , the master FSM state changes to "POS2", triggers the half bridge FSM assigned to T₅ and T₆, and resets an independent counter used to control the subsidiary FSM. In order to avoid an erroneous switching sequence, a flag indicating the current half-cycle, HC1_active_flag, is set (Figure E.1). As soon as the half-cycle counter reaches T_2 , the half bridge FSM assigned to T₃ and T₄ is triggered. The master state machine starts with the second half-cycle, as soon as the half-cycle counter elapses.

The use of the independent subsidiary state machine enables the time between the events $S_{5,6}$ and $S_{7,8}$ (which is equivalent to the duration D_2/f_S) to range from 0 to $T_S/2$ —on the assumption of a power transfer direction from the HV port to the LV port. With the extension to both power transfer directions, as shown in Figures E.1 and E.2, the proposed architecture allows to reproduce all required voltage sequences discussed in Chapters 3 and 5.

The use of independent FSMs for each half bridge finally allows the switching events $S_{i,j}$ to be delayed individually in order to compensate for the switching time delays discussed in Sections 4.3.2 and 4.3.3.



Figure E.1: Master state machine: employed for the DAB to control 3 out of 4 half bridge state machines (cf. Figure E.3). The 4th half bridge is controlled with the additional state machine depicted in Figure E.2, since those particular switching commands may occur after the half-cycle has elapsed (cf. Figure 4.29 and Figure 5.29). It is important to note, that the assignments related to each state are only performed at the instant when the FSM enters the respective state.



HC2 active flag $\leftarrow 0$

Figure E.2: Subsidiary part of the master state machine depicted in Figure E.1; controls the state machine of the 4^{th} half bridge (HV \rightarrow LV: $T_7, T_8; LV \rightarrow HV; T_3, T_4$). This additional state machine is required, since the switching action of the 4th half bridge may occur after the half-cycle in the master state machine has elapsed. Therefore, the control of the 4^{th} half bridge needs to be decoupled from the master state machine (cf. Figure 4.29 and Figure 5.29). The state machine uses the variables HC1 active flag and HC2 active flag to prevent failure modes during unsteady converter operation. It is important to note, that the assignments related to each state are only performed at the instant when the FSM enters the respective state.



Figure E.3: State machine $\text{HB}_{i,j}$ employed to generate the gate signals of a single half bridge (inclusive command delay and dead time interval generation). The indices *i* and *j* denote the indices of the MOSFET's reference designators given in the respective circuit diagrams (e.g. Figure 2.10 for the DAB). In case of the DAB, the FPGA runs four of these state machines, i.e. $\text{HB}_{1,2}$, $\text{HB}_{3,4}$, $\text{HB}_{5,6}$, and $\text{HB}_{7,8}$, simultaneously. It is important to note, that the assignments related to each state are only performed at the instant when the FSM enters the respective state.
Appendix F

Supplement to the Small-Signal Model of the DAB

F.1 Small-Signal Model

F.1.1 Matrix Expressions (from [120])

The state space matrix **A** for phase shift modulation (6.10) contains the derivatives of all system state variables at the end of the half-cycle, $\vec{x}(t = T_{\text{DAB}})$, with respect to the state variables at the beginning of the half-cycle, $\vec{x}(t = 0)$,

	$\frac{\partial i_L}{\partial i_{L,0}}$	$\frac{\partial i_L}{\partial i_{f1a,0}}$	$\frac{\partial i_L}{\partial i_{f1b,0}}$	$\frac{\partial i_L}{\partial (i_{\rm f2a},0/n)}$	$\frac{\partial i_L}{\partial (i_{\rm f2b}, 0/n)}$	$\frac{\partial i_L}{\partial v_{\rm f1,0}}$	$\frac{\partial i_L}{\partial (nv_{\mathrm{f2},0})}$	
	$\frac{\partial i_{f1a}}{\partial i_{L,0}}$	$\frac{\partial i_{\rm f1a}}{\partial i_{\rm f1a,0}}$	$\frac{\partial i_{\rm f1a}}{\partial i_{\rm f1b,0}}$	$\frac{\partial i_{\rm f1a}}{\partial (i_{\rm f2a,0}/n)}$	$\frac{\partial i_{\rm f1a}}{\partial (i_{\rm f2b,0}/n)}$	$\frac{\partial i_{\rm f1a}}{\partial v_{\rm f1,0}}$	$\frac{\partial i_{\rm f1a}}{\partial (nv_{\rm f2,0})}$	
	$\frac{\partial i_{\rm f1b}}{\partial i_{L,0}}$	$\frac{\partial i_{\rm f1b}}{\partial i_{\rm f1a,0}}$	$\frac{\partial i_{\rm f1b}}{\partial i_{\rm f1b,0}}$	$\frac{\partial i_{\rm f1b}}{\partial (i_{\rm f2a}, 0/n)}$	$\frac{\partial i_{\rm f1b}}{\partial (i_{\rm f2b}, 0/n)}$	$\frac{\partial i_{\rm f1b}}{\partial v_{\rm f1,0}}$	$\frac{\partial i_{\rm f1b}}{\partial (nv_{\rm f2,0})}$	
$\mathbf{A} = \lim$	$\frac{\partial(i_{f2a}/n)}{\partial i_{L}}$	$\frac{\partial(i_{f2a}/n)}{\partial i_{f1a}}$	$\frac{\partial(i_{f2a}/n)}{\partial i_{f1b}}$	$\frac{\partial(i_{f2a}/n)}{\partial(i_{f2a}, o/n)}$	$\frac{\partial(i_{f2a}/n)}{\partial(i_{f2b}, o/n)}$	$\frac{\partial(i_{f2a}/n)}{\partial v_{f1}}$	$\frac{\partial(i_{f2a}/n)}{\partial(nv_{f2a})}$,
$t \rightarrow T_{\text{DAB}}$	$\frac{\partial(i_{f2b}/n)}{\partial i}$	$\frac{\partial(i_{f2b}/n)}{\partial i_{f2b}}$	$\frac{\partial(i_{f2b}/n)}{\partial(i_{f2b}/n)}$	$\frac{\partial(i_{f2b}/n)}{\partial(i_{f2b}/n)}$	$\frac{\partial(i_{f2b}/n)}{\partial(i_{f2b}/n)}$	$\frac{\partial(i_{f2b}/n)}{\partial m}$	$\frac{\partial(i_{f2b}/n)}{\partial(i_{f2b}/n)}$	
	$\frac{\partial v_{f1}}{\partial v_{f1}}$	$\frac{\partial v_{f1a,0}}{\partial v_{f1}}$	$\frac{\partial v_{f1}}{\partial v_{f1}}$	$\frac{\partial v_{f1}}{\partial v_{f1}}$	$\frac{\partial v_{f1}}{\partial v_{f1}}$	$\frac{\partial v_{f1,0}}{\partial v_{f1}}$	$\frac{\partial v_{f1}}{\partial v_{f1}}$	
	$\frac{\partial (nv_{f2})}{\partial i}$	$\frac{\partial (nv_{f2})}{\partial i}$	$\frac{\partial (nv_{f2})}{\partial (nv_{f2})}$	$\frac{\partial(nv_{f2})}{\partial(i}$	$\frac{\partial(nv_{f2})}{\partial(i_{f2},0/n)}$	$\frac{\partial v_{f1,0}}{\partial (nv_{f2})}$	$\frac{\partial(nv_{f2,0})}{\partial(nv_{f2})}$	
	$\Box Di_{L,0}$	$o_{i_{\mathrm{f1a},0}}$	$o_{i_{\rm f1b},0}$	$O(i_{f2a,0}/n)$	$O(i_{f2b}, 0/n)$	$\partial v_{\mathrm{f1,0}}$	$O(nv_{f2,0})$	1)

and is evaluated at the steady-state operating point $\vec{x}_{st}(0)$. **B**_{PS} and **C** contain the derivatives of $\vec{x}(t = T_{\text{DAB}})$ with respect to the input variables,

$$\mathbf{B}_{\mathbf{PS}} = \lim_{t \to T_{\mathrm{DAB}}} \begin{bmatrix} \frac{\partial i_L}{\partial T_{\varphi,0}} \\ \frac{\partial i_{\mathrm{fla}}}{\partial T_{\varphi,0}} \\ \frac{\partial i_{\mathrm{fla}}}{\partial T_{\varphi,0}} \\ \frac{\partial (i_{\mathrm{fla}}/n)}{\partial T_{\varphi,0}} \\ \frac{\partial (i_{\mathrm{fla}/n}/n)}{\partial T_{\varphi,0}} \\ \frac{\partial (i$$

For the alternative modulation (6.17), **A** and **C** are again derived according to (F.1) and (F.2), only $\mathbf{B_{PS}}$ changes to $\mathbf{B_{alt}}$,

$$\mathbf{B}_{\mathbf{alt}} = \lim_{t \to T_{\mathrm{DAB}}} \begin{bmatrix} \frac{\partial i_L}{\partial T_{1,0}} & \frac{\partial i_L}{\partial T_{2,0}} & \frac{\partial i_L}{\partial T_{3,0}} \\ \frac{\partial i_{\mathrm{f1a}}}{\partial T_{1,0}} & \frac{\partial i_{\mathrm{f1a}}}{\partial T_{2,0}} & \frac{\partial i_{\mathrm{f1a}}}{\partial T_{3,0}} \\ \frac{\partial i_{\mathrm{f1b}}}{\partial T_{1,0}} & \frac{\partial i_{\mathrm{f1b}}}{\partial T_{2,0}} & \frac{\partial i_{\mathrm{f1b}}}{\partial T_{3,0}} \\ \frac{\partial (i_{\mathrm{f2a}/n})}{\partial T_{1,0}} & \frac{\partial (i_{\mathrm{f2a}/n})}{\partial T_{2,0}} & \frac{\partial (i_{\mathrm{f2a}/n})}{\partial T_{3,0}} \\ \frac{\partial (i_{\mathrm{f2b}/n})}{\partial T_{1,0}} & \frac{\partial (i_{\mathrm{f2b}/n})}{\partial T_{2,0}} & \frac{\partial (i_{\mathrm{f2b}/n})}{\partial T_{3,0}} \\ \frac{\partial (i_{\mathrm{f2b}/n})}{\partial T_{1,0}} & \frac{\partial (i_{\mathrm{f2b}/n})}{\partial T_{2,0}} & \frac{\partial (i_{\mathrm{f2b}/n})}{\partial T_{3,0}} \\ \frac{\partial (nv_{\mathrm{f2}})}{\partial T_{1,0}} & \frac{\partial (nv_{\mathrm{f2}})}{\partial T_{2,0}} & \frac{\partial (nv_{\mathrm{f2}})}{\partial T_{3,0}} \\ \frac{\partial (nv_{\mathrm{f2}})}{\partial T_{3,0}} & \frac{\partial (nv_{\mathrm{f2}})}{\partial T_{3,0}} & \frac{\partial (nv_{\mathrm{f2}})}{\partial T_{3,0}} \end{bmatrix}.$$
(F.3)

B_{PS}, **B**_{alt}, and **C** are again evaluated at the steady-state operating point $\vec{x}_{st}(0)$.

F.1.2 Small-Signal Transfer Function

The z-domain control to output transfer function is derived with equations (71) and (74) in [120],

$$\mathbf{G} = \mathbf{E}^{\mathrm{T}} \cdot \left[z_{\mathrm{DAB}} \mathbf{I} - \mathbf{Q}(k) \mathbf{RP}(k) \mathbf{AP}(k) \mathbf{Q}(k) \right]^{-1} \cdot \mathbf{Q}(k) \mathbf{RP}(k) \mathbf{B}_{\mathbf{PS}}, \quad (\mathrm{F.4})$$

with **R** being defined in (6.12) in Chapter 6, an integer number k, and

$$\mathbf{P}_{7\times7}(k) = \begin{bmatrix} (-1)^k & \mathbf{0} \\ & 1\times6 \\ \mathbf{0} & \mathbf{I} \\ & 6\times1 & 6\times6 \end{bmatrix}, \ \mathbf{Q}_{7\times7}(k) = \begin{bmatrix} (-1)^k \cdot \operatorname{sgn}(i_{L,0}) & \mathbf{0} \\ & 1\times6 \\ & \mathbf{0} & \mathbf{I} \\ & 6\times1 & 6\times6 \end{bmatrix}.$$
(F.5)

The products $\mathbf{P}(k)\mathbf{Q}(k)$ and $\mathbf{Q}(k)\mathbf{RP}(k)$ are equal to

$$\mathbf{Q}(k)\mathbf{RP}(k) = \begin{bmatrix} -\operatorname{sgn}(i_{L,0}) & \mathbf{0} \\ \mathbf{0} & \mathbf{I} \\ 6 \times 1 & 6 \times 6 \end{bmatrix} = \mathbf{Q}(0)\mathbf{R},$$

$$\mathbf{P}(k)\mathbf{Q}(k) = \begin{bmatrix} \operatorname{sgn}(i_{L,0}) & \mathbf{0} \\ \mathbf{0} & \mathbf{I} \\ 6 \times 1 & 6 \times 6 \end{bmatrix} = \mathbf{Q}(0) \text{ for } k \in \mathbb{Z},$$
(F.6)

and thus independent on k which allows for the simplified representation (6.11), (6.12).

F.2 Sampling Rate Reduction

For a given discrete time system with sampling rate T, step response function h(kT), $k \in \mathbb{N}_0$, and corresponding z-domain transfer function G(z), the transfer function $G_r(z)$ with increased sampling time $T_r = N \cdot T$, $N \in \mathbb{N}$ and similar step response function $h_r(kT_r) = h(kNT)$, is derived.

The original z-domain transfer function

$$G(z) = \frac{a(z)}{b(z)} \tag{F.7}$$

contains two polynomials a(z) and b(z) with constant and real coefficients and deg [a(z)] < deg [b(z)].

Based on the relation between the *i*-th pole $p_{s,i}$ of a continuous time transfer function and the *i*-th pole $p_{z,i}$ of the corresponding discrete time transfer function [151],

$$p_{z,i} = e^{p_{s,i}T},\tag{F.8}$$

the *i*-th pole $p_{rz,i}$ of $G_r(z)$ can be expressed as

$$p_{rz,i} = e^{p_{s,i}T_r} = e^{p_{s,i}NT} = p_{z,i}^N.$$
 (F.9)

The method discussed in [151] then represents a simple way to obtain the numerator $a_r(z)$ of $G_r(z)$ based on the step response $h_r(kT_r)$ of the corresponding time domain function which is equal to the step response of the already known function $h(k_N T)$ for $k_N = N \cdot k$ (this can be achieved with the Matlab command d2d as well).

F.3 Derivation of G_{DAB}

For the sake of clarity, the derivation of G_{DAB} is based on a simplified singleinput, single-output system with input X_{in} and output Y_{out} in **Figure F.1**.



Figure F.1: Single-input, single-output system with internal feedback used to derive G_{DAB} .

It contains the multiple-input, multiple-output transfer function $\mathbf{G}(z)$ in the forward path and $\mathbf{H}(z)$ in the feedback path. Vector \vec{X} denotes the input of $\mathbf{G}(z)$,

$$\vec{X} = \left(X_{\text{in}} \quad \vec{X}_{\text{r}}^{\text{T}}\right)^{\text{T}}, \quad \vec{X}_{\text{r}} \in \mathbb{C}^{n},$$
 (F.10)

with the system input value X_{in} and internal system values in \vec{X}_r . The output of $\mathbf{G}(z)$,

$$\vec{Y} = \mathbf{G} \cdot \vec{X} = \left(Y_{\text{out}} \quad \vec{Y}_{\text{s}}^{\text{T}}\right)^{\text{T}}, \quad \vec{Y}_{\text{s}} \in \mathbb{C}^{m},$$
 (F.11)

contains Y_{out} and another set of internal system values \vec{Y}_{s} that denote the inputs to $\mathbf{H}(z)$. The output of $\mathbf{H}(z)$ is then equal to \vec{X}_{r} ,

$$\vec{X}_{\rm r} = \mathop{\rm \mathbf{H}}_{n \times m} \cdot \vec{Y}_{\rm s},\tag{F.12}$$

in order to close the feedback loop. The transfer function matrix ${\bf G}$ is split according to

$$\mathbf{G} = \begin{bmatrix} G_{00} & \mathbf{G_{0r}} \\ & 1 \times n \\ \mathbf{G_{s0}} & \mathbf{G_{sr}} \\ & m \times 1 & m \times n \end{bmatrix}.$$
 (F.13)

With (F.10), (F.11), (F.12), and (F.13) the system output value $Y_{\rm out}$ and the vector $\vec{Y}_{\rm s}$ become

$$Y_{\rm out} = G_{00}X_{\rm in} + \mathbf{G_{0r}}\vec{X_{\rm r}} = G_{00}X_{\rm in} + \mathbf{G_{0r}}\mathbf{H}\vec{Y_{\rm s}}$$
(F.14)

$$\vec{Y}_{\rm s} = \mathbf{G_{s0}} X_{\rm in} + \mathbf{G_{sr}} \vec{X}_{\rm r} = \mathbf{G_{s0}} X_{\rm in} + \mathbf{G_{sr}} \mathbf{H} \vec{Y}_{\rm s}$$
(F.15)

and thus

$$\vec{Y}_{s} = \left(\mathbf{I} - \mathbf{G}_{sr}\mathbf{H}\right)^{-1}\mathbf{G}_{s0}X_{in}.$$
 (F.16)

Based on (F.14) and (F.16) the system transfer function G_{DAB} results:

$$G_{\text{DAB}} = Y_{\text{out}} / X_{\text{in}} = G_{00} + \mathbf{G}_{0\mathbf{r}} \mathbf{H} \left(\mathbf{I} - \mathbf{G}_{\mathbf{sr}} \mathbf{H} \right)^{-1} \mathbf{G}_{\mathbf{s0}}.$$
 (F.17)

Glossary

Commonly Used Abbreviations

HV	High Voltage
LV	Low Voltage
$_{ m HF}$	High Frequency
ZVS	Zero Voltage Switching
DAB	Dual Active Bridge
SRC	Series Resonant Converter
PRC	Parallel Resonant Converter
SPRC	Series-Parallel Resonant Converter
LLC	Series-Parallel Resonant Converter: the resonant network consists of two inductors $(L \text{ and } L_{\text{M}})$ and one capacitor C (cf. Figures 2.17 and 2.18)
LCC	Series-Parallel Resonant Converter: the resonant network consists of one inductor L and two capacitors (C and C_P ; cf. Figures 2.16 and 2.19).

Commonly Used Designators of Physical Properties

Designator	Description
$f_{ m S}$	switching frequency
$T_{\rm S}$	switching period, $T_{\rm S} = 1/f_{\rm S}$
n	transformer turns ratio
L	series inductance of the DAB and LLC converter inductor

Designator	Description
C	series capacitance of the LLC converter capacitor
L_{M}	transformer magnetizing inductance
R	resistor in series to L ; R models the converter conduction
	losses.
$C_{\rm DC1}$	HV side DC capacitor
L_{f1a}	main filter inductor, HV side
R_{f1a}	resistor in series to L_{fla} ; R_{fla} models the inductor's con-
_	duction losses
$L_{\rm f1b}$	HV side filter inductor, damping circuit
$R_{ m f1b}$	HV side filter, damping resistor
$C_{\rm f1}$	HV side filter capacitor
$C_{\rm DC2}$	LV side DC capacitor
L_{f2a}	main filter inductor, LV side
$R_{\rm f2a}$	resistor in series to $L_{\rm f2a}$; $R_{\rm f2a}$ models the inductor's con-
	duction losses
$L_{\rm f2b}$	LV side filter inductor, damping circuit
$R_{\rm f2b}$	LV side filter, damping resistor
$C_{\rm f2}$	LV side filter capacitor
V_1	HV port voltage
V_2	LV port voltage
$V_{1,\min}$	minimal HV port voltage (within the specified range)
$V_{1,\max}$	maximal HV port voltage (within the specified range)
$V_{2,\min}$	minimal LV port voltage (within the specified range)
$V_{2,\max}$	maximal LV port voltage (within the specified range)
p_1	instantaneous HV port power (> 0 for power transfer from
	the HV port to the LV port)
p_2	instantaneous LV port power (> 0 for power transfer from
	the HV port to the LV port)
$v_{\rm AC1}$	instantaneous bridge voltage, HV side
$v_{\rm AC2}$	instantaneous bridge voltage, LV side
$v_{ m R}$	instantaneous voltage applied to the HF networks of the
	DAB and the LLC converters, $v_{\rm R} = v_{\rm AC1} - n v_{\rm AC2}$
\imath_L	DAB models without magnetizing inductance)

continued from previous page

Designator	Description
$\tilde{i}_L = -i_L$	negative instantaneous inductor current, HV side (used by the DAB models without magnetizing inductance), used in Section 4.3.3
i_{L_1}	instantaneous inductor current, HV side (used by the DAB model with magnetizing inductance)
i_{L_2}	instantaneous inductor current, LV side (used by the DAB model with magnetizing inductance)
v_{f1}	instantaneous voltage applied to $C_{\rm f1}$
v_{f2}	instantaneous voltage applied to $C_{\rm f2}$
$i_{ m f1}$	sum of the instantaneous currents through $L_{\rm f1a}$ and $L_{\rm f1b}$
$i_{ m f2}$	sum of the instantaneous currents through $L_{\rm f2a}$ and $L_{\rm f2b}$
P_1	average HV port power obtained from the electric DAB model (> 0 for power transfer from the HV port to the LV port)
P_2	average LV port power obtained from the electric DAB model (> 0 for power transfer from the HV port to the LV port)
$V_{\rm AC1}$	RMS bridge voltage, HV side
$V_{\rm AC2}$	RMS bridge voltage, LV side
I_L	RMS value of the HV side inductor and transformer cur- rent (magnetizing inductance neglected)
I_{L_1}	RMS values of the inductor and transformer currents on the HV side (magnetizing inductance considered)
I_{L_2}	RMS values of the inductor and transformer currents on the LV side (magnetizing inductance considered)
$V_{\rm f1}$	RMS value of the DAB filter capacitor voltage, HV side
$V_{\rm f2}$	RMS value of the DAB filter capacitor voltage, LV side
I_{f1}	RMS value of the DAB filter inductor current, HV side
$I_{\rm f2}$	RMS value of the DAB filter inductor current, LV side
P _{prec,1}	precise HV port power obtained from the accurate DAB loss model (> 0 for power transfer from the HV port to the LV port) precise LV port power obtained from the accurate DAB
Γprec,2	loss model (> 0 for power transfer from the HV port to the LV port)

continued from previous page

Designator	Description	
P_{t}	total losses $(P_{\rm t} = P_{\rm prec,1} - P_{\rm prec,2})$	
$P_{\rm cond}$	total conduction and copper losses	
$P_{\rm tr,cond}$	transformer copper losses	
$P_{L_{\rm HV},{\rm cond}}$	inductor copper losses	
$P_{1,\mathrm{cond}}$	total conduction losses of the HV side of the converter (including the HV transformer winding)	
$P_{2,\mathrm{cond}}$	total conduction losses of the LV side of the converter	
,	(including the LV transformer winding)	
$P_{\rm S1,cond}$	conduction losses, all HV switches	
$P_{\rm S2,cond}$	conduction losses, all LV switches	
$P_{\rm tr1,cond}$	conduction losses, HV transformer winding	
$P_{\rm tr2,cond}$	conduction losses, LV transformer winding	
$P_{\rm S1,sw}$	total switching losses, HV side	
$P_{\rm S2,sw}$	total switching losses, LV side	
$P_{\rm S1,gate}$	gate driver losses, HV side	
$P_{\rm S2,gate}$	gate driver losses, LV side	
$P_{\rm tr,core}$	transformer core losses	
$P_{1,\mathrm{aux}}$	power demand of the HV side auxiliary power supply	
$P_{2,\mathrm{aux}}$	power demand of the LV side auxiliary power supply	
$V_{\rm GS,1}$	gate driver voltage, HV side	
$V_{\rm GS,2}$	gate driver voltage, LV side	
$Q_{\mathrm{G},1}$	gate charge of a single MOSFET, HV side	
$Q_{\mathrm{G},2}$	gate charge of a single MOSFET, LV side	
$v_{\mathrm{DS,T}_x}$	drain-to-source voltage applied to the switch T_x	
R_{S1}	MOSFETs' on-state resistances, HV side	
R_{S2}	MOSFETs' on-state resistances, LV side	
$R_{L_{\rm HV}}$	copper resistance of the DAB inductor, HV side	
R_{tr1}	copper resistance of the HV side transformer winding	
$R_{\mathrm{tr}2}$	copper resistance of the LV side transformer winding	
$R_{\rm PCB,a}$	LV side PCB resistance, HF path	
$R_{\rm PCB,b}$	LV side PCB resistance between LV side full bridge and the (ceramic) DC capacitor	

continued from previous page

Designator	Description
$V_{\rm tr,core}$	transformer core volume
$V_{L_{\rm HV}, \rm core}$	inductor core volume
i	index counter
\overline{x}	average value of placeholder x
\hat{x}	small-signal value of placeholder x

continued from previous page

Bibliography

- J. R. Wilson and G. Burgh, *Energizing Our Future: Rational Choices* for the 21st Century. Hoboken, NJ: Wiley & Sons, 2008.
- [2] M. Ehsani, A. Gao, S. E. Gay, and A. Emadi, Modern Electric, Hybrid Electric, and Fuel Cell Vehicles: Fundamentals, Theory, and Design. Boca Raton: CRC Press, 2005.
- [3] U. Ν. F. C. on Climate Change (UNFCCC). (1998)Accessed: Kvoto protocol. Feb. 2011.[Online]. Available: http://unfccc.int/resource/docs/convkp/kpeng.pdf
- Α. [4]Β. California R. Α. Environmental Protec-(2009,June) California's emission tion Agency. zero ve-Feb. hicle program. Accessed: 2011.[Online]. Available: http://www.arb.ca.gov/msprog/zevprog/factsheets/zev_tutorial.pdf
- [5] F. Renken and R. Knorr, "High temperature electronics for future hybrid powertrain applications," 11th European Conference on Power Electronics and Applications (EPE 2005), Dresden, Germany, 11–14 Sept. 2005.
- [6] A. Emadi, S. S. Williamson, and A. Khaligh, "Power electronics intensive solutions for advanced electric, hybrid electric, and fuel cell vehicular power systems," *IEEE Transactions on Power Electronics*, vol. 21, no. 3, pp. 567–577, May 2006.
- [7] S. Basu, Recent Trends in Fuel Cell Science and Technology. New York: Springer, 2007.
- [8] A. Emadi, K. Rajashekara, S. S. Williamson, and S. M. Lukic, "Topological overview of hybrid electric and fuel cell vehicular power system architectures and configurations," *IEEE Transactions on Vehicular Technology*, vol. 54, no. 3, pp. 763–770, May 2005.

- [9] J. Laeuffer, "Automotive requirements for power electronic converters in traction advanced control systems," *Proc. of the 43rd International Power Electronics Conference (PCIM 2001)*, Nuremberg, Germany, 19– 21 June 2001, pp. 207–215.
- [10] T. Franke, H. Glonner, D. Nowak, and F. Österreicher, "Electrified power train – challenges and opportunities for the electrical industry," 11th European Conference on Power Electronics and Applications (EPE 2005), Dresden, Germany, 11–14 Sept. 2005.
- [11] L. Spiegel, D. Kraxner, and H.-J. Neußer, "Das Hybridkonzept von Porsche," 16. Aachener Kolloquium "Fahrzeug- und Motorentechnik", Aachen, Germany, 8–10 Oct. 2007, pp. 1–17.
- [12] Robert Bosch GmbH, Automotive Electrics Automotive Electronics. Hoboken, NJ: Wiley, 2007.
- [13] C. C. Chan, "The state of the art of electric, hybrid, and fuel cell vehicles," *Proceedings of the IEEE*, vol. 95, no. 4, pp. 704–718, April 2007.
- [14] G. Tamai, M. A. Jeffers, C. Lo, C. Thurston, S. Tarnowsky, and S. Poulos, "Development of the hybrid system for the saturn VUE hybrid," *SAE 2006 World Congress*, Detroit, MI, 3–6 April 2006.
- [15] F. Renken and J. Wolf, "Power electronics for hybrid-drive systems," 12th European Conference on Power Electronics and Applications (EPE 2007), Aalborg, Denmark, 2–5 Sept. 2007.
- [16] M. Duoba, H. Ng, and R. Larsen, "Characterization and comparison of two hybrid electric vehicles (HEVs) – Honda Insight and Toyota Prius," *SAE 2001 World Congress*, Detroit, MI, 5–8 March 2001.
- [17] A. R. Gale and M. W. Degner, "Voltage trade-off evaluation for electric and hybrid electric vehicle applications," *Proc. of the 7th IEEE Workshop on Power Electronics in Transportation*, Auburn Hills, MI, 24–25 Oct. 2002, pp. 11–15.
- [18] M. Ehsani, Y. Gao, and J. M. Miller, "Hybrid electric vehicles: Architecture and motor drives," *Proceedings of the IEEE*, vol. 95, no. 4, pp. 719–728, April 2007.
- [19] J. Erjavec and J. Arias, *Hybrid, electric, and fuel-cell vehicles*. Clifton Park, NY: Thomson Delmar Learning, 2007.
- [20] G. R. Flohr, "Performance considerations of a bi-directional DC/DC converter for fuel cell powered vehicles," SAE 2002 Future Car Congress, Arlington, VA, 3–5 June 2002.

- [21] J. M. Miller, "Multiple voltage electrical power distribution system for automotive applications," Proc. of the 31st Intersociety Energy Conversion Engineering Conference (IECEC 1996), vol. 3, Washington, DC, 11–16 Aug. 1996, pp. 1930–1937.
- [22] H.-P. Schöner and P. Hille, "Automotive power electronics new challenges for power electronics," Proc. of the 31st IEEE Annual Power Electronics Specialists Conference (PESC 2000), vol. 1, Galway, Ireland, 18–23 June 2000, pp. 6–11.
- [23] W. Wondrak, "General requirements on quality and reliability from the view of an OEM." Presentation at the ECPE-HOPE Symposium Automotive Power Electronics, Sindelfingen, Germany, 7–8 Oct. 2008.
- [24] A. Emadi, Y. J. Lee, and K. Rajashekara, "Power electronics and motor drives in electric, hybrid electric, and plug-in hybrid electric vehicles," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 6, pp. 2237– 2245, June 2008.
- [25] M. Gerber, J. A. Ferreira, N. Seliger, and I. W. Hofsajer, "Integral 3-D thermal, electrical and mechanical design of an automotive DC/DC converter," *IEEE Transactions on Power Electronics*, vol. 20, no. 3, pp. 566–575, May 2005.
- [26] A. F. Burke, "Batteries and ultracapacitors for electric, hybrid, and fuel cell vehicles," *Proceedings of the IEEE*, vol. 95, no. 4, pp. 806–820, April 2007.
- [27] A. Affanni, A. Bellini, G. Franceschini, P. Guglielmi, and C. Tassoni, "Battery choice and management for new-generation electric vehicles," *IEEE Transactions on Industrial Electronics*, vol. 52, no. 5, pp. 1343– 1349, Oct. 2005.
- [28] U. Gottwick, A. Schäfert, A. Wach, and I. Faye, "Future power trains and fuel cell," 16th International Conference on Automotive Technologies (ICAT 2006), Istanbul, Turkey, 17 Nov. 2006, pp. 1–10.
- [29] F. Profumo, A. Tenconi, M. Cerchio, R. Bojoi, and G. Gianolio, "Fuel cells for electric power generation: Peculiarities and dedicated solutions for power electronic conditioning system," 11th International Power Electronics and Motion Control Conference (EPE-PEMC 2004), Riga, Latvia, 2–4 Sept. 2004.
- [30] S.-J. Jang, T.-W. Lee, W.-C. Lee, and C.-Y. Won, "Bi-directional DC-DC converter for fuel cell generation system," *Proc. of the 35th IEEE Annual Power Electronics Specialists Conference (PESC 2004)*, vol. 6, Aachen, Germany, 20–25 June 2004, pp. 4722–4728.

- [31] A. C. Loyns, K. G. Ellis, and J. M. Hill, "High voltage lead-acid battery modules," Proc. of the 27th International Telecommunications Conference (INTELEC 2005), Berlin, Germany, 18–22 Sept. 2005, pp. 277–281.
- [32] J.-S. Lai and D. J. Nelson, "Energy management power converters in hybrid electric and fuel cell vehicles," *Proceedings of the IEEE*, vol. 95, no. 4, pp. 766–777, April 2007.
- [33] M. E. Glavin, P. K. W. Chan, S. Armstrong, and W. G. Hurley, "A stand-alone photovoltaic supercapacitor battery hybrid energy storage system," Proc. 13th Power Electronics and Motion Control Conference (EPE-PEMC 2008), Poznań, Poland, 1–3 Sept. 2008, pp. 1688–1695.
- [34] A. Nasiri, Z. Nie, S. B. Bekiarov, and A. Emadi, "An on-line UPS system with power factor correction and electric isolation using BIFRED converter," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 2, pp. 722–730, Feb. 2008.
- [35] R. L. Steigerwald, "A comparison of half-bridge resonant converter topologies," *IEEE Transactions on Power Electronics*, vol. 3, no. 2, pp. 174–182, April 1988.
- [36] J. W. Kolar, U. Drofenik, J. Biela, M. L. Heldwein, H. Ertl, T. Friedli, and S. D. Round, "PWM converter power density barriers," *Proc. of* the 4th Power Conversion Conference (PCC 2007), Nagoya, Japan, 2– 5 April 2007, pp. 9–29.
- [37] W.-J. Gu and R. Liu, "A study of volume and weight vs. frequency for high-frequency transformers," Conf. Rec. of the 24th IEEE Annual Power Electronics Specialists Conference (PESC 1993), Seattle, WA, 20-24 June 1993, pp. 1123-1129.
- [38] J. Biela, "Optimierung des elektromagnetisch integrierten Serien-Parallel Resonanzkonverters mit eingeprägtem Ausgangsstrom," Ph.D. dissertation, Swiss Federal Institute of Technology Zurich (ETH Zürich), 2005.
- [39] J. Biela, U. Badstübner, and J. W. Kolar, "Design of a 5kW, 1 U, 10 kW/ltr. resonant DC-DC converter for telecom applications," *Proc. of the 29th International Telecommunications Energy Conference* (INTELEC 2007), Rome, Italy, 30 Sept.-4 Oct. 2007, pp. 824–831.
- [40] H. G. Langer and H.-C. Skudelny, "DC to DC converters with bidirectional power flow and controllable voltage ratio," *Proc. of the 3rd European Conference on Power Electronics and Applications (EPE 1989)*, Aachen, Germany, 9–12 Oct. 1989, pp. 1245–1250.

- [41] G. Chen, Y.-S. Lee, S. Y. R. Hui, D. Xu, and Y. Wang, "Actively clamped bidirectional flyback converter," *IEEE Transactions on Industrial Electronics*, vol. 47, no. 4, pp. 770–779, Aug. 2000.
- [42] R. Mecke, D. Fischer, and C. Rathge, "Isolated Cuk converter for bidirectional operation in fuel cell applications," *Proc. of the 45th Power Electronics Conference (PCIM 2002)*, Nuremberg, Germany, 14– 16 May 2002, pp. 69–74.
- [43] R. L. Steigerwald, "High-frequency resonant transistor DC-DC converters," *IEEE Transactions on Industrial Electronics*, vol. IE-31, no. 2, pp. 181–191, May 1984.
- [44] J. Biela, U. Badstübner, and J. W. Kolar, "Impact of power density maximization on efficiency of DC–DC converter systems," *IEEE Transactions on Power Electronics*, vol. 24, no. 1, pp. 288–300, Jan. 2009.
- [45] M. H. Kheraluwala and R. W. De Doncker, "Single phase unity power factor control for dual active bridge converter," Conf. Rec. of the IEEE Industry Applications Society Annual Meeting (IAS 1993), Toronto, Canada, 2–8 Oct. 1993, pp. 909–916.
- [46] K. Wang, C. Y. Lin, L. Zhu, D. Qu, F. C. Lee, and J. S. Lai, "Bidirectional DC to DC converters for fuel cell systems," *Proc. of the IEEE Workshop on Power Electronics in Transportation*, Dearborn, MI, 22–23 Oct. 1998, pp. 47–51.
- [47] M. Cacciato, F. Caricchi, F. Giulii, and E. Santini, "A critical evaluation and design of bi-directional DC/DC converters for super-capacitors interfacing in fuel cell applications," *Conf. Rec. of the 39th IEEE Industry Applications Society Annual Meeting (IAS 2004)*, vol. 2, Seattle, WA, 3–7 Oct. 2004, pp. 1127–1133.
- [48] F. Zhang, L. Xiao, and Y. Yan, "Bi-directional forward-flyback DC-DC converter," *Proc. of the 35th IEEE Annual Power Electronics Specialists Conference (PESC 2004)*, vol. 5, Aachen, Germany, 20–25 June 2004, pp. 4058–4061.
- [49] R. W. De Doncker, D. M. Divan, and M. H. Kheraluwala, "A three-phase soft-switched high power density DC/DC converter for high power applications," Conf. Rec. of the IEEE Industry Applications Society Annual Meeting (IAS 1988), Pittsburgh, PA, 2–7 Oct. 1988, pp. 796–805.
- [50] M. H. Kheraluwala, R. W. Gascoigne, D. M. Divan, and E. D. Baumann, "Performance characterization of a high-power dual active bridge DC-to-DC converter," *IEEE Transactions on Industry Applications*, vol. 28, no. 6, pp. 1294–1301, Nov./Dec. 1992.

- [51] J. Walter and R. W. De Doncker, "High-power galvanically isolated DC/DC converter topology for future automobiles," *Proc. of the 34th IEEE Annual Power Electronics Specialist Conference (PESC 2003)*, vol. 1, Acapulco, Mexico, 15–19 June 2003, pp. 27–32.
- [52] S. Inoue and H. Akagi, "A bidirectional DC-DC converter for an energy storage system with galvanic isolation," *IEEE Transactions on Power Electronics*, vol. 22, no. 6, pp. 2299–2306, Nov. 2007.
- [53] S. Inoue and H. Akagi, "A bidirectional isolated DC/DC converter as a core circuit of the next-generation medium-voltage power conversion system," *IEEE Transactions on Power Electronics*, vol. 22, no. 2, pp. 535–542, March 2007.
- [54] R. W. A. A. De Doncker, D. M. Divan, and M. H. Kheraluwala, "A three-phase soft-switched high-power-density DC/DC converter for high-power applications," *IEEE Transactions on Industry Applications*, vol. 27, no. 1, pp. 63–73, Jan./Feb. 1991.
- [55] K. Wang, F. C. Lee, and J. Lai, "Operation principles of bi-directional full-bridge DC/DC converter with unified soft-switching scheme and soft-starting capability," *Proc. of the 15th IEEE Annual Applied Power Electronics Conference and Exposition (APEC 2000)*, vol. 1, New Orleans, LA, 6–10 Feb. 2000, pp. 111–118.
- [56] L. Zhu, K. Wang, F. C. Lee, and J.-S. Lai, "New start-up schemes for isolated full-bridge boost converters," *IEEE Transactions on Power Electronics*, vol. 18, no. 4, pp. 946–951, July 2003.
- [57] L. Zhu, "A novel soft-commutating isolated boost full-bridge ZVS-PWM DC-DC converter for bidirectional high power applications," *IEEE Transactions on Power Electronics*, vol. 21, no. 2, pp. 422–429, March 2006.
- [58] L. A. Flores, O. Garcia, J. A. Oliver, and J. A. Cobos, "High-frequency bi-directional DC/DC converter using two inductor rectifier," *Proc. of* the 10th IEEE International Power Electronics Congress (CIEP 2006), Puebla City, Mexico, 16–18 Oct. 2006, pp. 1–6.
- [59] H. Xiao, D. Chen, and S. Xie, "A ZVS bi-directional DC-DC converter for vehicular electronics," *Proc. of the IEEE Vehicle Power and Propul*sion Conference (VPPC 2005), Chicago, IL, 7–9 Sept. 2005, pp. 582– 588.
- [60] S.-S. Lee and G.-W. Moon, "Full ZVS-range transient current buildup half-bridge converter with different ZVS operations to load variation,"

IEEE Transactions on Industrial Electronics, vol. 55, no. 6, pp. 2557–2559, June 2008.

- [61] H. Xiao and S. Xie, "A ZVS bidirectional DC-DC converter with phaseshift plus PWM control scheme," *IEEE Transactions on Power Electronics*, vol. 23, no. 2, pp. 813–823, March 2008.
- [62] G. K. Schoneman, "500-W zero-voltage-switched full-bridge twoquadrant power modulator," Proc. of the 8th IEEE Annual Applied Power Electronics Conference and Exposition (APEC 1993), San Diego, CA, 7–11 March 1993, pp. 700–706.
- [63] G. Chen, Y.-S. Lee, D. Xu, and Y. Wang, "A novel soft-switching and low-conduction-loss bidirectional DC-DC converter," *Proc. of the* 3rd International Power Electronics and Motion Control Conference (IPEMC 2000), vol. 3, Beijing, China, 15–18 Aug. 2000, pp. 1166–1171.
- [64] A. D. Swingler and W. G. Dunford, "Development of a bi-directional DC/DC converter for inverter/charger applications with consideration paid to large signal operation and quasi-linear digital control," Proc. of the 33rd IEEE Annual Power Electronics Specialists Conference (PESC 2002), vol. 2, Cairns, Australia, 23–27 June 2002, pp. 961–966.
- [65] M. Jain, M. Daniele, and P. K. Jain, "A bidirectional DC-DC converter topology for low power application," *IEEE Transactions on Power Electronics*, vol. 15, no. 4, pp. 595–606, July 2000.
- [66] G.-J. Su, F. Z. Peng, and D. J. Adams, "Experimental evaluation of a soft-switching DC/DC converter for fuel cell vehicle applications," *Proc. of the 7th IEEE Workshop on Power Electronics in Transportation*, Auburn Hills, MI, 24–25 Oct. 2002, pp. 39–44.
- [67] H. Li, F. Z. Peng, and J. S. Lawler, "A natural ZVS medium-power bidirectional DC-DC converter with minimum number of devices," *IEEE Transactions on Industry Applications*, vol. 39, no. 2, pp. 525–535, March/April 2003.
- [68] F. Z. Peng, H. Li, G.-J. Su, and J. S. Lawler, "A new ZVS bidirectional DC-DC converter for fuel cell and battery application," *IEEE Transactions on Power Electronics*, vol. 19, no. 1, pp. 54–65, Jan. 2004.
- [69] H.-J. Chiu and L.-W. Lin, "A bidirectional DC-DC converter for fuel cell electric vehicle driving system," *IEEE Transactions on Power Electronics*, vol. 21, no. 4, pp. 950–958, July 2006.

- [70] B.-R. Lin, J.-J. Chen, Y.-E. Lee, and H.-K. Chiang, "Analysis and implementation of a bidirectional ZVS DC-DC converter with active clamp," *Proc. of the 3rd IEEE Conference on Industrial Electronics and Applications (ICIEA 2008)*, Singapore, Singapore, 3–5 June 2008, pp. 382–387.
- [71] J. B. Klaassens, "Steady-state analysis of a series-resonant DC-DC converter with a bipolar power flow," *IEEE Transactions on Industrial Electronics*, vol. 36, no. 1, pp. 48–55, Feb. 1989.
- [72] I. W. Hofsajer, J. A. Ferreira, and J. D. van Wyk, "A new manufacturing and packaging technology for the integration of power electronics," *Conf. Rec. of the IEEE Industry Applications Society Annual Meeting (IAS* 1995), vol. 1, Lake Buena Vista, FL, 8–12 Oct. 1995, pp. 891–897.
- [73] Y. Cheron, Soft commutation. London: Chapman & Hall, 1992.
- [74] R. Lenke, F. Mura, and R. W. De Doncker, "Comparison of non-resonant and super-resonant dual-active ZVS-operated high-power DC-DC converters," Proc. of the 13th European Conference on Power Electronics and Applications (EPE 2009), Barcelona, Spain, 8–10 Sept. 2009.
- [75] A. Isurin and A. Cook, "Achieving a wide input voltage and output power load range step-down DC-DC conversion with good full range efficiency at 4 kW," Proc. of the 21st IEEE Annual Applied Power Electronics Conference and Exposition (APEC 2006), Dallas, TX, 19– 23 March 2006, pp. 67–71.
- [76] A. Isurin and A. Cook, "A novel resonant converter topology and its application," *Proc. of the 32nd IEEE Annual Power Electronics Specialists Conference (PESC 2001)*, vol. 2, Orlando, FL, 17–21 June 2001, pp. 1039–1044.
- [77] M. Pavlovsky, S. W. H. de Haan, and J. A. Ferreira, "Concept of 50 kW DC/DC converter based on ZVS, quasi-ZCS topology and integrated thermal and electromagnetic design," *Proc. of the 11th European Conference on Power Electronics and Applications (EPE 2005)*, Dresden, Germany, 11–14 Sept. 2005, pp. 1–9.
- [78] O. García, L. A. Flores, J. A. Oliver, J. A. Cobos, and J. de la Peña, "Bi-directional DC/DC converter for hybrid vehicles," *Proc. of the 36th IEEE Power Electronics Specialists Conference (PESC 2005)*, Recife, Brazil, 12–16 June 2005, pp. 1881–1886.
- [79] M. Nowak, J. Hildebrandt, and P. Luniewski, "Converters with AC transformer intermediate link suitable as interfaces for supercapacitor energy storage," Proc. of the 35th IEEE Annual Power Electron-

ics Specialists Conference (PESC 2004), vol. 5, Aachen, Germany, 20–25 June 2004, pp. 4067–4073.

- [80] P. J. Wolfs, "A current-sourced DC-DC converter derived via the duality principle from the half-bridge converter," *IEEE Transactions on Industrial Electronics*, vol. 40, no. 1, pp. 139–144, Feb. 1993.
- [81] A. Levy, "Comparing the push-pull, forward converter and full bridge topologies in low voltage, high power DC-DC converter," *Proc. of the High Frequency Power Conversion Conference (HFPC 2001)*, Rosemont, IL, 9–13 Sept. 2001, pp. 470–476.
- [82] Y. Jang and M. M. Jovanovic, "A new family of full-bridge ZVS converters," *IEEE Transactions on Power Electronics*, vol. 19, no. 3, pp. 701– 708, May 2004.
- [83] Y. Jang and M. M. Jovanovic, "A new PWM ZVS full-bridge converter," *IEEE Transactions on Power Electronics*, vol. 22, no. 3, pp. 987–994, May 2007.
- [84] R. P. Severns, "Topologies for three-element resonant converters," *IEEE Transactions on Power Electronics*, vol. 7, no. 1, pp. 89–98, Jan. 1992.
- [85] R. L. Steigerwald, R. W. De Doncker, and M. H. Kheraluwala, "A comparison of high-power DC-DC soft-switched converter topologies," *IEEE Transactions on Industry Applications*, vol. 32, no. 5, pp. 1139–1145, Sept./Oct. 1996.
- [86] J. F. Lazar and R. Martinelli, "Steady-state analysis of the LLC series resonant converter," Proc. of the 16th IEEE Annual Applied Power Electronics Conference and Exposition (APEC 2001), vol. 2, Anaheim, CA, 4–8 March 2001, pp. 728–735.
- [87] F. Caricchi, F. Crescimbini, F. G. Capponi, and L. Solero, "Study of bidirectional buck-boost converter topologies for application in electrical vehicle motor drives," *Proc. of the 13th IEEE Annual Applied Power Electronics Conference and Exposition (APEC 1998)*, vol. 1, Anaheim, CA, 15–19 Feb. 1998, pp. 287–293.
- [88] S. Waffler and J. W. Kolar, "A novel low-loss modulation strategy for high-power bi-directional buck+boost converters," *Proc. of the 7th International Conference on Power Electronics (ICPE 2007)*, Daegu, South Korea, 22–26 Oct. 2007, pp. 889–894.
- [89] S. W. Anderson, R. W. Erickson, and R. A. Martin, "An improved automotive power distribution system using nonlinear resonant switch converters," *IEEE Transactions on Power Electronics*, vol. 6, no. 1, pp. 48– 54, Jan. 1991.

- [90] K. Vangen, T. Melaa, and A. K. Ådnanes, "Soft-switched high-frequency, high power DC/AC converter with IGBT," Conf. Rec. of the 23rd IEEE Annual Power Electronics Specialists Conference (PESC 1992), Toledo, Spain, 29 June–3 July 1992, pp. 26–33.
- [91] H. L. Chan, K. W. E. Cheng, and D. Sutanto, "An extended load range ZCS-ZVS bi-directional phase-shifted DC-DC converter," *Proc. of the* 8th IEE International Conference on Power Electronics and Variable Speed Drives, London, UK, 18–19 Sept. 2000, pp. 74–79.
- [92] J. W. Kolar, Skriptum zur Vorlesung Leistungselektronische Systeme II. Swiss Federal Institute of Technology Zurich (ETH Zürich), Frühjahrssemester 2010.
- [93] F. Krismer, S. Round, and J. W. Kolar, "Performance optimization of a high current dual active bridge with a wide operating voltage range," *Proc. of the 37th IEEE Annual Power Electronics Specialists Conference* (*PESC 2006*), Jeju, South Korea, 18–22 June 2006, pp. 1–7.
- [94] K. Vangen, T. Melaa, S. Bergsmark, and R. Nilsen, "Efficient highfrequency soft-switched power converter with signal processor control," *Proc. of the 13th IEEE International Telecommunications Energy Conference (INTELEC 1991)*, Kyoto, Japan, 5–8 Nov. 1991, pp. 631–639.
- [95] G. G. Oggier, G. O. García, and A. R. Oliva, "Switching control strategy to minimize dual active bridge converter losses," *IEEE Transactions on Power Electronics*, vol. 24, no. 7, pp. 1826–1838, July 2009.
- [96] H. Tao, A. Kotsopoulos, J. L. Duarte, and M. A. M. Hendrix, "Transformer-coupled multiport ZVS bidirectional DC-DC converter with wide input range," *IEEE Transactions on Power Electronics*, vol. 23, no. 2, pp. 771–781, March 2008.
- [97] N. Schibli, "Symmetrical multilevel converters with two quadrant DC-DC feeding," Ph.D. dissertation, Swiss Federal Institute of Technology Lausanne (EPFL), 2000.
- [98] A.-C. Rufer, N. Schibli, and C. Briguet, "A direct coupled 4-quadrant multilevel converter for 16 2/3 Hz traction systems," Proc. of the 6th IEE Conference on Power Electronics and Variable Speed Drives, Nottingham, UK, 23–25 Sept. 1996, pp. 448–453.
- [99] H. Bai and C. Mi, "Eliminate reactive power and increase system efficiency of isolated bidirectional dual-active-bridge DC-DC converters using novel dual-phase-shift control," *IEEE Transactions on Power Electronics*, vol. 23, no. 6, pp. 2905–2914, Nov. 2008.

- [100] A. K. Jain and R. Ayyanar, "PWM control of dual active bridge: comprehensive analysis and experimental verification," Proc. of the 34th IEEE Annual Industrial Electronics Conference (IECON 2008), Orlando, FL, 10–13 Nov. 2008, pp. 909–915.
- [101] H. Zhou and A. M. Khambadkone, "Hybrid modulation for dual-active bridge bidirectional converter with extended power range for ultracapacitor application," *IEEE Transactions on Industrial Electronics*, vol. 45, no. 4, pp. 1434–1442, July/Aug. 2009.
- [102] Y. Wang, S. W. H. de Haan, and J. A. Ferreira, "Optimal operating ranges of three modulation methods in dual active bridge converters," *Proc. of the 6th IEEE International Power Electronics and Motion Control Conference (IPEMC 2009)*, Wuhan, China, 17–20 May 2009, pp. 1397–1401.
- [103] Y. Xie, J. Sun, and J. S. Freudenberg, "Power flow characterization of a bidirectional galvanically isolated high-power DC/DC converter over a wide operating range," *IEEE Transactions on Power Electronics*, vol. 1, no. 1, pp. 54–66, Jan. 2010.
- [104] Y. Wang, S. de Haan, and J. A. Ferreira, "Methods for experimental assessment of component losses to validate the converter loss model," Proc. of the 13th IEEE International Power Electronics and Motion Control Conference (EPE-PEMC 2008), Poznań, Poland, 1–3 Sept. 2008, pp. 187–194.
- [105] C. Buccella, C. Cecati, and F. de Monte, "A coupled electrothermal model for planar transformer temperature distribution computation," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 10, pp. 3583– 3590, Oct. 2008.
- [106] J. L. Russi, M. L. S. Martins, and H. L. Hey, "Coupled-filter-inductor soft-switching techniques: principles and topologies," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 9, pp. 3361–3373, Sept. 2008.
- [107] M. Reisch, Halbleiter-Bauelemente. Berlin: Springer, 2005.
- [108] J. Reinert, A. Brockmeyer, and R. W. De Doncker, "Calculation of losses in ferro- and ferrimagnetic materials based on the modified Steinmetz equation," *IEEE Transactions on Industry Applications*, vol. 37, no. 4, pp. 1055–1061, July/Aug. 2001.
- [109] Y. Han and Y.-F. Liu, "A practical transformer core loss measurement scheme for high-frequency power converter," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 2, pp. 941–948, June 2008.

- [110] O. D. Patterson and D. M. Divan, "Pseudo-resonant full bridge DC/DC converter," *IEEE Transactions on Power Electronics*, vol. 6, no. 4, pp. 671–678, Oct. 1991.
- [111] J. A. Ferreira, A. van Ross, and J. D. van Wyk, "A generic soft switching converter topology with a parallel nonlinear network for high-power application," *IEEE Transactions on Power Electronics*, vol. 7, no. 2, pp. 324–331, April 1992.
- [112] R. W. De Doncker and J. P. Lyons, "The auxiliary resonant commutated pole converter," Conf. Rec. of the 25th IEEE Industry Applications Society Annual Meeting (IAS 1990), Seattle, WA, 7–12 Oct. 1990, pp. 1228–1235.
- [113] W. McMurray, "Resonant snubbers with auxiliary switches," *IEEE Transactions on Industry Applications*, vol. 29, no. 2, pp. 355–362, March/April 1993.
- [114] G. W. Wester and R. D. Middlebrook, "Low-frequency characterization of switched DC–DC converters," *IEEE Transactions on Aerospace and Electronic Systems*, vol. AES-9, no. 3, pp. 376–385, May 1973.
- [115] R. D. Middlebrook and S. Ćuk, "A general unified approach to modelling switching-converter power stages," *Proc. of the IEEE Power Electronics Specialists Conference (PESC 1976)*, Cleveland, OH, 8–10 June 1976, pp. 18–34.
- [116] R. W. Erickson and D. Maksimović, Fundamentals of power electronics, 2nd ed. New York, NY: Springer Science+Business Media, 2001.
- [117] D. J. Packard, "Discrete modeling and analysis of switching regulators," Ph.D. dissertation, California Institute of Technology, 1976.
- [118] A. R. Brown and R. D. Middlebrook, "Sampled-data modeling of switching regulators," Proc. of the IEEE Power Electronics Specialists Conference (PESC 1981), Boulder, CO, 29 June–3 July 1981, pp. 349–369.
- [119] G. C. Verghese, M. E. Elbuluk, and J. G. Kassakian, "A general approach to sampled-data modeling for power electronic circuits," *IEEE Transactions on Power Electronics*, vol. PE-1, no. 2, pp. 76–89, April 1986.
- [120] I. Batarseh and K. Siri, "Generalized approach to the small signal modelling of DC-to-DC resonant converters," *IEEE Transactions on Aerospace and Electronic Systems*, vol. 29, no. 3, pp. 894–909, July 1993.

- [121] D. Maksimovic and R. Zane, "Small-signal discrete-time modeling of digitally controlled PWM converters," *IEEE Transactions on Power Electronics*, vol. 22, no. 6, pp. 2552–2556, Nov. 2007.
- [122] H. Bai, C. C. Mi, and S. Gargies, "The short-time-scale transient processes in high-voltage and high-power isolated bidirectional DC–DC converters," *IEEE Transactions on Power Electronics*, vol. 23, no. 6, pp. 2648–2656, Nov. 2008.
- [123] G. F. Franklin, J. D. Powell, and M. L. Workman, *Digital control of dynamic systems*, 3rd ed. Half Moon Bay, CA : Ellis-Kagle Press, 1998.
- [124] J. W. Umland and M. Safiuddin, "Magnitude and symmetric optimum criterion for the design of linear control systems: what is it and how does it compare with the others?" *IEEE Transactions on Industry Applications*, vol. 26, no. 3, pp. 489–497, May/June 1990.
- [125] S. Bontemps, A. Calmels, S. D. Round, and J. W. Kolar, "Low profile power module combined with state of the art MOSFET switches and SiC diodes allows high frequency and very compact three-phase sinusoidal input rectifiers," *Proc. of the International Power Electronics Conference (PCIM 2007)*, Nuremberg, Germany, 22–24 May 2007.
- [126] A. Prechtl, Vorlesungen über die Grundlagen der Elektrotechnik, Band 1. Wien: Springer, 1994.
- [127] Y.-C. Chen, "Ceramic disc nonlinear capacitor composed of ZrO₂-doped barium strontium titanate for application in electronic switching circuits," *Japanese Journal of Applied Physics*, vol. 45, pp. 5869–5871, July 2006.
- [128] G. Niedermeier and M. Esguerra, "Measurement of power losses with DC-bias – the displacement factor," Proc. of the 41st International Power Electronics Conference (PCIM 2000), Nuremberg, Germany, 6– 8 June 2000.
- [129] X. Li and A. K. S. Bhat, "Analysis and design of high-frequency isolated dual-bridge series resonant DC/DC converter," *IEEE Transactions on Power Electronics*, vol. 25, no. 4, pp. 850–862, April 2010.
- [130] F. Krismer, J. Biela, and J. W. Kolar, "A comparative evaluation of isolated bi-directional DC/DC converters with wide input and output voltage range," Conf. Rec. of the 40th IEEE Annual Industry Applications Conference (IAS 2005), vol. 1, Hong Kong, China, 2–6 Oct. 2005, pp. 599–606.

- [131] Y. Xie, R. Ghaemi, J. Sun, and J. S. Freudenberg, "Implicit model predictive control of a full bridge DC–DC converter," *IEEE Transactions* on Power Electronics, vol. 24, no. 12, pp. 2704–2713, Dec. 2009.
- [132] B. Fässler and D. Hösli, "Aufbau eines hochkompakten Phase-Shift DC-DC-Konverters," Master's thesis, Power Electronic Systems Laboratory (PES), Swiss Federal Institute of Technology Zurich (ETH Zürich), 2008.
- [133] C. Wang, Y. Kang, B. Lu, J. Sun, M. Xu, W. Dong, F. C. Lee, and W. C. Tipton, "A high power-density, high efficiency front-end converter for capacitor charging applications," *Proc. of the 20th IEEE Annual Applied Power Electronics Conference and Exposition (APEC 2005)*, vol. 2, Austin, TX, 6–10 March 2005, pp. 1258–1264.
- [134] M. Nymand and M. A. E. Andersen, "High-efficiency isolated boost DC–DC converter for high-power low-voltage fuel-cell applications," *IEEE Transactions on Industrial Electronics*, vol. 57, no. 2, pp. 505– 514, Feb. 2010.
- [135] L. Zhu, X. Xu, and F. Flett, "A 3 kW isolated bi-directional DC/DC converter for fuel cell electric vehicle application," Proc. of the 43th International Power Electronics Conference (PCIM 2001), Nuremberg, Germany, 19–21 June 2001, pp. 77–82.
- [136] S. Waffler, J. Biela, and J. W. Kolar, "Output ripple reduction of an automotive multi-phase bi-directional DC-DC converter," *Proc. of the* 1st IEEE Energy Conversion Congress and Exposition (ECCE 2009), San Jose, CA, 20–24 Sept. 2009, pp. 2184–2190.
- [137] S. Waffler and J. W. Kolar, "Efficiency optimization of an automotive multi-phase bi-directional DC-DC converter," Proc. of the 6th IEEE International Power Electronics and Motion Control Conference (IPEMC 2009), Wuhan, China, 15–17 Mai 2009, pp. 566–572.
- [138] C. Qiao, J. Zhang, P. Parto, and D. Jauregui, "Output capacitor comparison for low voltage high current applications," *Proc. of the 35th IEEE Power Electronics Specialists Conference (PESC 2004)*, vol. 1, Aachen, Germany, 20–25 June 2004, pp. 622–628.
- [139] C. R. Sullivan, "Optimal choice for number of strands in a litz-wire transformer winding," *IEEE Transactions on Power Electronics*, vol. 14, no. 2, pp. 283–291, March 1999.
- [140] Y. Wang, W. Roodenburg, and S. W. H. de Haan, "Comparative study of three transformer concepts for high current dual active bridge converters," Proc. of the 5th International Conference on Integration of

Power Electronics Systems (CIPS 2008), Nuremberg, Germany, 11–13 March 2008.

- [141] M. H. Kheraluwala, D. W. Novotny, and D. M. Divan, "Coaxially wound transformers for high-power high-frequency applications," *IEEE Transactions on Power Electronics*, vol. 7, no. 1, pp. 54–62, Jan. 1992.
- [142] G. Skutt, F. Lee, R. Ridley, and D. Nicol, "Leakage inductance and termination effects in a high-power planar magnetic structure," *Proc. of the* 9th IEEE Annual Applied Power Electronics Conference and Exposition (APEC 1994), vol. 1, Orlando, FL, 13–17 Feb. 1994, pp. 295–301.
- [143] R. Redl, N. O. Sokal, and L. Balogh, "A novel soft-switching full-bridge DC/DC converter: analysis, design considerations, and experimental results at 1.5 kW, 100 kHz," *IEEE Transactions on Power Electronics*, vol. 6, no. 3, pp. 408–418, July 1991.
- [144] B. Wrzecionko, "Entwurf und Realisierung eines bidirektionalen DC/DC-Konverters für denEinsatz in Hvbridund Brennstoffzellenfahrzeugen," Master's thesis, Power Electronic Systems Laboratory (PES), Swiss Federal Institute of Technology Zurich (ETH Zürich), 2008.
- [145] A. van den Bossche and V. C. Valchev, Inductors and transformers for power electronics. New York, NY: Taylor & Francis, 2005.
- [146] W. Benz, P. Heinks, and L. Starke, *Tabellenbuch Elektronik*, 6th ed. Frankfurt (Main): Frankfurter Fachverlag; Dossenheim: Kohl und Noltemayer, 1989.
- [147] H. Fricke and P. Vaske, *Elektrische Netzwerke*, 17th ed. Stuttgart: Teubner, 1982.
- [148] J. A. Ferreira, "Improved analytical modeling of conductive losses in magnetic components," *IEEE Transactions on Power Electronics*, vol. 9, no. 1, pp. 127–131, Jan. 1994.
- [149] W. Roshen, "Fringing field formulas and winding loss due to an air gap," *IEEE Transactions on Magnetics*, vol. 43, no. 8, pp. 3387–3394, Aug. 2007.
- [150] J. Lutz, Halbleiter-Leistungsbauelemente: Physik, Eigenschaften, Zuverlässigkeit (German Edition), 1st ed. Springer, Jun. 2006.
- [151] P. N. Robinson and G. S. Robinson, "A computer method for obtaining z transforms," *IEEE Transactions on Audio and Electroacoustics*, vol. 20, no. 1, pp. 98–99, March 1972.

- [152] Agilent Technologies, Agilent 4294A Precision Impedance Analyzer Operation Manual, 7th ed., Feb. 2003.
- [153] Infineon Technologies AG. Datasheet information on power semiconductors. [Online]. Available: http://www.infineon.com
- [154] EPCOS AG. (2006, 17 July) AC characteristics of HighCap ceramic capacitors. Accessed: May, 2010. [Online]. Available: http://www.epcos.com
- [155] EPCOS AG. Datasheet information on planar E cores. [Online]. Available: http://www.epcos.com
- [156] Ferroxcube. Datasheet information on planar E cores. [Online]. Available: http://www.ferroxcube.com/prod/assets/planar.htm
- [157] Vishay. (2009, 11 Dec.) Datasheet information on the metallized polypropylene capacitors KP/MKP 375. Accessed: Feb., 2011. [Online]. Available: http://www.vishay.com/docs/28127/mkp375
- [158] IXYS Corporation. Datasheet information on power semiconductors. [Online]. Available: http://www.ixys.com
- [159] Robert Pack GmbH & Co. KG. Technical data of the RUPALIT[®] high frequency litz wire. Accessed: Feb. 2011. [Online]. Available: http://www.pack-feindraehte.de/de/technische daten
- [160] EPCOS AG. Datasheet information on ferrite materials. [Online]. Available: http://www.epcos.com
- [161] Ferroxcube. (2008, 1 Sept.) 3F3 material specification. [Online]. Available: http://www.ferroxcube.com/prod/assets/3f3.pdf
- [162] C. Blake and C. Bull. IGBT or MOSFET: choose wisely. International Rectifier. Accessed: Feb., 2011. [Online]. Available: http://www.irf.com/technical-info/whitepaper/choosewisely.pdf
- [163] STMicroelectronics. Datasheet information on power semiconductors. [Online]. Available: http://www.st.com
- [164] International Rectifier. Datasheet information on power semiconductors. [Online]. Available: http://www.irf.com

Curriculum Vitae

Personal Details

Name	Florian Krismer		
Date of Birth	11th April 1978		
Place of Birth	Feldkirch, Austria		
Citizen of	Meiningen, Austria		
Nationality	Austria		
Civil State	Married with Olivia Daub Krismer		
Education			
1984 - 1988	Elementary school, Meiningen		
1988 - 1992	Secondary school, Feldkirch-Levis		
1992 - 1997	Higher Institute of Technical Education		
	(HTL), Rankweil		

University 1998 - 2004 Electrical engineering at the Vienna University of Technology, field of study: automation technology 26. 1. 2001 Intermediate examination, passed with honor 17. 6. 2004 Diploma examination, passed with honor Doctorate Intermediate examination, passed with honor

Doctorate	
2004 - 2010	Doctorate at the Power Electronic Systems
	Laboratory (PES), ETH Zürich